Edited by Helga Rübsamen-Schaeff and Helmut Buschmann

New Drug Development for Known and Emerging Viruses



Volume 80

Series Editors: R. Mannhold, H. Buschmann, J. Holenz





New Drug Development for Known and Emerging Viruses

Methods and Principles in Medicinal Chemistry

Edited by R. Mannhold, H. Buschmann, J. Holenz

Editorial Board G. Folkers, H. Kubinyi, H. Timmermann, H. van de Waterbeemd, J. Bondo Hansen Previous Volumes of the Series

Gruss, M., (ed.)

Solid State Development and Processing of Pharmaceutical Molecules Salts, Cocrystals, and Polymorphism

2021 ISBN: 978-3-527-34635-6 Vol. 79

Plowright, A.T. (Ed.)

Target Discovery and Validation Methods and Strategies for Drug Discovery

2020 ISBN: 978-3-527-34529-8 Vol. 78

Swinney, D., Pollastri, M. (Eds.)

Neglected Tropical Diseases Drug Discovery and Development

2019 ISBN: 978-3-527-34304-1 Vol. 77

Bachhav, Y. (Ed.)

Innovative Dosage Forms Design and Development at Early Stage

2019 ISBN: 978-3-527-34396-6 Vol. 76

Gervasio, F. L., Spiwok, V. (Eds.)

Biomolecular Simulations in Structure-based Drug Discovery

2018 ISBN: 978-3-527-34265-5 Vol. 75 Sippl, W., Jung, M. (Eds.) Epigenetic Drug Discovery

2018 ISBN: 978-3-527-34314-0 Vol. 74

Giordanetto, F. (Ed.)

Early Drug Development

2018 ISBN: 978-3-527-34149-8 Vol. 73

Handler, N., Buschmann, H. (Eds.)

Drug Selectivity

2017 ISBN: 978-3-527-33538-1 Vol. 72

Vaughan, T., Osbourn, J., Jalla, B. (Eds.)

Protein Therapeutics

2017 ISBN: 978-3-527-34086-6 Vol. 71

Ecker, G. F., Clausen, R. P., and Sitte, H. H. (Eds.)

Transporters as Drug Targets

2017 ISBN: 978-3-527-33384-4 Vol. 70 New Drug Development for Known and Emerging Viruses

Edited by Helga Rübsamen-Schaeff and Helmut Buschmann



Volume Editors

Prof. Dr. Helga Rübsamen-Schaeff Altestadt 2 40213 Düsseldorf Germany

Dr. Helmut Buschmann Sperberweg 15 52076 Aachen Germany

Series Editors

Prof. Dr. Raimund Mannhold Rosenweg 7 40489 Düsseldorf Germany

Dr. Helmut Buschmann

Sperberweg 15 52076 Aachen Germany

Dr. Jörg Holenz

GRT Therapeutics Grunenthal Boston Pain Innovation Hub 1 Broadway, Cambridge MA 02142 United States

Cover Image © Erhan Ergin / Fotolia.com for the symbols used in the margin.

All books published by **WILEY-VCH** are carefully produced. Nevertheless, authors, editors, and publisher do not warrant the information contained in these books, including this book, to be free of errors. Readers are advised to keep in mind that statements, data, illustrations, procedural details or other items may inadvertently be inaccurate.

Library of Congress Card No.: applied for

British Library Cataloguing-in-Publication Data A catalogue record for this book is available from the British Library.

Bibliographic information published by the Deutsche Nationalbibliothek

The Deutsche Nationalbibliothek lists this publication in the Deutsche Nationalbibliografie; detailed bibliographic data are available on the Internet at <http://dnb.d-nb.de>.

© 2022 WILEY-VCH GmbH, Boschstr. 12, 69469 Weinheim, Germany

All rights reserved (including those of translation into other languages). No part of this book may be reproduced in any form – by photoprinting, microfilm, or any other means – nor transmitted or translated into a machine language without written permission from the publishers. Registered names, trademarks, etc. used in this book, even when not specifically marked as such, are not to be considered unprotected by law.

 Print ISBN:
 978-3-527-34337-9

 ePDF ISBN:
 978-3-527-81071-0

 ePub ISBN:
 978-3-527-81070-3

 oBook ISBN:
 978-3-527-81069-7

Cover Design SCHULZ Grafik-Design, Fußgönheim, Germany Typesetting Straive, Chennai, India

Printed on acid-free paper.

10 9 8 7 6 5 4 3 2 1

Contents

Preface xv In Memoriam: Farah Elawar xvii Introduction xix References xx

1 HIV-Disease Overview, Targets for Therapy and Open Issues 1

v

Christoph Stephan

- 1.1 HIV—Disease Overview 1
- 1.2 Targets for Antiretroviral Therapy 2
- 1.3 Currently Open Issues in HIV/AIDS Research *10* References *10*

2 Curing Hepatitis C with Direct-Acting Antiviral Therapy 13

Michael J. Sofia

- 2.1 Introduction 13
- 2.2 Tools to Enable Drug Discovery 14
- 2.3 Drug Discovery Targets 15
- 2.4 NS5B Polymerase Inhibitors 15
- 2.4.1 Nucleoside and Nucleotide Inhibitors of NS5B Polymerase 15
- 2.4.2 Non-nucleoside NS5B Inhibitors 19
- 2.5 HCV NS3/4A Protease Inhibitors 22
- 2.5.1 HCV NS3/4A Protease Acyclic Reversible Inhibitors 25
- 2.5.2 HCV NS3/4A Protease Acyclic Covalent Binding Inhibitors 25
- 2.5.3 HCV NS3/4A Protease Macrocyclic Reversible Inhibitors 30
- 2.5.4 HCV NS3/4 Protease P2–P4 Macrocyclic Inhibitors 30
- 2.6 HCV NS5A Inhibitors 32
- 2.7 The Evolution of DAA Combination Therapies 35
- 2.8 Conclusion 41
 - Acknowledgment 41
 - References 41

vi Contents

3	Antiviral Drugs Against Influenza Virus 59 Oliver Planz			
3.1	The Influenza Virus 59			
3.2	The Pathogenesis of Influenza 60			
3.3	Influenza Drugs and Targets 62			
3.4	Adamantanes and Derivatives 63			
3.5	Neuraminidase Inhibitors 64			
3.5.1	Oseltamivir 65			
3.5.2	Zanamivir 66			
3.5.3	Peramivir 67			
3.5.4	Laninamivir 69			
3.5.5	Summary 70			
3.6	Polymerase-Inhibitors 70			
3.6.1	Baloxavir Marboxil (BAM) 70			
3.6.2	Favipiravir 73			
3.6.3	Pimodivir 75			
3.6.4	Summary 76			
3.7	Monoclonal Antibodies 76			
3.7.1	MHAA4549A 77			
3.7.2	MEDI8852 77			
3.7.3	VIS410 78			
3.7.4	Summary 78			
3.8	Host-Targeting Candidates 79			
3.8.1	DAS181 79			
3.8.2	Nitazoxanide 80			
3.8.3	LASAG 81			
3.8.4	Summary 82			
	References 82			
4	Respiratory Syncytial Virus Immunoreactivity, Vaccine Development, and Therapeutics 97 Farab Elawar Matthias Gotte, Nelson Lee, Rachel Fearns, and David I. Marchant			
4.1	Introduction 97			
4.1.1	The Burden of RSV Infection on Human Health 97			
4.1.2	RSV Transmission 98			
4.1.3	The Current Therapeutic Options and Opportunity for the Development			
111	The Unique Challenge of PSV Vaccine Development 00			
415	The Virus and the Replication Cycle 00			
416	The Nucleonrotein_RNA Complex 100			
0 117	The L Polymerase Protein 101			
/ 418	The P Phosphonrotein 101			
410	The M2-1 Protein 102			
т.1.9 				

- 4.1.10 The RSV-G Glycoprotein 102
- 4.1.11 The RSV-F Fusion Glycoprotein 102

Contents vii

- 4.1.12 The RSV-SH Glycoprotein 104
- 4.1.13 Antigenic Variation of RSV Surface Glycoproteins 108
- 4.1.14 Adaptation of RSV in the Community by Emergent High Titer Clades 108
- 4.2 RSV Longevity, Immune Evasion, and the Role of IgA 108
- 4.2.1 Evasion or Suppression of Immune Memory 108
- 4.3 The Impact of Immunoprophylaxis on the Health Burden of Respiratory Syncytial Virus *110*
- 4.4 Distinct RSV Symptoms 110
- 4.4.1 Wheeze 110
- 4.4.2 The Effect of Immunoprophylaxis on the Development of Asthma 111
- 4.4.3 Epidemiology and Clinical Aspects of Adult RSV Disease 111
- 4.5 History of RSV and Vaccine Development 112
- 4.5.1 The Tragic History of RSV Vaccine Development in the First Failed RSV Vaccine Trials of the 1960s *112*
- 4.6 New Developments in RSV Vaccine Development 113
- 4.6.1 Conformational Breathing of RSV-F Affects Neutralization Sensitivity 113
- 4.6.2 Measles Versus RSV Vaccine Development 113
- 4.6.3 The Lack of Immunogenicity, Immune Evasion, or Immune Suppression 114
- 4.6.4 Recent RSV Vaccines 114
- 4.6.4.1 Success in Secondary Endpoints 114
- 4.6.5 A Resurgence in Attenuated Replication-Competent Vaccines 114
- 4.6.5.1 Mechanisms of Protection from RSV that Require Attention in Vaccine Development *115*
- 4.7 Antivirals and Therapeutic Antibodies 115
- 4.7.1 Immunoprophylaxis 115
- 4.7.2 RSV-IGIV 115
- 4.7.3 Palivizumab 116
- 4.8 Therapeutics for Treating Active RSV Infections 116
- 4.8.1 Immunotherapy under Clinical Development 117
- 4.9 Drug Targets 118
- 4.9.1 Novel Antivirals Undergoing Clinical Development 118
- 4.9.2 Other Investigational Treatments and Broad-Spectrum Antivirals 119
- 4.10 Conclusions 120 References 120

5 Herpes Simplex Viruses 133

Alexander Birkmann

- 5.1 Introduction 133
- 5.2 Overview of the Viral Replication Cycle 134
- 5.3 Treatment of HSV Infections 135
- 5.4 Approved Anti-HSV Drugs 136
- 5.4.1 5-Substituted 2-Deoxyuridine Analogues 136
- 5.4.2 Vidarabine 137
- 5.4.3 Acyclic Guanosine Analogues 137
- 5.4.4 Foscarnet 139

viii Contents

- 5.4.5 Docosanol 139
- 5.5 Anti-HSV Drugs in Advanced Development or Recently Entering the Market 140
- 5.5.1 Helicase–Primase Inhibitors 140
- 5.5.2 Monoclonal Antibodies 141
- 5.5.3 Therapeutic Vaccines 141
- 5.5.4 Immunomodulators 142
- 5.6 HSV Resistance to Antiviral Drugs 142
- 5.6.1 Epidemiology and Manifestation 142
- 5.6.2 Resistance Mechanisms 143
- 5.6.3 Management of Resistant HSV 143
- 5.7 HSV and Alzheimer's Disease 144
- 5.8 Conclusion 144
 - References 144

6 Antiviral Strategies Against the Human Cytomegalo Virus Inhibitors of Viral Terminase 151

Helga Rübsamen-Schaeff

- 6.1 The Need for Novel Drugs against CMV and Attempts of the Past 151
- 6.2 The Strategy for the Discovery of Letermovir 154
- 6.3 The Link between Preclinical Models and Clinical Efficacy 156
- 6.4 Clinical Experience 159
- 6.4.1 Resistance Mutations and Resistance Development 164
- 6.4.2 Resistance Observed in Clinical Studies 165
- 6.5 Other Potential Indications for Letermovir 167
- 6.6 Conclusions 169 References 170

7 Antiviral Targeting of the Complex Epstein Barr Virus Life Cycle 175

Christian Münz

- 7.1 Disease Overview 175
- 7.2 Antiviral Strategies 177
- 7.2.1 Pharmacological Inhibition 177
- 7.2.2 B Cell Depleting Therapy 179
- 7.2.3 Adoptive T Cell Transfer and Immune Checkpoint Blockade 180
- 7.2.4 Vaccination 181
- 7.3 Open Issues 182
 - Acknowledgements 183
 - References 183
- 8 Kaposi's Sarcoma-associated Herpesvirus-Antiviral Treatment 191
 - Thomas F. Schulz
- 8.1 Introduction to Kaposi's Sarcoma-associated Herpesvirus (KSHV) 191
- 8.2 Epidemiological Considerations 191

Contents ix

- 8.3 Disease Overview 192
- 8.3.1 Kaposi's Sarcoma 192
- 8.3.2 Primary Effusion Lymphoma (PEL) 194
- 8.3.3 Multicentric Castleman's Disease (MCD) 194
- 8.3.4 KSHV and Polyclonal Post-transplant Lymphoproliferative Disease 195
- 8.3.5 Kaposi Sarcoma Herpesvirus-Associated Inflammatory Cytokine Syndrome (KICS) 195
- 8.3.6 KSHV and Hemophagocytosis, Bone Marrow Failure, and Hepatitis 196
- 8.4 Antiviral Strategies 196
- 8.4.1 Established Antiviral Strategies 196
- 8.4.1.1 Inhibitors of the KSHV DNA Polymerase 196
- 8.4.1.1.1 Activity in Tissue Culture 196
- 8.4.1.1.2 Activity in KSHV-infected Patients 199
- 8.4.1.1.3 Treatment of Kaposi Sarcoma 200
- 8.4.1.1.4 Treatment of PEL 200
- 8.4.1.1.5 Treatment of MCD 201
- 8.4.2 Kinase Inhibitors Against KSHV-associated Disease 201
- 8.4.2.1 Clinical Experience 201
- 8.4.2.2 Preclinical Studies 201
- 8.4.3 Other Established and Experimental Therapies Targeting KSHV-infected Cells 206
- 8.4.3.1 Interfering with Cellular Cytokines—Clinical Experience 206
- 8.4.3.2 Preclinical Data on Other Cellular Targets 207
- 8.5 New Antiviral Strategies Against KSHV in Preclinical Development 208
- 8.5.1 New Approaches to Target Productive (Lytic) KSHV DNA Replication and/or Packaging 209
- 8.5.2 Targeting KSHV Capsid Assembly 209
- 8.5.3 Targeted Immunotoxins 210
- 8.5.4 Targeting the Latency Phase of the KSHV Life Cycle 210
- 8.5.4.1 LANA (Latency-associated Nuclear Antigen) 211
- 8.5.4.2 vFLIP (Viral FLICE-inhibitory Protein) 212
- 8.5.4.2.1 Outlook 212

References 213

9 Direct-Acting Antivirals for the Treatment of Chronic Hepatitis B Virus (HBV) Infection 229

Angelica Corcuera, Alastair Donald, and Andreas Urban

- 9.1 Nucleos(t)ide Analog Reverse Transcriptase Inhibitors 229
- 9.2 HBV Entry Inhibitors 232
- 9.3 HBV Capsid Assembly Modulators (CAMs) 232
- 9.3.1 HBV Core Protein 232
- 9.3.2 First-Generation CAMs 234
- 9.3.3 Recent Progress in Novel CAMs 236
- 9.3.4 Modes of Action of CAMs 237

x Contents

- 9.3.5 Resistance Profile of CAMs 238
- 9.3.6 CAMs in Clinical Trials 242
- 9.4 Concluding Remarks 244 References 244

10 Hepatitis E Virus–Current Developments in Antiviral Strategies 253

Daniel Todt, Patrick Behrendt, and Eike Steinmann

- 10.1 Introduction 253
- 10.2 Genetic Diversity and Molecular Virology of HEV 254
- 10.3 Clinical Course of HEV Infections 256
- 10.4 HEV Therapy 258
- 10.5 Development of Novel Antivirals Against HEV 259
- 10.6 Prevention of Infection and Vaccination Strategies 264
- 10.7 Conclusions 265 References 266

11 Antiviral Therapy of Adenovirus Infections 277

Peter Lischka

- 11.1 Human Adenovirus 277
- 11.2 Adenovirus in Human Stem Cell Transplantation 279
- 11.2.1 Incidence, Transmission, and Clinical Manifestation 279
- 11.2.2 Current Therapy Strategies and Antiviral Agents 281
- 11.2.2.1 Ribavirin 282
- 11.2.2.2 (Val)Ganciclovir 282
- 11.2.2.3 Cidofovir 283
- 11.2.3 Novel Antiviral Approaches in Clinical Development 283
- 11.2.3.1 Brincidofovir 283
- 11.3 Ocular Adenovirus Infections 285
- 11.3.1 Incidence, Transmission, and Clinical Manifestation 285
- 11.3.2 Current Therapy Strategies and Antiviral Agents 286
- 11.3.2.1 Corticosteroids 286
- 11.3.2.2 Povidone-iodine and SHP640 287
- 11.3.2.3 Nucleoside Analogues 287
- 11.3.2.4 Palliative Treatment 288
- 11.3.3 Novel Antiviral Approaches in Clinical Development 288
- 11.3.3.1 OKG-0301 288
- 11.3.3.2 iVIEW-1201 288
- 11.4 Drug Targets for Direct Acting Antivirals 289
- 11.4.1 Virus Attachment and Entry 289
- 11.4.2 Virus Genome Replication 290
- 11.4.3 Particle Maturation 290
- 11.5 Conclusion 291
 - References 291

```
12 ssDNA-Viruses: Human Parvovirus Infection 297
Susanne Modrow
```

- 12.1 Introduction 297
- 12.2 Classification 297
- 12.3 Molecular Biology 298
- 12.4 Parvovirus Replication 300
- 12.5 Diseases Associated with Parvovirus Infection 301
- 12.5.1 Parvovirus B19 301
- 12.5.2 HBoV 302
- 12.6 Antiviral Chemotherapy of Parvovirus B19-infection 303
- 12.7 Therapeutic Options and Recommendations 303
- 12.7.1 Acute B19V-infection Associated with Transient Aplastic Crisis (TAC) 303
- 12.7.2 Prolonged B19V-symptoms in Immunocompetents 304
- 12.7.3 Fetal Disease 304
- 12.7.4 Immunocompromised Patients 304 References 304
- **13** Antiviral Targets and Strategies to Treat and Prevent Human Norovirus Infections *313*
 - Jana Van Dycke, Arno Cuvry, Johan Neyts, and Joana Rocha-Pereira
- 13.1 Introduction 313
- 13.1.1 Disease Burden and Pathogenesis 313
- 13.1.2 Viral Genome 315
- 13.1.3 Replication Cycle 315
- 13.1.4 In Vitro Models 316
- 13.1.5 In Vivo Models 317
- 13.1.5.1 Human Norovirus 317
- 13.1.5.2 The Murine Norovirus as a Surrogate for In Vivo Studies 318
- 13.2 Antiviral Targets 318
- 13.2.1 Binding and Entry 318
- 13.2.2 Nonstructural Proteins 320
- 13.2.2.1 NS1/2 (p48) 320
- 13.2.2.2 NS3 (NTPase/Helicase) 320
- 13.2.2.3 NS4 (p22) 321
- 13.2.2.4 NS5 (VPg) 321
- 13.2.2.5 NS6 (Protease) 322
- 13.2.2.6 NS7 (RNA-Dependent RNA Polymerase) 322
- 13.2.3 Host Factors 325
- 13.2.4 Antibodies 328
- 13.2.5 Gut Microbiota 328
- 13.2.6 Unknown Targets 330
- 13.3 Vaccine Development 331
- 13.4 Conclusion and Perspectives 332 References 332

xii Contents

14 Antiviral Strategies Against (Non-polio) Picornaviruses 347

Rana Abdelnabi, Jim Baggen, Carmen Mirabelli, and Johan Neyts

- 14.1 Classification and Clinical Impact 347
- 14.2 The Enterovirus Replication Cycle 347
- 14.2.1 Virion Structure 347
- 14.2.2 Genome Structure 348
- 14.2.3 Replication Cycle Stages 349
- 14.2.3.1 Entry 349
- 14.2.3.2 Translation and Replication 350
- 14.2.3.3 Assembly and Release 350
- 14.3 Prevention 351
- 14.4 Antiviral Strategies Against Enteroviruses 352
- 14.4.1 Directly Acting Antivirals 352
- 14.4.1.1 Early-Stage Inhibitors 352
- 14.4.1.2 Viral Protease Inhibitors 355
- 14.4.1.3 2C Targeting Molecules 356
- 14.4.1.4 RNA-dependent RNA Polymerase (RdRp) "(3D) Inhibitors" 356
- 14.4.2 Host-targeting Antivirals 357
- 14.4.2.1 Inhibitors of Lipid Processing 357
- 14.4.2.2 Assembly Inhibitors 357
- 14.4.3 Monoclonal Antibodies 358
- 14.4.4 Alternative Strategies 358
- 14.4.4.1 Direct Targeting of the RNA Genome 358
- 14.4.4.2 Dual Target Therapeutics 358
- 14.5 Conclusions 359 References 359

15 Novel Antiviral Strategies Against Emerging Arbovirus Infections 367

Yannick Brüggemann, Carina Elsner, and Stephanie Pfaender

- 15.1 Introduction 367
- 15.1.1 Zika Virus *368*
- 15.1.2 Yellow Fever Virus 368
- 15.1.2.1 Dengue Virus 369
- 15.1.3 West Nile Virus 369
- 15.1.4 Chikungunya Virus 370
- 15.2 Intervention Strategies 371
- 15.3 Genome Organization and Viral Replication Cycle 371
- 15.4 Targets For Antiviral Therapy 374
- 15.5 Direct-acting Antivirals (DAAs) 374
- 15.6 RNA-dependent RNA Polymerase Inhibitors 374
- 15.7 Protease/Helicase Inhibitors 377
- 15.8 Envelope Protein Inhibitors 378
- 15.9 Capsid Protein Inhibitors 379
- 15.10 NS4B Inhibitors 379
- 15.11 Methyltransferase Inhibitors 379
- 15.12 Inhibitors with Nonspecific Action 380

Contents xiii

- 15.13 Host Targeting Antivirals 380
- 15.14 Host Cell Nucleoside Biosynthesis Inhibitors 381
- 15.15 Host Cell Lipid Biosynthesis Inhibitors 382
- 15.16 Host Kinase Inhibitors 384
- 15.17 Protein Metabolism Inhibitors 385
- 15.18 Endocytosis and Membrane Fusion Inhibitors 387
- 15.19 Conclusion and Future Perspectives 387 References 390

Lucie Fénéant, Bianca Bodmer, Thomas C. Mettenleiter, Allison Groseth, and Thomas Hoenen

- 16.1 Introduction 409
- 16.2 Filoviruses 410
- 16.2.1 Virus and Disease Overview 410
- 16.2.2 Antiviral Strategies 411
- 16.2.2.1 Immunotherapies 411
- 16.2.2.2 Postexposure Prophylaxis 412
- 16.2.2.3 Small Molecules 413
- 16.2.2.4 Antisense Therapy 418
- 16.2.2.5 Host-directed Therapies 418
- 16.3 Henipaviruses 418
- 16.3.1 Disease Overview 419
- 16.3.2 Antiviral Strategies 419
- 16.3.2.1 Small Molecules 419
- 16.3.2.2 Immunotherapies 420
- 16.4 Arenaviruses 420
- 16.4.1 Old World Arenaviruses 420
- 16.4.1.1 Disease Overview 423
- 16.4.1.2 Antiviral Strategies 423
- 16.4.1.2.1 Small Molecules 423
- 16.4.1.2.2 Immunotherapies 425
- 16.4.2 New World Arenaviruses 426
- 16.4.2.1 Disease Overview 426
- 16.4.2.2 Antiviral Strategies 427
- 16.4.2.2.1 Immunotherapies 427
- 16.4.2.2.2 Small Molecules 428
- 16.5 Bunyaviruses 429
- 16.5.1 Disease Overview 429
- 16.5.2 Antiviral Strategies 430
- 16.5.2.1 Immunotherapies 430
- 16.5.2.2 Small Molecules 430
- 16.6 Considerations for the Development of Treatment Strategies Against Viral Hemorrhagic Fever Viruses 433 References 435

¹⁶ Current Therapies for Biosafety Level 4 Pathogens 409

xiv Contents

17 A Focus on Severe Acute Respiratory Syndrome (SARS) Coronavirus (SARS-CoVs) 1 and 2 449

Rolando Cannalire, Enzo Tramontano, and Vincenzo Summa

- 17.1 Overview on Coronavirus (CoV) 449
- 17.1.1 CoV Epidemiology 449
- 17.1.2 HCoV Replication 450
- 17.1.3 Human Corona Virus (HCoV) Diseases Natural History 453
- 17.2 Licensed and Clinical Investigational Drugs Against CoVs 454
- 17.2.1 Licensed Drugs Against CoVs 455
- 17.2.2 Clinical Investigational Drugs Against CoVs 456
- 17.3 Medicinal Chemistry Approaches Toward the Identification of New Drugs 459
- 17.3.1 SARS-CoVs RdRp: Structure, Function, and Inhibition 460
- 17.3.2 The CoVs 3CL^{pro} (or Main Protease): Structure, Function, and Inhibitors 463
- 17.3.3 The S Protein: Structure, Function, and Inhibition 472
- 17.4 Conclusions 477 References 478

Index 487

Preface

The area of Medicinal Chemistry aimed at fighting viral diseases has seen enormous progress in recent decades, making this updated reference from Helga Rübsamen-Schaeff and her team of expert authors more than overdue, after Eric De Clercq published last in this series back in 2010.

Many important, novel antiviral drugs have reached patients in recent years, making diseases such as AIDS or Hepatitis C manageable today, while only some years back, their diagnosis had been comparable to a death sentence. While we, the series editors, write this preface, our world is in the midst of a global pandemic caused by SARS CoV-2 (COVID-19), and promising, novel small molecule-based antiviral medicines have reached the clinical stage in record time, complementing vaccine and drug repurposing efforts to manage a global health crisis.

This impressive evolution is the result of the fascinating and relentless efforts of creative expert drug hunters, not accepting the status quo and pushing the boundaries of Medicinal Chemistry to combat the fascinating, yet often lethal efficiency of small fragments of viral RNA or DNA. Their success stories will be told in this fascinating book by Helga and her fellow authors, along with a thorough background review of all relevant viral biology and clinical pathology for the most important viral diseases. The reader will learn about cutting-edge drug discovery strategies to successfully develop antiviral agents, ranging from the use on non-classic elements, selective and kinetically controlled inhibition of viral enzymes, other viral proteins, and even of their RNA or capsids, often supported by the intelligent use of structure-enabled Medicinal Chemistry design.

The first two chapters focus on the science leading to the breakthrough in treating patients with HIV and Hepatitis C, followed by comprehensive overviews on Research and Development, and market entry for drugs against important viruses such as Influenza Virus, Respiratory Syncytial Virus, Herpes Simplex Virus, Human Cytomegalo Virus, Epstein Barr Virus, Kaposi's Sarcoma-associated Herpesvirus, Hepatitis B Virus, and Hepatitis E Virus (Chapters 3–10). Chapters 11–15 focus on Adeno-, Parvo-, Noro-, Picorna-, and Arboviruses RnD, respectively. Finally, a chapter on Biosafety Level 4 Pathogen Therapies and one on SARS CoV-2 virus conclude the reference book.

The combination of background, strategic insights, and applied case studies makes this volume a must read for every scientist involved in antiviral RnD, but also for the wider Medicinal Chemist and drug hunter community, interested in broadening expertise and skillsets.

xvi Preface

Helga Rübsamen-Schaeff led Bayer's Virology Research functions from 1994 until 2001 and Bayer's whole Infectious Diseases Research from 2001 to 2006, when she became founder CEO of AiCuris, a German biotech company dedicated to delivering innovative antiviral and antibacterial drugs to patients. Helga has received numerous awards in her career, e.g. recently the Loeffler-Frosch-Medal of the German Society for Virology. Together with her research teams, she has delivered several important medicines and drug candidates, such as recently Letermovir against the cytomegalovirus. For the work on Letermovir, she received the Innovation Award of Northrhine-Westphalia and the Price for Technology and Innovation by the German President of State in 2018. She is member of the National Academy of Sciences, Leopoldina.

The editors would like to thank Helga and all contributing authors for what we believe to be the most complete and up-to-date reference book for antiviral research and development.

Boston, Aachen, and Düsseldorf July 2021

Jörg Holenz Helmut Buschmann Raimund Mannhold In Memoriam: Farah Elawar



On the day we received the proof pages for Chapter 4 of this book, we were shocked and deeply saddened to learn about the tragic passing of Farah Elawar. Farah is the lead author of our contribution: Respiratory Syncytial Virus Immunoreactivity, Vaccine Development, and Therapeutics. She was in the final stages of her Ph.D. studies on the diversity and drug resistance of Respiratory Syncytial Virus in the community, in the Department of Medical Microbiology and Immunology at the University of Alberta. From the beginning of her studies with the Marchant laboratory, she impressed all of us with her enthusiasm and everlasting energy. She was always three steps ahead of her supervisor with the results of her experiments. It was almost as though she could read minds. With this ability, she provided the leadership to conceptualize, write, and finalize this book chapter. With her warm personality, she has made so many friends and has given us so many joyous moments. Farah will be dearly missed and always remembered.

David Marchant and Matthias Gotte

xvii

Introduction

More than 60 years ago, in 1959, the first antiviral drug to be licensed was described by Prusoff [1]. It was marketed for the topical treatment of herpes simplex virus infections of the eye. In 1972, Ribavirin was described as a broad-spectrum antiviral agent [2]; however, it had significant side-effects. In contrast, Acyclovir, discovered in 1977, was very well tolerated [3, 4] and became the gold standard for treating infections caused by herpes viruses.

With the discovery of HIV in 1983 as the virus causing AIDS [5], a very active era of search for anti-HIV drugs began. In 1985, the antiviral activity of an existing drug, AZT (zidovudine, a polymerase inhibitor) was first described against HIV [6]. While up to that time, all drugs, with the exception of ribavirin, to which multiple modes of action are being ascribed, were inhibitors of viral polymerases (in case of AZT, the reverse transcriptase), the following years witnessed a very active search for drugs inhibiting other targets of HIV like the protease, fusion, or integrase or non-nucleosidic inhibitors of HIV reverse transcriptase as novel drug classes [7]. Finally, about 24 years after HIV had first been discovered, by combining drugs with different modes of action, the HIV-infection, which had been a death sentence, became a treatable, although chronic disease allowing patients to live a nearly normal lifespan. Likewise, for Hepatitis C (HCV), the discovery of the virus led to a worldwide search for specific antiviral drugs targeting its polymerase, protease, or NS5A. In this case, combining drugs with different modes of action even allowed to cure the chronic HCV infection in the vast majority of patients, 25 years after HCV had first been described. These unprecedented and major achievements against two of the most dangerous small RNA viruses demonstrate the enormous power of academic and industrial research, when combined and targeted toward a specific virus. They are described in the first two chapters of the book. Obviously, while HCV infections can now be cured this goal is still to be reached for HIV and many approaches are being pursued.

The next two chapters deal with two RNA viruses, which also pose significant health problems: Influenza and the Respiratory Syncytial Virus and describe the existing high medical need for therapies in these indications, as well as starting points for novel therapeutic options.

Chapters 5–8 deal with large DNA viruses like Herpes Simplex, Cytomegalovirus, Epstein–Barr Virus, and the Human Herpes Virus 8, all widespread and implicated in a number of severe or fatal conditions. The successful development of novel generations of drugs against Herpes Simplex and Cytomegalovirus using novel targets for attacking these viruses will be described (Chapters 5 and 6). The next chapters highlight potential targets

xx Introduction

and strategies to be addressed in the search of therapeutics against the Epstein–Barr Virus or the Human Herpes Virus 8 (Chapters 7 and 8).

Chapter 9 describes antiviral efforts against a small DNA virus, Hepatitis B, which causes significant morbidity and mortality worldwide, especially in Asia, and against which strategies for a cure are being sought as well. Here, inhibitors against the viral capsid could be one potential avenue.

The following chapters address small DNA viruses (Hepatitis E, Adeno, and Parvo) and RNA viruses (Noro, Picorna), followed by chapters on emerging viruses like arbovirus infections and biosafety 4 level viruses like EBOLA.

Outbreaks of HIV, SARS-1 Coronavirus, MERS Coronavirus, EBOLA, and Zika clearly have indicated that the globalized world has become very vulnerable to epidemics of often zoonotic viruses infecting humans and spreading quickly due to the strong and multiple connections in to-day's world. While containment of several outbreaks or eventually treatment (HIV) and even cure (HCV) has been possible in the past, we are now facing an unprecedented outbreak of SARS CoV-2 causing the disease COVID-19. Infections with the virus were first documented in 2019, and the next two years saw a rapid spread leading to a pandemic with millions of cases worldwide and 5 million deaths by the fall of 2021. Efficient transmission by air makes containment of SARS CoV-2 particularly difficult and resulted in very significant economical downturns worldwide. SARS-CoV-2 causes severe respiratory symptoms, but also pathological inflammation and multi-organ-dysfunction, including the acute respiratory distress symptom, cardiovascular events, coagulopathies, nephropathy, and neurological symptoms [8-11]. While several highly active vaccines have meanwhile been discovered and vaccination campaigns are pushed worldwide, there is still a great need for highly potent and well-tolerated direct acting antiviral agents. We will give medicinal chemists insights into targets and strategies for the discovery of these urgently needed therapies against SARS-Cov-2 in the final chapter of this book.

Most of this book will deal with small molecular weight drugs and their targets, but where appropriate and potentially also the better strategy, immune modulators or immune therapies will be discussed as well.

References

- 1 Prusoff, W.H. (1959). Synthesis and biological activities of iodo deoxyuridine, an analog of thymidine. *Biochimica and Biophysica Acta*, 32: 295–296.
- Sidwell, R.W. Huffmann, J.H., Kahare, G.P., Allen, L.B., Witkowski, J.T. and Robins, R.K. (1972). Broad-spectrum antiviral activity of virazole: 1-B-D-ribofuranosyl-1,2,4triazole-3-carboxamide. *Science* 177: 705–706.
- Elion, G.B., Furman, P.A., Fyfe, J.A., de Miranda, P., Beauchamp, L., and Schaeffer,
 H.J. (1977). Selectivity of action of an antiherpetic agent, 9-(2-hydroxyethoxymethyl)
 guanine. *Proceedings of the National Academy of Sciences of the United States of America*,
 74: 5716–5720.

- **4** Schaeffer, H.J., Beauchamp, L., de Miranda, P., Elion, G.B., Bauer, D.J., and Collins, P. (1978). 9-(2-Hydroxyethoxymethyl) guanine activity against viruses of the herpes group. *Nature* 272: 583–585.
- 5 Barré-Sinoussi, F., Chermann, J.C., Rey, F., Nugeyre, M. T., Chamaret, S., Gruest, J., Dauguet, C., Axler-Blin, C., Vézinet-Brun, F., Rouzioux, C., Rozenbaum, W., Montagnier, L. (1983). Isolation of a T-lymphotropic retrovirus from a patient at risk for acquired immune deficiency syndrome (AIDS). *Science* 220(4599): 868–71, doi: 10.1126/ science.6189183
- 6 Mitsuya, H., Weinhold, K.J., Furman, P.A., St. Clair, M.H., Lehrman, S.N., Gallo, R.C., Bolognesi, D., Barry, D.W. and Broder, S. (1985). 30-Azido-30-deoxythymidine (BW A509U): an antiviral agent that inhibits the infectivity and cytopathic effect of human T lymphotropic virus type III/lymphadenopathy-associated virus in vitro. *Proceedings of the National Academy of Sciences of the United States of America*, 82: 7096–7100.
- 7 De Clercq, E. (2010). Outlook of the antiviral drug era, now more than 50 years after description of the first antiviral drug, in Antiviral Drug Strategies (De Clercq, E. Ed.), Wiley-VCH, Weinheim, Page 1.
- 8 Zhou, P., Yang, X.L., Wang, X.G., Hu, B., Zhang, L., Zhang, W., Si, H.R., Zhu, Y., Li, B., Huang, C.L., Hui-Chen, H.-D, Chen, J., Luo, Y., Guo, H., Jiang, R.-D., Liu, M.-Q., Chen, Y., Shen, X.-R., Wang, X., Zheng, X.-S., Zhao, K., Chen, Q.-J., Deng, F., Liu, L.-L., Yan, B., Zhan, F.-X., Wang, Y.-Y., Xiao, G.-F. and Shi, Z.-L. (2020). A pneumonia outbreak associated with a new coronavirus of probable bat origin. *Nature* 579: 270–273.
- **9** Helms, J., Kremer, S., Merdji, H., Clere-Jehl, R., Schenck, M., Kummerlen, C., Collange, O., Boulay, C., Fafi-Kremer, S., Ohana, M. and Anheim, M. (2020). Neurologic Features in Severe SARS-CoV-2 Infection. *New England Journal of Medicine* 382: 2268–2270.
- 10 Zhu, N., Zhang, D., Wang, W., Li, X., Yang, B., Song, J., Zhao, X., Huang, B., Shi, W., Lu, R., Niu, P., Zhan, F., Ma, X., Wang, D., Xu, W., Wu, G., Gao, G.F., and Tan, W. for the China Novel Coronavirus Investigating and Research Team. (2020). A novel coronavirus from patients with pneumonia in China, 2019. *New England Journal of Medicine* 382: 727–733.
- 11 Gupta, A., Madhavan, M.V., Sehgal, K., Nair, N., Mahajan, S., Sehravat, T.S., Bikdeli, B., Ahluvalia, N., Ausiello, J.C., Wan, E.Y., Freedberg, D.E., Kirtane, A.J., Parikh, S.A., Maurer, M.S. Nordvig, A.S., Accili, D., Bathon, J.M., Mohan, S., Bauer, K.A., Leon, M.B., Krumholz, H.M., Uriel, N., Mehra, M.R., Elkind, M.S.V., Stone, G.W., Schwartz, A., Ho, D.D., Bilezikian, J.P. and Landry, D.W. (2020). Extrapulmonary manifestations of COVID-19. *Nature Medicine*, 26: 1017–1032.

HIV—Disease Overview, Targets for Therapy and Open Issues

Christoph Stephan

Internal Medicine & Infectious Diseases, Medical Center/Infectious Diseases Unit, Goethe University Hospital Frankfurt, Frankfurt am Main, Germany

Nearly 40 years have passed since the first patients were discovered to have the previously unknown "acquired immunodeficiency syndrome" (AIDS) [1] and 38 years, since the causative virus was discovered [2]. The time since, has been characterized by a dynamically evolving body of epidemiological and basic science that is unique in medical history. Preliminary result is the development of antiretroviral treatments and social medicine progress, accompanied and catalyzed by the predominantly affected risk groups' emancipation, with reintegration into the society.

1

1.1 HIV–Disease Overview

There has not been a significant change in the **natural history of HIV-infection**; however, today in industrialized countries, disease manifestation in the form of AIDS can only be observed in undiagnosed, late presenting patients, who already show manifestations of immunological deterioration. Figure 1.1 demonstrates the course of the disease, according to the two most important measurable surrogate markers—CD4-cell count and HIV-RNA (viral load).

After initial HIV transmission, the retrovirus spreads throughout the human body and infects potentially all CD4-receptor positive cells. Consequently, during the first weeks of infection, there is a substantial fall in CD4-positive T-lymphocyte count and a rise in viral load—up to a turning point. Thereafter, CD4-cells rise again and viral load decreases, due to regain of a partial immunological control. At this time, anti-HIV antibodies can be found in plasma, and the patient will now respond positively to serological HIV-tests. Elimination of HIV, however, will not occur due to the rapid variation of viral surface receptors which may hide infected cells from the immune system and lead to divergent virus populations, including in a single patient [3, 4, 5, 6]. This continuous change of HIV stems from proofreading failures which lead to the evolution of many



Figure 1.1 Shows the course of both most important measurable surrogate markers from blood count, i.e. CD4-cell count and HIV-RNA (viral load), during untreated HIV infection.

HIV quasispecies. Another reason for the inability to eliminate HIV is the infection of durable reservoir and "archive" cells, e.g. edaphic CD4-receptor-positive macrophage and monocyte cells, leading to the chronic phase of the infection. For approximately 3–10 years, the patient will experience a relatively stable period, marked by individually solid CD4-cell count- and HIV-RNA viral load- "setpoints" [7]. However, after months or years, an immunological exhaustion will occur. Then, the CD4-cell count is substantially decreasing and viral load is rising again. The result may be AIDS, as defined by the emergence of at least one of 26 opportunistic infections and/or tumors, including pneumocystis pneumonia, cerebral toxoplasmosis, tuberculosis, cytomegalovirus retinitis, Kaposi's sarcoma, or B-cell non-Hodgkin lymphoma.

1.2 Targets for Antiretroviral Therapy

The replication of the retrovirus in the human host cell is well described and offers targeted treatment options, in order to prevent viral replication. Figure 1.2 shows the passage of HIV through the human host CD4-receptor positive T-cell. Antiretroviral drugs are able to address specific points in the HIV replication cycle and used in combination antiretroviral therapy (cART) aims to completely suppress HIV-1 replication long term. This will give the immune system a chance to recover and overcome opportunistic infections and tumors and/or to avoid significant deterioration from the beginning, when applied early after infection.

During the retrovirus replication in the human cell, specific cART-drug classes offer to interfere with different therapeutic intervention targets (see Table 1.1, presently favored drugs are printed in bold and Figure 1.2). Such interventions comprise: inhibition of first contact of HIV with the CD4-positive cell (attachment), the cell entry, intracellular reverse

2 1 HIV-Disease Overview, Targets for Therapy and Open Issues

transcriptase-enzyme and -activity, DNA-integration, virus assembling by proteases, and virus maturation; the latter leaves immature, noninfectious virus particles. The chemical structures of the different HIV drugs can be found in Table 1.2.



Figure 1.2 Shows the viral replication cycle for HIV in the human target cell, i.e. the CD4-receptor positive cell, and six treatment targets for antiretroviral therapy classes (for numbers in red, see Table 1.1/row 1).

Table 1.1	Explains the mode of action for available cART options, according to the target area
in HIV cell	passage (for numbers: see Figure 1.2).

No. (Fig- ure 1.2)	Drug class: inhibitor of	Generic names of available cART-drugs	Mode of action: inhibition of the	Formula	CAS registry number
1	Attachment	Fostemsavir ^a	HIV-binding site gp120, used by HIV for first contact with CD4-receptor	$C_{25}H_{26}N_7O_8P$	864953-29-7
		Ibalizumab (TNX-355)	Human CD4-receptor binding site for HIV (whole antibody)	n/a	680188-33-4
2	Entry-/ fusion	Enfuvirtid (T20)	HIV-gp41-fusion protein (36 amino acids- containing polypeptide)	$C_{204}H_{301}N_{51}O_{64}$	159519-65-0
		Maraviroc (MVC)	Human CCR5-coreceptor	$C_{29}H_{41}F_2N_5O$	376348-65-1

(Continued)

4 1 HIV–Disease Overview, Targets for Therapy and Open Issues

Table 1.1 (Continued)

No. (Fig- ure 1.2)	Drug class: inhibitor of	Generic names of available cART-drugs	Mode of action inhibition of th	I: Ie	Formula	CAS registry number
3	Nucleosidal reverse	Zidovudin (AZT)	Thymidin- analogue	Nucleobase replacement	$C_{10}H_{13}N_5O_4$	30516-87-1
	$\begin{array}{c} \mbox{transcriptase} \\ (NRTI) & \begin{tabular}{lllllllllllllllllllllllllllllllllll$	Lamivudin (3TC) ^b	Cytidin- analogue	during RNA–DNA-	$C_8H_{11}N_3O_3S$	134678-17-4
		as faulty chip, leading to	$\mathrm{C_8H_{10}FN_3O_3S}$	143491-57-0		
		early chain termination	$C_9H_{14}N_5O_4P$	147127-20-6		
		Abacavir (ABC) ^b	Gunaosine- analogue		$C_{14}H_{18}N_6O$	136470-78-5
4	Non- nucleosidal	Efavirenz (EFV)	Reverse transo enzyme bindi	criptase- ng site	$\begin{array}{c} C_{14}H_9ClF_3\\ NO_2 \end{array}$	154598-52-4
1	reverse transcriptase (NNRTI)	Nevirapine (NVP)			$C_{15}H_{14}N_4O$	129618-40-2
		Rilpivirine (RPV) ^b			$C_{22}H_{18}N_6$	500287-72-9
		Doravirin (DOR) ^{b?>}			$C_{17}H_{11}ClF_3 \\ N_5O_3$	1338225-97-0
5	Integrase strand	Raltegravir (RGV)	Viral DNA int human host D	egration in DNA, in cell	$C_{20}H_{21}FN_6O_5$	518048-05-0
	transfer (INSTI)	Elvitegravir (EVG)	nucleus		C ₂₃ H ₂₃ ClFNO ₅	697761-98-1
		Dolutegravir (DGT) ^b			$C_{20}H_{19}F_2N_3O_5\\$	1051375-16-6
		Bictegravir (BTG) ^b			$C_{21}H_{18}F_{3}N_{3}O_{5}$	1611493-60-7
		Cabotegravir ^a			$C_{19}H_{17}F_2N_3O_5\\$	1051375-10-0
6	Protease (PI)	Darunavir (DRV) ^b	gag-pol-polyp cleavage	rotein	$C_{27}H_{37}N_3O_7S$	206361-99-1
		Atazanavir (ATV)			$C_{38}H_{52}N_6O_7\\$	198904-31-3
		Lopinavir (LPV)			$C_{37}H_{48}N_4O_5\\$	192725-17-0
7	Maturation/ capsid	Lenacapavir ^a	Extracellular of arrangement ^a	capsid	$\begin{array}{c} C_{39}H_{32}ClF_{10} \\ N_7O_5S_2 \end{array}$	2189684-44-2
		GSK3640254 ^a	Last protease event: CA-p24	cleavage /SP1 ^a	n/a	n/a

a) In clinical study development—also refer to public study registry online-resource, available at: https:// www.clinicaltrials.gov.

b) Modern, recommended first-line combination antiretroviral therapy components: printed in **bold**. For treatment guidelines from the European AIDS Clinical Society (EACS), version 10, from November 2019, please refer to online-resource, available at: https://www.eacsociety.org/files/2019_guidelines-10.0_final.pdf.

 Table 1.2
 Illustrates chemical formula details and CAS registry numbers for available cART-drugs.

Drug class: inhibitor of	Generic drug name/chemical formula/CAS registry no./graphic structure formula	Generic drug name/chemical formula/CAS registry no./graphic structure formula
Attachment	Fostemsavir ⁹ $C_{23}H_{26}N_7O_8P$ 864953-29-7 $G = \begin{pmatrix} HO & O \\ HO & O \\ HO & OH $	Ibalizumab (TNX-355) n/a 680188-33-4 n/a—chemical formula not yet published, the drug is a humanized mouse whole antibody
Entry-/fusion	Enfuvirid (T20) $C_{20H}H_{30} N_{5} O_{64}$ 159519-65-0 $\int_{0}^{H} \left(\sum_{k=0}^{n} \sum_{k$	Maraviroc (MVC) $C_{29}H_{41}F_2N_5O$ 376348-65-1 F_{F}

Table 1.2 (Continued)





Table 1.2 (Continued)





Source: WIKIPEDIA, the free encyclopedia, as accessed online on 30 December 2019, please also see https://en.wikipedia.org. a) In clinical study development—refer to public study registry online-resource, available at: https://www.clinicaltrials.gov.

1.3 Currently Open Issues in HIV/AIDS Research

Important milestones in cART development have been achieved in the recent decades as standard of care: complete virus suppression, side-effect- and drug-interaction control, and convenience in taking antiretroviral regimens. Recently observed trends in HIV-treatment include the development of long-acting cART drugs, which are administered alternatively, e.g. injected every eight weeks, or once even less frequently implanted periodically.

When in July 2015 the first results from the START-study (Strategic Timing of Antiretroviral Treatment) were published, the benefit from modern cART for patients with early HIV-infection was evident for the first time [8]. Subsequently, antiretroviral therapy guidelines have changed worldwide and recommend cART for everybody with an HIV-infection, independent from the individual clinical category and CD4-cell count. Thereafter, the global focus of interest was to establish programs that could allow every infected person access to cART. Therefore, the United Nations Program on HIV/AIDS (UNAIDS) have established the 90–90–90-targets, in order to end AIDS as a disease and to control HIV transmissions on a society level [9]. Data on beneficial effects of programs, which lowered the barriers to cART, i.e. linked to HIV transmission control, were published before [10]. Moreover, the exciting confirmation of the SWISS STATEMENT hypothesis (Undetectable HIV leads to zero transmissions) [11] was a major step forward to realize antiretroviral treatment as most effective prevention [12] in the absence of a protective vaccine.

Another major open issue is cure from HIV/AIDS. Albeit individual cases of cure from HIV have been reported [13, 14], e.g. by stem cell transplantation from donors with the rare, intrinsic HIV-resistance due to homozygous CCR5- Δ 32/ Δ 32-gene mutations, stem cell transplantation will hardly be feasible for many patients, as this is associated with substantial risks. Alternative therapeutic procedures using the CRISPR-CASP-technique have been tried, but still require further developments [15]. Beyond stem cell manipulation and/or transplantation, the efforts to induce broadly neutralizing antibodies (BNAPs) remain a second scientific approach to achieve at least "functional cure" from HIV [16].

References

- 1 CDC (1981). Pneumocystis pneumonia—Los Angeles. MMWR 30: 250–252.
- **2** Barré-Sinoussi, F., Chermann, J.C., Rey, F. et al. (1983). Isolation of a T-lymphotropic retrovirus from a patient at risk for acquired immune deficiency syndrome (AIDS). *Science* 220 (4599): 868–871.
- **3** Rübsamen-Waigmann, H., Becker, W.B., Helm, E.B. et al. (1986). Isolation of variants of lymphocytopathic retroviruses from the peripheral blood and cerebrospinal fluid of patients with ARC or AIDS. *J. Med. Virol.* 19: 335–344.
- **4** von Briesen, H., Becker, W.B., Henco, K. et al. (1987). Isolation frequency and growth properties of HIV-variants: multiple simultaneous variants in a patient demonstrated by molecular cloning. *J. Med. Virol.* 23: 51–66.

- 5 Schwartz, S., Fenyo, E.-M., Felber, B.K., and George, N. (1989). Pavlakis rapidly and slowly replicating human immunodeficiency virus type-1 isolates can be distinguished according to target-cell tropism in T-cell and monocyte cell-lines. *Proceedings of the National Academy of Sciences*, October 1989. doi:https://doi.org/10.1073/pnas.86.18.7200. *Source*: PubMed.
- **6** Asjö, B., Albert, J., Karlson, A. et al. (1986). *Lancet* ii: 660–662. Tersmette, M., Lange, I.M.A., de Goede, R.E.Y. et al. (1989). *Lancet* i: 983–985.
- 7 O'Brien, T.R., Rosenberg, P.S., Yellin, F., and Goedert, J.J. (1998). Longitudinal HIV-1 RNA levels in a cohort of homosexual men. *J. Acquir. Immune Defic. Syndr. Hum. Retrovirol.* 18 (2): 155–161.
- **8** INSIGHT START Study Group, Lundgren, J.D., Babiker, A.G. et al. (2015). Initiation of antiretroviral terapy in early asymptomatic HIV infection. *N. Engl. J. Med.* 373 (9): 795–807.
- **9** WHO (2017). *UNAIDS documents: ending AIDS: progress towards the 90–90–90 targets*, published 20 July 2017. https://www.unaids.org/sites/default/files/media_asset/ Global_AIDS_update_2017_en.pdf.
- **10** Nosyk, B., Min, J., Lima, V.D. et al. (2013). HIV-1 disease progression during highly active antiretroviral therapy: an application using population-level data in British Columbia: 1996–2011. *J. Acquir. Immune Defic. Syndr.* 63 (5): 653–659.
- **11** Vernazza, P., Hirschel, B., Bernasconi, E., and Flepp, M. (2008). HIV-infizierte Menschen ohne andere STD sind unter wirksamer antiretroviraler Therapie sexuell nicht infektiös. *Bull. Med. Suisses* 89: 165–169.
- **12** Cohen, M.S., Chen, Y.Q., McCauley, M. et al. (2011). Prevention of HIV-1 infection with early antiretroviral therapy. *N. Engl. J. Med.* 365 (6): 493–505.
- Hütter, G., Nowak, D., Mossner, M. et al. (2009). Long-term control of HIV by CCR5 Delta32/Delta32 stem-cell transplantation. *N. Engl. J. Med.* 360 (7): 692–698.
- Gupta, R.K., Abdul-Jawad, S., McCoy, L.E. et al. (2019 Apr). HIV-1 remission following CCR5∆32/∆32 haematopoietic stem-cell transplantation. *Nature* 568 (7751): 244–248.
- **15** Xu, L., Wang, J., Liu, Y. et al. (2019). CRISPR-edited stem cells in a patient with HIV and acute lymphocytic leukemia. *N. Engl. J. Med.* 381 (13): 1240–1247.
- **16** Klein, F., Mouquet, H., Dosenovic, P. et al. (2013). Antibodies in HIV-1 vaccine development and therapy. *Science* 341 (6151): 1199–1204.
Michael J. Sofia

2

Arbutus Biopharma, Inc., Warminster, PA, USA

2.1 Introduction

Viral hepatitis is an inflammation of the liver that progressively leads to liver cirrhosis and eventually liver cancer. Every third person on the planet shows evidence of infection with viral hepatitis and some 500 million individuals are chronically infected. It is the leading cause of liver cancer worldwide and accounts for more than 80% of liver transplants globally. While mortality rates are dropping for many other infectious diseases, there has been an increase in mortality for those individuals diagnosed with chronic viral hepatitis [1].

It was not until the 1960s and 1970s that through a series of epidemiological studies and the work of several laboratories the hepatitis B virus (HBV) was discovered and linked to the occurrence of liver cancer. The development of a diagnostic test and ultimately of a vaccine dramatically reduced the incidence of HBV infection in the western world even though a cure remained far off on the horizon [2]. However, it turns out that HBV was not the only culprit that caused a chronic viral hepatitis that led to liver cancer. A mysterious non-A non-B hepatitis (NANBH) was identified by Harvey Alter and coworkers as a transfusion-associated hepatitis [2]. However, it was not until 1989 and the work of Michael Houghton and coworkers that the hepatitis C virus (HCV) was ultimately identified and its genome was mapped (Figure 2.1) [3, 4]. This work led to a diagnostic test that would allow for the screening of the donated blood supply and virtually eliminate acquiring HCV from blood transfusions [5]. Even with this breakthrough, acquisition of HCV infection via injection drug use and other means of contact with infected blood continued to propagate the disease such that the worldwide prevalence remained significant at approximately 75 million individuals. Because the development of a vaccine was and still is elusive and no cure was available, the spread of HCV continued and those who were infected had little hope that their future would be better.

In 1991, the first approved therapy for treating HCV was the cytokine interferon- α (IFN). First administered three times weekly, this therapy produced very low cure rates that came along with significant side effects such as anemia, neurological complications, and flu-like

14 *2* Curing Hepatitis C with Direct-Acting Antiviral Therapy



Figure 2.1 HCV Genome organization. The open reading frame (ORF) codes for a ~3000 amino acid precursor polyprotein that lies between the 3' and 5' nontranslated region (NTR). The NS2/3 protease and NS3/4A protease are involved in the polyprotein processing.

symptoms. The introduction of the nucleoside ribavirin (RBV), a nonspecific antiviral agent, led to improved cure rates but also added to the side-effect profile of this combination. The advent of pegylated-IFN (PEG-IFN) in combination with RBV, further improved cure rates and reduced the frequency of administration. However, cure rates remained modest and the serious side effects persisted [6].

Although IFN-containing regimens did provide a modest cure rate, the side effects were sufficiently severe that many patients could not complete the 48-week regimen. It became an aspirational objective to eliminate IFN from drug regimens that cured HCV, but in the early 2000s, it was not obvious how to accomplish this objective. In addition, there were six different viral genotypes (GT) known for HCV and the virus was shown to have a very high mutation rate because of the poor proof-reading function of the viral polymerase. These other factors complicated the outlook and increased the complexity of the task because the most desirable cure therapy would have the characteristics of being pangenotypic, with a high barrier to the formation of resistant virus and IFN-free.

2.2 Tools to Enable Drug Discovery

With the genome of the HCV virus fully delineated, the field was now ready to attempt to identify agents that could impact the virus directly with direct-acting antivirals (DAAs) and see if DAAs could deliver on some of the objectives. Early progress was slow because drug hunters were using surrogate systems such as bovine virus diarrhea virus (BVDV) as a way to evaluate potential drug candidates. Therefore, a new tool needed to be developed, a cell-based assay that drug discovery scientist could use to screen for inhibitors targeting the nonstructural proteins essential for the HCV virus to replicate itself. This was eventually achieved with the development of the HCV replicon cell system pioneered by Bartenschlager and coworkers [7]. By eliminating the gene sequence for the structural proteins, this noninfectious cell system allowed for screening of inhibitors of viral replication against the druggable nonstructural proteins. It also provided a convenient way to

assess drug resistance. Eventually, the JFH-1 cell line was developed and provided a way to assess drug candidates in a real infectious system [7].

Another challenge that hampered the discovery of curative therapies was the availability of an accessible and cost-effective animal model. Because of the narrow host tropism, the chimpanzee was the only animal model available for testing new therapeutic agents, but the cost, availability, and ethical concerns limited its use and therefore limited the impact on the discovery and development of new drugs. Although other animal models were eventually developed, their impact was not significant in the scheme of HCV drug discovery.

2.3 Drug Discovery Targets

HCV is a positive-sense single-stranded RNA virus whose genome is approximately 9.6 kb in length. The genome encodes over 3000 amino acids which after polyprotein processing produces three structural proteins and seven nonstructural proteins. The mapping of the HCV virus genome by Houghton and coworkers provided an initial list of viral targets against which drug discovery efforts could be launched (Figure 2.1) [8]. Several of the seven viral nonstructural proteins were considered druggable targets. They included the NS3/4A protease and NS5B viral polymerase around which most of the early drug discovery work focused. Subsequently, efforts targeting the NS4B and NS5A proteins began to emerge with significant focus on agents that bind to NS5A.

In addition to directly targeting the HCV nonstructural proteins, a number of efforts were initiated to target host-related targets that had effects on viral replication or on host immune responses. These targets included cyclophilins, MiR122, and interferon- γ [9–11]. Although targeting these nonviral targets did produce clinical agents that led to reductions in HCV viral load in early clinical trials, it was the work on the DAAs targeting NS3/4A protease, NS5A and NS5B polymerase that ultimately, in various combinations, produced the highly efficacious, safe, and short duration curative interferon-free therapies. The very rapid and successful development of DAA combination therapies made it difficult for host targeting agents to be competitive, especially since they would also have had to be combined with a DAA and could not stand alone.

2.4 NS5B Polymerase Inhibitors

2.4.1 Nucleoside and Nucleotide Inhibitors of NS5B Polymerase

The HCV NS5B RNA-dependent RNA polymerase (RdRp) is responsible for two RNA polymerization steps that are necessary for replicating the viral genome. The RdRp uses the HCV genomic RNA as a template from which the complementary negative RNA strand intermediate is generated and then this negative RNA strand becomes a template for the synthesis of a positive RNA strand [8]. The HCV NS5B polymerase is well conserved across all GTs and contains characteristics common among known viral RdRps [12]. It maintains the characteristic finger, palm, and thumb domains and contains the Asp, Gly, and Asp catalytic triad at the active site with the requirement of needing two divalent metal ions to initiate polymerization (Figure 2.2) [15].



Figure 2.2 HCV NS5B RdRp crystal structure with palm, finger, and thumb domains designated. PDB code 1C2P (1.9Å) [13, 14].

To target the HCV NS5B polymerase, two distinct approaches were pursued. These included the investigation of nucleosides and nucleotides as alternate substrates and the pursuit of non-nucleoside small-molecule allosteric modulators that bind to either the palm or thumb domains of NS5B [16].

The development of nucleos(t) ide inhibitors largely revolved around $2'-\alpha$ -fluoro-2'-Cmethyl and $2' - \alpha$ -hydroxy-2' - C-methyl substitution on the furanose ring system of the nucleos(t)ide (Figures 2.3 and 2.4, Tables 2.1 and 2.2) [16]. The 2'-methyl substitution was shown to be important for anti-HCV activity. Particularly in combination with the $2'-\alpha$ -F substitution, 2'-C-methyl substitution induced a level of selectivity for HCV versus other viruses and human polymerases. Also, it was demonstrated that the 2'- α -F substitution provided a profound benefit as it related to specificity for HCV and safety profile. Some of the early 2'-hydroxyl nucleosides having an adenosine or cytosine base, MK-0608 and NM-283 (valopicitabine), suffered from adverse safety observations either in preclinical testing or in human clinical studies resulting in termination of their further development (Figure 2.3 and Table 2.1) [17-19, 48-50]. This is in contrast to the early fluorinated cytosine nucleosides PSI-6130 and its prodrug RG7128 which were not hampered by safety concerns but were ultimately not taken forward in development because of efficacy and other development challenges (Figure 2.3 and Table 2.1). However, it was the study with RG7128 that first demonstrated clinical efficacy in GT 1, 2 and 3 HCV patients, thus establishing the possibility for development of a pangenotypic DAA HCV cure strategy [51].

Resistance was not a significant issue with the 2'-methyl-2'-F or 2'-methyl-2'-OH nucleosides. The S282T amino acid substitution that conferred resistance to these



Figure 2.3 Nucleoside HCV NS5B inhibitors.

nucleosides was shown not to be a pre-existing variant and was also shown to be quite unfit [20, 52–54].

It was the discovery of the liver-targeted nucleotide prodrug PSI-7851 and its single isomer PSI-7977 (sofosbuvir, SOF) that had a profound impact on the development of nucleos(t)ide drugs for HCV and ultimately on the development of IFN-free HCV cure regimens (Figure 2.4 and Table 2.2) [25, 55, 56]. Sofosbuvir was a 2'- α -F-2'-*C*-methyluridine 5'-phosphoramidate prodrug that leveraged liver first-pass metabolism to remove the promoiety and deliver the 2'- α -F-2'-*C*-methyluridine monophosphate to the liver. This prodrug strategy produced high concentrations of the active drug in the liver with virtually no systemic exposure. It demonstrated a high barrier to resistance and was shown to be active against all viral GTs.

Fast follow-on approaches based on sofosbuvir produced several uridine nucleotide prodrug clinical agents. These included IDX21427 (uprifosbuvir) and AL-335. Each of these agents demonstrated proof-of-concept in the clinic but were terminated because of a lack of competitive clinical efficacy (Figure 2.4 and Table 2.2).

Attempts to leverage the liver-targeted phosphoramidate prodrug strategy employing purine bases in place of the uridine base of sofosbuvir either with the 2'- α -fluoro-2'-C-methyl or 2'- α -hydroxy-2'-C-methyl furanose substitution proved unsuccessful. The drug INX-08189 was ultimately terminated because of fatal cardiovascular toxicity in the clinic



Figure 2.4 Nucleotide HCV NS5B polymerase inhibitors.

and PSI-352938 was terminated because of liver toxicity (Figure 2.4 and Table 2.2) [28, 33, 34, 57, 58]. Development of IDX-184 was ultimately discontinued because of lack of competitive efficacy [36–38, 59].

Drug	Replicon potency GT1	HCV RNA reduction Log ₁₀ IU/mL	Dosage	Dosing period	GT	Status
NM283 [17–19] Valopicitabine	$EC_{50}=1.23\mu M$	1.2	800 mg b.i.d	14 days	1	Discont'd PhII GI tox
PSI-6130 [20-22]	$EC_{90}=4.5~\mu M$	ND	ND	ND	ND	Discont'd PhI Poor PK
RG7128 [18, 19, 22] Mericitabine	$EC_{90} = ~4.5 \ \mu M$	2.7	1500 mg b.i.d.	14 days	1	Discont'd PhIII, Efficacy
R1626 [19, 23, 24] Balapiravir	$EC_{50} = 1.28 \ \mu M$	3.7	4500 mg b.i.d.	14 days	1	Discont'd PhII Hematological tox

 Table 2.1
 Nucleoside HCV NS5B polymerase inhibitors.

ND, not determined.

Replicon potency, phase I monotherapy data with development status.

Over a 12-year period, a number of other nucleos(t)ides (see Tables 2.1 and 2.2) were taken into development without success. These failures were attributed to either safety, efficacy, or other development issues such as lack of a competitive market profile.

In a 2011 phase II clinical study, sofosbuvir in combination with RBV was shown to deliver a 100% cure rate, sustained virological response (SVR), after only 12 weeks of oncea-day oral therapy [60]. This groundbreaking result led to the further development of SOF. In a phase III clinical study, SOF plus RBV delivered greater than a 93% cure rate in GT2 and 3. For GT1, 4, 5, or 6 patients, SOF plus RBV was not sufficient to deliver high cure rates; therefore, the triple combination of SOF plus PEG-INF/RBV for 12 weeks gave 89% SVR in GT1 patients and 97% SVR in GT4, 5, and 6 patients [61, 62]. Sofosbuvir was approved by the US FDA on 6 December 2013 as the first IFN-free HCV curative regimen for GT2 and 3 patients and to this day stands as the only nucleos(t)ide HCV cure agent approved for clinical use.

2.4.2 Non-nucleoside NS5B Inhibitors

The identification of non-nucleoside NS5B inhibitors was fueled by high-throughput screening and the use of structure-based drug design tools. The availability of HCV NS5B polymerase protein co-crystal structures with small molecule inhibitors allowed for identification of the allosteric binding sites and subsequent optimization of inhibitor binding. Ultimately, four allosteric binding sites were identified for non-nucleoside inhibitors.

		Phase I monotherapy				
Drug	Replicon potency GT1	HCV RNA reduction Log ₁₀ IU/mL	Dosage	Dosing period	GT	Status
PSI-7851 [25, 26]	$EC_{90} = 0.42 \ \mu M$	1.95	400 mg q.d.	3 days	1	Discont'd Moved to single isomer PSI-7977
PSI-7977 [25, 27] GS-7977 Sofosbuvir	$EC_{90} = 0.42 \ \mu M$	4.7	400 mg q.d.	14 days	1	Approved Sovaldi [®] Combo: Harvoni [®] , Epclusa [®] , Vosevi [®]
PSI-352938 [28-32]	$EC_{90} = 1.37 \ \mu M$	4.65	200 mg q.d.	7 days	1	Discont'd PhI Liver tox
INX-08189 [33–35]	$EC_{50} = 0.01 \ \mu M$	1.31	25 mg q.d.	3 days	1	Discont'd PhII Cardiac tox
IDX184 [36–39]	$EC_{50}=0.4~\mu M$	0.74	100 mg q.d.	3 days	1	Discont'd PhII Efficacy
GS-6620 [40-42]	$EC_{50} = 0.36 \ \mu M$	1.73	900 mg b.i.d	5 days	1	Discont'd PhI PK/PD Variability
IDX21437 [43–45] MK-3682 Uprifosbuvir	$EC_{50} = 0.32 \ \mu M$	4.2	300 mg q.d.	7 days	1	Discont'd PhIII Efficacy
AL-335 [46, 47] JNJ-64146212	$EC_{50} = 0.075 \ \mu M$	2.76	400 mg q.d.	7 days	1	Discont'd PhII Efficacy

Table 2.2 Nucleotide inhibitors of HCV NS5B polymerase.

Replicon potency, phase I data, and development status.

These sites are located in the palm or thumb domains of the NS5B polymerase [16]. No allosteric inhibitors have been identified that bind to the finger domain; however, several inhibitors with undisclosed binding sites have been reported [16].

The development of non-nucleoside HCV NS5B inhibitors faced a number of challenges, most notably narrow genotype coverage and rapid development of drug-resistant variants. Yet, there was significant activity in the exploration of non-nucleoside inhibitors because of the ease with which they could be identified by biochemical or phenotypic screening approaches (Figures 2.5–2.8 and Tables 2.3–2.6). The first non-nucleoside to enter the clinic and demonstrate clinical proof-of-concept was HCV-796 (nesbuvir) (Figure 2.8 and

Table 2.6) [106–108]. It demonstrated a 1.2–1.5 \log_{10} IU/ml drop in HCV RNA levels; however, development was halted because of liver toxicity.

Ultimately, two approved drugs emerged from the non-nucleoside class of inhibitors. The first was beclabuvir (Figure 2.5 and Table 2.3), which was approved only in Japan as part of the fixed-dose combination (Ximency[®]) that included the NS3 protease inhibitor asunaprevir and the NS5A inhibitor daclatasvir [113, 114]. The other was dasabuvir (Exviera[®]) (Figure 2.7 and Table 2.5), which was part of a complex multi-tablet combination therapy (Viekira Pak[®]) that included the NS5A inhibitor ombitasivir, the NS3/4 protease inhibitor paritaprevir, and ritonavir [115]. These agents were only approved for the treatment of GT1 patient populations, and ultimately, neither played a significant role in



Figure 2.5 Non-nucleoside HCV NS5B polymerase thumb domain inhibitors.



Figure 2.6 Non-nucleoside HCV NS5B polymerase thumb domain inhibitors.

widely adopted curative combination therapies nor were their market penetration significant. In general, because of their limited genotype coverage and high occurrence of resistance, the non-nucleoside class of inhibitors could not compete with the nucleotide sofosbuvir as the NS5B polymerase component of curative combination regimens and ultimately faded from use.

2.5 HCV NS3/4A Protease Inhibitors

The discovery and development of small-molecule HCV NS3/4A protease inhibitors was aggressively pursued with 10 drugs receiving regulatory approval either as single agents to be used in combination with IFN or as part of DAA combination regimens. The NS3/4A protease is a member of the chymotrypsin family of serine proteases. It is formed by



Figure 2.7 Non-nucleoside HCV NS5B polymerase palm domain inhibitors.

proteolytic cleavage of the HCV genome-derived ~3000 amino acid viral polypeptide by the HCV NS2/3 cysteine protease. It plays a critical role in viral polyprotein processing to generate the nonstructural proteins NS3, NS4A, NS4B, NS5A, and NS5B. The NS3 protein is bifunctional having a serine protease domain and an NTPase/helicase domain. It associates noncovalently with the NS4A cofactor forming the NS3/4A protease [8]. The crystal structure of the NS3/4A protease has been solved including structures having truncated NS3 domains (Figure 2.9). The substrate binding groove is extremely shallow, elongated, and solvent exposed with significant hydrophobic character making inhibitor design and optimization challenging [116].



Figure 2.8 Non-nucleoside HCV NS5B polymerase palm domain inhibitors.

		Pha				
Drug	Replicon potency GT1	HCV RNA reduction Log ₁₀ IU/mL	Dosage	Dosing period	GТ	Status
BILB 1941 [63, 64]	$EC_{50} = 84 \text{ nM}$	>1	450 mg t.i.d.	5 days	1	Discont'd PhII GI & Liver Tox
BI-207127 [65, 66] Deleobuvir	$EC_{50} = 11-23 \text{ nM}$	3.8	600 mg t.i.d.	5 days	1	Discont'd PhIII Efficacy
BMS-791325 [67–69] Beclabuvir	$EC_{50} = 3-6 \text{ nM}$	2.5	300 mg q.d.	24 h	1	Approved Combo Regimen Japan Ximency [®]
TMC647055 [70, 71]	$EC_{50} = 166 \text{ nM}$	2.4-3.3	1000 mg b.i.d.	6 days	1	Discont'd PhI
MK-3281 [72, 73]	$EC_{50} = 38 \text{ nM}$	1.95	800 mg b.i.d.	7 days	1	Discont'd PhI

 Table 2.3
 Non-nucleoside HCV NS5B polymerase thumb domain inhibitors.

Replicon potency, phase I data, and development status.

Development of early generation HCV NS3/4A protease inhibitors was limited by several factors. Their genotype coverage was restricted to GT1, and they suffered from the rapid emergence of clinical resistance. Through tenacious medicinal chemistry optimization, subsequent generation protease inhibitors were able to achieve broader genotype coverage and an increased barrier to resistance.

		Phase I monotherapy				
Drug	Replicon potency GT1	HCV RNA reduction Log ₁₀ IU/mL	Dosage	Dosing period	GT	Status
PF-00868554 [74–77] Filibuvir	EC ₅₀ = 75 nM	2.13	300 mg t.i.d.	8 days	1	Discont'd PhII
HCV-371 [78]	$EC_{50}=4.8~\mu M$	Not significant	_	_	1	Discont'd PhI Efficacy
VX-759 [79-81]	$EC_{50} = 300 \text{ nM}$	2.5	800 mg t.i.d.	10 days	1	Discont'd
VX-222 [82, 83] Lomibuvir	$EC_{50} = 5.9 \text{ nM}$	3.5	750 mg b.i.d.	3 days	1	Discont'd PhII
GS-9669 [84–86] Radalbuvir	$EC_{50} = 11 \text{ nM}$	>3	500 mg q.d.	3 days	1	Discont'd PhII
GS-9190 [87–91] Tegobuvir	$EC_{50} = 0.7 \text{ nM}$	1.61–1.95	40–120 mg b.i.d.	8 days	1	Discont'd PhII

 Table 2.4
 Non-nucleoside NS5B thumb domain inhibitors.

Replicon potency, phase I data, and development status.

2.5.1 HCV NS3/4A Protease Acyclic Reversible Inhibitors

The discovery of HCV NS3/4A protease inhibitors employed both peptidomimetic approaches and co-crystal structure-guided approaches. Early work in substrate-based peptide optimization led to the identification of several key compounds (Figure 2.10) that spawned an entire class of acyclic reversible inhibitors shown in Figure 2.11 and Table 2.7. A number of these agents entered clinical development demonstrating proof-of-concept antiviral activity in HCV patients. The only approved drug that emerged from this acyclic peptidomimetic class was asunaprevir (Figure 2.11 and Table 2.7). Asunaprevir was eventually launched in Japan as part of a combination regimen for a narrow GT1b patient population [114]. However, this class paved the way for next-generation inhibitors.

2.5.2 HCV NS3/4A Protease Acyclic Covalent Binding Inhibitors

Simultaneously, with the development of reversible acyclic peptidomimetic inhibitors, a class of acyclic peptidic inhibitors that bound covalently to the active-site serine residue of HCV protease were being developed. This class is represented primarily by the approved agents telaprevir (VX-950) and boceprevir (SCH503034) (Figure 2.12 and Table 2.8). As part of their design, each employed a reactive ketoamide warhead that reacted with the active-site serine. These two agents were developed as combination regimens with PEG-IFN/RBV, the standard of care at the time and were the first DAAs approved for therapy. They

		Phase I monotherapy				
Drug	Replicon potency GT1	HCV RNA reduction Log ₁₀ IU/mL	Dosage	Dosing period	GT	Status
GSK625433 [92–94]	$EC_{50} = 3 \text{ nM}$	NA	NA	NA	1	Discont'd PhI
SB-750330 [95]	$EC_{50} = 2 nM$	NA	NA	NA	1	Discont'd PhI
IDX375 [96]	$EC_{50} = 18 \text{ nM}$	0.5 – 1.1	200 mg q.d.	2 days	1	Discont'd
ANA-598 [97, 98] Setrobuvir	$EC_{50} = 3-18 \text{ nM}$	2.9	800 mg b.i.d.	3 days	1	Discont'd PhII
ABT-333 [99–102] Dasabuvir	$EC_{50} = 1.8-7.7 \text{ nM}$	NA	300 – 1200 mg b.i.d.	2 days	1	Approved as combo Viekira Pak [®]
ABT-072 [103, 104]	$EC_{50} = 0.3-1.1 \text{ nM}$	NA	400 mg q.d.	NA	1	Discont'd
RG7109 [105]	$EC_{50} = 1.1 \text{ nM}$	NA	NA	NA	NA	Discont'd PhI

Table 2.5 Non-nucleoside HCV NS5B polymerase palm domain inhibitors.

NA, data not available.

Replicon potency, phase I data, and development status.

		Pł	_			
Drug	Replicon potency GT1	HCV RNA reduction Log ₁₀ IU/mL	Dosage	Dosing period	GT	Status
HCV-796 [106–108] Nesbuvir	$EC_{50} = 13 \text{ nM}$	1.2–1.5	50–1500 mg b.i.d.	14 days	1	Discont'd PhII Safety
BMS-929075 [94, 109]	$EC_{50} = 3-20 \text{ nM}$	NA	NA	NA	NA	Discont'd PhIa
MK-8876 [94, 110]	$EC_{50} = 1-7 \text{ nM}$	3.4	800 mg q.d.	7 days	1	Discont'd PhII
GSK2485852 [111, 112]	$EC_{50} = 1 - 8 \text{ nM}$	NA	NA	NA	NA	Discont'd PhIa

NA, data not available.

Replicon potency, phase I data, and development status.



Figure 2.9 Crystal structure of the protease domain of the HCV NS3/4A protease showing the shallow substrate binding groove in gray (PDB Code: 3RC6).



Figure 2.10 HCV NS3/4A protease inhibitor early leads that derived from substrate-based peptide optimization.



Figure 2.11 HCV NS3/4A protease acyclic reversible binding inhibitors.

displayed limited genotype coverage and thus were approved to treat only GT1 patients but were able to achieve a ~70–80% SVR (sustained virological response off therapy) cure in this patient population [134, 139]. However, because they required twice-daily or three times-a-day administration, exhibited significant side effects in addition to those presented with PEG-IFN/RBV co-therapy and demonstrated a low barrier to resistance with V36, T54, R155, and A156 clinically observed mutations, they presented significant limitations in the clinic [140].

		Pha				
Drug	Replicon potency GT1	HCV RNA reduction Log ₁₀ IU/mL	Dosage	Dosing period	GT	Status
BI 201335 [117–120] Faldaprevir	$EC_{50} = 3 \text{ nM}$	4.2	240 mg q.d.	14 days	1	Discont'd PhIII Efficacy
BMS-605339 [121]	$EC_{50} = 12 \text{ nM}$	1.8	120 mg q.d.	12 h	1	Discont'd Ph1 Cardio tox
BMS-650032 [122–124] Asunaprevir	$EC_{50} = 6 \text{ nM}$	2.7 - 3.5	200–600 mg b.i.d	3 days	1	Approved Combo Ximency [®]
ACH-1625 [125, 126] Sovaprevir	$EC_{50} = 11 \text{ nM}$	3.81	600 mg q.d.	5 days	1	Discont'd PhI Liver tox
GS-9451 [127–130] Vedroprevir	$EC_{50} = 2 \text{ nM}$	3.6	400 mg q.d.	3 days	1	Discont'd PhI

 Table 2.7
 HCV NS3/4 protease acyclic reversible inhibitors.

Replicon potency, phase I data, and development status.



Figure 2.12 HCV NS3/4A protease covalent binding inhibitors.

		Phase I monotherapy				
Drug	Replicon potency GT1	HCV RNA reduction Log ₁₀ IU/mL	Dosage	Dosing period	GT	Status
VX-950 [131–133] Telaprevir	$EC_{50} = 83-280 \text{ nM}$	4.4	750 mg t.i.d.	14 days	1	Approved Incivek [®]
SCH-503034 [134, 135] Boceprevir	EC ₅₀ = 220–233 nM	2.06	400 mg t.i.d.	7 days	1	Approved Victrelis [®]
SCH-900518 [136–138] Narlaprevir	$EC9_0 = 40 \text{ nM}$	4.5	800 mg t.i.d.	7 days	1	Discont'd PhII

 Table 2.8
 HCV NS3/4 protease covalent binding inhibitors.

Replicon potency, phase I data, and development status.

2.5.3 HCV NS3/4A Protease Macrocyclic Reversible Inhibitors

Further work on the acyclic peptidic inhibitors and particularly on inhibitor B (Figure 2.10) led to ciluprevir (BILN-2061, Figure 2.13 and Table 2.9), a cyclic inhibitor. Ciluprevir became the foundation for all subsequent work on cyclic reversible inhibitors of HCV NS3/4A protease (Figure 2.13 and Table 2.9). A number of these agents entered clinical development but only two, simeprevir (TMC435) and paratepravir (ABT-450) (Figure 2.13 and Table 2.9), were ultimately approved for clinical use either in combination with PEG-IFN/RBV or as part of a DAA combination regimen.

2.5.4 HCV NS3/4 Protease P2-P4 Macrocyclic Inhibitors

The other major class of macrocyclic HCV NS3/4A protease inhibitors were the P2–P4 macrocycles (Figure 2.14 and Table 2.10). This class was identified by structure-based approaches and the realization that there was potential to access additional interactions with the helicase domain of NS3/4A. Developments in this class of inhibitors led to the identification of molecules that expanded the genotype coverage beyond GT1 and increased the barrier to resistance. However, resistance concerns would still remain and as with other protease inhibitors, this class required combination with either PEG-IFN/RBV or other potent DAAs such as the nucleotide sofosbuvir or a potent NS5A inhibitor which could protect against resistance and enhance genotype coverage. The approved commercial combinations became vaniprevir + PEG-IFN/RBV (Vanihep[®], Japan only, GT1), grazoprevir + elbasvir (NS5A) (Zepatier[®], GT1 and 4), voxilaprevir + sofosbuvir (NS5B) + velpatasvir(NS5A) (Vosevi[®], pangenotypic), and glecaprevir + pibrentasvir (NS5A) (Mavyret[®], pangenotypic).



Figure 2.13 HCV NS3/4A protease macrocyclic reversible binding inhibitors.

		Р				
Drug	Replicon potency GT1	HCV RNA reduction Log ₁₀ IU/mL	Dosage	Dosing period	GT	Status
BILN-2061 [141] Ciluprevir	$EC_{50} = 1.2 \text{ nM}$	>2	200 mg b.i.d.	2 days	1	Discont'd PhI, Cardio tox
ITMN-191 [51, 142, 143] RG7227 Danoprevir	$EC_{50} = 1.8 \text{ nM}$	3.8	200 mg t.i.d.	14 days	1	Approved Ganovo [®] (China)
TMC435 [144, 145] Simperevir	$EC_{50} = 9.4 \text{ nM}$	3.91	150 mg q.d.	6 days	1	Approved Olysio [®]
IDX-320 [146, 147]	EC ₅₀ = 0.5–3.4 nM	3.3	400 mg q.d.	3 days	1	Discont'd PhII Liver tox
ABT-450 [148–150] Paritaprevir	EC ₅₀ = 0.21–1.0 nM	4.0	200 mg q.d.	3 days	1	Approved Viekira Pak [®] , Technivie [®]
ACH-2684 [151–153] Deldeprevir, Neceprevir	EC ₅₀ = 0.04–0.7 nM	3.73	400 mg q.d.	3 days	1	Discont'd PhII
GS-9256 [154–156]	$EC_{50} = 20 \text{ nM}$	2.8	450 mg b.i.d.	1 day	1	Discont'd PhII
PHX1766 [157]	$EC_{50} = 8 \text{ nM}$	1.5	800 mg b.i.d.	6 days	1	Discont'd PhI Efficacy

Table 2.9 HCV NS3/4 protease macrocyclic reversible	bindinc	j inhibitors.
---	---------	---------------

Replicon potency, phase I data, and development status.

2.6 HCV NS5A Inhibitors

The HCV NS5A nonstructural protein has been shown to play several roles in the virus life cycle, yet it has no known enzymatic activity (Figure 2.15). It was shown to be important in the process of viral genome replication and modulation of host cell factors that support viral persistence [8]. Small molecule agents that inhibit the function of NS5A were identified by phenotypic screening of large compound libraries and resistance mapping of early leads (Figure 2.16) [172]. Additional studies that included pull-down experiments, X-ray crystallography, and affinity labeling studies helped solidify the nature of the inhibitor mode of binding and complex stoichiometry of the inhibition [171, 173–175].



Figure 2.14 HCV NS3/4A protease P2–P4 macrocyclic reversible binding inhibitors.

Through intensive medicinal chemistry efforts, the early small molecule leads (Figure 2.16) were transformed into exceptionally potent inhibitors of viral replication (Figures 2.17 and 2.18, Tables 2.11 and 2.12). Ultimately, this work led to the approval of five agents that became integral parts of marketed curative combination regimens. The pivotal clinical study that established the excitement around this mechanism of action and prompted a frenzy of development efforts by many groups occurred with daclatasvir (Figure 2.17 and Table 2.11) [176, 177]. In a phase I study, daclatasvir delivered a 1.8, 3.2, and 3.3 log₁₀ IU/mL drop in serum HCV RNA at 24 hours which persisted to 144 hours for

		P	_			
Drug	Replicon potency GT1	HCV RNA reduction Log ₁₀ IU/mL	Dosage	Dosing period	GT	Status
MK-7009 [158–161] Vaniprevir	$EC_{50} = 4.5 \text{ nM}$	1.8–4.6	600 mg q.d. 700 mg b.i.d.	8 days	1	Approved Japan As combo Vanihep [®]
MK-5172 [162–165] Grazoprevir	$EC_{50} = 7.4 \text{ nM}$	3	30–800 mg q.d.	7 days	1	Approved As combo Elbasvir [®]
GS-9857 [166–168] Voxilaprevir	EC ₅₀ = 1.5–6.6 nM	>3	100 mg q.d.	3 days	1	Approved As combo Vosevi [®]
ABT-493 [169, 170] Glecaprevir	EC ₅₀ = 0.85–0.94 nM	3.8-4.36	100–700 mg q.d.	3 days	1	Approved As combo Mavyret [®]

 Table 2.10
 HCV NS3/4 protease P2-P4 macrocyclic reversible binding inhibitors.

Replicon potency, phase I data, and development status.



Figure 2.15 Crystal structure of HCV NS5A with potential binding residues highlighted. PDB Code 1ZH1 [171]. With permission of John Wiley & Sons.

the 100 mg dose. This exceptional efficacy was attributed to the dual effect of the drug on viral replication and virion assembly [199].

Even with exceptional clinical efficacy, first-generation NS5A inhibitors were plagued with rapid onset of resistance and limited genotype coverage. For daclatasvir, when administered as monotherapy for 14 days at doses from 1 to 100 mg q.d., most subjects experienced viral breakthrough with major mutations observed at M28T/A/V, Q30H/R/K/E, L31M/V, and Y93H/N/C in GT1a infected patients and L31M/V and Y93H/C in GT1b



Figure 2.16 Early NS5A protein inhibitor screening leads.

patients [172, 200, 201]. Most NS5A inhibitors suffered from these same resistance issues; however, a select group of later generation inhibitors such as pibrentasvir and valpatasvir were able to overcome many of the resistance and genotype coverage problems (Figure 2.18 and Table 2.12). NS5A inhibitors became integral components of many DAA HCV cure combinations that include Harvoni[®], Epclusa[®], Vosevi[®], Viekira Pak[®], and Mavyret[®] providing 95–100% SVR cure rates in only 8–12 weeks of therapy.

2.7 The Evolution of DAA Combination Therapies

With the objective to achieve higher cure rates, early attempts to study drug combinations focused on the addition of DAAs to the standard of care PEG-IFN/RBV. Early phase clinical studies demonstrated that these combinations delivered deeper viral load reductions, faster kinetics of viral load declines, and a higher percentage of patients that reached a rapid virological response (RVR), i.e., undetectable HBV RNA levels after four weeks on therapy, than PEG-IFN/RBV alone. Triple combinations that included either protease inhibitors, nucleos(t)ide inhibitors, or non-nucleoside inhibitors with PEG-IFN/RBV were all shown to deliver additive effects clinically [18, 202]. In 2010, combinations of a NS3/4 protease inhibitor, telaprevir (Incivek®) or boceprevir (Victrelis®), with either PEG-IFN/RBV or PEG-INTRON were approved by the US FDA and European EMEA as the first DAA-containing regimens to cure HCV [202-204]. They delivered on an improved cure rate (~72%) and in some cases a shortened duration of therapy (24-28 weeks) for treatment-naïve patients. However, they became difficult regimens to administer because of the added adverse events these protease inhibitors introduced to the regimens including exacerbation of anemia and adverse cutaneous effects with telaprevir coupled with the potential for the emergence of resistance. In addition, these first DAA-containing combinations were only approved for treating GT1 patients, thus requiring patient genotyping. It remained an aspirational goal to eliminate the use of subcutaneously delivered PEG-IFN/RBV and its associated serious adverse side effects. Ultimately, with the emergence of IFN-free DAA regimens, the clinicaluse lifespan of both telaprevir and boceprevir was only three years.



Figure 2.17 First-generation HCV NS5A inhibitors.



Figure 2.18 Second-generation NS5A inhibitors.

One pivotal study that was first to investigate the possibility of eliminating PEG-IFN/ RBV was called INFORM-1 [51]. This small investigational study attempted to study whether a nucleoside NS5B inhibitor, RG7128 (mericitabine, Figure 2.3), and a protease inhibitor, RG7227 (danoprevir, Figure 2.13), combination would produce a clinical response

		Phase I monotherapy				
Drug	Replicon potency GT1	HCV RNA reduction Log ₁₀ IU/mL	Dosage	Dosing period	GT	Status
BMS-790052 [176–178] Daclatasvir	$EC_{50} = 0.009 - 0.05$ nM	3.6 - 4.5	100 mg q.d.	14 days	1	Approved As combo Daklinza [®]
GS-5885 [179, 180] Ledipasvir	EC ₅₀ = 0.004– 0.031 nM	3.2 - 3.3	10 mg q.d.	3 days	1	Approved As combo Harvoni [®]
MK-8742 [181–183] Elbasvir	EC ₅₀ = 0.003– 0.004 nM	4.2 - 5.1	50 mg q.d.	5 days	1	Approved As combo Zepatier [®]
GSK2336805 [184, 185]	$EC_{50} = 0.008 - 0.06$ nM	2.96	60 mg q.d.	1 day	1	Discont'd PhII
ABT-267 [102, 186, 187] Ombitasvir	EC ₅₀ = 0.005– 0.135 nM	2.9	5, 50 mg q.d.	3 days	1	Approved As combo Viekira Pak [®]
PPI-668 [188–190] Ravidasvir	EC ₅₀ = 0.016– 0.014 nM	3.5	100 mg q.d.	3 days	1	Unknown Developed in China

Table 2.11	First-generation HCV NS5A inhibito	rs

Replicon potency, phase I data, and development status.

similar to that seen when each of these individual agents were combined with PEG-IFN/ RBV. In a 14-day study in treatment-naïve GT1-infected patients, a median change in HCV RNA from baseline of $-5.1 \log_{10}$ IU/ml was observed, thus setting the stage for future all DAA combination studies.

It was not until the results reported from two key Phase II studies that the field realized PEG-IFN/RBV-containing regimens were going to be a thing of the past. The first study examined the combination of daclatasvir (NS5A inhibitor) 60 mg q.d. and asunaprevir (NS3/4 protease inhibitor) 600 mg b.i.d. in GT1 null responder patients for 24 weeks [205]. Of 11 patients that received the dual DAA combination, 5 reached undetectable HCV RNA at end of treatment and 4 of them achieved SVR at weeks 12 and 24 post-treatment period. Of those that did not reach SVR, all six were GT1a subjects who experienced viral breakthrough and exhibited resistant mutations in both the NS3 (R155K, D168K/E/T/V/Y) and NS5A (Q30R, L31M/V, Y93C/N) proteins. The second watershed study was referred to as ELECTRON and involved a multi-arm study evaluating sofosbuvir (NS5B nucleotide inhibitor) 400 mg q.d. in combination with RBV alone or in combination with PEG-IFN/RBV for 12 weeks of therapy in GT2/3 subjects [206]. All combination arms of the study including the IFN-free sofosbuvir/RBV arm resulted in

		Phase I monotherapy				
Drug	Replicon potency GT1	HCV RNA reduction Log ₁₀ IU/mL	Dosage	Dosing period	GT	Status
GS-5816 [191, 192] velpatasvir	EC ₅₀ = 0.015- 0.031 nM	3.6	100 mg q.d.	3 days	1	Approved As combo Epclusa [®] & Vosevi [®]
ABT-530 [169, 193, 194] pibrentasvir	EC ₅₀ = 0.002– 0.004 nM	4.1	40 mg q.d.	3 days	1	Approved As combo Mavyret [®]
IDX-719 [195, 196] samatasvir	EC ₅₀ = 0.003– 0.008 nM	3.3-4.3	25–100 mg q.d.	3 days	1	Discont'd PhII
ACH-3102 [172, 197, 198] odalasvir	EC ₅₀ = 0.005– 0.26 nM	3.5-4.0	25–300 mg q.d.	3 days	1	Discont'd PhII

 Table 2.12
 Second-generation HCV NS5A inhibitors.

Replicon potency, phase I data, and development status.

SVR12/24 of 100% with no observed viral breakthroughs. The era of IFN-containing HCV cure regimens had come to an end.

Because of its high barrier to resistance and broad genotype coverage, sofosbuvir became the backbone agent of choice for many DAA combinations and was married in clinical studies with numerous NS5A inhibitors including daclatasvir, ledipasvir, and velpatasvir as well as protease inhibitor simeprevir [207–211]. However, other non-sofosbuvir-containing DAA combination regimens also emerged combining a NS5A inhibitor, a NS3/4 protease inhibitor with or without a non-nucleoside NS5B inhibitor. In some cases, adding RBV to the combination appeared to be beneficial.

The first IFN-free combination to be approved by the US FDA (6 December 2013) and subsequently by the European EMEA (January 2014) was 12 or 24 weeks of once-a-day sofosbuvir and RBV for the treatment of GT2, 3 patients. The sofosbuvir/RBV combination resulted in a >90% SVR rate in Phase III clinical trials with minimal side effects mostly associated with RBV [61, 62]. Subsequently, the fixed-dose combination of sofosbuvir and ledipasvir (NS5A, Figure 2.17) (Harvoni[®]) with or without RBV was approved in October 2014 for all GT1 patient populations after demonstrating SVR12 rates of >97% in Phase III clinical studies [212, 213]. In addition, this Phase III study showed that SVR12 rates were similar irrespective of whether the patient was non-cirrhotic or cirrhotic. Also, in 2014, it was reported that sofosbuvir was combined with the NS3/4 protease inhibitor simeprevir (Figure 2.13) with or without RBV for 12 weeks in GT1 patients [211, 214]. Here again a >97% SVR12 was observed; however, SVR12 rates for patients with the protease Q80K polymorphism were somewhat lower.

Although the marriage of sofosbuvir and the NS5A inhibitor daclatasvir was the first Phase II study to evaluate SVR rates with a nucleotide/NS5A combination demonstrating a 98% SVR24 in GT1 patients, 92% in GT2, and 89% in GT3 patients, it was not until 2015 that daclatasvir plus sofosbuvir received regulatory approval [207]. This combination became the first DAA regimen to be approved for treating GT3 patients without the use of RBV [215]. Subsequently, approval of this regimen for treating GT1 patients was achieved.

Further development of the sofosbuvir backbone franchise led to the development of the fixed-dose combinations Epclusa[®] and Vosevi[®]. Epclusa[®], the combination of sofosbuvir (NS5B nucleotide) with the pangenotypic NS5A inhibitor velpatasvir (Figure 2.18, Table 2.12), provided the first true pangenotypic high barrier to resistance q.d. 12-week therapy. This regimen showed SVR12 rates of 99% in GT1 and 2 patients and 98% in hard-to-treat GT3-naïve patients [191, 216, 217]. The triple combination regimen, Vosevi[®], containing sofosbuvir (NS5B nucleotide), velpatasvir (NS5A) and the NS3/4 protease inhibitor voxilaprevir, was evaluated in an 8-week q.d. regimen and delivered SVR12 of 95% and 96% in GT1 and GT3 patients, respectively [218–220]. Vosevi[®] has been used primarily to treat patients who have failed other HCV cure regimens.

Combination regimens lacking the nucleotide sofosbuvir backbone were also shown to be effective at delivering high cure rates. The first of these was a dual combination that included daclatasvir (NS5A inhibitor) and asunaprevir (NS3/4 protease inhibitor). This combination was studied and approved in 2014 only in Japan because of the prevalence of GT1b patients in the Japanese population and the limited genotype coverage of this combination. The Phase III clinical trial evaluated 24 weeks of dual therapy and achieved and SVR12 of 81% in IFN nonresponders and 87% in those intolerant of or ineligible for IFN use [124]. It was also shown that the T93H polymorphism in NS5A led to a reduction in clinical efficacy. Subsequently, addition of a third agent, beclabuvir (NS5B non-nucleoside inhibitor) to create a b.i.d. fixed-dose 12-week triple therapy, Ximency[®], was able to capture GT1a patients in the Japanese market and deliver a 93% SVR12 in naïve patients [114].

The three-drug combination paritaprevir (NS3/4 protease inhibitor) /r (ritonavir boosting), ombitasvir (NS5A inhibitor), and dasabuvir (NS5B non-nucleoside inhibitor) branded as Viekira Pak[®] or Technivie[®] was a multi-pill regimen that delivered a 96% SVR12 after 12 weeks of therapy in GT1 patients [115, 221]. In 2016, the DAA combination of elbasvir (NS5A inhibitor) and grazoprevir (NS3/4 protease inhibitor) was launched as Zepatier[®] [222]. In a Phase III study, this combination delivered an overall SVR rate of 95% against a GT1, 4, and 6 patient population but suffered from lower SVR12 rates in patients with NS5A resistance-associated substitutions (RASs).

The first true pan-genotypic regimen, Mavyret[®], that lacked the use of sofosbuvir combined the NS5A inhibitor pibrentasvir (Figure 2.18 and Table 2.12) and the NS3/4 protease inhibitor glecaprevir (Figure 2.14 and Table 2.10) [221]. This DAA combination delivered a 99% SVR12 in GT1 patients with either an 8- or 12-week course of therapy. In addition, SVR12 rates in GT2-7 non-cirrhotic patients ranged from 93 to 98% for an 8-week course of therapy.

Further, Phase III and post-approval studies with a number of these DAA combination regimens expanded the patient population that could benefit from these curative therapies. It became possible to now effectively cure difficult-to-treat patient populations such as cirrhotics, HIV and HBV coinfected patients, difficult-to-treat ethnicities, liver transplant patients, and pediatric patient populations [223–232].

With short duration therapies that delivered reliably high cure rates with minimal side effects, it became possible to consider the idea of transplanting organs from HCV-infected patients, thus expanding the pool of donors. Thus livers, kidneys, and hearts from HCV-infected donors can now be transplanted and the recipient successfully treated with a short course of curative therapy [229, 233, 234].

As the HCV cure field matured, the dominant therapies were characterized by q.d. dosing, pangenotypic coverage, high barrier to resistance, and fixed-dose regimens. That resulted in Harvoni[®], Epclusa[®], and Mavyret[®] being the dominant first-line cure therapies. Each of these regimens utilized a NS5A inhibitor that complements other agents in the cocktail. For Harvoni[®] and Epclusa[®], the nucleotide sofosbuvir provided the backbone with the high barrier to resistance and pangenotypic coverage necessary to protect for NS5A resistance-associated variants (RAVs). By using both a potent pangenotypic NS5A inhibitor and a potent pangenotypic NS3/4A protease inhibitor, cross protection for RAVs is accomplished in Mavyret[®].

2.8 Conclusion

In a span of 25 years, the virus that causes a debilitating and deadly liver disease was identified and oral, safe, effective, and short duration curative therapies were delivered to patients. This achievement is the first example of a chronic viral disease being cured and a story that is unparalleled in modern medicine. Today, millions of patients now live normal productive lives without HCV thanks to these new drugs. However, this accomplishment was not easy and required intense commitment from researchers and the availability of significant resources. Analysis showed that the success rate in HCV drug development was only 2%, less than half of the industry average for successful drug development, thus putting into perspective the magnitude of the challenge and the achievement [235]. It was because of this achievement that the World Health Organization (WHO) set the goal of eliminating HCV by the year 2030 [1]. Many countries around the world have established HCV elimination programs and are beginning to make measurable progress against the WHO goal.

Acknowledgment

The author would like to thank Dr. Kristi Fan of Arbutus Biopharma for generating Figures 2.2, 2.9, and 2.15.

References

- 1 WHO (2017). Global Hepatitis Report 2017. France: World Health Organization.
- **2** Alter, H.J. (2019). From B to non-B to C: the hepatitis C virus in historical perspective. In: *HCV: The Journey from Discovery to a Cure*, vol. I (ed. M.J. Sofia), 4–15. Switzerland: Springer Nature.

- 42 2 Curing Hepatitis C with Direct-Acting Antiviral Therapy
 - **3** Choo, Q.L., Kuo, G., Weiner, A.J. et al. (1989). Isolation of a cDNA clone derived from a blood-borne non-A, non-B viral hepatitis genome. *Science* 244 (4902): 359–362.
 - **4** Houghton, M. (2009). The long and winding road leading to the identification of the hepatitis C virus. *Journal of Hepatology* 51 (5): 939–948.
 - **5** Alter, H.J., Purcell, R.H., Shih, J.W. et al. (1989). Detection of antibody to hepatitis C virus in prospectively followed transfusion recipients with acute and chronic non-A, non-B hepatitis. *The New England Journal of Medicine* 321 (22): 1494–1500.
 - **6** Alqahtani, S.A. and Sulkowski, M. (2019). The role of interferon for the treatment of chronic hepatitis C virus infection. In: *HCV: The Journey from Discovery to a Cure*, vol. 1 (ed. M.J. Sofia), 98–113. Switzerland: Springer Nature.
 - 7 Lohmann, V. and Bartenschlager, R. (2014). On the history of hepatitis C virus cell culture systems. *Journal of Medicinal Chemistry* 57 (5): 1627–1642.
 - **8** Saeed, M., Billerbeck, E., and Rice, C.M. (2019). HCV Molecular Virology and Animal Models. In: *HCV: The Journey from Discovery to a Cure*, vol. 1 (ed. M.J. Sofia), 30–68. Switzerland: Springer Nature.
 - **9** Frese, M., Schwarzle, V., Barth, K. et al. (2002). Interferon-gamma inhibits replication of subgenomic and genomic hepatitis C virus RNAs. *Hepatology* 35 (3): 694–703.
 - Sendi, H., Mehrab-Mohseni, M., Foureau, D.M. et al. (2015). MiR-122 decreases HCV entry into hepatocytes through binding to the 3' UTR of OCLN mRNA. *Liver International* 35 (4): 1315–1323.
 - 11 Stanciu, C., Trifan, A., Muzica, C., and Sfarti, C. (2019). Efficacy and safety of alisporivir for the treatment of hepatitis C infection. *Expert Opinion on Pharmacotherapy* 20 (4): 379–384.
 - Simmonds, P. (2004). Genetic diversity and evolution of hepatitis C virus –15 years on. *The Journal of General Virology* 85 (Pt 11): 3173–3188.
 - 13 Biswal, B.K., Cherney, M.M., Wang, M. et al. (2005). Crystal structures of the RNAdependent RNA polymerase genotype 2a of hepatitis C virus reveal two conformations and suggest mechanisms of inhibition by non-nucleoside inhibitors. *The Journal of Biological Chemistry* 280 (18): 18202–18210.
 - **14** Lesburg, C.A., Cable, M.B., Ferrari, E. et al. (1999). Crystal structure of the RNAdependent RNA polymerase from hepatitis C virus reveals a fully encircled active site. *Nature Structural Biology* 6 (10): 937–943.
 - 15 Lohmann, V., Korner, F., Herian, U., and Bartenschlager, R. (1997). Biochemical properties of hepatitis C virus NS5B RNA-dependent RNA polymerase and identification of amino acid sequence motifs essential for enzymatic activity. *Journal of Virology* 71 (11): 8416–8428.
 - **16** Sofia, M.J., Chang, W., Furman, P.A. et al. (2012). Nucleoside, nucleotide and nonnucleside inhibitors of hepatitis C virus NS5B RNA-dependent RNA polymerase. *Journal of Medicinal Chemistry* 55: 2481–2531.
 - Pierra, C., Amador, A., Benzaria, S. et al. (2006). Synthesis and pharmacokinetics of valopicitabine (NM283), an efficient prodrug of the potent anti-HCV agent 2'-C-methylcytidine. *Journal of Medicinal Chemistry* 49 (22): 6614–6620.
 - **18** Brown, N.A. (2009). Progress towards improving antiviral therapy for hepatitis C with hepatitis C virus polymerase inhibitors. Part I: Nucleoside analogues. *Expert Opinion on Investigational Drugs* 18 (6): 709–725.

- **19** Burton, J.R. (2009). HCV NS5B polymerase inhibitors. *Clinics in Liver Disease* 13: 12.
- 20 Stuyver, L.J., McBrayer, T.R., Tharnish, P.M. et al. (2006). Inhibition of hepatitis C replicon RNA synthesis by beta-D-2'-deoxy-2'-fluoro-2'-C-methylcytidine: a specific inhibitor of hepatitis C virus replication. *Antiviral Chemistry & Chemotherapy* 17 (2): 79–87.
- **21** Clark, J.L., Hollecker, L., Mason, J.C. et al. (2005). Design, synthesis, and antiviral activity of 2'-deoxy-2'-fluoro-2'-C-methyl-cytidine, a potent inhibitor of hepatitis C virus replication. *Journal of Medicinal Chemistry* 48 (17): 5504–5508.
- Sofia, M.J., Furman, P.A., and Symonds, W.T. (2010). 2'-F-2'-C-methyl nucleosides and nucleotides for the treatment of hepatitis C virus: from discovery to the clinic. In: *RSC Drug Discovery Ser. Accounts in Drug Discovery*, vol. 4 (eds. J.C. Barrish, P.H. Carter, P.T.W. Cheng and R. Zahler), 238–266. UK: Royal Society of Chemistry.
- **23** Toniutto, P., Fabris, C., Bitetto, D. et al. (2008). R-1626, a specific oral NS5B polymerase inhibitor of hepatitis C virus. *IDrugs* 11 (10): 738–749.
- 24 Zeuzem, S., Nelson, D., Andreone, P. et al. (2010). Phase IIB study of balapiravir (RG1626; nucleoside analogue inhibitor of HCV polymerase) plus peginterferon Alfa-2A (40KD) and ribavirin for chronic hepatitis C genotype 1: final results. In 45th Annual Meeting of the European Association for the Study of the Liver, Vienna, Austria.
- **25** Sofia, M.J., Bao, D., Chang, W. et al. (2010). Discovery of a beta-*D*-2'-deoxy-2'-alpha-fluoro-2'-b-*C*-methyluridine nucleotide prodrug (PSI-7977) for the treatment of hepatitis C virus. *Journal of Medicinal Chemistry* 53 (19): 7202–7218.
- 26 Lawitz, E., Rodriguez-Torres, M., Denning, J. et al. (2013). Pharmacokinetics, pharmacodynamics, and tolerability of GS-7851 a nucleotide analog polymerase inhibitor, following multiple ascending doses in patients with chronic hepatits C infection. *Antimicrobial Agents and Chemotherapy* 57 (3): 1209–1217.
- **27** Lawitz, E., Lalezari, J., Rodriguez-Torres, M. et al. (2010). High rapid virologic response (RVR) with PSI-7977 daily dosing plus PEG-INF/RBV in a 28-day phase 2a trial. In *61st Annual Meeting of the American Association of the Study of Liver Diseases*, Boston, MA.
- 28 Lam, A.M., Espiritu, C., Murakami, E. et al. (2011). Inhibition of hepatitis C virus replicon RNA synthesis by PSI-352938, a cyclic phosphate prodrug of {beta}-D-2'-deoxy-2'-{alpha}fluoro-2'-{beta}-C-methylguanosine. *Antimicrobial Agents and Chemotherapy* 55: 2566–2575.
- 29 Du, J., Bao, D., Chun, B.K. et al. (2012). Beta-D-2'-alpha-F-2'-beta-C-methyl-6-O-substituted 3',5'-cyclic phosphate nucleotide prodrugs as inhibitors of hepatitis C virus replication: a structure-activity relationship study. *Bioorganic & Medicinal Chemistry Letters* 22 (18): 5924–5929.
- **30** Lam, A.M., Espiritu, C., Bansal, S. et al. (2011). HCV nucleotide inhibitors PSI-352938 and PSI-353661 exhibit a novel mechanism of resistance requiring multiple mutations within replicon RNA. *Journal of Virology* 85 (23): 12334–12342.
- 31 Lawitz, E., Rodriguez-Torres, M., Denning, J.M. et al. (2011). Once daily dual-nucleotide combination of PSI-938 and PSI-7977 provides 94% HCV RNA < LOD at day 14: first purine/pyrimidine clinical combination data (The Nuclear Study). In 46th Annual Meeting of the European Association for the Study of the Liver, Berlin, Germany.</p>
- **32** PRNewswire (2010). Pharmasset reports positive preliminary antiviral data with PSI-938 for the treatment of hepatitis C. *COMTEX News Network*.

- 44 2 Curing Hepatitis C with Direct-Acting Antiviral Therapy
 - McGuigan, C., Madela, K., Aljarah, M. et al. (2010). Design, synthesis and evaluation of a novel double prodrug: INX-08189. A new clinical candidate for hepatitis C virus. *Bioorganic & Medicinal Chemistry Letters* 20 (16): 4850–4854.
 - 34 Vernachio, J.H., Bleiman, B., Bryant, K.D. et al. (2011). INX-08189, a phosphoramidate prodrug of 6-O-methyl-2'-C-methyl guanosine, is a potent inhibitor of hepatitis C virus replication with excellent pharmacokinetic and pharmacodynamic properties. *Antimicrobial Agents and Chemotherapy* 55 (5): 1843–1851.
 - **35** Patti, J.M., Matson, M., Goehlecke, B. et al. (2011). A study of the safety and pharmacokinetics of single ascending oral doses of INX-08189, a nucleotide polymerase inhibitor, in healthy subjects. In *46th Annual Meeting of the European Association for the Study of the Liver*, Berlin, Germany.
 - **36** Gentile, I., Buonomo, A.R., Zappulo, E., and Borgia, G. (2015). Discontinued drugs in 2012–2013: hepatitis C virus infection. *Expert Opinion on Investigational Drugs* 24 (2): 239–251.
 - Zhou, X.J., Pietropaolo, K., Chen, J. et al. (2011). Safety and pharmacokinetics of IDX184, a liver-targeted nucleotide polymerase inhibitor of hepatitis C virus, in healthy subjects. *Antimicrobial Agents and Chemotherapy* 55 (1): 76–81.
 - **38** Lalezari, J., Asmuth, D., Casiro, A. et al. (2009). Antiviral activity, safety, and pharmacokinetics of IDX184, a liver-targeted nucleotide HCV polymerase inhibitor, in patients with chronic hepatitis C. In *60th Annual Meeting of the American Association for the Study of Liver Diseases*, Boston, MA.
 - **39** Lalezari, J., Asmuth, D., Casiro, A. et al. (2012). Short-term monotherapy with IDX184, a liver-targeted nucleotide polymerase inhibitor, in patients with chronic hepatitis C virus infection. *Antimicrobial Agents and Chemotherapy* 56 (12): 6372–6378.
 - **40** Cho, A., Zhang, L., Xu, J. et al. (2014). Discovery of the first C-nucleoside HCV polymerase inhibitor (GS-6620) with demonstrated antiviral response in HCV infected patients. *Journal of Medicinal Chemistry* 57 (5): 1812–1825.
 - **41** Feng, J.Y., Cheng, G., Perry, J. et al. (2014). Inhibition of hepatitis C virus replication by GS-6620, a potent C-nucleoside monophosphate prodrug. *Antimicrobial Agents and Chemotherapy* 58 (4): 1930–1942.
 - **42** Lawitz, E., Hill, J., Marbury, T. et al. (2012). GS-6620, a liver-targeted nucleotide prodrug, exhibits antiviral activity and favorable safety profile over 5 days in treatment naive chronic HCV genotype 1 subjects. In *47th Annual Meeting of the European Association for the Study of the Liver*, Barcelona, Spain.
 - **43** Alexandre, F.R., Badaroux, E., Bilello, J.P. et al. (2017). The discovery of IDX21437: design, synthesis and antiviral evaluation of 2'-alpha-chloro-2'-beta-C-methyl branched uridine pronucleotides as potent liver-targeted HCV polymerase inhibitors. *Bioorganic & Medicinal Chemistry Letters* 27 (18): 4323–4330.
 - **44** Lawitz, E., Gane, E., Feld, J.J. et al. (2019). Efficacy and safety of a two-drug direct-acting antiviral agent regimen ruzasvir 180 mg and uprifosbuvir 450 mg for 12 weeks in adults with chronic hepatitis C virus genotype 1, 2, 3, 4, 5 or 6. *Journal of Viral Hepatitis* 26 (9): 1127–1138.
 - **45** Lawitz, E., Poordad, F., Anderson, L.J. et al. (2019). Efficacy and safety of ruzasvir 60 mg and uprifosbuvir 450 mg for 12 weeks in adults with chronic hepatitis C virus genotype 1, 2, 3, 4 or 6 infection. *Journal of Viral Hepatitis* 26 (6): 675–684.

- **46** Wang, G., Dyatkina, N., Prhavc, M. et al. (2019). Synthesis and anti-HCV activities of 4'-fluoro-2'-substituted uridine triphosphates and nucleotide prodrugs: discovery of 4'-fluoro-2'-C-methyluridine 5'-phosphoramidate prodrug (AL-335) for the treatment of hepatitis C infection. *Journal of Medicinal Chemistry* 62 (9): 4555–4570.
- **47** McClure, M.W., Berliba, E., Tsertsvadze, T. et al. (2018). Safety, tolerability, and pharmacokinetics of AL-335 in healthy volunteers and hepatitis C virus-infected subjects. *PLoS One* 13 (10): e0204974.
- **48** Carroll, S.S., Ludmerer, S., Handt, L. et al. (2009). Robust antiviral efficacy upon administration of a nucleoside analog to hepatitis C virus-infected chimpanzees. *Antimicrobial Agents and Chemotherapy* 53 (3): 926–934.
- **49** Olsen, D.B., Davies, M.E., Handt, L. et al. (2011). Sustained viral response in a hepatitis C virus-infected chimpanzee via a combination of direct-acting antiviral agents. *Antimicrobial Agents and Chemotherapy* 55 (2): 937–939.
- **50** Pierra, C., Benzaria, S., Amador, A. et al. (2005). NM 283, an efficient prodrug of the potent anti-HCV agent 2'-C-methylcytidine. *Nucleosides, Nucleotides & Nucleic Acids* 24 (5–7): 767–770.
- 51 Gane, E.J., Roberts, S.K., Stedman, C.A. et al. (2010). Oral combination therapy with a nucleoside polymerase inhibitor (RG7128) and danoprevir for chronic hepatitis C genotype 1 infection (INFORM-1): a randomised, double-blind, placebo-controlled, dose-escalation trial. *Lancet* 376 (9751): 1467–1475.
- 52 Le Pogam, S., Jiang, W.R., Leveque, V. et al. (2006). In vitro selected Con1 subgenomic replicons resistant to 2'-C-methyl-cytidine or to R1479 show lack of cross resistance. *Virology* 351 (2): 349–359.
- **53** Ludmerer, S.W., Graham, D.J., Boots, E. et al. (2005). Replication fitness and NS5B drug sensitivity of diverse hepatitis C virus isolates characterized by using a transient replication assay. *Antimicrobial Agents and Chemotherapy* 49 (5): 2059–2069.
- 54 Le Pogam, S., Yan, J.M., Kosaka, A. et al. (2010). No evidence of drug resistance or baseline S282T resistance mutation among GT1 and GT4 HCV infected patients on nucleoside polymerase inhibitor RG7128 and PEG-INF/RBV combination treatment for up to 12 weeks: interim analysis from the propel study. In *61st Annual Meeting of the American Association for the Study of Liver Diseases*, Boston, MA.
- **55** Lam, A.M., Murakami, E., Espiritu, C. et al. (2010). PSI-7851, a pronucleotide of beta-*D*-2'deoxy-2'-fluoro-2'-*C*-methyluridine monophosphate, is a potent and pan-genotype inhibitor of hepatitis C virus replication. *Antimicrobial Agents and Chemotherapy* 54 (8): 3187–3196.
- 56 Lam, A.M., Espiritu, C., Bansal, S. et al. (2012). Genotype and subtype profiling of PSI-7977 as a nucleotide inhibitor of hepatitis C virus. *Antimicrobial Agents and Chemotherapy* 56 (6): 3359–3368.
- 57 Reddy, P.G., Bao, D., Chang, W. et al. (2010). 2'-Deoxy-2'-alpha-fluoro-2'-beta-C-methyl 3',5'-cyclic phosphate nucleotide prodrug analogs as inhibitors of HCV NS5B polymerase: discovery of PSI-352938. *Bioorganic & Medicinal Chemistry Letters* 20 (24): 7376–7380.
- **58** Ahmad, T., Yin, P., Saffitz, J. et al. (2015). Cardiac dysfunction associated with a nucleotide polymerase inhibitor for treatment of hepatitis C. *Hepatology* 62 (2): 409–416.
- **59** Lalezari, J., O'Riordan, W., Poordad, F. et al. (2010). A phase IIa study of IDX184 in combination with pegylated interferon (pegIFN) and ribavirin (RBV) in treatment-naive HCV genotype 1-infected subjects. In *61st Annual Meeting of the American Association for the Study of Liver Diseases*, Boston, MA.

- 46 2 Curing Hepatitis C with Direct-Acting Antiviral Therapy
 - **60** Gane, E.J., Stedman, C.A., Hyland, R. et al. (2012). Once daily sofosbuvir (GS-7977) plus ribavirin in patients with HCV genotypes 1,2, and 3: the electron trial. In *63rd Annual Meeting of the American Association for the Study of Liver Diseases*, Boston, MA.
 - **61** Jacobson, I.M., Gordon, S.C., Kowdley, K.V. et al. (2013). Sofosbuvir for hepatitis C genotype 2 or 3 in patients without treatment options. *New England Journal of Medicine* 368 (20): 1867–1877.
 - **62** Lawitz, E., Mangia, A., Wyles, D. et al. (2013). Sofosbuvir for previously untreated chronic hepatitis C infection. *New England Journal of Medicine* 368 (20): 1878–1887.
 - **63** Brochu, C. (2011). Discovery of BILB 1941, a non-nucleoside polymerase inhibitor for the treatment of hepatitis C virus infection. In *242nd National Meeting of the American Chemical Society*, Denver, CO.
 - **64** Erhardt, A., Deterding, K., Benhamou, Y. et al. (2009). Safety, pharmacokinetics and antiviral effect of BILB 1941, a novel hepatitis C virus RNA polymerase inhibitor, after 5 days oral treatment. *Antiviral Therapy* 14 (1): 23–32.
 - 65 LaPlante, S.R., Bos, M., Brochu, C. et al. (2014). Conformation-based restrictions and scaffold replacements in the design of hepatitis C virus polymerase inhibitors: discovery of deleobuvir (BI 207127). *Journal of Medicinal Chemistry* 57 (5): 1845–1854.
 - **66** Larrey, D., Lohse, A.W., Trepo, C. et al. (2013). Antiviral effect, safety, and pharmacokinetics of five-day oral administration of Deleobuvir (BI 207127), an investigational hepatitis C virus RNA polymerase inhibitor, in patients with chronic hepatitis C. *Antimicrobial Agents and Chemotherapy* 57 (10): 4727–4735.
 - **67** Gentile, I., Zappulo, E., Buonomo, A.R. et al. (2015). Beclabuvir for the treatment of hepatitis C. *Expert Opinion on Investigational Drugs* 24 (8): 1111–1121.
 - **68** Tatum, H., Thuluvath, P.J., Lawitz, E. et al. (2015). A randomized, placebo-controlled study of the NS5B inhibitor beclabuvir with peginterferon/ribavirin for HCV genotype 1. *Journal of Viral Hepatitis* 22 (8): 658–664.
 - 69 Gentles, R.G. (2019). Discovery of beclabuvir: a potent allosteric inhibitor of the hepatitis C virus polymerase. In: *HCV: The Journey from Discovery to a Cure*, vol. 1 (ed. M.J. Sofia), 193–228. Switzerland: Springer Nature.
 - **70** Cummings, M.D., Lin, T.I., Hu, L. et al. (2014). Discovery and early development of TMC647055, a non-nucleoside inhibitor of the hepatitis C virus NS5B polymerase. *Journal of Medicinal Chemistry* 57 (5): 1880–1892.
 - **71** Devogelaere, B., Berke, J.M., Vijgen, L. et al. (2012). TMC647055, a potent nonnucleoside hepatitis C virus NS5B polymerase inhibitor with cross-genotypic coverage. *Antimicrobial Agents and Chemotherapy* 56 (9): 4676–4684.
 - 72 Narjes, F., Crescenzi, B., Ferrara, M. et al. (2011). Discovery of (7R)-14-cyclohexyl-7-{[2-(dimethylamino)ethyl](methyl) amino}-7,8-dihydro-6H-indolo[1,2-e]
 [1,5]benzoxazocine-11-carboxylic acid (MK-3281), a potent and orally bioavailable finger-loop inhibitor of the hepatitis C virus NS5B polymerase. *Journal of Medicinal Chemistry* 54: 289–301.
 - 73 Brainard, D.M., Anderson, M.S., Petry, A.S. et al. (2009). Safety and antiviral activity of NS5B polymerase inhibitor MK-3281in genotype 1 and 3 HCV-infected patients In 60th Annual Meeting of the American Association for the Study of Liver Diseases, Boston, MA.
 - 74 Li, H., Tatlock, J., Linton, A. et al. (2009). Discovery of (R)-6-cyclopentyl-6-(2-(2,6-diethylpyridin-4-yl)ethyl)-3-((5,7-dimethyl-[1,2,4]triazolo[1,5-a]pyrimidin-2-yl)

methyl)-4-hydroxy-5,6-dihydropyran-2-one (PF-00868554) as a potent and orally available hepatitis C virus polymerase inhibitor. *Journal of Medicinal Chemistry* 52: 1255–1258.

- 75 Shi, S.T., Herlihy, K.J., Graham, J.P. et al. (2009). Preclinical characterization of PF-00868554, a potent nonnucleoside inhibitor of the hepatitis C virus RNA-dependent RNA polymerase. *Antimicrobial Agents and Chemotherapy* 53 (6): 2544–2552.
- 76 Hammond, J.L., Rosario, M.C., Wagner, F. et al. (2008). Antiviral activity of the HCV polymerase inhibitor PF-00868554 administered as monotherapy in HCV genotype 1 infected subjects. In 59th Annual Meeting of the American Association for the Study of Liver Diseases, San Francisco, CA.
- 77 Jacobson, I., Pockros, P.J., Lalezari, J. et al. (2009). Antiviral activity of filibuvir in combination with pegylated interferon alfa-2a and ribavirin for 28 days in treatment-naive patients chronically infected with HCV genotype 1 In 44th Meeting of the European Association for the Study of Liver Diseases, Copenhagen, Denmark.
- 78 Howe, A.Y., Bloom, J., Baldick, C.J. et al. (2004). Novel nonnucleoside inhibitor of hepatitis C virus RNA-dependent RNA polymerase. *Antimicrobial Agents and Chemotherapy* 48 (12): 4813–4821.
- **79** Le Pogam, S., Kang, H., Harris, S.F. et al. (2006). Selection and characterization of replicon variants dually resistant to thumb- and palm-binding nonnucleoside polymerase inhibitors of the hepatitis C virus. *Journal of Virology* 80 (12): 6146–6154.
- **80** Cooper, C., Lawitz, E.J., Ghali, P. et al. (2009). Evaluation of VCH-759 monotherapy in hepatitis C infection. *Journal of Hepatology* 51 (1): 39–46.
- **81** Cooper, C., Lawitz, E.J., Ghali, P. et al. (2007). Antiviral activity of the non-nucleoside polymerase inhibitor, VCH-759, in chronic Hepatitis C patients:Results from a randomized, double-blind, placebo -controlled, ascending multiple dose study. In *58th Annual Meeting of the American Association for the Study of Liver Diseases* Boston, MA.
- Rodriguez-Torres, M., Lawitz, E., Conway, B. et al. (2010). Safety and antiviral activity of the HCV non-nucleoside polymerase inhibitor VX-222 in treatment-naive genotype 1 HCV-infected patients. In 45th Meeting of the European Association for the Study of Liver Diseases, Vienna, Austria.
- **83** Di Bisceglie, A.M., Sulkowski, M., Gane, E. et al. (2014). VX-222, a non-nucleoside NS5B polymerase inhibitor, in telaprevir-based regimens for genotype 1 hepatitis C virus infection. *European Journal of Gastroenterology and Hepatology* 7 (7): 761–773.
- **84** Lazerwith, S.E., Lew, W., Zhang, J. et al. (2014). Discovery of GS-9669, a thumb site II non-nucleoside inhibitor of NS5B for the treatment of genotype 1 chronic hepatitis C infection. *Journal of Medicinal Chemistry* 57 (5): 1893–1901.
- **85** Fenaux, M., Eng, S., Leavitt, S.A. et al. (2013). Preclinical characterization of GS-9669, a thumb site II inhibitor of the hepatitis C virus NS5B polymerase. *Antimicrobial Agents and Chemotherapy* 57 (2): 804–810.
- **86** Borgia, G., Maraolo, A.E., Nappa, S. et al. (2018). NS5B polymerase inhibitors in phase II clinical trials for HCV infection. *Expert Opinion on Investigational Drugs* 27 (3): 243–250.
- **87** Bavisotto, L., Wang, C., Jacobson, I. et al. (2007). Antiviral, pharmacokinetic and safety data for GS-9190, a non-nucleoside HCV NS5B polymerase inhibitor, in a phase-1 trial in HCV genotype 1 infected patients. In *58th Annual Meeting of the American Association for the Study of Liver Diseases*, Boston, MA.

- **48** 2 Curing Hepatitis C with Direct-Acting Antiviral Therapy
 - 88 Harris, J., Bae, A., Sun, S. et al. (2010). Antiviral response and resistance analysis of treatment-naïve HCV infected subjects receiving single and multiple doses of GS-9190. In 61th Annual Meeting of the American Association for the Study of Liver Diseases Boston, MA, Hynes Convention Center, Boston, MA.
 - **89** Mo, H., Hedskog, C., Svarovskaia, E. et al. (2016). Antiviral response and resistance analysis of treatment-naive HCV-infected patients receiving single and multiple doses of GS-9190. *Journal of Viral Hepatitis* 23 (8): 644–651.
 - **90** Shih, I.H., Vliegen, I., Peng, B. et al. (2011). Mechanistic characterization of GS-9190 (Tegobuvir), a novel nonnucleoside inhibitor of hepatitis C virus NS5B polymerase. *Antimicrobial Agents and Chemotherapy* 55 (9): 4196–4203.
 - **91** Wong, K.A., Xu, S., Martin, R. et al. (2012). Tegobuvir (GS-9190) potency against HCV chimeric replicons derived from consensus NS5B sequences from genotypes 2b, 3a, 4a, 5a, and 6a. *Virology* 429 (1): 57–62.
 - **92** Burton, G., Ku, T.W., Carr, T.J. et al. (2007). Studies on acyl pyrrolidine inhibitors of HCV RNA-dependent RNA polymerase to identify a molecule with replicon antiviral activity. *Bioorganic & Medicinal Chemistry Letters* 17: 1930–1933.
 - **93** Slater, M.J., Amphlett, E.M., Andrews, D.M. et al. (2007). Optimization of novel acyl pyrrolidine inhibitors of hepatitis C virus RNA-dependent RNA polymerase leading to a development candidate. *Journal of Medicinal Chemistry* 50: 897–900.
 - **94** Watkins, W.J. (2019). Evolution of HCV NS5B non-nucleoside Inhibitors. In: *HCV: The Journey from Discovery to a Cure*, vol. 1 (ed. M.J. Sofia), 171–192. Switzerland: Springer Nature.
 - 95 Shaw, A.N., Tedesco, R., Bambal, R. et al. (2009). Substituted benzothiadizine inhibitors of Hepatitis C virus polymerase. *Bioorganic & Medicinal Chemistry Letters* 19: 4350–4353.
 - **96** de Bruijne, J., van de Wetering de Rooij, J., van Vliet, A.A. et al. (2012). First-in-human study of the pharmacokinetics and antiviral activity of IDX375, a novel nonnucleoside hepatitis C virus polymerase inhibitor. *Antimicrobial Agents and Chemotherapy* 56 (8): 4525–4528.
 - **97** Mallalieu, N.L., Rahimy, M.H., Crowley, C.A. et al. (2014). Pharmacokinetics and pharmacodynamics of setrobuvir, an orally administered hepatitis C virus non-nucleoside analogue inhibitor. *Clinical Therapeutics* 36 (12): 2047–2063 e3.
 - 98 Kirkovsky, L., Zhou, Y., Norris, D. et al. (2007). ANA598, a novel non-nucleoside inhibitor of HCV NS5B polymerase, exhibits favorable pharmacokinetic properties in multiple preclinical species. In 58th Annual Meeting of the American Association for the Study of Liver Diseases, Boston, MA.
 - **99** Maring, C.W.R., Hutchinson, D., Flentge, C. et al. (2009). Preclinical potency, pharmacokinetic and ADME characterization of ABT-333, a novel non-nucleoside HCV polymerase inhibitor. In *44th Annual Meeting of the European Association for the Study of the Liver*, Copenhagen, Denmark.
 - 100 Gaultier, I.A., Cohen, D.E., Dumas, E.O. et al. (2011). 12-Week efficacy and saftey of ABT-072 or ABT-333 with pegylated interferon + ribavirin, following 3-day monotherapy in genotype 1 HCv-infected treatment-naive subjects. In *The 21st Conference of the Asian Pacific Association for the Study of Liver*, Bangkok, Thailand.
- 101 El Kassas, M., Elbaz, T., Hafez, E. et al. (2017). Discovery and preclinical development of dasabuvir for the treatment of hepatitis C infection. *Expert Opinion on Drug Discovery* 12 (6): 635–642.
- **102** Trivella, J.P., Gutierrez, J., and Martin, P. (2015). Dasabuvir: a new direct antiviral agent for the treatment of hepatitis C. *Expert Opinion on Pharmacotherapy* 16 (4): 617–624.
- 103 Randolph, J.T., Krueger, A.C., Donner, P.L. et al. (2018). Synthesis and biological characterization of aryl uracil inhibitors of hepatitis C virus NS5B polymerase: discovery of ABT-072, a trans-stilbene analog with good oral bioavailability. *Journal of Medicinal Chemistry* 61 (3): 1153–1163.
- **104** De Clercq, E. (2014). Current race in the development of DAAs (direct-acting antivirals) against HCV. *Biochemical Pharmacology* 89 (4): 441–452.
- 105 Talamas, F.X., Abbot, S.C., Anand, S. et al. (2014). Discovery of N-[4-[6-tert-butyl-5-methoxy-8-(6-methoxy-2-oxo-1H-pyridin-3-yl)-3-quinolyl]phenyl]methanesulfonamide (RG7109), a potent inhibitor of the hepatitis C virus NS5B polymerase. *Journal of Medicinal Chemistry* 57 (5): 1914–1931.
- 106 Howe, A.Y., Cheng, H., Johann, S. et al. (2008). Molecular mechanism of hepatitis C virus replicon variants with reduced susceptibility to a benzofuran inhibitor, HCV-796. Antimicrobial Agents and Chemotherapy 52 (9): 3327–3338.
- **107** Li, H. and Shi, S.T. (2010). Non-nucleoside inhibitors of hepatitis C virus polymerase: current progress and future challenges. *Future Medicinal Chemistry* 2 (1): 121–141.
- 108 Hang, J.Q., Yang, Y., Harris, S.F. et al. (2009). Slow binding inhibition and mechanism of resistance of non-nucleoside polymerase inhibitors of hepatitis C virus. *The Journal of Biological Chemistry* 284 (23): 15517–15529.
- **109** Yeung, K.S., Beno, B.R., Parcella, K. et al. (2017). Discovery of a hepatitis C virus NS5B replicase palm site allosteric inhibitor (BMS-929075) advanced to phase 1 clinical studies. *Journal of Medicinal Chemistry* 60 (10): 4369–4385.
- **110** McComas, C.C., Palani, A., Chang, W. et al. (2017). Development of a new structural class of broadly acting HCV non-nucleoside inhibitors leading to the discovery of MK-8876. *ChemMedChem* 12 (17): 1436–1448.
- 111 Voitenleitner, C., Crosby, R., Walker, J. et al. (2013). In vitro characterization of GSK2485852, a novel hepatitis C virus polymerase inhibitor. *Antimicrobial Agents and Chemotherapy* 57 (11): 5216–5224.
- **112** Wilfret, D.A., Walker, J., Voitenleitner, C. et al. (2014). A randomized, double blind, dose escalation, first time in human study to assess the safety, tolerability, pharmacokinetics, and antiviral activity of single doses of GSK2485852 in chronically infected hepatitis C subjects. *Clinical Pharmacology in Drug Development* 3 (6): 439–448.
- **113** Iwata, H. and Ishikawa, H. (2017). Pharmacological properties and clinical efficacy of Ximency((R)) combination tablets. *Nihon Yakurigaku Zasshi* 150 (3): 153–164.
- **114** Zappulo, E., Scotto, R., Buonomo, A.R. et al. (2020). Efficacy and safety of a fixed dose combination tablet of asunaprevir + beclabuvir + daclatasvir for the treatment of Hepatitis C. *Expert Opinion on Pharmacotherapy* 21 (3): 261–273.
- **115** Deeks, E.D. (2015). Ombitasvir/paritaprevir/ritonavir plus dasabuvir: a review in chronic HCV genotype 1 infection. *Drugs* 75 (9): 1027–1038.

- **50** *2* Curing Hepatitis C with Direct-Acting Antiviral Therapy
 - **116** Romano, K.P., Laine, J.M., Deveau, L.M. et al. (2011). Molecular mechanisms of viral and host cell substrate recognition by hepatitis C virus NS3/4A protease. *Journal of Virology* 85 (13): 6106–6116.
 - **117** Llinas-Brunet, M., Bailey, M.D., Goudreau, N. et al. (2010). Discovery of a potent and selective noncovalent linear inhibitor of the hepatitis C virus NS3 protease (BI 201335). *Journal of Medicinal Chemistry* 53 (17): 6466–6476.
 - **118** White, P.W., Llinas-Brunet, M., Amad, M. et al. (2010). Preclinical characterization of BI 201335, a C-terminal carboxylic acid inhibitor of the hepatitis C virus NS3-NS4A protease. *Antimicrobial Agents and Chemotherapy* 54 (11): 4611–4618.
 - **119** Agarwal, K. and Barnabas, A. (2015). Faldaprevir for the treatment of genotype-1 hepatitis C virus. *Expert Review of Gastroenterology & Hepatology* 9 (3): 277–288.
 - **120** Zeuzem, S., Soriano, V., Asselah, T. et al. (2013). Faldaprevir and deleobuvir for HCV genotype 1 infection. *New England Journal of Medicine* 369 (7): 630–639.
 - 121 Scola, P.M., Wang, A.X., Good, A.C. et al. (2014). Discovery and early clinical evaluation of BMS-605339, a potent and orally efficacious tripeptidic acylsulfonamide NS3 protease inhibitor for the treatment of hepatitis C virus infection. *Journal of Medicinal Chemistry* 57 (5): 1708–1729.
 - **122** McPhee, F., Sheaffer, A.K., Friborg, J. et al. (2012). Preclinical profile and characterization of the hepatitis C virus NS3 protease inhibitor asunaprevir (BMS-650032). *Antimicrobial Agents and Chemotherapy* 56 (10): 5387–5396.
 - **123** Pasquinelli, C., McPhee, F., Eley, T. et al. (2012). Single- and multiple-ascending-dose studies of the NS3 protease inhibitor asunaprevir in subjects with or without chronic hepatitis C. *Antimicrobial Agents and Chemotherapy* 56 (4): 1838–1844.
 - 124 Suzuki, Y., Ikeda, K., Suzuki, F. et al. (2013). Dual oral therapy with daclatasvir and asunaprevir for patients with HCV genotype 1b infection and limited treatment options. *Journal of Hepatology* 58 (4): 655–662.
 - 125 Agarwal, A., Zhang, B., Olek, E. et al. (2012). Rapid and sharp decline in HCV upon monotherapy with NS3 protease inhibitor, ACH-1625. *Antiviral Therapy* 17 (8): 1533–1539.
 - 126 Liverton, N. (2019). Evolution of NS3/4A protease inhibitors. In: HCV: The Journey from Discovery to a Cure, vol. 1 (ed. M.J. Sofia), 231–260. Switzerland: Springer Nature.
 - 127 Sheng, X.C., Appleby, T., Butler, T. et al. (2012). Discovery of GS-9451: an acid inhibitor of the hepatitis C virus NS3/4A protease. *Bioorganic & Medicinal Chemistry Letters* 22 (7): 2629–2634.
 - **128** Yang, H., Robinson, M., Corsa, A.C. et al. (2014). Preclinical characterization of the novel hepatitis C virus NS3 protease inhibitor GS-9451. *Antimicrobial Agents and Chemotherapy* 58 (2): 647–653.
 - **129** Abutaleb, A. and Kottilil, S. (2017). Vedroprevir in the management of hepatitis C virus infection. *Expert Opinion on Investigational Drugs* 26 (12): 1399–1402.
 - 130 Wyles, D.L., Rodriguez-Torres, M., Lawitz, E. et al. (2014). All-oral combination of ledipasvir, vedroprevir, tegobuvir, and ribavirin in treatment-naive patients with genotype 1 HCV infection. *Hepatology* 60 (1): 56–64.
 - 131 Lin, C., Kwong, A.D., and Perni, R.B. (2006). Discovery and development of VX-950, a novel, covalent, and reversible inhibitor of hepatitis C virus NS3.4A serine protease. Infectious Disorders Drug Targets 6 (1): 3–16.

- **132** Lin, K., Perni, R.B., Kwong, A.D., and Lin, C. (2006). VX-950, a novel hepatitis C virus (HCV) NS3-4A protease inhibitor, exhibits potent antiviral activities in HCv replicon cells. *Antimicrobial Agents and Chemotherapy* 50 (5): 1813–1822.
- **133** Reesink, H.W., Zeuzem, S., Weegink, C.J. et al. (2006). Rapid decline of viral RNA in hepatitis C patients treated with VX-950: a phase Ib, placebo-controlled, randomized study. *Gastroenterology* 131 (4): 997–1002.
- 134 Venkatraman, S. (2019). Discovery of boceprevir, a ketoamide-derived HCV NS3 protease inhibitor, for treatment of genotype 1 infections. In: *HCV: The Journey from Discovery to a Cure*, vol. 1 (ed. M.J. Sofia), 293–316. Switzerland: Springer Nature.
- 135 Kwo, P.Y. (2012). Boceprevir: a novel nonstructural 3 (NS3) protease inhibitor for the treatment of chronic hepatitis C infection. *Therapeutic Advances in Gastroenterology* 5 (3): 179–188.
- 136 Arasappan, A., Bennett, F., Bogen, S.L. et al. (2010). Discovery of narlaprevir (SCH 900518): a potent, second generation HCV NS3 serine protease inhibitor. ACS Medicinal Chemistry Letters 1 (2): 64–69.
- 137 de Bruijne, J., Bergmann, J.F., Reesink, H.W. et al. (2010). Antiviral activity of narlaprevir combined with ritonavir and pegylated interferon in chronic hepatitis C patients. *Hepatology* 52 (5): 1590–1599.
- **138** Tong, X., Arasappan, A., Bennett, F. et al. (2010). Preclinical characterization of the antiviral activity of SCH 900518 (narlaprevir), a novel mechanism-based inhibitor of hepatitis C virus NS3 protease. *Antimicrobial Agents and Chemotherapy* 54 (6): 2365–2370.
- 139 Kwong, A.D., Perni, R.B., and Graham, C.S. (2019). Development and marketing of INCIVEK (Telaprevir; VX-950): a first-generation HCV protease inhibitor, in combination with PEGylated interferon and ribavirin. In: *HCV: The Journey from Discovery to a Cure*, vol. 1 (ed. M.J. Sofia), 261–292. Switzerland: Springer Nature.
- **140** Rong, L., Dahari, H., Ribeiro, R.M., and Perelson, A.S. (2010). Rapid emergence of protease inhibitor resistance in hepatitis C virus. *Science Translational Medicine* 2 (30): 30ra32.
- **141** Lamarre, D., Anderson, P.C., Bailey, M. et al. (2003). An NS3 protease inhibitor with antiviral effects in humans infected with hepatitis C virus. *Nature* 426 (6963): 186–189.
- **142** Seiwert, S.D., Andrews, S.W., Jiang, Y. et al. (2008). Preclinical characteristics of the hepatitis C virus NS3/4A protease inhibitor ITMN-191 (R7227). *Antimicrobial Agents and Chemotherapy* 52 (12): 4432–4441.
- 143 Deutsch, M. and Papatheodoridis, G.V. (2010). Danoprevir, a small-molecule NS3/4A protease inhibitor for the potential oral treatment of HCV infection. *Current Opinion in Investigational Drugs* 11 (8): 951–963.
- 144 Rosenquist, A., Samuelsson, B., Johansson, P.O. et al. (2014). Discovery and development of simeprevir (TMC435), a HCV NS3/4A protease inhibitor. *Journal of Medicinal Chemistry* 57 (5): 1673–1693.
- **145** Zeuzem, S., Berg, T., Gane, E. et al. (2014). Simeprevir increases rate of sustained virologic response among treatment-experienced patients with HCV genotype-1 infection: a phase IIb trial. *Gastroenterology* 146 (2): 430–41 e6.
- HIVandHepatits.com FDA suspends trials of experimental HCV regimen IDX184 plus IDX320 due to liver toxicity concerns (accessed 7 September 2010) http://www. hivandhepatitis.com/hcv-treatment/experimental-hcv-drugs/1989-fda-suspendstrials-of-experimental-hcv-regimen-idx184-plus-idx320-due-to-liver-toxicity-concerns.

- **52** 2 Curing Hepatitis C with Direct-Acting Antiviral Therapy
 - 147 Parsy, C.C., Alexandre, F.R., Bidau, V. et al. (2015). Discovery and structural diversity of the hepatitis C virus NS3/4A serine protease inhibitor series leading to clinical candidate IDX320. *Bioorganic & Medicinal Chemistry Letters* 25 (22): 5427–5436.
 - **148** Gentile, I., Borgia, F., Buonomo, A.R. et al. (2014). ABT-450: a novel protease inhibitor for the treatment of hepatitis C virus infection. *Current Medicinal Chemistry* 21 (28): 3261–3270.
 - 149 Andreone, P., Colombo, M.G., Enejosa, J.V. et al. (2014). ABT-450, ritonavir, ombitasvir, and dasabuvir achieves 97% and 100% sustained virologic response with or without ribavirin in treatment-experienced patients with HCV genotype 1b infection. *Gastroenterology* 147 (2): 359–365 e1.
 - 150 Gamal, N., Vitale, G., and Andreone, P. (2015). ABT-450: a novel agent for the treatment of CHC genotype 1: focus on treatment-experienced patients. *Expert Review of Anti-Infective Therapy* 13 (3): 295–304.
 - **151** De Luca, A., Bianco, C., and Rossetti, B. (2014). Treatment of HCV infection with the novel NS3/4A protease inhibitors. *Current Opinion in Pharmacology* 18: 9–17.
 - **152** Huang, M.; Podos, S.; Patel, D. et al. (2010). ACH-2684: HCV NS3 protease inhibitor with potent activity against multiple genotypes and known resistant variants. In *American Association for the Study of Liver Disease*, Boston, MA.
 - **153** Levin, J. (2012). Achillion Pharmaceuticals, Inc. announces additional proof-of-concept data with ACH-2684 for the treatment of hepatitis C. FierceBiotech.
 - **154** Sheng, X.C., Casarez, A., Cai, R. et al. (2012). Discovery of GS-9256: a novel phosphinic acid derived inhibitor of the hepatitis C virus NS3/4A protease with potent clinical activity. *Bioorganic & Medicinal Chemistry Letters* 22 (3): 1394–1396.
 - **155** Yang, H., Yang, C., Wang, Y. et al. (2017). Preclinical characterization of the novel HCV NS3 protease inhibitor GS-9256. *Antiviral Therapy* 22 (5): 413–420.
 - 156 Mo, H., Hedskog, C., Lawitz, E. et al. (2017). Antiviral response and resistance analysis of treatment-naive HCV infected patients receiving multiple doses of the NS3 protease inhibitor GS-9256. *Antiviral Research* 140: 151–157.
 - **157** Hotho, D.M., de Bruijne, J., O'Farrell, A.M. et al. (2012). Pharmacokinetics and antiviral activity of PHX1766, a novel HCV protease inhibitor, using an accelerated phase I study design. *Antiviral Therapy* 17 (2): 365–375.
 - McCauley, J.A., McIntyre, C.J., Rudd, M.T. et al. (2010). Discovery of vaniprevir (MK-7009), a macrocyclic hepatitis C virus NS3/4a protease inhibitor. *Journal of Medicinal Chemistry* 53 (6): 2443–2463.
 - **159** Liverton, N.J., Carroll, S.S., Dimuzio, J. et al. (2010). MK-7009, a potent and selective inhibitor of hepatitis C virus NS3/4A protease. *Antimicrobial Agents and Chemotherapy* 54 (1): 305–311.
 - **160** Lawitz, E., Sulkowski, M., Jacobson, I. et al. (2013). Characterization of vaniprevir, a hepatitis C virus NS3/4A protease inhibitor, in patients with HCV genotype 1 infection: safety, antiviral activity, resistance, and pharmacokinetics. *Antiviral Research* 99 (3): 214–220.
 - **161** Manns, M.P., Gane, E., Rodriguez-Torres, M. et al. (2012). Vaniprevir with pegylated interferon alpha-2a and ribavirin in treatment-naive patients with chronic hepatitis C: a randomized phase II study. *Hepatology* 56 (3): 884–893.
 - 162 Harper, S., McCauley, J.A., Rudd, M.T. et al. (2012). Discovery of MK-5172, a macrocyclic hepatitis C virus NS3/4a protease inhibitor. ACS Medicinal Chemistry Letters 3 (4): 332–336.

- **163** Summa, V., Ludmerer, S.W., McCauley, J.A. et al. (2012). MK-5172, a selective inhibitor of hepatitis C virus NS3/4a protease with broad activity across genotypes and resistant variants. *Antimicrobial Agents and Chemotherapy* 56 (8): 4161–4167.
- **164** Gane, E., Ben Ari, Z., Mollison, L. et al. (2016). Efficacy and safety of grazoprevir + ribavirin for 12 or 24 weeks in treatment-naive patients with hepatitis C virus genotype 1 infection. *Journal of Viral Hepatitis* 23 (10): 789–797.
- **165** Lagging, M., Brown, A., Mantry, P.S. et al. (2016). Grazoprevir plus peginterferon and ribavirin in treatment-naive patients with hepatitis C virus genotype 1 infection: a randomized trial. *Journal of Viral Hepatitis* 23 (2): 80–88.
- 166 Taylor, J.G., Zipfel, S., Ramey, K. et al. (2019). Discovery of the pan-genotypic hepatitis C virus NS3/4A protease inhibitor voxilaprevir (GS-9857): a component of Vosevi((R)). Bioorganic & Medicinal Chemistry Letters 29 (16): 2428–2436.
- **167** Rodriguez-Torres, M., Glass, S., Hill, J. et al. (2016). GS-9857 in patients with chronic hepatitis C virus genotype 1-4 infection: a randomized, double-blind, dose-ranging phase 1 study. *Journal of Viral Hepatitis* 23 (8): 614–622.
- 168 Gane, E.J., Schwabe, C., Hyland, R.H. et al. (2016). Efficacy of the combination of sofosbuvir, velpatasvir, and the NS3/4A protease inhibitor GS-9857 in treatment-naive or previously treated patients with hepatitis C virus genotype 1 or 3 infections. *Gastroenterology* 151 (3): 448–456 e1.
- **169** Lawitz, E.J., O'Riordan, W.D., Asatryan, A. et al. (2015). Potent antiviral activities of the direct-acting antivirals ABT-493 and ABT-530 with three-day monotherapy for hepatitis C virus genotype 1 infection. *Antimicrobial Agents and Chemotherapy* 60 (3): 1546–1555.
- **170** Ng, T.I., Tripathi, R., Reisch, T. et al. (2018). In vitro antiviral activity and resistance profile of the next-generation hepatitis C virus NS3/4A protease inhibitor glecaprevir. *Antimicrobial Agents and Chemotherapy* 62 (1). DOI: 10.1128/AAC.01620-17
- 171 Lambert, S.M., Langley, D.R., Garnett, J.A. et al. (2014). The crystal structure of NS5A domain 1 from genotype 1a reveals new clues to the mechanism of action for dimeric HCV inhibitors. *Protein Science* 23 (6): 723–734.
- 172 Belema, M., Lopez, O.D., Bender, J.A. et al. (2014). Discovery and development of hepatitis C virus NS5A replication complex inhibitors. *Journal of Medicinal Chemistry* 57 (5): 1643–1672.
- 173 Tellinghuisen, T.L., Marcotrigiano, J., and Rice, C.M. (2005). Structure of the zinc-binding domain of an essential component of the hepatitis C virus replicase. *Nature* 435 (7040): 374–379.
- 174 Love, R.A., Brodsky, O., Hickey, M.J. et al. (2009). Crystal structure of a novel dimeric form of NS5A domain I protein from hepatitis C virus. *Journal of Virology* 83 (9): 4395–4403.
- 175 O'Boyle, D.R. 2nd and Gao, M. (2019). NS5A as a target for HCV drug discovery. In: *HCV: The Journey from Discovery to a Cure*, vol. 2 (ed. M.J. Sofia), 3–25. Switzerland: Springer Nature.
- **176** Wang, C., Jia, L., Huang, H. et al. (2012). In vitro activity of BMS-790052 on hepatitis C virus genotype 4 NS5A. *Antimicrobial Agents and Chemotherapy* 56 (3): 1588–1590.
- 177 Nettles, R.E., Gao, M., Bifano, M. et al. (2011). Multiple ascending dose study of BMS-790052, a nonstructural protein 5A replication complex inhibitor, in patients infected with hepatitis C virus genotype 1. *Hepatology* 54 (6): 1956–1965.

- 54 2 Curing Hepatitis C with Direct-Acting Antiviral Therapy
 - **178** Belema, M. and Meanwell, N.A. (2014). Discovery of daclatasvir, a pan-genotypic hepatitis C virus NS5A replication complex inhibitor with potent clinical effect. *Journal of Medicinal Chemistry* 57 (12): 5057–5071.
 - 179 Link, J.O., Taylor, J.G., Xu, L. et al. (2014). Discovery of ledipasvir (GS-5885): a potent, once-daily oral NS5A inhibitor for the treatment of hepatitis C virus infection. *Journal of Medicinal Chemistry* 57 (5): 2033–2046.
 - **180** Lawitz, E.J., Gruener, D., Hill, J.M. et al. (2012). A phase 1, randomized, placebocontrolled, 3-day, dose-ranging study of GS-5885, an NS5A inhibitor, in patients with genotype 1 hepatitis C. *Journal of Hepatology* 57 (1): 24–31.
 - **181** Coburn, C.A., Meinke, P.T., Chang, W. et al. (2013). Discovery of MK-8742: an HCV NS5A inhibitor with broad genotype activity. *ChemMedChem* 8 (12): 1930–1940.
 - **182** Cheung, T.T., Yan Chiu, J.W., Yuen, M.F. et al. (2018). A phase I, single- and multipledose study to evaluate the pharmacokinetics of elbasvir and grazoprevir in healthy Chinese participants. *Clinical Therapeutics* 40 (5): 719–732 e1.
 - **183** Yeh, W.W., Fraser, I.P., Jumes, P. et al. (2018). Antiviral activity, safety, and tolerability of multiple ascending doses of elbasvir or grazoprevir in participants infected with hepatitis C virus genotype-1 or -3. *Clinical Therapeutics* 40 (5): 704–718 e6.
 - 184 Gardner, S., Cutrell, A., Elko-Simms, C. et al. (2014). A double-blind, randomized, placebo-controlled study to assess the safety, antiviral activity and pharmacokinetics of GSK2336805 when given as monotherapy and in combination with peginterferon alfa-2a and ribavirin in hepatitis C virus genotype 1-infected treatment-naive subjects. *Liver International* 34 (6): e89–e95.
 - 185 Walker, J., Crosby, R., Wang, A. et al. (2014). Preclinical characterization of GSK2336805, a novel inhibitor of hepatitis C virus replication that selects for resistance in NS5A. *Antimicrobial Agents and Chemotherapy* 58 (1): 38–47.
 - 186 Krishnan, P., Beyer, J., Mistry, N. et al. (2015). In vitro and in vivo antiviral activity and resistance profile of ombitasvir, an inhibitor of hepatitis C virus NS5A. *Antimicrobial Agents and Chemotherapy* 59 (2): 979–987.
 - **187** Kowdley, K.V., Lawitz, E., Poordad, F. et al. (2014). Phase 2b trial of interferon-free therapy for hepatitis C virus genotype 1. *New England Journal of Medicine* 370 (3): 222–232.
 - 188 Zhong, M., Peng, E., Huang, N. et al. (2016). Discovery of ravidasvir (PPI-668) as a potent pan-genotypic HCV NS5A inhibitor. *Bioorganic & Medicinal Chemistry Letters* 26 (18): 4508–4512.
 - 189 Hafez, E., Elbaz, T., El Kassas, M., and Esmat, G. (2018). A new potent NS5A inhibitor in the management of hepatitis C virus: ravidasvir. *Current Drug Discovery Technologies* 15 (1): 24–31.
 - 190 Esmat, G., Elbaz, T., El Raziky, M. et al. (2017). Effectiveness of ravidasvir plus sofosbuvir in interferon-naive and treated patients with chronic hepatitis C genotype-4. *Journal of Hepatology*. DOI: 10.1016/j.hep.2017.09.006
 - **191** Link, J.O., Taylor, J.G., Trejo-Martin, A. et al. (2019). Discovery of velpatasvir (GS-5816): a potent pan-genotypic HCV NS5A inhibitor in the single-tablet regimens Vosevi((R)) and Epclusa((R)). *Bioorganic & Medicinal Chemistry Letters* 29 (16): 2415–2427.
 - **192** Lawitz, E., Freilich, B., Link, J. et al. (2015). A phase 1, randomized, dose-ranging study of GS-5816, a once-daily NS5A inhibitor, in patients with genotype 1-4 hepatitis C virus. *Journal of Viral Hepatitis* 22 (12): 1011–1019.

- **193** Ng, T.I., Krishnan, P., Pilot-Matias, T. et al. (2017). In vitro antiviral activity and resistance profile of the next-generation hepatitis C virus NS5A inhibitor pibrentasvir. *Antimicrobial Agents and Chemotherapy* 61 (5). DOI:10.1128/AAC.02558-16
- Gane, E., Poordad, F., Wang, S. et al. (2016). High efficacy of ABT-493 and ABT-530 treatment in patients with HCV genotype 1 or 3 infection and compensated cirrhosis. *Gastroenterology* 151 (4): 651–659 e1.
- Bilello, J.P., Lallos, L.B., McCarville, J.F. et al. (2014). In vitro activity and resistance profile of samatasvir, a novel NS5A replication inhibitor of hepatitis C virus. *Antimicrobial Agents and Chemotherapy* 58 (8): 4431–4442.
- Vince, B., Hill, J.M., Lawitz, E.J. et al. (2014). A randomized, double-blind, multiple-dose study of the pan-genotypic NS5A inhibitor samatasvir in patients infected with hepatitis C virus genotype 1, 2, 3 or 4. *Journal of Hepatology* 60 (5): 920–927.
- Gane, E.J., Stedman, C.A., Schwabe, C. et al. (2018). Short-duration AL-335, odalasvir, with or without simeprevir, in patients with HCV GT1 or 3 infection without cirrhosis. *Hepatology* 68 (6): 2145–2157.
- Murayama, A., Fujiwara, K., Yamada, N. et al. (2018). Evaluation of antiviral effects of novel NS5A inhibitors in hepatitis C virus cell culture system with full-genome infectious clones. *Antiviral Research* 158: 161–170.
- 199 Fridell, R.A., Qiu, D., Valera, L. et al. (2011). Distinct functions of NS5A in hepatitis C virus RNA replication uncovered by studies with the NS5A inhibitor BMS-790052. *Journal of Virology* 85 (14): 7312–7320.
- Karino, Y., Toyota, J., Ikeda, K. et al. (2013). Characterization of virologic escape in hepatitis C virus genotype 1b patients treated with the direct-acting antivirals daclatasvir and asunaprevir. *Journal of Hepatology* 58 (4): 646–654.
- Nakamoto, S., Kanda, T., Wu, S. et al. (2014). Hepatitis C virus NS5A inhibitors and drug resistance mutations. *World Journal of Gastroenterology* 20 (11): 2902–2912.
- Alkhouri, N. and Zein, N.N. (2012). Protease inhibitors: silver bullets for chronic hepatitis C infection? *Cleveland Clinic Journal of Medicine* 79 (3): 213–222.
- Ascione, A. (2012). Boceprevir in chronic hepatitis C infection: a perspective review. *Therapeutic Advances in Chronic Disease* 3 (3): 113–121.
- Perry, C.M. (2012). Telaprevir: a review of its use in the management of genotype 1 chronic hepatitis C. *Drugs* 72 (5): 619–641.
- Lok, A.S., Gardiner, D.F., Lawitz, E. et al. (2012). Preliminary study of two antiviral agents for hepatitis C genotype 1. *The New England Journal of Medicine* 366 (3): 216–224.
- 206 Gane, E.J., Stedman, C.A., Hyland, R.H. et al. (2013). Nucleotide polymerase inhibitor sofosbuvir plus ribavirin for hepatitis C. *New England Journal of Medicine* 368 (1): 34–44.
- Sulkowski, M.S., Gardiner, D.F., Rodriguez-Torres, M. et al. (2014). Daclatasvir plus sofosbuvir for previously treated or untreated chronic HCV infection. *New England Journal of Medicine* 370 (3): 211–221.
- Lawitz, E., Poordad, F.F., Pang, P.S. et al. (2014). Sofosbuvir and ledipasvir fixed-dose combination with and without ribavirin in treatment-naive and previously treated patients with genotype 1 hepatitis C virus infection (LONESTAR): an open-label, randomised, phase 2 trial. *Lancet* 383 (9916): 515–523.

- 56 2 Curing Hepatitis C with Direct-Acting Antiviral Therapy
 - Voaklander, R. and Jacobson, I.M. (2017). Sofosbuvir, velpatasvir and voxilaprevir combination for the treatment of hepatitis C. *Expert Review of Gastroenterology & Hepatology* 11 (9): 789–795.
 - Kwo, P., Gitlin, N., Nahass, R. et al. (2016). Simeprevir plus sofosbuvir (12 and 8 weeks) in hepatitis C virus genotype 1-infected patients without cirrhosis: OPTIMIST-1, a phase 3, randomized study. *Hepatology* 64 (2): 370–380.
 - Lawitz, E., Sulkowski, M.S., Ghalib, R. et al. (2014). Simeprevir plus sofosbuvir, with or without ribavirin, to treat chronic infection with hepatitis C virus genotype 1 in non-responders to pegylated interferon and ribavirin and treatment-naive patients: the COSMOS randomised study. *Lancet* 384 (9956): 1756–1765.
 - Keating, G.M. (2015). Ledipasvir/Sofosbuvir: a review of its use in chronic hepatitis C. *Drugs* 75 (6): 675–685.
 - Osinusi, A. and McHutchinson, J.G. (2019). The clinical development of ledipasvir/ sofosbuvir (LDV/SOF, Harvoni[®]). In: *HCV: The Journey from Discovery to a Cure*, vol. 2 (ed. M.J. Sofia), 237–280. Switzerland: Springer Nature.
 - Gimeno-Ballester, V., Simon, M.A., Trigo, C. et al. (2016). Sofosbuvir plus simeprevir for the treatment of genotype 1 chronic hepatitis C: a review of evidence. *Expert Review of Gastroenterology & Hepatology* 10 (11): 1289–1303.
 - Pol, S., Vallet-Pichard, A., and Corouge, M. (2014). Treatment of hepatitis C virus genotype 3-infection. *Liver International* 34 (Suppl 1): 18–23.
 - Greig, S.L. (2016). Sofosbuvir/velpatasvir: a review in chronic hepatitis C. *Drugs* 76 (16): 1567–1578.
 - 217 Brainard, D.M. and McHutchinson, J.G. (2019). The clinical development of sofosbuvir/velpatasvir (SOF/VEL, Epclusa[®]). In: *HCV: The Journey from Discovery to a Cure*, vol. 2 (ed. M.J. Sofia), 281–316. Switzerland: Springer Nature.
 - Pearlman, B., Perrys, M., and Hinds, A. (2019). Sofosbuvir/velpatasvir/voxilaprevir for previous treatment failures with glecaprevir/pibrentasvir in chronic hepatitis C infection. *The American Journal of Gastroenterology* 114 (9): 1550–1552.
 - Heo, Y.A. and Deeks, E.D. (2018). Sofosbuvir/velpatasvir/voxilaprevir: a review in chronic hepatitis C. *Drugs* 78 (5): 577–587.
 - 220 Mathur, P., Kottilil, S., and Wilson, E. (2019). Sofosbuvir/velpatasvir/voxilaprevir: a highly effective option for retreatment of hepatitis C in difficult-to-treat patients. *Antiviral Therapy* 24 (1): 1–10.
 - Cohen, D.E. (2019). Clinical development of Viekira Pak to Mavyret. In: *HCV: The Journey from Discovery to a Cure*, vol. 2 (ed. M.J. Sofia), 347–368. Switzerland: Springer Nature.
 - **222** Robertson, M.N. and Barr, E. (2019). Development of ZEPATIER[®]. In: *HCV: The Journey from Discovery to a Cure*, vol. 2 (ed. M.J. Sofia), 369–408. Switzerland: Springer Nature.
 - Lledo, G.M., Carrasco, I., Benitez-Gutierrez, L.M. et al. (2018). Regression of liver fibrosis after curing chronic hepatitis C with oral antivirals in patients with and without HIV coinfection. *AIDS* 32 (16): 2347–2352.
 - El-Raziky, M., Khairy, M., Fouad, A. et al. (2018). Effect of direct-acting agents on fibrosis regression in chronic hepatitis C virus patients' treatment compared with interferon-containing regimens. *Journal of Interferon & Cytokine Research* 38 (3): 129–136.

- **225** Foster, G.R., Irving, W.L., Cheung, M.C. et al. (2016). Impact of direct acting antiviral therapy in patients with chronic hepatitis C and decompensated cirrhosis. *Journal of Hepatology* 64 (6): 1224–1231.
- **226** Pradat, P., Virlogeux, V., and Trepo, E. (2018). Epidemiology and elimination of HCV-related liver disease. *Viruses* 10 (10).
- **227** Younossi, Z.M., Stepanova, M., Henry, L. et al. (2018). The effect of interferon-free regimens on health-related quality of life in East Asian patients with chronic hepatitis C. *Liver International* 38 (7): 1179–1187.
- **228** Juanbeltz, R., Martinez-Baz, I., San Miguel, R. et al. (2018). Impact of successful treatment with direct-acting antiviral agents on health-related quality of life in chronic hepatitis C patients. *PLoS One* 13 (10): e0205277.
- 229 van Tilborg, M., Maan, R., van der Meer, A.J., and de Knegt, R.J. (2017). Interferon-free antiviral therapy for chronic hepatitis C among patients in the liver transplant setting. Best Practice & Research. Clinical Gastroenterology 31 (2): 219–225.
- **230** Sulkowski, M., Naggie, S., Lalezari, J. et al. (2014). Sofosbuvir and ribavirin for hepatitis C in patients with HIV coinfection. *Journal of the American Medical Association* 312 (4): 353–361.
- **231** Hawkins, C., Grant, J., Ammerman, L.R. et al. (2016). High rates of hepatitis C virus (HCV) cure using direct-acting antivirals in HIV/HCV-coinfected patients: a real-world perspective. *The Journal of Antimicrobial Chemotherapy* 71 (9): 2642–2645.
- **232** FDA (2019). FDA approves label changes for Sovaldi, Harvoni in children with HCV. https://www.healio.com/hepatology/hepatitis-c/news/online/%7B5edae8d8c4f3-4334-8883-7cc06a04c750%7D/fda-approves-label-changes-for-sovaldi-harvoni-inchildren-with-hcv (accessed 15 March 2020).
- **233** Potluri, V.S., Goldberg, D.S., Mohan, S. et al. (2019). National trends in utilization and 1-year outcomes with transplantation of HCV-viremic kidneys. *Journal of the American Society of Nephrology* 30 (10): 1939–1951.
- 234 Schlendorf, K.H., Zalawadiya, S., Shah, A.S. et al. (2019). Expanding heart transplant in the era of direct-acting antiviral therapy for hepatitis C. *JAMA Cardiology*. DOI: 10.1001/jamacardio.2019.4748
- **235** Calcoen, D., Elias, L., and Yu, X. (2015). What does it take to produce a breakthrough drug? *Nature Reviews* 14 (3): 161–162.

Oliver Planz

3

Department of Immunology, Interfaculty Institute for Cell Biology, Eberhard Karls University, Tuebingen, Germany

3.1 The Influenza Virus

Influenza virus is a negative sense, segmented RNA virus that causes severe respiratory disease and up to 650 000 deaths annually worldwide [1]. Two types are responsible for major disease burden: while influenza A virus causes most influenza cases and causes pandemics, influenza B virus is a major source of morbidity and mortality in influenza during interpandemic periods [2]. Seasonal influenza epidemics alone affect up to 9 million people in the United States resulting in 12000–56000 deaths annually and in a high economic burden with an estimated average of \$11.2 billion annually [1, 3]. In addition to the seasonal epidemic influenza, worldwide pandemic outbreaks emerge at unpredictable intervals causing an enormous increase in morbidity and mortality and consequently an increase of the negative economic impact [4]. Thus, developing safe and effective strategies for prevention, prophylaxis, and treatment against influenza is crucial.

Influenza virus harbors three surface proteins, the hemagglutinin (HA), the neuraminidase (NA), and the matrix 2 (M2) protein. By phylogenic similarity, the HA proteins are divided into two groups. For cell attachment and entry, the HA binds to terminal Neu5Ac a(2,3)- and Neu5Ac a(2,6)-Gal linkages of sialic acid on the cell surface [5]. Activation of the M2 proton channel is then induced by lowering the pH of the endosome. This leads to an acid environment to the virus and triggers the release of the viral ribonucleoprotein (RNP). Replication requires the viral polymerase complex, which is highly conserved between different strains. This heterotrimer is composed of three protein subunits: polymerase basic protein 1 (PB1), polymerase basic protein 2 (PB2), and polymerase acidic protein (PA) [6]. The influenza A and B virions contain eight viral ribonucleoprotein (vRNP) segments, which have a double-helical hairpin structure. Each is carrying one polymerase heterotrimer and is stabilized by viral NP molecules [6–8]. Viral RNA synthesis occurs in the cell nucleus. After replication, new virions are released by budding. The NA of the ensembled virions cleaves sialic acids from the host cell receptors and from the glycosylated viral HA and NA proteins to prevent self-aggregation and to support the release of progeny virions from the cell surface [9].

Vaccines represent the first choice to protect against the burden of influenza [10]. However, due to antigenic drifts and occasional antigenic shifts due to exchange of RNP segments the virus is able to escape the pre-existing immunity [11]. This results in reduced efficacies of the vaccines and limits their usage. Thus, continuous adaptions of the vaccines to the current influenza virus strains and consequently annual immunization are needed. Antigenic shifts can result in influenza virus strains that cause pandemics [12]. In such a pandemic situation, the available vaccines are completely ineffective, and it takes several months after the emergence and spread of a pandemic influenza virus until a new vaccine, effective against the new emerging virus, is available [13]. Moreover, only roughly 50% of the vaccinated individuals develop an optimal immune response, but especially in younger children and the elderly, who are at risk of severe influenza, the immune response to the virus is suboptimal. Also, the influenza vaccination coverage is only about 50% of the population. This shows a second problem, which limits the prevention of influenza by vaccination [10]. Thus, because of these limitations of the current vaccines to protect during seasonal influenza, antivirals play an important role for the treatment of the disease. Furthermore, in a pandemic situation, virus-specific vaccine is unlikely to be available until after many months; thus in the early phase of a pandemic, antivirals are needed to control it.

Today, the standard of care drugs are neuraminidase inhibitors (NAIs), which show limitations regarding the treatment window for clinical effectiveness and the emergence of NAI-resistant influenza virus strains. Several attempts have been made in recent years to develop new compounds directed against influenza viruses to overcome the limitations of NAIs. Among these preparations, only baloxavir marboxil (BAM) and favipiravir (FP), both polymerase inhibitors, have reached the market (the first in Japan and the United States, and the second only in Japan). Other antiviral agents and monoclonal antibodies are in advanced stages of development, but, at present, none of these new drugs and monoclonal antibodies have satisfactory characteristics to substitute for NAIs. However, a completely new strategy in the development of antivirals has reached the clinical development stage. In contrast to target the virus directly, cellular factors were identified which are needed by the virus to ensure its propagation. Consequently, inhibiting these cellular factors leads to an inhibition of the formation of new progeny virus.

Given the inherent limitations of the currently licensed drugs against influenza, there is still an urgent need for developing new anti-influenza drugs. The focus of novel drug development should aim at the following characteristics: (i) effective, when start of treatment is late (>48 hours), (ii) low susceptibility for developing resistant virus variants, (iii) broad range of activity (influenza A and B virus), (iv) improved effectiveness compared to the standard of care (at the present: NAIs), suitable for uncomplicated as well as complicated (severe) cases of influenza [14, 15]. This chapter provides an overview of antiviral drugs against influenza that have either been licensed in at least one country or are at least under phase 2 clinical development.

3.2 The Pathogenesis of Influenza

The transmission of influenza virus occurs primarily through droplets when sneezing or coughing. In addition, it is discussed whether droplets generated during normal breathing

or speech will also contain infectious influenza virus. Furthermore, transmission occurs through direct contact, for example, by shaking hands [16].

Influenza virus infects epithelial cells of the upper respiratory tract, which represent the primary compartment of virus propagation. In a second step, the virus can spread to the distal respiratory tract to cause severe disease like pneumonia. Not the high viral load but rather an excessive persistent immune response is the determinant of a severe pneumonia with respiratory failure [17]. Lung failure or a high severity of the disease is associated with a strong activation or recruitment of monocytes/macrophages and neutrophils [18–21]. These cells are the main source for the secretion of cytokines and chemokines, which can lead to an overwhelming cytokine/chemokine response, a hypercytokinemia also known as "cytokine storm" (Figure 3.1) [22, 23].

Influenza is associated with a wide range of clinical symptoms, varying from courses with mild signs of a cold to severe pneumonia requiring ventilation. After influenza virus infection and an incubation period of one to two days, the onset of symptoms often starts rapidly with high fever and chills. Signs and symptoms like loss of appetite, dry cough, sore throat, headache, myalgia, and arthralgia are often associated with influenza. Dyspnea and pronounced exhaustion are signs for a pneumonia. Depending on the age group, the disease can develop differently. While adults often show only a slight increase in temperature after infection with a particular influenza virus strain, the same strain can lead to abdominal pain, diarrhea, and vomiting in young children [24]. Even before the onset of first clinical symptoms, the infected person can spread the virus. In immune-competent patients, this infectivity lasts for about four to five days after the onset of symptoms. In



Figure 3.1 Mechanism of acute inflammation induction after influenza virus infection.

immunocompromised patients, the infectivity can last up to more than 30 days [25, 26]. Usually in an uncomplicated influenza, the symptoms disappear spontaneously within a week. The most common complication of an influenza infection is the influenza-associated pneumonia, both as primary viral pneumonia but also as a secondary bacterial coinfection [27]. The mortality risk increases with a bacterial coinfection [28]. Next to bacterial coinfections, pulmonary aspergillosis (fungal infection from the species *Aspergillus*) is another complication in influenza that is accompanied with a high mortality rate [29].

3.3 Influenza Drugs and Targets

In 2020, three major classes of antivirals were licensed for the treatment and prophylaxis of influenza: the M2 ion channel blockers (adamantanes), the NAIs, and the viral polymerase inhibitors, which have been recently progressed through late-stage clinical trials. Two of them (BAM and FP) have been licensed in Japan and BAM also in the United States. Table 3.1 provides an overview of the licensed antiviral drugs available to treat influenza.

Antiviral drugs have been developed for a long time to reduce the influenza-related risks and to overcome the challenges and problems caused by vaccines. The adamantane derivatives (rimantadine and amantadine) *a polymerase inhibitor (Baloxavir Marboxil)* and the

Antiviral drug	Target	Route of administration	Treatment Prophylactic	Therapeutic	Effectiveness
Amantadine/ Rimantadine	M2	Oral	100 mg twice daily	100 mg twice daily	Influenza A virus
Symmetrel®/ Flumadine®					
Zanamivir	Neuraminidase	Inhalative	10 mg	10 mg twice daily	Influenza A virus
Relenza®			once daily up to 28 days		Influenza B virus
Oseltamivir	Neuraminidase	Oral	75 mg	75 mg twice daily	Influenza A virus
Tamiflu®			once daily		Influenza B virus
			up to six weeks		
Peramivir	Neuraminidase	Intravenous		600 mg	Influenza A virus
Rapivap®			single dose	Influenza B virus	
Laninamivir	Neuraminidase	Inhalative		20–40 mg single dose	Influenza A virus
Inavir®					Influenza B virus
Baloxavir Marboxil	Polymerase	Oral	40–80 mg 40–8 single dose singl 600– twice	40-80 mg	Influenza A virus
Xofluza®				single dose	Influenza B virus
Favipiravir	Polymerase	Oral		600–1600 mg twice daily	Influenza A virus
Avigan®					Influenza B virus

 Table 3.1
 Current licensed drugs against influenza.

NAIs (oseltamivir, zanamivir) have been the only drugs licensed worldwide for influenza treatment and prevention. In recent years, laninamivir and peramivir were licensed in Japan, China, Japan, South Korea, and the United States. All these NAIs differ in their pharmacokinetic (PK) characteristics, routes of administration, and the age of the targeted patients [10, 30]. However, there exist limitations of NAIs in their efficacy to treat influenza. Furthermore these NAIs, mainly oseltamivir, are under discussion in terms of the risk of emergence of resistant strains and doubts regarding their efficacy in severe influenza in high-risk patients, like influenza virus-induced exacerbation of COPD [31–33]. The development of the third class of antivirals against influenza, namely the polymerase inhibitors, was aimed to overcome these limitations. BAM is the first polymerase inhibitor that was approved for uncomplicated influenza and for the treatment of influenza in high-risk patients. However, until now clinical trials have excluded hospitalized patients with severe influenza [34]. The use of BAM has already been associated with emergence of resistant influenza virus variants [35, 36].

Thus, several new approaches and strategies have been attempted to achieve influenza prevention and control. Monoclonal antibodies against the highly conserved stem region of the HA molecule of influenza A viruses and drugs that target different stages of the influenza virus life cycle in the infected human cells have been developed and tested [37]. Moreover, drugs that interfere with host cell factors are in development [38, 39]. Up to date, none of these new drugs or monoclonal antibodies that are in development has adequate characteristics to be superior over NAIs. Especially, oseltamivir remains the drug of choice for influenza treatment, even though the limitations of the drug are obvious [40].

3.4 Adamantanes and Derivatives

Influenza A virus encodes an integral membrane protein, M2 that forms a proton channel, which is required for viral replication [41]. The first anti-influenza compounds licensed were the adamantane derivatives rimantadine and amantadine, which inhibit viral propagation by blocking the proton conductivity of the M2 ion channel and preventing the migration of the vRNP complex into the nucleus of the host cell [15, 30, 42]. However, amantadine and rimantadine act only on the M2 ion channel of influenza A viruses, since the M2 protein of influenza B viruses is structurally different reducing the sensitivity to the M2 inhibitors [43, 44]. M2 inhibitors bind the viral matrix2 protein on two potential sites. One is located in the ion channel pore and has a high affinity for the inhibitor, while the second is located on the lipid face of the pore and shows a low-affinity for the inhibitor [45, 46]. M2 inhibitors have been around for almost 50 years. Their synthesis is cheap but their use for the treatment of influenza has been limited (Figure 3.2) [47].

This is partially due to the fact that amantadine- and rimantadine-resistant viruses emerge rapidly in treated patients, and in a single passage in tissue culture [48, 49]. Already in the 1980s, amantadine- and rimantadine-resistant influenza A virus strains were found [50]. Notably, the resistant virus variants showed no decline in replicative fitness and transmissibility [51, 52]. The ion channel pore harbors the two most common mutations (V27A and



Figure 3.2 Structure of amantadine and rimantadine.

S31N), confirming this as the pharmacologically important site [46]. The pandemic A(H1N1) pdm09 virus was already resistant shortly after appearance in 2009 [53] and amantadineresistant A(H3N2) viruses retaining a S31N substitution in the M2 protein spread globally during the 2000s [54, 55]. Many of the circulating avian Influenza A(H5N1) strains, which have led to "birdflu" in Southeast Asia, also developed resistance to M2 inhibitors [56, 57]. The levels of resistance to amantanes among circulating influenza A(H3N2) and influenza A(H1N1)pdm09 viruses is >99%. Moreover, adamantanes show poor tolerability [58]. Thus, since the 2004–2005 influenza season, the use of adamantane derivatives for antiviral treatment or chemoprophylaxis of currently circulating influenza A viruses was no longer recommended by the Centers for Disease Control and Prevention (CDC) [10, 59].

3.5 Neuraminidase Inhibitors

The recommended antiviral drugs for influenza prevention and treatment are NAIs [60]. The development of NAI started in the early 1970s with derivatives of 2,3-dehydro-2-deoxy-*N*-acetylneuraminic acid (DANA) (Figure 3.3), which were known to be weak inhibitors of NA [61, 62].

A single substitution of the C4-OH with a 4-guanidino group, which led to zanamivir, enhanced binding by more than 10 000-fold over DANA [61, 63]. It was not until 1999 that zanamivir was approved as first in class (July 1999) followed by oseltamivir (October 1999) for use in humans [61, 64, 65]. The NAI bind to the NA of the virus and prevent the budding of influenza viruses from the host cell by inhibiting the enzymatic cleavage (which would facilitate the release of newly formed virions from the cell) and reduce the spread of infection in the respiratory tract [30, 40, 61, 66–68]. Based on the results of several clinical trials, NAIs have been considered efficient for treatment and for prevention of influenza virus infection [69, 70]. Four different NAIs are licensed: oseltamivir and zanamivir are approved globally for the treatment and the prevention of influenza, peramivir is licensed in various countries, while laninamivir was only approved in Japan [10].

NAIs have been found to be relatively safe and well tolerated. In particular, oseltamivir, which is the most frequently used NAI, is not only licensed for the use by otherwise healthy adults as well as subjects with severe underlying disease but also in neonates, younger infants, and pregnant women. Here, doses vary and are dependent on age and weight [69, 70].

One of the challenges of NAIs is that, in order to be effective, treatment should start within 48 hours postexposure, while first clinical symptoms will appear between 24 and 36 hours after initial infection. Since influenza virus replication is fast during the time of incubation, the effectivity of NAIs depends on the timing of the antiviral intake. Oseltamivir



Figure 3.3 Structure of DANA.

reduced viral shedding and shortened the time to symptom alleviation by up to 1.5 days, when treatment started during the first 48 hours postinfection [71, 72].

The second challenge is the emergence of influenza virus strains that are resistant to NAIs. In this regard, most available studies were carried out with oseltamivir and zanamivir, which were approved before the other NAIs [40, 71]. The most dominant mutation which confers reduced susceptibility to both oseltamivir and peramivir is a H274Y substitution in the NA [73]. Until to date, only influenza A(H1N1) viruses harboring this mutation have circulated widely at the community level [74]. Moreover, different NA amino acid substitutions for influenza A and B viruses were found leading to a reduced sensitivity of peramivir, zanamivir, and laninamivir, but these are rarely detected and some are cell culture-only-derived substitutions [75].

The effectiveness of NAIs has been documented in meta-analyses, including those by Dobson et al. [71] and the Cochrane Report by Jefferson et al. [40]. They concluded that NAIs are suitable to limit the severity and duration of an uncomplicated influenza in patients of any age. However, the efficacy depends on the start of treatment within the first 48 hours of symptom onset. Unfortunately, there are no firm conclusions available regarding the effect of NAIs on severe influenza [10]. Here, the present results are even conflicting [40, 76]. The other NAIs, like laninamivir and peramivir, do not seem to have substantially different efficacies compared to those of oseltamivir or zanamivir. Thus, NAIs cannot be considered completely satisfactory for the treatment of influenza and the development of new drugs against influenza viruses has been considered mandatory [10].

3.5.1 Oseltamivir

Oseltamivir phosphate is a prodrug indicated for oral administration. Once in the gastrointestinal tract, the prodrug is rapidly cleaved to the active metabolite, oseltamivir carboxylate (Figure 3.4) [77].



Figure 3.4 Structure of the oseltamivir phosphate (prodrug) and oseltamivir carboxylate (active metabolite).

Oseltamivir was approved in 2000, shortly after zanamivir and is the most commonly prescribed NAI globally. Oseltamivir is considered the standard of care in many countries for the treatment of hospitalized influenza cases [78]. A large number of preclinical studies were conducted to investigate the efficacy of oseltamivir in cell culture and in animal models. Influenza A/H3N2 viruses are slightly more sensitive to oseltamivir than Influenza A/H1N1 virus subtypes, whereas oseltamivir demonstrated a 10–20-fold lower efficacy for Influenza B viruses compared with influenza A viruses [61, 79]. For the treatment of adults, oseltamivir is taken orally twice daily, with a dose of 75 mg (Table 3.1).

In various randomized controlled trials, which were mainly conducted in otherwise healthy adults, it was shown that oseltamivir shortens the duration of symptoms in uncomplicated influenza by approximately 1 day, if treatment was initiated within 48 hours of symptom onset [40, 71, 80]. Next to studies in uncomplicated influenza, observational studies in severely ill or hospitalized influenza patients indicated that treatment reduces the risk of pneumonia and mortality [71, 81, 82]. All of these studies demonstrated that the start of treatment is crucial and the greatest benefit is seen when start of treatment was as early as possible after symptom onset [81]. Prophylactic administration of oseltamivir, for as long as 10 days to 6 weeks demonstrated 68–90% efficacy in preventing infection during an influenza activity period [31, 83]. Nevertheless, to-date, the role of oseltamivir in reducing uncomplicated influenza symptoms and severe complications are not entirely clear, since reports with contradictory results exist. These reports were summarized in different meta-analyses like the Cochrane report in 2014. Here, it was stated that the administration of oseltamivir in adults with uncomplicated influenza decreased the time to first alleviation of symptoms by only 16.8–25.2 hours [40, 71].

Oseltamivir-resistant viruses can emerge either in the course of influenza virus evolution (naturally) or under drug-selection force (with drug intervention). The immune status of the patient and the severity of illness can also affect the frequency of resistance. Here, higher rates are often observed in hospitalized children and immunocompromised patients in whom virus replication is prolonged [84-86]. The frequency of virus resistance to oseltamivir is lower than that seen with adamantanes [74]. In the 2007–2008 and 2008–2009 influenza seasons, roughly 90% of the circulating influenza A H1N1 strains developed resistance to oseltamivir [87-89]. In contrast, the 2009 pandemic strain A H1N1pdm09, which is circulating since then, rarely (<1-3%) contained the mutations in the NA leading to resistance to oseltamivir [10, 75]. Many amino acid substitutions in the NA confer reduced susceptibility to oseltamivir [33, 75]. The most frequent amino acid substitutions in influenza A viruses of N1 NA subtypes are H274Y and N294S [90]. E119V and R292K mutations in the NA are most frequent in influenza viruses of the N2 subtypes, while influenza B viruses harbor R152K and D198N amino acid substitutions in the NA [10, 91, 92]. There is no cross-resistance among NAIs. Most of these substitutions leading to reduced susceptibility to oseltamivir do not influence the sensitivity of other NAIs. Patients infected with an oseltamivir-resistant strain can be treated successfully with zanamivir [93].

3.5.2 Zanamivir

Zanamivir was developed as GG167 (4-guanidino-2,4-dideoxy-2,3-dehydro-*N*-acetylneuraminic acid) in the 1990s as a novel inhibitor of the influenza virus NA (Figure 3.5) [94–96].



Figure 3.5 Chemical structure of zanamivir.

It exhibits a higher affinity for the NA binding site than the cellular sialic acid and therefore blocks the NA-sialic acid interaction [61]. In 1999, zanamivir was the first NAI approved for use in adults and children >7 years of age for the treatment and prevention of uncomplicated influenza. For treatment, a 10 mg dose is inhaled twice daily for 5 days, whereas for prevention, zanamivir is given once daily for 10-28 days (Table 3.1) [83]. In preclinical studies, zanamivir has demonstrated efficacy against influenza A and B virus strains including avian influenza virus strains (H5N1) in cell culture and in animal models [68, 94, 95, 97]. Various studies demonstrated that zanamivir has a higher activity against influenza B and H1N1 strains than oseltamivir, but oseltamivir is more active against H3N2 [83]. Due to its polarity, zanamivir is not well absorbed resulting in a poor oral bioavailability. Thus, zanamivir is administered as an inhaled powder [30]. Inhaled zanamivir shows a protective efficacy of 68% for seasonal prophylaxis in healthy adults [98]. Furthermore, inhaled zanamivir is effective in seasonal prophylaxis in at-risk adolescents and adults, but in older people, this effect is not significant [99]. In children, inhaled zanamivir had no significant effect on the reduction in duration of clinical symptoms [40, 100]. Inhaled zanamivir has not been found to reduce pneumonia complications during severe influenza [40]. In patients with pulmonary disease, zanamivir can precipitate bronchospasm and therefore, it is not suitable to treat patients who require mechanical ventilation [101]. Postexposure prophylaxis using inhaled zanamivir was effective in preventing transmission to adults and children [102]. Zanamivir is also available in an intravenous (IV) form for compassionate use and is currently undergoing clinical trials to determine optimal dosing [83]. IV zanamivir showed a similar safety profile as oseltamivir, but the time to clinical response was not superior to oseltamivir [103].

The frequency of resistance to zanamivir has constantly remained low, most probably due to its infrequent use and its closer homology to the natural sialic acid substrate [66]. Of note, zanamivir remains active against influenza strains that contain the H274Y NA mutation, leading to reduced susceptibility of oseltamivir and peramivir [101]. Nevertheless, since 2009, there are increased reports of oseltamivir/zanamivir-resistant A(H1N1)pdm09 and H3N2 isolates [61, 92].

3.5.3 Peramivir

Peramivir, a novel cyclopentane and structurally distinct from other NAIs, was developed as BCX-1812 and RWJ-27020 and was approved by the Food and Drug Administration in December 2014 (Figure 3.6) [104].



Peramivir

Figure 3.6 Chemical structure of peramivir.

Compared to other NAIs, peramivir shows structural substitutions, which support a tighter interaction between peramivir and the NA active site and leads to a slower dissociation rate ($t_{1/2} > 24$ hours) when compared to oseltamivir or zanamivir ($t_{1/2} = 1.25$ hours) [105, 106]. Peramivir is a highly selective inhibitor of influenza A and B virus NAs and an effective inhibitor of various influenza A and B virus strains including those with pandemic potential [107]. *In vitro*, the inhibitory potency for influenza B virus is higher than that for influenza A virus. It also appears to be greater than that of either zanamivir or oseltamivir due to the tighter binding of peramivir to the NA [108, 109].

In mice and ferrets infected with influenza A virus and in influenza B virus-infected mice, peramivir was highly effective in viral load reduction, preventing death and limiting disease symptoms. Peramivir was still effective in mice when therapy began 24 hours after virus infection. Preclinical pharmacology revealed that peramivir was non-toxic in mice and rats at doses of $\geq 1000 \text{ mg/kg/day}$ and $\geq 100 \text{ mg/kg/day}$ in ferrets [110]. Even though peramivir demonstrated a solid efficacy in human challenge studies after oral application, human clinical trials revealed a low oral bioavailability and thus low efficacy [111]. Peramivir has been approved for IV administration only. Due to high plasma and respiratory tissue concentrations, it is administered as a single-dose infusion. Thus, peramivir provides a valuable therapeutic alternative for critically ill patients or those unable to take medication via the oral or inhalative routes [112]. Several clinical trials showed that peramivir is well tolerated in adult and pediatric populations. Only mild-to-moderate adverse events were found. Common side effects include gastrointestinal disorders, electrocardiographic (ECG) abnormalities, and decreased neutrophil counts. Compared to oseltamivir, zanamivir, and other NAIs, the antiviral efficacy of peramivir is slightly lower in humans [101]. A large variety of clinical studies were performed to demonstrate the antiviral efficacy of peramivir after IV administration [112–115]. In a phase 2 trial, a significant reduction of time to alleviation of symptoms and significant viral load reduction compared to placebo was demonstrated, when peramivir was administered at doses of 300 and 600 mg, respectively [113].

Development of resistance to peramivir among seasonal influenza A/H1N1pdm09, influenza A/H3N2 and B viruses is low (1-3%) with a higher tendency for influenza A/H1N1pdm09 viruses [116–118]. The most commonly detected H274Y framework

mutation in A/H1N1pdm09 viruses conferring resistance to oseltamivir also results in cross-resistance to peramivir but not to zanamivir. This is due to the fact that resistant variants with mutations in the enzyme framework can retain susceptibility to other NAIs, while those in the functional or catalytic residues tend to cause cross-resistance to most of the others NAIs [119]. The H274Y mutation has been shown to rapidly emerge upon IV treatment with peramivir in immunocompromised or hematopoietic cell-transplanted recipients [120].

3.5.4 Laninamivir

Laninamivir octanoate, which was developed as CS-8958, is the octanoyl ester prodrug of laninamivir (Figure 3.7).

Like zanamivir, laninamivir is an inhaled NAI, but currently approved for the treatment and prevention of influenza A and B infection only in Japan [121]. The laninamivir octanoate prodrug is converted to the active form in the respiratory tract [122]. This allows to treat influenza following a single inhaled dose. Laninamivir exhibits NA inhibitory activity against influenza A and B viruses, including the 2009 pandemic H1N1 (H1N1pdm09) viruses and has good activity against strains that contain the H274Y NA mutation [123-125]. Laninamivir octanoate has long-lasting antiviral activities, due to its long half-life and its high concentration within lung tissues [121]. Preclinical studies in mice demonstrated that the efficacy of a single administration of laninamivir octanoate was comparable to the multiple treatment regimen of zanamivir or oseltamivir [125, 126]. In humans, laninamivir showed a slow elimination lasting for up to 6 days after a single inhalation [125]. For treatment, laninamivir is approved as a single inhalation dose of 40 mg for individuals \geq 10 years of age and 20 mg for children (Table 3.1). For prevention, a single inhalation of 20 mg daily for two days is recommended in adults and children and was associated with roughly 75% protective efficacy. The efficacy of laninamivir is comparable to the other approved NAIs. Like zanamivir, the structure of laninamivir is most similar to the natural target, and therefore resistance is seen less frequently, because it would lead to greater changes in the virus, which could lead to decreased fitness [61, 127]. Clinical studies found lower rates of adverse events when compared with oseltamivir. Unlike zanamivir, laninamivir was not associated with bronchospasm [128].



Figure 3.7 Chemical structures of laninamivir (active metabolite) and laninamivir octanoate (prodrug).

3.5.5 Summary

NAIs represent a valuable drug class for treatment and prevention of influenza. They are considered as standard of care treatment against influenza. Unfortunately, the antiviral efficacy is controversially discussed due to the narrow time window for treatment and the appearance of drug-induced resistant influenza virus variants. This emphasizes the need to evaluate the potential benefits of combinations of NAIs in reducing the rates of resistance and to improve treatment efficacy. Furthermore, this also shows that there is a need to develop drugs that have different targets than the NA.

3.6 Polymerase-Inhibitors

The structure and function of the influenza virus polymerase complex has been described in detail [6, 7, 129, 130]. It is a heterotrimer composed of three protein subunits: PB1, PB2, and PA. Viral RNAs are imported into the nucleus where they are transcribed to mRNA. Here, PB2 is involved in the import of viral RNA into the nucleus and catalyzes the replication and transcription [131, 132]. A PA-dependent "capsnatching" reaction on cellular-capped RNAs is required to generate primers for the primary transcription of vRNA to mRNA [133, 134]. The elongation of the viral RNA is performed by the RNA-dependent RNA polymerase, PB1 [6, 74, 135–137].

The influenza polymerase complex is highly conserved and plays an essential role in the influenza virus replication cycle [6]. This renders the polymerase complex an attractive target for the development of small molecule inhibitors as new antiviral compounds [6, 74]. There is great hope that this group of antivirals against influenza will help to overcome the limitations found with NAIs and to increase the repertoire of antiviral options in particular for an effective treatment of severe influenza (e.g. in hospitalized patients), but also for treatment to start >48 hours post-symptom onset and with a lower tendency for resistance [6, 74]. However, as monotherapy, both licensed polymerase-inhibitors are associated with the frequent emergence of variants with reduced susceptibility. Preliminary data suggest that combination therapy with NAIs limit the emergence of antiviral-resistant strains. This represents an attractive approach for future studies:

3.6.1 Baloxavir Marboxil (BAM)

In vitro studies have demonstrated antiviral activity without cytotoxicity even in cells infected with various influenza virus strains including avian subtypes and NAI-resistant influenza viruses [138]. Mice infected with influenza virus were protected from clinical signs and mortality even when treatment started four days postinfection. Moreover, a combination of oseltamivir with a subtherapeutic dose of BAM protected mice from infection and mortality [139]. The *in vitro* susceptibility profiles are summarized by Hayden and Shindo [140].

Phase 1 clinical trials revealed that a dose escalation from 6 to 80 mg of BAM was well tolerated in healthy adults and no serious adverse events or deaths were reported [141]. Treatmentemergent adverse events included headache, increased alanine aminotransferase and aspartate aminotransferase levels, elevated eosinophil count and white blood cell count, which were present in 8 of 55 BAM-exposed individuals. These adverse events were mild and resolved spontaneously within a few days [141]. In phase 2 and 3 clinical trials involving outpatients aged 12–64 years with symptoms of uncomplicated influenza-like illness, BAM was also well tolerated including in high-risk patients with comorbidities such as asthma or chronic lung disease [141, 142]. Adverse events were reported in $\sim 2\%$ of the patients and were not considered severe [10]. A recent update to BAM precautions in Japan stated that bloody stool, epistaxis, hematuria, or other forms of bleeding might occur [74]. In phase 2 and 3 clinical trials including the CAPSTONE 1 (phase 3) clinical study, different amounts of BAM treatment reduced the time to alleviation of symptoms in predominantly A(H1N1)pdm09 and A(H3N2) infected patients from 28.2 to 23.5 hours when compared to placebo [142].

BAM, also known as S-033188, was licensed in 2018 in Japan and the United States as a first-in-class polymerase inhibitor for oral treatment of uncomplicated influenza in subjects aged \geq 12 years with influenza manifestations for \leq 48 hours [74]. In addition, in October 2019, the FDA approved BAM for the treatment of uncomplicated influenza in people that are at high risk of developing flu-related complications. BAM is a prodrug that is hydrolyzed in the intestine, blood, and liver to form baloxavir acid (BXA), which is the active compound to inhibit cap-dependent endonuclease (Figure 3.8) [143]. This prevents the initiation of mRNA synthesis of the influenza virus [144]. The drug shows antiviral activity against several influenza A viruses, including oseltamivir-resistant viruses as well as B viruses [138].

PK data confirmed that due to a long plasma retention time (half-life of 49–91 hours), BAM could be delivered as a single dose in patients with uncomplicated influenza [141]. This avoids compliance issues for a possible combination therapy with NAIs like oseltamivir or inhaled zanamivir, which are administered twice daily for five days [74]. BXA concentrations were lower in subjects after a meal compared to the exposure measured before a meal, indicating that food influences the metabolism of the drug. Nevertheless, food uptake had no influence on BAX-mediated virus inhibition, because the BAX plasma concentration exceeded the target concentration to inhibit viral propagation, which was estimated in preclinical studies [141].



Baloxavir marboxil

Baloxavir acid

Figure 3.8 Chemical structures of baloxavir marboxil (prodrug) and its active metabolite baloxavir acid.

However, no significant difference in the clinical efficacy was observed in patients receiving oseltamivir or baloxavir although BAM administration showed greater impact on viral load reduction [142].

The CAPSTONE 2 trial (phase 3) in outpatients who were at increased risk of influenza complications due to comorbidities, again demonstrated no significant difference in the reduction of the time to improvement of influenza symptoms compared to oseltamivir treatment, although a slight difference was present [145]. However, this study demonstrated a significantly faster reduction of the time to improvement of influenza symptoms than placebo and oseltamivir, when patients were infected with influenza B viruses [145]. Another benefit of BAM in the CAPSTONE 2 study was the significantly lower use of antibiotics and the lower incidence of influenza-related complications in high-risk patients treated with baloxavir [145]. In the clinical trials, BAM resulted in a 100- to 1000-fold-reduction in viral titers [142]. Additional phase 3 clinical trials were conducted in 2019 to evaluate the efficacy of BAM treatment: in children between 1 and 12 years of age, in hospitalized patients with influenza and in a study to evaluate the prophylactic benefits of baloxavir [74].

In summary, the administration as a single oral dose and the antiviral efficacy against NAI-resistant virus strains are important advantages of BAM over other available antiinfluenza virus drugs [138]. Even though oseltamivir is applied via the oral route as well, the other NAIs are either administered by inhalation (zanamivir and laninamivir) or via the IV route (peramivir). Despite these clear advantages, there are some challenges under debate for the usage of BAM in clinical practice as an alternative to NAIs. The costs of influenza treatment with BAM are roughly three times higher than the cost for oseltamivir treatment [146]. Another challenge for BAM is the emergence of variants resistant to the polymerase inhibitor due to mutations in the PA polymerase gene leading to an I38T and E23K exchange in the protein [147]. None of these mutations were detected in 2018/2019 prior to the time when BAM was licensed, thus they are likely drug-induced [35]. Both mutations, however, have been already encountered during the BAM clinical trials used to obtain the license [142]. Here, 2.2–9.7% of the patients shed mutant influenza virus and a pediatric study showed that mutant virus was found in roughly 20% of treated children [142, 144, 148]. The viruses, which were insensitive to BAM remained sensitive to NAIs and had an impaired replicative capacity in vitro [148]. Patients harboring mutant influenza virus shed the viruses and remained symptomatic for a longer period than patients infected with wild-type virus [142]. This clearly demonstrates a potential negative effect to BAM and up to now the routine use of BAM for treatment of uncomplicated influenza in adults is being debated [10, 35, 148-151].

Combination therapy of BAM with oseltamivir was examined in healthy subjects indicating that there was no significant drug-drug interaction in healthy subjects [141]. Studies investigating the effectiveness of oseltamivir and baloxavir combination therapy in humans, with influenza virus infections have recently been completed, results are being awaited. In influenza virus-infected mice, the combination therapy of oseltamivir/baloxavir delivered four days postinfection showed no significant clinical benefit compared to either oseltamivir or baloxavir monotherapy but showed a synergistic response with respect to reduction of viral load and lower pro-inflammatory cytokine (IL-6 and MCP-1) levels [139]. Further research and clinical studies will need to determine the potential of BAM as antiviral against uncomplicated influenza.

3.6.2 Favipiravir

FP also known as T-705, acts as a purine nucleoside analogue and therefore – like BAM – inhibits the activity of the RNA-dependent RNA polymerase directly [152, 153]. FP is a prodrug that after oral ingestion requires intracellular phosphoribosylation to be transformed into its active form, FP ribofuranosyl-5'-triphosphate (FRTP) (Figure 3.9) [10].

FP has a broad-spectrum antiviral activity, which is not limited to all influenza subtypes including those resistant to NA and M2 inhibitors [153]. FP is also able to inhibit virus propagation of several other RNA viruses, including West Nile virus, poliovirus, Ebola virus, and norovirus [10, 154–156]. Currently, FP has conditional marketing approval in Japan; restricted for the use to treatment of novel influenza viruses that are resistant to other available antivirals [153]. Favipiravir showed antiviral activity against SARS-CoV-2 (see Chapter 17). Favipiravir has also shown in vitro activity against SARS-CoV-2 and has been tested in several clinical studies. A meta-analysis revealed "a significant clinical improvement in the Favipiravir group versus the control group during seven days after hospitalization (RR = 1.24, 95% CI: 1.09–1.41; P=0.001)." However, Favipiravir appeared not to exert a significant beneficial effect in terms of mortality in the general group of patients with mild to moderate COVID-19 (https://doi.org/10.1038/s41598-021-90551-6). The dosage for adults is 1600 mg administered orally twice daily on Day 1, followed by 600 mg orally twice daily from Day 2 to Day 5. The total treatment duration should be five days. A recent in vitro study showed that FP is primarily acting as a guanine and an adenine analogue; thus leading to mutations and that it consequently may harbor a risk for teratogenicity and embryotoxicity [153].

In vitro studies demonstrated a broad antiviral spectrum of FP against different influenza virus subtypes, including avian influenza viruses H5N1 and H7N9 and those that are insensitive or resistant to NAIs [84, 157, 158]. When treatment started 72 hours after infection, FP showed a significantly better outcome than oseltamivir in preventing influenza or reducing its severity in mice infected with lethal doses [158, 159]. Moreover, when used in combination with oseltamivir, protection due to the NAI was increased and the treatment efficacy window was extended to 96 hours after symptom onset [10, 160]. Combination therapy studies of FP with NAIs in mice infected with a pandemic H1N1 influenza virus or with an avian influenza virus H5N1 demonstrated that both drugs act synergistically in terms of viral load reduction, reduction of body weight loss, and in protection against a lethal challenge [161, 162].

Preclinical studies in different animal species showed that the safety and tolerability of FP was good, however, with the exception that some data revealed a teratogenicity potential of FP



Figure 3.9 Chemical structure of favipiravir and its active form.

[153, 163]. The dosage that caused teratogenicity in preclinical animal studies was similar to the dose of the drug that was required for influenza treatment in humans. Although *in vitro* studies have demonstrated the potential of FP to induce FP-resistant influenza virus variants, *in vivo* the use of FP was only rarely associated with mutations [153], in contrast to BAM.

A phase 1 clinical study to investigate the safety, tolerability, and pharmacokinetics (PKs) of FP showed that doses ranging from 30 to 1600 mg/kg were well tolerated and led to a rapid absorption and elimination via urine as T-705M1, the major metabolite and hydroxide of T-705. Moreover, no serious adverse incidents were reported [164]. FP-associated adverse events included diarrhea (6.3%), nausea (0.8%), vomiting (0.5%), and elevated uric acid levels in the blood (5.6%) [164].

Various phase 2 and 3 studies have been performed with FP to determine the efficacy and safety for treatment of uncomplicated influenza. The efficacy of FP to reduce the time to alleviation of symptoms varied in the different trials with patients suffering from uncomplicated influenza. A dose regimen of 1800 mg BID on day 1 and 800 mg BID on days 2–5 applied via the oral route in a phase 2 and two phase 3 trials, resulted in a reduced time to alleviation of symptoms of only 15.0, 14.2, and 6.1 hours compared to placebo-treated patients [74, 140]. FP treatment led to reduction of viral load. In patients treated with FP, a 10-fold lower viral load was measured on days 2 and 3 of the study compared to placebo-treated patients. Virus samples isolated from patients before and after treatment during the clinical trials demonstrated amino acid substitutions in proteins of the polymerase complex of post-treatment viruses. These substitutions had no influence on viral fitness. *In vitro* experiments gave no clear answers to the question on the potency of FP to induce mutations. In one study, no specific resistance mutations that led to reduced FP susceptibility [166, 167].

Preclinical combination therapy studies of FP with oseltamivir have been performed in mice. The combination therapy showed a synergistic effect of FP and oseltamivir monotherapy and also allowed a prolonged treatment window [160]. The combination therapy was also successful to protect immunocompromised mice from a lethal infection, but viruses could be detected that developed oseltamivir-resistance via the classical H274Y NA mutation. In one study, the viruses remained sensitive to oseltamivir [168], while in another study, FP did not suppress the emergence of oseltamivir-resistant variants [169].

Based on the rather modest efficacy of FP to reduce the time of disease (time to alleviation of symptoms), the ability of FP to induce resistant virus variants and the teratogenicity concerns, it appears unlikely that FP will be further developed for use in uncomplicated influenza [74].

In the current COVID-19 pandemic (as of March 2021), FP has demonstrated antiviral activity against SARS-CoV-2 *in vitro* [170] and in an animal model [171] and is being investigated in several trials with different results. Benefit after FP treatment was shown in some studies in SARS-CoV-2 infected patients, while other trails demonstrated no advantage of FP treatment. Furthermore, there are contradictory statements on the dose justification, for FP use in patients infected with SARS-CoV-2 [172]. A meta-analysis revealed a significant clinical improvement in the Favipiravir group versus the control group during seven days after hospitalization (RR = 1.24, 95% CI: 1.09 –1.41; P = 0.001). However, Favipiravir appeared not to exert a significant beneficial effect in terms of mortality in the general group of patients with mild to moderate COVID-19 [173].

3.6.3 Pimodivir

Pimodivir, an oral drug, also known as JNJ-63623872 or VX-787, is a non-nucleoside polymerase inhibitor that inhibits viral gene transcription by preventing the PB2 subunit from binding the 7-methyl GTP cap structures on the host-capped RNA (Figure 3.10) [174, 175]. Pimodivir is metabolized by CYP 3A4, but has no effect on cytochrome P450 activity [140]. Due to structural differences in the PB2 cap-binding pocket, pimodivir is only effective against influenza A viruses. The antiviral activity can vary significantly depending on the influenza virus A subtypes [176], but it is inactive against influenza B viruses [6, 164]. Preclinical studies in cell culture and in mice have shown that pimodivir is effective against a diverse group of influenza A viruses including the H1N1pdm, H5N1, H7N9, and strains resistant to NAIs [177]. When pimodivir was given to mice as late as four days postinfection with H1N1pdm09 or H5N1, it was still able to protect against a lethal virus challenge [177]. Combination studies in mice using pimodivir and oseltamivir suggest a potential benefit of the combination [178].

In a phase 2a clinical challenge study of uncomplicated influenza, healthy volunteers were infected with an influenza A virus (H3N2). Four groups were treated once daily for a total of 5 days with different dose levels: 100 mg, 400 mg, loading dose 900/600 mg, and loading dose 1200/600 mg [178]. In the 1200/600 mg group of pimodivir-treated individuals, pimodivir significantly reduced viral shedding and clinical symptoms by 43.0 hours compared to placebo-treated individuals [178]. In this study, pimodivir was generally safe and well tolerated with no serious adverse events or adverse events [178].

A phase 2b clinical study, TOPAZ investigated the antiviral efficacy of pimodivir in influenza virus infected, but otherwise healthy patients suffering from uncomplicated influenza. Pimodivir 300 or 600 mg, or pimodivir 600 mg plus oseltamivir 75 mg was given twice daily for five consecutive days. Antiviral activity, safety, and PKs of pimodivir alone or in combination were evaluated [179]. Dose-related diarrhea (16.9%, 25/148) was the most commonly reported adverse event, with nausea (4.0%, 6/148) and vomiting (2.7%, 4/148) [179]. Viral load reduction was the primary endpoint of the study, demonstrating that 600 mg pimodivir given twice daily for five days was adequate to reduce viral load. Combination therapy (pimodivir/oseltamivir) was more effective than administration of pimodivir alone [179]. Pimodivir has recently received FDA Fast Track designation due to its potential to address an unmet medical need in those who develop influenza A infection



Pimodivir

Figure 3.10 Chemical structure of pimodivir.

complications [15]. Further clinical studies with pimodivir alone or in combination with oseltamivir in high-risk patients are ongoing [15].

The use of pimodivir in clinical studies has also been associated with the emergence of mutations in 11 of 172 patients (6.4%), resulting in virus variants with a several-fold decrease in susceptibility to the drug [179]. The mutated viruses did not affect the clinical conditions and the patients did not spread the mutated virus after treatment, still, this finding deserves attention [10]. The finding of pimodivir efficacy could not be confirmed in a hospital-based study, where pimodivir was combined with oseltamivir [140].

3.6.4 Summary

Next to NAIs, the polymerase-inhibitors represent a class of influenza antivirals, most of which show efficacy against influenza A and B virus strains. Problems and challenges found with NAIs seem to be also present with the polymerase-inhibitors: narrow treatment window, potential to resistance, and at present not approved for the treatment of severe influenza. Thus, new strategies are required.

3.7 Monoclonal Antibodies

As described above in detail, the current antivirals for influenza are suboptimal due to the limited therapeutic window and an increasing incidence of resistance [160, 180-182]. Consequently, new preventive and therapeutic strategies to tackle influenza are being sought. In this respect, monoclonal Abs (mAbs) got into focus. mAbs are highly specific and may have very limited off-target effects in line with a favorable safety profile [183–185]. They are in development for the treatment and prevention of various infectious diseases. One of these mAbs is palivizumab, which was licensed for the prevention of Respiratory Syncytial Virus (RSV) infection [186, 187]. The development of the technologies for the production of mAbs has also stimulated the use of novel mAbs-based therapies for influenza [188]. Studies of natural immune responses to influenza virus infection provided evidence that the HA is the main target of the postvaccination immune response. HA is composed of two domains: a immunodominant globular head (HA1), which varies between different strains and a stalk (HA2) that is relatively well-conserved between strains and subtypes [189]. Thus, a large number of mAbs are currently under development targeting the stem region of influenza A viruses since they should be effective against various influenza virus strains [190, 191]. Moreover, a non-neutralizing mAb, TCN032, targeting the viral matrix protein is in clinical trials as well. In a phase 2a challenge study, a single dose of TCN032 was found to be well tolerated [192]. In addition, patients with severe influenza A virus infection were treated with anti-influenza immune plasma in a phase 3 trial. However, the benefit was insufficient to justify the use of immune plasma for treating patients with severe influenza [193]. Thus, more specific antibodies are likely needed.

Three monoclonal antibodies, which have reached the phase 2 stage of development, MHAA4549A, MEDI8852, and VIS 410 will be described. Three other mAbs, CR626, CR8020, and CT-P27 [15], which are also in clinical development, will not be covered here, because of comparable results to the mAbs introduced in this chapter.

3.7.1 MHAA4549A

This is a human monoclonal IgG1 antibody targeting a highly conserved epitope on the stalk region of the HA [194]. The mAb was isolated by using an *in vivo* enrichment technique that identified neutralizing human antibodies. Antigen-specific plasma-cells (antibody-producing B cells) were identified, isolated, and enriched from human peripheral blood mononuclear cells (PBMCs). MHAA4549A was identified by screening only 840 human antibodies and has demonstrated a neutralizing activity against all tested seasonal human influenza A strains [195]. The antibody has two mechanisms of action. First, it binds to the stalk region of the HA and thus blocks the HA-mediated membrane fusion in the endosome. Second, MHAA4549A can recognize virus-infected cells and mediate antibody-dependent cell-mediated cytotoxicity. This leads to killing of the infected cells [194].

Preclinical studies demonstrated therapeutic efficacy in a mouse model against diverse influenza viruses even when administered 72 hours postinfection. Studies in ferrets demonstrated protection against an H5N1 virus challenge and synergism was observed in a combination treatment of MHAA4549A with oseltamivir at 48 hours postinfection [15, 195].

PKs of MHAA4549A were determined in mice and in cynomolgus monkeys using a single IV dose. The PK data in monkeys were used for projection of the human PK profile. Here, it was predicted that a single IV dose ranging from 15 to 45 mg/kg should achieve efficacious exposure in humans [196, 197]. Thus, clinical trials have been conducted to evaluate the efficacy of MHAA4549A but also safety and PKs. Moreover, studies were performed with MHAA4549A in combination with oseltamivir [194, 197, 198]. MHAA4549A was well tolerated with dose-proportional serum PKs, a long terminal half-life (21.9–24.6 days), and a slow clearance (152–240 ml/day) [198]. However, nasopharyngeal swab PKs were not dose proportional, which should guide future dose selection to achieve the high drug concentrations needed at the site of action for efficacy [194, 198]. MHAA4549A treatment significantly reduced viral loads in a human influenza virus challenge model [194]. In another clinical trial in adult participants hospitalized with severe influenza A, using a single IV dose of MHAA4549A in combination with oseltamivir vir indicated no advantage for any of the primary clinical outcomes evaluated and for the virologic outcomes evaluated when compared with the standard of care [15].

3.7.2 MEDI8852

MEDI8852 is a potent, broadly neutralizing investigational human IgG1 kappa mAb, derived from human memory B cells that also targets the HA stalk with a broad spectrum of activity by neutralizing all 16 influenza A HA subtypes [199] and lacking efficacy against influenza B. Preclinical studies in animal models have demonstrated efficacy for prophylaxis or therapy against influenza virus, including such with pandemic potential, when used alone or with oseltamivir. MEDI8852 treatment prevented death of mice and ferrets infected with a pandemic influenza virus strain, even when treatment was started up to three days postinfection. MEDI8852 treatment was also shown to be superior to oseltamivir [199]. Additionally, MEDI8852 blocked the transmission of influenza virus in a ferret model, a unique finding among influenza-specific mAbs [199, 200].

MEDI8852 was shown to be well tolerated in phase 1 clinical trials, when subjects received 250–3000 mg MEDI8852. The terminal half-life ranged from 19.4 to 22.6 days [201]. A phase 2a study was conducted to evaluate the safety of MEDI8852 in adults with acute, uncomplicated influenza. This study demonstrated an acceptable safety profile [202]. In 2016, MEDI8852 received Fast Track designation by the FDA that supports the drug for an advanced development and review [15, 203].

3.7.3 VIS410

VIS410 is a neutralizing IgG1 monoclonal antibody that is engineered like MHAA4549A/ MEDI8852. Here, a novel approach called *atomic interaction network analysis* was used [204]. This method characterizes and targets functionally conserved epitopes within the influenza HA glycoprotein. In the present approach, it targets the stem region of influenza virus HA glycoprotein and has demonstrated binding to both groups HAs of influenza A viruses [205]. VIS410 is broadly active against various influenza A virus strains. In preclinical studies, VIS410 was effective at protecting mice from a challenge with a pandemic strain, when VIS410 was administered prophylactically, 12 hours prior to infection (50 mg/ kg). In addition, a single therapeutic dose of 2, 10, or 50 mg/kg VIS-410 given 24 hours after infection protected almost 100% of mice, that were either infected with a NAI-susceptible or -resistant virus. Improved viral clearance and limited spread of virus in the lungs was found, when VIS410 was delivered prophylactically [206].

In a phase 1 clinical trial, the safety, tolerability, and PKs of VIS410 against Influenza A virus was investigated. The results were used to implement a mathematical modeling approach to investigate whether VIS410 could be used to treat seasonal influenza and to protect at-risk groups [207]. In this trial, VIS410 applied in the range of 2–50 mg/kg was found to be safe and well tolerated [207]. VIS410 was associated with adverse events, including diarrhea, nausea, and vomiting. In a phase 2a influenza A virus human challenge study, a single-dose IV VIS410 administration 24 hours after virus-inoculation has led to a statistically significant decrease in viral load compared with placebo. VIS410 was safe and well tolerated. This demonstrated that VIS410 treatment of patients with uncomplicated influenza may lead to favorable effects on virus replication and hopefully symptom resolution [208].

3.7.4 Summary

Monoclonal antibody preparations are a different approach compared to the NA and polymerase inhibitors harboring superior PK properties. With an average half-life of roughly three weeks, one dose would be sufficient to protect and could also control infection and clinical symptoms, even if administration occurs 48 hours after onset of clinical symptoms [199, 202, 206]. Moreover, mAb treatment did not result in mutation in the viral genome and in the appearance of resistant virus strains [196, 202, 208]. This makes these compounds potentially valuable prophylactic and therapeutic tools against influenza as monotherapy or in combination with either NA or polymerase inhibitors. However, further studies in patients with severe influenza or with patients at high risk need to be performed to finally conclude on the potential of mAbs.

3.8 Host-Targeting Candidates

In contrast to targeting the virus directly, as described above for various small molecule inhibitors and monoclonal antibodies, another strategy would be to interfere with host cell functions, which the virus needs to fulfill its replication. Small RNA viruses such as influenza viruses require extensive interactions with host-cell functions to support their life cycle. At the same time, the infected cell induces a machinery of defense mechanisms to fight the invader [209]. These processes are mediated by a variety of intracellular signaling cascades and present a great opportunity for pharmaceutical intervention. Since it is almost impossible for the virus to escape the cellular dependency, the formation of resistant viral variants should be a rare event. However, interfering with cellular functions could have an influence on cellular integrity and thus lead to adverse events. A successful antiviral will need to balance toxicity to the cell and high antiviral efficacy.

Here, three concepts that have reached phase 2 of clinical trials are summarized: (i) DAS181, a protein that cleaves the cellular receptor for the virus; (ii) nitazoxanide that interferes with the trafficking of the HA in the cell; and (iii) LASAG that functions as an NFkappaB inhibitor. Activation of the NFkappaB pathway is required by influenza virus to ensure RNP-export.

3.8.1 DAS181

DAS181, also known as Fludase, is a recombinant sialidase fusion protein composed of the catalytic domain of a bacterial sialidase (*Actinomyces viscosus*) and the epithelium-anchoring domain of the human protein amphiregulin [15, 210, 211]. DAS181 cleaves both the Neu5Ac a(2,3)- and Neu5Ac a(2,6)-Gal linkages of sialic acid, which are the primary receptors for both human and avian/equine influenza viruses to bind and enter the host cell [5]. This renders the host cells inaccessible to influenza virus infection [212]. It could be shown that DAS181 treatment removed about 90% of sialic acid receptors on human airway epithelial cells within 15 minutes after treatment [213]. At least three days were required until the amount of sialic acid receptors returned to normal [213]. DAS181 is administered by dry powder inhalation with microparticles of 5–10 µm in size. The particle size leads to a deposition of the drug to the upper and central respiratory tract, but not the lower respiratory tract [14].

Preclinical studies have demonstrated that DAS181 is functional against a broad range of human and avian influenza viruses, as well as parainfluenza viruses and human metapneumoviruses (A2, B1, and B2 strains) with no cellular toxicity [213–217]. After treatment with DAS181, mice could be protected from lethal influenza virus infection [218, 219]. In a randomized, double-blind, placebo-controlled phase 2 study, it was shown that DAS181 was effective in decreasing viral load and viral shedding in humans, with no significant adverse effects in the patients that received the treatment [220]. The active domain of the sialidase is derived from a bacterial species that is also part of the human flora. Thus, immunological reactions by the host should be unlikely [211]. Nevertheless, in a phase 1 clinical trial, it was observed that when treatment was performed for more than seven days, patients developed an immune response against the compound. Antibodies directed against DAS181 led to a reduced efficacy [15, 221]. DAS181 has also been tested in a phase

1 trial in adults with well-controlled asthma. While the generalizability of the results is limited because of a small sample size, the drug was associated with some adverse events and it was stated that caution should be employed when administering DAS181 to individuals with less stable reactive airway disease [222].

3.8.2 Nitazoxanide

Nitazoxanide (NTZ) was originally developed and licensed as an antiprotozoal/helminth drug for the treatment of enteritis caused by *Cryptosporidium* and *Giardia* infections [223]. NTZ is a compound of the thiazolide class (Figure 3.11) that is orally administered and rapidly deacetylated in the blood to tizoxanide, which represents the active metabolic form [224, 225]. In addition to its antiparasitic activity, NTZ is also effective against a broad range of bacterial and viral pathogens [226–230].

Since viruses have different replication strategies, the mode of action of NTZ against different viruses varies. The mechanism of action against influenza virus is achieved by interference of NTZ with the viral HA. NTZ inhibits the trafficking of the HA from the endoplasmic reticulum to the Golgi apparatus and thus prevents the assembly and release of viral particles from the host cell [224, 225].

Preclinical studies have demonstrated that NTZ has an antiviral activity against a range of different strains of influenza A and B virus strains from human and avian origin with IC₅₀₈ t.l ranging from 0.2 to 1.5 µg/mL [224, 227, 230, 231]. In addition, a combination of NTZ with oseltamivir demonstrated a synergistic effect against influenza A viruses, including oseltamivir-resistant virus [227]. The potential benefit of the combination of a compound targeting the virus directly (oseltamivir) with a host-targeting antiviral (NTZ) may be the improvement in the effectiveness in particular against oseltamivir-resistant strains and the reduction in the likelihood of selecting for resistance [14]. There are currently no published data available of the antiviral efficacy of NTZ in animal models [14, 15]. A large number of clinical studies in humans were performed with NTZ, demonstrating the potential of this repurposed drug as an antiviral against influenza [224]. Results from a phase 2b/3 human clinical trial with participants suffering of acute uncomplicated influenza demonstrated that treatment with 600 mg NTZ twice daily (started within 48 hours of symptom onset) for 5 days was associated with a reduction of the duration of symptoms of 21.2 hours. The drug was well tolerated and reduced the viral load [232]. Various phase 2 or 3 clinical trials with NTZ are currently underway, including a comparison to oseltamivir and an oseltamivir/NTZ combination therapy [14].



Nitazoxanide

Tizoxanide

Figure 3.11 Chemical structures of nitazoxanide and its active metabolite tizoxanide. *Source*: Radi et al. [244].

3.8.3 LASAG

LASAG (D,L-lysine acetylsalicylate \cdot glycine) also known as BAY 81-8781 is approved as Aspirin i.v. for IV application (Figure 3.12). In addition to LASAG's function as an inhibitor of cyclooxygenases (COX), it was described that acetylsalicylic acid (ASA) and other salicylates inhibit the activation of the NF-kappaB pathway in a low millimolar range [233–235].

Newly synthesized vRNP complexes must be exported from the nucleus into the cytoplasm [129], which requires the activation of the NF-kappaB pathway by Influenza A and B viruses [236]. In cell culture, it could be demonstrated that LASAG is effective against numerous influenza A and B virus strains including highly pathogenic avian strains [237, 238]. In the mouse infection model, pharmacological studies demonstrated that the oral route of administration was not suitable to reach the sufficient concentrations of LASAG for a successfully reduction of viral load in the lung. This could only be achieved by inhalation of LASAG [239]. Inhalation of pure ASA is clinically not suitable, since it causes respiratory irritation due to its acidic properties [240, 241]. However, the D,L-lysine increases the stability and tolerability of inhaled ASA, it prevents ASA from hydrolyzing and promotes the formation of a salt. Moreover, the addition of the glycine to ASA prevents discoloration and further leads to an increase of the stability.

Both lysine and glycine are essential amino acids and are considered to have no relevant pharmacodynamic or toxic effects. They present no risk to human health. As LASAG immediately dissociates into ASA, the pharmacodynamics of LASAG are equivalent to those of ASA [242]. In mice infected with a $5\times$ lethal dose of influenza A virus, LASAG treatment was still effective in protecting 50% of the animals from death when administration started as late as 48 hours after infection [239].

A phase 1 clinical trial demonstrated that inhalation of LASAG was well tolerated and did not lead to adverse events [243]. Patients with a reported duration of illness of less than 120 h were included in the study. Other inclusion criteria included: presence of at least one respiratory symptom (nasal congestion, sore throat, or cough) of any severity and hospital admission due to (suspected) influenza; presence of at least one constitutional symptom (aches/myalgia, fatigue, headache or feverishness/chills/sweats) of any severity and presence of fever (temperature \geq 38.0 °C orally, or \geq 38.5 °C rectally) at the time of screening. In a phase 2a clinical trial, hospitalized patients received inalation of LASAG for seven days, twice daily. LASAG significantly (17.2 hours) improved the time to alleviation of influenza symptoms in hospitalized patients compared to standard of care (oseltamivir) and demonstrated an influence on the viral load [243]. The LASAG group received 800 mg of LASAG/4 ml of fill dose, equivalent to 400 mg of ASA/4 ml of fill dose, resulting in an alveolar dose of 45 mg ASA. The placebo group received 4 ml saline solution (0.9% sodium chloride (NaCl))



LASAG - acetylsalicyclic acid DL lysine

Figure 3.12 Chemical structure of LASAG.

dissolved in water. This phase 2 proof-of-concept (PoC) study demonstrated that targeting an intracellular signaling pathway using aerosolized LASAG, which represents a different host cell strategy compared to DAS181 or nitazoxanide, improved influenza symptoms and reduced viral load. The mode of action of LASAG against influenza is not completely solved. Due to the fact that COX-inhibitors have anti-inflammatory properties, it might be that the influence of the disease outcome is due to an anti-inflammatory effect. However, in preclinical studies it could be demonstrated that indomethacin, a pure COX inhibitor, showed no inhibition of virus propagation and was not able to inhibit NFkappaB [237]. Also as LASAG therapy is expected to last only for seven days and the drug is inhaled, the exposure is low and typical side effects of oral ASA therapies are unlikely.

3.8.4 Summary

Host cell targets represent a potential, new strategy for the development of antivirals against influenza. One of the advantages is the reduced resistance formation due to the fact that the virus needs the particular host cell function in order to ensure its propagation. Nevertheless, targeting host cell components or host cell factors could lead to an increased risk for adverse events. This risk needs to be scrutinized in detail for each particular host cell target. However, as influenza only requires short-term interventions, it may be one of the viruses where this strategy may find a tolerable balance between attacking the virus and inhibiting essential host functions.

References

- **1** WHO (2017). Influenza in focus: up to 650000 people die of respiratory diseases linked to seasonal flu each year. http://www.who.int/en/news-room/detail/14-12-2017-up-to-650-000-people-die-of-respiratory-diseases-linked-to-seasonal-flu-each-year (accessed 9 January 2020).
- **2** Caini, S., Kusznierz, G., Garate, V.V. et al. (2019). The epidemiological signature of influenza B virus and its B/Victoria and B/Yamagata lineages in the 21st century. *PLoS ONE* 14 (9):e0222381. https://doi.org/10.1371/journal.pone.0222381.
- **3** Putri, W., Muscatello, D.J., Stockwell, M.S., and Newall, A.T. (2018). Economic burden of seasonal influenza in the United States. *Vaccine* 36: 3960–3966.
- **4** Erbelding, E.J., Post, D.J., Stemmy, E.J. et al. (2018). A universal influenza vaccine: the strategic plan for the National Institute of Allergy and Infectious Diseases. *J. Infect. Dis.* 218: 347–354.
- 5 Ito, T. (2000). Interspecies transmission and receptor recognition of influenza A viruses. *Microbiol. Immunol.* 44: 423–430.
- **6** Stevaert, A. and Naesens, L. (2016). The influenza virus polymerase complex: an update on its structure, functions, and significance for antiviral drug design. *Med. Res. Rev.* 36: 1127–1173.
- **7** Fodor, E. (2013). The RNA polymerase of influenza a virus: mechanisms of viral transcription and replication. *Acta Virol.* 57: 113–122.
- **8** Gabriel, G., Dauber, B., Wolff, T. et al. (2005). The viral polymerase mediates adaptation of an avian influenza virus to a mammalian host. *Proc. Natl. Acad. Sci. USA*. 102: 18590–18595.

- **9** Rossman, J.S. and Lamb, R.A. (2011). Influenza virus assembly and budding. *Virology* 411: 229–236.
- **10** Principi, N., Camilloni, B., Alunno, A. et al. (2019). Drugs for influenza treatment: is there significant news? *Front. Med. (Lausanne)* 6: 109.
- **11** Kim, H., Webster, R.G., and Webby, R.J. (2018). Influenza virus: dealing with a drifting and shifting pathogen. *Viral Immunol.* 31: 174–183.
- Webster, R.G. and Govorkova, E.A. (2014). Continuing challenges in influenza. *Ann. N. Y. Acad. Sci.* 1323: 115–139. https://www.who.int/influenza/vaccines/virus/ recommendations/2020-21_north/en/
- **13** Chua, J.V. and Chen, W.H. (2010). Bench-to-bedside review: vaccine protection strategies during pandemic flu outbreaks. *Crit. Care* 14: 218.
- **14** Koszalka, P., Tilmanis, D., and Hurt, A.C. (2017). Influenza antivirals currently in latephase clinical trial. *Influenza Other Respir. Viruses* 11: 240–246.
- **15** Behzadi, M.A. and Leyva-Grado, V.H. (2019). Overview of current therapeutics and novel candidates against influenza, respiratory syncytial virus, and middle east respiratory syndrome coronavirus infections. *Front. Microbiol.* 10: 1327.
- **16** Tellier, R. (2006). Review of aerosol transmission of influenza A virus. *Emerg. Infect. Dis.* 12: 1657–1662.
- 17 Herold, S., Becker, C., Ridge, K.M., and Budinger, G.R. (2015). Influenza virus-induced lung injury: pathogenesis and implications for treatment. *Eur. Respir. J.* 45: 1463–1478.
- 18 Brandes, M., Klauschen, F., Kuchen, S., and Germain, R.N. (2013). A systems analysis identifies a feedforward inflammatory circuit leading to lethal influenza infection. *Cell* 154: 197–212.
- **19** Dunning, J., Blankley, S., Hoang, L.T. et al. (2018). Progression of whole-blood transcriptional signatures from interferon-induced to neutrophil-associated patterns in severe influenza. *Nat. Immunol.* 19: 625–635.
- 20 Hogner, K., Wolff, T., Pleschka, S. et al. (2013). Macrophage-expressed IFN-beta contributes to apoptotic alveolar epithelial cell injury in severe influenza virus pneumonia. *PLoS Pathog.* 9: e1003188.
- **21** Peteranderl, C., Herold, S., and Schmoldt, C. (2016). Human influenza virus infections. *Semin. Respir. Crit. Care Med.* 37: 487–500.
- **22** de Jong, M.D., Simmons, C.P., Thanh, T.T. et al. (2006). Fatal outcome of human influenza A (H5N1) is associated with high viral load and hypercytokinemia. *Nat. Med.* 12: 1203–1207.
- **23** Droebner, K., Reiling, S.J., and Planz, O. (2008). Role of hypercytokinemia in NF-kappaB p50-deficient mice after H5N1 influenza A virus infection. *J. Virol.* 82: 11461–11466.
- **24** Ruf, B.R. and Knuf, M. (2014). The burden of seasonal and pandemic influenza in infants and children. *Eur. J. Pediatr.* 173: 265–276.
- **25** Lehners, N., Tabatabai, J., Prifert, C. et al. (2016). Long-term shedding of influenza virus, parainfluenza virus, respiratory syncytial virus and nosocomial epidemiology in patients with hematological disorders. *PLoS One* 11: e0148258.
- **26** Lessler, J., Reich, N.G., Cummings, D.A. et al. (2009). Outbreak of 2009 pandemic influenza A (H1N1) at a New York City school. *N. Engl. J. Med.* 361: 2628–2636.
- 27 (a) Mancini, D.A., Alves, R.C., Mendonca, R.M. et al. (2008). Influenza virus and proteolytic bacteria co-infection in respiratory tract from individuals presenting respiratory manifestations. *Rev. Inst. Med. Trop. Sao Paulo* 50: 41–46. (b) Klein, E. Y., Monteforte, B.,

Gupta, A., Jiang, W., May, L., Hsieh, Y., & Dugas, A. F. (2016). The Frequency of Influenza and Bacterial Co-infection: A systematic review and Meta-analysis. *Influenza and Other Respiratory Viruse*. http://dx.doi.org/10.1111/irv.12398.

- 28 von Baum, H., Schweiger, B., Welte, T. et al. (2011). How deadly is seasonal influenzaassociated pneumonia? The German Competence Network for Community-Acquired Pneumonia. *Eur. Respir. J.* 37: 1151–1157.
- **29** Schauwvlieghe, A., Rijnders, B.J.A., Philips, N. et al. (2018). Invasive aspergillosis in patients admitted to the intensive care unit with severe influenza: a retrospective cohort study. *Lancet. Respir. Med.* 6: 782–792.
- **30** Ison, M.G. (2017). Antiviral treatments. *Clin. Chest. Med.* 38: 139–153.
- **31** Welte, T. and Miravitlles, M. (2014). Viral, bacterial or both? Regardless, we need to treat infection in COPD. *Eur. Respir. J.* 44: 11–13.
- **32** Mallia, P. and Johnston, S.L. (2007). Influenza infection and COPD. *Int. J. Chron. Obstruct. Pulmon. Dis.* 2: 55–64.
- **33** Hurt, A.C. and Kelly, H. (2016). Debate regarding oseltamivir use for seasonal and pandemic influenza. *Emerg. Infect. Dis.* 22: 949–955.
- 34 Shah, S., McManus, D., Bejou, N. et al. (2019). 2645. Clinical outcomes of oseltamivir vs. baloxavir in patients hospitalized with influenza A. Open Forum Infect. Dis. 6:S925. https://doi.org/10.1093/ofid/ofz360.2323.
- 35 Gubareva, L.V., Mishin, V.P., Patel, M.C. et al. (2019). Assessing baloxavir susceptibility of influenza viruses circulating in the United States during the 2016/17 and 2017/18 seasons. *Euro. Surveill.* 24: 1–5.
- **36** Imai, M., Yamashita, M., Sakai-Tagawa, Y. et al. (2020). Influenza A variants with reduced susceptibility to baloxavir isolated from Japanese patients are fit and transmit through respiratory droplets. *Nat. Microbiol.* 5: 27–33.
- 37 Davidson, S. (2018). Treating influenza infection, from now and into the future. *Front. Immunol.* 9: 1946.
- **38** Planz, O. (2013). Development of cellular signaling pathway inhibitors as new antivirals against influenza. *Antivir. Res.* 98: 457–468.
- **39** van de Wakker, S.I., Fischer, M.J.E., and Oosting, R.S. (2017). New drug-strategies to tackle viral-host interactions for the treatment of influenza virus infections. *Eur. J. Pharmacol.* 809: 178–190.
- **40** Jefferson, T., Jones, M.A., Doshi, P. et al. (2014). Neuraminidase inhibitors for preventing and treating influenza in healthy adults and children. *Cochrane Database Syst. Rev.* 2014: CD008965.
- **41** Lamb, R.A., Zebedee, S.L., and Richardson, C.D. (1985). Influenza virus M2 protein is an integral membrane protein expressed on the infected-cell surface. *Cell* 40: 627–633.
- **42** Schnell, J.R. and Chou, J.J. (2008). Structure and mechanism of the M2 proton channel of influenza A virus. *Nature* 451: 591–595.
- **43** Wang, J., Pielak, R.M., McClintock, M.A., and Chou, J.J. (2009). Solution structure and functional analysis of the influenza B proton channel. *Nat. Struct. Mol. Biol.* 16: 1267–1271.
- **44** Pielak, R.M. and Chou, J.J. (2011). Influenza M2 proton channels. *Biochim. Biophys. Acta* 1808: 522–529.
- **45** Rosenberg, M.R. and Casarotto, M.G. (2010). Coexistence of two adamantane binding sites in the influenza A M2 ion channel. *Proc. Natl. Acad. Sci. USA* 107: 13866–13871.
- **46** McKimm-Breschkin, J.L., Jiang, S.B., Hui, D.S. et al. (2018). Prevention and treatment of respiratory viral infections: presentations on antivirals, traditional therapies and host-directed interventions at the 5th ISIRV Antiviral Group conference. *Antivir. Res.* 149: 118–142.
- **47** Davies, W.L., Grunert, R.R., Haff, R.F. et al. (1964). Antiviral activity of 1-adamantanamine (amantadine). *Science* 144: 862–863.
- **48** Monto, A.S. (2008). Antivirals and influenza: frequency of resistance. *Pediatr. Infect. Dis. J.* 27: S110–S112.
- **49** Deyde, V.M., Xu, X., Bright, R.A. et al. (2007). Surveillance of resistance to adamantanes among influenza A(H3N2) and A(H1N1) viruses isolated worldwide. *J. Infect. Dis.* 196: 249–257.
- **50** Hayden, F.G., Belshe, R.B., Clover, R.D. et al. (1989). Emergence and apparent transmission of rimantadine-resistant influenza A virus in families. *N. Engl. J. Med.* 321: 1696–1702.
- **51** Hayden, F.G., Sperber, S.J., Belshe, R.B. et al. (1991). Recovery of drug-resistant influenza A virus during therapeutic use of rimantadine. *Antimicrob. Agents Chemother.* 35: 1741–1747.
- **52** Sweet, C., Hayden, F.G., Jakeman, K.J. et al. (1991). Virulence of rimantadine-resistant human influenza A (H3N2) viruses in ferrets. *J. Infect. Dis.* 164: 969–972.
- 53 Gubareva, L.V., Trujillo, A.A., Okomo-Adhiambo, M. et al. (2010). Comprehensive assessment of 2009 pandemic influenza A (H1N1) virus drug susceptibility in vitro. *Antivir. Ther.* 15: 1151–1159.
- 54 Saito, R., Suzuki, Y., Li, D. et al. (2008). Increased incidence of adamantane-resistant influenza A(H1N1) and A(H3N2) viruses during the 2006–2007 influenza season in Japan. *J. Infect. Dis.* 197: 630–632; author reply 632–633.
- 55 Bright, R.A., Shay, D.K., Shu, B. et al. (2006). Adamantane resistance among influenza A viruses isolated early during the 2005–2006 influenza season in the United States. *JAMA* 295: 891–894.
- **56** Hurt, A.C., Selleck, P., Komadina, N. et al. (2007). Susceptibility of highly pathogenic A(H5N1) avian influenza viruses to the neuraminidase inhibitors and adamantanes. *Antivir. Res.* 73: 228–231.
- **57** Cheung, C.L., Rayner, J.M., Smith, G.J. et al. (2006). Distribution of amantadine-resistant H5N1 avian influenza variants in Asia. *J. Infect. Dis.* 193: 1626–1629.
- 58 Duwe, S. (2017). Influenza viruses antiviral therapy and resistance. *GMS Infect. Dis.* 5: Doc04.
- **59** CDC (2018). Influenza antiviral medications: summary for clinicians. https://www.cdc.gov/flu/professionals/antivirals/summary-clinicians.htm (accessed 5 January 2020).
- **60** Parra-Rojas, C., Nguyen, V.K., Hernandez-Mejia, G., and Hernandez-Vargas, E.A. (2018). Neuraminidase inhibitors in influenza treatment and prevention(-)is it time to call it a day? *Viruses* 10 (9): 454.
- **61** McKimm-Breschkin, J.L. (2013). Influenza neuraminidase inhibitors: antiviral action and mechanisms of resistance. *Influenza Other Respir. Virus.* 7 (Suppl 1): 25–36.
- **62** Palese, P., Bodo, G., and Tuppy, H. (1970). Quantitative determination of neuraminidaseactive foci in cell monolayer cultures infected with influenza or newcastle disease virus. *J. Virol.* 6: 556–558.

86 3 Antiviral Drugs Against Influenza Virus

- **63** Varghese, J.N., McKimm-Breschkin, J.L., Caldwell, J.B. et al. (1992). The structure of the complex between influenza virus neuraminidase and sialic acid, the viral receptor. *Proteins* 14: 327–332.
- **64** Gupta, Y.K., Meenu, M., and Mohan, P. (2015). The Tamiflu fiasco and lessons learnt. *Indian J. Pharm.* 47: 11–16.
- 65 Chaudhuri, S., Symons, J.A., and Deval, J. (2018). Innovation and trends in the development and approval of antiviral medicines: 1987–2017 and beyond. *Antivir. Res.* 155: 76–88.
- **66** Moscona, A. (2005). Neuraminidase inhibitors for influenza. *N. Engl. J. Med.* 353: 1363–1373.
- **67** Democratis, J., Pareek, M., and Stephenson, I. (2006). Use of neuraminidase inhibitors to combat pandemic influenza. *J. Antimicrob. Chemother.* 58: 911–915.
- **68** Gubareva, L.V., Kaiser, L., and Hayden, F.G. (2000). Influenza virus neuraminidase inhibitors. *Lancet* 355: 827–835.
- **69** Esposito, S. and Principi, N. (2016). Oseltamivir for influenza infection in children: risks and benefits. *Expert. Rev. Respir. Med.* 10: 79–87.
- **70** Oboho, I.K., Reed, C., Gargiullo, P. et al. (2016). Benefit of early initiation of influenza antiviral treatment to pregnant women hospitalized with laboratory-confirmed influenza. *J. Infect. Dis.* 214: 507–515.
- 71 Dobson, J., Whitley, R.J., Pocock, S., and Monto, A.S. (2015). Oseltamivir treatment for influenza in adults: a meta-analysis of randomised controlled trials. *Lancet* 385: 1729–1737.
- 72 McClellan, K. and Perry, C.M. (2001). Oseltamivir: a review of its use in influenza. *Drugs* 61: 263–283.
- 73 Pizzorno A., Bouhy, X., Abed, Y., Boivin, G. (2011). Generation and characterization of recombinant pandemic influenza A(H1N1) viruses resistant to neuraminidase inhibitors. *J. Infect. Dis.* 203 (1):25–31. https://doi.org/10.1093/infdis/jiq010.
- **74** Mifsud, E.J., Hayden, F.G., and Hurt, A.C. (2019). Antivirals targeting the polymerase complex of influenza viruses. *Antivir. Res.* 169: 104545.
- **75** Lackenby, A., Besselaar, T.G., Daniels, R.S. et al. (2018). Global update on the susceptibility of human influenza viruses to neuraminidase inhibitors and status of novel antivirals, 2016–2017. *Antivir. Res.* 157: 38–46.
- 76 Wang, K., Shun-Shin, M., Gill, P. et al. (2012). Neuraminidase inhibitors for preventing and treating influenza in children (published trials only). *Cochrane Database Syst. Rev.* 4. 2012;4:CD002744. PMID: 22258949.
- **77** He, G., Massarella, J., and Ward, P. (1999). Clinical pharmacokinetics of the prodrug oseltamivir and its active metabolite Ro 64-0802. *Clin. Pharmacokinet.* 37: 471–484.
- **78** Uyeki, T.M., Fowler, R.A., and Fischer, W.A. 2nd. Gaps in the clinical management of influenza: a century since the 1918 pandemic. *JAMA* 2018 (320): 755–756.
- **79** McKimm-Breschkin, J., Trivedi, T., Hampson, A. et al. (2003). Neuraminidase sequence analysis and susceptibilities of influenza virus clinical isolates to zanamivir and oseltamivir. *Antimicrob. Agents Chemother.* 47: 2264–2272.
- **80** Heneghan, C.J., Onakpoya, I., Jones, M.A. et al. (2016). Neuraminidase inhibitors for influenza: a systematic review and meta-analysis of regulatory and mortality data. *Health Technol. Assess.* 20: 1–242.

- 81 Muthuri, S.G., Venkatesan, S., Myles, P.R. et al. (2014). Effectiveness of neuraminidase inhibitors in reducing mortality in patients admitted to hospital with influenza A H1N1pdm09 virus infection: a meta-analysis of individual participant data. *Lancet Respir. Med.* 2: 395–404.
- Venkatesan, S., Myles, P.R., Leonardi-Bee, J. et al. (2017). Impact of outpatient neuraminidase inhibitor treatment in patients infected with influenza A(H1N1)pdm09 at high risk of hospitalization: an individual participant data metaanalysis. *Clin. Infect. Dis.* 64: 1328–1334.
- Kamali, A. and Holodniy, M. (2013). Influenza treatment and prophylaxis with neuraminidase inhibitors: a review. *Infect. Drug Resist.* 6: 187–198.
- Kiso, M., Mitamura, K., Sakai-Tagawa, Y. et al. (2004). Resistant influenza A viruses in children treated with oseltamivir: descriptive study. *Lancet* 364: 759–765.
- Memoli, M.J., Hrabal, R.J., Hassantoufighi, A. et al. (2010). Rapid selection of a transmissible multidrug-resistant influenza A/H3N2 virus in an immunocompromised host. *J. Infect. Dis.* 201: 1397–1403.
- 86 Zheng, Y., He, Y., Deng, J. et al. (2011). Hospitalized children with 2009 influenza a (H1N1) infection in Shenzhen, China, November–December 2009. *Pediatr. Pulmonol.* 46: 246–252.
- Dharan, N.J., Gubareva, L.V., Meyer, J.J. et al. (2009). Infections with oseltamivir-resistant influenza A(H1N1) virus in the United States. *JAMA* 301: 1034–1041.
- Hurt, A.C., Holien, J.K., Parker, M.W., and Barr, I.G. (2009). Oseltamivir resistance and the H274Y neuraminidase mutation in seasonal, pandemic and highly pathogenic influenza viruses. *Drugs* 69: 2523–2531.
- Esposito, S., Molteni, C.G., Colombo, C. et al. (2010). Oseltamivir-induced resistant pandemic A/H1N1 influenza virus in a child with cystic fibrosis and *Pseudomonas aeruginosa* infection. *J. Clin. Virol.* 48: 62–65.
- Adams, S.E., Lee, N., Lugovtsev, V.Y. et al. (2019). Effect of influenza H1N1 neuraminidase V116A and I117V mutations on NA activity and sensitivity to NA inhibitors. *Antivir. Res.* 169: 104539.
- Gaymard, A., Charles-Dufant, A., Sabatier, M. et al. (2016). Impact on antiviral resistance of E119V, I222L and R292K substitutions in influenza A viruses bearing a group 2 neuraminidase (N2, N3, N6, N7 and N9). *J. Antimicrob. Chemother.* 71: 3036–3045.
- **92** Zürcher, T., Yates, P.J., Daly, J. et al. (2006). Mutations conferring zanamivir resistance in human influenza virus N2 neuraminidases compromise virus fitness and are not stably maintained in vitro. *J. Antimicrob. Chemother.* 58: 723–732.
- Abed, Y., Baz, M., and Boivin, G. (2006). Impact of neuraminidase mutations conferring influenza resistance to neuraminidase inhibitors in the N1 and N2 genetic backgrounds. *Antivir. Ther.* 11: 971–976.
- Ryan, D.M., Ticehurst, J., and Dempsey, M.H. (1995). GG167 (4-guanidino-2,4-dideoxy-2,3-dehydro-*N*-acetylneuraminic acid) is a potent inhibitor of influenza virus in ferrets. *Antimicrob. Agents Chemother.* 39: 2583–2584.
- Hayden, F.G., Osterhaus, A.D., Treanor, J.J. et al. (1997). Efficacy and safety of the neuraminidase inhibitor zanamivir in the treatment of influenzavirus infections. GG167 Influenza Study Group. *N. Engl. J. Med.* 337: 874–880.
- von Itzstein, M., Wu, W.Y., Kok, G.B. et al. (1993). Rational design of potent sialidase-based inhibitors of influenza virus replication. *Nature* 363: 418–423.

88 3 Antiviral Drugs Against Influenza Virus

- **97** Gubareva, L.V., McCullers, J.A., Bethell, R.C., and Webster, R.G. (1998). Characterization of influenza A/HongKong/156/97 (H5N1) virus in a mouse model and protective effect of zanamivir on H5N1 infection in mice. *J. Infect. Dis.* 178: 1592–1596.
- Monto, A.S., Robinson, D.P., Herlocher, M.L. et al. (1999). Zanamivir in the prevention of influenza among healthy adults: a randomized controlled trial. *JAMA* 282: 31–35.
- LaForce, C., Man, C.Y., Henderson, F.W. et al. (2007). Efficacy and safety of inhaled zanamivir in the prevention of influenza in community-dwelling, high-risk adult and adolescent subjects: a 28-day, multicenter, randomized, double-blind, placebo-controlled trial. *Clin. Ther.* 29: 1579–1590; discussion 1577–1578.
- Heneghan, C.J., Onakpoya, I., Thompson, M. et al. (2014). Zanamivir for influenza in adults and children: systematic review of clinical study reports and summary of regulatory comments. *BMJ: Brit. Med. J.* 348: g2547.
- **101** Ison, M.G. (2013). Clinical use of approved influenza antivirals: therapy and prophylaxis. *Influenza Other Respir. Virus.* 7 (Suppl 1): 7–13.
- Kaiser, L., Henry, D., Flack, N.P. et al. (2000). Short-term treatment with zanamivir to prevent influenza: results of a placebo-controlled study. *Clin. Infect. Dis.* 30: 587–589.
- Marty, F.M., Vidal-Puigserver, J., Clark, C. et al. (2017). Intravenous zanamivir or oral oseltamivir for hospitalised patients with influenza: an international, randomised, double-blind, double-dummy, phase 3 trial. *Lancet Respir. Med.* 5: 135–146.
- McLaughlin, M.M., Skoglund, E.W., and Ison, M.G. (2015). Peramivir: an intravenous neuraminidase inhibitor. *Expert. Opin. Pharmacother.* 16: 1889–1900.
- Bantia, S., Arnold, C.S., Parker, C.D. et al. (2006). Anti-influenza virus activity of peramivir in mice with single intramuscular injection. *Antivir. Res.* 69: 39–45.
- Kim, C.U., Lew, W., Williams, M.A. et al. (1997). Influenza neuraminidase inhibitors possessing a novel hydrophobic interaction in the enzyme active site: design, synthesis, and structural analysis of carbocyclic sialic acid analogues with potent anti-influenza activity. *J. Am. Chem. Soc.* 119: 681–690.
- Alame, M.M., Massaad, E., and Zaraket, H. (2016). Peramivir: a novel intravenous neuraminidase inhibitor for treatment of acute influenza infections. *Front. Microbiol.* 7: 450. https://doi.org/10.3389/fmicb.2016.00450.
- Bantia, S., Parker, C.D., Ananth, S.L. et al. (2001). Comparison of the anti-influenza virus activity of RWJ-270201 with those of oseltamivir and zanamivir. *Antimicrob. Agents Chemother.* 45: 1162.
- Chairat, K., Tarning, J., White, N.J., and Lindegardh, N. (2013). Pharmacokinetic properties of anti-influenza neuraminidase inhibitors. *J. Clin. Pharmacol.* 53: 119–139.
- Amarelle, L., Lecuona, E., and Sznajder, J.I. (2017). Anti-influenza treatment: drugs currently used and under development. *Arch. Bronconeumol.* 53: 19–26.
- **111** Barroso, L., Treanor, J., Gubareva, L., and Hayden, F.G. (2005). Efficacy and tolerability of the oral neuraminidase inhibitor peramivir in experimental human influenza: randomized, controlled trials for prophylaxis and treatment. *Antivir. Ther.* 10: 901–910.
- Kohno, S., Yen, M.Y., Cheong, H.J. et al. (2011). Phase III randomized, double-blind study comparing single-dose intravenous peramivir with oral oseltamivir in patients with seasonal influenza virus infection. *Antimicrob. Agents Chemother.* 55: 5267–5276.

- **113** Kohno, S., Kida, H., Mizuguchi, M. et al. (2010). Efficacy and safety of intravenous peramivir for treatment of seasonal influenza virus infection. *Antimicrob. Agents Chemother.* 54: 4568–4574.
- **114** de Jong, M.D., Ison, M.G., Monto, A.S. et al. (2014). Evaluation of intravenous peramivir for treatment of influenza in hospitalized patients. *Clin. Infect. Dis.* 59: e172–e185.
- 115 Ison, M.G. (2011). Antivirals and resistance: influenza virus. Curr. Opin. Virol. 1: 563–573.
- **116** Leang, S.K., Kwok, S., Sullivan, S.G. et al. (2014). Peramivir and laninamivir susceptibility of circulating influenza A and B viruses. *Influenza Other Respir. Viruses* 8: 135–139.
- **117** Meijer, A., Rebelo-de-Andrade, H., Correia, V. et al. (2014). Global update on the susceptibility of human influenza viruses to neuraminidase inhibitors, 2012–2013. *Antivir. Res.* 110: 31–41.
- **118** Takashita, E., Ejima, M., Itoh, R. et al. (2014). A community cluster of influenza A(H1N1) pdm09 virus exhibiting cross-resistance to oseltamivir and peramivir in Japan, November to December 2013. *Euro. Surveill.* 19. (1):pii=20666. https://doi.org/10.2807/1560-7917. ES2014.19.1.20666.
- **119** Gubareva, L.V., Webster, R.G., and Hayden, F.G. (2001). Comparison of the activities of zanamivir, oseltamivir, and RWJ-270201 against clinical isolates of influenza virus and neuraminidase inhibitor-resistant variants. *Antimicrob. Agents Chemother.* 45: 3403–3408.
- **120** Memoli, M.J., Hrabal, R.J., Hassantoufighi, A. et al. (2010). Rapid selection of oseltamivirand peramivir-resistant pandemic H1N1 virus during therapy in 2 immunocompromised hosts. *Clin. Infect. Dis.* 50: 1252–1255.
- **121** Samson, M., Pizzorno, A., Abed, Y., and Boivin, G. (2013). Influenza virus resistance to neuraminidase inhibitors. *Antivir. Res.* 98: 174–185.
- 122 Koyama, K., Takahashi, M., Nakai, N. et al. (2010). Pharmacokinetics and disposition of CS-8958, a long-acting prodrug of the novel neuraminidase inhibitor laninamivir in rats. *Xenobiotica* 40: 207–216.
- 123 Yamashita, M., Tomozawa, T., Kakuta, M. et al. (2009). CS-8958, a prodrug of the new neuraminidase inhibitor R-125489, shows long-acting anti-influenza virus activity. *Antimicrob. Agents Chemother.* 53: 186–192.
- **124** Itoh, Y., Shinya, K., Kiso, M. et al. (2009). In vitro and in vivo characterization of new swine-origin H1N1 influenza viruses. *Nature* 460: 1021–1025.
- **125** Watanabe, A., Chang, S.-C., Kim, M.J. et al. (2010). Long-acting neuraminidase inhibitor laninamivir octanoate versus oseltamivir for treatment of influenza: a double-blind, randomized, noninferiority clinical trial. *Clin. Infect. Dis.* 51: 1167–1175.
- **126** Kubo, S., Tomozawa, T., Kakuta, M. et al. (2010). Laninamivir prodrug CS-8958, a long-acting neuraminidase inhibitor, shows superior anti-influenza virus activity after a single administration. *Antimicrob. Agents Chemother.* 54: 1256–1264.
- **127** Hurt, A.C., Lee, R.T., Leang, S.K. et al. (2011). Increased detection in Australia and Singapore of a novel influenza A(H1N1)2009 variant with reduced oseltamivir and zanamivir sensitivity due to a S247N neuraminidase mutation. *Euro. Surveill.* 16.
- **128** Watanabe, A. (2013). A randomized double-blind controlled study of laninamivir compared with oseltamivir for the treatment of influenza in patients with chronic respiratory diseases. *J. Infect. Chemother.* 19: 89–97.
- **129** Eisfeld, A.J., Neumann, G., and Kawaoka, Y. (2015). At the centre: influenza A virus ribonucleoproteins. *Nat. Rev. Microbiol.* 13: 28–41.

- 90 3 Antiviral Drugs Against Influenza Virus
 - **130** Guilligay, D., Tarendeau, F., Resa-Infante, P. et al. (2008). The structural basis for cap binding by influenza virus polymerase subunit PB2. *Nat. Struct. Mol. Biol.* 15: 500–506.
 - **131** Resa-Infante, P., Jorba, N., Coloma, R., and Ortin, J. (2011). The influenza virus RNA synthesis machine: advances in its structure and function. *RNA Biol.* 8: 207–215.
 - **132** Gabriel, G., Klingel, K., Otte, A. et al. (2011). Differential use of importin-alpha isoforms governs cell tropism and host adaptation of influenza virus. *Nat. Commun.* 2: 156.
 - Krug, R.M., Morgan, M.A., and Shatkin, A.J. (1976). Influenza viral mRNA contains internal N6-methyladenosine and 5'-terminal 7-methylguanosine in cap structures. *J. Virol.* 20: 45–53.
 - **134** Fodor, E., Pritlove, D.C., and Brownlee, G.G. (1994). The influenza virus panhandle is involved in the initiation of transcription. *J. Virol.* 68: 4092–4096.
 - **135** Lamb, R.A. and Choppin, P.W. (1983). The gene structure and replication of influenza virus. *Annu. Rev. Biochem.* 52: 467–506.
 - 136 Su, C.Y., Cheng, T.J., Lin, M.I. et al. (2010). High-throughput identification of compounds targeting influenza RNA-dependent RNA polymerase activity. *Proc. Natl. Acad. Sci. USA* 107: 19151–19156.
 - **137** Krammer, F., Smith, G.J.D., Fouchier, R.A.M. et al. (2018). Influenza. *Nat. Rev. Dis. Primers* 4: 3.
 - **138** Noshi, T., Kitano, M., Taniguchi, K. et al. (2018). In vitro characterization of baloxavir acid, a first-in-class cap-dependent endonuclease inhibitor of the influenza virus polymerase PA subunit. *Antivir. Res.* 160: 109–117.
 - **139** Fukao, K., Noshi, T., Yamamoto, A. et al. (2019). Combination treatment with the cap-dependent endonuclease inhibitor baloxavir marboxil and a neuraminidase inhibitor in a mouse model of influenza A virus infection. *J. Antimicrob. Chemother.* 74: 654–662.
 - **140** Hayden, F.G. and Shindo, N. (2019). Influenza virus polymerase inhibitors in clinical development. *Curr. Opin. Infect. Dis.* 32: 176–186.
 - **141** Koshimichi, H., Ishibashi, T., Kawaguchi, N. et al. (2018). Safety, tolerability, and pharmacokinetics of the novel anti-influenza agent baloxavir marboxil in healthy adults: phase I study findings. *Clin. Drug. Investig.* 38: 1189–1196.
 - **142** Hayden, F.G., Sugaya, N., Hirotsu, N. et al. (2018). Baloxavir marboxil for uncomplicated influenza in adults and adolescents. *N. Engl. J. Med.* 379: 913–923.
 - **143** Tomassini, J., Selnick, H., Davies, M.E. et al. (1994). Inhibition of cap (m7GpppXm)dependent endonuclease of influenza virus by 4-substituted 2,4-dioxobutanoic acid compounds. *Antimicrob. Agents Chemother.* 38: 2827–2837.
 - 144 Takashita, E., Morita, H., Ogawa, R. et al. (2018). Susceptibility of influenza viruses to the novel cap-dependent endonuclease inhibitor baloxavir marboxil. *Front. Microbiol.* 9: 3026.
 - 145 Ison, M.G., Portsmouth, S., Shishido, T. et al. (2018). LB16. Phase 3 trial of baloxavir marboxil in high-risk influenza patients (CAPSTONE-2 study). Open Forum Infect. Dis. 5 (Suppl 1): 2.
 - **146** Yang, T. (2019). Baloxavir marboxil: the first cap-dependent endonuclease inhibitor for the treatment of influenza. *Ann. Pharmacother.* 53: 754–759.
 - 147 Jones, J.C., Kumar, G., Barman, S. et al. (2018). Identification of the I38T PA substitution as a resistance marker for next-generation influenza virus endonuclease inhibitors. *mBio* 9 (2):e00430-18. https://doi.org/10.1128/mBio.00430-18.

- **148** Omoto, S., Speranzini, V., Hashimoto, T. et al. (2018). Characterization of influenza virus variants induced by treatment with the endonuclease inhibitor baloxavir marboxil. *Sci. Rep.* 8: 9633.
- **149** Koszalka, P., Tilmanis, D., Roe, M. et al. (2019). Baloxavir marboxil susceptibility of influenza viruses from the Asia-Pacific, 2012-2018. *Antivir. Res.* 164: 91–96.
- 150 Takashita, E., Kawakami, C., Morita, H. et al. (2019). On behalf of the influenza virus surveillance group of detection of influenza A(H3N2) viruses exhibiting reduced susceptibility to the novel cap-dependent endonuclease inhibitor baloxavir in Japan, December 2018. *Euro. Surveill.* 24 (3):1800698. https://doi.org/10.2807/1560-7917.ES.2019.24.3.1800698
- 151 Takashita, E., Kawakami, C., Ogawa, R. et al. (2019). Influenza A(H3N2) virus exhibiting reduced susceptibility to baloxavir due to a polymerase acidic subunit I38T substitution detected from a hospitalised child without prior baloxavir treatment, Japan, January 2019. *Euro. Surveill*. 24 (12):1900170. https://doi.org/10.2807/1560-7917. ES.2019.24.12.1900170.
- **152** Furuta, Y., Takahashi, K., Kuno-Maekawa, M. et al. (2005). Mechanism of action of T-705 against influenza virus. *Antimicrob. Agents Chemother.* 49: 981–986.
- (a) Furuta, Y., Komeno, T., and Nakamura, T. (2017). Favipiravir (T-705), a broad spectrum inhibitor of viral RNA polymerase. *Proc. Jpn. Acad. Ser. B Phys. Biol. Sci.* 93: 449–463. (b) Goldhill, D.H., Langat, P., Xie, H. et al. (2019). Determining the mutation bias of favipiravir in influenza virus using next-generation sequencing. *J. Virol.* 93 (2):e01217–e01218. http://doi.org/10.1128/JVI.01217-18.
- **154** Furuta, Y., Gowen, B.B., Takahashi, K. et al. (2013). Favipiravir (T-705), a novel viral RNA polymerase inhibitor. *Antivir. Res.* 100: 446–454.
- **155** Rocha-Pereira, J., Jochmans, D., Dallmeier, K. et al. (2012). Favipiravir (T-705) inhibits in vitro norovirus replication. *Biochem. Biophys. Res. Commun.* 424: 777–780.
- 156 Smither, S.J., Eastaugh, L.S., Steward, J.A. et al. (2014). Post-exposure efficacy of oral T-705 (favipiravir) against inhalational Ebola virus infection in a mouse model. *Antivir. Res.* 104: 153–155.
- 157 Stephenson, I., Democratis, J., Lackenby, A. et al. (2009). Neuraminidase inhibitor resistance after oseltamivir treatment of acute influenza A and B in children. *Clin. Infect. Dis.* 48: 389–396.
- **158** Watanabe, T., Kiso, M., Fukuyama, S. et al. (2013). Characterization of H7N9 influenza A viruses isolated from humans. *Nature* 501: 551–555.
- 159 Sidwell, R.W., Barnard, D.L., Day, C.W. et al. (2007). Efficacy of orally administered T-705 on lethal avian influenza A (H5N1) virus infections in mice. *Antimicrob. Agents Chemother.* 51: 845.
- **160** Marathe, B.M., Wong, S.-S., Vogel, P. et al. (2016). Combinations of oseltamivir and T-705 extend the treatment window for highly pathogenic influenza A(H5N1) virus infection in mice. *Sci. Rep.* 6: 26742.
- **161** Tarbet, E.B., Maekawa, M., Furuta, Y. et al. (2012). Combinations of favipiravir and peramivir for the treatment of pandemic influenza A/California/04/2009 (H1N1) virus infections in mice. *Antivir. Res.* 94: 103–110.
- 162 Smee, D.F., Hurst, B.L., Wong, M.H. et al. (2010). Effects of the combination of favipiravir (T-705) and oseltamivir on influenza A virus infections in mice. *Antimicrob. Agents Chemother.* 54: 126–133.

92 *3* Antiviral Drugs Against Influenza Virus

- **163** Nagata, T., Lefor, A.K., Hasegawa, M., and Ishii, M. (2015). Favipiravir: a new medication for the Ebola virus disease pandemic. *Disas. Med. Publ. Health Prepar.* 9: 79–81.
- **164** Kobayashi, O., Noto, M., Sakurai, T., and Iwamoto, A. (2008). T-705, a novel antiinfluenza virus compound – the safety, tolerability and pharmacokinetics in human. *Int. J. Infect. Dis.* 12: e299.
- **165** Baranovich, T., Wong, S.S., Armstrong, J. et al. (2013). T-705 (favipiravir) induces lethal mutagenesis in influenza A H1N1 viruses in vitro. *J. Virol.* 87: 3741–3751.
- **166** Ormond, L., Liu, P., Matuszewski, S. et al. (2017). The combined effect of oseltamivir and favipiravir on influenza A virus evolution. *Genome Biol. Evol.* 9: 1913–1924.
- **167** Goldhill, D.H., Te Velthuis, A.J.W., Fletcher, R.A. et al. (2018). The mechanism of resistance to favipiravir in influenza. *Proc. Natl. Acad. Sci. USA*. 115: 11613–11618.
- 168 Baz, M., Carbonneau, J., Rheaume, C. et al. (2018). Combination therapy with oseltamivir and favipiravir delays mortality but does not prevent oseltamivir resistance in immunodeficient mice infected with pandemic A(H1N1) influenza virus. *Viruses* 10 (11):610. https://doi.org/10.3390/v10110610. PMID: 30400276; PMCID: PMC6266789.
- 169 Kiso, M., Iwatsuki-Horimoto, K., Yamayoshi, S. et al. (2017). Emergence of oseltamivirresistant H7N9 influenza viruses in immunosuppressed cynomolgus macaques. J. Infect. Dis. 216: 582–593.
- **170** Li, G. and De Clercq, E. (2020). Therapeutic options for the 2019 novel coronavirus (2019-nCoV). Nat. Rev. Drug Discov. https://doi.org/10.1038/d41573-020-00016-0.
- 171 Driouich, J.S., Cochin, M., Lingas, G. et al. (2021). Favipiravir antiviral efficacy against SARS-CoV-2 in a hamster model. *Nat. Commun.* 12: 1735. https://doi.org/10.1038/ s41467-021-21992-w.
- 172 Eloy, P., Solas, C., Touret, F. et al. (2020). Dose rationale for Favipiravir use in patients infected with SARS-CoV-2. *Clin. Pharmacol. Ther.* 108: 188–188. https://doi.org/10.1002/ cpt.1877.
- Hassanipour, S., Arab-Zozani, M., Amani, B. et al. (2021). The efficacy and safety of Favipiravir in treatment of COVID-19: a systematic review and meta-analysis of clinical trials. *Sci. Rep.* 11: 11022. https://doi.org/10.1038/s41598-021-90551-6.
- 174 Clark, M.P., Ledeboer, M.W., Davies, I. et al. (2014). Discovery of a novel, first-in-class, orally bioavailable azaindole inhibitor (VX-787) of influenza PB2. *J. Med. Chem.* 57: 6668–6678.
- 175 Boyd, M.J., Bandarage, U.K., Bennett, H. et al. (2015). Isosteric replacements of the carboxylic acid of drug candidate VX-787: effect of charge on antiviral potency and kinase activity of azaindole-based influenza PB2 inhibitors. *Bioorg. Med. Chem. Lett.* 25: 1990–1994.
- 176 Smee, D.F., Barnard, D.L., and Jones, S.M. (2016). Activities of JNJ63623872 and oseltamivir against influenza A H1N1pdm and H3N2 virus infections in mice. *Antivir. Res.* 136: 45–50.
- 177 Byrn, R.A., Jones, S.M., Bennett, H.B. et al. (2015). Preclinical activity of VX-787, a first-in-class, orally bioavailable inhibitor of the influenza virus polymerase PB2 subunit. *Antimicrob. Agents Chemother.* 59: 1574–1587.
- **178** Trevejo, J.M., Asmal, M., Vingerhoets, J. et al. (2018). Pimodivir treatment in adult volunteers experimentally inoculated with live influenza virus: a Phase IIa, randomized, double-blind, placebo-controlled study. *Antivir. Ther.* 23: 335–344.

- 179 Finberg, R.W., Lanno, R., Anderson, D. et al. (2018). Phase 2b study of pimodivir (JNJ-63623872) as monotherapy or in combination with oseltamivir for treatment of acute uncomplicated seasonal influenza A: TOPAZ trial. J. Infect. Dis. 219: 1026–1034.
- **180** Garcia-Sastre, A. (2006). Antiviral response in pandemic influenza viruses. *Emerg. Infect. Dis.* 12: 44–47.
- **181** Beigel, J. and Bray, M. (2008). Current and future antiviral therapy of severe seasonal and avian influenza. *Antivir. Res.* 78: 91–102.
- **182** Nogales, A., Piepenbrink, M.S., Wang, J. et al. (2018). A highly potent and broadly neutralizing H1 influenza-specific human monoclonal antibody. *Sci. Rep.* 8: 4374.
- **183** Reichert, J.M. (2017). Antibodies to watch in 2017. *MAbs* 9: 167–181.
- **184** Catapano, A.L. and Papadopoulos, N. (2013). The safety of therapeutic monoclonal antibodies: implications for cardiovascular disease and targeting the PCSK9 pathway. *Atherosclerosis* 228: 18–28.
- 185 Singh, J.A., Wells, G.A., Christensen, R. et al. (2011). Adverse effects of biologics: a network meta-analysis and cochrane overview. *Cochrane Database Syst. Rev.* 2011: CD008794.
- **186** Shriver, Z., Trevejo, J.M., and Sasisekharan, R. (2015). Antibody-based strategies to prevent and treat influenza. *Front. Immunol.* 6.
- **187** Sparrow, E., Friede, M., Sheikh, M. et al. (2016). Passive immunization for influenza through antibody therapies, a review of the pipeline, challenges and potential applications. *Vaccine* 34: 5442–5448.
- **188** Jin, Y., Lei, C., Hu, D. et al. (2017). Human monoclonal antibodies as candidate therapeutics against emerging viruses. *Front. Med.* 11: 462–470.
- **189** Wang, T.T. and Palese, P. (2011). Catching a moving target. *Science* 333: 834.
- **190** Corti, D., Voss, J., Gamblin, S.J. et al. (2011). A neutralizing antibody selected from plasma cells that binds to Group 1 and Group 2 influenza A hemagglutinins. *Science* 333: 850.
- **191** DiLillo, D.J., Palese, P., Wilson, P.C., and Ravetch, J.V. (2016). Broadly neutralizing anti-influenza antibodies require Fc receptor engagement for in vivo protection. *J. Clin. Invest.* 126: 605–610.
- **192** Ramos, E.L., Mitcham, J.L., Koller, T.D. et al. (2015). Efficacy and safety of treatment with an anti-m2e monoclonal antibody in experimental human influenza. *J. Infect. Dis.* 211: 1038–1044.
- **193** Beigel, J.H., Aga, E., Elie-Turenne, M.C. et al. (2019). Anti-influenza immune plasma for the treatment of patients with severe influenza A: a randomised, double-blind, phase 3 trial. *Lancet Respir. Med.* 7: 941–950.
- **194** McBride, J.M., Lim, J.J., Burgess, T. et al. (2017). Phase 2 randomized trial of the safety and efficacy of MHAA4549A, a broadly neutralizing monoclonal antibody, in a human influenza A virus challenge model. *Antimicrob. Agents Chemother.* 61: e01154-17.
- 195 Nakamura, G., Chai, N., Park, S. et al. (2013). An in vivo human-plasmablast enrichment technique allows rapid identification of therapeutic influenza A antibodies. *Cell Host Microbe* 14: 93–103.
- Gupta, P., Kamath, A.V., Park, S. et al. (2016). Preclinical pharmacokinetics of MHAA4549A, a human monoclonal antibody to influenza A virus, and the prediction of its efficacious clinical dose for the treatment of patients hospitalized with influenza A. *mAbs* 8: 991–997.

- 94 3 Antiviral Drugs Against Influenza Virus
 - Lim, J.J., Deng, R., Derby, M.A. et al. (2016). Two phase 1, randomized, double-blind, placebo-controlled, single-ascending-dose studies to investigate the safety, tolerability, and pharmacokinetics of an anti-influenza A virus monoclonal antibody, MHAA4549A, in healthy volunteers. *Antimicrob. Agents Chemother*. 60: 5437–5444.
 - Deng, R., Lee, A.P., Maia, M. et al. (2018). Pharmacokinetics of MHAA4549A, an anti-influenza A monoclonal antibody, in healthy subjects challenged with influenza A virus in a phase IIa randomized trial. *Clin. Pharmacokinet.* 57: 367–377.
 - Kallewaard, N.L., Corti, D., Collins, P.J. et al. (2016). Structure and function analysis of an antibody recognizing all influenza A subtypes. *Cell* 166: 596–608.
 - Paules, C.I., Lakdawala, S., McAuliffe, J.M. et al. (2017). The hemagglutinin A stem antibody MEDI8852 prevents and controls disease and limits transmission of pandemic influenza viruses. *J. Infect. Dis.* 216: 356–365.
 - Mallory, R.M., Ali, S.O., Takas, T. et al. (2017). A phase 1 study to evaluate the safety and pharmacokinetics of MEDI8852, an anti-influenza A monoclonal antibody, in healthy adult volunteers. *Biologicals* 50: 81–86.
 - Ali, O., Takas, T., Nyborg, A. et al. (2017). A phase 2a study to evaluate the safety of MEDI8852 in outpatient adults with acute, uncomplicated influenza A. *Open Forum Infect. Dis.* 4: S519–S519.
 - Ali, S.O., Takas, T., Nyborg, A. et al. (2018). Evaluation of MEDI8852, an anti-influenza A monoclonal antibody, in treating acute uncomplicated influenza. *Antimicrob. Agents Chemother.* 62.
 - Viswanathan, K., Shriver, Z., and Babcock, G.J. (2015). Amino acid interaction networks provide a new lens for therapeutic antibody discovery and anti-viral drug optimization. *Curr. Opin. Virol.* 11: 122–129.
 - Tharakaraman, K., Subramanian, V., Viswanathan, K. et al. (2015). A broadly neutralizing human monoclonal antibody is effective against H7N9. *Proc. Natl. Acad. Sci.* 112: 10890.
 - Baranovich, T., Jones, J.C., Russier, M. et al. (2016). The hemagglutinin stem-binding monoclonal antibody VIS410 controls influenza virus-induced acute respiratory distress syndrome. *Antimicrob. Agents Chemother.* 60: 2118–2131.
 - 207 Wollacott, A.M., Boni, M.F., Szretter, K.J. et al. (2016). Safety and upper respiratory pharmacokinetics of the hemagglutinin stalk-binding antibody VIS410 support treatment and prophylaxis based on population modeling of seasonal influenza A outbreaks. *Ebiomedicine* 5: 147–155.
 - Hershberger, E., Sloan, S., Narayan, K. et al. (2019). Safety and efficacy of monoclonal antibody VIS410 in adults with uncomplicated influenza A infection: results from a randomized, double-blind, phase-2, placebo-controlled study. *Ebiomedicine* 40: 574–582.
 - Ludwig, S., Pleschka, S., Planz, O., and Wolff, T. (2006). Ringing the alarm bells: signalling and apoptosis in influenza virus infected cells. *Cell. Microbiol.* 8: 375–386.
 - Salvatore, M., Satlin, M.J., Jacobs, S.E. et al. (2016). DAS181 for treatment of parainfluenza virus infections in hematopoietic stem cell transplant recipients at a single center. *Biol. Blood Marrow Transpl.* 22: 965–970.
 - Malakhov, M.P., Aschenbrenner, L.M., Smee, D.F. et al. (2006). Sialidase fusion protein as a novel broad-spectrum inhibitor of influenza virus infection. *Antimicrob. Agents Chemother.* 50: 1470–1479.

- Triana-Baltzer, G.B., Sanders, R.L., Hedlund, M. et al. (2011). Phenotypic and genotypic characterization of influenza virus mutants selected with the sialidase fusion protein DAS181. *J. Antimicrob. Chemother.* 66: 15–28.
- Triana-Baltzer, G.B., Babizki, M., Chan, M.C. et al. (2010). DAS181, a sialidase fusion protein, protects human airway epithelium against influenza virus infection: an in vitro pharmacodynamic analysis. *J. Antimicrob. Chemother.* 65: 275–284.
- Triana-Baltzer, G.B., Gubareva, L.V., Klimov, A.I. et al. (2009). Inhibition of neuraminidase inhibitor-resistant influenza virus by DAS181, a novel sialidase fusion protein. *PLoS One* 4: e7838.
- Drozd, D.R., Limaye, A.P., Moss, R.B. et al. (2013). DAS181 treatment of severe parainfluenza type 3 pneumonia in a lung transplant recipient. *Transpl. Infect. Dis.* 15: E28–E32.
- Chan, R.W., Chan, M.C., Wong, A.C. et al. (2009). DAS181 inhibits H5N1 influenza virus infection of human lung tissues. *Antimicrob. Agents Chemother*. 53: 3935–3941.
- Thammawat, S., Sadlon, T.A., Adamson, P., and Gordon, D.L. (2015). Effect of sialidase fusion protein (DAS 181) on human metapneumovirus infection of Hep-2 cells. *Antivir. Chem. Chemother.* 24: 161–165.
- Belser, J.A., Lu, X., Szretter, K.J. et al. (2007). DAS181, a novel sialidase fusion protein, protects mice from lethal avian influenza H5N1 virus infection. *J. Infect. Dis.* 196: 1493–1499.
- Marjuki, H., Mishin, V.P., Chesnokov, A.P. et al. (2014). An investigational antiviral drug, DAS181, effectively inhibits replication of zoonotic influenza A virus subtype H7N9 and protects mice from lethality. *J. Infect. Dis.* 210: 435–440.
- Moss, R.B., Hansen, C., Sanders, R.L. et al. (2012). A phase II study of DAS181, a novel host directed antiviral for the treatment of influenza infection. *J. Infect. Dis.* 206: 1844–1851.
- Zenilman, J.M., Fuchs, E.J., Hendrix, C.W. et al. (2015). Phase 1 clinical trials of DAS181, an inhaled sialidase, in healthy adults. *Antivir. Res.* 123: 114–119.
- Colombo, R.E., Fiorentino, C., Dodd, L.E. et al. (2016). A phase 1 randomized, doubleblind, placebo-controlled, crossover trial of DAS181 (Fludase(R)) in adult subjects with well-controlled asthma. *BMC Infect. Dis.* 16: 54.
- White, C.A. Jr. (2004). Nitazoxanide: a new broad spectrum antiparasitic agent. *Expert Rev. Anti-Infect. Ther.* 2: 43–49.
- Rossignol, J.F., Elfert, A., El-Gohary, Y., and Keeffe, E.B. (2009). Improved virologic response in chronic hepatitis C genotype 4 treated with nitazoxanide, peginterferon, and ribavirin. *Gastroenterology* 136: 856–862.
- Broekhuysen, J., Stockis, A., Lins, R.L. et al. (2000). Nitazoxanide: pharmacokinetics and metabolism in man. *Int. J. Clin. Pharmacol. Ther.* 38: 387–394.
- Hoffman, P.S., Sisson, G., Croxen, M.A. et al. (2007). Antiparasitic drug nitazoxanide inhibits the pyruvate oxidoreductases of *Helicobacter pylori*, selected anaerobic bacteria and parasites, and *Campylobacter jejuni*. *Antimicrob*. *Agents Chemother*. 51: 868–876.
- Belardo, G., Cenciarelli, O., La Frazia, S. et al. (2015). Synergistic effect of nitazoxanide with neuraminidase inhibitors against influenza A viruses in vitro. *Antimicrob. Agents Chemother.* 59: 1061–1069.

3 Antiviral Drugs Against Influenza Virus

- Tilmanis, D., van Baalen, C., Oh, D.Y. et al. (2017). The susceptibility of circulating human influenza viruses to tizoxanide, the active metabolite of nitazoxanide. *Antivir. Res.* 147: 142–148.
- Stachulski, A.V., Santoro, M.G., Piacentini, S. et al. (2018). Second-generation nitazoxanide derivatives: thiazolides are effective inhibitors of the influenza A virus. *Future Med. Chem.* 10: 851–862.
- Rossignol, J.F. (2014). Nitazoxanide: a first-in-class broad-spectrum antiviral agent. *Antivir. Res.* 110: 94–103.
- Sleeman, K., Mishin, V.P., Guo, Z. et al. (2014). Antiviral susceptibility of variant influenza A(H3N2)v viruses isolated in the United States from 2011 to 2013. *Antimicrob. Agents Chemother.* 58: 2045–2051.
- Haffizulla, J., Hartman, A., Hoppers, M. et al. (2014). Effect of nitazoxanide in adults and adolescents with acute uncomplicated influenza: a double-blind, randomised, placebo-controlled, phase 2b/3 trial. *Lancet Infect. Dis.* 14: 609–618.
- Yin, M.J., Yamamoto, Y., and Gaynor, R.B. (1998). The anti-inflammatory agents aspirin and salicylate inhibit the activity of I(kappa)B kinase-beta. *Nature* 396: 77–80.
- Kopp, E. and Ghosh, S. (1994). Inhibition of NF-kappa B by sodium salicylate and aspirin. *Science* 265: 956–959.
- Grilli, M., Pizzi, M., Memo, M., and Spano, P. (1996). Neuroprotection by aspirin and sodium salicylate through blockade of NF-kappaB activation. *Science* 274: 1383–1385.
- Wurzer, W.J., Ehrhardt, C., Pleschka, S. et al. (2004). NF-kappaB-dependent induction of tumor necrosis factor-related apoptosis-inducing ligand (TRAIL) and Fas/FasL is crucial for efficient influenza virus propagation. *J. Biol. Chem.* 279: 30931–30937.
- Mazur, I., Wurzer, W.J., Ehrhardt, C. et al. (2007). Acetylsalicylic acid (ASA) blocks influenza virus propagation via its NF-kappaB-inhibiting activity. *Cell. Microbiol.* 9: 1683–1694.
- Kumar, N., Xin, Z.T., Liang, Y., and Ly, H. (2008). NF-kappaB signaling differentially regulates influenza virus RNA synthesis. *J. Virol.* 82: 9880–9889.
- Droebner, K., Haasbach, E., Dudek, S.E. et al. (2017). Pharmacodynamics, pharmacokinetics, and antiviral activity of BAY 81-8781, a novel NF-kappaB inhibiting anti-influenza drug. *Front. Microbiol.* 8: 2130.
- Le Pham, D., Lee, J.H., and Park, H.S. (2017). Aspirin-exacerbated respiratory disease: an update. *Curr. Opin. Pulm. Med.* 23: 89–96.
- 241 Hoak, J.C. (1983). Mechanisms of action: aspirin. Thromb. Res. Suppl. 4: 47-51.
- Kanani, K., Gatoulis, S.C., and Voelker, M. (2015). Influence of differing analgesic formulations of aspirin on pharmacokinetic parameters. *Pharmaceutics* 7: 188–198.
- Scheuch, G., Canisius, S., Nocker, K. et al. (2018). Targeting intracellular signaling as an antiviral strategy: aerosolized LASAG for the treatment of influenza in hospitalized patients. *Emerg. Microbes Infect.* 7: 21.
- Radi, A.-E., El-Ghany, N.A., and Wahdan, T. (2011). Anodic voltammetric methods for determination of the antiparasitic drug nitazoxanide in bulk form, pharmaceutical formulation, and its metabolite tizoxanide in human serum. *Monatshefte Für Chemie Chemical Monthly*, 143(5): 697–702.

Farah Elawar¹, Matthias Gotte¹, Nelson Lee², Rachel Fearns³, and David J. Marchant¹

¹Department of Medical Microbiology and Immunology, Li Ka Shing Institute of Virology, University of Alberta, Edmonton, Alberta, Canada

²Department of Medicine, Division of Infectious Diseases, University of Alberta, Edmonton, Alberta, Canada
³Department of Microbiology, School of Medicine, Boston University, MA, United States

4.1 Introduction

4

In the past decade, respiratory syncytial virus (RSV) therapeutics and vaccine research have increased significantly but with no new therapeutics' licensing since the FDA licensed Palivizumab in 1998. However, recently, there has been some progress in the development of an RSV vaccine, with nine candidate vaccines past phase 1 clinical trials. A recent clinical trial of a nanoparticle RSV-F vaccine demonstrated at least secondary endpoint efficacy; it reduced infant hospitalizations and hypoxia due to RSV infection [1]. Clinical trials of an experimental prefusion RSV-F vaccine have also demonstrated some partial success by the generation of IgA in the airway mucosae, but with mixed results between trial participants [2]. Furthermore, there are at least eight candidate RSV antivirals in clinical trials. These range from fusion inhibitors, nucleoside analogues, antibodies, and small interfering RNA (siRNA).

This chapter will discuss the healthcare burden posed by RSV, updated morbidity, and mortality data, and evasion of adaptive immunity by RSV. We will discuss how these aspects of RSV infection have likely hampered therapeutic and vaccine development. Among the therapeutic approaches that are currently in development, we will address primarily inhibitors of the fusion process between the viral and cellular membranes as well as nucleoside analogues and the importance of their structure.

4.1.1 The Burden of RSV Infection on Human Health

In healthy adults, RSV infection ranges from common cold-like symptoms to "full-blown flu," so the vast majority of these cases are subclinical and not reported. Hospitalization is likely when RSV infects the lower respiratory tract after establishing infection in the naso-pharynx. Lower respiratory tract infections are the leading cause of death in infants [3].

RSV is a significant contributor to this healthcare burden and is the second leading cause of death worldwide in children, second to malaria [4]. These infections, however, should not be considered as mutually exclusive. The current theory on the significant lethality of RSV is that it pushes those with an already high disease load, "over the edge" [5, 6], a theory that has been outlined in two publications by Caballero et al. [7–9] accurately labeling RSV an "opportunistic killer." This is particularly true for those with congenital heart and lung defects, and those with Down's syndrome. In developing nations and those regions plagued by a massive burden on health, natural and human-caused, there is a high opportunistic parasite and infection burden in the general population. Children in lower socioeconomic status residences are also more likely to succumb to RSV infection [7].

RSV thus is a significant infection that further complicates the health of those with other coinfections and maladies, including one of or a combination of tuberculosis, atypical mycobacterial infections, cholera, typhus, schistosomiasis, dengue virus infection, and malaria. All of the infections mentioned above do not always occur in isolation but sometimes as compounded pathogenesis. Thus, an acute RSV infection of the lower respiratory tract can be an overwhelming factor that kills those who are already suffering from other diseases.

The worldwide Pneumonia Etiology Research for Child Health (PERCH) studies on the causes of severe pneumonia showcase RSV as a critical pathogen that requires public health attention and investment. In the updated 2019 publication, the PERCH study examined children admitted to the hospital due to severe pneumonia. All hospitalized children in the study were age-matched with healthy controls from random communities around the study site. The study demonstrated that RSV was the leading pathogen causing severe pneumonia in 31% of all tested cases [10]. Overall, viruses made up 61% as the cause for severe pneumonia exceeding bacteria, which accounted for 27% of the cases. The study also noted that RSV was the most frequent virus detected in negative controls, supporting the hypothesis that RSV is an ubiquitous and opportunistic pathogen.

4.1.2 RSV Transmission

In developed nations, the virus is spread through the community and in nosocomial settings by aerosols, droplets, and by transmission from contaminated surfaces [11–14]. Standard hygienic measures, like shielding coughs and sneezes, hand washing, and social distancing play an essential role in limiting the spread of infections [15].

In 1981, Hall et al. showed the most common modes of transmission of RSV comes from close contact with large droplets or inoculation after touching contaminated surfaces [16]. Small aerosolized particles did not contribute to RSV spread and infection. These findings were then corroborated as well; more recent studies have shown that RSV can be transmitted through large droplet aerosols [11, 14].

Studies on RSV transmission in Kenya [17] and Italy [18] have identified older schoolaged siblings that bring RSV home with them as a source of infant RSV.

4.1.3 The Current Therapeutic Options and Opportunity for the Development of Treatments for RSV Infection

In infected individuals, the peak of RSV viral load reaches a slow crescendo that abates slowly. This is in contrast to influenza that must be treated with direct-acting antivirals at

the onset of symptoms for therapeutics to be effective [19, 20]. For RSV, there is a sufficiently long therapeutic window in which to identify and treat RSV infection. The peak of viral load typically happens on day 6–8 after infection with RSV, in contrast to 4 days postinfection with other respiratory viruses [21].

Currently, when a patient is admitted to the hospital with RSV, the hospital can only provide supportive care such as supplemental oxygen, mechanical ventilation, and fluid replacement. There are no suitable antivirals for the treatment of RSV. Although Ribavirin is licensed for use in RSV infection, it is hardly used because of a lack of efficacy and numerous side effects. There is no vaccine for RSV but one available prophylaxis is a monoclonal antibody called Palivizumab. This drug is costly and is only partially effective at preventing hospitalization of infants with RSV (reducing hospitalizations by about 50%). Of note, Palivizumab is only effective at preventing infection; it does not provide any therapeutic relief in an already established RSV infection. Palivizumab has been licensed since 1997 [22].

4.1.4 The Unique Challenge of RSV Vaccine Development

A vaccine is needed to at least partially counter the considerable health burden posed by RSV. Any such vaccine will likely still need to be supported by using monoclonal antibody prophylaxis and antiviral therapeutics to protect in cases of vaccine breakthrough. Vaccines have all remained elusive since the discovery of RSV 70 years ago. As discussed above, only the monoclonal antibody Palivizumab has been licensed for the prevention of severe RSV infection in the most vulnerable infants—usually those born prematurely and/or with other underlying conditions.

Relative to other viral infectious diseases, RSV has posed some significant challenges, among others, due to its apparent ability to evade the immune response and immune memory. This novel aspect of RSV immunology compared with other respiratory viruses like influenza, rhinoviruses, coronaviruses, and others are complicating vaccine development. Therefore, unlike other viruses, protection from RSV infection will likely require nontraditional vaccine approaches. However, apart from a few exceptions, the bulk of the RSV vaccine candidates in clinical trials is still based on traditional vaccine strategies. In the absence of pressure from immune memory, reinfection by the same strain can occur repeatedly, and studies have shown that RSV genomic adaptation occurs [23–25]. This most likely also means that RSV will adapt through mutation to evade vaccine pressure.

4.1.5 The Virus and the Replication Cycle

Viruses are classified by the International Committee on Taxonomy of Viruses (ICTV). RSV, formally known as human orthopneumovirus, belongs to the family *Pneumoviridae*, together with human metapneumovirus (HMPV), another pediatric respiratory pathogen. These viruses are in the order *Mononegavirales*, meaning that they are within the group of viruses with a non-segmented negative strand RNA genome [26]. Two types of RSV, type-A and type-B, share 95% sequence identity. RSV virion particles are between 100 and 300 nm in diameter and can have filamentous and other polymorphic shapes (Figure 4.1). The viral particle is composed of the nucleocapsid surrounded by a matrix protein lattice,



Figure 4.1 Transmission electron microscopy of RSV from tissue culture. (a and b) RSV particles imaged by transmission electron microscopy and (c) immuno-transmission electron microscopy. (a and b) Note the electron dense core. (c) Particles are in a fibrous-detritus matrix labeled with Anti-RSV-F antibody conjugated to 10 nm gold particles. The matrix confers resistance to antibody neutralization but has no effect on infectivity. Scale bar = 100 μ m.

surrounded in turn by a lipid envelope [27, 28]. The nucleocapsid consists of a helical ribonucleoprotein complex of viral genome RNA and nucleoprotein (N) [29, 30]. This N-RNA complex is associated with the viral polymerase, consisting of the large polymerase subunit (L) and phosphoprotein (P). It is also associated with the viral M2-1 protein, which also forms a lattice between the nucleocapsid and the matrix protein layer in virion particles [27, 28]. Three glycosylated spike proteins, glycoprotein (G), fusion protein (F), and small-hydrophobic protein (SH), protrude from the lipid envelope [27, 28]. The nucleocapsid and glycoprotein components of the virus particle and their roles in the viral replication cycle are described in the sections below. For a more detailed description of the molecular biology of RSV, see a review by Griffiths et al. [31].

4.1.6 The Nucleoprotein-RNA Complex

As noted above, the RSV genome is a single strand of negative sense RNA. It is 15.2 kb in length and codes for 10 mRNAs and 11 known proteins [32, 33]. In addition to the proteins that comprise the viral particle, it codes for nonstructural proteins 1 and 2 (NS1 and NS2), which antagonize innate immune responses [34], and M2-2, which regulates viral polymerase activity [35]. The 3' end of the genome contains an extragenic leader promoter region, which recruits the polymerase to begin transcription or genome replication [36, 37]. Each of the viral genes is flanked with conserved cis-acting elements, which direct the polymerase during mRNA transcription [38, 39]. The 5' end of the genome contains an extragenic trailer region. In the replicative intermediate RNA, known as the antigenome, the complement of the trailer acts as a promoter to recruit the polymerase to begin genome RNA synthesis [40–42]. The genome is encapsidated along its length with oligomeric N protein, such that each 43 kDa N monomer contacts seven nucleotides [30]. The genome RNA remains encapsidated throughout the viral replication cycle and so presumably the polymerase transiently displaces N protein as it moves along the template during transcription and replication.

4.1.7 The L Polymerase Protein

The RSV polymerase is responsible for generating capped and polyadenylated mRNAs and encapsidated antigenome and genome RNAs [43-45]. The essential nature and distinctive features of the polymerase make it an attractive target for antiviral drug development [44, 46]. The core polymerase is a complex of the L and P proteins [41, 47–49]. L is a 250 kDa protein that possesses the enzymatic activities required for transcription and replication. It has five domains: an RNA-dependent RNA polymerase (RdRp) domain, a capping domain. a connector domain, methyltransferase domain, and carboxyl-terminal (C-terminal) domain [44, 45]. The RdRp shares a number of features with the RdRps of other RNA viruses, reflected by the fact that there are broad-spectrum RdRp inhibitors that function against the RSV polymerase [50, 51]. The capping domain functions to add a guanosine cap to the 5' end of mRNAs [52, 53]. Although the cap structure is identical to that of cellular mRNAs, the enzymatic reaction is distinctive, occurring through a GDP polyribonucleotidyltransferase activity, rather than a guanylyltransferase activity [54–57]. By analogy with the L protein of related viruses, the capping domain might also play a structural role in initiation of RNA synthesis by providing a priming loop to stabilize the initiation complex [58–60]. The methyltransferase domain is responsible for methylating the mRNA cap, and the CTD also likely contributes to this activity [61, 62]. The role of the connector domain is not well characterized, but it is thought to aid the structural rearrangements necessary for the different enzymatic activities to come into play. The structure of the RSV polymerase and that of HMPV, its close relative, were recently solved using cryo-electron microscopy [47, 48, 63]. These studies revealed the structures of the RdRp and capping domains. These domains associate together to create the central cavity of the RdRp catalytic site. While the RdRp has features conserved with those of other RdRps, the capping domain of RSV L (and that of other non-segmented negative strand RNA viruses polymerases) is structurally unique, reflecting its distinctive enzymatic activity [45, 47, 48, 63]. The other three domains were not visible, indicating considerable structural flexibility. This is consistent with structural analysis of other non-segmented negative strand RNA virus polymerases, in which the connector, methyltransferase, and C-terminal domains have been observed juxtaposed in different positions relative to each other [58, 59, 64, 65]. These structural changes are likely to be necessary for the polymerase to be regulated between its different activities.

4.1.8 The P Phosphoprotein

The 27 kDa P protein functions as an essential cofactor for the L protein [49]. P has three domains, a central oligomerization domain that allows tetramer formation, and intrinsically disordered N- and C-terminal domains [66, 67]. P stabilizes the L protein and acts as an adaptor allowing the polymerase to associate with the other viral proteins involved in transcription and genome replication. The C-terminal domain of P binds to N-RNA, allowing the polymerase to associate with the template [68–70]. Its N-terminal domain binds to soluble N protein, presumably to enable encapsidation of the nascent RNA replication products [71, 72]. Its N-terminal domain also binds to M2-1, a transcription elongation factor [73, 74]. The cryo-electron microscopy analysis of the RSV L-P complex revealed that

the P tetramer adopts an unusual arrangement covering a relatively large surface area of the L protein [47, 48]. A similar arrangement was observed in the HMPV L-P structure [63]. P also functions independently of L to associate with N protein and form the cytoplasmic inclusion bodies that are the sites of RSV transcription and replication [75, 76]. It has recently been shown that a compound targeting these biomolecular condensate inclusion bodies has antiviral activity [77].

4.1.9 The M2-1 Protein

M2-1 is a 22 kDa protein that also forms a homo-tetramer [74, 78]. It is not required for RSV genome replication or initiation of transcription, but it is required as an elongation factor for the viral polymerase to generate full-length mRNA transcripts [79–81]. Because the RSV genes are transcribed sequentially, such that transcription of each downstream gene is dependent on transcription of all the upstream genes, M2-1 is essential for the entire viral genome to be transcribed [82]. M2-1 also binds to viral mRNAs posttranscriptionally and to select cellular mRNAs [76, 83]. However, it is not known what effect this posttranscriptional RNA binding activity has.

4.1.10 The RSV-G Glycoprotein

RSV-G is a type II transmembrane glycoprotein with the transmembrane domain at the amino-terminal end [84]. The G protein is the most variable protein of RSV, and RSV types A and B were initially differentiated, based on antibody reactivity to the G protein [20]. RSV-G serves as an attachment factor during virus entry (Figure 4.2) and is shed from infected cells as a soluble glycoprotein. The soluble form is expressed several hours before the expression of the full-length membrane-anchored RSV-G protein [85]. In the soluble form, RSV-G serves as an immune-decoy [85, 86]. RSV G protein's attachment role secures the viral particle to the host cellular membrane to prepare the fusion process to occur (Figure 4.2).

Despite its variability, RSV-G contains a central conserved domain that includes a CX3C motif [84]. Due to the highly conserved central conserved domain, which enhances infectivity and may have roles in modulating the host immune response through its interaction with CX3CR1, the G protein may emerge in the coming years as an attractive target for vaccine and therapeutic antibody development as reviewed in [84].

4.1.11 The RSV-F Fusion Glycoprotein

The fusion (RSV-F) protein is the most highly conserved glycoprotein on the surface of the RSV virion, and it is the principal neutralizing determinant on the RSV particle (Figure 4.3). Palivizumab and the most promising vaccine strategies have been developed against RSV-F. It exists on the surface of infectious virions in a metastable "prefusion" conformation that harbors the principal neutralization site, \emptyset [87]. The structure of RSV-F, its prefusion conformation, and the \emptyset neutralizing determinant were all elucidated recently [88], and some of the most promising experimental prophylactic, therapeutic, and vaccine strategies have come from these studies (Table 4.1) [87].



Figure 4.2 The RSV life cycle. The RSV life cycle and where potential RSV inhibitors will work to inhibit steps in the replication cycle.



(a) The principle antibody neutralization sites on a prefusion RSV-F monomer

Figure 4.3 The principal neutralizing determinants of RSV-F fusion glycoprotein, conformational breathing, and the fusion reaction. (a) The neutralization sites on RSV-F are shown on an RSV-F monomer. (b) Conformational breathing of the RSV-F trimer results in transient exposure and concealment of some epitopes on RSV-F. (c) The process of the fusion reaction showing extension of the fusion peptide (i) from within the RSV-F protein into the host-cell target membrane (ii). The complementary determining region (CDR) domains are brought together that brings the virus and host cell membrane within in close apposition (iii) and promotes virus-host cell membrane fusion (iv).

4.1.12 The RSV-SH Glycoprotein

RSV-SH is the smallest of the RSV membrane glycoproteins and exists in four forms in infected cells: SH_0 , SH_g , SH_p , and SH_t [89]. The predominantly expressed form is SH_0 , which is 7.5 kDa and is unglycosylated. SH_g is a 13–15 kDa glycosylated form that is the precursor of the more highly glycosylated SH_p . From that is between 21 and 40 kDa. The SH_t is a truncated form with approximately 7.5 kDa [90].

Immunotherapy under clinical development	Туре	Drug target	Stage	Target population
Ribavirin NH_2 N	Small-molecule inhibitor	e Broad- spectrum nucleoside analogue	Marketed compound	Ribavirin is used mainly when the outcome of an RSV LRTI could be fatal, typically in the immunocom- promised
Palivizumab	Monoclonal antibody	RSV-F	Marketed compound	Prevention of serious lower respiratory tract disease requiring hospitalization in children at high risk for RSV disease
Immunotherapy under clinical development	Туре	Drug target	Stage	Target population
RI-00I	IVIG	RSV proteins	s Phase 2a	Immunosup- pressed RSV- infected patients at risk for LRTI
RI-002	IVIG	RSV proteins	s For RSV- unknown	Primary humoral immunodeficiency disease in patients aged >12 years
Motavizumab (MEDI-524)	Monoclonal antibody	RSV-F	Failed to obtain FDA approval	High-risk infants
MEDI-8897	Monoclonal antibody	RSV-F	Phase 2b	High-risk infants
Novel antivirals undergoing clinical development	Туре	Drug target	Stage	Target population
Presatovir (GS-5806) ^{<i>a</i>} $CI \longrightarrow -NH CH_3$	Small-molecule inhibitor	e RSV-F	Phase 2-currently on hold	RSV-infected patients

Table 4.1 Antiviral compounds against RSV.

(Continued)

Table 4.1(Continued)

Novel antivirals undergoing clinical development	Туре	Drug target	Stage	Target population
Lumicitabine (ALS-008176 or JNJ-64041575) ^{<i>a</i>} H_2N H_2N H_3C H_3C CH_3 CH_3 CH_3 CH_3 CH_3 CH_3	Nucleoside analogue	RSV-L	Currently on hold	RSV-infected patients
Rilematovir (JNJ-53718678) ^{<i>a</i>} $O_{S}^{CH_3}$ O_{F} F_{F} F_{F} F_{F} N N N Cl Cl Cl Cl N N Cl N Cl N Cl N N N N N N N N	Small-molecule inhibitor	e RSV-F	Phase 2	RSV-infected patients
Sisunatovir (RV521) ^{<i>a</i>} $F \xrightarrow{V} V$ $F \xrightarrow{V} V$ F	Small-molecule inhibitor	e RSV-F	Phase 2a	RSV-infected patients
Ziresovir (AK0529) ^{<i>a</i>} (H_3) $(H_$	Fusion inhibitor	RSV-F	Phase 2	RSV-infected patients

Other investigational treatments and broad-spectrum antivirals	Туре	Drug target	Stage	Target population
RSV M2-1	Transcription anti- termination factor	RSV polymerase complex		RSV-infected patients
EDP-938	Replication inhibitor	RSV polymerase complex	Phase 2	RSV-infected patients

Table 4.1 (Continued)

Other investigational treatments and broad-spectrum antivirals	Туре	Drug target	Stage	Target population
EIDD-1931 ^{<i>a</i>} HN,OH HO,OH HO,OH	Ribonucleoside analogue inhibitor	RSV L	Preclinical	RSV-infected patients
VH244	Interfering particle		Preclinical	RSV-infected patients
Remdesivir ^a H_3C C CH_8 O NH_2 H_3C NH_2	Nucleoside analogue	RSV L polymerase complex	Marketed for SARS- Coronavirus 2	RSV-infected patients 2
Nitazoxanide ^a $\bar{O} = N^{\uparrow}$ S = N $H_{3}C = O$ O	Post viral entry interference	RSV L glycoprotein folding	Preclinical	RSV-infected patients
Favipiravir ^a $V \to V \to V$ $V \to V$ V $V \to V$ V $V \to V$ V $V \to V$ V V $V \to V$ V V V V V V V V V	Purine nucleotide analogue	RSV L	Marketed for influenza and SARS- coronavirus 2	RSV-infected patients

^a Chemical structures (Source: Modified from DrugBank).

The function of RSV-SH is less understood; it is not necessary for viral replication *in vitro*, but it is a necessary factor for pathogenicity *in vivo*. It is known to have roles as a viroporin [91], which enhances membrane permeability [92] and potentially acts as a protein that mediates cellular homeostasis during viral infection [93]. *In vitro* studies have also shown that SH protein inhibits apoptosis by inhibiting NF- κ B [94]. A recent study of Bovine RSV (BRSV) demonstrates that it shares a high level of genetic identity to Human RSV, immune-modulatory effects of SH, particularly in the inhibition of NF- κ B p65 phosphorylation [95]. Since 2015, there have been advances in a vaccine derived from the SH protein [96]. Langley et al. have recently published a successful preclinical study with a vaccine formulated with a single subunit based on the SH protein of RSV. Currently, this candidate is in phase 1 clinical trials.

4.1.13 Antigenic Variation of RSV Surface Glycoproteins

Antigenic variation of the RSV surface glycoproteins is not the dominant factor that plays a role in the global persistence of RSV infection, as it does with influenza. Nevertheless, a significant mutation of RSV-G has occurred over the last 50 years and resulted in the divergence of RSV type-B as a distinct subtype. There is some evidence that genetic adaptation leads to the emergence of dominant strains in the community that replace pre-existing strains as the predominant RSV in the population [23, 97]. For example, the current dominant RSV type-A strain, called ON1 that was identified in 2010 contained, most notably, a 72 bp duplication in the RSV-G glycoprotein. This strain has overtaken other strains of RSV to become the predominant RSV type-A strain worldwide [97]. In tissue culture, these viruses replicated to the highest levels compared with other RSV-A strains from patients [23]. RSV persists in the community, and the same strain can reinfect the same population over again. Thus, it is the nature of RSV to evade or suppress memory B cell development and not necessarily genetic variation that has resulted in RSV persistence and has complicated RSV vaccine development.

4.1.14 Adaptation of RSV in the Community by Emergent High Titer Clades

RSV continues to adapt to selection pressure by mutation. In 2017, we identified emergent clades of RSV types-A and -B that were associated with higher patient titers in the 2014–2016 RSV seasons. Isolates from these clades replicated to higher titers in culture than the lower titer "resident clades" of RSV [23]. These results suggest an ongoing adaptation of RSV to naïve populations. We suggest that if a genuinely efficacious vaccine were to be realized, vaccine breakthrough and therapy resistance could occur while adapting to naïve populations.

4.2 RSV Longevity, Immune Evasion, and the Role of IgA

Environmental factors independent of viral genetics play a significant role in determining the sensitivity of RSV to antibody neutralization and longevity. Susceptibility of RSV to antibody neutralization *in vitro* varies when comparing the different methods of purification [98].

The air pollution level may also play a role in determining how long RSV can remain infectious on a dry surface. RSV is predominantly spread by aerosol and contact with contaminated surfaces [99]. RSV infectivity can be prolonged significantly, effectively preserved on a dry surface from hours to months upon complexing of RSV particles with diesel pollution [100]. This study challenges the dogma that virus infectivity will be lost upon desiccation in the environment and has implications for hygiene, sterilization procedures, and infection control during RSV seasons in urban centers.

4.2.1 Evasion or Suppression of Immune Memory

Bronchiolitis has been associated with the IgG1 subtype [101]. Secreted IgA is transported across the epithelium into the respiratory mucosa, where it confers sterilizing

protection against viral infection [2, 102, 103] (Figure 4.4). IgA, IgG, and IgM are provided in breast milk [104]. It has been suggested that breast milk regurgitation will coat the upper respiratory tract (URT), which lends protection from initial URT infection for infants.

A significant hindrance to RSV vaccine development is that RSV infection does not confer immune memory; the same virus strain can reinfect the same individual multiple times [105]. This would suggest that RSV is not immunogenic. In adult challenge studies with influenza and RSV, it was shown that although individuals generated healthy levels of IgG memory B cells, a paucity of memory IgA B cells existed after RSV infection, unlike with influenza [102]. A more recent study showed there was no enrichment of anti-RSV IgA memory B cells in adenectomies from children compared with peripheral blood [106], while there were relatively higher IgG memory B cells in the adenoids compared with peripheral blood. These results further speak to the evasion of or suppression of memory IgA B cells in the URT.



Figure 4.4 Factors that prevent and help clear RSV infection from the airway. The locations and factors protect from RSV infection in the upper, lower respiratory tracts, and in the gut.

4.3 The Impact of Immunoprophylaxis on the Health Burden of Respiratory Syncytial Virus

RSV is the most common cause of lower respiratory tract infections and hospitalizations in infants worldwide [50]. The first and only currently licensed drug to prevent RSV infection is Palivizumab. Although everyone is expected to have had an RSV infection by their second birthday [107], in certain well-recognized high-risk groups, such as preterm infants and infants born with chronic lung disease or congenital heart disease, there is a high risk of severe respiratory illness [108, 109].

4.4 Distinct RSV Symptoms

RSV causes symptoms in infants that distinguish it from other respiratory viruses. The most common is wheezing due to infant bronchiolitis. This is because RSV is highly successful at infiltrating the lower airways compared with other respiratory viruses [110]: RSV accounts for 90% of bronchiolitis in infants and 75% of all bronchiolitis cases worldwide [111]. It is also a leading cause of infant hospitalization, with an estimated 33.1 million episodes of lower respiratory tract infection (LRTI), 3.2 million hospital admissions, and as many as 118 200 deaths worldwide in 2015 [4]. Recent studies have also suggested that hospitalization due to RSV in the first three years of life is associated with long-term respiratory illness, especially asthma or reoccurring wheeze. These conditions may also persist into adulthood [112–115]. There is, however, more research needed to parse out the involvement of RSV in these longer-term chronic sequelae, like asthma. There is a relationship with the prevention of RSV in the first 12 months of life and an associated reduction in wheeze. However, this relationship is a mere association, and so mechanistic causality has not yet been demonstrated [116].

4.4.1 Wheeze

Wheeze is the result of the narrowing of the airways in the lung and is caused by inflammation due to infection, asthma, or chronic obstructive pulmonary disease. Its hallmark presentation is a whistling sound that is generated upon inspiration/expiration due to the airway narrowing [117].

The question remains as to whether preventing RSV infection with immunoprophylaxis can prevent the onset of chronic wheeze, as mixed results have been published in the literature. There are some accounts that recurrent wheeze is correlated with infants who receive Palivizumab as well as reports of a reduction in wheeze in infants that receive Palivizumab [116, 118, 119]. These results may differ, however, when confounding factors are considered. For example, those infants that receive Palivizumab are at a higher risk of recurrent wheeze because of a variety of risk factors such as prematurity, cystic fibrosis, congenital heart disease, and other factors that qualify them for Palivizumab. Studies have also only noted recurrent wheeze correlated to infants in families with a history of atopy. Recurrent wheeze is common in infants born prematurely before 33 weeks; the risk is not consistently reduced when those infants receive Palivizumab [112, 120].

RSV also poses a significant health burden on the elderly population worldwide, especially those with pre-existing conditions like COPD and heart disease. Unlike in infants, RSV does not have distinct symptoms in adults, like a wheeze that readily distinguishes it from other respiratory pathogens [121]. RSV is also a significant contributor to pneumonia—both community and ventilator acquired. There is no treatment available for RSV in elderly populations. The prophylactic antibody, Palivizumab that prevents RSVassociated hospitalization, is only administered to high-risk infants, and Palivizumab has only been tested in clinical trials for high-risk infants. There are candidates in preclinical and clinical trials to prophylax and vaccinate the elderly; however, nothing has been approved.

4.4.2 The Effect of Immunoprophylaxis on the Development of Asthma

A recent single-blinded study published in 2018 followed the development of asthma in premature infants prophylaxed with Palivizumab in their first year of life. Their health was followed through to the age of six years [122]. After unblinding, no association of Palivizumab versus placebo on the development of asthma by six years of age was found. These results might suggest superficially that RSV infection prevention may not affect the development of asthma. However, one must remember that for the intervention with Palivizumab, efficacy hovers around a 50% reduction in hospitalizations. This is not a reduction in infections, for which, Palivizumab is even less efficacious. With this in mind, a study with an impressive participant power of 400 may still be underpowered to determine the effect of RSV infection on the development of asthma. A recent article also found that when looking at children with six years of age, there is no correlation between having received Palivizumab in infancy and the risk of decreased pulmonary function or asthma [122].

Therefore, to truly understand the link between RSV infection and asthma, considerably better interventions like those that target, e.g. the Ø neutralization site on RSV-F may have to be used. A more recent study found a significant benefit from Palivizumab prophylaxis on the development of asthma [123]. Here, there was a significant discord in the diagnosis of asthma between physicians. This may be due to a diagnosis of asthma from signs alone, which is unreliable compared with methylcholine challenge diagnosis of asthma.

4.4.3 Epidemiology and Clinical Aspects of Adult RSV Disease

Emerging data show that RSV is also a significant cause of hospitalization and deaths in adults. It has been estimated that RSV annually infects 3–10% of the adult population [124]. A recent global estimate in 2015 showed that there were about 1.5 million episodes (95%CI 0.3–6.9 million) of acute respiratory infections due to RSV in older adults in industrialized countries (data for developing countries were missing). Of these, 14.5%

or an estimated 336000 required hospitalization (range 186000-614000), and there were about 14000 deaths (range 5000-50000). Hospitalization rate and fatality were higher for those aged ≥ 65 years than for those aged 50–64 years and among those with underlying conditions (e.g. chronic lung diseases, chronic cardiovascular diseases) [125-128]. RSV has been shown to account for 5-15% of community-acquired pneumonia, 9-10% of hospital admissions for acute cardiorespiratory diseases and excessive deaths among adults during seasonal peaks. Outbreaks among nursing home residents are frequent but under-recognized [125, 127]. In hospitalized adults, the disease burden of RSV and clinical severity (as indicated by mortality 8-10%, ICU admission 10-18%, prolonged duration of hospitalization) are at least similar to, if not higher than, seasonal influenza [124, 127-129]. In particular, profoundly immunosuppressed adults, such as hematopoietic stem cell transplant (HSCT) recipients, are at high risk for progression to severe LRTI by RSV, which rapidly can be fatal (17-40%) [130]. Active pneumonic changes are evident radiographically in about 50-60% of hospitalized adults. Typically, changes include patchy consolidations and ground-glass opacities involving the lower zones, which are predictive of adverse outcomes [131]. Notably, high viral load and longer duration of viral shedding are significantly associated with clinical severity, indicating potential roles for antiviral treatment [132, 133]. Despite the availability of rapid molecular diagnostics, adult RSV infections are often not clinically suspected or are diagnosed late in the course of illness, posing challenges to potential interventions and infection control [134].

4.5 History of RSV and Vaccine Development

4.5.1 The Tragic History of RSV Vaccine Development in the First Failed RSV Vaccine Trials of the 1960s

The first vaccine clinical trials began in 1964, which is where RSV disease enhanced by the vaccine was realized. Following the success of the development of other formalin-inactivated (FI) vaccines, the RSV (FI RSV) vaccine trial in the 1960s ended in a tragedy. Most of the vaccinated children in the study had an increased risk of hospitalization due to RSV infection, and there were two deaths of vaccinees due to vaccine-enhanced RSV disease [135]. Health regulatory bodies were thus reticent to grant permission to subsequent RSV vaccine clinical trials.

It was not until 40 years after the ill-fated FI RSV vaccine trial in the 1960s that RSV research groups elucidated the reasons for the failure of the vaccine [136, 137]. Primary causes were a weak antibody avidity to the RSV F protein and inadequate Toll-Like Receptor (TLR) stimulation by the inactivated RSV vaccines, which ultimately lead to poorly neutralizing antibodies [136]. Further research identified an absence of prefusion F protein in the FI RSV particles [137]. This further explains weak antibodies that were developed in patients given the formalin-inactivated vaccine.

In the 1990s and 2000s, there were several clinical trials of RSV subunit vaccines that failed to confer robust protection from RSV infection or the preparation was not brought to market (tested in high-risk groups). However, these trials lacked the previously

noted negative consequences [138–141]. Furthermore, replication-competent attenuated vaccines proved too attenuated and not to elicit an immune response that could confer protective immune memory, even when given with an injected RSV subunit vaccine [142].

Research had led to the idea that the potential for using TLR stimulants as adjuvants will potentiate RSV vaccine reactivity. There is some evidence that stimulating with TLR agonists is a solution to an effective vaccine. TLR agonists have been shown to increase RSV and Influenza vaccine immunogenicity [143, 144]. However, such strategies must be approached with caution. In a study where TLRs were stimulated in mice, the mice responded with an increase in disease severity [145].

4.6 New Developments in RSV Vaccine Development

The *in vitro* sensitivity of RSV to antibody neutralization varies depending upon the method of virus stock purification [98]. It is, therefore, possible that the proportion of prefusion versus post-fusion RSV-F on the surface of RSV virions could be altered by different techniques and chemicals involved in the purification process.

4.6.1 Conformational Breathing of RSV-F Affects Neutralization Sensitivity

A recent study has demonstrated the conformational "breathing" of the RSV-F trimer on the surface of RSV virions (Figure 4.3) [146]. RSV-F breathing leads to conformational alterations in the principal neutralization site \emptyset [147]. Neutralization site \emptyset is recognized by two different human antibodies AM22 and RSD5, that each bind to distinct conformations of site \emptyset [146]. Given that the most promising vaccine strategies include prefusion RSV-F, discoveries such as these help to inform vaccine development and the mechanistic understanding of neutralization.

4.6.2 Measles Versus RSV Vaccine Development

After the failures of vaccine development for RSV in the 1960s, the development of a live attenuated measles vaccine occurred that is still used as a part of the MMR vaccine today. However, so far, live-attenuated RSV vaccines have remained elusive as they have not been immunogenic or conferred immune memory against RSV infections.

The critical difference in success between attenuated RSV and measles vaccines could be found in the mode of replication. Measles infections produce a viremia with detectable high levels of virus in the blood, whereas there is by and large no detectable RSV in the bloodstream during infection; measles virus infection is systemic whereas RSV is restricted to the airways. The lungs are an immune-privileged organ. Even alveolar macrophages are a distinct subset of white blood cells of the lung that act to moderate immune responses to microbial infections. It is, therefore, possible that attenuated measles vaccines work because there is whole-body involvement during the infection. This would include major immune organs like the spleen and the thymus. Lastly, there is a lack of data about RSV strains and epidemiology, but presently many research groups are contributing to epidemiological data on RSV.

4.6.3 The Lack of Immunogenicity, Immune Evasion, or Immune Suppression

Research on the immune response of humans to RSV shows high levels of antibodies to RSV but no protection from infection. For reasons still unexplained, humans cannot raise the antibody titer high enough to gain complete immunity to the virus. In addition, mucosal antibody levels and memory are inadequate or nonexistent [102].

Currently, vaccine strategies use novel adjuvants, virus-like particles (VLPs), and vector vaccines. Most current vaccines use a purified F protein. Currently, there are 6 RSV vaccines in phase 2 and 14 in phase 1 with many in early development stages. The pipeline of RSV vaccine candidates ranges from whole-inactivated, live attenuated, particle- and subunit-based or different RSV glycoproteins, nucleic acid vaccines, recombinant vectors, and more immunoprophylactic monoclonal antibodies in the hope of improving the RSV treatment and prophylaxis field.

4.6.4 Recent RSV Vaccines

In the current pipeline, we have seen the failure of ResVAX-Novavax's RSV F subunit vaccine in early 2019. This preparation is a nanoparticle vaccine with recombinant RSV-F. In its prefusion conformation, the RSV-F is in a quasi-stable state so that it will rapidly and spontaneously undergo conversion to the post-fusion form. Therefore, the native RSV-F of ResVax will likely be predominantly post-fusion, thus minimizing presentation of the principal Ø neutralization epitope. The FDA has required another trial to test clinical equipoise for the reappropriation of the product as one that reduces the hospitalization of infants.

4.6.4.1 Success in Secondary Endpoints

The maternal immunization trial with ResVax concluded in Spring 2019 and achieved its secondary endpoint, which was a reduction in hospitalizations due to LRTI. Here, ResVax also protected from the serious cyanosis hypoxia complication of RSV LRTI in neonates. As discussed earlier, there was a reduction in all-cause LRTI. This is consistent with RSV as a complication of other infections and health problems, i.e. a major complication of preexisting disease burden.

4.6.5 A Resurgence in Attenuated Replication-Competent Vaccines

The development of the RSV reverse genetics system brought with it opportunities for attenuated RSV vaccine development [148–151]. Being a negative-stranded virus with relatively complex molecular biology, this system requires ectopic co-expression of numerous RSV genes with a helper virus. The helper virus is needed to provide the RNA-dependent RNA polymerase activity specific to negative-stranded virus replication. Thus, this system provides greater control over the viral genetic material that goes into making attenuated

vaccines and allows for inputting several attenuations and safety mechanisms to make reversion to fully virulent forms less likely.

One technology involves codon deoptimization by altering codons of RSV NS1 and NS2 virulence genes from a virus to a human codon bias [148]. The attenuated deoptimized virus protected mice from challenge strains and was more immunogenic than WT RSV. Currently, PATH tracks the RSV vaccine and antibody progress that occurs.

4.6.5.1 Mechanisms of Protection from RSV that Require Attention

in Vaccine Development

Mice are currently the most prominent model for T cell research in humans. T cells can clear RSV and shorten the duration of viral shedding. Paradoxically, T cells can also enhance disease. There have been studies showing that when T cells are depleted, symptoms are reduced in certain conditions [152]. Also, adoptive transfer T cells can cause more disease. In conclusion, it is not clear what T cells' role in RSV infection is.

Once the lower airways become infected with RSV, CD8 T cells are essential for viral clearance [153]. However, during Th2 responses, poor outcomes may be associated, such as an enhanced respiratory disease. Enhanced respiratory disease is also partially due to an inappropriate T cell response, more specifically due to Th2 cells [136]. To assess the safety of the vaccine candidate, developers are required to study Th1 and Th2 responses.

Another issue is the duration of protective immunity: When looking at correlates of protection, Hall et al. rechallenged 15 individuals with RSV for two years [105]. Test persons were susceptible to the same strain of RSV even after a few months of clearing their infection, although antibodies were detected in their responses. There appears to be a plateau for the boosting of antibodies to a point where total immunity to the virus cannot be acquired.

4.7 Antivirals and Therapeutic Antibodies

4.7.1 Immunoprophylaxis

The only success in getting a pharmaceutical product to market.

4.7.2 RSV-IGIV

Since vaccine progress was slowing down due to the 1960s' vaccine incident with two infant deaths, RSV treatment development turned to safer options. In the 1980s, RSV researchers were testing cotton rat models for intravenous immunoglobulin [154]. In the early 1990s, the first trials for RSV hyper immunoglobin (RSV-IGIV), a polyclonal human intravenous antibody pooled preparation, took place. The pilot study showed that IGIV is safe and effective in infants for the prevention of severe RSV disease.

IVIG was the first successful prophylaxis developed for severe RSV infection in infants [155, 156]. Furthermore, this became the first time that IGIV was used to prevent a viral infection. In 1996, the American Food and Drug Administration (FDA) agency approved

RSV-IGIV [157]. IGIV became the first approved agent for RSV in high-risk infants and to treat children under the age of 2 at risk for severe RSV.

In some patient populations, there were concerns over the safety and efficacy of RSV-IGIV. The official guidelines in 1997 recommended only using the preparation in infants under 32 weeks because of data showing the highest efficacy in infants under 6 months and potential complications in children undergoing surgery [158].

In 2003, RSV-IGIV was pulled from the market after the licensing of the first monoclonal antibody for RSV-Palivizumab in 1998. However, RSV-IGIV had paved the way for humanized monoclonal antibodies to be used in treatment regimens.

4.7.3 Palivizumab

In the late 1990s, after the efficacy of RVS-IGIV against RSV severe LRTI and hospitalization was shown, pharmaceutical companies began making monoclonal antibodies against the RSV F protein. They also sought to make the injection intramuscular as opposed to IV, which is more difficult and prone to complications in infants. Three monoclonal antibodies were developed for RSV. The first by SmithKline Beecham (now GSK) [138], SB 209763, showed safety and good tolerability but lacked efficacy due to the low dosing schedule in their phase 3 trial. OraVax made the second antibody, HNK20, which displayed a lack of efficacy in a rhesus monkey trial [159]. Lastly, MedImmune created MEDI-493, a monoclonal antibody that was potent and broadly reactive to the F protein of RSV and was more potent than RSV-IGIV. MEDI-493 (Palivizumab) was later marketed for the prevention of RSV hospitalization [160].

Palivizumab (Synagis[®]) was approved in 1998 by the FDA; it is a humanized monoclonal antibody administered through intramuscular injections (as opposed to intravenous IGIV) at 15 mg/kg monthly for infants, who according to the guidelines have a high risk for bronchiolitis.

We are expecting a soon to be approved, new mAb from MedImmune, that is 100 times more potent than Palivizumab and would be given once a season rather than monthly. Palivizumab has a half-life of 20–30 days; therefore, high-risk infants have to be brought into hospitals and clinics every month during the RSV season to make sure they have a high-enough concentration of antibodies in their bodies to be protected from severe RSV infection. With the new mAb from MedImmune, the half-life is said to be 70–100 days, which would mean one visit to the clinic during the RSV season would be sufficient [161].

Although Palivizumab reduces hospitalization by as much as 80%, it does not prevent the URT infection [162]. Palivizumab does, however, protect against severe lower respiratory tract disease, lowering the incidence of infant hospitalization, though the cost to benefit ratio provided by widespread Palivizumab use has been questioned in the literature [163–165].

4.8 Therapeutics for Treating Active RSV Infections (Table 4.1)

Currently, only two drugs are approved by the US FDA for RSV infections: Palivizumab for prophylaxis and aerosolized Ribavirin for treatment. Ribavirin is a broad-spectrum

nucleoside analogue that was created in 1972 and found to affect infections with poliovirus, hepatitis C virus (HCV), RSV, and several other RNA viruses. Different mechanisms have been associated with the use of Ribavirin in the context of HCV treatment [166]. The triphosphate form of Ribavirin is accommodated by the RNA-dependent RNA polymerase, which can affect both efficacy and fidelity of viral genome synthesis. In 1986, the FDA approved Ribavirin for aerosol treatment of RSV infection in children in the United States.

Evidence for the clinical benefit of aerosolized Ribavirin has been inconclusive. Health care providers' perceptions of limited benefits and the cumbersome requirement for scavenging apparatus and concerns about mutagenicity, carcinogenicity, and teratogenicity have resulted in infrequent use. Currently, Ribavirin is used mainly when the outcome of an RSV LRTI could be fatal, typically in the immunocompromised. Recent studies showed similar outcomes comparing aerosolized versus oral Ribavirin, which was proposed as an alternative with better tolerance, lower cost, and easier application [130, 167].

Available data suggest that immunocompromised patients, who receive these treatments in various combinations may have a lower rate of progression to LRTI and better clinical outcomes. Treatment can be considered, though the level of evidence is weak to moderate [168, 169]. It is unknown whether these approaches can be applied to older, non-immunocompromised adults. A recent randomized placebo-controlled trial on RSV bronchiolitis in 420 infants showed that single-dose intravenous Palivizumab (15 mg/kg) did not result in faster hospital discharge or in preventing readmission [170]. Another placebo-controlled trial (n = 429) showed that Palivizumab for RSV prophylaxis in infants did not have a significant impact on current asthma or lung function at the age of sixyears [122].

4.8.1 Immunotherapy under Clinical Development

RI-001 (Table 4.1) is an IVIG prepared from plasma donors who have high titers of neutralizing anti-RSV antibodies. RI-001 was evaluated in a phase 2a placebo-controlled trial in immunosuppressed RSV-infected patients at risk of LRTI, but no data have been published. Compassionate use experience in 15 immunocompromised patients with RSV LRTI was associated with a significant increase in neutralizing antibody titers and clinical improvement, especially if given early [171].

RI-002 (Table 4.1) is an Intravenous Immune Globulin (IVIG) approved by the US FDA in April 2019 for the treatment of Primary Humoral Immunodeficiency Disease in adults and adolescents (12–17 years of age). It also has standardized elevated levels of RSV-neutralizing antibodies (from plasma donors who have high titers of neutralizing anti-RSV antibodies), and elevated antibody titers to other respiratory viruses (e.g. influenza, parainfluenza, coronavirus, metapneumovirus). In a multicenter, phase 3 trial (n = 59), its regular infusion over one year was associated with significantly reduced risk of severe infections, and it was well tolerated [172]. Plans on its further study for RSV prophylaxis or treatment are unknown.

Motavizumab (MEDI-524, Table 4.1), a second-generation monoclonal antibody derived from Palivizumab, was studied in a placebo-controlled trial in infants hospitalized for RSV LRTI. It was not shown to impact on viral load or clinical severity [173]. In a phase 3 prophylaxis trial in high-risk children, it was associated with hypersensitivity reactions in

some infants and consequently failed to obtain US FDA approval, although Motavizumab was shown to be more effective than Palivizumab.

MEDI-8897, Table 4.1 is a third-generation RSV-specific monoclonal antibody that targets the prefusion conformation of the RSV F protein, with an extended half-life (62.5–72.9 days). Data from a phase 1 trial in healthy adults showed a favorable safety profile [174]. In a phase 1b/2a dose-escalation study (n = 89), healthy preterm infants, who received a single dose showed a significant rise of serum RSV-neutralizing antibody levels compared with placebo [175]. A phase 2b trial on efficacy and safety of "once-per-season" prophylactic regimen against medically attended RSV LRTI in healthy preterm infants has been completed (NCT02878330; granted Breakthrough Therapy Designation by US FDA in February 2019).

4.9 Drug Targets

There have been decades of research on the replication and the structural components of RSV. The virus encodes 10 genes that are responsible for being translated into 11 proteins.

4.9.1 Novel Antivirals Undergoing Clinical Development

Presatovir (GS-5806, Table 4.1) is a potent small-molecule inhibitor that targets the RSV F protein by inhibiting F protein-mediated cell-to-cell fusion. Prior studies with GS-5806 in healthy adults challenged with intranasal RSV demonstrated reduced viral load and severity of clinical disease [130, 176–178]. Preliminary results from a recent phase 2 study conducted in HSCT recipients with either upper respiratory (URTI, *n* = 189) or lower respiratory tract RSV infection (LRTI, *n* = 60) were encouraging. URTI patients treated with GS-5806 demonstrated a significant decrease in time-weighted average nasal RSV RNA load over nine days compared with placebo, and reduced LRT complication progression rates (9 vs. 20%, *p* = 0.04), especially when being treated early. However, virological and clinical endpoints were insignificant in patients with established LRTI. Treatment is generally well tolerated. Several newly identified treatment-emergent RSV F substitutions (e.g. T400A/I, F140I, L138I) were found to map to interaction sites between the drug and RSV F. Resistance occurs rapidly to drugs by RNA viruses such as RSV, if the drug targets a non-conserved site on the protein. Human clinical trials were declared suspended, "on hold," in autumn of 2018 for undeclared reasons.

Lumicitabine (ALS-008176 or JNJ-64041575, Table 4.1) is an oral nucleoside analogue prodrug. The triphosphate form of the parent compound is a selective inhibitor of the RSV RNA polymerase that causes chain termination [179]. A human RSV challenge model showed rapid viral clearance and significant reduction of viral load, accompanied by improvements in the severity of clinical disease compared with negative controls [180, 181]. Human clinical trials were suspended, "on hold," in autumn of 2018 for undeclared reasons. Data on ongoing/completed clinical trials in adults and infants have not been reported (NCT02673476, NCT02935673, EudraCT number 2013-005104-33).

Rilematovir (JNJ-53718678, Table 4.1) is a small-molecule RSV fusion inhibitor. In a phase 2a adult RSV challenge study [182], oral treatment was shown to reduce viral load and mean overall symptom scores [183]. Reported treatment-emergent adverse events (grade 1 or 2 electrocardiogram change) require further study. Phase 1b and phase 2 placebo-controlled trials in adults and infants have been initiated (NCT03379675, NCT03656510).

SiSunatovir (RV521, Table 4.1) is an oral small-molecule fusion inhibitor designed to target RSV-F that mediates RSV binding to cellular receptors. Preliminary results from a randomized, placebo-controlled phase 2a study showed that oral treatment decreased RSV viral load and reduced disease severity in 66 adult participants in a virus challenge model (NCT03258502) [184].

Ziresovir (AK0529, Table 4.1) is a fusion inhibitor, active against RSV A & B. Preliminary results from a randomized, double-blind, placebo-controlled phase 2 trial in children 1–24 months and hospitalized with RSV showed a dose-dependent decrease in symptom score at 24 hours after a single dose, and no major safety signal was reported (NCT02654171). A phase 2 trial is ongoing (NCT03699202) [177].

4.9.2 Other Investigational Treatments and Broad-Spectrum Antivirals

RSV M2-1 inhibitor (Table 4.1), a transcription anti-termination factor, has been identified as a new target for RSV therapeutic intervention. Small molecules such as cyclopamine have been shown *in vitro* to inhibit the postentry phase of viral replication, reduce transcription of downstream genes in RSV replication, and disrupt IBAG (inclusion bodyassociated granules formation). The drug suppresses RSV lung infection in the mouse model by blocking transcription [177, 185, 186].

EDP-938 (Table 4.1) is a novel non-fusion replication inhibitor of RSV with a high barrier to resistance. Phase 1 and 2 trials are ongoing (NCT03691623).

EIDD-1931 (Table 4.1) (β -d-N⁴-hydroxycytidine or NHC) is an orally available ribonucleoside analogue inhibitor with activity against RSV and several other RNA viruses. The putative mechanism of action is the induction of excessive mutations that lead to error catastrophe and, ultimately, the inability to replicate. The drug was shown to have a high resistance barrier. Animal models indicated efficacy in reducing viral load and symptoms [187].

VH244 (Table 4.1) is a novel broad-spectrum antiviral. The mechanism of action is a "Therapeutic Interfering Particle," modeled after defective, nonreplicating, influenza virus particles [188]. It has been shown *in vitro* to reduce replication of influenza viruses through genomic interference, and inhibit RSV and human rhinoviruses through host innate immunity and activation of the cell antiviral state. A surrogate murine model for RSV showed protection from the disease if given within four days postinfection [177].

Other broad-spectrum antivirals with *in vitro* activity against RSV include **Remdesivir**, **Favipiravir**, **and Nitazoxanide (Table 4.1)**. However, data specific to RSV infection is lacking [189–191]. Remdesivir is an adenosine analogue prodrug that is incorporated by the viral RNA polymerase and causes delayed chain-termination further downstream [192]. Remdesivir was developed for the treatment of Ebola Virus Disease (EVD) and is currently assessed in clinical trials for Covid-19 [177, 193]. In October 2020, the FDA approved

Remdesivir for use against SARS CoV-2 to combat Covid-19. Remdesivir is currently used worldwide with reports stating quicker recovery for patients who are hospitalized with Covid-19 [193]. Favipiravir is approved for the treatment of Influenza in Japan. The drug is structurally related to Ribavirin. Favipiravir was assessed in clinical trials against SARS-CoV-2 with minimal-to-no success. More studies are ongoing and necessary to conclude this is an effective drug against Covid 19 [194].

4.10 Conclusions

The recent partial successes in RSV vaccine development constitute a significant step forward in realizing a therapy that can confer, at least, passive immunity to the most vulnerable. However, the research community has not yet shed light on the mechanism for memory IgA B cell deficiency in the context of RSV infection. This aspect of RSV replication and biology will likely have to be elucidated before transmission-limiting vaccines can be developed. With the potential for RSV antivirals and vaccines coming along in the next decade, there may be viral factors that mutate in response to drug pressure. For example, even with the limited use of Palivizumab in high-risk infants, RSV already has shown mutations in the Palivizumab binding site that confers viral resistance to the antibody [195]. To create a vaccine: the prefusion RSV-F confirmation with a focus on anti-RSV nasal IgA levels and new live-attenuated vaccine approaches with adjuvant strategies are likely the way forward.

References

- **1** Jares Baglivo, S. and Polack, F.P. (2019). The long road to protecting infants against severe RSV lower respiratory tract illness. F1000Res 8.
- **2** Ascough, S., Vlachantoni, I., Kalyan, M. et al. (2019). Local and systemic immunity against RSV induced by a novel intranasal vaccine: a randomised, double-blind, placebo-controlled trial. *Am. J. Respir. Crit. Care Med.* 200 (4): 481–492.
- **3** Bryce, J., Boschi-Pinto, C., Shibuya, K. et al. (2005). WHO estimates of the causes of death in children. *Lancet* 365: 1147–1152.
- **4** Shi, T., McAllister, D.A., O'Brien, K.L. et al. (2017). Global, regional, and national disease burden estimates of acute lower respiratory infections due to respiratory syncytial virus in young children in 2015: a systematic review and modelling study. *Lancet* 390: 946–958.
- 5 Nair, H., Nokes, D.J., Gessner, B.D. et al. (2010). Global burden of acute lower respiratory infections due to respiratory syncytial virus in young children: a systematic review and meta-analysis. *Lancet* 375: 1545–1555.
- **6** Stein, R.T., Bont, L.J., Zar, H. et al. (2017). Respiratory syncytial virus hospitalization and mortality: systematic review and meta-analysis. *Pediatr. Pulmonol.* 52: 556–569.
- **7** Caballero, M.T., Bianchi, A.M., Nuno, A. et al. (2019). Mortality associated with acute respiratory infections among children at home. *J. Infect. Dis.* 219: 358–364.
- **8** Caballero, M.T. and Polack, F.P. (2018). Respiratory syncytial virus is an "opportunistic" killer. *Pediatr. Pulmonol.* 53: 664–667.
- **9** Scheltema, N.M., Gentile, A., Lucion, F. et al. (2017). Global respiratory syncytial virusassociated mortality in young children (RSV GOLD): a retrospective case series. *Lancet Glob. Health* 5: e984–e991.
- **10** Pneumonia Etiology Research for Child Health Study Group (2019). Causes of severe pneumonia requiring hospital admission in children without HIV infection from Africa and Asia: the PERCH multi-country case-control study. *Lancet.* 394: 757–779.
- **11** La Rosa, G., Fratini, M., Della Libera, S. et al. (2013). Viral infections acquired indoors through airborne, droplet or contact transmission. *Ann. Ist. Super. Sanita*. 49: 124–132.
- 12 Shiu, E.Y.C., Leung, N.H.L., and Cowling, B.J. (2019). Controversy around airborne versus droplet transmission of respiratory viruses: implication for infection prevention. *Curr. Opin. Infect. Dis.* 32: 372–379.
- **13** Kodama, F., Nace, D.A., and Jump, R.L.P. (2017). Respiratory syncytial virus and other noninfluenza respiratory viruses in older adults. *Infect. Dis. Clin. N. Am.* 31: 767–790.
- 14 Lindsley, W.G., Blachere, F.M., Davis, K.A. et al. (2010). Distribution of airborne influenza virus and respiratory syncytial virus in an urgent care medical clinic. *Clin. Infect. Dis.* 50: 693–698.
- **15** Noor, A. and Krilov, L.R. (2018). Respiratory syncytial virus vaccine: where are we now and what comes next? *Expert. Opin. Biol. Ther.* 18: 1247–1256.
- **16** Hall, C.B. and Douglas, R.G. Jr. (1981). Modes of transmission of respiratory syncytial virus. *J. Pediatr.* 99: 100–103.
- 17 Munywoki, P.K., Koech, D.C., Agoti, C.N. et al. (2014). The source of respiratory syncytial virus infection in infants: a household cohort study in rural Kenya. *J. Infect. Dis.* 209: 1685–1692.
- **18** Pandolfi, E., Gesualdo, F., Rizzo, C. et al. (2019). Breastfeeding and respiratory infections in the first 6 months of life: A case control Study. *Front. Pediatr.* 7: 152.
- 19 Brint, M.E., Hughes, J.M., Shah, A. et al. (2017). Prolonged viral replication and longitudinal viral dynamic differences among respiratory syncytial virus infected infants. *Pediatr. Res.* 82: 872–880.
- **20** DeVincenzo, J.P., Wilkinson, T., Vaishnaw, A. et al. (2010). Viral load drives disease in humans experimentally infected with respiratory syncytial virus. *Am. J. Respir. Crit. Care Med.* 182: 1305–1314.
- **21** Graham, B.S. and Anderson, L.J. (2013). Challenges and opportunities for respiratory syncytial virus vaccines. *Curr. Top. Microbiol. Immunol.* 372: 391–404.
- **22** Anonymous (1998). Palivizumab, a humanized respiratory syncytial virus monoclonal antibody, reduces hospitalization from respiratory syncytial virus infection in high-risk infants. The IMpact-RSV Study Group. *Pediatrics* 102: 531–537.
- **23** Elawar, F., Griffiths, C.D., Zhu, D. et al. (2017). A Virological and phylogenetic analysis of the emergence of new clades of respiratory syncytial virus. *Sci. Rep.* 7: 12232.
- **24** Agoti, C.N., Munywoki, P.K., Phan, M.V.T. et al. (2017). Transmission patterns and evolution of respiratory syncytial virus in a community outbreak identified by genomic analysis. *Virus Evol.* 3: vex006.
- **25** Agoti, C.N., Phan, M.V.T., Munywoki, P.K. et al. (2019). Genomic analysis of respiratory syncytial virus infections in households and utility in inferring who infects the infant. *Sci. Rep.* 9: 10076.

- Kuhn, J.H., Adkins, S., Alioto, D. et al. (2020). 2020 Taxonomic update for phylum Negarnaviricota (Riboviria: Orthornavirae), including the large orders Bunyavirales and Mononegavirales. *Arch. Virol.* 165: 3023–3072.
- Kiss, G., Holl, J.M., Williams, G.M. et al. (2014). Structural analysis of respiratory syncytial virus reveals the position of M2-1 between the matrix protein and the ribonucleoprotein complex. *J. Virol.* 88: 7602–7617.
- 28 Liljeroos, L., Krzyzaniak, M.A., Helenius, A., and Butcher, S.J. (2013). Architecture of respiratory syncytial virus revealed by electron cryotomography. *Proc. Natl. Acad. Sci.* U. S. A. 110: 11133–11138.
- Bakker, S.E., Duquerroy, S., Galloux, M. et al. (2013). The respiratory syncytial virus nucleoprotein-RNA complex forms a left-handed helical nucleocapsid. *J. Gen. Virol.* 94: 1734–1738.
- Tawar, R.G., Duquerroy, S., Vonrhein, C. et al. (2009). Crystal structure of a nucleocapsid-like nucleoprotein-RNA complex of respiratory syncytial virus. *Science* 326: 1279–1283.
- Griffiths, C., Drews, S.J., and Marchant, D.J. (2017). Respiratory syncytial virus: infection, detection, and new options for prevention and treatment. *Clin. Microbiol. Rev.* 30: 277–319.
- Collins, P.L., Dickens, L.E., Buckler-White, A. et al. (1986). Nucleotide sequences for the gene junctions of human respiratory syncytial virus reveal distinctive features of intergenic structure and gene order. *Proc. Natl. Acad. Sci. U. S. A.* 83: 4594–4598.
- 33 Collins, P.L. and Wertz, G.W. (1983). cDNA cloning and transcriptional mapping of nine polyadenylylated RNAs encoded by the genome of human respiratory syncytial virus. *Proc. Natl. Acad. Sci. U. S. A.* 80: 3208–3212.
- Sedeyn, K., Schepens, B., and Saelens, X. (2019). Respiratory syncytial virus nonstructural proteins 1 and 2: Exceptional disrupters of innate immune responses. *PLoS Pathog.* 15: e1007984.
- Bermingham, A. and Collins, P.L. (1999). The M2-2 protein of human respiratory syncytial virus is a regulatory factor involved in the balance between RNA replication and transcription. *Proc. Natl. Acad. Sci. U. S. A.* 96: 11259–11264.
- Fearns, R., Peeples, M.E., and Collins, P.L. (2002). Mapping the transcription and replication promoters of respiratory syncytial virus. *J. Virol.* 76: 1663–1672.
- **37** Tremaglio, C.Z., Noton, S.L., Deflube, L.R., and Fearns, R. (2013). Respiratory syncytial virus polymerase can initiate transcription from position 3 of the leader promoter. *J. Virol.* 87: 3196–3207.
- Kuo, L., Fearns, R., and Collins, P.L. (1997). Analysis of the gene start and gene end signals of human respiratory syncytial virus: quasi-templated initiation at position 1 of the encoded mRNA. *J. Virol.* 71: 4944–4953.
- Kuo, L., Grosfeld, H., Cristina, J. et al. (1996). Effects of mutations in the gene-start and gene-end sequence motifs on transcription of monocistronic and dicistronic minigenomes of respiratory syncytial virus. *J. Virol.* 70: 6892–6901.
- Fearns, R., Collins, P.L., and Peeples, M.E. (2000). Functional analysis of the genomic and antigenomic promoters of human respiratory syncytial virus. *J. Virol.* 74: 6006–6014.

- **41** Noton, S.L., Deflube, L.R., Tremaglio, C.Z., and Fearns, R. (2012). The respiratory syncytial virus polymerase has multiple RNA synthesis activities at the promoter. *PLoS Pathog.* 8: e1002980.
- **42** Peeples, M.E. and Collins, P.L. (2000). Mutations in the 5' trailer region of a respiratory syncytial virus minigenome which limit RNA replication to one step. *J. Virol.* 74: 146–155.
- **43** Cao, D. and Liang, B. (2021). Cryo-electron microscopy structures of the pneumoviridae polymerases. *Viral Immunol.* 34: 18–26.
- **44** Fearns, R. and Plemper, R.K. (2017). Polymerases of paramyxoviruses and pneumoviruses. *Virus Res.* 234: 87–102.
- 45 Liang, B. (2020). Structures of the mononegavirales polymerases. J. Virol. 94.
- **46** Fearns, R. and Deval, J. (2016). New antiviral approaches for respiratory syncytial virus and other mononegaviruses: inhibiting the RNA polymerase. *Antivir. Res.* 134: 63–76.
- **47** Cao, D., Gao, Y., Roesler, C. et al. (2020). Cryo-EM structure of the respiratory syncytial virus RNA polymerase. *Nat. Commun.* 11: 368.
- **48** Gilman, M.S.A., Liu, C., Fung, A. et al. (2019). Structure of the respiratory syncytial virus polymerase complex. *Cell* 179: 193–204 e14.
- **49** Mazumder, B., Adhikary, G., and Barik, S. (1994). Bacterial expression of human respiratory syncytial viral phosphoprotein P and identification of Ser237 as the site of phosphorylation by cellular casein kinase II. *Virology* 205: 93–103.
- 50 Deval, J., Hong, J., Wang, G. et al. (2015). Molecular basis for the selective inhibition of respiratory syncytial virus RNA polymerase by 2'-fluoro-4'-chloromethyl-cytidine triphosphate. *PLoS Pathog.* 11: e1004995.
- 51 Warren, T.K., Jordan, R., Lo, M.K. et al. (2016). Therapeutic efficacy of the small molecule GS-5734 against Ebola virus in rhesus monkeys. *Nature* https://doi.org/10.1038/ nature17180.
- 52 Barik, S. (1993). The structure of the 5' terminal cap of the respiratory syncytial virus mRNA. J. Gen. Virol. 74 (Pt 3): 485–490.
- **53** Liuzzi, M., Mason, S.W., Cartier, M. et al. (2005). Inhibitors of respiratory syncytial virus replication target cotranscriptional mRNA guanylylation by viral RNA-dependent RNA polymerase. *J. Virol.* 79: 13105–13115.
- **54** Braun, M.R., Deflube, L.R., Noton, S.L. et al. (2017). RNA elongation by respiratory syncytial virus polymerase is calibrated by conserved region V. *PLoS Pathog.* 13: e1006803.
- **55** Ogino, T. and Banerjee, A.K. (2007). Unconventional mechanism of mRNA capping by the RNA-dependent RNA polymerase of vesicular stomatitis virus. *Mol. Cell* 25: 85–97.
- **56** Ogino, T. and Banerjee, A.K. (2011). An unconventional pathway of mRNA cap formation by vesiculoviruses. *Virus Res.* 162: 100–109.
- 57 Ogino, T., Yadav, S.P., and Banerjee, A.K. (2010). Histidine-mediated RNA transfer to GDP for unique mRNA capping by vesicular stomatitis virus RNA polymerase. *Proc. Natl. Acad. Sci. U. S. A.* 107: 3463–3468.
- 58 Horwitz, J.A., Jenni, S., Harrison, S.C., and Whelan, S.P.J. (2020). Structure of a rabies virus polymerase complex from electron cryo-microscopy. *Proc. Natl. Acad. Sci. U. S. A.* 117: 2099–2107.
- **59** Liang, B., Li, Z., Jenni, S. et al. (2015). Structure of the L protein of vesicular stomatitis virus from electron cryomicroscopy. *Cell* 162: 314–327.

- Ogino, M., Gupta, N., Green, T.J., and Ogino, T. (2019). A dual-functional priming-capping loop of rhabdoviral RNA polymerases directs terminal de novo initiation and capping intermediate formation. *Nucleic Acids Res.* 47: 299–309.
- Paesen, G.C., Collet, A., Sallamand, C. et al. (2015). X-ray structure and activities of an essential Mononegavirales L-protein domain. *Nat. Commun.* 6: 8749.
- Sutto-Ortiz, P., Tcherniuk, S., Ysebaert, N. et al. (2021). The methyltransferase domain of the Respiratory Syncytial Virus L protein catalyzes cap N7 and 2'-O-methylation. *PLoS Pathog.* 17: e1009562.
- Pan, J., Qian, X., Lattmann, S. et al. (2020). Structure of the human metapneumovirus polymerase phosphoprotein complex. *Nature* 577: 275–279.
- 64 Abdella, R., Aggarwal, M., Okura, T. et al. (2020). Structure of a paramyxovirus polymerase complex reveals a unique methyltransferase-CTD conformation. *Proc. Natl. Acad. Sci.* U. S. A. 117: 4931–4941.
- Jenni, S., Bloyet, L.M., Diaz-Avalos, R. et al. (2020). Structure of the vesicular stomatitis virus L protein in complex with its phosphoprotein cofactor. *Cell Rep.* 30: 53–60 e5.
- Noval, M.G., Esperante, S.A., Molina, I.G. et al. (2016). Intrinsic disorder to order transitions in the scaffold phosphoprotein P from the respiratory syncytial virus RNA polymerase complex. *Biochemistry* 55: 1441–1454.
- Pereira, N., Cardone, C., Lassoued, S. et al. (2017). New insights into structural disorder in human respiratory syncytial virus phosphoprotein and implications for binding of protein partners. *J. Biol. Chem.* 292: 2120–2131.
- Alvarez Paggi, D., Esperante, S.A., Salgueiro, M. et al. (2019). A conformational switch balances viral RNA accessibility and protection in a nucleocapsid ring model. *Arch. Biochem. Biophys.* 671: 77–86.
- **69** Ouizougun-Oubari, M., Pereira, N., Tarus, B. et al. (2015). A druggable pocket at the nucleocapsid/phosphoprotein interaction site of human respiratory syncytial virus. *J. Virol.* 89: 11129–11143.
- Tran, T.L., Castagne, N., Bhella, D. et al. (2007). The nine C-terminal amino acids of the respiratory syncytial virus protein P are necessary and sufficient for binding to ribonucleoprotein complexes in which six ribonucleotides are contacted per N protein protomer. *J. Gen. Virol.* 88: 196–206.
- 71 Esneau, C., Raynal, B., Roblin, P. et al. (2019). Biochemical characterization of the respiratory syncytial virus N(0)-P complex in solution. *J. Biol. Chem.* 294: 3647–3660.
- **72** Galloux, M., Gabiane, G., Sourimant, J. et al. (2015). Identification and characterization of the binding site of the respiratory syncytial virus phosphoprotein to RNA-free nucleoprotein. *J. Virol.* 89: 3484–3496.
- **73** Esperante, S.A., Paris, G., and de Prat-Gay, G. (2012). Modular unfolding and dissociation of the human respiratory syncytial virus phosphoprotein p and its interaction with the m(2-1) antiterminator: a singular tetramer-tetramer interface arrangement. *Biochemistry* 51: 8100–8110.
- **74** Tran, T.L., Castagne, N., Dubosclard, V. et al. (2009). The respiratory syncytial virus M2-1 protein forms tetramers and interacts with RNA and P in a competitive manner. *J. Virol.* 83: 6363–6374.

- **75** Galloux, M., Risso-Ballester, J., Richard, C.A. et al. (2020). Minimal Elements Required for the Formation of Respiratory Syncytial Virus Cytoplasmic Inclusion Bodies in vivo and in vitro. *MBio*: 11.
- **76** Rincheval, V., Lelek, M., Gault, E. et al. (2017). Functional organization of cytoplasmic inclusion bodies in cells infected by respiratory syncytial virus. *Nat. Commun.* 8: 563.
- 77 Risso-Ballester, J., Galloux, M., Cao, J. et al. (2021). A condensate-hardening drug blocks RSV replication in vivo. *Nature* 595: 596–599.
- 78 Tanner, S.J., Ariza, A., Richard, C.A. et al. (2014). Crystal structure of the essential transcription antiterminator M2-1 protein of human respiratory syncytial virus and implications of its phosphorylation. *Proc. Natl. Acad. Sci. U. S. A.* 111: 1580–1585.
- **79** Collins, P.L., Hill, M.G., Cristina, J., and Grosfeld, H. (1996). Transcription elongation factor of respiratory syncytial virus, a nonsegmented negative-strand RNA virus. *Proc. Natl. Acad. Sci. U. S. A.* 93: 81–85.
- **80** Fearns, R. and Collins, P.L. (1999). Role of the M2-1 transcription antitermination protein of respiratory syncytial virus in sequential transcription. *J. Virol.* 73: 5852–5864.
- **81** Hardy, R.W. and Wertz, G.W. (1998). The product of the respiratory syncytial virus M2 gene ORF1 enhances readthrough of intergenic junctions during viral transcription. *J. Virol.* 72: 520–526.
- 82 Collins, P.L., Hill, M.G., Camargo, E. et al. (1995). Production of infectious human respiratory syncytial virus from cloned cDNA confirms an essential role for the transcription elongation factor from the 5' proximal open reading frame of the M2 mRNA in gene expression and provides a capability for vaccine development. *Proc. Natl. Acad. Sci. U. S. A.* 92: 11563–11567.
- **83** Braun, M.R., Noton, S.L., Blanchard, E.L. et al. (2021). Respiratory syncytial virus M2-1 protein associates non-specifically with viral messenger RNA and with specific cellular messenger RNA transcripts. *PLoS Pathog.* 17: e1009589.
- **84** Tripp, R.A., Power, U.F., Openshaw, P.J.M., and Kauvar, L.M. (2018). Respiratory syncytial virus: targeting the G protein provides a new approach for an old problem. *J. Virol.* 92: e01302–17.
- 85 Hendricks, D.A., Baradaran, K., McIntosh, K., and Patterson, J.L. (1987). Appearance of a soluble form of the G protein of respiratory syncytial virus in fluids of infected cells. *J. Gen. Virol.* 68 (Pt 6): 1705–1714.
- **86** Gebber, G.L. and Barman, S.M. (1977). Brain stem vasomotor circuits involved in the genesis and entrainment of sympathetic nervous rhythms. *Prog. Brain Res.* 47: 61–75.
- **87** McLellan, J.S., Chen, M., Joyce, M.G. et al. (2013). Structure-based design of a fusion glycoprotein vaccine for respiratory syncytial virus. *Science* 342: 592–598.
- **88** McLellan, J.S., Chen, M., Leung, S. et al. (2013). Structure of RSV fusion glycoprotein trimer bound to a prefusion-specific neutralizing antibody. *Science* 340: 1113–1117.
- **89** Rixon, H.W., Brown, G., Aitken, J. et al. (2004). The small hydrophobic (SH) protein accumulates within lipid-raft structures of the Golgi complex during respiratory syncytial virus infection. *J. Gen. Virol.* 85: 1153–1165.
- **90** Collins, P.L. and Wertz, G.W. (1985). The 1A protein gene of human respiratory syncytial virus: nucleotide sequence of the mRNA and a related polycistronic transcript. *Virology* 141: 283–291.

- **126** 4 Respiratory Syncytial Virus Immunoreactivity, Vaccine Development, and Therapeutics
 - **91** Bukreyev, A., Whitehead, S.S., Murphy, B.R., and Collins, P.L. (1997). The recombinant respiratory syncytial virus from which the entire SH gene has been deleted grows efficiently in cell culture and exhibits site-specific attenuation in the mouse's respiratory tract. *J. Virol.* 71: 8973–8982.
 - **92** Carter, S.D., Dent, K.C., Atkins, E. et al. (2010). Direct visualization of the small hydrophobic protein of human respiratory syncytial virus reveals the structural basis for membrane permeability. *FEBS Lett.* 584: 2786–2790.
 - **93** Araujo, G.C., Silva, R.H., Scott, L.P. et al. (2016). Structure and functional dynamics characterization of the ion channel of the human respiratory syncytial virus (hRSV) small hydrophobic protein (SH) transmembrane domain by combining molecular dynamics with excited normal modes. *J. Mol. Model.* 22: 286.
 - **94** Fuentes, S., Tran, K.C., Luthra, P. et al. (2007). Function of the respiratory syncytial virus small hydrophobic protein. *J. Virol.* 81: 8361–8366.
 - **95** Pollock, N., Taylor, G., Jobe, F., and Guzman, E. (2017). Modulation of the transcription factor NF-kappaB in antigen-presenting cells by bovine respiratory syncytial virus small hydrophobic protein. *J. Gen. Virol.* 98: 1587–1599.
 - **96** Langley, J.M., MacDonald, L.D., Weir, G.M. et al. (2018). A respiratory syncytial virus vaccine based on the small hydrophobic protein ectodomain presented with a novel lipid-based formulation is highly immunogenic and safe in adults: a first-in-humans Study. *J. Infect. Dis.* 218: 378–387.
 - **97** Liang, X., Liu, D.H., Chen, G.L. et al. (2019). Gradual replacement of all previously circulating respiratory syncytial virus A strain with the novel ON1 genotype in Lanzhou from 2010 to 2017. *Medicine (Baltimore)* 98: e15542.
 - **98** Bilawchuk, L.M., Griffiths, C.D., Jensen, L.D. et al. (2017). The susceptibilities of respiratory syncytial virus to nucleolin receptor blocking and antibody neutralization are dependent upon the method of virus purification. *Viruses* 9: 207.
 - **99** Hall, C.B., Douglas, R.G. Jr., and Geiman, J.M. (1980). Possible transmission by fomites of respiratory syncytial virus. *J. Infect. Dis.* 141: 98–102.
 - **100** Cruz-Sanchez, T.M., Haddrell, A.E., Hackett, T.L. et al. (2013). Formation of a stable mimic of ambient particulate matter containing viable infectious respiratory syncytial virus and its dry-deposition directly onto cell cultures. *Anal. Chem.* 85: 898–906.
 - 101 Carlsen, K.H., Mellbye, O.J., Fuglerud, P. et al. (1993). Serum immunoglobulin G subclasses and serum immunoglobulin A in acute bronchiolitis in infants. *Pediatr. Allergy Immunol.* 4: 20–25.
 - **102** Habibi, M.S., Jozwik, A., Makris, S. et al. (2015). Impaired antibody-mediated protection and defective IgA B-cell memory in experimental infection of adults with respiratory syncytial virus. *Am. J. Respir. Crit. Care Med.* 191: 1040–1049.
 - 103 Suzuki, T., Ainai, A., and Hasegawa, H. (2017). Functional and structural characteristics of secretory IgA antibodies elicited by mucosal vaccines against influenza virus. *Vaccine* 35: 5297–5302.
 - **104** Downham, M.A., Scott, R., Sims, D.G. et al. (1976). Breast-feeding protects against respiratory syncytial virus infections. *Br. Med. J.* 2: 274–276.
 - **105** Hall, C.B., Walsh, E.E., Long, C.E., and Schnabel, K.C. (1991). Immunity to and frequency of reinfection with respiratory syncytial virus. *J. Infect. Dis.* 163: 693–698.

- **106** Shehata, L., Wieland-Alter, W.F., Maurer, D.P. et al. (2019). Systematic comparison of respiratory syncytial virus-induced memory B cell responses in two anatomical compartments. *Nat. Commun.* 10: 1126.
- **107** Glezen, W.P., Taber, L.H., Frank, A.L., and Kasel, J.A. (1986). Risk of primary infection and reinfection with respiratory syncytial virus. *Am. J. Dis. Child.* 140: 543–546.
- 108 Figueras-Aloy, J., Manzoni, P., Paes, B. et al. (2016). Defining the risk and associated morbidity and mortality of severe respiratory syncytial virus infection among preterm infants without chronic lung disease or congenital heart disease. *Infect. Dis. Ther.* 5: 417–452.
- **109** Paes, B., Fauroux, B., Figueras-Aloy, J. et al. (2016). Defining the risk and associated morbidity and mortality of severe respiratory syncytial virus infection among infants with chronic lung disease. *Infect. Dis. Ther.* 5: 453–471.
- **110** Piedimonte, G. and Perez, M.K. (2014). Respiratory syncytial virus infection and bronchiolitis. *Pediatr. Rev.* 35: 519–530.
- **111** Henrickson, K.J., Hoover, S., Kehl, K.S., and Hua, W. (2004). National disease burden of respiratory viruses detected in children by polymerase chain reaction. *Pediatr. Infect. Dis. J.* 23: S11–S18.
- **112** Yoshihara, S., Kusuda, S., Mochizuki, H. et al. (2013). Effect of palivizumab prophylaxis on subsequent recurrent wheezing in preterm infants. *Pediatrics* 132: 811–818.
- **113** Sigurs, N., Aljassim, F., Kjellman, B. et al. (2010). Asthma and allergy patterns over 18 years after severe RSV bronchiolitis in the first year of life. *Thorax* 65: 1045–1052.
- **114** Carbonell-Estrany, X., Perez-Yarza, E.G., Garcia, L.S. et al. (2015). Long-term burden and respiratory effects of respiratory syncytial virus hospitalization in preterm infants-the SPRING study. *PLoS One* 10: e0125422.
- **115** Fauroux, B., Simoes, E.A.F., Checchia, P.A. et al. (2017). The burden and long-term respiratory morbidity associated with respiratory syncytial virus infection in early childhood. *Infect. Dis. Ther.* 6: 173–197.
- **116** Blanken, M.O., Rovers, M.M., Molenaar, J.M. et al. (2013). Respiratory syncytial virus and recurrent wheeze in healthy preterm infants. *N. Engl. J. Med.* 368: 1791–1799.
- **117** Ducharme, F.M., Tse, S.M., and Chauhan, B. (2014). Diagnosis, management, and prognosis of preschool wheeze. *Lancet* 383: 1593–1604.
- **118** Simoes, E.A., Groothuis, J.R., Carbonell-Estrany, X. et al. (2007). Palivizumab prophylaxis, respiratory syncytial virus, and subsequent recurrent wheezing. *J. Pediatr.* 151 (34–42): 42 e1.
- **119** Simoes, M., Inoue, Y., Matsunaga, N.Y. et al. (2018). Recurrent wheezing in preterm infants: prevalence and risk factors. *J. Pediatr*.
- **120** Simoes, E.A., Carbonell-Estrany, X., Rieger, C.H. et al. (2010). The effect of respiratory syncytial virus on subsequent recurrent wheezing in atopic and nonatopic children. *J. Allergy Clin. Immunol.* 126: 256–262.
- **121** Walsh, E.E. and Falsey, A.R. (2012). Respiratory syncytial virus infection in adult populations. *Infect. Disord. Drug Targets* 12: 98–102.
- **122** Scheltema, N.M., Nibbelke, E.E., Pouw, J. et al. (2018). Respiratory syncytial virus prevention and asthma in healthy preterm infants: a randomised controlled trial. *Lancet Respir. Med.* 6: 257–264.

- Igde, M., Kabasakal, H., Ozturk, O. et al. (2018). Palivizumab prophylaxis, respiratory syncytial virus and subsequent development of asthma. *Minerva Pediatr.* 70: 252–259.
- Lee, N., Lui, G.C., Wong, K.T. et al. (2013). High morbidity and mortality in adults hospitalized for respiratory syncytial virus infections. *Clin. Infect. Dis.* 57: 1069–1077.
- Lee, N. and Qureshi, S.T. (2013). Other viral pneumonias: coronavirus, respiratory syncytial virus, adenovirus, hantavirus. *Crit. Care Clin.* 29: 1045–1068.
- 126 Shi, T., Denouel, A., Tietjen, A.K. et al. (2019). Global disease burden estimates of respiratory syncytial virus-associated acute respiratory infection in older adults in 2015: a systematic review and meta-analysis. J. Infect. Dis. 222 (Suppl 7): 577–583.
- Zhou, J.A., Schweinle, J.E., Lichenstein, R. et al. (2019). Severe illnesses associated with outbreaks of respiratory syncytial virus and influenza in adults. *Clin. Infect. Dis.*
- Saravanos, G.L., Sheel, M., Homaira, N. et al. (2019). Respiratory syncytial virusassociated hospitalisations in Australia, 2006-2015. *Med. J. Aust.* 210: 447–453.
- Ackerson, B., Tseng, H.F., Sy, L.S. et al. (2018). Severe morbidity and mortality associated with respiratory syncytial virus versus influenza infection in hospitalized older adults. *Clin. Infect. Dis.* 69: 197–203.
- Batista, M.V., El Haddad, L., and Chemaly, R.F. (2018). Paramyxovirus infections in hematopoietic cell transplant recipients. *Curr. Opin. Infect. Dis.* 31: 542–552.
- Wong, S.S., Yu, J.W., Wong, K.T. et al. (2014). Initial radiographic features as outcome predictor of adult respiratory syncytial virus respiratory tract infection. *AJR Am. J. Roentgenol.* 203: 280–286.
- Lee, N., Chan, M.C., Lui, G.C. et al. (2015). High viral load and respiratory failure in adults hospitalized for respiratory syncytial virus infections. *J. Infect. Dis.* 212: 1237–1240.
- Walsh, E.E., Peterson, D.R., Kalkanoglu, A.E. et al. (2013). Viral shedding and immune responses to respiratory syncytial virus infection in older adults. *J. Infect. Dis.* 207: 1424–1432.
- Lee, N., Walsh, E.E., Sander, I. et al. (2019). Delayed diagnosis of respiratory syncytial virus infections in hospitalized adults: individual patient data, chart review analysis and physician survey in the USA. *J. Infect. Dis.* 220: 969–979.
- 135 Kim, H.W., Canchola, J.G., Brandt, C.D. et al. (1969). Respiratory syncytial virus disease in infants despite prior administration of antigenic inactivated vaccine. *Am. J. Epidemiol.* 89: 422–434.
- Delgado, M.F., Coviello, S., Monsalvo, A.C. et al. (2009). Lack of antibody affinity maturation due to poor toll-like receptor stimulation leads to enhanced respiratory syncytial virus disease. *Nat. Med.* 15: 34–41.
- Killikelly, A.M., Kanekiyo, M., and Graham, B.S. (2016). Pre-fusion F is absent on the surface of formalin-inactivated respiratory syncytial virus. *Sci. Rep.* 6: 34108.
- Groothuis, J.R., King, S.J., Hogerman, D.A. et al. (1998). Safety and immunogenicity of a purified F protein respiratory syncytial virus (PFP-2) vaccine in seropositive children with bronchopulmonary dysplasia. *J. Infect. Dis.* 177: 467–469.
- Piedra, P.A., Glezen, W.P., Kasel, J.A. et al. (1995). Safety and immunogenicity of the PFP vaccine against respiratory syncytial virus (RSV): the western blot assay helps distinguish immune responses from the PFP vaccine from RSV infection. *Vaccine* 13: 1095–1101.

- Piedra, P.A., Grace, S., Jewell, A. et al. (1996). Purified fusion protein vaccine protects against lower respiratory tract illness during respiratory syncytial virus season in children with cystic fibrosis. *Pediatr. Infect. Dis. J.* 15: 23–31.
- Piedra, P.A., Cron, S.G., Jewell, A. et al. (2003). Immunogenicity of a new purified fusion protein vaccine to respiratory syncytial virus: a multi-center trial in children with cystic fibrosis. *Vaccine* 21: 2448–2460.
- Gonzalez, I.M., Karron, R.A., Eichelberger, M. et al. (2000). The live attenuated cpts 248/404 RSV vaccine is evaluated in combination with a subunit RSV vaccine (PFP-2) in healthy young and older adults. *Vaccine* 18: 1763–1772.
- Falloon, J., Talbot, H.K., Curtis, C. et al. (2017). Dose selection for an adjuvanted respiratory syncytial virus F protein vaccine for older adults based on humoral and cellular immune responses. *Clin. Vaccine Immunol.* 24: e00157–17.
- 144 Van Hoeven, N., Fox, C.B., Granger, B. et al. (2017). A formulated TLR7/8 agonist is a flexible, highly potent and effective adjuvant for pandemic influenza vaccines. *Sci. Rep.* 7: 46426.
- Johnson, T.R., Rao, S., Seder, R.A. et al. (2009). TLR9 agonist, but not TLR7/8, functions as an adjuvant to diminish FI-RSV vaccine-enhanced disease, while either agonist used as therapy during primary RSV infection increases disease severity. *Vaccine* 27: 3045–3052.
- Gilman, M.S.A., Furmanova-Hollenstein, P., Pascual, G. et al. (2019). Transient opening of trimeric prefusion RSV F proteins. *Nat. Commun.* 10: 2105.
- Jones, H.G., Battles, M.B., Lin, C.C. et al. (2019). Alternative conformations of a major antigenic site on RSV F. *PLoS Pathog.* 15: e1007944.
- Meng, J., Lee, S., Hotard, A.L., and Moore, M.L. (2014). Refining the balance of attenuation and immunogenicity of respiratory syncytial virus by targeted codon deoptimization of virulence genes. *MBio* 5: e01704–e01714.
- 149 Stobart, C.C., Rostad, C.A., Ke, Z. et al. (2016). A live RSV vaccine with engineered thermostability is immunogenic in cotton rats despite high attenuation. *Nat. Commun.* 7: 13916.
- Rostad, C.A., Stobart, C.C., Todd, S.O. et al. (2018). Enhancing the thermostability and immunogenicity of a respiratory syncytial virus (RSV) live-attenuated vaccine by incorporating unique RSV line19F protein residues. *J. Virol.* 92.
- **151** Rostad, C.A., Stobart, C.C., Gilbert, B.E. et al. (2016). A recombinant respiratory syncytial virus vaccine candidate attenuated by a low-fusion F protein is immunogenic and protective against challenge in cotton rats. *J. Virol.* 90: 7508–7518.
- Alwan, W.H., Record, F.M., and Openshaw, P.J. (1992). CD4+ T cells clear virus but augment disease in mice infected with respiratory syncytial virus. Comparison with the effects of CD8+ T cells. *Clin. Exp. Immunol.* 88: 527–536.
- Liu, J., Haddad, E.K., Marceau, J. et al. (2016). A numerically subdominant CD8 T cell response to matrix protein of respiratory syncytial virus controls infection with limited immunopathology. *PLoS Pathog.* 12: e1005486.
- 154 Olmsted, R.A., Elango, N., Prince, G.A. et al. (1986). Expression of the F glycoprotein of respiratory syncytial virus by a recombinant vaccinia virus: comparison of the individual contributions of the F and G glycoproteins to host immunity. *Proc. Nat. Acad. Sci. U. S. A.* 83 (19): 7462–7466.

- **130** 4 *Respiratory Syncytial Virus Immunoreactivity, Vaccine Development, and Therapeutics*
 - 155 Groothuis, J.R., Levin, M.J., Rodriguez, W. et al. (1991). Use of intravenous gamma globulin to passively immunize high-risk children against respiratory syncytial virus: safety and pharmacokinetics. The RSVIG Study Group. *Antimicrob. Agents Chemother.* 35: 1469–1473.
 - 156 Groothuis, J.R., Simoes, E.A., Levin, M.J. et al. (1993). Prophylactic administration of respiratory syncytial virus immune globulin to high-risk infants and young children. The Respiratory Syncytial Virus Immune Globulin Study Group. N. Engl. J. Med. 329: 1524–1530.
 - 157 Rodriguez, W.J., Gruber, W.C., Welliver, R.C. et al. (1997). Respiratory syncytial virus (RSV) immune globulin intravenous therapy for RSV lower respiratory tract infection in infants and young children at high risk for severe RSV infections: Respiratory Syncytial Virus Immune Globulin Study Group. *Pediatrics* 99: 454–461.
 - **158** Anonymous (1997). Respiratory syncytial virus immune globulin intravenous: indications for use. American Academy of Pediatrics Committee on Infectious Diseases, Committee on Fetus and Newborn. *Pediatrics* 99: 645–650.
 - **159** Weltzin, R., Traina-Dorge, V., Soike, K. et al. (1996). Intranasal monoclonal IgA antibody to respiratory syncytial virus protects rhesus monkeys against upper and lower respiratory tract infection. *J. Infect. Dis.* 174: 256–261.
 - **160** Johnson, S., Oliver, C., Prince, G.A. et al. (1997). Development of a humanized monoclonal antibody (MEDI-493) with potent in vitro and in vivo activity against respiratory syncytial virus. *J. Infect. Dis.* 176: 1215–1224.
 - **161** Zhu, Q., McLellan, J.S., Kallewaard, N.L. et al. (2017). A highly potent extended half-life antibody as a potential RSV vaccine surrogate for all infants. *Sci. Transl. Med.*: 9.
 - **162** Hu, J. and Robinson, J.L. (2010). Treatment of respiratory syncytial virus with Palivizumab: a systematic review. *World J. Pediatr.* 6: 296–300.
 - **163** Fenton, C., Scott, L.J., and Plosker, G.L. (2004). Palivizumab: a review of its use as prophylaxis for serious respiratory syncytial virus infection. *Paediatr. Drugs* 6: 177–197.
 - **164** Anonymous (1998). Palivizumab, a humanized respiratory syncytial virus monoclonal antibody, reduces hospitalization from respiratory syncytial virus infection in high-risk infants. *Pediatrics* 102: 531–537.
 - **165** Munoz-Durango, N., Pizarro-Ortega, M.S., Rey-Jurado, E. et al. (2018). Patterns of antibody response during natural hRSV infection: insights for the development of new antibody-based therapies. *Expert Opin. Investig. Drugs* 27: 721–731.
 - **166** Feld, J.J., and Hoofnagle, J.H. (2005). Mechanism of action of interferon and ribavirin in treatment of hepatitis C. *Nature* 436 (7053): 967–972.
 - **167** Foolad, F., Aitken, S.L., Shigle, T.L. et al. (2018). Oral versus aerosolized ribavirin for the treatment of respiratory syncytial virus infections in hematopoietic cell transplantation recipients. *Clin. Infect. Dis.* 68: 1641–1649.
 - **168** Manuel, O., Estabrook, M., and American Society of Transplantation Infectious Diseases Community of P (2019). RNA respiratory viral infections in solid organ transplant recipients: guidelines from the American Society of Transplantation Infectious Diseases Community of Practice. *Clin. Transpl.* 33: e13511.
 - **169** Shah, J.N. and Chemaly, R.F. (2011). Management of RSV infections in adult recipients of hematopoietic stem cell transplantation. *Blood* 117: 2755–2763.
 - **170** Alansari, K., Toaimah, F.H., Almatar, D.H. et al. (2019). Monoclonal antibody treatment of RSV bronchiolitis in young infants: a randomized trial. *Pediatrics* 143: e20182308.

- **171** Falsey, A.R., Koval, C., DeVincenzo, J.P., and Walsh, E.E. (2017). Compassionate use experience with high-titer respiratory syncytical virus (RSV) immunoglobulin in RSV-infected immunocompromised persons. *Transpl. Infect. Dis.* 19.
- **172** Wasserman, R.L., Lumry, W., Harris, J. 3rd et al. (2016). Efficacy, safety, and pharmacokinetics of a new 10% liquid intravenous immunoglobulin containing high titer neutralizing antibody to RSV and other respiratory viruses in subjects with primary immunodeficiency disease. *J. Clin. Immunol.* 36: 590–599.
- **173** Ramilo, O., Lagos, R., Saez-Llorens, X. et al. (2014). Motavizumab treatment of infants hospitalized with respiratory syncytial virus infection does not decrease viral load or severity of illness. *Pediatr. Infect. Dis. J.* 33: 703–709.
- 174 Griffin, M.P., Khan, A.A., Esser, M.T. et al. (2017). Safety, tolerability, and pharmacokinetics of MEDI8897, the respiratory syncytial virus Prefusion F-targeting monoclonal antibody with an extended half-life, in healthy adults. *Antimicrob. Agents Chemother*. 61: e01714–16.
- 175 Domachowske, J.B., Khan, A.A., Esser, M.T. et al. (2018). Safety, tolerability and pharmacokinetics of MEDI8897, an extended half-life single-dose respiratory syncytial virus prefusion F-targeting monoclonal antibody administered as a single dose to healthy preterm infants. *Pediatr. Infect. Dis. J.* 37: 886–892.
- **176** DeVincenzo, J.P., Whitley, R.J., Mackman, R.L. et al. (2014). Oral GS-5806 activity in a respiratory syncytial virus challenge study. *N. Engl. J. Med.* 371: 711–722.
- 177 Beigel, J.H., Nam, H.H., Adams, P.L. et al. (2019). Advances in respiratory virus therapeutics a meeting report from the 6th isirv antiviral group conference. *Antivir. Res.* 167: 45–67.
- 178 Perron, M., Stray, K., Kinkade, A. et al. (2015). GS-5806 inhibits a broad range of respiratory syncytial virus clinical isolates by blocking the virus-cell fusion process. *Antimicrob. Agents Chemother.* 60: 1264–1273.
- 179 Deval, J., Hong, J., Wang, G. et al. (2015). Molecular basis for the selective inhibition of respiratory syncytial virus RNA polymerase by 2'-fluoro-4'-chloromethyl-cytidine triphosphate. *PLoS Pathog.* 11 (6): e1004995.
- **180** DeVincenzo, J.P., McClure, M.W., Symons, J.A. et al. (2015). Activity of Oral ALS-008176 in a respiratory syncytial virus challenge study. *N. Engl. J. Med.* 373: 2048–2058.
- **181** Patel, K., Kirkpatrick, C.M., Nieforth, K.A. et al. (2019). Respiratory syncytial virus-A dynamics and the effects of lumicitabine, a nucleoside viral replication inhibitor, in experimentally infected humans. *J. Antimicrob. Chemother*. 74: 442–452.
- **182** Stevens, M., Rusch, S., DeVincenzo, J. et al. (2018). Antiviral activity of oral JNJ-53718678 in healthy adult volunteers challenged with respiratory syncytial virus: a placebocontrolled study. *J. Infect. Dis.* 218: 748–756.
- **183** Roymans, D., Alnajjar, S.S., Battles, M.B. et al. (2017). Therapeutic efficacy of a respiratory syncytial virus fusion inhibitor. *Nat. Commun.* 8: 167.
- 184 DeVincenzo, J., Tait, D., Oluwayi, O. et al. Safety and efficacy of oral RV521 in a human respiratory syncytial virus (RSV) phase 2a challenge study. *Am. J. Respir. Crit. Care Med.* 197: A7715.
- **185** Selvaraj, M., Yegambaram, K., Todd, E. et al. (2018). The structure of the human respiratory syncytial virus M2-1 protein bound to the interaction domain of the phosphoprotein P defines the orientation of the complex. *MBio* 9: e01554–18.

132 4 Respiratory Syncytial Virus Immunoreactivity, Vaccine Development, and Therapeutics

- 186 Richard, C.A., Rincheval, V., Lassoued, S. et al. (2018). RSV hijacks cellular protein phosphatase 1 to regulate M2-1 phosphorylation and viral transcription. *PLoS Pathog.* 14: e1006920.
- 187 Yoon, J.J., Toots, M., Lee, S. et al. (2018). Orally efficacious broad-spectrum ribonucleoside analog inhibitor of influenza and respiratory syncytial viruses. *Antimicrob. Agents Chemother.* 62: e00766–18.
- **188** Henle, W. and Henle, G. (1943). Interference of inactive virus with the propagation of virus of influenza. *Science* 98: 87–89.
- **189** Piacentini, S., La Frazia, S., Riccio, A. et al. (2018). Nitazoxanide inhibits paramyxovirus replication by targeting the fusion protein folding: role of glycoprotein-specific thiol oxidoreductase ERp57. *Sci. Rep.* 8: 10425.
- **190** Brendish, N.J. and Clark, T.W. (2017). Antiviral treatment of severe non-influenza respiratory virus infection. *Curr. Opin. Infect. Dis.* 30: 573–578.
- **191** Tchesnokov, E.P., Feng, J.Y., Porter, D.P., and Gotte, M. (2019). Mechanism of inhibition of Ebola virus RNA-dependent RNA polymerase by Remdesivir. *Viruses* 11: 326.
- **192** Warren, T., Jordan, R., Lo, M. et al. (2016). Therapeutic efficacy of the small molecule GS-5734 against Ebola virus in rhesus monkeys. *Nature*. 531: 381–385.
- **193** Beigel, J.H., Tomashek, K.M., Dodd, L.E. et al. (2020). Remdesivir for the treatment of Covid-19 final report. *N. Engl. J. Med.* 383: 1813–1826.
- **194** Hassanipour, S., Arab-Zozani, M., Amani, B. et al. (2021). The efficacy and safety of Favipiravir in treatment of COVID-19: a systematic review and meta-analysis of clinical trials. *Sci. Rep.* 11: 11022.
- **195** Zhu, Q., Patel, N.K., McAuliffe, J.M. et al. (2012). Natural polymorphisms and resistanceassociated mutations in the fusion protein of respiratory syncytial virus (RSV): effects on RSV susceptibility to Palivizumab. *J. Infect. Dis.* 205: 635–638.
- **196** Bagga, B., Cehelsky, J.E., Vaishnaw, A. et al. (2015). Effect of preexisting serum and mucosal antibody on experimental respiratory syncytial virus (RSV) challenge and infection of adults. *J. Infect. Dis.* 212: 1719–1725.
- **197** Iyengar, L. and Selvaraj, R.J. (1972). Intestinal absorption of immunoglobulins by newborn infants. *Arch. Dis. Child.* 47: 411–414.
- 198 Gopalakrishna, K.P., Macadangdang, B.R., Rogers, M.B. et al. (2019). Maternal IgA protects against the development of necrotizing enterocolitis in preterm infants. *Nat. Med.* 25: 1110–1115.
- **199** Weemaes, C., Klasen, I., Goertz, J. et al. (2003). Development of immunoglobulin A in infancy and childhood. *Scand. J. Immunol.* 58: 642–648.

5

Herpes Simplex Viruses

Alexander Birkmann

AiCuris Anti-Infective Cures AG, Wuppertal, Germany

5.1 Introduction

Like all members of the family of the herpesviridae, herpes simplex virus type 1 (HSV-1) and type 2 (HSV-2) are enveloped, double-stranded DNA viruses [1]. Their genome size is approximately 152 kb and comprises at least 74 open reading frames. The genome consists of two unique segments, the unique long (U_1) and the unique short (U_3) regions, flanked by inverted repeat sequences. Together with the closely related varicella zoster virus (VZV), HSV-1 and -2 represent the human pathogenic members of the alphaherpesvirinae subfamily which share the prominent feature of latency in neurons [2]. HSV is transmitted via infectious viral particles derived from either active lesions, or (especially in the case of HSV-2) from asymptomatic shedding [3, 4]. The primary infection typically occurs during childhood in the case of HSV-1, or later during adolescence in the case of HSV-2, since the latter virus is mainly transmitted sexually [5]. HSV normally enters the body via skin or mucosa and can lead to local lesions at the site of infection, which may or may not be accompanied by clinical signs of disease. Once it has entered the host, the virus spreads to the neurons of the peripheral nervous system, where it establishes a latent and life-long infection (a notable feature of all herpesviruses) [1]. Historically, HSV-1 was associated with infection "above the belt" (i.e. mouth and eye, where the virus persists in the trigeminal ganglion), whereas HSV-2 was attributed to infection "below the belt," specifically genital herpes, by infecting the sacral ganglion [6]. However, nowadays, there is a significant overlap between the sites of HSV infection with an increasing proportion of genital herpes caused by HSV-1, which may be attributed to a decrease in childhood HSV-1 infections and an increase in the practice of oral sex [7, 8]. Once neuronal latency has been established, HSV may be reactivated by an unknown molecular trigger and spread from the ganglion to the lower epidermal layers at or near the site of the initial infection where it replicates. The severity of such a recurrence can range from asymptomatic (only recognized by the ability to detect the virus on the skin) to painful ulcers, such as the typical orofacial cold sores or genital lesions. HSV outbreaks can substantially impact patients'

quality of life; they can be painful and stigmatizing, leading to distress and even psychosocial problems [9, 10]. Ocular HSV-1 infections can result in visual impairment [11]. Especially for genital herpes, it could be shown by quantitative polymerase chain reaction (PCR) analysis that these recurrences can occur frequently as asymptomatic episodes of viral shedding [3]. If occurring frequently, these shedding events can result in a high risk of transmission, even when no symptoms are present. At least 70% of HSV-2 transmissions take place during periods of asymptomatic shedding [12].

Normally, an HSV recurrence is self-limiting as it is rapidly cleared by the immune system, mainly via the cellular immune response [13, 14]. However, HSV reactivation can cause severe and sometimes even life-threatening infections in immunocompromised individuals (especially in those with drug-resistant infection, see Section 5.7) that can last for several months [15]. Furthermore, newborns born to an HSV-positive mother are at risk of acquiring a congenital infection that can manifest in a so-called skin, eye, and mouth infection, herpes simplex encephalitis, or a disseminated disease affecting multiple organs. Notably, even immunocompetent patients can develop herpes simplex encephalitis which mainly affects either patients between 6 months and 20 years of age or patients that are older than 50 years [16]. Especially, these severe infections, but also the stigmatizing labial or genital manifestations, represent a high medical need requiring effective treatment options to control the virus. The overall global prevalence for HSV-1 and HSV-2 infections is estimated to be 67 and 11%, respectively, with the highest rate in the age group 45–49 years (79 and 17%, respectively) [7, 8].

5.2 Overview of the Viral Replication Cycle

Like all herpesviruses, HSV is an enveloped virus, i.e. the virus particle is surrounded by a membranous lipid bilayer. The virus enters its target cell by attaching to cell surface heparan sulfate proteoglycans followed by specific and sequentially aligned interaction of viral glycoproteins with their cellular receptors and then by fusion of the viral and cellular membrane either directly at the plasma membrane or after endosomal uptake [17, 18]. The main viral proteins involved in these processes are the glycoproteins gB (responsible for attachment, together with gC), gD (required for receptor binding), and the heterodimeric complex gH/gL essential for fusion of the virion with the host cell membrane, which is triggered by a cascade of interactions with gB and gD. The cell-cell adhesion molecules nectin-1 and nectin-2, the tumor necrosis factor receptor-related molecule herpesvirus entry mediator (HVEM), and 3-O-sulfonated heparan sulfate have been identified as cellular receptors for HSV. After entering the host cell, the tegument-coated capsid is transported to the nuclear pores and thus into the nucleus where transcription and replication of the viral genome and assembly of progeny capsids take place. The viral particle includes several proteins essential for viral DNA replication, such as the DNA polymerase and the helicase-primase complex, both of which are involved in generation of new genomic DNA. It also includes several nonessential enzymes required for nucleotide biosynthesis and DNA metabolism, such as the viral thymidine kinase (TK) and uracil-DNA glycosylase [19]. After initiation, viral DNA synthesis follows a rolling-circle mechanism [20], producing concatemeric molecules that are later cleaved during nucleocapsid assembly by the terminase, another essential viral protein complex, see also Chapter 6 in this edition. Assembly of the viral capsids and packaging of the viral genome occur in the nucleus and are followed by nucleocapsid egress. In the current model, readily formed capsids bud at the inner nuclear membrane into the perinuclear cleft, thus acquiring an envelope membrane. This primary envelope is then fused with the outer nuclear membrane leading to a release of naked capsids into the cytosol (summarized by [21]). Assembly of the final particle including secondary envelopment and tegumentation occurs by budding into cellular vesicles containing viral glycoproteins on the lumen side and tegument proteins on the cytosolic side. Virions are then released from the cell by fusion of the cellular vesicles with the plasma membrane.

5.3 Treatment of HSV Infections

HSV establishes persistent infections that can recur from time to time. Therefore, treatment can be required for the first infection, called primary (infection of a HSV negative person), for initial/non-primary (if an individual is already infected by one HSV type and acquires the other type), or for a recurrent infection. Since the primary or initial infection can be asymptomatic or display rather nonspecific symptoms [22], most prescribed treatments are typically used for the recurrences. Nonetheless, especially for genital herpes, the first episode may be prolonged and more painful compared to later recurrences [5]. Therefore, if recognized sufficiently early, antiviral treatment should be given to shorten these episodes. In recurrent infections, the majority of viral replication occurs during the first 24 hours of an acute outbreak. Therefore, it is important to treat a recurrent episode as early and effectively as possible ("Hit 'em early, hit 'em hard" strategy) to prevent the virus from further replication and thus to minimize tissue damage [23]. However, if the antiviral medication is not taken early enough, the course of infection can no longer be influenced by direct acting antivirals. Therefore, for labial herpes, one approved treatment option combines the antiviral drug acyclovir (see Section 5.4.3) with hydrocortisone in order to treat both the viral replication and the accompanying inflammation [24].

It is important to note that the reduced immune response in immunocompromised patients means that HSV recurrences can be longer and more severe than recurrences in the immunocompetent, and increases the likelihood of resistance development, as there is no additional defense against the virus besides antiviral treatment [15]. An alternative to the so-called "episodic" treatment described above (i.e. initiating treatment as soon as an active outbreak is recognized) is the use of suppressive treatment (i.e. to treat with an antiviral drug even in the absence of symptoms to prevent reactivation of viral replication and the associated potential recurrence from starting). For immunocompetent patients with genital herpes, especially those with frequent recurrences, suppressive treatment has been shown to be effective [25, 26]. HSV suppression is also a standard clinical practice for patients undergoing transplantation in order to prevent HSV replication under immuno-suppression [27]. Furthermore, it could be shown that suppressive HSV treatment with valacyclovir is accompanied by a reduction of asymptomatic shedding leading to reduced transmission of genital HSV of about 50% [28].

5.4 Approved Anti-HSV Drugs

As described in the following section, apart from 1-docosanol, which targets the cellular uptake of the virus, almost all approved anti-HSV drugs target the viral DNA replication. This applies to previously and currently approved medications as well as to small molecule drugs currently in development. Since the virus comes with its own replicative machinery, this feature can be exploited to enhance selectivity and specificity of the treatment and thereby to reduce the risk of unwanted off-target toxicities, provided that no activity against the human enzymes can be demonstrated. An overview of the targets of antiviral drugs in the HSV replication cycle is shown in Figure 5.1.

5.4.1 5-Substituted 2-Deoxyuridine Analogues

Though not the first antiviral compound ever described, idoxuridine (5-iodo-2-deoxyuridine, Figure 5.2) was the first antiviral drug to gain marketing authorization [29]. Originally described as an antitumor agent for treatment of leukemia, it was later empirically determined that the thymidine analogue is also active against DNA viruses [30]. While it was too toxic for systemic administration, it could be shown that when administered locally at high concentrations, it was able to treat HSV keratitis, i.e. HSV infection of corneal cells [31]. Thereafter, the anti-HSV keratitis activity of trifluridine (trifluorothymidine, ViropticTM, Figure 5.2), another member of the same group, was also demonstrated [32]. However, idoxuridine and trifluridine cannot be considered as specific antiviral agents. Although there may be some specific dependence on the viral TK, both compounds can be



Figure 5.1 Schematic overview of the HSV replication cycle highlighting steps targeted by antiviral drugs.

phosphorylated by viral and cellular kinases to the 5-triphosphate (TP) (i.e., idoxuridine) or 5-monophosphate (i.e., trifluridine) forms, respectively. Both idoxuridine and trifluridine can thus inhibit viral as well as the cellular DNA synthesis, leading to unwanted toxicities (summarized by [33]). Trifluridine is still on the market in the United States for HSV keratitis (https://www.accessdata.fda.gov/scripts/cder/daf/ [02 March 2021]). Interestingly, in combination with the thymidine phosphorylase inhibitor tipiracil, trifluridine is also approved for oral treatment of colon cancer (LonsurfTM, [34]). Idoxuridine is no longer available due to its toxicity. In contrast to idoxuridine and trifluridine, the thymidine analogue brivudine (5-[2-bromovinyl]-2'-deoxyuridine, BVDU, ZostexTM, Figure 5.2) is highly specific in its activity against HSV-1 and VZV (but not HSV-2) [35]. This specificity is derived from the fact that brivudine is specifically phosphorylated by the TK of HSV-1 and VZV, converting it to either the mono- or di-phosphate forms, which are subsequently transformed to the active TP by cellular kinases. Brivudine is approved in several European countries for treatment of VZV infections (shingles) and HSV keratitis.

5.4.2 Vidarabine

The first drug licensed for systemic treatment of herpesvirus infections (HSV and VZV) was the adenosine analogue vidarabine (9-beta-D-arabinofuranosyladenine, Ara-A, Figure 5.2), which is phosphorylated by cellular enzymes to the TP [36, 37]. It has been shown that besides activity against VZV, vidarabine was also effective in herpes simplex encephalitis in a clinical trial [38]. The half-life of the active TP metabolite (ara-ATP) is three times longer in HSV-infected cells compared to uninfected cells, though the mechanism for this selectivity is unknown. However, vidarabine is more toxic and less metabolically stable than many of the other current antivirals such as acyclovir (see Section 5.4.3), and is therefore no longer available for systemic application. Of note, a topical preparation still remains on the market for the treatment of HSV keratitis in some countries.

5.4.3 Acyclic Guanosine Analogues

Brivudine is not the first drug that is specifically activated by the viral TK enzyme. The breakthrough came with the discovery of the acyclic guanosine analogue acyclovir (acyclo-guanosine, ZoviraxTM, Figure 5.2), which became the gold standard for HSV therapy [39, 40]. In HSV- or VZV-infected cells, the viral TK, normally responsible for phosphoryl-ating the pyrimidine base thymidine, surprisingly converts the purine analogue acyclovir to its monophosphate (nicely summarized in [41]). Besides mediating specificity to virus-infected cells, this conversion into a charged molecule has another beneficial effect, as it is trapped inside the cells leading to prolongation of its intracellular half-life. Acyclovir is very well tolerated and for systemic application, it can be used both orally and intravenously. The latter has the advantage that much higher systemic exposure can be reached since acyclovir has a poor oral bioavailability of only approximately 20% [42]. An intravenous formulation of acyclovir for the treatment of severe cases, e.g. immunocompromised patients, herpes simplex encephalitis, and neonatal infection, is available [43, 44].

To overcome the issue of low oral bioavailability, the valine ester prodrug valacyclovir (Valtrex[™]; Figure 5.2) was developed, which exhibits an improved bioavailability of



⁽SADBE, SQX770)

Figure 5.2 Chemical structures of anti-HSV drugs.

approximately 55% [45]. Acyclovir and valacyclovir are approved for the treatment of primary and recurrent labial herpes, as well as for both episodic and suppressive treatment of genital herpes (see prescribing information for ZoviraxTM and ValtrexTM, https://www. accessdata.fda.gov/scripts/cder/daf/ [02 March 2021]). Moreover, based on a large trial with almost 1500 HSV-discordant couples, valacyclovir has also been approved in the United States for reduction of transmission of genital herpes [28]. Besides systemic use, acyclovir is also available in topical formulations for the treatment of cold sores (ointment) and HSV keratitis (eye drops).

Another acyclic guanosine drug with a very similar profile to acyclovir that is also specifically activated by the viral TK is penciclovir (Denavir[™], Figure 5.2). Penciclovir has a somewhat lower antiviral activity against HSV in cell culture compared to acyclovir. This is compensated, however, by the higher intracellular stability of the TP ester of penciclovir, leading to an overall comparable efficacy of both nucleoside analogues [46]. Since the oral bioavailability of penciclovir is even lower than that of acyclovir, it is mainly used as topical treatment for labial herpes. Only the diacetyl 6-deoxy prodrug of penciclovir, famciclovir (Famvir[™], Figure 5.2) is used for systemic treatment [47], being approved for treatment of labial herpes as well as for treatment and suppression of genital herpes (see prescribing information for Denavir[™] and Famvir[™], https://www.accessdata.fda.gov/scripts/cder/ daf/ [02 March 2021]).

5.4.4 Foscarnet

Foscarnet (trisodium phosphonoformate, FoscavirTM, Figure 5.2) was approved for clinical use in the United States in 1991 [48]. Current indications include therapy of cytomegalovirus (CMV) retinitis and acyclovir-resistant HSV infections in immunocompromised individuals (see prescribing information for FoscavirTM, https://www.accessdata.fda.gov/ scripts/cder/daf/ [02 March 2021]). Foscarnet selectively acts at the pyrophosphate binding site on virus-specific DNA polymerases at concentrations that do not affect cellular DNA polymerases [49]. Foscarnet does not require activation by TK or other kinases, and is therefore active against HSV strains carrying mutations in the TK gene that mediate resistance to acyclovir (see Sections 5.6.2 and 5.6.3). However, cross-resistance to acyclovir mediated by mutations within the HSV DNA polymerase itself can still occur [50]. Foscarnet use is limited by poor oral bioavailability of less than 22% necessitating intravenous administration 2-3 times a day [51]. Furthermore, foscarnet can cause severe side effects, mainly nephrotoxicity, mineral and electrolyte abnormalities, as well as seizures (see prescribing information for FoscavirTM, https://www.accessdata.fda.gov/scripts/cder/ daf/ [02 March 2021]). The mechanism behind the toxicities of foscarnet is not fully understood but its ability to form complexes of divalent ions such as Ca^{2+} is considered as potential reason at least for the mineral and electrolyte imbalance [52, 53].

5.4.5 Docosanol

The only approved anti-HSV drug that does not target the viral replication directly is 1-docosanol (Abreva[™]; Figure 5.2). The proposed mechanism of action is inhibition of viral entry by preventing cellular uptake of the virus, i.e. the fusion of the viral envelope

with the cell membrane [54]. Docosanol has been approved by the FDA for topical treatment of labial herpes even though its clinical efficacy remains controversial [55]. Interestingly, in contrast to all other approved anti-HSV drugs, it is available over-the-counter in the United States (see prescribing information for Abreva[™], https://www. accessdata.fda.gov/scripts/cder/daf/ [02 March 2021]).

5.5 Anti-HSV Drugs in Advanced Development or Recently Entering the Market

Except for the helicase–primase inhibitors, the focus of new anti-HSV drugs at advanced stages of development is immunological, using either a direct (monoclonal antibodies) or indirect (therapeutic vaccines and immunomodulators) mechanism of action.

5.5.1 Helicase-Primase Inhibitors

As the name says, this class of compounds target the viral helicase-primase enzyme complex which herpesviruses need to open the DNA strands and keep them separated during DNA replication (Figure 5.1) [56]. As there is no cellular counterpart for this machinery, which herpesviruses share—at least mechanistically—with certain bacteriophages [57], it is a good target for new anti-HSV drugs. Furthermore, as for foscarnet, helicase-primase inhibitors do not require activation by TK or any other viral enzyme. Consequently, helicase-primase inhibitors can be protective to uninfected cells and are active against HSV strains with mutations in the TK or DNA polymerase genes that mediate resistance against acyclovir, penciclovir, or foscarnet [58] (see Section 5.6.3). Two helicase-primase inhibitors are currently in clinical development: amenamevir and pritelivir. Amenamevir (ASP2151, Amenalief™, Figure 5.2), which was originally discovered by Astellas, is an oxadiazolephenyl derivative active against both HSV and VZV [59]. In a Phase 2 study in patients with genital herpes, the efficacy of once-daily amenamevir in shortening the HSV episode duration was comparable to that of valacyclovir [60]. However, Astellas discontinued the development of amenamevir for HSV infections in the United States and the drug was subsequently licensed to Maruho, who conducted a Phase 3 trial in herpes zoster patients and gained approval for this indication in Japan [61]. At the time of this review, three clinical trials with amenamevir were registered in Japan, one Phase 3 trial in patients with an acute episode of genital herpes (JapicCTI-194955), one Phase 3 trial in patients with an acute episode of labial herpes (JapicCTI-194954), and one pilot trial investigating oral amenamevir against epithelial herpetic keratitis resistant to acyclovir ointment therapy (jRCTs061190001).

The second helicase–primase inhibitor is pritelivir (BAY 57-1293, AIC316, Figure 5.2), which was originally discovered at Bayer AG [62] and is currently in clinical development by AiCuris Anti-Infective Cures GmbH. In contrast to amenamevir, the thiazolylamide pritelivir specifically inhibits HSV-1 and HSV-2 and has no activity against VZV. In two Phase 2 trials in otherwise healthy adults with recurrent genital herpes, 28 days oral treatment with pritelivir once-daily demonstrated superior suppression of viral shedding and genital lesions compared to both placebo and oral valacyclovir [63, 64]. A Phase 2 trial for

the treatment of acyclovir-resistant mucocutaneous HSV infections in immunocompromised patients was ongoing in the United States at the time of writing and also included patients that are either acyclovir and foscarnet resistant or acyclovir resistant and foscarnet intolerant (NCT03073967). Furthermore, several different solid formulations are existing for potential use in various other sub-indications [65].

5.5.2 Monoclonal Antibodies

According to the Cortellis Database [Clarivate Analytics, https://www.cortellis.com/ (02 March 2021)], two monoclonal antibodies are currently in development for HSV infections, HDIT101 and UB621 (Figure 5.1). HDIT101 (Herpevizumab, Heidelberg ImmunoTherapeutics) targets an epitope on gB present on both HSV-1 and HSV-2, and is reported to neutralize cell-free HSV particles as well as to prevent cell-to-cell propagation of the virus [66]. At the time of this article two trials with HDIT101 were ongoing in Germany. The first compared one single intravenous dose of HDIT101 vs. valacyclovir in patients with anogenital HSV-2 infection with the primary endpoint "percentage of days with lesion(s)" (NCT04165122), the second compared a topical formulation applied four times over two days vs. placebo in patients with cold sores with the primary endpoint "number of recurrences after 12 months" (NCT04539483). In contrast to HDIT101, UB621 (United BioPharma) targets the viral glycoprotein gD of both HSV-1 and HSV-2 (United BioPharma company web site, http://www.unitedbiopharma.com/ [02 March 2021]). Safety has been demonstrated in a Phase 1 clinical trial. At the time of this review, a Phase 2 trial in patients with genital herpes was ongoing in the United States with the primary endpoint of viral shedding at two different doses of subcutaneously administered UB621 vs. placebo (NCT03595995).

5.5.3 Therapeutic Vaccines

Several attempts have been made to develop a therapeutic vaccine against HSV. One of the most advanced candidates so far was GEN003 by Genocea Biosciences. GEN003 comprises recombinant HSV-2 gD and ICP4 proteins together with Matrix[™] M2 adjuvant [67]. The respective antigens were selected by screening of T-cell responses in asymptomatic HSV-2 positive or HSV-2-exposed but uninfected subjects to identify potential targets involved in controlling and limiting an active infection. In several clinical trials in patients with genital herpes, a moderate reduction of viral shedding vs. placebo could be shown [68–70]. However, for strategic reasons, Genocea discontinued the development of GEN003 in 2017 (Genocea press release of 25 September 2017, https://ir.genocea.com/news-releases/news-release-details/genocea-announces-strategic-shift-immuno-oncology-and [02 March 2021]).

HSV529 is a replication-defective HSV-2 vaccine that has two genes deleted, UL5 and UL29 [71]. The results of a Phase 1 trial in both HSV-negative and HSV-positive participants conducted by the National Institute of Allergy and Infectious Diseases (NIAID) and Sanofi Pasteur have been reported recently [72]. HSV529 elicited neutralizing antibodies and a modest CD4+ T-cell response in HSV-seronegative vaccines. However, no results have been reported yet from a completed trial investigating the effect of HSV529 vaccination on immunological parameters and viral shedding in HSV-2 seropositive adults after valacyclovir treatment (NCT02571166).

5.5.4 Immunomodulators

Besides antibodies and therapeutic vaccines, immunomodulators have also been developed to tackle HSV infections. Notably, the imidazoquinolinamine immunomodulators and TLR-7 antagonists imiquimod (Aldara[™], Figure 5.2), and resiguimod (Figure 5.2) were tested against HSV in the clinic [73]. The impact of topical resiguimod on genital herpes has been investigated in several clinical trials with generally rather moderate outcomes but has shown a certain degree of post-treatment efficacy after intermittent administration [74, 75]. There are many reports in the literature describing imiguimod as a treatment option for acyclovir-resistant HSV infections, though no conclusive data are available from controlled clinical trials [76–79]. Another immunomodulator currently in clinical development for HSV-1 infections is SQX770 (Squarex LLC; Figure 5.2), a topical formulation of squaric acid dibutyl ester (SADBE) for prevention of recurrent cold sores [80]. At the time of this review, two randomized, double-blind, placebo-controlled clinical studies had been conducted in patients with labial herpes (https://biotuesdays.com/2019/12/03/squarexhopes-to-begin-pivotal-testing-to-prevent-recurring-cold-sores-in-2020/ [02 March 2021]). According to company data, it could be demonstrated that SQX770 was effective in extending the time between recurrences and to reduce severity of outbreaks when a single dose is applied to the upper arm (https://www.trialsitenews.com/squarex-announces-positiveresults-from-phase-2-study-of-topical-sqx770-in-prevention-of-recurrent-herpes-labialis/ [02 March 2021]). The proposed mode of action is that SQX770 shifts the immune response toward a stronger type 1 cellular immune response against the virus. Shifts in gene expression profiles were also reported, from those associated with frequent recurrences to those associated with few or no HSV recurrences [81].

5.6 HSV Resistance to Antiviral Drugs

5.6.1 Epidemiology and Manifestation

Resistance to nucleoside analogues is rare in the immunocompetent, but their prevalence is higher in immunocompromised patients [82]. In otherwise healthy individuals with labial and genital herpes, resistance to the commonly used nucleoside analogues is rare with resistant virus found in 0.1–0.7% of the cases. Notably, higher rates of acyclovir resistance have been reported for HSV keratitis after prophylactic use of acyclovir, possibly because of the immune-privileged status of the eye, i.e. a limited inflammatory response [83]. The situation is different in immunocompromised patients such as human stem cell transplant recipients or solid organ transplant recipients and in HIV patients on suboptimal antiretroviral treatment. Here, an average resistance rate to nucleoside analogues of 3.5–14% has been reported with some publications noting rates up to 36% [84]. HSV outbreaks in immunocompromised patients can manifest as atypical lesions which can be enlarged with deeper and more extensive ulceration and which may develop in atypical areas [15]. HSV may spread to other target sites in those patients to cause pneumonitis, esophagitis, hepatitis, retinal necrosis, disseminated infection, and encephalitis [85, 86].

5.6.2 Resistance Mechanisms

Resistance to the widely used nucleoside analogues is primarily mediated by mutations in the UL23 gene coding for the viral TK (ca. 95% of resistant isolates [87]) either preventing successful production of TK (TK-deficient virus) or resulting in alteration of TK substrate specificity (TK-altered virus). The remaining 5% of resistant isolates exhibit alteration of the DNA polymerase activity stemming from mutations within the UL30 gene. As mentioned earlier (see Section 5.4.4), the mutations in the DNA polymerase gene may also result in cross-resistance to foscarnet [50]. Whereas the DNA polymerase gene is an essential enzyme for the virus, TK mutants retain full replication competence in dividing cells with sufficiently high nucleoside triphosphate precursor pools such as epithelial cells; however, TK enzyme function is essential for viral replication in nondividing cells such as neurons, where nucleoside triphosphate precursors for DNA synthesis are limited [88, 89]. As a consequence, it might be assumed that acyclovir-resistance mediated by TK alterations or TK deficiency is always newly induced with every recurrence/treatment cycle. However, reports of the presence of both acyclovir-resistant and acyclovir-sensitive virus in latently infected neurons support the hypothesis that the functional TK of a sensitive virus strain might compensate for the reactivation deficiency of the mutant in trans, leading to successful replication of the resistant virus in epithelial cells after it spreads from the neuron [90, 91]. In which case, acyclovir resistance could derive from both newly induced mutations in the viral genome or from reactivation of already resistant HSV.

5.6.3 Management of Resistant HSV

Depending on which resistance-mediating mutations are present, the virus may remain susceptible to higher doses of nucleoside analogues. Therefore, high doses of intravenous acyclovir are frequently administered as a first step in cases of suspected clinical resistance [92]. In parallel, the resistance profile can be determined by genotypic (mainly by PCR and sequence analysis) or phenotypic resistance testing (plaque reduction assay as gold standard). In case of a confirmed acyclovir resistance that does not respond to high-dose acyclovir, the only approved treatment options in many countries including the United States is intravenous foscarnet (see Section 5.4.4). However, as mentioned above, foscarnet treatment is frequently accompanied by severe toxicities and resistance can occur through cross-resistance (in case of DNA polymerase mutations mediating the resistance to acyclovir as well, see Section 5.6.2) or by acquisition of additional mutations, leading to a situation without further treatment options for the patient. In cases of dual resistance to acyclovir and foscarnet or of acyclovir resistance with foscarnet intolerance, off-label treatments such as imiquimod (see Section 5.5.4) or cidofovir [93] are used as alternatives to foscarnet. However, efficacy of imiquimod is controversial and treatment with cidofovir can also be toxic.

A promising alternative to foscarnet for the treatment of acyclovir-resistant infections could be the helicase-primase inhibitors (see Section 5.5.1). As stated before, due to the different mode of action, these molecules are active against both TK and polymerase mutant HSV, respectively [58]. Resistance to helicase-primase inhibitors, as reported from

nonclinical studies, is mediated through changes in the UL5 (coding for the helicase) and/ or UL52 (coding for the primase) genes [94, 95]. At the time of this review, a clinical trial of the helicase–primase inhibitor pritelivir in immunocompromised patients with acyclovirresistant, mucocutaneous HSV infections was ongoing (NCT03073967).

5.7 HSV and Alzheimer's Disease

Before closing the chapter on HSV, one additional topic which is gaining increasing prominence in the scientific literature should briefly be addressed: the potential association of HSV (especially HSV-1) with Alzheimer's disease (AD) [96, 97]. Though many viruses have been connected with AD, there is an interesting line of evidence that HSV could contribute to or even be the underlying cause of AD: (i) HSV is known to trigger amyloid aggregation and its DNA is commonly found in amyloid plaques. (ii) Anti-HSV drugs reduce $A\beta$ and p-tau accumulation in brains of infected mice. (iii) Clinical studies show cognitive impairment is associated with HSV seropositivity in various patient groups and in healthy adults. At the time of this article, a clinical trial was ongoing to assess the efficacy of 500 mg valacyclovir as a treatment for patients with mild AD (NCT03282916).

5.8 Conclusion

HSV infections are frequently found in the human population. Since it is a self-limiting infection and since treatments are available, it is not considered to be a major health issue in otherwise healthy, immunocompetent individuals. However, for immunocompromised patients such as transplant recipients or HIV patients, HSV infection can be severe and may even become life-threatening. This is especially concerning since resistance to the commonly used nucleoside analogues is more common in this patient population. Foscarnet, the only approved rescue medication in such cases, is active against acyclovir-resistant virus but is toxic, and intolerance and/or dual resistance can occur. Novel treatment options are therefore still urgently needed for HSV infections, potentially also including indications that are not yet so obvious such as AD. Helicase–primase inhibitors, such as pritelivir, may prove very useful against nucleoside analogue-resistant HSV, but also for other indications due to their superior activity compared with polymerase inhibitors [63].

References

- 1 Roizman, B., Knipe, D.M., and Whitley, R.J. (2006). Herpes simplex viruses. In: *Fields Virology*, 5e (eds. D.M. Knipe and P.M. Howley), 2502–2601. Lippincott Williams & Wilkins.
- **2** Davison, A.J. (2007). Overview of classification. In: *Human Herpesviruses: Biology, Therapy, and Immunoprophylaxis* (eds. A. Arvin, G. Campadelli-Fiume, E. Mocarski, et al.). Cambridge: Cambridge University Press.
- **3** Mark, K.E., Wald, A., Magaret, A.S. et al. (2008). Rapidly cleared episodes of herpes simplex virus reactivation in immunocompetent adults. *J. Infect. Dis.* 198: 1141–1149.

- 4 Sacks, S.L., Griffiths, P.D., Corey, L. et al. (2004). HSV-2 transmission. Antivir. Res. 63 (Suppl 1): S27–S35.
- **5** Gupta, R., Warren, T., and Wald, A. (2007). Genital herpes. Lancet 370: 2127–2137.
- **6** Whitley, R. and Baines, J. (2018). Clinical management of herpes simplex virus infections: past, present, and future. *F1000 Res* 7: 1726.
- **7** Looker, K.J., Magaret, A.S., May, M.T. et al. (2015). Global and regional estimates of prevalent and incident Herpes simplex virus type 1 infections in 2012. *PLoS ONE* 10: e0140765.
- **8** Looker, K.J., Magaret, A.S., Turner, K.M. et al. (2015). Global estimates of prevalent and incident herpes simplex virus type 2 infections in 2012. *PLoS ONE* 10: e114989.
- **9** Mindel, A. and Marks, C. (2005). Psychological symptoms associated with genital herpes virus infections: epidemiology and approaches to management. *CNS Drugs* 19: 303–312.
- **10** Ross, K., Johnston, C., and Wald, A. (2011). Herpes simplex virus type 2 serological testing and psychosocial harm: a systematic review. *Sex. Transm. Infect.* 87: 594–600.
- **11** Rowe, A.M., St Leger, A.J., Jeon, S. et al. (2013). Herpes keratitis. *Prog. Retin. Eye Res.* 32: 88–101.
- **12** Mertz, G.J. (2008). Asymptomatic shedding of herpes simplex virus 1 and 2: implications for prevention of transmission. *J. Infect. Dis.* 198: 1098–1100.
- 13 Koelle, D.M., Posavad, C.M., Barnum, G.R. et al. (1998). Clearance of HSV-2 from recurrent genital lesions correlates with infiltration of HSV-specific cytotoxic T lymphocytes. *J. Clin. Invest.* 101: 1500–1508.
- 14 Zhu, J., Koelle, D.M., Cao, J. et al. (2007). Virus-specific CD8+ T cells accumulate near sensory nerve endings in genital skin during subclinical HSV-2 reactivation. *J. Exp. Med.* 204: 595–603.
- 15 Bacon, T.H., Levin, M.J., Leary, J.J. et al. (2003). Herpes simplex virus resistance to acyclovir and penciclovir after two decades of antiviral therapy. *Clin. Microbiol. Rev.* 16: 114–128.
- **16** James, S.H., Kimberlin, D.W., and Whitley, R.J. (2009). Antiviral therapy for herpesvirus central nervous system infections: neonatal herpes simplex virus infection, herpes simplex encephalitis, and congenital cytomegalovirus infection. *Antivir. Res.* 83: 207–213.
- Weed, D.J. and Nicola, A.V. (2017). Herpes simplex virus membrane fusion. *Adv. Anat. Embryol. Cell Biol.* 223: 29–47.
- **18** Agelidis, A.M. and Shukla, D. (2015). Cell entry mechanisms of HSV: what we have learned in recent years. *Futur. Virol.* 10: 1145–1154.
- **19** Weller, S.K. and Coen, D.M. (2012). Herpes simplex viruses: mechanisms of DNA replication. *Cold Spring Harb. Perspect. Biol.* 4: a013011.
- **20** Skaliter, R. and Lehman, I.R. (1994). Rolling circle DNA replication in vitro by a complex of herpes simplex virus type 1-encoded enzymes. *Proc. Natl. Acad. Sci. USA* 91: 10665–10669.
- **21** Mettenleiter, T.C., Muller, F., Granzow, H., and Klupp, B.G. (2013). The way out: what we know and do not know about herpesvirus nuclear egress. *Cell. Microbiol.* 15: 170–178.
- **22** Stoopler, E.T., Kuperstein, A.S., and Sollecito, T.P. (2012). How do I manage a patient with recurrent herpes simplex? *J. Can. Dent. Assoc.* 78: c154.
- **23** Spruance, S., Aoki, F.Y., Tyring, S. et al. (2007). Short-course therapy for recurrent genital herpes and herpes labialis. *J. Fam. Pract.* 56: 30–36.

- **24** Harmenberg, J.G., Awan, A.R., Alenius, S. et al. (2003). ME-609: a treatment for recurrent herpes simplex virus infections. *Antivir. Chem. Chemother.* 14: 205–215.
- Patel, R., Bodsworth, N.J., Woolley, P. et al. (1997). Valaciclovir for the suppression of recurrent genital HSV infection: a placebo controlled study of once daily therapy. International Valaciclovir HSV Study Group. *Genitourin. Med.* 73: 105–109.
- **26** Az-Mitoma, F., Sibbald, R.G., Shafran, S.D. et al. (1998). Oral famciclovir for the suppression of recurrent genital herpes: a randomized controlled trial. Collaborative Famciclovir Genital Herpes Research Group. *JAMA* 280: 887–892.
- **27** Dignani, M.C., Mykietiuk, A., Michelet, M. et al. (2002). Valacyclovir prophylaxis for the prevention of Herpes simplex virus reactivation in recipients of progenitor cells transplantation. *Bone Marrow Transplant.* 29: 263–267.
- **28** Corey, L., Wald, A., Patel, R. et al. (2004). Once-daily valacyclovir to reduce the risk of transmission of genital herpes. *N. Engl. J. Med.* 350: 11–20.
- **29** Kaufman, H.E. (1962). Clinical cure of herpes simplex keratitis by 5-iodo-2'-deoxyuridine. *Proc. Soc. Exp. Biol. Med.* 109: 251–252.
- 30 Bauer, D.J. (1973). Antiviral chemotherapy: the first decade. Br. Med. J. 3: 275–279.
- **31** Calabresi, P. (1963). Current status of clinical investigations with 6-azauridine, 5-iodo-2'- deoxyuridine, and related derivatives. *Cancer Res* 23: 1260–1267.
- Kaufman, H.E. and Heidelberger, C. (1964). Therapeutic antiviral action of 5-trifluoromethyl-2'-deoxyuridine in Herpes simplex keratitis. *Science* 145: 585–586.
- 33 Drach, J.C. (1980). Antiviral agents. In: Annual Reports in Medicinal Chemistry, vol. 15, 149–161. Elsevier.
- 34 Peeters, M., Cervantes, A., Moreno Vera, S., and Taieb, J. (1629-1645). Trifluridine/tipiracil: an emerging strategy for the management of gastrointestinal cancers. *Future Oncol.* 2018: 14.
- **35** De Clercq, E. (2004). Discovery and development of BVDU (brivudin) as a therapeutic for the treatment of herpes zoster. *Biochem. Pharmacol.* 68: 2301–2315.
- 36 Schwartz, P.M., Novack, J., Shipman, C. Jr., and Drach, J.C. (1984). Metabolism of arabinosyladenine in herpes simplex virus-infected and uninfected cells: correlation with inhibition of DNA synthesis and role in antiviral selectivity. *Biochem. Pharmacol.* 33: 2431–2438.
- **37** De, C.E. Looking back in 2009 at the dawning of antiviral therapy now 50 years ago an historical perspective. *Adv. Virus Res.* 2009 (73): 1–53.
- **38** Whitley, R., Soong, S., Dolin, R. et al. (1977). Adenine arabinoside therapy of biopsy-proved herpes simplex encephalitis. *N. Engl. J. Med.* 297: 289–294.
- **39** Elion, G.B., Furman, P.A., Fyfe, J.A. et al. (1977). Selectivity of action of an antiherpetic agent, 9-(2-hydroxyethoxymethyl) guanine. *Proc. Natl. Acad. Sci. USA* 74: 5716–5720.
- **40** Schaeffer, H.J., Beauchamp, L., de Miranda, P. et al. (1978). 9-(2-hydroxyethoxymethyl) guanine activity against viruses of the herpes group. *Nature* 272: 583–585.
- **41** Vere Hodge, R.A. and Field, H.J. (2013). Antiviral agents for herpes simplex virus. *Adv. Pharmacol.* 67: 1–38.
- **42** de Miranda, P. and Blum, M.R. (1983). Pharmacokinetics of acyclovir after intravenous and oral administration. *J. Antimicrob. Chemother.* 12: 29–37.
- **43** Whitley, R. (2004). Neonatal Herpes simplex virus infection. *Curr. Opin. Infect. Dis.* 17: 243–246.

- **44** Whitley, R.J. (2006). Herpes simplex encephalitis: adolescents and adults. *Antivir. Res.* 71: 141–148.
- **45** Beutner, K.R. (1995). Valacyclovir: a review of its antiviral activity, pharmacokinetic properties, and clinical efficacy. *Antivir. Res.* 28: 281–290.
- **46** Hodge, R.A. and Perkins, R.M. (1989). Mode of action of 9-(4-hydroxy-3-hydroxymethylbut-1-yl)guanine (BRL 39123) against herpes simplex virus in MRC-5 cells. *Antimicrob. Agents Chemother.* 33: 223–229.
- 47 Mubareka, S., Leung, V., Aoki, F.Y., and Vinh, D.C. (2010). Famciclovir: a focus on efficacy and safety. *Expert Opin. Drug Saf.* 9: 643–658.
- **48** De, C.E. and Li, G. (2016). Approved antiviral drugs over the past 50 years. *Clin. Microbiol. Rev.* 29: 695–747.
- **49** Crumpacker, C.S. (1992). Mechanism of action of foscarnet against viral polymerases. *Am. J. Med.* 92: 3s–7s.
- **50** Schmit, I. and Boivin, G. (1999). Characterization of the DNA polymerase and thymidine kinase genesof Herpes simplex virus isolates from AIDS patients in whom acyclovirand foscarnet therapy sequentially failed. *J. Infect. Dis.* 180: 487–490.
- 51 Lietman, P.S. (1992). Clinical pharmacology: foscarnet. Am. J. Med. 92: 8s-11s.
- **52** Wallin, A. and Ryrfeldt, A. (1995). The toxicity of Foscarnet (phosphonoformic acid, trisodium salt; PFA) studied in cultured dog renal tubular cells. *Toxicol. Vitro* 9: 237–244.
- **53** Matarese, L.E., Speerhas, R., Seidner, D.L., and Steiger, E. (2000). Foscarnet-induced electrolyte abnormalities in a bone marrow transplant patient receiving parenteral nutrition. *JPEN J. Parenter. Enteral Nutr.* 24: 170–173.
- 54 Katz, D.H., Marcelletti, J.F., Khalil, M.H. et al. (1991). Antiviral activity of 1-docosanol, an inhibitor of lipid-enveloped viruses including Herpes simplex. *Proc. Natl. Acad. Sci. USA* 88: 10825–10829.
- 55 Spruance, S.L. (2002). *N*-Docosanol (Abreva) for herpes labialis: problems and questions. *J. Am. Acad. Dermatol.* 47: 457–458.
- **56** Weller, S.K. and Kuchta, R.D. (2013). The DNA helicase–primase complex as a target for herpes viral infection. *Expert Opin. Ther. Targets* 17: 1119–1132.
- 57 Marintcheva, B. and Weller, S.K. (2001). A tale of two HSV-1 helicases: roles of phage and animal virus helicases in DNA replication and recombination. *Prog. Nucleic Acid Res. Mol. Biol.* 70: 77–118.
- **58** Field, H.J., Huang, M.L., Lay, E.M. et al. (2013). Baseline sensitivity of HSV-1 and HSV-2 clinical isolates and defined acyclovir-resistant strains to the helicase-primase inhibitor pritelivir. *Antivir. Res.* 100: 297–299.
- **59** Chono, K., Katsumata, K., Kontani, T. et al. (2010). ASP2151, a novel helicase-primase inhibitor, possesses antiviral activity against varicella-zoster virus and herpes simplex virus types 1 and 2. *J. Antimicrob. Chemother.* 65: 1733–1741.
- **60** Tyring, S., Wald, A., Zadeikis, N. et al. (2012). ASP2151 for the treatment of genital herpes: a randomized, double-blind, placebo- and valacyclovir-controlled, dose-finding study. *J. Infect. Dis.* 205: 1100–1110.
- **61** Shiraki, K. (2017). Helicase-primase inhibitor amenamevir for herpesvirus infection: towards practical application for treating herpes zoster. *Drugs Today (Barc)* 53: 573–584.
- **62** Kleymann, G., Fischer, R., Betz, U.A. et al. (2002). New helicase-primase inhibitors as drug candidates for the treatment of herpes simplex disease. *Nat. Med.* 8: 392–398.

- 148 5 Herpes Simplex Viruses
 - **63** Wald, A., Timmler, B., Magaret, A. et al. (2016). Effect of Pritelivir compared with valacyclovir on genital HSV-2 shedding in patients with frequent recurrences: a randomized clinical trial. *JAMA* 316: 2495–2503.
 - **64** Wald, A., Corey, L., Timmler, B. et al. (2014). Helicase–primase inhibitor pritelivir for HSV-2 infection. *N. Engl. J. Med.* 370: 201–210.
 - 65 Ruebsamen-Schaeff, H. and Buschmann, H. (1867-1870). Different solid forms for optimizing route of administration of the herpes drug Pritelivir. *Med. Chem. Commun.* 2019: 10.
 - 66 Schaller, T., Seyfizadeh, N., Exner, E. et al. (2018). Herpevizumab, a potent humanized antibody to treat anogenital herpes simplex virus (HSV-1/2) infection–Summary of preclinical data and perspectives of an ongoing clinical trial. *Geburtshilfe Frauenheilkd*. 78: 222.
 - **67** Skoberne, M., Cardin, R., Lee, A. et al. (2013). An adjuvanted herpes simplex virus 2 subunit vaccine elicits a T cell response in mice and is an effective therapeutic vaccine in Guinea pigs. *J. Virol.* 87: 3930–3942.
 - **68** Van Wagoner, N., Fife, K., Leone, P.A. et al. (2018). Effects of different doses of GEN-003, a therapeutic vaccine for genital HSV-2, on viral shedding and lesions: results of a randomized placebo-controlled trial. *J. Infect. Dis.* 218 (12): 1890–1899.
 - **69** Bernstein, D.I., Flechtner, J.B., McNeil, L.K. et al. (2019). Therapeutic HSV-2 vaccine decreases recurrent virus shedding and recurrent genital herpes disease. *Vaccine* 37: 3443–3450.
 - **70** Bernstein, D.I., Wald, A., Warren, T. et al. (2017). Therapeutic vaccine for genital Herpes simplex virus-2 infection: findings from a randomized trial. *J. Infect. Dis.* 215: 856–864.
 - 71 Da Costa, X., Kramer, M.F., Zhu, J. et al. (2000). Construction, phenotypic analysis, and immunogenicity of a UL5/UL29 double deletion mutant of Herpes simplex virus 2. *J. Virol.* 74: 7963–7971.
 - **72** Dropulic, L.K., Oestreich, M.C., Pietz, H.L. et al. (2019). A randomized, double-blinded, placebo-controlled, phase 1 study of a replication-defective Herpes simplex virus (HSV) type 2 vaccine, HSV529, in adults with or without HSV infection. *J. Infect. Dis.* 220: 990–1000.
 - **73** Dockrell, D.H. and Kinghorn, G.R. (2001). Imiquimod and resiquimod as novel immunomodulators. *J. Antimicrob. Chemother.* 48: 751–755.
 - **74** Mark, K.E., Corey, L., Meng, T.C. et al. (2007). Topical resiquimod 0.01% gel decreases herpes simplex virus type 2 genital shedding: a randomized, controlled trial. *J. Infect. Dis.* 195: 1324–1331.
 - **75** Spruance, S.L., Tyring, S.K., Smith, M.H., and Meng, T.C. (2001). Application of a topical immune response modifier, resiquimod gel, to modify the recurrence rate of recurrent genital herpes: a pilot study. *J. Infect. Dis.* 184: 196–200.
 - 76 Perkins, N., Nisbet, M., and Thomas, M. (2011). Topical imiquimod treatment of aciclovirresistant Herpes simplex disease: case series and literature review. *Sex. Transm. Infect.* 87: 292–295.
 - 77 Bernstein, D.I., Spruance, S.L., Arora, S.S. et al. (2005). Evaluation of imiquimod 5% cream to modify the natural history of herpes labialis: a pilot study. *Clin. Infect. Dis.* 41: 808–814.

- **78** Schacker, T.W., Conant, M., Thoming, C. et al. (2002). Imiquimod 5-percent cream does not alter the natural history of recurrent herpes genitalis: a phase II, randomized, double-blind, placebo-controlled study. *Antimicrob. Agents Chemother.* 46: 3243–3248.
- **79** Syed, T.A., Ahmadpour, O.A., Ahmad, S.A., and Shamsi, S. (1998). Treatment of genital herpes in males with imiquimod 1% cream: a randomised, double-blind, placebo-controlled study. *Clin. Drug Investig.* 16: 187–191.
- **80** Palli, M.A., McTavish, H., Kimball, A., and Horn, T.D. (2017). Immunotherapy of recurrent Herpes Labialis with squaric acid. *JAMA Dermatol.* 153: 828–829.
- 81 McTavish, H., Zerebiec, K.W., Zeller, J.C. et al. (2019). Immune characteristics correlating with HSV-1 immune control and effect of squaric acid dibutyl ester on immune characteristics of subjects with frequent herpes labialis episodes. *Immun. Inflamm. Dis.* 7: 22–40.
- **82** Piret, J. and Boivin, G. (2016). Antiviral resistance in herpes simplex virus and varicella-zoster virus infections: diagnosis and management. *Curr. Opin. Infect. Dis.* 29: 654–662.
- **83** van Velzen, M., Missotten, T., van Loenen, F.B. et al. (2013). Acyclovir-resistant herpes simplex virus type 1 in intra-ocular fluid samples of herpetic uveitis patients. *J. Clin. Virol.* 57: 215–221.
- **84** Piret, J. and Boivin, G. (2011). Resistance of herpes simplex viruses to nucleoside analogues: mechanisms, prevalence, and management. *Antimicrob. Agents Chemother.* 55: 459–472.
- **85** Herget, G.W., Riede, U.N., Schmitt-Gräff, A. et al. (2005). Generalized Herpes simplex virus infection in an immunocompromised patient—report of a case and review of the literature. *Pathol. Res. Pract.* 201: 123–129.
- **86** Saylor, D., Thakur, K., and Venkatesan, A. (2015). Acute encephalitis in the immunocompromised individual. *Curr. Opin. Infect. Dis.* 28: 330–336.
- **87** Hill, E.L., Hunter, G.A., and Ellis, M.N. (1991). In vitro and in vivo characterization of herpes simplex virus clinical isolates recovered from patients infected with human immunodeficiency virus. *Antimicrob. Agents Chemother.* 35: 2322–2328.
- **88** Coen, D.M., Kosz-Vnenchak, M., Jacobson, J.G. et al. (1989). Thymidine kinase-negative herpes simplex virus mutants establish latency in mouse trigeminal ganglia but do not reactivate. *Proc. Natl. Acad. Sci.* 86: 4736–4740.
- **89** Tenser, R.B. and Dunstan, M.E. (1979). Herpes simplex virus thymidine kinase expression in infection of the trigeminal ganglion. *Virology* 99: 417–422.
- **90** Wang, K., Mahalingam, G., Hoover, S.E. et al. (2007). Diverse herpes simplex virus type 1 thymidine kinase mutants in individual human neurons and Ganglia. *J. Virol.* 81: 6817–6826.
- **91** van Velzen, M., van Loenen, F.B., Meesters, R.J. et al. (2012). Latent acyclovir-resistant Herpes simplex virus type 1 in trigeminal ganglia of immunocompetent individuals. *J. Infect. Dis.* 205: 1539–1543.
- **92** Piret, J. and Boivin, G. (2014). Antiviral drug resistance in Herpesviruses other than cytomegalovirus. *Rev. Med. Virol.* 24: 186–218.
- **93** Martinez, C. and Luks-Golger, D. (1997). Cidofovir use in acyclovir-resistant herpes infection. *Ann. Pharmacother.* 31: 1519–1521.

- **94** Field, H.J. and Biswas, S. (2011). Antiviral drug resistance and helicase-primase inhibitors of herpes simplex virus. *Drug Resist. Updat.* 14: 45–51.
- **95** Chono, K., Katsumata, K., Kontani, T. et al. (2012). Characterization of virus strains resistant to the Herpes virus helicase-primase inhibitor ASP2151 (Amenamevir). *Biochem. Pharmacol.* 84: 459–467.
- **96** Itzhaki, R.F. (2014). Herpes simplex virus type 1 and Alzheimer's disease: increasing evidence for a major role of the virus. *Front. Aging Neurosci.* 6: 202.
- **97** Itzhaki, R.F. (2017). Herpes simplex virus type 1 and Alzheimer's disease: possible mechanisms and signposts. *FASEB J.* 31: 3216–3226.

Antiviral Strategies Against the Human Cytomegalo Virus Inhibitors of Viral Terminase

Helga Rübsamen-Schaeff

AiCuris Anti-Infective Cures AG, Wuppertal, Germany

6.1 The Need for Novel Drugs against CMV and Attempts of the Past

Human cytomegalo virus (CMV) is a beta herpesvirus. It is widespread in the human population and prevalence ranges from about 50% to close to 100% depending on geographic location and socioeconomic status [1]. It is a large, double-stranded DNA virus harboring in its 236 kb genome at least 165 genes, 4 noncoding RNAs, and 14 miRNAs. Of the protein-coding genes, 44 are replication genes common to all herpes viruses, while about 30 are unique to beta herpes viruses. As an ancient virus, which is thought to have coevolved with its host over millions of years, it replicates systemically and has adapted to various biological niches in the host, with many genes involved in evading detection by the immune response, while others direct cell tropism [2]. Replication and molecular biology of CMV has been reviewed, e.g. in Ref. [3].

While CMV causes unapparent chronic infections lifelong, any condition of a weak or nonexistent immune system can lead to reactivation often resulting in severe disease, which may have a fatal outcome. Patients at risk include recipients of stem cells [4, 5] or solid organs [6], newborns [7], HIV/AIDS patients [8], patients in intensive care [9], or patients treated aggressively against certain autoimmune disorders and with certain cancers [10]. The cost for society, which is associated with untreated CMV infections, can be very high: Untreated congenital CMV infection alone is estimated to cost more than US\$3 billion annually, adjusted for 2015 dollars [11]. This is due to the fact that congenital CMV infection may cause a number of severe disabilities, including mental retardation or hearing loss. There is a very high medical need for well-tolerated, efficacious drugs to treat the conditions, where CMV is an opportunistic pathogen.

As reviewed by Andrei et al. [12], marketed drugs against CMV, which target the viral polymerase, are limited in their use by toxic side effects, pharmacokinetic drawbacks, and resistance development. Moreover, the fact that all low molecular weight drugs licensed before 2017 share a common target (except for Fomivirsen, an antisense oligonucleotide),

6

152 6 Antiviral Strategies Against the Human Cytomegalo Virus Inhibitors of Viral Terminase

also leads to cross-resistance among the polymerase inhibitors. Furthermore, at least the nucleosidic drugs like ganciclovir are prodrugs and require a viral enzyme for their activation within cells [13]. Hence, in uninfected cells, they will only become activated after viral replicative steps have taken place and infection cannot be prevented.

Due to the high medical need in some conditions, the existing CMV-drugs have been used for interventions against CMV outside their indication and in spite of their limitations: for example, Ganciclovir and Valganciclovir, in spite of their toxicities, have been applied to treat congenital CMV infections and are presently being investigated in clinical trials in congenitally infected children [14, 15]. However, although there obviously is a high medical need, no drug has been licensed to date for this indication, nor for any of the other conditions beyond transplantation medicine mentioned above.

Due to the urgent need to develop safer, more effective treatments, ideally also addressing different targets, a number of approaches have been taken and RNAi molecules, monoclonal antibodies, or CMV vaccines have been investigated, apart from low molecular weight drugs: synthetic siRNA against essential gene products of CMV like UL54 or UL97 or UL122/123 was shown to trigger RNAi in infected cells, leading to viral inhibition [16, 17], but clinical development of therapeutics based on this technology has not been reported. A combination of monoclonal antibodies, e.g. CSJ148, consisting of two anti-CMV human monoclonal antibodies (LJP538 and LJP539) that bind to and inhibit the function of viral CMV glycoprotein B (gB) and the pentameric complex (consisting of glycoproteins gH, gL, UL128, UL130, and UL131) have been evaluated in the clinics [18, 19]. While phase I showed a good safety and pharmacokinetic profile, there have been no positive reports on the outcome of phase II in stem cell recipients. Efforts to develop a CMV vaccine began more than 30 years ago, but so far, these efforts failed as well. Most recently, a phase 3 trial by Astellas/Vical was reported. It was designed to evaluate the efficacy of ASP0113 compared with placebo in CMV-seropositive recipients undergoing an allogeneic stem cell transplant. Efficacy was assessed using a primary composite endpoint of overall mortality and CMV end-organ disease through the first year following the transplant. Unfortunately, the endpoint was not met [20].

A number of small molecular weight drugs were also investigated, either by screening for anti-CMV compounds in cell culture or by screening for compounds in target-based biochemical assays.

Out of these efforts, to date, only one new inhibitor of CMV, which targets the viral terminase (BAY 73-6327, AIC246, Letermovir, PrevymisTM, Figure 6.1) has been licensed for prophylaxis against CMV in patients receiving stem cell transplantations [21]. Maribavir, a benzimidazole riboside (Figure 6.1), which was shown to inhibit the CMV protein kinase UL97 (a potent inhibitor of histone phosphorylation catalyzed by wild-type pUL97), with an IC50 of 3–35 nM *in vitro* [22] was developed in the clinics up to phase III. However, in a prophylactic phase III study in patients receiving stem cells, the drug failed [23]. Similarly, great hopes were connected with CMX001 or Brincidofovir, an oral prodrug hexadecyloxypropyl-ester of Cidofovir (Figure 6.1). It was developed up to a prophylactic phase III study in stem cell transplanted patients, but failed in this study as well [24].

In an effort to repurpose drugs, the multi-targeted kinase inhibitor sorafenib has been shown to inhibit CMV in cell culture at concentrations in the one-digit μ M range [25]. Development for this indication, however, has not been reported for sorafenib.

















(e)

(d)

Figure 6.1 Small molecular weight inhibitors of the human cytomegalo virus. (a) Letermovir, BAY 73-6327, AIC246, trade name Prevymis, (2-[(4*S*)-8-fluoro-2-[4-(3-methoxyphenyl)piperazin-1-yl]-3-[2-methoxy-5-(trifluoromethyl)phenyl]-4*H*-quinazolin-4-yl]acetic acid). (b) Maribavir, Benzimidavir; Bzurea; Camvia; 1263W94; BW1263W94; GW257406X). (c) CMX001, Brincidofovir, prodrug of cidofovir. (d) Cidofovir, trade name Vistide, CAS Registry Number: 113852-37-2. (e) Hydroxypyridonecarboxylic acid, compound 7r from Ref. [26].

Recently, in a new approach, hydroxypyridonecarboxylic acids [26] (Figure 6.1e) have been investigated as inhibitors of CMV pUL89 endonuclease, which provides the enzymatic functions for the CMV terminase complex in viral packaging. In addition, five new hits from a cellular screen were described by Kapoor et al. [27].

154 6 Antiviral Strategies Against the Human Cytomegalo Virus Inhibitors of Viral Terminase

The difficulty to protect patients with a very weak or missing immune system against CMV reactivation is highlighted by the failure of the multiple approaches discussed above. For Maribavir [28, 29], some clinical studies are ongoing investigating a different application, namely treatment of resistant/refractory CMV infections. Brincidofovir was licensed to SymBio and may be developed further for HHV-6 in transplanted patients. In spite of all efforts, Letermovir is the only drug that could be licensed for the prophylaxis against CMV in stem cell recipients to date. Therefore, the strategy which resulted in its discovery will be discussed.

6.2 The Strategy for the Discovery of Letermovir

When researching for antiviral compounds, approaches can be to screen compounds in virus-infected cell cultures or screening compounds in target-based biochemical assays and by using molecular modeling. Target-based biochemical assays have been successful in the past for small viruses like HIV or HCV and for single enzymes like proteases. However, biochemical target-based screens bear the inherent problem that there is no guaranty that a highly active inhibitor in the biochemical assay will also pass the cell membrane and be active inside an infected cell. To date, there is no rule, which would allow medicinal chemists to make the inhibitor enter the cell by adequate rational chemical modifications. In addition, CMV, which is one of the largest viruses [2, 3], contains many more potential targets than, e.g. HIV or HCV and there is an uncertainty how to choose the "right" targets for target-based screens. Even for a small virus like HCV, inhibitors against the HCV NS5A nonstructural protein, which plays several important roles in the virus life cycle (but has no known enzymatic activity), were identified by phenotypic screening of large compound libraries and resistance mapping of early leads and might not have been discovered otherwise (see Chapter 2). Generally, for a target with many complex interactions or for multienzyme complexes, biochemical target-screens are difficult to establish and such screens may not reflect the true composition and conformation of the proteins or protein-nucleic acid complexes in a living cell. Such screens may thus yield hits, which cannot be optimized for in vivo antiviral activity and will not be able to deliver the relevant activity in a living cell.

In theory, screening chemical molecules in an infected cell will allow to probe inhibition of all targets simultaneously, which are responsible for the replication of the virus, including multi-protein complexes or complexes of proteins with nucleic acid. This approach may also reveal targets, which are particularly important for the kinetics of viral replication.

Based on all these considerations, for CMV, cellular screens were performed apart from target-based screens. Interestingly, apart from several CMV core-inhibitors with distinct chemical structures (unpublished), which were discovered in cellular screens, the viral terminase complex (Figure 6.2a) was hit three times in such screens, with compounds from different chemical classes, suggesting its essential function for the replication kinetics of the virus. The terminase inhibitors inhibited neither viral DNA synthesis nor viral transcription or translation (Figure 6.2a), but particles lacking a DNA core accumulated within infected cells together with unprocessed high molecular weight DNA. Inhibition of functional cleavage at viral intergenomic transitions in inhibitor-treated cells pointed to interference with the viral DNA cleavage machinery.

The first terminase-inhibitor to be discovered was BAY38-4766 [30–32] (Figure 6.2b). In addition to inhibiting the human virus, it had strong antiviral activity *in vitro* and *in vivo* against the murine cytomegalovirus and allowed to prove its activity in a lethal challenge model in mice [30, 31]. Analysis of mutants generated *in vitro* revealed resistance mutations to BAY38-4766 which mapped to the two proteins, UL89 and UL56 [32]. UL89 and UL56 function as two subunits of the viral terminase [32]. Since human DNA does not undergo similar maturation steps and hence interaction with human enzymes was not to be expected, a good tolerability in humans was expected from targeting these structures.



(a)



(b)



(c)



(d)

(Continued)

156 6 Antiviral Strategies Against the Human Cytomegalo Virus Inhibitors of Viral Terminase





Figure 6.2 Mode of action of the terminase and terminase inhibitors of CMV. (a) The mode of action of the CMV terminase. (b) The first terminase inhibitor, drug candidate Bay 38-4766, tomeglovir. (c) The second terminase inhibitor class (phenylsulfonamide aminopyridines) drug candidate: Bay 64-8030. (d) Letermovir, representing a third class (dihydroquinazoline derivatives) BAY 73-6327, AIC246, Prevymis. (e) Letermovir: Structure–activity relationship.

After demonstrating high efficacy in the murine lethal challenge model [31] and good tolerability in toxicity studies, BAY 38-4766 (tomeglovir) underwent phase I testing in humans. As expected, a good tolerability was demonstrated in man. Unfortunately, however, BAY 38-4766 was found to induce CYP enzymes leading to its own degradation. A prodrug approach was not able to improve this situation and the class was abandoned.

A second class of terminase-inhibitors to be discovered were the phenylsulfonamide aminopyridines (Figure 6.2c), with the drug candidate BAY 64-8030. Again, phase I testing in humans demonstrated good tolerability at all doses tested, but BAY 64-8030 had an unexpected high metabolic cleavage, compared with the metabolism seen in the preceding animal studies. Development was thus suspended as well and the compound was modified chemically to stabilize the position of metabolic attack.

Finally, letermovir, (BAY 73-6327, AIC 246) a dihydroquinazoline-derivative, was discovered (Figure 6.2d). It showed cross-resistance to viruses resistant to previous terminase inhibitors and hence was assumed to be a terminase-inhibitor as well. This was confirmed in dedicated studies of *in vitro* resistance-induction and marker transfer [33]. Like the other anti-terminase molecules, BAY 73-6327 was very well tolerated in animals. Figure 6.2e shows some aspects of the structure-activity relationship of letermovir.

6.3 The Link between Preclinical Models and Clinical Efficacy

As described before, three classes of terminase inhibitors were studied in phase I tests in humans and two failed due to the fact that at that time, models for predicting human pharmacokinetics by *in vitro* tests were just evolving. However, all terminase development
candidates confirmed the predictions from theoretical considerations with respect to tolerability: all three classes of terminase inhibitors were very well tolerated not only in animals, but also in humans, demonstrating that targeting an enzymatic complex which does not exist in humans is likely to confer an advantage with respect to tolerability. In addition, due to the fact that some of the marketed drugs are prodrugs (and hence dependent on cellular and viral kinase activities), it was assumed that a compound, which does not need such activation, will have a good efficacy across different cell types and across different multiplicities of infection (MOI). Table 6.1 [34, 35] shows for letermovir that the EC₅₀ is in the nanomolar range (EC₅₀ 4–5 nM) with a very steep dose–response curve: EC₉₀ is increased only slightly compared with EC₅₀ (EC₉₀ 5–7 nM). In contrast, EC₅₀ of ganciclovir is in the micromolar range and the compound has a much flatter dose–response curve (EC₅₀ 3–4 μ M, EC₉₀ 11–18 μ M).

Furthermore, upon increase of the multiplicity of infection (Table 6.2) [34, 35], the EC_{50} of letermovir increases by a factor of about 3 that of ganciclovir by a factor of 5 and maribavir was found to be inactive at higher viral inputs into the cell culture.

As letermovir – in contrast to BAY 38-4766 – is specific for CMV only, *in vivo* activity was assessed in an engineered mouse xenograft model, in which mice were transplanted with a gelfoam sponge carrying CMV-infected human cells. The animals were treated once daily via oral gavage (Figure 6.3) [35, 36]. Treatment lead to a dose-dependent reduction of CMV titer in the cells within the sponge transplanted to the mice, compared with a placebotreated group. Statistical analysis revealed significant antiviral effects for the 10, 30, and 100 mg/kg/day treatment by letermovir as well as for the 100 mg/kg/day valganciclovir group (p < 0.006), Table 6.3.

		EC ₅₀ (μM)ª		EC ₉₀ (μM)ª		No. of independent expts
Assay	CMV strain	Letermovir	Ganciclovir	Letermovir	Ganciclovir	
CPE-RA ^b	Davis	0.0040 ± 0.0010	2.70 ± 0.70	ND ^c	ND	7
	AD169	0.0050 ± 0.0010	4.30 ± 1.80	ND	ND	13
GFP_RA ^d	AD169-GFP	0.0038 ± 0.0009	1.73 ± 0.93	0.0051 ± 0.0014	10.7 ± 2.5	18
	RV-HG	0.0049 ± 0.0009	2.33 ± 2.75	0.0071 ± 0.0025	18.3 ± 22.7	5

Table 6.1Sensitivities of different CMV laboratory strains to Letermovir and Ganciclovirin fibroblast cells.

 a EC₅₀ and EC₉₀ values were determined by the indicated antiviral assay. Nonlinear regression analysis was performed, and the resulting graphs were used to calculate the respective values. Results are expressed as means \pm standard deviations.

^b CPE reduction assay.

^c ND, not determined.

^d Fluorescence reduction assay.

Multiplicity of infection (MOI)	EC ₅₀ (μM) Letermovir	EC ₅₀ (μM) Ganciclovir	EC ₅₀ (μM) Maribavir
0.003	0.0013	0.99	0.29
0.01	0.0015	0.68	0.17
0.03	0.0029	1.74	0.38
0.1	0.0034	2.21	0.94
0.3	0.0036	6.51	No activity
1	0.0042	5.26	No activity

Table 6.2 Comparison of the effect of increasing CMV viral load on EC ₅	EC ₅₀ .
---	--------------------

Source: From Refs. [34, 35], CMV strain AD169-GFP, EC_{50} values determined by fluorescence reduction of a GFP-labeled virus, means from 2 to 8 independent experiments.



Figure 6.3 In vivo antiviral activity of letermovir in a mouse xenotransplant model [35, 36]. The effect of the therapy by valganciclovir (VGCV) or letermovir (AIC246) on CMV replication in a mouse xenograft model: viral titers in CMV-infected gel-sponges harvested from transplanted mice receiving antiviral treatment as indicated or placebo [35, 36]. Drugs were given once daily per os for nine days. Results are expressed as means ± standard errors of the means. Letermovir showed superior *in vivo* activity in the mouse xenograft gelfoam model. In addition, as in cell culture, *in vivo*, a remarkably steep dose–response with $ED_{50} = 3 \text{ mg/kg/day}$ and $ED_{90} = 8 \text{ mg/kg/day}$ day was observed (Table 6.3) [35, 36].

Table 6.3	In vivo antiviral activit	y of letermovir in a mouse	xenotransplant model	[35, 36]
lable 0.5	III VIVO antivitat activit		Acholiansplant mouce	, , , , , , , , , , , , , , , , , , , ,

	EC ₅₀	EC ₉₀ [mg/kg/d]
Letermovir	3	8
VGCV	16	>100

 ED_{50} and ED_{90} values. Since ED_{90} was not reached for valganciclovir at 100 mg/kg/day, it was set to >100 mg/kg/day.

6.4 Clinical Experience

After letermovir had shown efficacy *in vitro* and in the animal model and was well tolerated in toxicological studies [37], a number of phase I studies confirmed the good tolerability and revealed a pharmacokinetic profile suggesting that a once daily application as well as a p.o. and an i.-v. formulation was possible [38–42].

The first phase II trial was conducted in kidney-transplanted patients, who had reactivated CMV [43] and who were treated orally for 14 days with letermovir either 40 mg b.i.d., 80 mg o.d., or with local standard of care. The primary endpoint was the decrease of CMV DNA copy number from baseline. This endpoint was achieved in all groups with statistical significance, proving for the first time in humans that letermovir was active against CMV. Furthermore, letermovir was well tolerated in the kidney-transplanted patients. Interestingly, as this had not been excluded by the protocol, some of the patients had a CMV infection with a proven resistance genotype against polymerase inhibitors and achieved suppression of their virus upon letermovir treatment as well [43].

While the kidney-transplanted recipients in the above study had reactivated CMV, but had not developed CMV disease yet, a single patient in whom disease due to a multiresistant virus had developed in several organs was treated under an emergency IND obtained from the FDA [44] (Figures 6.4 and 6.5). After application of letermovir (AIC246), the viral load became undetectable and all organs healed [44] (Figures 6.4 and 6.5), including the retinitis. This demonstrated for the first time that CMV disease can be treated efficiently by letermovir as well and that the drug is active against polymerase-resistant virus causing clinical disease.

As CMV-positive patients receiving stem cell transplantations have a very high medical need for suppression of CMV reactivation immediately after transplantation, a phase 2 dose-finding study was conducted to investigate the effect of letermovir on the incidence and time to onset of prophylaxis failure in CMV-seropositive patients, who had received allogeneic hematopoetic-cell transplants from matched related or unrelated donors. Patients were treated with oral letermovir at doses of 60, 120, or 240 mg/day, respectively, for 12 weeks after engraftment, or with matching placebo. The primary endpoint was all-cause prophylaxis failure, defined as discontinuation of the study drug because of CMV antigen or DNA detection, end-organ disease, or any other cause. In this study, the reduction of the incidence of all-cause prophylaxis failure was clearly dose-dependent [45] (Figure 6.6), matching the dose-dependent efficacy, which had been seen before *in vitro* and in animals. Importantly, the safety profile of letermovir was similar to placebo (Table 6.4) [45], with no indication of hematologic toxicity or nephrotoxicity. The ECGs showed no significant findings in the central laboratory readings, the vital parameters showed no statistically significant cases of predefined abnormalities between the letermovir and the placebo groups [45].

These results let to a phase III trial [46] (sponsored by MSD, Figure 6.7). In this phase 3, double-blind trial, CMV-seropositive transplant recipients, 18 years of age or older, were assigned in a 2:1 ratio to receive letermovir or placebo, administered orally or intravenously, through week 14 after transplantation. Letermovir was administered at a dose of 480 mg/day (or 240 mg/day in patients taking cyclosporine). Patients in whom clinically significant CMV infection (CMV disease or CMV viremia leading to preemptive treatment) developed, discontinued the trial regimen and received anti-CMV treatment. The primary end point was the proportion of patients, who had clinically significant CMV infection



Figure 6.4 Successful treatment of a lung-transplanted patient with resistance against polymerase inhibitors and CMV-pneumonitis, retinitis, and colitis [44]. Antiviral treatment (top) and viral load and CMV disease (bottom). *Source*: Kaul et al. [44]. With permission of John Wiley & Sons.



Figure 6.5 Organ disease before and after treatment with letermovir. Chest X-ray before (a) and after (b) treatment with letermovir, colonoscopy before (c) and after (d) treatment with letermovir. *Source*: Kaul et al. [44]. With permission of John Wiley & Sons.



Figure 6.6 CMV prophylaxis by letermovir in patients with hematopoietic stem cell transplantation, NCT01063829: dose-dependent protection from CMV reactivation. *Source*: Based on [45].

<i>N</i> (%) Patients with at least one predefined change post baseline	Letermovir 60 mg qd <i>N</i> = 33	Letermovir 120 mg qd <i>N</i> = 31	Letermovir 240 mg qd <i>N</i> = 34	Placebo N = 33
Hemoglobin: $PC = -2g/dl$	10 (30)	11 (35.5)	8 (23.5)	7 (21)
WBC: $PC = -2000/mm^3$	11 (33)	13 (42)	10 (29)	9 (27)
Eosinophils: $PC = +20\%$	2(6)	1(3)	1 (3)	2(6)
Neutrophils: $PC = -20\%$	9 (27)	9 (29)	6 (18)	10 (30)
Platelets: $PC = -100000/mm^3$	4 (12)	0	6 (18)	4 (12)

Table 6.4CMV prophylaxis by letermovir in patients with hematopoietic stem celltransplantation: clinical laboratory. Source: Data from [45].

through week 24 after transplantation. Patients who discontinued the trial or had missing end-point data at week 24 were imputed as having a primary endpoint event. Patients were followed through week 48 after transplantation.

A total of 565 patients underwent randomization and received letermovir or placebo beginning a median of nine days after transplantation. Significantly fewer patients in the letermovir group than in the placebo group had clinically significant CMV infection or were imputed as having a primary end-point event by week 24 after transplantation, P < 0.001 (Figure 6.8). The frequency and severity of adverse events were similar in the two groups overall. The rates of myelotoxic and nephrotoxic events were similar in the letermovir group and the placebo group. In conclusion, letermovir prophylaxis resulted in a significantly lower risk of clinically significant CMV infection than placebo.



Figure 6.7 Prophylaxis by letermovir against CMV reactivation in CMV-seropositive recipients of hematopoietic stem cells: Study design. *Source*: Based on [46].



Figure 6.8 MSD phase 3 study: time to clinically significant HCMV infection through week 24 post-transplant (FAS). *Source*: Based on [46].



Figure 6.9 MSD phase 3 study: all-cause mortality through week 24 post-transplant (FAS, w 24 data including vital status collected post-study). *Source*: Based on [46].

It is noteworthy, that in-depth analysis of mortality in this study revealed a significant advantage for letermovir-treated patients at week 24, Figure 6.9. All cause mortality at week 24 after transplantation was significantly lower among letermovir recipients than among recipients of placebo. This is in line with earlier observations from preemptive



Figure 6.10 Letermovir phase 3 study (MSD): time to engraftment through week 24 post-HSCT, NCT02137772.

therapy regimens that permitting early CMV viremia – even at low copy numbers – is associated with an increased risk of overall mortality within the first year of HSCT and with a positive dose–response relationship regarding the virus, even in patients, in whom preemptive therapy has been appropriately initiated [5, 47].

Furthermore, in contrast to the toxicities seen with polymerase inhibitors, there was no negative effect on engraftment (Figure 6.10) and the hematological lab parameters were similar between letermovir and placebo [46].

This study led to the market authorization for letermovir in the United States in 2017 [21] and later in other jurisdictions as well. It is the first phase III study successfully demonstrating functional prophylaxis against reactivation of the virus for CMV carriers, who undergo stem cell transplantations.

As discussed above, all other attempts to generate prophylactic regimens in this indication have failed to date. Apart from good tolerability, letermovir has a clear dose–response relationship in *in vitro* studies, in animals and in humans and this dose–response relationship is steep (Tables 6.1–3, Figures 6.3, 6.6). Therefore, apart from carefully choosing a target not existing in humans, a clear dose–response relationship and high efficacy against increasing viral loads in cell culture as well as in animal studies appear key parameters to be optimized in the early selection of potential drug candidates.

6.4.1 Resistance Mutations and Resistance Development

The mode of action of letermovir, i.e. inhibition of CMV terminase, was proven by crossresistance to previous terminase inhibitors, *in vitro* resistance selection, and marker-transfer studies [33]. Therefore, and due to additional studies [48, 49], the loci of terminase mutations, which are responsible for resistance against letermovir, are well known.

The terminase consists of the UL56 and UL89 proteins, associated with UL51 in the functional packaging holocomplex [32, 50, 51]. The UL56 subunit of the terminase complex is believed to play a role in DNA packaging through sequence-specific binding of DNA packaging motifs in CMV genome concatemers [32, 51].



Figure 6.11 Resistance loci in UL 56 (Source: Courtesy of J. Strizki, MSD).

DNA sequencing of viruses generated in cell culture revealed resistance mutations primarily in UL56. Apart from CMV mutations associated with letermovir resistance that map to amino acids in UL56, a few mutations mapping to UL89 and UL51 were also detected. Most mutations occur in UL56 at 19 sites, two sites are mutated in UL89 and one site in UL51 [33, 52, 53]. Almost all letermovir resistance-associated variants (RAVs) map to the UL56 gene "hot spot", i.e. in the region encoding amino acids (AA) 229–369 [33, 49] (Figure 6.11). 18 of 19 UL56 mutation sites are clustered between AA 229-369 of UL56 [33, 53], the 19th mutation maps to AA 25 [49]. The UL56 mutations result in a loss of letermovir susceptibility ranging from minimal (1.8-fold) to highly resistant (>9000 fold), while the UL89 and UL51 mutations confer very modest resistance (1.6- to 5.4-fold). These letermovir mutations do not have a substantial impact on CMV fitness. Of note, common UL56 polymorphisms in wildtype virus do not impact susceptibility to letermovir nor does the subtype of glycoprotein B [54].

6.4.2 Resistance Observed in Clinical Studies

The resistance pattern emerging in clinical use may differ from the laboratory situation. Furthermore, the frequency and nature of emerging resistance determines on the long run how useful a drug proves to be in the clinics and whether it needs to be combined with other antiviral agents. Obviously, the lack of cross-resistance to CMV DNA polymerase inhibitors [33, 43, 44, 53] is a valuable feature of letermovir in this context.

Of 98 subjects receiving LET in the Phase 2b dose-finding prophylaxis trial [45], only one, who received a suboptimal 60 mg daily dose, had a mutation (pUL56 substitution V236M) that confers reduced susceptibility to LET [55].

The analysis of mutations occurring in the Phase 3 trial of CMV prophylaxis for hematopoietic stem cell transplant recipients [46] was performed using amplicon-based nextgeneration sequencing of UL86 and UL 56 in patients, who received 14 weeks letermovir prophylaxis or placebo and in whom clinically significant CMV infection developed through week 24 after HSCT [56]. A low incidence of resistance was detected. The RAVs that were detected, mapped to the CMV UL56 gene at positions associated with reduced susceptibility to letermovir based on resistance selection in cell culture (Figure 6.12).

Out of 373 subjects who received letermovir prophylaxis, resistance-associated mutations were identified in 3 subjects. RAVs encoding V236M and C325W were detected



Figure 6.12 pUL56 variants observed in subjects with clinically significant CMV infection receiving prophylactic letermovir in Phase 3 Study NCT02137772 [56] (Source: Courtesy of J. Strizki, MSD).

independently in subjects 1 and 3, who experienced a clinically significant CMV infection while receiving letermovir prophylaxis. Two other variants, E237G and R369T, were detected >3 weeks after subjects 2 and 3, respectively, had discontinued letermovir prophylaxis and received preemptive therapy with ganciclovir. While V236M was seen in a patient with early noncompliance, the patient, in whom C325W was detected was viremic on day 1. The E237G variant was a minority variant (4% frequency in NGS reads) and its relevance is unclear. The mutation E237G and R369T conferred reduced susceptibility to letermovir with EC50 changes of 13- and 52-fold, respectively. Concerning reactivation of CMV during prophylaxis extended duration letermovir beyond day 100 has recently been shown to be efficacious in preventing clincally significant CMV in patients with GVHD [57].

While the above studies systematically analyzed the occurrence of letermovir resistance mutants during prophylaxis, only small patient numbers have been analyzed for resistance-development in cases, where the virus had reactivated and treatment by letermovir was initiated to achieve viral suppression.

The first trial involving 27 viremic recipients of a kidney transplant [43] met its primary endpoint with respect to viral suppression, but due to the short treatment period of 14days, an analysis of letermovir resistance mutations was not performed. During treatment of a lungtransplanted patient who received a 49 day course with 120-240 mg letermovir daily and was suffering from CMV viremia and CMV disease in several organs due to polymerase-resistant virus [44], letermovir successfully suppressed the virus with healing of the organs, demonstrating efficacy against polymerase-resistant virus. In this case, development of resistance to letermovir was not detected. On the other hand, in another lung-transplanted patient, who was treated with 480 mg letermovir after developing ganciclovir resistance, there was only a transient reduction of viral load below detection and analysis of the rebounded virus revealed the C325Y mutation [58]. A case report of development of the C325Y mutation after treatment of CMV viremia was also described in an HSCT patient. Again, letermovir first suppressed CMV and upon relapse, the C325Y mutation was detected [59]. Four cases of letermovir use were described in patients, who received solid organ transplantations and had developed ganciclovirresistant CMV retinitis [60]. All patients showed clinical and fundoscopical improvements with resolution of retinitis, but three patients failed to achieve sustained virologic suppression, with

one exhibiting low-grade, intermittent DNAemia, and the other two patients with high-grade DNAemia after more than one month of therapy with letermovir. While the patient with the low-grade viremia did not reveal any resistance mutation at the UL 56 site, which was sequenced, the two patients with high viremia had mutations in the 325 locus: C325F and C325Y, respectively. For one of them, a reuse of valganciclovir and subsequent virologic suppression was possible, since the virus had reverted to UL-54 wildtype. The other patient could be treated again with foscarnet and achieved virologic suppression as well. None of the patients experienced any recurrent retinitis or vision loss while on letermovir for ongoing suppression.

A first pediatric case worth mentioning here was a 14 year old girl, who was transplanted due to sickle cell disease [61]. In spite of prophylaxis against CMV and HHV6 with foscarnet and later with ganciclovir, her clinical condition worsened and on day 90, a ganciclovir-resistance mutation was found. She was transitioned back to iv foscarnet, but her clinical condition including enteric involvement remained severe and viremia rose. Iv letermovir (480 mg) was added to iv foscarnet for dual salvage therapy through day 188. During this time, her CMV levels declined to undetectable and her clinical condition improved. Iv foscarnet was stopped and the patient was managed on oral letermovir (480 mg) until discharge from the hospital. After discharge, the patient first did very well. Later, the clinical condition worsened and on day 257, a UL 56 mutation R369S was discovered. This mutation is at a locus that was seen in the prophylactic phase 3 study (R369T, see above). The authors discuss, whether while switching from iv letermovir to oral drug, the enteric absorption could have been reduced by her severe intestinal GvHD.

To date, letermovir is only licensed for the prophylaxis against CMV in stem cell recipients. Unfortunately, except for Ref. [43], there is no clinical study published for the therapy of viremia or CMV disease. While the forgoing case reports are a selection from the literature, they demonstrate clinical utility in a number of cases, but also the possibility of emergence of letermovir resistance. These case reports, however, do not allow to draw general conclusions on the frequency of resistance upon treatment of CMV reactivation, since the number of treated and nonresistant patients is not known as it is not being reported. In addition, various risk factors of the transplanted patients themselves may modify their ability to control CMV. In the absence of guidance from clinical studies, it appears very important to ensure compliance with the dose chosen and to monitor the patients for exposure to some immunosuppressants is influenced in the presence of letermovir and needs to be monitored carefully [62] in these patients as well.

6.5 Other Potential Indications for Letermovir

As discussed above, CMV is a very widespread virus and may cause severe infections in all conditions with a weak or missing immune system. Patients at risk include recipients of stem cells, for which the drug is presently licensed. However, recipients of solid organs may also experience CMV reactivation [6] and a study investigating letermovir-prophylaxis in kidney recipients is ongoing (NCT03443869).

Table 6.5 Summarizes the conditions, where CMV may be a significant pathogen or copathogen beyond the transplantation field. Outside the transplantation field, newborns, who are congenitally infected by CMV, are at particular risk to develop life-long **168** 6 Antiviral Strategies Against the Human Cytomegalo Virus Inhibitors of Viral Terminase

Potential indication	Reason for treatment	References
Congenital CMV infection	Congenital CMV infection represents the most common congenital viral infection, estimated incidence in developed countries: 0.6–0.7% of all live births, i.e. approximately 60 000 neonates born every year with congenital CMV infection in the United States and the European Union combined. Children infected early in pregnancy have highest risk to develop severe neurological impairment and may die	[7, 11, 63–65]
Active CMV replication in HIV patients on HAART treatment	CMV (and/or other herpesvirus) replication is a significant cause of immune activation in HIV-infected individuals with incomplete antiretroviral therapy-mediated CD4 ⁺ T cell recovery. In the AIDS stage, CMV can lead to blindness, severe colitis, and death	[8, 66]
Active CMV replication in patients in	Active CMV infection is associated with a significantly higher mortality rate compared with critically ill patients without active CMV infection;	[9, 67]
intensive care	There is an independant correlation between CMV reactivation and increased morbidity in non-immunosuppressed patients with severe sepsis	[68]
CMV and glioblastoma	From [69]: "Classical and novel antiviral therapies against CMV should be revisited as they may represent a great promise for halting tumor progression and lower cancer deaths"	[10, 69]
Detection of CMV in other autoimmune diseases	Predictors of mortality with concurrent cytomegalovirus detection	[70]
CMV and inflammatory bowel disease	CMV colitis may affect up to one-third of patients with acute severe ulcerative colitis (ASUC) refractory to corticosteroid therapy (Refs. 165, 166 in Ref. [71]). Higher rate of treatment refractoriness and need for colectomy in patients with	[71]
	demonstrable CMV colitis. Identification of this disease should prompt treatment with antiviral therapy in the setting of refractoriness to steroids or biologic therapy	Refs. 74–78 in Ref. [72]

 Table 6.5
 Potential indications for treating CMV infection outside transplantation medicine.

disabilities like mental retardation or hearing loss or even to die (Table 6.5, refs. [7, 11, 63, 64, 65]). As mentioned above, apart from the significant disease burden in these children, there is a very high financial burden for society due to congenital infections with estimates of the overall economic burden exceeding US\$3 billion annually [11]. A study using ganciclovir - in spite of its tolerability issues - in neonates with symptomatic disease has demonstrated that start of therapy in the neonatal period prevented hearing deterioration at six months [63]. However, two-thirds of treated infants had significant neutropenia during therapy. A small study, also using ganciclovir concluded [64]: "Asymptomatic congenital cytomegalovirus infection is likely to be a leading cause of sensorineural hearing loss in young children. Intravenous ganciclovir therapy seems to offer a medical option to prevent subsequent sensorineural hearing loss. Further studies including a greater number of children are needed. Cytomegalovirus screening models are mandatory, if medical

therapy is to be implemented in time" [64]. As screening systems are available and letermovir offers high antiviral potency together with good tolerability, time seems to be ripe for clinical trials testing the use of letermovir in CMV-infected symptomatic and asymptomatic newborns.

There are a number of other conditions where CVM is being suspected to be a pathogen with its own negative impact on the patients' health. One example is HAART-treated HIV infection [8, 66], another patient group are individuals in intensive care with active CMV infection (Table 6.5) [9, 67, 68]. A study by Limaye et al. [67], who treated critically ill CMV seropositive adults with ganciclovir noted that treated patients had fewer days on ventilation, both, in the intention-to-treat population and in the prespecified sepsis subgroup. Similarly, CMV may be a co-pathogen in certain cancers (Table 6.5) [10, 69]. Foster et al. reported recently that CMV seropositivity is associated with decreased survival in glioblastoma patients [10]. A multicenter randomized double-blinded controlled phase 2 study evaluating the efficacy of valganciclovir as add-on therapy in glioblastoma patients is ongoing (NCT04116411).

Also in autoimmune diseases, CMV may play an important role [70]: CMV-DNA copy numbers and concurrent infections are predictors of in-hospital mortality in CMVinfected patients with autoimmune diseases (incl. SLE, RA). Therefore, serial measurements of CMV-DNA copy numbers and close observation for signs of other infections were recommended for patients with autoimmune diseases, who have concurrent CMV infection [70]. In addition, there is a strong correlation of worsening symptoms in inflammatory bowel disease, when CMV replication is present (Table 6.5) [71, 72]. Interestingly, a retrospective study of compassionate therapeutic use of letermovir for CMV infections included a patient with systemic lupus erythematosus and a patient with a multicentric form of Castleman disease. In both cases, CMV infections did not recurr [73].

The above examples for a potential benefit for patients, who might receive therapy for CMV suppression beyond the transplantation field, are not exhaustive. The existance of letermovir, a well-tolerated and efficacious drug, now allows to investigate the role of this ubiqitous virus in much more detail in these conditions, without the toxicities of the drugs that needed to be used up to now. This may open new therapeutic concepts, which may include the control of CMV among other interventions, also outside the transplantation field.

6.6 Conclusions

In conclusion, letermovir stands for a new class of CMV inhibitors, which is well tolerated and very efficacious. It is the first novel and non-nucleosidic drug for CMV since decades, which has been licensed with the exception of foscarnet and fomivirsen (for intravitreal injection), both of which have significant limitations.

The availability of letermovir allows for the first time to treat stem cell recipients prophylactically against reactivation of CMV, which translates into a survival benefit. Studies in kidney transplant recipients are ongoing. Letermovir may also be highly useful in a number of other indications outside the transplantation field.

170 6 Antiviral Strategies Against the Human Cytomegalo Virus Inhibitors of Viral Terminase

Letermovir and its predecessors have been discovered by cellular screens. Due to the complexity of the terminase complex, a biochemical assay for the terminase function was impossible at that time. However, at least in our hands and compared with target-based screens, cellular screens have also proven to be more effective for the discovery of other drugs, such as a novel Herpes Simplex drug, Pritelivir [74] (see Birkmann, Chapter 5) or the Hepatitis B drug BAY 41-4109 (AIC 429) [75], which inhibits core formation and leads to degradation of HBV core protein. While a number of other core-inhibitors have meanwhile been discovered Bay 41-4109 and congeners remain the only drugs with the degradation mechanism of core (Chapter 9).

References

- 1 Cannon, M.J., Schmid, D.S., and Hyde, T.B. (2010). Review of cytomegalovirus seroprevalence and demographic characteristics associated with infection. *Reviews in Medical Virology* 20: 202–213. https://doi.org/10.1002/rmv.655.
- **2** Forte, E., Zhang, Z., Thorp, E.B., and Hummel, M. (2020). Cytomegalovirus latency and reactivation: an intricate interplay with the host immune response. *Frontiers in Cellular and Infection Microbiology* 10: 130.
- **3** Mocarski, E.S. Jr. (1988). Biology and replication of cytomegalovirus. *Transfusion Medicine Reviews* 2: 229–234. https://doi.org/10.1016/S0887-7963(88)70050-4.
- **4** Ljungman, P., Perez-Bercoff, L., Jonsson, J. et al. (2006). Risk factors for the development of cytomegalovirus disease after allogeneic stem cell transplantation. *Haematologica* 91: 78–83.
- **5** Teira, P., Battiwalla, M., Ramanathan, M. et al. (2016). Early cytomegalovirus reactivation remains associated with increased transplant-related mortality in the current era: a CIBMTR analysis. *Blood* 127: 2427–2438.
- **6** Freeman, R.B., Paya, C., Pescovitz, M.D. et al. (2004). Risk factors for cytomegalovirus viremia and disease developing after prophylaxis in high-risk solid-organ transplant recipients. *Transplantation* 78: 1765–1773.
- 7 Nassetta, L., Kimberlin, D., and Whitley, R. (2009). Treatment of congenital cytomegalovirus infection: implications for future therapeutic strategies. *The Journal of Antimicrobial Chemotherapy* 63: 862–867.
- **8** Christensen-Quick, A., Vanpouille, C., Lisco, A., and Gianella, S. (2017). Cytomegalovirus and HIV persistence: pouring gas on the fire. *AIDS Research and Human Retroviruses* 33: 23–30.
- **9** Kalil, A.C. and Florescu, D.F. (2011). Is cytomegalovirus reactivation increasing the mortality of patients with severe sepsis? *Critical Care* 15: 1–38.
- **10** Foster, H., Piper, K., DePledge, L. et al. (2019). Human cytomegalovirus seropositivity is associated with decreased survival in glioblastoma patients. *Neuro-Oncology Advances* 1 (1): 1–7.
- **11** Acosta, E., Bowlin, T., Brooks, J. et al. (2020). Advances in the development of therapeutics for cytomegalovirus infections. *The Journal of Infectious Diseases* 221: 32–44.
- 12 Andrei, G., Snoeck, R., and De Clercq, E. (eds.) (2011). Anti-HCMV compounds. In: *Antiviral Drug Strategies*, 227–282. Weinheim: Wiley-VCH.
- 13 Sullivan, V., Talaricof, C.L., Stanatt, S.C. et al. (1992). A protein kinase homologue controls phosphorylation of ganciclovir in human cytomegalovirus-infected cells. *Nature* 358: 162–164.

- 14 https://clinicaltrials.gov/ct2/show/NCT03301415?term=NCT03301415&draw=2&rank=1 (accessed 18.April 2021, Sponsor and Information by: National Institute of Allergy and Infectious Diseases (NIAID))
- **15** https://clinicaltrials.gov/ct2/show/NCT03107871?term=NCT03107871&draw=2&rank=1 (accessed 18 April 2021, Sponsor and responsible for information: Albert Park)
- **16** Wiebusch, L., Truss, M., and Hagemeier, C. (2004). Inhibition of human cytomegalovirus replication by small interfering RNAs. *The Journal of General Virology* 85: 179–184.
- Hamilton, S.T., Milbradt, J., Marschall, M., and Rawlinson, W.D. (2014). Human cytomegalovirus replication is strictly inhibited by siRNAs targeting UL54, UL97 or UL122/123 gene transcripts. *PLoS One* 9 (6): 1–12.
- **18** Dole, K., Pereyra Segal, F., Feire, A. et al. (2016). A first-in-human study to assess the safety and pharmacokinetics of monoclonal antibodies against human cytomegalovirus in healthy volunteers. *Antimicrobial Agents and Chemotherapy* 60: 2881–2887.
- **19** https://clinicaltrials.gov/ct2/show/NCT02268526 (accessed 18 April 2021, Sponsor and responsible for information: Novartis Pharmaceuticals).
- **20** https://www.astellas.com/en/news/10206 (accessed 18 April 2021, information by Astellas Pharma Inc., Bernhardt G. Zeiher, president of Development, Astellas).
- 21 https://www.businesswire.com/news/home/20171109005397/en/Merck-Receives-FDA-Approval-of-PREVYMIS[™]-letermovir-for-Prevention-of-Cytomegalovirus-CMV-Infection-and-Disease-in-Adult-Allogeneic-Stem-Cell-Transplant-Patients (accessed 18. April 2021).
- **22** Biron, K.K., Harvey, R.J., Chamberlain, S.C. et al. (2002). Potent and selective inhibition of human cytomegalovirus replication by 1263W94, a benzimidazole L-riboside with a unique mode of action. *Antimicrobial Agents and Chemotherapy* 46 (8): 2365–2372.
- **23** Marty, F.M., Ljungman, P., Papanicolaou, G.A. et al. (2011). Maribavir prophylaxis for prevention of cytomegalovirus disease in recipients of allogeneic stem-cell transplants: a phase 3, double-blind, placebo-controlled, randomised trial. *The Lancet Infectious Diseases* 11 (4): 284–292.
- 24 Marty, F.M., Winston, D.J., Chemaly, R.F. et al. (2019). A randomized, double-blind, placebocontrolled phase 3 trial of oral brincidofovir for cytomegalovirus prophylaxis in allogeneic hematopoietic-cell transplantation. *Biology of Blood and Marrow Transplantation* 25: 369–381.
- **25** Michaelis, M., Paulis, C., Löschmann, N. et al. (2011). The multi-targeted kinase inhibitor sorafenib inhibits human cytomegalovirus replication. *Cellular and Molecular Life Sciences* 68: 1079–1090.
- **26** Kankanala, J., Wang, Y., Geraghty, R.J., and Wang, Z. (2018). Hydroxypyridonecarboxylic acids as inhibitors of human cytomegalovirus pUL89 endonuclease. *ChemMedChem* 13: 1–7.
- 27 Kapoor, A., Ghosh, A.K., Forman, M. et al. (2020). Validation and characterization of five distinct novel inhibitors of human cytomegalovirus. *Journal of Medicinal Chemistry* 63 (8): 3896–3907.
- **28** NCT02931539: Salvage treatment Maribavir 400mg bid vs SOC in CMV refractory or resistant to Ganciclovir, Valganciclovir, Foscarnet or Cidofovir, N=351.
- **29** NCT02927067: Maribavir 400 mg bid vs Valganciclovir 900 mg bid for CMV treatment in HSCT. Initiation of PET with >900 virus count, endpoint: reduction in virus load.
- **30** Reefschläger, J., Bender, W., Hallenberger, S. et al. (2001). Novel-nonnucleoside inhibitors of cytomegaloviruses (BAY 38-4766): in vitro and in vivo antiviral activity and mechanism of action. *The Journal of Antimicrobial Chemotherapy* 48: 757–767.

172 6 Antiviral Strategies Against the Human Cytomegalo Virus Inhibitors of Viral Terminase

- **31** Weber, O., Bender, W., Eckenberg, P. et al. (2001). Inhibition of murine cytomegalovirus and human cytomegalovirus by a novel non-nucleosidic compound in vivo. *Antiviral Research* 49: 179–189.
- **32** Buerger, I., Reefschläger, J., Bender, W. et al. (2001). A novel nonnucleosidic inhibitor specifically targets cytomegalovirus DNA maturation via the UL89 and UL56 gene products. *Journal of Virology* 75: 9077–9086.
- 33 Goldner, T., Hewlett, G., Ettischer, N. et al. (2011). The novel anticytomegalovirus compound AIC246 (Letermovir) inhibits human cytomegalovirus replication through a specific antiviral mechanism that involves the viral terminase. *Journal of Virology* 85: 10884–10893.
- **34** Rübsamen-Schaeff, H. (2010). HIV DARTConference 8th December 2010, Los Cabos/Mexico.
- Lischka, P., Hewlett, G., Wunberg, T. et al. (2010). In vitro und in vivo activities of the novel anticytomegalovirus compound AIC246. *Antimicrobial Agents and Chemotherapy* 54 (3): 1290–1297.
- 36 Rübsamen-Schaeff, H. (2010). 2nd Antiinfective Summit San Francisco, 28 September 2010.
- **37** Zimmermann, H., Lischka, P., and Ruebsamen-Schaeff, H. (2011). AIC246 a novel drug under development for prevention and treatment of cytomegalovirus infections acting via a novel mechanism of action. *European Infectious Disease* 5 (2): 112–114.
- **38** Kropeit, D., Scheuenpflug, J., Erb-Zohar, K. et al. (2017). Pharmacokinetics and safety of letermovir, a novel anti-human cytomegalovirus drug, in patients with renal impairment. *British Journal of Clinical Pharmacology* 83: 1944–1953.
- 39 Erb-Zohar, K., Kropeit, D., Scheuenpflug, J. et al. (2017). Intravenous hydroxypropyl β-cyclodextrin formulation of letermovir: a phase 1, randomized, single-ascending, and multiple-dose trial. *Clinical and Translational Science* 10: 487–495.
- **40** Kropeit, D., McCormick, D., Erb-Zohar, K. et al. (2017). Pharmacokinetics and safety of the anti-human cytomegalovirus drug letermovir in subjects with hepatic impairment. *British Journal of Clinical Pharmacology* 83: 2678–2686.
- **41** Scheuenpflug, J., Kropeit, D., Erb-Zohar, K., et al. The effect of oral letermovir administration on the pharmacokinetics of a single oral dose of P-glycoprotein (MDR1) substrate digoxin in healthy volunteers AiCuris, data on file.
- **42** Kropeit, D., McCormick, D., Erb-Zohar, K. et al. (2021). Pharmacokinetics and safety of letermovir and midazolam co-administration in healthy subjects. *Clinical Pharmacology in Drug Development,* in press.
- **43** Stoelben, S., Arns, W., Renders, L. et al. (2014). Preemptive treatment of cytomegalovirus infection in kidney transplant recipients with letermovir: results of a Phase 2a study. *Transplant International* 27 (1): 77–86.
- 44 Kaul, D.R., Stoelben, S., Cober, E. et al. (2011). First report of successful treatment of multidrug-resistant cytomegalovirus disease with the novel anti-CMV compound AIC246. *Amercian Journal of Transplantation* 11: 1079–1084.
- **45** Chemaly, R.F., Ullmann, A.J., Stoelben, S. et al. (2014). Letermovir for cytomegalovirus prophylaxis in hematopoietic-cell transplantation. *The New England Journal of Medicine* 370: 1781–1789.
- **46** Marty, F.M., Ljungman, P., Chemaly, R.F. et al. (2017). Letermovir prophylaxis for cytomegalovirus in hematopoietic-cell transplantation. *The New England Journal of Medicine* 377: 2433–2444.

- **47** Green, M.L., Leisenring, W., Xie, H. et al. (2016). CMV viral load and mortality after hematopoietic cell transplantation: a cohort study in the era of preemptive therapy. *Lancet Haematology* 3: 119–127.
- **48** Chou, S. (2020). Advances in the genotypic diagnosis of cytomegalovirus antiviral drug resistance. Antiviral research doi: https:// doi.org/10.1016/j.antiviral.2020.104711.
- **49** Chou, S., Satterwhite, L.E., and Ercolani, R.J. (2018). New locus of drug resistance in the human cytomegalovirus UL56 gene revealed by in vitro exposure to letermovir and ganciclovir. *Antimicrobial Agents and Chemotherapy* 62: 1–6.
- **50** Ligat, G., Cazal, R., Hantz, S., and Alain, S. (2018). The human cytomegalovirus terminase complex as an antiviral target: a close-up view. *FEMS Microbiology Reviews* 42: 137–145.
- **51** Gentry, B.G., Bogner, E., and Drach, J.C. (2019). Targeting the terminase: an important step forward in the treatment and prophylaxis of human cytomegalovirus infections. *Antiviral Research* 161: 116–124.
- **52** Chou, S. (2017). A third component of the human cytomegalovirus terminase complex is involved in letermovir resistance. *Antiviral Research* 148: 1–4.
- **53** Goldner, T., Hempel, C., Ruebsamen-Schaeff, H. et al. (2014). Geno- and phenotypic characterization of human cytomegalovirus mutants selected in vitro after letermovir (AIC246) exposure. *Antimicrobial Agents and Chemotherapy* 58: 610–613.
- **54** Lischka, P., Zhang, D., Holder, D., and Zimmermann, H. (2016). Impact of glycoprotein B genotype and naturally occurring ORF UL56 polymorphisms upon susceptibility of clinical human cytomegalovirus isolates to letermovir. *Antiviral Research* 132: 204–209.
- 55 Lischka, P., Michel, D., and Zimmermann, H. (2016). Characterization of cytomegalovirus breakthrough events in a phase 2 prophylaxis trial of letermovir (AIC246, MK 8228). *The Journal of Infectious Diseases* 213: 23–30.
- **56** Douglas, C.M., Barnard, R., Holder, D. et al. (2019). Letermovir resistance analysis in a clinical trial of cytomegalovirus prophylaxis for hematopoietic stem cell transplant recipients. *The Journal of Infectious Diseases* 221: 1117–1126.
- 57 Bansal, R., Gordillo, C.A., Abramova, R.Assal, A. et al. (2020). Extended letermovir administration, beyond day 100, is effective for CMV prophylaxis in patients with graft versus host disease. Transpl Infect Dis Oct 9; e13487. https://doi:10.1111/tid.13487.
- 58 Cherrier, L., Nasar, A., Goodlet, K. et al. (2018). Emergence of letermovir resistance in a lung transplant recipient with ganciclovir-resistant cytomegalovirus infection. *American Journal of Transplantation* 18: 3060–3064.
- 59 Frietsch, D., Michel, D., Stamminger, T. et al. (2019). Resistance of cytomegalovirus towards letermovir due to C325Y mutation in UL56 in a patient after hematopoietic stem cell transplantation. *Mediterranean Journal of Hematology and Infectious Diseases* 11: 1–4.
- **60** Turner, N., Strand, A., Grewal, D.S. et al. (2019). Use of letermovir for drug-resistent CMV retinitis: a case series Antimicrobila Agents Chemother. *American Society for Microbiology* 63: 1–6.
- **61** Kilgore, J.T., Becken, B., Varga, M.G. et al. (2019). Use of letermovir for salvage therapy for resistant cytomegalovirus in a pediatric hematopoietic stem cell transplant recipient. *Journal of the Pediatric Infectious Diseases Society*: piz050. https://doi.org/10.1093/jpids/piz050.
- **62** Mc Crea, J., Macha, S., Adedoyin, A. et al. (2019). Pharmakokinetic drug-drug interaction between letermovir and the immunesuppressants cyclosporine, tacrolimus, sirolimus and mycophenolate. *Journal of Clinical Pharmacology* 59: 1331–1339.

- 174 6 Antiviral Strategies Against the Human Cytomegalo Virus Inhibitors of Viral Terminase
 - **63** Kimberlin, D., Lin, C.-Y., Sanchez, P. et al. (2003). Effect of ganciclovir therapy on hearing in symptomatic congenital CMV disease involving the central nervoussystem: a randomized, controlled trial. *The Journal of Pediatrics* 143: 16–25.
 - **64** Lackner, A., Acham, A., Alborno, T. et al. (2009). Effect on hearing of ganciclovir therapy for asymptomatic congenital cytomegalovirus infection: four to 10 year follow up. *The Journal of Laryngology and Otology* 123: 391–396.
 - **65** Marsico, C. and Kimberlin, D.W. (2017). Congenital cytomegalovirus infection: advances and challenges in diagnosis, prevention and treatment. *Italian Journal of Pediatrics* 43: 38.
 - 66 Hunt, P.W., Martin, J.N., Sinclair, E. et al. (2011). Valganciclovir reduces T cell activation in HIV-infected individuals with incomplete CD4⁺ T cell recovery on antiretroviral therapy. *The Journal of Infectious Diseases* 203: 1474–1483.
 - 67 Limaye, A.P., Stapleton, R.D., Peng, L. et al. (2017). Effect of ganciclovir on IL-6 levels among cytomegalovirus-seropositive adults with critical illness a randomized clinical trial. *JAMA* 318 (8): 731–740.
 - **68** Heininger, A., Haeberle H., Fischer, I. et al. (2011). Cytomegalovirus reactivation and associated outcome of critically ill patients with severe sepsis. *Critical Care* 15:R77.
 - **69** Söderberg-Nauclér, C. and Johnsen, J.I. (2015). Cytomegalovirus in human brain tumors: role in pathogenesis and potential treatment options (2015) Karolinska Institute. *World Journal of Experimental Medicine* 5 (1): 1–10.
 - **70** Lee, K.Y., Yoo, B.-W., Ahn, S.S. et al. (2017). Predictors of mortality in autoimmune disease patients with concurrent cytomegalovirus infections detected by quantitative real-time PCR. *PLoS ONE* 12 (7):1–12.
 - 71 Rubin, D.T., Ananthakrishnan, A.N., Siegel, C.A., Sauer, B.G., and Long, M.D. (2019). ACG clinical guideline: ulcerative colitis in adults. *American Journal of Gastroenterology* 114: 384–413, cited: Kambham, N., Vij, R., Cartwright, C.A. et al. (2004). Cytomegalovirus infection in steroid-refractory ulcerative colitis: a case-control study. *The American Journal of Surgical Pathology* 28: 365–373.
 - **72** Harbord, M., Eliakim, R., Bettenworth, D. et al. (2017). Third European evidence-based consensus on diagnosis and management of ulcerative colitis. Part 2: current management. *Journal of Crohn's and Colitis* 2017: 769–784.
 - 73 Schubert, L., Fisecker, L., Thalhammer, F., Burgmann, H., and Steininger, C. (2020). Letermovir for the compassionate use of cytomagalovirus infection. *European Journal of Clinical Microbiology & Infectious Dieseases*. https://doi.org/10.1007/s10096-020-03990-w.
 - **74** Kleymann, G., Fischer, R., Betz, U. et al. (2002). New helicase-primase inhibitors as drug candidates for the treatment of herpes simplex disease. *Nature Medicine* 8 (4): 392–398.
 - **75** Deres, K., Schröder, C.H., Paessens, A. et al. (2003). Inhibition of hepatitis B virus replication by drug-induced depletion of nucleocapsids. *Science* 299: 893–896.

Antiviral Targeting of the Complex Epstein Barr Virus Life Cycle

Christian Münz

7

Viral Immunobiology, Institute of Experimental Immunology, University of Zürich, Zürich, Switzerland

7.1 Disease Overview

Epstein Barr virus (EBV) is surprisingly at the same time one of the most common pathogens and the most oncogenic virus that infects humans [1]. More than 95% of the human adult population carries this γ -herpesvirus as a persistent asymptomatic infection. Yet, it readily transforms human B cells in culture and is associated with a variety of lymphocyte, epithelial, and smooth muscle cell-derived malignancies that fortunately occur only at low incidence rates [2, 3]. They nevertheless are estimated to amount to around 200000 new cancers in humans every year [4]. In addition to these tumors, EBV is also the cause of several immune pathologies. These include infectious mononucleosis (IM) as the symptomatic primary EBV infection, more frequently observed when the virus is first encountered later in life during adolescence or early adulthood [5]. Furthermore, myeloid cell activation that results in hemophagocytic lymphohistiocytosis (HLH) results from inefficient killing of EBV-infected cells, primarily B cells as the main host cell of the virus, triggering then pathogenic cytokine production by lymphocytes that stimulate myeloid cells [6]. Uncontrolled EBV infection then often also spreads from B cells to other lymphocyte compartments like natural killer (NK) and T cells [7] and might also home to the central nervous system (CNS) to stimulate autoimmunity resulting in multiple sclerosis (MS) [8]. Transition to these pathologies is fortunately avoided in most EBV carriers by immune control by cytotoxic lymphocytes, mainly NK and CD8⁺ T cells [9]. Accordingly, the reestablishment of this immune control by antiviral therapies would be desirable for the treatment of EBV-associated diseases. In order to devise respective strategies, it is, however, necessary to understand which aspects of the viral life cycle need to be targeted.

EBV is usually transmitted via saliva exchange, although transmission by organ transplantation and maybe sexual intercourse is also possible [10]. Most likely, it enters submucosal secondary lymphoid tissues of the oropharynx, like the tonsils, via transcytosis across the mucosal epithelium [11, 12]. At these sites, it infects B cells, initiating growth transforming gene expression called EBV latency [13]. This includes at the protein level up

176 7 Antiviral Targeting of the Complex Epstein Barr Virus Life Cycle

to six Epstein Barr virus nuclear antigens (EBNAs) and two latent membrane proteins (LMPs). In addition, as non-translated RNAs, two Epstein-Barr virus-encoded small RNAs (EBERs) and more than 40 miRNAs are expressed [14]. B cell transformation by these latent EBV gene products was suggested to drive infected B cells into memory differentiation for long-term persistence [15], and only if T cell-mediated immune control is missing like after immune suppression following organ transplantation or due to human immune deficiency virus (HIV) coinfection, latently EBV-infected B cells emerge as lymphomas [16] (Figure 7.1). From infected memory B cells that do not express any latent EBV proteins, lytic replication for infectious virion production can be induced and seems to be stimulated after plasma cell differentiation that is stimulated after B cell receptor stimulation by the cognate antigen [17]. Thus, latent EBV proteins have been the focus of therapeutic interventions to target EBV-associated malignancies. However, it was recently noted that early, but most likely not late lytic EBV gene expression contributes to viral oncogenesis [1, 18, 19]. While late lytic gene expression and virion production lyses the respective host cells, early lytic gene products like viral IL-10 but also chemokines that are induced at this infection stage might condition the microenvironment for more efficient growth of EBV-associated malignancies (Figure 7.1). Furthermore, IM is primarily driven by the immune pathologic expansion of cytotoxic CD8⁺ T cells that recognize early lytic antigens [20]. Thus, early, but



Figure 7.1 Both latent and early lytic EBV infection contribute to virus-associated tumorigenesis. EBV-associated lymphomagenesis is driven by viral oncogenes, such as EBNA2, EBNA3C, and LMP1. These are expressed during EBV latent infection with up to 6 nuclear antigens (EBNAs), 2 membrane antigens (LMPs), 2 small noncoding RNAs, and more than 40 miRNAs during latency III. Lymphomagenesis is further promoted by early lytic EBV infection, conditioning the tumor microenvironment by factors such as viral IL-10 (vIL-10). However, late lytic EBV infection with virion replication actually destroys tumor cells and thereby diminishes lymphomagenesis. *Source*: Servier Medical Art templates, Licensed under a Creative Commons Attribution 3.0 unported license: https://smart.servier.com.

not late lytic EBV proteins should also be targeted therapeutically in addition to latent gene products to treat EBV-associated tumors and some of the virus-associated immune pathologies.

7.2 Antiviral Strategies

7.2.1 Pharmacological Inhibition

Following the above discussed considerations, it is probably not too surprising that inhibition of EBV's DNA polymerase, required for infectious virion production, on its own has no therapeutic effects against EBV-induced malignancies like post-transplant lymphoproliferative disease (PTLD) [21]. Only in combination with B cell depleting and chemotherapy, viral DNA polymerase inhibitors like ganciclovir caused durable clinical responses in a small number of primary CNS PTLDs [22]. These combination therapies might induce lytic EBV replication and virion production, making the respective cells susceptible to viral DNA polymerase inhibition (Figure 7.2), while at the same time removing latently infected cells with B cell depleting therapy. Along these lines, histone deacetylase (HDAC) inhibitors and DNA damage-inducing drugs can be combined with ganciclovir for efficient EBV-associated tumor cell lysis [23, 24]. However, the pharmacological agents that have so far been used for this purpose in patients suffer from low specificity and efficacy for lytic EBV reactivation. Butyrates and romidepsin inhibit HDACs, but demonstrated either poor pharmacokinetics or intolerable toxicity during treatment of EBV-associated lymphomas [25–27]. Therefore, more specific inducers of lytic replication that progresses all the way to virion production need to be identified [28] and other targets of EBV infection should be explored for pharmacological inhibition of EBV-associated diseases. One such target is EBNA1 [29]. EBNA1 is required to maintain the EBV genome extra-chromosomally as a circular, multi-copy episome in infected cells by anchoring it to mitotic chromosomes during cell division and amplifying it prior to mitosis (Figure 7.2a). Without EBNA1, the virus transforms human B cells several thousand-fold less efficiently [30]. EBNA1 is therefore the only viral protein that is expressed in all EBV-associated tumors, including Burkitt lymphoma that expresses it as the sole viral protein [31]. Thus, targeting the viral genome maintenance function of EBNA1 could address an Achilles heel of EBV and be therapeutically useful against all associated malignancies. The EBNA1 domain that has mainly been explored is the C-terminal portion binding the viral DNA as a homodimer. Its structure bound to DNA had been previously solved [32] and viral DNAderived palindromic oligonucleotide binding by recombinant versions of this domain have been explored for inhibition by small molecular compound libraries [33, 34]. Furthermore, EBNA1 phosphorylation inhibition and DNA modifications to prevent EBNA1 binding have been investigated to cause viral episome loss [35, 36]. Based on these encouraging results, four druggable pockets in the C-terminal domain of EBNA1 were predicted [37]. Two narrow sites with potential for allosteric inhibition of EBNA1 function and two wide pockets in the DNA binding or dimerization surfaces were identified. Molecular docking simulations were performed on the wide and narrow pocket in proximity to the viral DNA binding domain with a small molecular compound library. These



Figure 7.2 Therapeutic interventions against EBV-associated malignancies. (a) EBV-infected tumor cells express transforming latent viral proteins such as EBNA1 and LMP1. Pharmacological inhibitors against these are currently developed. Alternatively, their differentiation into late lytic EBV-infected cells has been therapeutically explored which makes these cells then susceptible to viral DNA polymerase inhibition by, for example, ganciclovir (b). Furthermore, EBV-associated lymphomas are treated via antibody-mediated depletion targeting their surface molecules CD20, CD19, and CD30. Moreover, adoptive T cell therapies have been explored against EBV-associated malignancies for more than 25 years. In addition to EBV-specific T cell receptor (TCR) specificities, also chimeric antigen receptors (CARs) targeting CD19, CD20, CD30, or LMP1 are investigated. Finally, EBV-specific vaccination is being developed with recombinant viral envelope antigen formulations like glycoprotein multimers and virus like particles (VLPs) or recombinant viral vectors expressing latent EBV antigens. (c) One of the lead compounds for an EBNA1 inhibitor [38]. *Source:* Servier Medical Art templates, Licensed under a Creative Commons Attribution 3.0 unported license: https://smart.servier.com.

studies identified lead compounds that have shown promising results in xenograft models of the EBV-associated epithelial cell cancer nasopharyngeal carcinoma (NPC) [38]. The so far most promising of these lead compounds is shown in Figure 7.2c. In addition to EBNA1, also LMP1 as the main viral oncogene [39] is being explored for pharmacological inhibition. This protein constitutively signals similar to the tumor necrosis factor (TNF) receptor superfamily member CD40 upon auto-aggregation [40]. Indeed, homomultimerization sequences have been suggested as targets to inhibit LMP1 function and peptides that inhibit them or binding of TNF receptor-associated factors (TRAFs) are currently being explored [41] (Figure 7.2a). Thus, pharmacological targeting of latent EBV gene products, mainly EBNA1 and LMP1, or reactivation of lytic virion production are probably more suitable to target EBV-associated malignancies than viral DNA polymerase inhibition (Figure 7.2b).

7.2.2 B Cell Depleting Therapy

The greatest success in the treatment of EBV-associated lymphomas has so far been achieved with B cell depleting therapies, deploying reagents such as the anti-CD20 antibody rituximab (Figure 7.2a). Particularly for PTLD, B cell depleting therapies have reduced the cumulative incidence at least 10-fold from around 20% to slightly above 1% [42, 43]. This has currently even led to pretreatment with rituximab in transplant patients that are at risk to develop PTLD due to MHC, sex, and EBV serostatus mismatch [44, 45]. In addition to PTLD, other CD20-positive EBV-associated lymphomas, like diffuse large B cell lymphoma and Burkitt lymphoma, respond to rituximab treatment [46]. Interestingly, B cell depletions with the anti-CD20 antibodies rituximab and ocrelizumab demonstrated also clinical efficacy in MS patients [47–51] (Figure 7.3). MS risk is two-fold elevated after IM and 3.6-fold with elevated EBNA1-specific antibodies [52]. Both risk factors synergize with the main genetic risk factor for this autoimmune disease, the MHC class II molecule HLA-DRB1*1501, for odds ratios of 7 and 15, respectively. Moreover, in a longitudinal study, all EBV seronegative individuals seroconverted prior to MS onset [53]. The high viral titers during IM and the persistence of elevated EBNA1-specific antibody responses might set up and indicate, respectively, a reservoir of EBV-infected B cells in MS patients. Since LMP1 expression during latent EBV infection confers antigen presentation functions to B cells [54, 55], this reservoir in MS patients might stimulate autoimmune CD4⁺ T cells [8] (Figure 7.3). B cell depletion might eliminate this reservoir of EBV-infected B cells and ameliorate disease. However, CD20 targeting eliminates all B cells prior to plasma cell differentiation and therefore might provoke a general loss of humoral immune control, especially after long-term application. Thus, a more selective targeting of EBV-infected cells is desirable. Along these lines, anti-CD30 antibodies like brentuximab have been explored in Hodgkin lymphoma and diffuse large B cell lymphoma, of which 10 or 40% are EBV positive [56]. Especially in its auristatin E conjugated form, inhibiting tumor cell mitosis via disruption of microtubule polymerization, CD30 targeting has resulted in promising results [57, 58] (Figure 7.2a). Enrichment of auristatin E in the tumor microenvironment might even be achieved if only a subset of the tumor cells is CD30 positive [59, 60]. In contrast to depletion of CD20 positive cells, CD30 targeted therapies show also promising results for some T cell lymphomas, like systemic anaplastic large-cell lymphoma (ALCL) [61, 62]. In individual patients, drug-conjugated brentuximab has also been reported to provide clinical benefits for EBV-associated extranodal NK/T cell lymphomas [63, 64]. Thus, CD30 targeting could deplete EBV transformed B and NK/T cells more selectively in patients with the respective lymphomas.

180 7 Antiviral Targeting of the Complex Epstein Barr Virus Life Cycle



Figure 7.3 Therapeutic interventions against EBV-associated immunopathologies. Early lytic EBV antigens drive immunopathological CD8⁺ T cell expansion during IM, the primary symptomatic EBV infection that occurs more often upon the first EBV encounter later in life. Inefficient cytotoxic immune control of EBV-infected cells allows this immune pathology then to progress to hemophagocytic lymphohistiocytosis (HLH) during which lymphocyte-derived cytokines activate myeloid cells to cause tissue damage and erythrocyte phagocytosis. IM also predisposes for multiple sclerosis (MS) by possibly establishing a reservoir of EBV-infected B cells that efficiently stimulates autoimmune CD4⁺ T cell responses. The benefit of B cell depletion in MS and adoptive transfer of EBV-specific T cells might eliminate this detrimental reservoir. *Source*: Servier Medical Art templates, Licensed under a Creative Commons Attribution 3.0 unported license: https://smart.servier.com.

7.2.3 Adoptive T Cell Transfer and Immune Checkpoint Blockade

EBV-associated PTLD was one of the first clinical settings in which therapeutic adoptive transfer of antigen specific T cells was explored [65, 66] (Figure 7.2a). These T cell lines were originally generated by in vitro stimulation of donor-derived T cells with the autologous EBV-transformed B cell lines prior to transfer into the bone marrow recipient suffering from PTLD. However, this protocol required generation of the autologous EBV-transformed B cell line or lymphoblastoid cell line (LCL) from the bone marrow donor and expansion of EBV-specific T cells from the same donor for a month. In order to shorten the time period until transfer into the patient, EBV antigen-derived peptide pulsed DCs were explored next [67, 68]. These allowed focusing the transferred T cell product on distinct EBV antigens, some of which are also expressed in lymphomas, like Hodgkin lymphoma, with reduced latent EBV gene expression. The LMPs, primarily LMP2, were targeted in some of these approaches [69, 70]. Alternatively, EBNA1-specific T cells were selected prior to adoptive transfer with clinical responses in PTLD patients [71]. EBNA1,

LMP1, and LMP2-specific T cell lines were also explored with some clinical benefit in MS patients to improve immune control of virus-induced antigen presenting cells [72, 73] (Figure 7.3). In order to further reduce the time to treatment and choose both MHC restriction elements that are matched between bone marrow donor and recipient during allogeneic transplantation, banks of EBV-specific T cell lines and clones have been developed [74-76]. These different adoptive T cell transfer protocols achieved durable clinical responses in the majority, often two thirds, of treated patients suffering from EBV-associated lymphomas either expressing all eight latent EBV antigens or only EBNA1, LMP1, and LMP2. For further improvement of these therapies, especially to protect the adoptively transferred T cells from the immune suppressive tumor microenvironment and to endow them with additional specificities, genetic modifications were explored. For example, transgenic expression of a dominant negative receptor for the immune suppressive cytokine TGF- β seemed to confer increased clinical potency of LMP-specific T cell transfer into Hodgkin lymphoma patients [77]. Alternatively, blocking or deleting inhibitory receptors like PD-1 on these T cell products could further increase their clinical potency. Indeed, PD-1 blocking antibodies have led to significant clinical responses in Hodgkin lymphoma patients [78].

Finally, EBV-specific T cell lines might also be extremely well suitable for the expression of chimeric antigen receptors (CARs) or additional T cell receptors for mainly three reasons (Figure 7.2a). (i) For the treatment of EBV-associated lymphoma CARs that target CD19, CD20, CD30, or LMP1, and TCRs against LMP1 and 2 presented by common MHC class I molecules like HLA-A2 are explored [79–84]. The first benefit to express these additional antigen receptors on EBV-specific T cell lines is the benefit from re-stimulation upon viral antigen encounter after transfer into patients [79]. (ii) The second reason is that they can actually control reactivation of EBV infection at the same time as they mediate their antitumor effects [68]. (iii) Third, these EBV-specific T cell lines are depleted of alloreactive specificities, which minimizes cytokine release syndrome (CRS) after adoptive transfer [76]. Thus, adoptive T cell transfer is an efficient therapy of EBV-associated diseases and the use of allogeneic T cell banks and genetic modifications also reduces some of the logistic issues that were originally associated with autologous EBV-specific T cell line generation prior to transfer into patients.

7.2.4 Vaccination

Developing a vaccine against EBV to therapeutically reinstall immune control in patients with EBV-associated diseases and protect adolescents from IM is complicated due to the nature of the protective immune response against EBV. This immune control seems to nearly exclusively depend on cytotoxic lymphocytes, like CD8⁺ T cells and NK cells, that can sufficiently expand during EBV infection [6, 9]. In contrast, type I and II interferon (IFN) and antibody deficiencies do not predispose for EBV-associated diseases [6]. Furthermore, established immune control of EBV does not protect from reinfection and/or circulation between the oropharynx and the blood [85, 86]. Thus, mucosal immunity, including neutralizing antibody responses against EBV, is not sufficient to prevent reinfection.

Nevertheless, the first vaccine that was tried against EBV aimed for the glycoprotein 350 of the EBV envelope that the virus uses to attach to B cells via CD21 [87, 88]. This

182 7 Antiviral Targeting of the Complex Epstein Barr Virus Life Cycle

recombinant gp350 vaccine reduced IM incidence in young adults by 78%, but did not prevent EBV infection. Based on these encouraging findings multimeric gp350 vaccine formulations were developed [89] (Figure 7.2a). Furthermore, additional EBV glycoproteins were included in this vaccine formulation to elicit antibodies that block both entry into B and epithelial [90]. The neutralizing activity of the antibody responses that are raised against these vaccine candidates in mice and monkeys are quite encouraging, but any clinical effect of such a vaccination beyond IM prevention remains unclear. Therefore, additional vaccine formulations to elicit cytotoxic T cell responses have been explored. These include a recombinant modified vaccinia virus Ankara (MVA) that encodes for an EBNA1 fusion protein with LMP2 [91] (Figure 7.2a). It has been used in British and Chinese patients with the EBV-associated epithelial cell cancer NPC [92, 93]. Increased T cell responses to one or both of the vaccine antigens were observed in at least half of the treated individuals. In addition, an adenovirus encoding EBNA1 and a LMP1 and LMP2 derived polyepitope was developed [94]. It was used for expansion of EBV-specific T cell lines for adoptive transfer into NPC and MS patients [72, 73, 95]. The potency of these two recombinant vaccine formulations of adenoviral CD8⁺ and MVA-mediated CD4⁺ T cell priming can possibly be combined for a more potent induction of protective EBV-specific T cell responses [96]. EBNA1 as the sole vaccination antigen might be sufficient in this heterologous prime boost vaccination approach. In addition to recombinant viral vaccines, also virus-like particles (VLPs) of EBV itself have been explored to elicit both protective antibody and T cell responses, but have so far only been tested in preclinical models [97, 98] (Figure 7.2a). However, in one of these studies, a VLP containing EBNA1 in addition to the structural capsid, tegument and envelope proteins were able to elicit protective T cell responses that dampened EBV infection in mice with reconstituted human immune system components [98]. These studies with recombinant viruses and VLPs suggest that EBNA1 should be included as a vaccine component to elicit protective T cell responses [99].

7.3 Open Issues

More than 50 years after the discovery of EBV as the first human tumor virus [100, 101], investigators are starting to develop specific drugs and vaccines, encouraged by the success of virus-specific adoptive T cell transfer. While the above discussed treatments target latent transforming and late lytic structural EBV gene products, early lytic proteins that might contribute to tumorigenesis [1] are so far not explored for EBV treatments. Future treatment approaches might also want to consider this group of antigens. The implementation of the so-far developed treatments will in addition to addressing important clinical needs answer important questions of EBV immunobiology, namely if targeting of cellular transformation diminishes EBV persistence, if sterilizing immunity can be achieved with more potent antibody induction and if therapeutic reduction or maybe even elimination of persistent EBV infection has also negative effects on the human immune system. Along these lines, recent studies in preclinical models suggested that EBV persistence benefits from, but does not require cellular transformation [102].

The strong NIH program to develop an EBV vaccine that elicits envelope protein-specific antibody responses [89, 90] might elicit more potent humoral immunity than EBV infection

itself. However, the danger exists that if sterilizing immunity can only be achieved for a certain time period, primary infection is delayed and might then elicit more frequent and more severe IM symptoms. Finally, the EBV inhibitors in development could actually clear infection, but reinfection would probably occur prior to loss of EBV-specific immune memory and therefore without IM symptoms [85, 86]. However, during this time period of EBV loss, it would be interesting to investigate which overall changes occur in the human immune system and if our leucocytes benefit from having EBV as a sparring partner, as has been demonstrated for the β -herpesvirus cytomegalovirus [103].

Acknowledgements

Research in my laboratory is supported by Cancer Research Switzerland (KFS-4091-02-2017), KFSP-Precision^{MS} of the University of Zurich, the Vontobel Foundation, the Baugarten Foundation, the Sobek Foundation, the Swiss Vaccine Research Institute, Roche, ReiThera, and the Swiss National Science Foundation (310030B_182827 and CRSII5_180323).

References

- 1 Münz, C. (2019). Latency and lytic replication in the oncogenesis of the Epstein Barr virus. *Nat. Rev. Micobiol.* 17: 691–700.
- **2** Cesarman, E. (2014). Gammaherpesviruses and lymphoproliferative disorders. *Annu. Rev. Pathol.* 9: 349–372.
- **3** Shannon-Lowe, C. and Rickinson, A. (2019). The global landscape of EBV-associated tumors. *Front. Oncol.* 9: 713.
- 4 Cohen, J.I., Fauci, A.S., Varmus, H., and Nabel, G.J. (2011). Epstein–Barr virus: an important vaccine target for cancer prevention. *Sci. Transl. Med.* 3 (107): 107fs7.
- **5** Dunmire, S.K., Verghese, P.S., and Balfour, H.H. Jr. (2018). Primary Epstein–Barr virus infection. *J. Clin. Virol.* 102: 84–92.
- **6** Latour, S. and Fischer, A. (2019). Signaling pathways involved in the T-cell-mediated immunity against Epstein–Barr virus: lessons from genetic diseases. *Immunol. Rev.* 291 (1): 174–189.
- **7** Kim, W.Y., Montes-Mojarro, I.A., Fend, F., and Quintanilla-Martinez, L. (2019). Epstein– Barr virus-associated T and NK-cell lymphoproliferative diseases. *Front. Pediatr.* 7: 71.
- **8** Ascherio, A., Munger, K.L., and Lunemann, J.D. (2012). The initiation and prevention of multiple sclerosis. *Nat. Rev. Neurol.* 8 (11): 602–612.
- **9** Damania, B. and Münz, C. (2019). Immunodeficiencies that predispose to pathologies by human oncogenic gamma-herpesviruses. *FEMS Microbiol. Rev.* 43: 181–192.
- **10** Niederman, J.C., Miller, G., Pearson, H.A. et al. (1976). Infectious mononucleosis. Epstein–Barr-virus shedding in saliva and the oropharynx. *N. Engl. J. Med.* 294 (25): 1355–1359.
- **11** Tugizov, S.M., Berline, J.W., and Palefsky, J.M. (2003). Epstein–Barr virus infection of polarized tongue and nasopharyngeal epithelial cells. *Nat. Med.* 9 (3): 307–314.
- **12** Tugizov, S.M., Herrera, R., and Palefsky, J.M. (2013). Epstein–Barr virus transcytosis through polarized oral epithelial cells. *J. Virol.* 87 (14): 8179–8194.

- **13** Thorley-Lawson, D.A. (2001). Epstein–Barr virus: exploiting the immune system. *Nat. Rev. Immunol.* 1 (1): 75–82.
- **14** Cullen, B.R. (2013). MicroRNAs as mediators of viral evasion of the immune system. *Nat. Immunol.* 14 (3): 205–210.
- **15** Babcock, J.G., Hochberg, D., and Thorley-Lawson, A.D. (2000). The expression pattern of Epstein–Barr virus latent genes in vivo is dependent upon the differentiation stage of the infected B cell. *Immunity* 13 (4): 497–506.
- **16** Taylor, G.S., Long, H.M., Brooks, J.M. et al. (2015). The immunology of Epstein–Barr virus-induced disease. *Annu. Rev. Immunol.* 33: 787–821.
- 17 Laichalk, L.L. and Thorley-Lawson, D.A. (2005). Terminal differentiation into plasma cells initiates the replicative cycle of Epstein–Barr virus in vivo. *J. Virol.* 79 (2): 1296–1307.
- **18** Ma, S.D., Hegde, S., Young, K.H. et al. (2011). A new model of Epstein–Barr virus infection reveals an important role for early lytic viral protein expression in the development of lymphomas. *J. Virol.* 85 (1): 165–177.
- **19** Okuno, Y., Murata, T., Sato, Y. et al. (2019). Defective Epstein–Barr virus in chronic active infection and haematological malignancy. *Nat. Microbiol.* 4 (3): 404–413.
- Callan, M.F., Tan, L., Annels, N. et al. (1998). Direct visualization of antigen-specific CD8+ T cells during the primary immune response to Epstein–Barr virus in vivo. *J. Exp. Med.* 187 (9): 1395–1402.
- **21** AlDabbagh, M.A., Gitman, M.R., Kumar, D. et al. (2017). The role of antiviral prophylaxis for the prevention of Epstein–Barr virus-associated posttransplant lymphoproliferative disease in solid organ transplant recipients: a systematic review. *Am. J. Transplant.* 17 (3): 770–781.
- 22 Dugan, J.P., Haverkos, B.M., Villagomez, L. et al. (2018). Complete and durable responses in primary central nervous system post-transplant lymphoproliferative disorder with Zidovudine, Ganciclovir, Rituximab and Dexamethasone. *Clin. Cancer Res.* 24 (14): 3273–3281.
- Feng, W.H. and Kenney, S.C. (2006). Valproic acid enhances the efficacy of chemotherapy in EBV-positive tumors by increasing lytic viral gene expression. *Cancer Res.* 66 (17): 8762–8769.
- **24** Ghosh, S.K., Forman, L.W., Akinsheye, I. et al. (2007). Short, discontinuous exposure to butyrate effectively sensitizes latently EBV-infected lymphoma cells to nucleoside analogue antiviral agents. *Blood Cells Mol. Dis.* 38 (1): 57–65.
- 25 Perrine, S.P., Hermine, O., Small, T. et al. (2007). A phase 1/2 trial of arginine butyrate and ganciclovir in patients with Epstein–Barr virus-associated lymphoid malignancies. *Blood* 109 (6): 2571–2578.
- **26** Ghosh, S.K., Perrine, S.P., Williams, R.M., and Faller, D.V. (2012). Histone deacetylase inhibitors are potent inducers of gene expression in latent EBV and sensitize lymphoma cells to nucleoside antiviral agents. *Blood* 119 (4): 1008–1017.
- **27** Kim, S.J., Kim, J.H., Ki, C.S. et al. (2016). Epstein–Barr virus reactivation in extranodal natural killer/T-cell lymphoma patients: a previously unrecognized serious adverse event in a pilot study with romidepsin. *Ann. Oncol.* 27 (3): 508–513.
- **28** Tikhmyanova, N., Paparoidamis, N., Romero-Masters, J. et al. (2019). Development of a novel inducer for EBV lytic therapy. *Bioorg. Med. Chem. Lett.* 29 (16): 2259–2264.
- 29 Frappier, L. (2015). EBNA1. Curr. Top. Microbiol. Immunol. 391: 3-34.

- **30** Humme, S., Reisbach, G., Feederle, R. et al. (2003). The EBV nuclear antigen 1 (EBNA1) enhances B cell immortalization several thousandfold. *Proc. Natl. Acad. Sci. USA* 100 (19): 10989–10994.
- **31** Thorley-Lawson, D.A. and Allday, M.J. (2008). The curious case of the tumour virus: 50 years of Burkitt's lymphoma. *Nat. Rev. Microbiol.* 6 (12): 913–924.
- **32** Bochkarev, A., Barwell, J.A., Pfuetzner, R.A. et al. (1996). Crystal structure of the DNAbinding domain of the Epstein–Barr virus origin-binding protein, EBNA1, bound to DNA. *Cell* 84 (5): 791–800.
- **33** Li, N., Thompson, S., Schultz, D.C. et al. (2010). Discovery of selective inhibitors against EBNA1 via high throughput in silico virtual screening. *PLoS ONE* 5 (4): e10126.
- **34** Thompson, S., Messick, T., Schultz, D.C. et al. (2010). Development of a high-throughput screen for inhibitors of Epstein–Barr virus EBNA1. *J. Biomol. Screen*. 15 (9): 1107–1115.
- **35** Kang, M.S., Lee, E.K., Soni, V. et al. (2011). Roscovitine inhibits EBNA1 serine 393 phosphorylation, nuclear localization, transcription, and episome maintenance. *J. Virol.* 85 (6): 2859–2868.
- 36 Yasuda, A., Noguchi, K., Minoshima, M. et al. (2011). DNA ligand designed to antagonize EBNA1 represses Epstein–Barr virus-induced immortalization. *Cancer Sci.* 102 (12): 2221–2230.
- Gianti, E., Messick, T.E., Lieberman, P.M., and Zauhar, R.J. (2016). Computational analysis of EBNA1 "druggability" suggests novel insights for Epstein–Barr virus inhibitor design. *J. Comput. Aided Mol. Des.* 30 (4): 285–303.
- 38 Messick, T.E., Smith, G.R., Soldan, S.S. et al. (2019). Structure-based design of smallmolecule inhibitors of EBNA1 DNA binding blocks Epstein–Barr virus latent infection and tumor growth. *Sci. Transl. Med.* 11 (482): eaau5612.
- **39** Kulwichit, W., Edwards, R.H., Davenport, E.M. et al. (1998). Expression of the Epstein– Barr virus latent membrane protein 1 induces B cell lymphoma in transgenic mice. *Proc. Natl. Acad. Sci. USA* 95 (20): 11963–11968.
- **40** Kieser, A. and Sterz, K.R. (2015). The latent membrane protein 1 (LMP1). *Curr. Top. Microbiol. Immunol.* 391: 119–149.
- **41** Yasui, T., Luftig, M., Soni, V., and Kieff, E. (2004). Latent infection membrane protein transmembrane FWLY is critical for intermolecular interaction, raft localization, and signaling. *Proc. Natl. Acad. Sci. USA* 101 (1): 278–283.
- **42** Styczynski, J., Gil, L., Tridello, G. et al. (2013). Response to rituximab-based therapy and risk factor analysis in Epstein Barr Virus-related lymphoproliferative disorder after hematopoietic stem cell transplant in children and adults: a study from the Infectious Diseases Working Party of the European Group for Blood and Marrow Transplantation. *Clin. Infect. Dis.* 57 (6): 794–802.
- **43** Curtis, R.E., Travis, L.B., Rowlings, P.A. et al. (1999). Risk of lymphoproliferative disorders after bone marrow transplantation: a multi-institutional study. *Blood* 94 (7): 2208–2216.
- 44 Stocker, N., Labopin, M., Boussen, I. et al. (2020). Pre-emptive rituximab treatment for Epstein–Barr virus reactivation after allogeneic hematopoietic stem cell transplantation is a worthwhile strategy in high-risk recipients: a comparative study for immune recovery and clinical outcomes. *Bone Marrow Transplant.* 55 (3): 586–594.
- **45** Kim, B.K., Kang, H.J., Hong, K.T. et al. (2019). Successful Preemptive Therapy with single dose Rituximab for Epstein–Barr Virus Infection to Prevent

Post-Transplant-Lymphoproliferative Disease after Pediatric Hematopoietic Stem Cell Transplantation. *Transpl. Infect. Dis.* 21 (6): e13182.

- **46** Salles, G., Barrett, M., Foa, R. et al. (2017). Rituximab in B-cell hematologic malignancies: a review of 20 years of clinical experience. *Adv. Ther.* 34 (10): 2232–2273.
- **47** Naismith, R.T., Piccio, L., Lyons, J.A. et al. (2010). Rituximab add-on therapy for breakthrough relapsing multiple sclerosis: a 52-week phase II trial. *Neurology* 74 (23): 1860–1867.
- **48** Hawker, K., O'Connor, P., Freedman, M.S. et al. (2009). Rituximab in patients with primary progressive multiple sclerosis: results of a randomized double-blind placebo-controlled multicenter trial. *Ann. Neurol.* 66 (4): 460–471.
- **49** Hauser, S.L., Waubant, E., Arnold, D.L. et al. (2008). B-cell depletion with rituximab in relapsing-remitting multiple sclerosis. *N. Engl. J. Med.* 358 (7): 676–688.
- **50** Hauser, S.L., Bar-Or, A., Comi, G. et al. (2017). Ocrelizumab versus interferon beta-1a in relapsing multiple sclerosis. *N. Engl. J. Med.* 376 (3): 221–234.
- **51** Montalban, X., Hauser, S.L., Kappos, L. et al. (2017). Ocrelizumab versus placebo in primary progressive multiple sclerosis. *N. Engl. J. Med.* 376 (3): 209–220.
- **52** Olsson, T., Barcellos, L.F., and Alfredsson, L. (2017). Interactions between genetic, lifestyle and environmental risk factors for multiple sclerosis. *Nat. Rev. Neurol.* 13 (1): 25–36.
- 53 Levin, L.I., Munger, K.L., O'Reilly, E.J. et al. (2010). Primary infection with the Epstein– Barr virus and risk of multiple sclerosis. *Ann. Neurol.* 67 (6): 824–830.
- 54 Brooks, J.M., Lee, S.P., Leese, A.M. et al. (2009). Cyclical expression of EBV latent membrane protein 1 in EBV-transformed B cells underpins heterogeneity of epitope presentation and CD8+ T cell recognition. J. Immunol. 182 (4): 1919–1928.
- 55 Leung, C.S., Maurer, M.A., Meixlsperger, S. et al. (2013). Robust T-cell stimulation by Epstein–Barr virus-transformed B cells after antigen targeting to DEC-205. *Blood* 121 (9): 1584–1594.
- **56** Deutsch, Y.E., Tadmor, T., Podack, E.R., and Rosenblatt, J.D. (2011). CD30: an important new target in hematologic malignancies. *Leuk. Lymphoma* 52 (9): 1641–1654.
- 57 Younes, A., Gopal, A.K., Smith, S.E. et al. (2012). Results of a pivotal phase II study of brentuximab vedotin for patients with relapsed or refractory Hodgkin's lymphoma. *J. Clin. Oncol.* 30 (18): 2183–2189.
- 58 Moskowitz, C.H., Nademanee, A., Masszi, T. et al. (2015). Brentuximab vedotin as consolidation therapy after autologous stem-cell transplantation in patients with Hodgkin's lymphoma at risk of relapse or progression (AETHERA): a randomised, double-blind, placebo-controlled, phase 3 trial. *Lancet* 385 (9980): 1853–1862.
- **59** Jacobsen, E.D., Sharman, J.P., Oki, Y. et al. (2015). Brentuximab vedotin demonstrates objective responses in a phase 2 study of relapsed/refractory DLBCL with variable CD30 expression. *Blood* 125 (9): 1394–1402.
- **60** Bartlett, N.L., Chen, R., Fanale, M.A. et al. (2014). Retreatment with brentuximab vedotin in patients with CD30-positive hematologic malignancies. *J. Hematol. Oncol.* 7: 24.
- **61** Pro, B., Advani, R., Brice, P. et al. (2012). Brentuximab vedotin (SGN-35) in patients with relapsed or refractory systemic anaplastic large-cell lymphoma: results of a phase II study. *J. Clin. Oncol.* 30 (18): 2190–2196.
- 62 Prince, H.M., Kim, Y.H., Horwitz, S.M. et al. (2017). Brentuximab vedotin or physician's choice in CD30-positive cutaneous T-cell lymphoma (ALCANZA): an international, open-label, randomised, phase 3, multicentre trial. *Lancet* 390 (10094): 555–566.

- **63** Kim, H.K., Moon, S.M., Moon, J.H. et al. (2015). Complete remission in CD30-positive refractory extranodal NK/T-cell lymphoma with brentuximab vedotin. *Blood Res.* 50 (4): 254–256.
- **64** Poon, L.M. and Kwong, Y.L. (2016). Complete remission of refractory disseminated NK/T cell lymphoma with brentuximab vedotin and bendamustine. *Ann. Hematol.* **95** (5): 847–849.
- **65** Heslop, H.E., Brenner, M.K., and Rooney, C.M. (1994). Donor T cells to treat EBVassociated lymphoma. *N. Engl. J. Med.* 331 (10): 679–680.
- **66** Leen, A.M., Rooney, C.M., and Foster, A.E. (2007). Improving T cell therapy for cancer. *Annu. Rev. Immunol.* 25: 243–265.
- **67** Hanley, P.J., Cruz, C.R., Savoldo, B. et al. (2009). Functionally active virus-specific T cells that target CMV, adenovirus, and EBV can be expanded from naive T-cell populations in cord blood and will target a range of viral epitopes. *Blood* 114 (9): 1958–1967.
- 68 Papadopoulou, A., Gerdemann, U., Katari, U.L. et al. (2014). Activity of broad-spectrum T cells as treatment for AdV, EBV, CMV, BKV, and HHV6 infections after HSCT. *Sci. Transl. Med.* 6 (242): 242ra83.
- **69** Bollard, C.M., Straathof, K.C., Huls, M.H. et al. (2004). The generation and characterization of LMP2-specific CTLs for use as adoptive transfer from patients with relapsed EBV-positive Hodgkin disease. *J. Immunother.* 27 (4): 317–327.
- **70** Bollard, C.M., Gottschalk, S., Torrano, V. et al. (2014). Sustained complete responses in patients with lymphoma receiving autologous cytotoxic T lymphocytes targeting Epstein–Barr virus latent membrane proteins. *J. Clin. Oncol.* 32 (8): 798–808.
- **71** Icheva, V., Kayser, S., Wolff, D. et al. (2013). Adoptive transfer of Epstein–Barr virus (EBV) nuclear antigen 1-specific T cells as treatment for EBV reactivation and lymphoproliferative disorders after allogeneic stem-cell transplantation. *J. Clin. Oncol.* 31 (1): 39–48.
- 72 Pender, M.P., Csurhes, P.A., Smith, C. et al. (2014). Epstein–Barr virus-specific adoptive immunotherapy for progressive multiple sclerosis. *Mult. Scler.* 20 (11): 1541–1544.
- **73** Pender, M.P., Csurhes, P.A., Smith, C. et al. (2018). Epstein–Barr virus-specific T cell therapy for progressive multiple sclerosis. *JCI Insight*. 3 (22): e124714.
- **74** Leen, A.M., Bollard, C.M., Mendizabal, A.M. et al. (2013). Multicenter study of banked third-party virus-specific T cells to treat severe viral infections after hematopoietic stem cell transplantation. *Blood* 121 (26): 5113–5123.
- 75 O'Reilly, R.J., Prockop, S., Hasan, A., and Doubrovina, E. (2019). Therapeutic advantages provided by banked virus-specific T-cells of defined HLA-restriction. *Bone Marrow Transplant*. 54 (Suppl 2): 759–764.
- 76 Doubrovina, E., Oflaz-Sozmen, B., Prockop, S.E. et al. (2012). Adoptive immunotherapy with unselected or EBV-specific T cells for biopsy-proven EBV⁺ lymphomas after allogeneic hematopoietic cell transplantation. *Blood* 119 (11): 2644–2656.
- **77** Bollard, C.M., Tripic, T., Cruz, C.R. et al. (2018). Tumor-specific T-cells engineered to overcome tumor immune evasion induce clinical responses in patients with relapsed Hodgkin lymphoma. *J. Clin. Oncol.* 36 (11): 1128–1139.
- **78** Ansell, S.M., Lesokhin, A.M., Borrello, I. et al. (2015). PD-1 blockade with nivolumab in relapsed or refractory Hodgkin's lymphoma. *N. Engl. J. Med.* 372 (4): 311–319.
- **79** Lapteva, N., Gilbert, M., Diaconu, I. et al. (2019). T cell receptor stimulation enhances the expansion and function of CD19 chimeric antigen receptor-expressing T cells. *Clin. Cancer Res.* 25 (24): 7340–7350.

- Savoldo, B., Rooney, C.M., Di Stasi, A. et al. (2007). Epstein Barr virus specific cytotoxic T lymphocytes expressing the anti-CD30zeta artificial chimeric T-cell receptor for immunotherapy of Hodgkin disease. *Blood* 110 (7): 2620–2630.
- Chu, Y., Yahr, A., Huang, B. et al. (2017). Romidepsin alone or in combination with anti-CD20 chimeric antigen receptor expanded natural killer cells targeting Burkitt lymphoma in vitro and in immunodeficient mice. *Oncoimmunology* 6 (9): e1341031.
- Tang, X., Zhou, Y., Li, W. et al. (2014). T cells expressing a LMP1-specific chimeric antigen receptor mediate antitumor effects against LMP1-positive nasopharyngeal carcinoma cells in vitro and in vivo. *J. Biomed. Res.* 28 (6): 468–475.
- Cho, H.I., Kim, U.H., Shin, A.R. et al. (2018). A novel Epstein–Barr virus-latent membrane protein-1-specific T-cell receptor for TCR gene therapy. *Br. J. Cancer* 118 (4): 534–545.
- Jurgens, L.A., Khanna, R., Weber, J., and Orentas, R.J. (2006). Transduction of primary lymphocytes with Epstein–Barr virus (EBV) latent membrane protein-specific T-cell receptor induces lysis of virus-infected cells: a novel strategy for the treatment of Hodgkin's disease and nasopharyngeal carcinoma. *J. Clin. Immunol.* 26 (1): 22–32.
- 85 Sitki-Green, D., Covington, M., and Raab-Traub, N. (2003). Compartmentalization and transmission of multiple Epstein–Barr virus strains in asymptomatic carriers. *J. Virol.* 77 (3): 1840–1847.
- Walling, D.M., Brown, A.L., Etienne, W. et al. (2003). Multiple Epstein–Barr virus infections in healthy individuals. *J. Virol.* 77 (11): 6546–6550.
- Moutschen, M., Leonard, P., Sokal, E.M. et al. (2007). Phase I/II studies to evaluate safety and immunogenicity of a recombinant gp350 Epstein–Barr virus vaccine in healthy adults. *Vaccine* 25 (24): 4697–4705.
- 88 Sokal, E.M., Hoppenbrouwers, K., Vandermeulen, C. et al. (2007). Recombinant gp350 vaccine for infectious mononucleosis: a phase 2, randomized, double-blind, placebo-controlled trial to evaluate the safety, immunogenicity, and efficacy of an Epstein–Barr virus vaccine in healthy young adults. *J. Infect. Dis.* 196 (12): 1749–1753.
- Kanekiyo, M., Bu, W., Joyce, M.G. et al. (2015). Rational design of an Epstein–Barr virus vaccine targeting the receptor-binding site. *Cell* 162 (5): 1090–1100.
- **90** Bu, W., Joyce, M.G., Nguyen, H. et al. (2019). Immunization with components of the viral fusion apparatus elicits antibodies that neutralize Epstein–Barr virus in B cells and epithelial cells. *Immunity* 50 (5): 1305–1316. e6.
- **91** Taylor, G.S., Haigh, T.A., Gudgeon, N.H. et al. (2004). Dual stimulation of Epstein–Barr virus (EBV)-specific CD4⁺- and CD8⁺-T-cell responses by a chimeric antigen construct: potential therapeutic vaccine for EBV-positive nasopharyngeal carcinoma. *J. Virol.* 78 (2): 768–778.
- **92** Hui, E.P., Taylor, G.S., Jia, H. et al. (2013). Phase I trial of recombinant modified vaccinia ankara encoding Epstein–Barr viral tumor antigens in nasopharyngeal carcinoma patients. *Cancer Res.* 73 (6): 1676–1688.
- Taylor, G.S., Jia, H., Harrington, K. et al. (2014). A recombinant modified vaccinia ankara vaccine encoding Epstein–Barr virus (EBV) target antigens: a phase I trial in UK patients with EBV-positive cancer. *Clin. Cancer Res.* 20 (19): 5009–5022.
- **94** Duraiswamy, J., Bharadwaj, M., Tellam, J. et al. (2004). Induction of therapeutic T-cell responses to subdominant tumor-associated viral oncogene after immunization with replication-incompetent polyepitope adenovirus vaccine. *Cancer Res.* 64 (4): 1483–1489.

- **95** Smith, C., Tsang, J., Beagley, L. et al. (2012). Effective treatment of metastatic forms of Epstein–Barr virus-associated nasopharyngeal carcinoma with a novel adenovirus-based adoptive immunotherapy. *Cancer Res.* 72 (5): 1116–1125.
- **96** Rühl, J., Citterio, C., Engelmann, C. et al. (2019). Heterologous prime-boost vaccination protects from EBV antigen expressing lymphomas. *J. Clin. Invest.* 129: 2071–2087.
- **97** Ruiss, R., Jochum, S., Wanner, G. et al. (2011). A virus-like particle-based Epstein–Barr virus vaccine. *J. Virol.* 85 (24): 13105–13113.
- **98** van Zyl, D.G., Tsai, M.H., Shumilov, A. et al. (2018). Immunogenic particles with a broad antigenic spectrum stimulate cytolytic T cells and offer increased protection against EBV infection ex vivo and in mice. *PLoS Pathog.* 14 (12): e1007464.
- **99** Münz, C. (2004). Epstein–Barr virus nuclear antigen 1: from immunologically invisible to a promising T cell target. *J. Exp. Med.* 199: 1301–1304.
- **100** Epstein, M.A., Achong, B.G., and Barr, Y.M. (1964). Virus particles in cultured lymphoblasts from Burkitt's lymphoma. *Lancet* 1: 702–703.
- **101** Epstein, M.A., Henle, G., Achong, B.G., and Barr, Y.M. (1964). Morphological and biological studies on a virus in cultured lymphoblasts from Burkitt's lymphoma. *J. Exp. Med.* 121: 761–770.
- **102** Murer, A., McHugh, D., Caduff, N. et al. (2018). EBV persistence without its EBNA3A and 3C oncogenes in vivo. *PLoS Pathog.* 14 (4): e1007039.
- **103** Barton, E.S., White, D.W., Cathelyn, J.S. et al. (2007). Herpesvirus latency confers symbiotic protection from bacterial infection. *Nature* 447: 326–329.

Kaposi's Sarcoma-associated Herpesvirus—Antiviral Treatment

Thomas F. Schulz

8

Institute of Virology, Hannover Medical School, Hannover, Germany

8.1 Introduction to Kaposi's Sarcoma-associated Herpesvirus (KSHV)

Kaposi's sarcoma-associated herpesvirus (KSHV), taxonomical name human herpesvirus 8 (HHV8), is a human γ_2 -herpesvirus (rhadinovirus) and the cause of three human malignancies, a systemic inflammatory condition and rare cases of hemophagocytosis with bone marrow failure and hepatitis. The three malignant diseases are Kaposi's sarcoma (KS), in which this virus was originally discovered and from which the most commonly used name for this virus is derived [1], primary effusion lymphoma (PEL) [2], and the plasma cell variant of multicentric Castleman's disease (MCD) [3]. In addition, KSHV has been linked to rare cases of post-transplant polyclonal lymphoproliferative disorders with plasmacytic differentiation [4-6]. Among its nonmalignant clinical manifestations are KSHV-associated inflammatory cytokine syndrome (KICS) [7-10], KSHV-induced hemophagocytosis in transplant recipients [11–13], and rare cases of KSHV-associated hepatitis in transplanted patients [14, 15]. Substantial epidemiological evidence, together with a plethora of mechanistic experimental studies support the causative involvement of this virus in the pathogenesis of the three malignant tumors KS, PEL and MCD, and KSHV has consequently been classified as a class I human carcinogen by the International Agency for Research against Cancer (IARC), a WHO agency tasked with evaluating the carcinogenic risk to humans. The evidence leading to this classification has been summarized [16, 17] and has also been discussed in many recent reviews to which the reader is referred [18–21].

8.2 Epidemiological Considerations

Unlike most other human herpesviruses, KSHV shows a very uneven geographical distribution. It is common in sub-Saharan Africa, with prevalence rates of antibodies to KSHV usually higher than 50% in adult populations and particularly high seroprevalence rates in

192 8 Kaposi's Sarcoma-associated Herpesvirus-Antiviral Treatment

East and Central Africa (for a review, see [17]). In countries bordering on the Mediterranean, prevalence is intermediate, with seroprevalence rates in the 3–20% range and considerable regional variability. In contrast, the virus is very rare in Northern and Western Europe as well as most of Asia. Higher prevalence rates have also been reported in populations of African descent in the Americas, as well as in some native American populations [22, 23]; for detailed references, see [17]. Further examples of KSHV being associated with particular populations are the higher KSHV seroprevalence in Uighur populations in Xinjiang, China [24] and in the Buryat population in Southern Siberia [25], as well as the higher seroprevalence and presence of particular KSHV genotypes in Israel [26]. In addition, KSHV seroprevalence rates are often higher among men who have sex with men (MSM) than in the general population of the same country (detailed references in [17]).

In endemic countries or populations with a high KSHV seroprevalence, transmission of the virus occurs mainly in childhood before puberty, with mother–child transmission, as well as transmission among siblings with virus-containing saliva being considered an important route of spread [27–33]. In adults living in KSHV endemic countries, as well as for individuals at increased risk of sexually transmitted diseases, there is evidence of sexual transmission of KSHV [34–38]. KSHV-containing saliva is also thought to play an important role during sexual transmission [27, 37].

8.3 Disease Overview

8.3.1 Kaposi's Sarcoma

From a clinical point of view, four distinct variants of KS can be distinguished. First described by Moritz Kaposi in 1872, the "**classical**" form of KS is today defined as a mostly indolent, slowly progressing tumor on the lower extremities of elderly men from countries with a higher KSHV seroprevalence, such as countries around the Mediterranean. Patients with classic KS always have antibodies to KSHV, the virus can be detected by PCR in tumor biopsies and tumor cells latently infected with KSHV can be identified in tissue sections by immunohistochemistry staining for the KSHV latent nuclear antigen, LANA (see Figure 8.1 and Section 8.5.4.1). Classic KS is a rare tumor, with reported incidence rates ranging from less than 1 to about 3 per 100000 population, even in regions of Southern Europe with KSHV seroprevalence rates in the range of 10–20% [17]. In these geographic regions, the development of KS in a KSHV seropositive individual is therefore a rare event and the vast majority of KSHV-infected individuals will never experience any clinical manifestations.

Clinically more aggressive and often characterized by the involvement of internal organs such as the gut or the lung is the **African endemic** form of KS in HIV-negative individuals. First described in the 1920s and characterized clinically in the 1960s, this clinical variant occurs predominantly in East and Central Africa. Environmental cofactors are thought to contribute to the more aggressive behavior of these tumors by promoting virus replication. Suggested candidates for such cofactors include other infectious agents, in particular helminths or malaria, but also certain plant chemicals: there is substantial epidemiological evidence for helminth infections and/or malaria coinfection promoting KSHV productive replication and KSHV infection [39–42]. Furthermore, there is laboratory evidence to
suggest that extracts from certain plants found in East and Central Africa may promote KSHV replication in tissue culture [43].

The third clinical KS variant is **transplant-associated** or "**iatrogenic**" **KS**. Transplantassociated KS occurs several hundred fold more frequently in transplant recipients compared to the general population of the same country [44]. It is encountered mainly in countries with increased KSHV seroprevalence rates. Estimations of its frequency vary from <1 to 11% of solid organ transplant recipients, depending on the country and the transplanted organ, with post-transplant KS more common in kidney transplant recipients than recipients of other solid organs and very rare in hematopoetic stem cell transplant recipients [45–54]. KS occurs much more frequently in transplant recipients who are KSHV seropositive at the time of transplantation: a pooled analysis of six studies on kidney, liver, and heart transplant recipients from Italy and France suggests that approximately a quarter (25.6%) of transplant recipients who were KSHV seropositive before transplantation developed transplant-associated KS [6, 55–58]; more extensive data are reviewed in [45, 59]. The



Figure 8.1 Expression of KSHV LANA in a KS biopsy. (a) Immunohistochemistry staining using an antibody to KSHV LANA. KSHV-infected, LANA-expressing endothelial spindle cells are stained in dark brown color. (b) HE stain. The panel shows the atypical endothelial KS spindle cells, which are the histological hallmark of KS lesions. *Source*: Dr. Guntram Büsche, Dept. of Pathology, Hannover Medical School.

rate of post-transplant KS may have decreased in the last decade following the switch from calcineurin inhibitors as immunosuppressive agents in solid organ transplant recipients with KS to an mTOR inhibitor-based regimen (see Section 8.4.2.1; [59]). Most cases of transplant KS are thought to result from the reactivation of KSHV in patients who were already infected with KSHV prior to transplantation [45, 56, 60]. However, cases of donor-recipient transmission of KSHV have been noted [61, 62]. These findings indicate that transplant-associated KS is more of a problem in countries with a higher KSHV seroprevalence and in transplant recipients with pre-existing KSHV infection.

The incidence of KS increased markedly with the worldwide spread of HIV in the 1980s and **AIDS-associated KS**, or **epidemic KS**, now represents the commonest KS variant worldwide. KS is several thousand times more frequent in HIV-infected individuals than in the general population of the same country [17, 44]. In East Africa (Uganda), KS was, in the early 1990s, the commonest cancer in men, and is today the second most common cancer in men and the third most frequent cancer in women in this geographic region [63]. Incidence rates of KS in Ugandan men were in the range of 30 per 100 000 in the first decade of the twenty-first century, having decreased from even higher levels after the introduction of effective antiretroviral combination therapy (ART) [63]. KS incidence rates have also decreased strongly after the introduction of ART in Western countries but remain much higher among HIV-infected persons than in the general population [64, 65].

An important therapeutic challenge therefore is the HIV-infected patient with well-controlled HIV viral load whose KS tumor does not regress or who develops new KS lesions under effective antiretroviral therapy. In the United States, AIDS-associated KS cases in patients with well-controlled HIV now account for about a third of all KS presentations [66, 67]. Together with transplant recipients with KS, these are the KS patients who are most in need of innovative treatment approaches and for whom innovative drugs with activity against KSHV-associated disease are required.

8.3.2 Primary Effusion Lymphoma (PEL)

PEL, which is also referred to as body cavity-based lymphoma, is a very rare malignant B cell non-Hodgkin lymphoma occurring mainly in AIDS patients and rarely also in transplant recipients ([2, 5, 68, 69]; further literature in [17, 59]). It accounts for 2–5% of HIV-related lymphomas and presents as pleural, peritoneal, and pericardial lymphomatous effusions, often in the absence of a clinically detectable solid tumor mass. The presence of KSHV DNA in the tumor cells is considered a diagnostic criterion. Thus, other AIDS-related lymphomas presenting with body cavity effusion but lacking KSHV in the tumor cell are not classified as PEL. PEL tumor cells are of post-germinal center B cell origin, lack many typical B cell surface markers, and show a plasmacytic differentiation pattern, as assessed by their transcriptome and the expression of CD134 [70].

8.3.3 Multicentric Castleman's Disease (MCD)

MCD is a polyclonal lymphoproliferative disease. The vast majority of HIV-associated MCD cases and about half of HIV-negative cases are infected with KSHV ([3, 71, 72]; further references in [17]). Unlike localized Castleman disease (which is often KSHV

negative), the multicentric form behaves more aggressively and is often fatal. There are several reports of non-Hodgkin lymphomas, in particular plasmablastic lymphoma and EBV-negative PEL tumors, developing out of MCD lesions [68, 73]. KSHV-infected B cells in MCD express only lambda light chains and are therefore monotypic; they express CD38, IRF4, BLIMP1 but are commonly negative for CD30, CD40, and the plasma cell marker CD138. These cells also do not exhibit somatic mutations in the hypervariable regions of their immunoglobulin genes and are therefore of a pre-germinal center differentiation stage despite displaying evidence of a plasmacytic differentiation [70].

The clinical presentation of KSHV-associated MCD cases is often characterized by symptoms and laboratory evidence of systemic inflammation. Clinical symptoms include, in addition to lymphadenopathy, fever, night sweats, weight loss, diarrhea, and fatigue. Abnormal laboratory findings consist of hypergammaglobulinemia, cytopenia, hypoalbuminemia, elevated CRP levels as well as raised serum levels of human IL6, IL10, and the viral cytokine vIL6 [72, 74–76]. There is often abundant expression of human IL6 in the germinal centers of MCD lymph nodes and expression vIL6 is regularly seen in KSHV-infected plasmacytic B cells in these lesions [75, 77, 78].

In addition to vIL6, other lytic viral proteins may also be expressed in MCD lesions [77–79], in addition to the latent viral protein LANA (see Section 8.5.4.1). KSHV viral load in peripheral blood is often high in patients with active MCD tumors [70]. There are reports of the successful treatment of MCD cases with ganciclovir, an inhibitor of the KSHV DNA polymerase and thereby of lytic (productive) replication [80] (see Section 8.4.1.1.5). More recently, a combination treatment with high-dose ganciclovir and AZT has been shown to be beneficial in MCD cases [81] (see Section 8.4.1.1.5). These observations suggest that KSHV lytic (productive) replication and viral proteins expressed during the lytic replication cycle may contribute to the pathogenesis of MCD (see Section 8.4.1.1.5). Although KSHV-infected plasmablasts in MCD are often negative for CD20, rituximab (an antibody to CD20) is very effective in controlling MCD with response rates higher than 70% [59, 82–84]. Rituximab can be applied in combination with liposomal doxorubicin to prevent flares of KS [85].

8.3.4 KSHV and Polyclonal Post-transplant Lymphoproliferative Disease

There are several case reports of polyclonal lymphoproliferative disease in transplant recipients involving KSHV-infected B cells with a plasmacytic differentiation, but which do not meet the diagnostic criteria for PEL or MCD [4–6]. Their clinical presentation may include systemic symptoms, such as fever, anemia, skin rash, arthritis, and the diffuse lymphocytic infiltration of lymph nodes and visceral organs [6].

8.3.5 Kaposi Sarcoma Herpesvirus-Associated Inflammatory Cytokine Syndrome (KICS)

KICS is defined as a clinical syndrome of systemic inflammation in KSHV- and HIVcoinfected patients in the absence of MCD. KICS is characterized by clinical signs such as fever, night sweat, fatigue, diarrhea, weight loss, thrombocytopenia, hypoalbuminemia, markedly elevated CRP levels, high KSHV viral load in peripheral blood, and raised IL6 and IL10 levels [7–10, 14, 86]. Patients with KICS frequently also present with KS or PEL and are often severely ill with 50% survival rates of around a couple of years [9]. KICS can occur in patients with well-controlled HIV load [87]. As for MCD, treatment can be attempted with rituximab to eliminate KSHV-infected B-cells [88] or with monoclonal antibodies against the IL6 receptor such as tocilizumab [86].

8.3.6 KSHV and Hemophagocytosis, Bone Marrow Failure, and Hepatitis

KSHV has also been linked to cases of bone marrow failure, sometimes associated with evidence of hemophagocytosis, and rare cases of hepatitis in transplant recipients [10–15]. These clinical manifestations may be accompanied by a high KSHV viral load in peripheral blood and respond to treatment with antivirals or rituximab (see Section 8.4.1.1 and [59]).

8.4 Antiviral Strategies

As already touched on in the preceding sections, the treatment of KSHV-associated diseases may either be directed at the virus itself, or against KSHV-infected cells, tumor cells, or cellular cytokines secreted as part of the disease process. In the following section, the focus will be on established and experimental antiviral strategies.

8.4.1 Established Antiviral Strategies

8.4.1.1 Inhibitors of the KSHV DNA Polymerase (Table 8.1)

8.4.1.1.1 Activity in Tissue Culture

Several competitive nucleoside inhibitors developed against other herpesviral DNA polymerases and approved for clinical use also show activity against KSHV. In tissue culture, several studies have found cidofovir to be the most potent drug and to inhibit viral DNA replication with IC50 values ranging from 0.05 to $6\,\mu$ M, depending on the cell culture and virus detection assay used [89–92]. Ganciclovir (reported IC50 values $0.96-11\,\mu$ M), foscarnet ($34-177\,\mu$ M), brivudine ($0.6\,\mu$ M), and adefovir ($39\,\mu$ M) are also inhibitors, while aciclovir and penciclovir are less effective [89–93]. Among newer nucleoside inhibitors that are not yet in clinical use, HPMPA ((S)-9-[3-hydroxy-2-(phosphonomethoxy)-propyl] adenine; $0.6\,\mu$ M), S2242 ([(1,3-dihydroxy-2-propoxymethyl)purine]; $0.1\,\mu$ M), (E)-CVDC ((E)-5-(2-chlorovinyl)-2'deoxycytidine; $2.4\,\mu$ M), CVDU (5-(2-Chlorovinyl)-2'-deoxyuridine; $5.5\,\mu$ M), FIAC (2'-fluoro-5-iodo-aracytosine; $1\,\mu$ M), N-MCT (2'-exo-methanocarbathymidine; $0.08\,\mu$ M), and HDVD (1-[(2S,4S-2-(hydroxymethyl)-1,3-dioxolan-4-yl]5-vinylpyrimidine-2,4(1H,3H)-dione; $0.09\,\mu$ M) showed strong activity [91, 93–95]. Additional compounds with good activity against KSHV in tissue culture include cyclopropavir and several 6-alkoxy-substituted methylene cyclopropane nucleosides (cyclopropavir derivatives; [96]). For an overview, see Table 8.1.

With the exception of foscarnet, these compounds need to be activated by phosphorylation in the infected cell by virus-encoded kinases. In the case of KSHV, the viral protein kinase (vPK) encoded by open reading frame (ORF) 36, which is the homologue of the cytomegalovirus pUL97 vPK, mediates phosphorylation of ganciclovir and acyclovir [93, 112]. The second KSHV kinase is encoded by KSHV ORF21 and represents the

	Compound	Target	IC50	Chemical structure	References
Clinical trial or case report data available	Cidofovir	Viral DNA polymerase	0.05-6µM	HO-P-O-(S) HO-P-O-(S)	[89–92]
	Ganciclovir	Viral DNA polymerase	2.6-11 µM		[89–93]
	Foscarnet	Viral DNA polymerase	34–177 µM		[89–93]
	Brivudine	Viral DNA polymerase	0.6 μΜ		[97, 98]
	Adefovir	Viral DNA polymerase	39 µM		[98]
No clinical data available	HPMPA (<i>S</i>)-9-[3-hydroxy- 2-(phosphono- methoxy)-propyl] adenine	Viral DNA polymerase	0.6 µM	H N H N N O H O O O H	[91, 93]
	S2242 (1,3-dihydroxy- 2-propoxymethyl) purine	Viral DNA polymerase	0.1 μΜ	HO O N NH2 OH	[91, 93]
	(E)-CVDC (E)-5-(2- chlorovinyl)-2'- deoxycytidine	Viral DNA polymerase	2.4 µM		[91, 93]

 Table 8.1
 Directly acting antivirals (DAA) with activity against KSHV.

(Continued)

Table 8.1 (Continued)

Compound	Target	IC50	Chemical structure	References
CVDU 5-(2-Chlorovinyl)- 2'-deoxyuridine	Viral DNA polymerase	5.5 μΜ		[91, 93]
FIAC 2'-fluoro- 5-iodo- aracytosine	Viral DNA polymerase	1μM		[91, 93]
HDVD 1-[(2S,4S-2- (hydroxymethyl)- 1,3-dioxolan-4- yl]5- vinylpyrimidine- 2,4(1H,3H)-dione	Viral DNA polymerase	0.09 μΜ		[94]
Cyclopropavir	Viral DNA polymerase	~4µM		[96]
2'-Exo- methanocarba- thymidine	Viral DNA polymerase	~0.1 µM		[95]
NSC 373989	pORF9 (Pol)/ pORF59 (PAF) complex	~2µM		[99]
XZ45	pORF6 (ssDNA BP)? pORF29 (terminase)?	n.a.		[100]

Compound	Target	IC50	Chemical structure	References
Raltegravir	pORF29 (terminase)		N-N N H OH	[101]
Dolutegravir	pORF29 (terminase)		الله الله الله الله الله الله الله الله	[101]
Compound 14	KSHV protease	~24 µM		[102, 103]
Nelfinavir	KSHV protease? Cellular target?	~5µM		[104, 105]
K8.1-scFV- immunotoxin	Glycoprotein K8.1		scFV-exotoxin A conjugate	[106]
gH-scFV- immunotoxin	Glycoprotein gH		scFV-exotoxin A conjugate	[107]
LANA inhibitor I	LANA interaction with the viral latent origin of replication	17–19μM in <i>in vitro</i> DNA binding (fluorescence polarization) assay	HO J C C C C C C C C C C C C C C C C C C	[108, 109]
Cyclic ΙΚΚγ- mimetic peptide	vFLIP	Approx. 25–50 µM in apoptosis assays	Ac-LQVAYH X LFQ X YDNH IKSSC-NH ₂	[110, 111]

Table 8.1 (Continued)

homologue of the HSV UL23 and VZV ORF36 thymidine kinases (TK) [93, 97, 113]. Although KSHV TK is primarily a protein tyrosine kinase [113] and only a comparatively poor TK [114], it does mediate the phosphorylation and activation of brivudine and the anti-HIV nucleoside reverse transcriptase inhibitor azidothymidine (AZT) [93, 97]. In contrast to the α -herpesviral TK of HSV and VZV, KSHV TK does not efficiently phosphorylate thymidine compounds such as aciclovir [114].

8.4.1.1.2 Activity in KSHV-infected Patients

Reflecting their activity against KSHV in tissue culture, the three clinically approved nucleoside inhibitors of the herpesviral DNA polymerase (ganciclovir, cidofovir, and foscarnet) have been shown to reduce KSHV shedding or viral load in peripheral blood in patients. Oral valganciclovir has been reported to reduce KSHV shedding in oral samples in one study [115]. Valacyclovir and famciclovir treatment was associated with a moderate reduction in the number of treated individuals shedding KSHV in the oropharynx [116]. While two groups failed to observe a reduction of the KSHV load in peripheral blood leukocytes following treatment with ganciclovir or foscarnet [117, 118], others reported that foscarnet decreased the KSHV viral load in peripheral blood and led to a clinical improvement in two transplant patients with KSHV-associated pancytopenia and hemophagocytosis [11, 119]. However, in spite of this indication of a moderate *in vivo* effect, herpesviral DNA polymerase inhibitors have only limited clinical efficacy when used to treat KSHV-associated disease.

8.4.1.1.3 Treatment of Kaposi Sarcoma

When used in patients with KS, herpesviral DNA polymerase inhibitors do not seem to be effective against this tumor. Krown and colleagues did not observe any clinical improvement in five patients with classic KS after treatment with valganciclovir [120]. Similarly, cidofovir had no beneficial impact on disease progression in five cases of AIDS-KS and two classic KS cases [121]. However, Mazzi and colleagues reported a beneficial effect of cidofovir on KSHV viremia and disease progression in two patients with AIDS KS [122]. In contrast, intralesional injection of cidofovir in one patient with classic KS was ineffective [123]. This combined experience probably reflects the fact that the viral gene expression program in KS tumors is mostly restricted to the latent viral genes LANA, vcyc, vFLIP, and a cluster of viral miRNAs, although it may in some KS tumors also extend to a few early viral genes, such as K15 [77, 78, 124, 125]. Productive viral replication-which would be susceptible to inhibitors of the viral DNA polymerase and may be required for the activation of prodrugs by KSHV vPK/ORF36 or TK/ORF21 (see Section 8.4.1.1.1)—therefore probably only occurs in a small proportion of infected cells. It is worth noting, however, that the use of herpesviral DNA polymerase inhibitors as preventive treatment, or as treatment for other herpesviral diseases, has consistently been found to lower the incidence of KS [126, 127]. This observation suggests that herpesviral DNA polymerase inhibitors may reduce viral replication and the infection of new cells in KSHV-infected patients and thereby decrease the risk of infected endothelial cells developing into KS tumors.

Valganciclovir and cidofovir have been used successfully in combination with liposomal doxorubicin/daunorubicin, the conventional cancer chemotherapy for KS [128, 129], but the contribution of the herpesviral DNA polymerase inhibitors to the treatment success is difficult to ascertain from these case studies.

8.4.1.1.4 Treatment of PEL

Since PEL is a very rare non-Hodgkin lymphoma (see Section 8.3.2), there are only case reports or case series describing the use of herpesviral DNA polymerase inhibitors in individual cases. Parenteral ganciclovir, in some cases followed by oral valganciclovir, or cidofovir, sometimes injected into the body cavities affected by malignant effusions, produced disease remissions in some, but not all cases [130–132]. Ganciclovir in combination with chemotherapy was also beneficial in one case [133]. Overall, this experience suggests that herpesviral DNA polymerase inhibitors are not the mainstay of PEL treatment, but could be tried in combination with conventional cancer therapy in particular cases.

8.4.1.1.5 Treatment of MCD

Similar to the treatment of PEL with herpesviral DNA polymerase inhibitors, the experience with these drugs, when used on their own, in patients suffering from MCD has been mixed. While there are reports of the successful treatment of KSHV- and HIV-associated MCD with ganciclovir [80], cidofovir failed in five patients [134]. Several successful treatment attempts combined chemotherapy, rituximab, or the proteasome inhibitor bortezomib with ganciclovir [135–137].

A pilot trial of combining two nucleoside inhibitors, ganciclovir and the antiretroviral reverse transcriptase inhibitor AZT, in high doses yielded very promising results [81], with 12 of 14 treated patients showing a major clinical response and 7 experiencing a major improvement of biochemical parameters (levels of CRP, albumin, sodium, hemoglobin, and platelet count; see Section 8.3.3). Since AZT is phosphorylated and activated to a toxic moiety by KSHV TK [114], it is possible that this high-dose AZT/GCV combination therapy succeeded in eliminating KSHV-infected B-cells expressing TK, while GCV, activated by KSHV vPK/ORF36 (see Section 8.4.1.1.1), might have had an antiviral effect [81].

8.4.2 Kinase Inhibitors Against KSHV-associated Disease (Table 8.2)

8.4.2.1 Clinical Experience

Clinically approved tyrosine kinase inhibitors have been used to treat KSHV-associated diseases, in particular KS. The rationale for their use lies in the observation that several cellular receptor tyrosine kinases such as c-kit, PDGFR, and EphA2 play important roles in the KSHV life cycle (see reviews [19, 20]). Imatinib, an inhibitor of c-abl, c-kit, and PDGFR, has been used in clinical trials of KS and induced a partial tumor regression in about a third of patients [138, 139]. Similarly, sorafenib, which targets receptors of VEGF and PDGF as well as c-kit, has been tried with moderate success [140, 141].

The PI3K/AKT-mTOR pathway has emerged as an important player in the proliferation and angiogenesis of KSHV-infected endothelial cells as well as in KSHV productive replication (reviewed in [19–21]). Several viral proteins, including the nonstructural viral membrane proteins encoded by ORFs K1, ORF74 (the viral homologue of a G protein-coupled receptor; vGCR), ORF K15, the vPK encoded by ORF36, and the viral IL6 homologue (vIL6) contribute to the activation of the PI3/AKT pathway in KSHV-infected cells undergoing the early stages of viral reactivation or showing a "relaxed latency" viral transcriptional program [125, 142–146]. The mTOR kinase in the PI3K/AKT pathway is targeted by the FK506/rapamycin complex and rapamycin has proved to be successful in the treatment of transplant KS [59, 147, 148]. Rapamycin has also been tested successfully in patients with AIDS-KS [149].

8.4.2.2 Preclinical Studies

Ongoing preclinical research continues to explore cellular kinases as druggable targets to treat KSHV disease.

In addition to the PI3K/AKT/mTOR pathway (see Section 8.4.2.1 and below in this section), the ERK/MAPK pathway and its downstream effector kinases have turned out to be required for KSHV infection as well as reactivation [150, 151]. Several inhibitors targeting this pathway, including clinically approved drugs such as the MEK1/2 inhibitor trametinib,

Table 8.2Compounds or biologicals directed at cellular targets and with activity against KSHV orKSHV disease.

	Compound	Target	Structure	References
Clinical data available	Imatinib	c-abl, c-kit, PDGFR	HN CH ₃ HN CH ₃ O CH ₃	[138, 139]
	Sorafenib	VEGFR, PDGFR, c-kit	$\overset{CI}{\underset{F}{\overset{O}}}_{F} \overset{O}{\underset{F}{\overset{O}}}_{F} \overset{O}{\underset{H}{\overset{O}}}_{H} \overset{O}}{H}} \overset{O}{\underset{H}{\overset{O}}}_{H} \overset{O}}{H} \overset{O}{H}}_{H} \overset{O}{H}} \overset{O}{H}} \overset{O}{H}} \overset{O}{H} \overset{O}}{H} \overset{O}{H}} \overset{O}{H}} \overset{O}{H}} \overset{O}{H} \overset{O}{H}} \overset{O}{H}} \overset{O}{H}} \overset{O}{H}} \overset{O}{H} \overset{O}{H}} \overset{O}{H}} \overset{O}{H}} \overset{O}{H} \overset{O}{H}} \overset{O}{H} \overset{O}{H}} \overset{O}{H}} \overset{O}{H}} \overset{O}{H}} \overset{O}{H} \overset{O}{H}} \overset{O}{H}} \overset{O}{H}} \overset{O}{H} \overset{O}{H} \overset{O}{H}} \overset{O}{H}} \overset{O}{H}} \overset{O}{H} \overset{O}{H}} \overset{O}{H}} \overset{O}{H} \overset{O}{H} \overset{O}{H}} \overset{O}{H}} \overset{O}{H} \overset{O}{H}} \overset{O}{H} \overset{O}{H}} \overset{O}{H} \overset{O}{H} \overset{O}{H}} \overset{O}{H} \overset{O}{H} \overset{O}{H}} \overset{O}{H} \mathsf{$	[140, 141]
	Rapamycin/ Sirolimus	mTOR		[149, 166–170]
	Bortezomib	Proteasome		[184, 185, 187–189]
	Rituximab	CD20	Monoclonal antibody	
	Bevacizumab	VEGF	Monoclonal antibody	[173–175]
	Siltuximab	IL6 R	Monoclonal antibody	[176, 177]
	Tocilizumab	IL6 R	Monoclonal antibody	[176, 177]
Only preclinical experimental data available	Dasatinib	BCR-Abl, Src, Lck, Lyn, Yes, Fyn, c-Kit, EphA2, PDGFRβ	$\overbrace{CI}^{CH_3} \underset{N}{H_1} \overbrace{S}^{N} \underset{N}{\overset{NH}{\longrightarrow}} \underset{N}{\overset{N}{\overset{NH}{\longrightarrow}} \underset{N}{\overset{N}{\overset{NH}{\longrightarrow}} \underset{N}{\overset{N}{\overset{NH}{\longrightarrow}} \underset{N}{\overset{N}{\overset{NH}{\longrightarrow}} \underset{N}{\overset{N}{\overset{N}{\overset{N}{\overset{NH}{\longrightarrow}}} \underset{N}{\overset{N}{\overset{N}{\overset{N}{\overset{N}{\overset{N}{\overset{N}{\overset{N}{$	[97]
	Crizotinib (PF-2341066)	c-Met, ALK		[164]
			F	

C	Compound	Target	Structure	References
Γ	Frametinib	MEK		[97]
E	BI-D1870	RSK-1/2		[156]
τ	JNC3810A	Tyro 3		[165]
N S	MLN0128/ Sapanisertib	mTOR		[171]
C	Chloroquine	mTOR, p38-MAPK		[172]
1 7 1 1 1 0	17-DMAG/ Alvespimycin 17-Dimethyla ninoethylamino- 17-demethoxygel lanamycin	HSP90		[178–180]
F 8 5 5 (7 F 7 F 7	PU-H71 3-[(6-Iodo-1,3- penzodioxol-5-yl) sulfanyl]-9-[3- propan-2-ylamino) propyl] purin-6-amine	HSP90	N_{H_2} N_{H	[178–180]

Table 8.2 (Continued)

(Continued)

Table 8.2 (Continued)

Compound	Target	Structure	References
AUY-922/ Luminespib 5-(2,4-Dihydroxy- 5-isopropyl-phenyl)- <i>N</i> -ethyl-4-[4- (morpholinomethyl) phenyl] isoxazole-3- carboxamide	HSP90		[178–180]
VER-155008	HSP70	$\overset{NC}{\leftarrow}\overset{O}{\leftarrow}\mathsf$	[182]
MLN4924	NEDDylation	HO O O S NH ₂	[183]
Novobiocin	Topoisom- erase II	H ₂ N _V CH ₃ CH	[195] 3
(+)-Rutamarin	Topoisom- erase II		[195]
PX-478	HIF-1α		[205]
Tenovin-6	SIRT1		[206]
C10	APE1		[207]



Table 8.2 (Continued)

but also experimental compounds such as sangivamycin and capsaicin have been shown to inhibit KSHV reactivation or to induce apoptosis in PEL cells [97, 152, 153]. Several KSHV proteins such as the viral chemokine receptor homologue (vGCR) [154], the ORF36encoded vPK [155], and the viral tegument protein encoded by ORF45 [156, 157] have been shown to induce the MEK/ERK pathway or downstream RSK1/2 kinases and with it viral replication [158]. BI-D1870, a RSK1/2 inhibitor, suppresses KSHV lytic gene expression and virus production [156]. The ORF45 tegument protein interacts directly with RSK kinases and this protein complex also contains activated ERK [157, 159] and is required to sustain late viral gene expression and virus production [156]. The pORF45/RSK complex mediates the phosphorylation of the eukaryotic translation initiation factor eIF4B; the pORF45/RSK-mediated phosphorylation of eIF4B is resistant to rapamycin (i.e. not dependent on mTOR) and U1026 (i.e. not dependent on MEK) [160]. An attempt was therefore made to target the pOR45/RSK interaction directly, resulting in the development of a small pORF45-derived peptide that inhibits the formation of this complex as well as viral lytic gene expression and virus production [161].

The clinically approved tyrosine kinase inhibitors dasatinib, ponatinib and bosutinib are potent inhibitors of the protein kinase activity of KSHV TK and also of KSHV reactivation, but these two effects may not be directly linked, as dasatinib, ponatinib, and bosutinib target a range of cellular tyrosine kinases that are likely involved in KSHV reactivation [97]. Dasatinib reduced the in vivo tumor growth of KSHV-infected endothelial cells in a xenograft model to the same extent as imatinib [97]. Since these three compounds also efficiently inhibit the ability of KSHV TK to autophosphorylate tyrosine residues in its

aminoterminal end, they may be able to exert an additional inhibitory effect on KSHV replication *in vivo*, since the TK homologue of a murine gammaherpesvirus, MHV68, is required for efficient replication *in vivo* while being dispensable in tissue culture [97, 162, 163]. Ibrutinib, an inhibitor of Bruton's tyrosine kinase, also inhibited KSHV reactivation in B cells [97].

In the context of PEL, crizotinib (PF-2341066), a small molecule inhibitor of ALK and c-Met, the receptor for hepatocyte growth factor (HGF), induces PEL cell apoptosis through cell cycle arrest and DNA damage, and suppresses tumor progression in a mouse xenograft model [164]. Crizotinib also inhibited KSHV replication in KSHV-infected BJAB cells, a B-cell line [97]. The receptor tyrosine kinase Tyro 3, a member of the Tyro3/Axl/Mer TK (TAM) family that is known to promote proliferation, survival, and chemoresistance in several solid cancers, is specifically upregulated in PEL cells and essential for their survival [165]. A small molecule inhibitor, UNC3810A, inhibits PEL cell proliferation in tissue culture at nanomolar concentrations and in a mouse xenograft model [165].

As described in the preceding section, the PI3K/AKT/mTOR pathway plays an important role in the KSHV life cycle. In cultured primary endothelial cells, KSHV induces mTORC1 activation and mTOR inhibitors antagonize KSHV-induced cell proliferation and differentiation [166–168]. In tissue culture and in mouse xenograft models, mTOR inhibitors reduce the growth of PEL cell lines and the production of Il6 and IL8, in particular when combined with an AKT inhibitor [169–171]. An ATP-competitive inhibitor of mTOR, MLN0128 (Sapanisertib), which inhibits both the mTORC1 and mTORC2 complexes (unlike the allosteric inhibitor rapamycin, which only inhibits mTORC1), induces apoptosis in PEL cells and reduces growth of PEL in a xenograft model at nanomolar IC50 concentrations and is still effective against doxorubicin- or rapamycin-resistant PEL cell clones [171]. The antimalarial compound chloroquine inhibits KSHV lytic replication by suppressing the activation of mTOR and p38-MAPK signaling pathways [172].

8.4.3 Other Established and Experimental Therapies Targeting KSHVinfected Cells (Table 8.2)

8.4.3.1 Interfering with Cellular Cytokines-Clinical Experience

Vascular endothelial growth factor (VEGF) plays an important role in the pathogenesis of all three KSHV-associated neoplasms (reviewed in [19–21]). A phase 2 clinical trial of an antibody to VEGF, bevacizumab, in patients with AIDS-KS reported an overall response rate (complete and partial responses) of 31% [173]. This response rate increased, when bevacizumab was used in combination with liposomal daunorubicin, but this study did not resolve if bevacizumab provided an additional benefit over and above this standard chemotherapy treatment for AIDS-KS [174]. When injected into oropharyngeal KS lesions in addition to standard ART, bevacizumab did not increase the response rates in a small phase 2 trial [175].

Tocilizumab and siltuximab, humanized antibodies against the IL6 receptor, have been found to be effective in idiopathic (i.e. KSHV-negative) as well as KSHV-associated MCD [176, 177].

8.4.3.2 Preclinical Data on Other Cellular Targets

The cellular chaperone heat shock protein 90 (HSP90) has been shown to be involved in regulating the stability of the KSHV latency proteins LANA and vFLIP, as well as the vFLIP-interacting protein NEMO/IKK γ [178–181]. Small molecule HSP90 inhibitors such as 17-DMAG, PU-H71, AUY-922, and geldanamycin induce degradation of LANA and vFLIP, inhibit vFLIP-dependent activation of NFkB, and as a consequence induce cell death in PEL cell lines and inhibit the proliferation of KSHV-infected endothelial cell lines in vitro and in a mouse xenograft model [178–181].

Several isoforms of the HSP70 chaperone family have been reported to be involved in the formation of KSHV nuclear replication and transcription compartments (RTCs) during the early stages of viral DNA replication and a small molecule HSP70 inhibitor, VER-155008, antagonized KSHV RTC formation, and the relocalization of the RNA polymerase II to the viral genome [182]. Similarly, recruitment of the viral pre-replication complex consisting of RTA and the viral core replication proteins to the viral lytic origin of replication (ori lyt) was inhibited by MLN 4924, an inhibitor of NEDDylation, suggesting that NEDDylation is important for the formation of KSHV nuclear replication compartments [183]. MLN 4924 also inhibits NFkB activation in PEL cells and thereby induces apoptosis in PEL cells [183].

The proteasome inhibitor bortezomib has been shown to induce the reactivation of the KSHV lytic cycle and/or induce apoptosis in PEL cell lines in tissue culture [184–186] and of PEL cells in a xenograft model [187]. Bortezomib also increased the KSHV and EBV viral load in HIV-infected patients with KSHV- or EBV-associated lymphoma [188]. In a pilot trial, bortezomib induced a partial regression of KS lesions in 60% of patients with AIDS-associated KS [189].

The notch signaling pathway has also been implicated in the KSHV-induced lymphatic reprogramming and the endothelial to mesenchymal transition of KSHV-infected endothelial cells [190–192]. The γ -secretase inhibitors GSI and LY-411,575, which block the activation of the notch pathway, induced apoptosis in primary cells cultivated from KS lesions and in a KSHV-infected epithelial cell line [193]. However, to what extent these effects of γ -secretase inhibitors were specific to KSHV-infected cells remains to be clarified.

Cellular topoisomerases are recruited to the lytic replication origin in the KSHV genome and are required for lytic replication [194]. Several topoisomerase I and II inhibitors can inhibit lytic KSHV replication [195]; among these, two topoisomerase II inhibitors showed the best selectivity: novobiocin was reported to have a selectivity index around 30 [195], while a novel topoisomerase II inhibitor, (+)-rutamarin, inhibited KSHV productive replication in PEL cell lines with an IC50 concentration of around 1 μ M and a selectivity index of 84 [196].

Additional cellular targets have been validated in tissue culture and small molecule inhibitors against these cellular proteins have been reported. These include HIF1 α (hypoxia-inducible factor 1 alpha), which plays an important role in the KSHV life cycle by activating the expression of KSHV RTA, the main activator of the lytic replication cycle as well as other lytic viral genes; HIF1 α thereby mediates the induction of the KSHV lytic replication cycle by hypoxia [197, 198]. KSHV LANA binds to and stabilizes HIF1 α [199, 200], and at least two viral lytic proteins, the chemokine receptor homologue vGCR (ORF74) and the interferon regulatory factor homologue vIRF3 (ORF K10.5),

increase HIF1 α expression or stabilize the HIF1 α protein to thereby induce VEGF production [201, 202]. These viral proteins thus contribute to the activation of hypoxiainducible factors in, and the metabolic reprogramming of latently infected endothelial cells [203, 204]. HIF1 α has also been shown to play a role in maintaining an optimal metabolic state for the growth of PEL cells [205]. Reflecting the importance of HIF1 α in the KSHV life cycle and the metabolic state of KSHV-infected cells, a small molecule inhibitor of HIF1 α , PX-478, was shown to inhibit the growth of PEL cells in tissue culture at low micromolar concentrations [205].

Other metabolic targets include the metabolic sensor SIRT1, which was also found to be essential for PEL cell survival [206]. The small molecule SIRT1 inhibitor tenovin-6 was shown to inhibit PEL cell growth in tissue culture and to enhance the survival time of mice xenografted with a PEL cell line [206]. Furthermore, the multifunctional APE1 protein, which regulates the redox state of certain transcription factors and thereby their transition from an inactive to an active state, has been shown to regulate KSHV lytic replication and a small molecule APE1 inhibitor (C10) inhibits KSHV reactivation [207]. A heme oxygenase 1 (HO-1) inhibitor (SnPP) induces cell death in KSHV-infected endothelial cells and suppresses KSHV-mediated tumorigenesis in a KS nude mouse model employing KSHV-infected immortalized human endothelial cells [208].

A CRISPR-Cas9 screen of KSHV-transformed mesenchymal embryonic stem cells yielded XPO1/Crm1, a nuclear export factor, as a critical player in the proliferation of KSHV-transformed cells and a small molecule inhibitor of XPO1, KPT-8602, inhibits the proliferation of KSHV-infected mesenchymal embryonic stem cells and activates a p53-dependent cell cycle block [209].

A screen of a library of 1280 FDA-approved drugs yielded three compounds targeting histamine receptors or signaling [210]. Further analysis indicates that downstream MAPK and PI3K signaling pathways are involved in the activation of the KSHV lytic replication cycle by histamine receptors [210].

Attempts have also been made to target the interaction of the KSHV gH/gL glycoprotein complex with its receptor, EphA2. An inhibitor of the interaction of EphA2 with its physiological ligand, ephrin A5, was shown to inhibit KSHV infection of blood and lymphatic endothelial cells at two- to three-digit micromolar concentrations [211].

8.5 New Antiviral Strategies Against KSHV in Preclinical Development

The picture emerging from the experimental and clinical evidence summarized in the preceding sections is that existing antiviral drugs are mostly of moderate efficacy when used to treat KSHV-associated malignancies. Particularly in the case of KS, in which viral gene expression is mostly restricted and most tumor cells are latently infected, currently available herpesviral DNA polymerase inhibitors are largely ineffective. They may occasionally be beneficial in MCD, a KSHV-associated disease in which KSHV productive (lytic) replication plays a greater role. There is therefore a clinical need to explore new approaches to develop inhibitors against already validated as well as new targets in the KSHV life cycle.

8.5.1 New Approaches to Target Productive (Lytic) KSHV DNA Replication and/or Packaging (Table 8.1)

As is the case for other herpesviruses, the KSHV DNA polymerase, encoded by ORF9, works in concert with the processivity factor, encoded by ORF59, which promotes the elongation of newly synthesized viral DNA strands. A screen of the NCI Diversity Set library using an in vitro DNA synthesis assay based on the ORF9/ORF59 complex yielded several inhibitors with in vitro activity, of which one, NSC373989, also inhibited KSHV reactivation in a lytically induced PEL cell line (BCBL1); this observation indicates that it may be possible to develop inhibitors against the KSHV viral DNA polymerase complex with a different mode of action than the conventional nucleoside-based competitive DNA polymerase inhibitors [99].

Another approach is based on the observation that all herpesviruses contain at least two proteins with structural folds that are similar to RNAse H-like nucleotidyltransferases. These are the single-strand DNA (ssDNA) binding proteins required for viral DNA replication (ICP8/pUL29 in HSV; pUL57 in HCMV; pORF6 in KSHV) and the large terminase subunit, which contains the nuclease activity required for the trimming of the newly replicated viral DNA to the correct length and its packaging into preformed capsids (pUL15 in HSV; pUL89 in HCMV; the C-terminal domain of pORF29 in KSHV). The catalytic site of the HIV integrase is also contained within an RNAse H-like fold and HIV integrase inhibitors were therefore tested for their ability to inhibit herpesviral replication. One compound, XZ45, was found to inhibit HSV, HCMV, and KSHV replication at low micromolar concentrations [100]. This study did not resolve whether XZ45 targets primarily the KSHV ssDNA binding protein pORF6 or the KSHV large terminase subunit pORF29 [100].

In keeping with the structural similarity of the c-terminal domain of KSHV pORF29 with RNAse H-like nucleotidyltransferases, the HIV integration inhibitors raltegravir and dolutegravir inhibit the DNA nuclease activity of pORF29 in in vitro assays at 4 and <0.1 μ M, respectively [101]. Derivatives of α -hydroxytropolone show a similar inhibitory activity *in vitro* and also inhibit KSHV lytic replication in tissue culture [101].

As already mentioned in Section 8.4.3.1, several cellular HSP70 isoforms are involved in the formation of KSHV nuclear RTCs and a small molecule HSP70 inhibitor, VER-155008, antagonizes RTC formation and viral gene expression [182]. This observation suggests that there may also be "druggable" cellular targets that could be harnessed to inhibit KSHV productive (lytic) replication.

8.5.2 Targeting KSHV Capsid Assembly (Table 8.1)

The KSHV protease is part of the pORF17 scaffold/protease polyprotein and a serine protease [212], which has been explored as a drug target. The KSHV protease mediates the cleavage of the pORF17 polyprotein into the protease and the scaffold (assembly) protein, which fills preformed nuclear herpesviral capsids prior to their loading with viral DNA. "Maturation cleavage" of the scaffold protein is also mediated by the herpesviral protease and is required for its release from pre-formed capsids and their loading with viral DNA [212]. The KSHV protease forms a dimer, which can be disrupted by a small α -helical peptide [213] or small molecules [102, 214]. One such compound inhibited KSHV production in tissue culture with IC50 values in the low two-digit micromolar range [103].

Although herpesviral proteases are serine proteases, inhibitors of the HIV aspartyl protease have also been explored for their activity against KSHV replication. Of several HIV protease inhibitors tested, nelfinavir inhibits KSHV reactivation in tissue culture with an IC50 of about 5µM and also inhibits HSV and HCMV replication [104]. However, the activity of nelfinavir against KSHV may not be linked to an inhibition of the KSHV protease and could also be explained by off-target effects of nelfinavir, such as its inhibition of the PI3K/ AKT or STAT3 pathways [104]. There is also clinical evidence that HIV protease inhibitors administered as part of an ART may be beneficial against AIDS-KS, but this is likely to be linked to their activity against HIV and an improved immune response against KSHV in individuals undergoing efficient HIV therapy [215, 216]. However, an ART regimen containing HIV protease inhibitors was found to be associated with lower KSHV shedding rates in HIV-infected patients and this effect was independent of CD4 count and HIV viral load [105]. Also, a favorable outcome (partial/complete remission or stabilization of tumors) was also observed in 16 out of 26 HIV-negative classic KS patients treated with indinavir [217]. Whether these observations point to a direct effect of at least some HIV protease inhibitors against KSHV or can be explained by off-target effects, or possibly a direct anti-angiogenic effect of HIV protease inhibitors [217], is uncertain.

8.5.3 Targeted Immunotoxins (Table 8.1)

In an attempt to target KSHV proteins expressed on the surface of infected cells with monoclonal antibodies conjugated to toxic moieties, such targeted immunotoxins have been generated for the K8.1 and gH glycoproteins, after coupling single-chain variable region fragments of monoclonal antibodies to the respective specificity to *Pseudomonas aeruginosa* exotoxin A. These targeted immunotoxins were capable of killing KSHV-infected cells undergoing productive replication in tissue culture and synergized in their killing potential with ganciclovir [106, 107]. However, their effect would be limited to the late stage of productive (lytic) KSHV reactivation, which is likely to be encountered in only a small proportion of KSHV-infected tumor cells in vivo.

8.5.4 Targeting the Latency Phase of the KSHV Life Cycle

The—overall only moderate—efficacy of antiviral treatments directed against the lytic (productive) phase of the KSHV life cycle raises the question if targeting earlier stages of the productive replication cycle or latent KSHV persistence might represent a better approach to treat KSHV-associated neoplasia. Experimental evidence suggests that silencing the expression of either of the three latent viral protein-encoding genes, ORF73 (LANA), ORF72 (cyclin homologue, vcyc), ORF71/K13 (vFLIP), in PEL cell lines with the help of shRNA inhibits their growth and/or induces cell death [218, 219]. Likewise, deleting or silencing the expression of ORFs K1, K15, and vGCR interferes with virus reactivation at an early stage in the productive replication cycle [125, 220, 221]. Therefore, targeting any or several of these viral proteins therapeutically *in vivo* might provide a better route to treatment success. In the following sections, the role of the above candidate targets in the viral life cycle and ongoing attempts to exploit them therapeutically are briefly summarized.

8.5.4.1 LANA (Latency-associated Nuclear Antigen) (Table 8.1)

LANA is expressed in all latently infected tumor cells (Figure 8.1). It is essential for the replication of latent viral episomes and their distribution to dividing daughter cells during mitosis (reviewed in [222]). Its C-terminal DNA binding domain (DBD) binds to the latent KSHV replication origin, which is located in the terminal repeat subunits of the viral genome (Figure 8.2) [223–225]. The structure of the C-terminal DBD has been solved, alone and in complex with an oligonucleotide representing one of the LANA-binding sites in the latent origin of replication [226–228]. A short amino acid sequence in its N-terminal domain is responsible for attaching the complex of LANA and viral DNA to histones H2A and H2B on mitotic chromosomes (Figure 8.2). Mutation of either this N-terminal histone-binding motif or the DNA-binding motifs in the DBD compromises latent replication [225, 226, 229, 230]. Silencing LANA in PEL cells with the help of shRNA induces cell death [219]. Furthermore, decreasing the expression of LANA in PEL cells by treatment with glycyrrhizic acid induces apoptosis by restoring the function of p53, which is inhibited by LANA [231], and HSP90 inhibitors can induce LANA degradation and cell death in PEL cells [178] (see Section 8.4.3.2). Together, this experimental evidence suggests that LANA could represent a promising therapeutic target.

Recently, Kirsch and colleagues used the structure of the LANA DBD in complex with the latent replication origin [227] as a starting point for a fragment-based drug discovery project that yielded a first generation of small molecule LANA inhibitors [108, 109]. The



Figure 8.2 Schematic diagram of the KSHV genome and its latent origin of replication. (a) Diagram of the linear KSHV genome. The long unique coding region (LUR; 138 kbp) is shown as a dark solid line and flanked by multiple terminal repeat (TR) subunits of 801 bp each. The drawing is not to scale. Boxes represent selected KSHV genes discussed in the text and their transcriptional orientation. Numbers inside boxes denote the ORF number, the name of the corresponding viral protein is located above each box. ssDna BP, single-stranded DNA-binding protein; DNA Pol, DNA polymerase; TK, thymidine kinase; Terminase, large terminase subunit containing nuclease activity; vPK, viral serine/threonine kinase; PAF, polymerase-associated factor. Genes marked in yellow are expressed during latency, genes marked in blue show significant expression in most KS biopsies (see text), and genes marked in green are only expressed in KSHV-infected cells undergoing productive (lytic) replication. (b) A single TR subunit is enlarged to show its organization. A part of the TR subunit is chromatinized and arranged in nucleosomes, while the three LANA-binding sites (LBS) are free of histones. The position of the latent origin or replication (ori) is indicated. (c) ORF73 encodes the major latency protein LANA, which consists of an aminoterminal domain that interacts with histones H2A/B, an internal repeat region (IR), and a carboxyterminal region. The latter contains the DNA-binding domain (DBD), which binds to three specific DNA motifs (LANAbinding sites, LBS) in each TR subunit.

most promising compounds inhibited the interaction of LANA with the latent viral origin in fluorescence polarization and in electrophoretic mobility shift assays with IC50 values in the single- to double-digit micromolar range [109]. Ongoing work aims to improve the potency of these first compounds and to test them for antiviral activity in tissue culture.

8.5.4.2 vFLIP (Viral FLICE-inhibitory Protein) (Table 8.1)

Another latent protein, the KSHV homologue of cellular Fas-associated death domain (FADD) interleukin 1 β -converting enzyme (FLICE)-inhibitory proteins (FLIP) inhibits Fas-induced apoptosis and is a potent NFkB activator [232–234]. Transgenic mice with vFLIP expression targeted to B cells develop B-cell lymphomas [235], while transgenic expression in endothelial cells results in elongated fusiform endothelial cells reminiscent of KS spindle cells and an inflammatory phenotype characterized by increased expression of IL6 and IL10, similar to KICS (see Section 8.3.5; [236]). The activation of the NFkB pathway by vFLIP involves its ability to interact with IKK γ /NEMO [181, 218] and the structure of a fragment of the coiled-coil domains of IKK γ /NEMO in complex with vFLIP has been solved [237]. Silencing vFLIP expression in PEL cells lines [218], or blocking the NFkB pathway with small molecule NFkB inhibitors [233, 234], induces cell death, suggesting that vFLIP-induced activation of NFkB would be a worthwhile therapeutic target. In addition to targeting other proteins in KSHV-infected cells, HSP90 inhibitors also induce vFLIP degradation and inhibit vFLIP/IKK γ -dependent NFkB activation (see Section 8.4.3.2).

Recently, two groups have reported peptide-based small molecules capable of antagonizing the vFLIP-IKK γ interaction and promoting apoptosis in PEL cell lines. Briggs and colleagues designed a conformationally constrained, stapled IKK γ peptide derived from the IKK γ -vFLIP interaction site, which could interfere with the binding of IKK γ to vFLIP and enhance apoptosis in PEL cell lines [110]. Sadek and colleagues developed a tertiary protein structure mimic of the vFLIP-interaction site in the IKK γ /NEMO helix and showed its ability to induce cell death in PEL cell lines and to delay tumor growth in a PEL xenograft mouse model [111].

8.5.4.2.1 Outlook

Although a wide range of drugs and therapies, ranging from conventional cancer chemotherapy to antiviral drugs, have been tried with some success to treat KSHV-associated neoplasia, none of them are entirely satisfactory, because of side effects or moderate efficacy. There is therefore room for the development of new therapeutic approaches. Since KSHV-associated malignancies require the continuous presence of the KSHV genome and at least latent viral gene expression in the tumor cells, targeting appropriate KSHV proteins and/or their ability to activate cellular processes involved in pathogenesis could in theory provide a more specific and effective therapeutic approach than currently used treatment strategies. KSHV-associated diseases are mostly confined to people living with HIV and transplant recipients in countries with a higher KSHV seroprevalence. Even if new specific antiviral drugs, aimed at new targets and with good activity against KSHV, could be discovered, their preclinical and clinical development will therefore present an additional challenge.

References

- 1 Chang, Y., Cesarman, E., Pessin, M.S. et al. (1994). Identification of herpesvirus-like DNA sequences in AIDS-associated Kaposi's sarcoma. *Science* 266: 1865–1869.
- **2** Cesarman, E., Chang, Y., Moore, P.S. et al. (1995). Kaposi's sarcoma-associated herpesvirus-like DNA sequences in AIDS-related body-cavity-based lymphomas. *N. Engl. J. Med.* 332: 1186–1191.
- **3** Soulier, J., Grollet, L., Oksenhendler, E. et al. (1995). Kaposi's sarcoma-associated herpesvirus-like DNA sequences in multicentric Castleman's disease. *Blood* 86: 1276–1280.
- **4** Matsushima, A.Y., Strauchen, J.A., Lee, G. et al. (1999). Posttransplantation plasmacytic proliferations related to Kaposi's sarcoma-associated herpesvirus. *Am. J. Surg. Pathol.* 23: 1393–1400.
- Kapelushnik, J., Ariad, S., Benharroch, D. et al. (2001). Post renal transplantation human herpesvirus 8-associated lymphoproliferative disorder and Kaposi's sarcoma. *Br. J. Haematol.* 113: 425–428.
- **6** Marcelin, A.G., Roque-Afonso, A.M., Hurtova, M. et al. (2004). Fatal disseminated Kaposi's sarcoma following human herpesvirus 8 primary infections in liver-transplant recipients. *Liver Transpl.* 10: 295–300.
- 7 Uldrick, T.S., Wang, V., O'Mahony, D. et al. (2010). An interleukin-6-related systemic inflammatory syndrome in patients co-infected with Kaposi sarcoma-associated herpesvirus and HIV but without multicentric castleman disease. *Clin. Infect. Dis.* 51: 350–358.
- **8** Polizzotto, M.N., Uldrick, T.S., Hu, D., and Yarchoan, R. (2012). Clinical manifestations of Kaposi sarcoma herpesvirus lytic activation: multicentric castleman disease (KSHV-MCD) and the KSHV inflammatory cytokine syndrome. *Front. Microbiol.* 3: 73.
- **9** Polizzotto, M.N., Uldrick, T.S., Wyvill, K.M. et al. (2016). Clinical features and outcomes of patients with symptomatic Kaposi sarcoma herpesvirus (KSHV)-associated inflammation: prospective characterization of KSHV inflammatory cytokine syndrome (KICS). *Clin. Infect. Dis.* 62: 730–738.
- **10** Luppi, M., Barozzi, P., Schulz, T.F. et al. (2000). Bone marrow failure associated with human herpesvirus 8 infection after transplantation. *N. Engl. J. Med.* 343: 1378–1385.
- **11** Luppi, M., Barozzi, P., Rasini, V. et al. (2002). Severe pancytopenia and hemophagocytosis after HHV-8 primary infection in a renal transplant patient successfully treated with foscarnet. *Transplantation* 74: 131–132.
- **12** Karras, A., Thervet, E., Legendre, C., and Groupe Cooperatif de transplantation d'Ile de France (2004). Hemophagocytic syndrome in renal transplant recipients: report of 17 cases and review of literature. *Transplantation* 77: 238–243.
- **13** Thaunat, O., Mamzer-Bruneel, M.F., Agbalika, F. et al. (2006). Severe human herpesvirus-8 primary infection in a renal transplant patient successfully treated with anti-CD20 monoclonal antibody. *Blood* 107: 3009–3010.
- **14** Luppi, M., Barozzi, P., Schulz, T.F. et al. (2000). Nonmalignant disease associated with human herpesvirus 8 reactivation in patients who have undergone autologous peripheral blood stem cell transplantation. *Blood* 96: 2355–2357.
- **15** Pietrosi, G., Vizzini, G., Pipitone, L. et al. (2011). Primary and reactivated HHV8 infection and disease after liver transplantation: a prospective study. *Am. J. Transplant.* 11: 2715–2723.

- 214 8 Kaposi's Sarcoma-associated Herpesvirus-Antiviral Treatment
 - **16** Bouvard, V., Baan, R., Straif, K. et al. (2009). A review of human carcinogens—Part B: biological agents. *Lancet Oncol.* 10: 321–322.
 - 17 IARC (2012). Biological Agents: IARC Monographs on the Evaluation of Carcinogenic Risks to Humans Volume 100B. International Agency for Research on Cancer. ISBN 978 92 832 1319 2.
 - 18 Gramolelli, S., Weidner-Glunde, M., Abere, B. et al. (2015). Inhibiting the recruitment of PLCgamma1 to Kaposi's sarcoma herpesvirus K15 protein reduces the invasiveness and angiogenesis of infected endothelial cells. *PLoS Pathog.* 11: e1005105.
 - **19** Dittmer, D.P. and Damania, B. (2016). Kaposi sarcoma-associated herpesvirus: immunobiology, oncogenesis, and therapy. *J. Clin. Invest.* 126: 3165–3175.
 - **20** Mariggio, G., Koch, S., and Schulz, T.F. (2017). Kaposi sarcoma herpesvirus pathogenesis. *Phil. Trans. R. Soc. B* 372: 20160275. https://doi.org/10.1098/rstb.2016.0275.
 - **21** Cesarman, E., Damania, B., Krown, S.E. et al. (2019). Kaposi sarcoma. *Nat. Rev. Dis. Primers* 5: 9.
 - **22** Biggar, R.J., Whitby, D., Marshall, V. et al. (2000). Human herpesvirus 8 in Brazilian Amerindians: a hyperendemic population with a new subtype. *J. Infect. Dis.* 181: 1562–1568.
 - **23** Whitby, D., Marshall, V.A., Bagni, R.K. et al. (2004). Genotypic characterization of Kaposi's sarcoma-associated herpesvirus in asymptomatic infected subjects from isolated populations. *J. Gen. Virol.* 85: 155–163.
 - 24 Fu, B., Sun, F., Li, B. et al. (2009). Seroprevalence of Kaposi's sarcoma-associated herpesvirus and risk factors in Xinjiang, China. *J. Med. Virol.* 81: 1422–1431.
 - 25 Cassar O, Bassot S, Plancoulaine S, Quintana-Murci L, Harmant C, Gurtsevitch V, Senyuta NB, Yakovleva LS, de The G, Gessain A. 2010. Human herpesvirus 8, Southern Siberia. *Emerg. Infect. Dis.* 16:580–2.
 - **26** Davidovici, B., Karakis, I., Bourboulia, D. et al. (2001). Seroepidemiology and molecular epidemiology of Kaposi's sarcoma-associated herpesvirus among Jewish population groups in Israel. *J. Natl. Cancer Inst.* 93: 194–202.
 - 27 Cattani, P., Capuano, M., Cerimele, F. et al. (1999). Human herpesvirus 8 seroprevalence and evaluation of nonsexual transmission routes by detection of DNA in clinical specimens from human immunodeficiency virus-seronegative patients from central and southern Italy, with and without Kaposi's sarcoma. *J. Clin. Microbiol.* 37: 1150–1153.
 - **28** Plancoulaine S, Abel L, van Beveren M, Tregouet DA, Joubert M, Tortevoye P, de The G, Gessain A. 2000. Human herpesvirus 8 transmission from mother to child and between siblings in an endemic population. *Lancet* 356:1062–5.
 - **29** Dedicoat, M., Newton, R., Alkharsah, K.R. et al. (2004). Mother-to-child transmission of human herpesvirus-8 in South Africa. *J. Infect. Dis.* 190: 1068–1075.
 - **30** Brayfield, B.P., Kankasa, C., West, J.T. et al. (2004). Distribution of Kaposi sarcomaassociated herpesvirus/human herpesvirus 8 in maternal saliva and breast milk in Zambia: implications for transmission. *J. Infect. Dis.* 189: 2260–2270.
 - **31** Mbulaiteye, S., Marshall, V., Bagni, R.K. et al. (2006). Molecular evidence for mother-tochild transmission of Kaposi sarcoma-associated herpesvirus in Uganda and K1 gene evolution within the host. *J. Infect. Dis.* 193: 1250–1257.

- 32 Mancuso, R., Brambilla, L., Agostini, S. et al. (2011). Intrafamiliar transmission of Kaposi's sarcoma-associated herpesvirus and seronegative infection in family members of classic Kaposi's sarcoma patients. J. Gen. Virol. 92: 744–751.
- **33** Borges, J.D., Souza, V.A., Giambartolomei, C. et al. (2012). Transmission of human herpesvirus type 8 infection within families in american indigenous populations from the Brazilian Amazon. *J. Infect. Dis.* 205: 1869–1876.
- **34** Kedes, D.H., Operskalski, E., Busch, M. et al. (1996). The seroepidemiology of human herpesvirus 8 (Kaposi's sarcoma-associated herpesvirus): distribution of infection in KS risk groups and evidence for sexual transmission. *Nat. Med.* 2: 918–924.
- **35** Simpson, G.R., Schulz, T.F., Whitby, D. et al. (1996). Prevalence of Kaposi's sarcoma associated herpesvirus infection measured by antibodies to recombinant capsid protein and latent immunofluorescence antigen. *Lancet* 348: 1133–1138.
- **36** Martin, J.N., Ganem, D.E., Osmond, D.H. et al. (1998). Sexual transmission and the natural history of human herpesvirus 8 infection. *N. Engl. J. Med.* 338: 948–954.
- **37** Dukers, N.H., Renwick, N., Prins, M. et al. (2000). Risk factors for human herpesvirus 8 seropositivity and seroconversion in a cohort of homosexual men. *Am. J. Epidemiol.* 151: 213–224.
- **38** Renwick, N., Halaby, T., Weverling, G.J. et al. (1998). Seroconversion for human herpesvirus 8 during HIV infection is highly predictive of Kaposi's sarcoma. *AIDS* 12: 2481–2488.
- **39** Nalwoga, A., Cose, S., Wakeham, K. et al. (2015). Association between malaria exposure and Kaposi's sarcoma-associated herpes virus seropositivity in Uganda. *Tropical Med. Int. Health* 20: 665–672.
- **40** Nalwoga, A., Webb, E.L., Chihota, B. et al. (2019). Kaposi's sarcoma-associated herpesvirus seropositivity is associated with parasite infections in Ugandan fishing communities on Lake Victoria islands. *PLoS Negl. Trop. Dis.* 13: e0007776.
- **41** Wakeham, K., Webb, E.L., Sebina, I. et al. (2011). Parasite infection is associated with Kaposi's sarcoma associated herpesvirus (KSHV) in Ugandan women. *Infect Agent Cancer* 6: 15.
- **42** Wakeham, K., Webb, E.L., Sebina, I. et al. (2013). Risk factors for seropositivity to Kaposi sarcoma-associated herpesvirus among children in Uganda. *J. Acquir. Immune Defic. Syndr.* 63: 228–233.
- **43** Whitby, D., Marshall, V.A., Bagni, R.K. et al. (2007). Reactivation of Kaposi's sarcomaassociated herpesvirus by natural products from Kaposi's sarcoma endemic regions. *Int. J. Cancer* 120: 321–328.
- **44** Grulich, A.E., van Leeuwen, M.T., Falster, M.O., and Vajdic, C.M. (2007). Incidence of cancers in people with HIV/AIDS compared with immunosuppressed transplant recipients: a meta-analysis. *Lancet* 370: 59–67.
- **45** Lebbe, C., Legendre, C., and Frances, C. (2008). Kaposi sarcoma in transplantation. *Transplant Rev. (Orlando)* 22: 252–261.
- **46** Farge, D. (1993). Kaposi's sarcoma in organ transplant recipients. The Collaborative Transplantation Research Group of Ile de France. *Eur. J. Med.* 2: 339–343.
- **47** Frezza, E.E., Fung, J.J., and van Thiel, D.H. (1997). Non-lymphoid cancer after liver transplantation. *Hepato-Gastroenterology* 44: 1172–1181.

- **48** Garcia-Astudillo, L.A. and Leyva-Cobian, F. (2006). Human herpesvirus-8 infection and Kaposi's sarcoma after liver and kidney transplantation in different geographical areas of Spain. *Transpl. Immunol.* 17: 65–69.
- **49** Margolius, L., Stein, M., Spencer, D., and Bezwoda, W.R. (1994). Kaposi's sarcoma in renal transplant recipients. Experience at Johannesburg Hospital, 1966–1989. *S. Afr. Med. J.* 84: 16–17.
- **50** Mbulaiteye, S.M. and Engels, E.A. (2006). Kaposi's sarcoma risk among transplant recipients in the United States (1993–2003). *Int. J. Cancer* 119: 2685–2691.
- **51** Qunibi, W.Y., Barri, Y., Alfurayh, O. et al. (1993). Kaposi's sarcoma in renal transplant recipients: a report on 26 cases from a single institution. *Transplant. Proc.* 25: 1402–1405.
- **52** Serraino, D., Piselli, P., Angeletti, C. et al. (2005). Risk of Kaposi's sarcoma and of other cancers in Italian renal transplant patients. *Br. J. Cancer* 92: 572–575.
- **53** Shmueli, D., Shapira, Z., Yussim, A. et al. (1989). The incidence of Kaposi sarcoma in renal transplant patients and its relation to immunosuppression. *Transplant. Proc.* 21: 3209–3210.
- 54 Tessari, G., Naldi, L., Boschiero, L. et al. (2006). Incidence and clinical predictors of Kaposi's sarcoma among 1721 Italian solid organ transplant recipients: a multicenter study. *Eur. J. Dermatol.* 16: 553–557.
- 55 Cattani, P., Capuano, M., Graffeo, R. et al. (2001). Kaposi's sarcoma associated with previous human herpesvirus 8 infection in kidney transplant recipients. *J. Clin. Microbiol.* 39: 506–508.
- **56** Frances, C., Mouquet, C., Marcelin, A.G. et al. (2000). Outcome of kidney transplant recipients with previous human herpesvirus-8 infection. *Transplantation* 69: 1776–1779.
- 57 Andreoni, M., Goletti, D., Pezzotti, P. et al. (2001). Prevalence, incidence and correlates of HHV-8/KSHV infection and Kaposi's sarcoma in renal and liver transplant recipients. *J. Infect.* 43: 195–199.
- **58** Aseni, P., Vertemati, M., Minola, E. et al. (2001). Kaposi's sarcoma in liver transplant recipients: morphological and clinical description. *Liver Transpl.* 7: 816–823.
- **59** Riva, G., Luppi, M., Barozzi, P. et al. (2012). How I treat HHV8/KSHV-related diseases in posttransplant patients. *Blood* 120: 4150–4159.
- **60** Frances, C., Marcelin, A.G., Legendre, C. et al. (2009). The impact of preexisting or acquired Kaposi sarcoma herpesvirus infection in kidney transplant recipients on morbidity and survival. *Am. J. Transplant.* 9: 2580–2586.
- **61** Luppi, M., Barozzi, P., Santagostino, G. et al. (2000). Molecular evidence of organ-related transmission of Kaposi sarcoma-associated herpesvirus or human herpesvirus-8 in transplant patients. *Blood* 96: 3279–3281.
- **62** Barozzi, P., Luppi, M., Facchetti, F. et al. (2003). Post-transplant Kaposi sarcoma originates from the seeding of donor-derived progenitors. *Nat. Med.* 9: 554–561.
- **63** Wabinga, H.R., Nambooze, S., Amulen, P.M. et al. (2014). Trends in the incidence of cancer in Kampala, Uganda 1991–2010. *Int. J. Cancer* 135: 432–439.
- **64** Grulich, A.E., Li, Y., McDonald, A.M. et al. (2001). Decreasing rates of Kaposi's sarcoma and non-Hodgkin's lymphoma in the era of potent combination anti-retroviral therapy. *AIDS* 15: 629–633.
- **65** van Leeuwen, M.T., Vajdic, C.M., Middleton, M.G. et al. (2009). Continuing declines in some but not all HIV-associated cancers in Australia after widespread use of antiretroviral therapy. *AIDS* 23: 2183–2190.

- **66** Krown, S.E., Lee, J.Y., Dittmer, D.P., and Consortium, A.M. (eds.) (2008). More on HIV-associated Kaposi's sarcoma. *N. Engl. J. Med.* 358: 535–536. author reply 536.
- **67** Maurer, T., Ponte, M., and Leslie, K. (2007). HIV-associated Kaposi's sarcoma with a high CD4 count and a low viral load. *N. Engl. J. Med.* 357: 1352–1353.
- **68** Oksenhendler, E., Boulanger, E., Galicier, L. et al. (2002). High incidence of Kaposi sarcoma-associated herpesvirus-related non-Hodgkin lymphoma in patients with HIV infection and multicentric Castleman disease. *Blood* **99**: 2331–2336.
- **69** Komanduri, K.V., Luce, J.A., McGrath, M.S. et al. (1996). The natural history and molecular heterogeneity of HIV-associated primary malignant lymphomatous effusions. *J. Acquir. Immune Defic. Syndr. Hum. Retrovirol.* 13: 215–226.
- **70** Oksenhendler, E., Boutboul, D., and Galicier, L. (2019). Kaposi sarcoma-associated herpesvirus/human herpesvirus 8-associated lymphoproliferative disorders. *Blood* 133: 1186–1190.
- 71 Hernandez, J.L., Gomez-Roman, J., Ramos-Estebanez, C. et al. (2005). Human herpesvirus 8 and Epstein–Barr virus coinfection in localized Castleman disease during pregnancy. *Haematologica* (90 Suppl): ECR35.
- **72** Aoki, Y., Tosato, G., Fonville, T.W., and Pittaluga, S. (2001). Serum viral interleukin-6 in AIDS-related multicentric Castleman disease. *Blood* 97: 2526–2527.
- **73** Dupin, N., Diss, T.L., Kellam, P. et al. (2000). HHV-8 is associated with a plasmablastic variant of Castleman disease that is linked to HHV-8-positive plasmablastic lymphoma. *Blood* 95: 1406–1412.
- **74** Oksenhendler, E., Carcelain, G., Aoki, Y. et al. (2000). High levels of human herpesvirus 8 viral load, human interleukin-6, interleukin-10, and C reactive protein correlate with exacerbation of multicentric castleman disease in HIV-infected patients. *Blood* 96: 2069–2073.
- **75** Aoki, Y., Yarchoan, R., Wyvill, K. et al. (2001). Detection of viral interleukin-6 in Kaposi sarcoma-associated herpesvirus-linked disorders. *Blood* 97: 2173–2176.
- **76** Polizzotto, M.N., Uldrick, T.S., Wang, V. et al. (2013). Human and viral interleukin-6 and other cytokines in Kaposi sarcoma herpesvirus-associated multicentric Castleman disease. *Blood* 122: 4189–4198.
- **77** Katano, H., Sato, Y., Kurata, T. et al. (2000). Expression and localization of human herpesvirus 8-encoded proteins in primary effusion lymphoma, Kaposi's sarcoma, and multicentric Castleman's disease. *Virology* 269: 335–344.
- 78 Parravicini, C., Chandran, B., Corbellino, M. et al. (2000). Differential viral protein expression in Kaposi's sarcoma-associated herpesvirus-infected diseases: Kaposi's sarcoma, primary effusion lymphoma, and multicentric Castleman's disease. *Am. J. Pathol.* 156: 743–749.
- 79 Rainbow, L., Platt, G.M., Simpson, G.R. et al. (1997). The 222- to 234-kilodalton latent nuclear protein (LNA) of Kaposi's sarcoma-associated herpesvirus (human herpesvirus 8) is encoded by orf73 and is a component of the latency-associated nuclear antigen. *J. Virol.* 71: 5915–5921.
- **80** Casper, C., Nichols, W.G., Huang, M.L. et al. (2004). Remission of HHV-8 and HIV-associated multicentric Castleman disease with ganciclovir treatment. *Blood* 103: 1632–1634.
- **81** Uldrick, T.S., Polizzotto, M.N., Aleman, K. et al. (2011). High-dose zidovudine plus valganciclovir for Kaposi sarcoma herpesvirus-associated multicentric Castleman disease: a pilot study of virus-activated cytotoxic therapy. *Blood* 117: 6977–6986.

- **82** Powles, T., Stebbing, J., Montoto, S. et al. (2007). Rituximab as retreatment for rituximab pretreated HIV-associated multicentric Castleman disease. *Blood* 110: 4132–4133.
- **83** Bower, M., Powles, T., Williams, S. et al. (2007). Brief communication: rituximab in HIV-associated multicentric Castleman disease. *Ann. Intern. Med.* 147: 836–839.
- **84** Gerard, L., Berezne, A., Galicier, L. et al. (2007). Prospective study of rituximab in chemotherapy-dependent human immunodeficiency virus associated multicentric Castleman's disease: ANRS 117 CastlemaB Trial. *J. Clin. Oncol.* 25: 3350–3356.
- **85** Uldrick, T.S., Polizzotto, M.N., Aleman, K. et al. (2014). Rituximab plus liposomal doxorubicin in HIV-infected patients with KSHV-associated multicentric Castleman disease. *Blood* 124: 3544–3552.
- **86** Caro-Vegas, C., Sellers, S., Host, K.M. et al. (2020). Runaway Kaposi sarcoma-associated herpesvirus replication correlates with systemic IL-10 levels. *Virology* 539: 18–25.
- **87** Cantos, V.D., Kalapila, A.G., Ly Nguyen, M. et al. (2017). Experience with Kaposi sarcoma herpesvirus inflammatory cytokine syndrome in a large urban HIV clinic in the United States: case series and literature review. *Open Forum Infect. Dis.* 4: ofx196.
- **88** Karass, M., Grossniklaus, E., Seoud, T. et al. (2017). Kaposi sarcoma inflammatory cytokine syndrome (KICS): a rare but potentially treatable condition. *Oncologist* 22: 623–625.
- Kedes, D.H. and Ganem, D. (1997). Sensitivity of Kaposi's sarcoma-associated herpesvirus replication to antiviral drugs. Implications for potential therapy. *J. Clin. Invest.* 99: 2082–2086.
- **90** Medveczky, M.M., Horvath, E., Lund, T., and Medveczky, P.G. (1997). In vitro antiviral drug sensitivity of the Kaposi's sarcoma-associated herpesvirus. *AIDS* 11: 1327–1332.
- **91** Neyts, J. and De Clercq, E. (1997). Antiviral drug susceptibility of human herpesvirus 8. *Antimicrob. Agents Chemother.* 41: 2754–2756.
- **92** Sergerie, Y. and Boivin, G. (2003). Evaluation of susceptibility of human herpesvirus 8 to antiviral drugs by quantitative real-time PCR. *J. Clin. Microbiol.* 41: 3897–3900.
- 93 Coen, N., Duraffour, S., Topalis, D. et al. (2014). Spectrum of activity and mechanisms of resistance of various nucleoside derivatives against gammaherpesviruses. *Antimicrob. Agents Chemother.* 58: 7312–7323.
- **94** Coen, N., Singh, U., Vuyyuru, V. et al. (2013). Activity and mechanism of action of HDVD, a novel pyrimidine nucleoside derivative with high levels of selectivity and potency against gammaherpesviruses. *J. Virol.* 87: 3839–3851.
- **95** Zhu, W., Burnette, A., Dorjsuren, D. et al. (2005). Potent antiviral activity of northmethanocarbathymidine against Kaposi's sarcoma-associated herpesvirus. *Antimicrob. Agents Chemother.* 49: 4965–4973.
- **96** Prichard, M.N., Williams, J.D., Komazin-Meredith, G. et al. (2013). Synthesis and antiviral activities of methylenecyclopropane analogs with 6-alkoxy and 6-alkylthio substitutions that exhibit broad-spectrum antiviral activity against human herpesviruses. *Antimicrob. Agents Chemother.* 57: 3518–3527.
- **97** Beauclair, G., Naimo, E., Dubich, T. et al. (2019). Targeting the Kaposi sarcoma herpesvirus ORF 21 tyrosine kinase and viral lytic reactivation by tyrosine kinase inhibitors approved for clinical use. *J. Virol.* https://doi.org/10.1128/JVI.01791-19.
- **98** Coen, N., Duraffour, S., Snoeck, R., and Andrei, G. (2014). KSHV targeted therapy: an update on inhibitors of viral lytic replication. *Viruses* 6: 4731–4759.

- Dorjsuren, D., Burnette, A., Gray, G.N. et al. (2006). Chemical library screen for novel inhibitors of Kaposi's sarcoma-associated herpesvirus processive DNA synthesis. *Antivir. Res.* 69: 9–23.
- Yan, Z., Bryant, K.F., Gregory, S.M. et al. (2014). HIV integrase inhibitors block replication of alpha-, beta-, and gammaherpesviruses. *MBio* 5: e01318–e01314.
- Miller, J.T., Zhao, H., Masaoka, T. et al. (2018). Sensitivity of the C-terminal nuclease domain of Kaposi's sarcoma-associated herpesvirus ORF29 to two classes of active-site ligands. *Antimicrob. Agents Chemother.* 62: e00233-18.
- Shahian, T., Lee, G.M., Lazic, A. et al. (2009). Inhibition of a viral enzyme by a small-molecule dimer disruptor. *Nat. Chem. Biol.* 5: 640–646.
- Acker, T.M., Gable, J.E., Bohn, M.F. et al. (2017). Allosteric inhibitors, crystallography, and comparative analysis reveal network of coordinated movement across human herpesvirus proteases. *J. Am. Chem. Soc.* 139: 11650–11653.
- Gantt, S., Carlsson, J., Ikoma, M. et al. (2011). The HIV protease inhibitor nelfinavir inhibits Kaposi's sarcoma-associated herpesvirus replication in vitro. *Antimicrob. Agents Chemother.* 55: 2696–2703.
- Gantt, S., Cattamanchi, A., Krantz, E. et al. (2014). Reduced human herpesvirus-8 oropharyngeal shedding associated with protease inhibitor-based antiretroviral therapy. *J. Clin. Virol.* 60: 127–132.
- Chatterjee, D., Chandran, B., and Berger, E.A. (2012). Selective killing of Kaposi's sarcoma-associated herpesvirus lytically infected cells with a recombinant immunotoxin targeting the viral gpK8.1A envelope glycoprotein. *MAbs* 4: 233–242.
- Cai, Y. and Berger, E.A. (2011). An immunotoxin targeting the gH glycoprotein of KSHV for selective killing of cells in the lytic phase of infection. *Antivir. Res.* 90: 143–150.
- Kirsch, P., Jakob, V., Elgaher, W.A.M. et al. (2020). Discovery of novel latency-associated nuclear antigen inhibitors as antiviral agents against Kaposi's sarcoma-associated herpesvirus. *ACS Chem. Biol.* https://doi.org/10.1021/acschembio.9b00845.
- Kirsch, P., Jakob, V., Oberhausen, K. et al. (2019). Fragment-based discovery of a qualified hit targeting the latency-associated nuclear antigen of the oncogenic Kaposi's sarcoma-associated herpesvirus/human herpesvirus 8. *J. Med. Chem.* 62: 3924–3939.
- Briggs, L.C., Chan, A.W.E., Davis, C.A. et al. (2017). IKKgamma-mimetic peptides block the resistance to apoptosis associated with Kaposi's sarcoma-associated herpesvirus infection. *J. Virol.* 91: e01170-17.
- Sadek J, Wuo MG; Rooklin, D. et al. (2019). Modulating vFLIP levels and NF-kB signalling using NEMO coiled coli mimics. Abstr 22nd International Workshop on Kaposi's Sarcoma Herpesvirus (KSHV) and Related Agents, New York.
- Cannon, J.S., Hamzeh, F., Moore, S. et al. (1999). Human herpesvirus 8-encoded thymidine kinase and phosphotransferase homologues confer sensitivity to ganciclovir. *J. Virol.* 73: 4786–4793.
- Gill, M.B., Turner, R., Stevenson, P.G., and Way, M. (2015). KSHV-TK is a tyrosine kinase that disrupts focal adhesions and induces Rho-mediated cell contraction. *EMBO J.* 34: 448–465.
- **114** Gustafson, E.A., Schinazi, R.F., and Fingeroth, J.D. (2000). Human herpesvirus 8 open reading frame 21 is a thymidine and thymidylate kinase of narrow substrate specificity that efficiently phosphorylates zidovudine but not ganciclovir. *J. Virol.* 74: 684–692.

- **115** Casper, C., Krantz, E.M., Corey, L. et al. (2008). Valganciclovir for suppression of human herpesvirus-8 replication: a randomized, double-blind, placebo-controlled, crossover trial. *J. Infect. Dis.* 198: 23–30.
- **116** Cattamanchi, A., Saracino, M., Selke, S. et al. (2011). Treatment with valacyclovir, famciclovir, or antiretrovirals reduces human herpesvirus-8 replication in HIV-1 seropositive men. *J. Med. Virol.* 83: 1696–1703.
- **117** Boivin, G., Gaudreau, A., Toma, E. et al. (1999). Human herpesvirus 8 DNA load in leukocytes of human immunodeficiency virus-infected subjects: correlation with the presence of Kaposi's sarcoma and response to anticytomegalovirus therapy. *Antimicrob. Agents Chemother.* 43: 377–380.
- **118** Humphrey, R.W., O'Brien, T.R., Newcomb, F.M. et al. (1996). Kaposi's sarcoma (KS)associated herpesvirus-like DNA sequences in peripheral blood mononuclear cells: association with KS and persistence in patients receiving anti-herpesvirus drugs. *Blood* 88: 297–301.
- **119** Low, P., Neipel, F., Rascu, A. et al. (1998). Suppression of HHV-8 viremia by foscarnet in an HIV-infected patient with Kaposi's sarcoma and HHV-8 associated hemophagocytic syndrome. *Eur. J. Med. Res.* 3: 461–464.
- **120** Krown, S.E., Dittmer, D.P., and Cesarman, E. (2011). Pilot study of oral valganciclovir therapy in patients with classic Kaposi sarcoma. *J. Infect. Dis.* 203: 1082–1086.
- **121** Little, R.F., Merced-Galindez, F., Staskus, K. et al. (2003). A pilot study of cidofovir in patients with kaposi sarcoma. *J. Infect. Dis.* 187: 149–153.
- **122** Mazzi, R., Parisi, S.G., Sarmati, L. et al. (2001). Efficacy of cidofovir on human herpesvirus 8 viraemia and Kaposi's sarcoma progression in two patients with AIDS. *AIDS* 15: 2061–2062.
- 123 Simonart, T., Noel, J.C., De Dobbeleer, G. et al. (1998). Treatment of classical Kaposi's sarcoma with intralesional injections of cidofovir: report of a case. J. Med. Virol. 55: 215–218.
- **124** Hosseinipour, M.C., Sweet, K.M., Xiong, J. et al. (2014). Viral profiling identifies multiple subtypes of Kaposi's sarcoma. *MBio* 5: e01633–e01614.
- **125** Abere, B., Mamo, T.M., Hartmann, S. et al. (2017). The Kaposi's sarcoma-associated herpesvirus (KSHV) non-structural membrane protein K15 is required for viral lytic replication and may represent a therapeutic target. *PLoS Pathog.* 13: e1006639.
- 126 Martin, D.F., Kuppermann, B.D., Wolitz, R.A. et al. (1999). Oral ganciclovir for patients with cytomegalovirus retinitis treated with a ganciclovir implant. Roche Ganciclovir Study Group. N. Engl. J. Med. 340: 1063–1070.
- 127 Mocroft, A., Youle, M., Gazzard, B. et al. (1996). Anti-herpesvirus treatment and risk of Kaposi's sarcoma in HIV infection. Royal Free/Chelsea and Westminster Hospitals Collaborative Group. *AIDS* 10: 1101–1105.
- **128** Verucchi, G., Calza, L., Trevisani, F. et al. (2005). Human herpesvirus-8-related Kaposi's sarcoma after liver transplantation successfully treated with cidofovir and liposomal daunorubicin. *Transpl. Infect. Dis.* 7: 34–37.
- **129** Murphy, C., Hawkes, E., Chionh, F., and Chong, G. (2017). Durable remission of both multicentric Castleman's disease and Kaposi's sarcoma with valganciclovir, rituximab and liposomal doxorubicin in an HHV-8-positive, HIV-negative patient. *J. Clin. Pharm. Ther.* 42: 111–114.

- **130** Luppi, M., Trovato, R., Barozzi, P. et al. (2005). Treatment of herpesvirus associated primary effusion lymphoma with intracavity cidofovir. *Leukemia* 19: 473–476.
- **131** Marquet, J., Velazquez-Kennedy, K., Lopez, S. et al. (2018). Case report of a primary effusion lymphoma successfully treated with oral valganciclovir after failing chemotherapy. *Hematol. Oncol.* 36: 316–319.
- **132** Ozbalak, M., Tokatli, I., Ozdemirli, M. et al. (2013). Is valganciclovir really effective in primary effusion lymphoma: case report of an HIV(–) EBV(–) HHV8(+) patient. *Eur. J. Haematol.* 91: 467–469.
- **133** Pastore, R.D., Chadburn, A., Kripas, C., and Schattner, E.J. (2000). Novel association of haemophagocytic syndrome with Kaposi's sarcoma-associated herpesvirus-related primary effusion lymphoma. *Br. J. Haematol.* 111: 1112–1115.
- **134** Berezne, A., Agbalika, F., and Oksenhendler, E. (2004). Failure of cidofovir in HIVassociated multicentric Castleman disease. *Blood* 103: 4368–4369. author reply 4369.
- 135 Sbenghe, M.M., Besa, E., Mahipal, A. et al. (2012). HHV-8-associated multicentric Castleman's disease in HIV-negative patient: a novel therapy for an orphan disease. Oncologist 17: 145–146. author reply 147-8.
- **136** Barbarov, I., Koren-Michowitz, M., Schiby, G. et al. (2015). Fulminant HHV-8 associated Castleman's disease in a non-HIV, Kaposi sarcoma patient with borderline hemophagocytic syndrome. *Isr. Med. Assoc. J.* 17: 253–255.
- **137** Kantarci, F.E., Eren, R., Gundogan, C. et al. (2016). A HHV-8 positive, HIV negative multicentric Castleman disease treated with R-CEOP chemotherapy and valganciclovir combination. *J. Infect. Chemother.* 22: 483–485.
- **138** Koon, H.B., Bubley, G.J., Pantanowitz, L. et al. (2005). Imatinib-induced regression of AIDS-related Kaposi's sarcoma. *J. Clin. Oncol.* 23: 982–989.
- 139 Koon, H.B., Krown, S.E., Lee, J.Y. et al. (2014). Phase II trial of imatinib in AIDS-associated Kaposi's sarcoma: AIDS Malignancy Consortium Protocol 042. *J. Clin. Oncol.* 32: 402–408.
- **140** Ardavanis, A., Doufexis, D., Kountourakis, P., and Rigatos, G. (2008). A Kaposi's sarcoma complete clinical response after sorafenib administration. *Ann. Oncol.* 19: 1658–1659.
- **141** Uldrick, T.S., Goncalves, P.H., Wyvill, K.M. et al. (2017). A phase Ib study of sorafenib (BAY 43-9006) in patients with Kaposi sarcoma. *Oncologist* 22: 505–e49.
- 142 Bais, C., Santomasso, B., Coso, O. et al. (1998). G-protein-coupled receptor of Kaposi's sarcoma-associated herpesvirus is a viral oncogene and angiogenesis activator. *Nature* 391: 86–89.
- **143** Wang, L., Dittmer, D.P., Tomlinson, C.C. et al. (2006). Immortalization of primary endothelial cells by the K1 protein of Kaposi's sarcoma-associated herpesvirus. *Cancer Res.* 66: 3658–3666.
- Bala, K., Bosco, R., Gramolelli, S. et al. (2012). Kaposi's sarcoma herpesvirus K15 protein contributes to virus-induced angiogenesis by recruiting PLCgamma1 and activating NFAT1-dependent RCAN1 expression. *PLoS Pathog.* 8: e1002927.
- **145** Bhatt, A.P., Wong, J.P., Weinberg, M.S. et al. (2016). A viral kinase mimics S6 kinase to enhance cell proliferation. *Proc. Natl. Acad. Sci. USA* 113: 7876–7881.
- 146 Morris, V.A., Punjabi, A.S., Wells, R.C. et al. (2012). The KSHV viral IL-6 homolog is sufficient to induce blood to lymphatic endothelial cell differentiation. *Virology* 428: 112–120.

- 222 8 Kaposi's Sarcoma-associated Herpesvirus-Antiviral Treatment
 - **147** Stallone, G., Schena, A., Infante, B. et al. (2005). Sirolimus for Kaposi's sarcoma in renal-transplant recipients. *N. Engl. J. Med.* 352: 1317–1323.
 - 148 Yaich, S., Charfeddine, K., Zaghdane, S. et al. (2012). Sirolimus for the treatment of Kaposi sarcoma after renal transplantation: a series of 10 cases. *Transplant. Proc.* 44: 2824–2826.
 - 149 Krown, S.E., Roy, D., Lee, J.Y. et al. (2012). Rapamycin with antiretroviral therapy in AIDS-associated Kaposi sarcoma: an AIDS Malignancy Consortium study. J. Acquir. Immune Defic. Syndr. 59: 447–454.
 - **150** Pan, H., Xie, J., Ye, F., and Gao, S.J. (2006). Modulation of Kaposi's sarcoma-associated herpesvirus infection and replication by MEK/ERK, JNK, and p38 multiple mitogen-activated protein kinase pathways during primary infection. *J. Virol.* 80: 5371–5382.
 - **151** Yu, F., Harada, J.N., Brown, H.J. et al. (2007). Systematic identification of cellular signals reactivating Kaposi sarcoma-associated herpesvirus. *PLoS Pathog.* 3: e44.
 - **152** Wakao, K., Watanabe, T., Takadama, T. et al. (2014). Sangivamycin induces apoptosis by suppressing Erk signaling in primary effusion lymphoma cells. *Biochem. Biophys. Res. Commun.* 444: 135–140.
 - **153** Moriguchi, M., Watanabe, T., Kadota, A., and Fujimuro, M. (2019). Capsaicin Induces Apoptosis in KSHV-Positive Primary Effusion Lymphoma by Suppressing ERK and p38 MAPK Signaling and IL-6 Expression. *Front. Oncol.* 9: 83.
 - **154** Cannon, M., Philpott, N.J., and Cesarman, E. (2003). The Kaposi's sarcoma-associated herpesvirus G protein-coupled receptor has broad signaling effects in primary effusion lymphoma cells. *J. Virol.* 77: 57–67.
 - **155** Kim, H., Jang, J.H., Song, Y.E., and Seo, T. (2020). Kaposi's sarcoma-associated herpesvirus viral protein kinase phosphorylates extracellular signal-regulated kinase and activates MAPK/ERK signaling pathway. *Biochem. Biophys. Res. Commun.* 521: 1083–1088.
 - **156** Kuang, E., Tang, Q., Maul, G.G., and Zhu, F. (2008). Activation of p90 ribosomal S6 kinase by ORF45 of Kaposi's sarcoma-associated herpesvirus and its role in viral lytic replication. *J. Virol.* 82: 1838–1850.
 - 157 Kuang, E., Wu, F., and Zhu, F. (2009). Mechanism of sustained activation of ribosomal S6 kinase (RSK) and ERK by kaposi sarcoma-associated herpesvirus ORF45: multiprotein complexes retain active phosphorylated ERK AND RSK and protect them from dephosphorylation. J. Biol. Chem. 284: 13958–13968.
 - **158** Fu, B., Kuang, E., Li, W. et al. (2015). Activation of p90 ribosomal S6 kinases by ORF45 of Kaposi's sarcoma-associated herpesvirus is critical for optimal production of infectious viruses. *J. Virol.* 89: 195–207.
 - 159 Avey, D., Tepper, S., Li, W. et al. (2015). Phosphoproteomic analysis of KSHV-infected cells reveals roles of ORF45-activated RSK during lytic replication. *PLoS Pathog.* 11: e1004993.
 - 160 Kuang, E., Fu, B., Liang, Q. et al. (2011). Phosphorylation of eukaryotic translation initiation factor 4B (EIF4B) by open reading frame 45/p90 ribosomal S6 kinase (ORF45/RSK) signaling axis facilitates protein translation during Kaposi sarcoma-associated herpesvirus (KSHV) lytic replication. J. Biol. Chem. 286: 41171–41182.
 - **161** Li, X., Huang, L., Xiao, Y. et al. (2019). Development of an ORF45-derived peptide to inhibit the sustained RSK activation and lytic replication of Kaposi's sarcoma-associated herpesvirus. *J. Virol.* 93: e02154-18.

- **162** Coleman, H.M., de Lima, B., Morton, V., and Stevenson, P.G. (2003). Murine gammaherpesvirus 68 lacking thymidine kinase shows severe attenuation of lytic cycle replication in vivo but still establishes latency. *J. Virol.* 77: 2410–2417.
- 163 Song, M.J., Hwang, S., Wong, W.H. et al. (2005). Identification of viral genes essential for replication of murine gamma-herpesvirus 68 using signature-tagged mutagenesis. *Proc. Natl. Acad. Sci. USA* 102: 3805–3810.
- **164** Dai, L., Trillo-Tinoco, J., Cao, Y. et al. (2015). Targeting HGF/c-MET induces cell cycle arrest, DNA damage, and apoptosis for primary effusion lymphoma. *Blood* 126: 2821–2831.
- **165** Wong, J.P., Stuhlmiller, T.J., Giffin, L.C. et al. (2019). Kinome profiling of non-Hodgkin lymphoma identifies Tyro3 as a therapeutic target in primary effusion lymphoma. *Proc. Natl. Acad. Sci. USA* 116: 16541–16550.
- **166** Sodhi, A., Chaisuparat, R., Hu, J. et al. (2006). The TSC2/mTOR pathway drives endothelial cell transformation induced by the Kaposi's sarcoma-associated herpesvirus G protein-coupled receptor. *Cancer Cell* 10: 133–143.
- **167** Chang, H.H. and Ganem, D. (2013). A unique herpesviral transcriptional program in KSHV-infected lymphatic endothelial cells leads to mTORC1 activation and rapamycin sensitivity. *Cell Host Microbe* 13: 429–440.
- **168** Roy, D., Sin, S.H., Lucas, A. et al. (2013). mTOR inhibitors block Kaposi sarcoma growth by inhibiting essential autocrine growth factors and tumor angiogenesis. *Cancer Res.* 73: 2235–2246.
- **169** Sin, S.H., Roy, D., Wang, L. et al. (2007). Rapamycin is efficacious against primary effusion lymphoma (PEL) cell lines in vivo by inhibiting autocrine signaling. *Blood* 109: 2165–2173.
- **170** Bhatt, A.P., Bhende, P.M., Sin, S.H. et al. (2010). Dual inhibition of PI3K and mTOR inhibits autocrine and paracrine proliferative loops in PI3K/Akt/mTOR-addicted lymphomas. *Blood* 115: 4455–4463.
- **171** Caro-Vegas, C., Bailey, A., Bigi, R. et al. (2019). Targeting mTOR with MLN0128 overcomes rapamycin and chemoresistant primary effusion lymphoma. *MBio* 10: e02871-18.
- 172 Yang, M., Huang, L., Li, X., and Kuang, E. (2016). Chloroquine inhibits lytic replication of Kaposi's sarcoma-associated herpesvirus by disrupting mTOR and p38-MAPK activation. *Antivir. Res.* 133: 223–233.
- Uldrick, T.S., Wyvill, K.M., Kumar, P. et al. (2012). Phase II study of bevacizumab in patients with HIV-associated Kaposi's sarcoma receiving antiretroviral therapy. *J. Clin. Oncol.* 30: 1476–1483.
- **174** Ramaswami, R., Uldrick, T.S., Polizzotto, M.N. et al. (2019). A pilot study of liposomal doxorubicin combined with bevacizumab followed by bevacizumab monotherapy in patients with advanced Kaposi sarcoma. *Clin. Cancer Res.* 25: 4238–4247.
- **175** Ablanedo-Terrazas, Y., Alvarado-de la Barrera, C., Ormsby, C.E. et al. (2015). Intralesional bevacizumab in patients with human immunodeficiency virus-associated Kaposi's sarcoma in the upper airway. *Laryngoscope* 125: E132–E137.
- **176** van Rhee, F., Greenway, A., and Stone, K. (2018). Treatment of idiopathic castleman disease. *Hematol. Oncol. Clin. North Am.* 32: 89–106.
- **177** Nagao, A., Nakazawa, S., and Hanabusa, H. (2014). Short-term efficacy of the IL6 receptor antibody tocilizumab in patients with HIV-associated multicentric Castleman disease: report of two cases. *J. Hematol. Oncol.* 7: 10.

- 224 8 Kaposi's Sarcoma-associated Herpesvirus-Antiviral Treatment
 - **178** Chen, W., Sin, S.H., Wen, K.W. et al. (2012). Hsp90 inhibitors are efficacious against Kaposi sarcoma by enhancing the degradation of the essential viral gene LANA, of the viral co-receptor EphA2 as well as other client proteins. *PLoS Pathog.* 8: e1003048.
 - **179** Nayar, U., Lu, P., Goldstein, R.L. et al. (2013). Targeting the Hsp90-associated viral oncoproteome in gammaherpesvirus-associated malignancies. *Blood* 122: 2837–2847.
 - 180 Gopalakrishnan, R., Matta, H., and Chaudhary, P.M. (2013). A purine scaffold HSP90 inhibitor BIIB021 has selective activity against KSHV-associated primary effusion lymphoma and blocks vFLIP K13-induced NF-kappaB. *Clin. Cancer Res.* 19: 5016–5026.
 - **181** Field, N., Low, W., Daniels, M. et al. (2003). KSHV vFLIP binds to IKK-gamma to activate IKK. *J. Cell Sci.* 116: 3721–3728.
 - **182** Baquero-Perez, B. and Whitehouse, A. (2015). Hsp70 isoforms are essential for the formation of Kaposi's sarcoma-associated herpesvirus replication and transcription compartments. *PLoS Pathog.* 11: e1005274.
 - **183** Hughes, D.J., Wood, J.J., Jackson, B.R. et al. (2015). NEDDylation is essential for Kaposi's sarcoma-associated herpesvirus latency and lytic reactivation and represents a novel anti-KSHV target. *PLoS Pathog.* 11: e1004771.
 - **184** Reid, E.G. (2011). Bortezomib-induced Epstein–Barr virus and Kaposi sarcoma herpesvirus lytic gene expression: oncolytic strategies. *Curr. Opin. Oncol.* 23: 482–487.
 - **185** Granato, M., Romeo, M.A., Tiano, M.S. et al. (2017). Bortezomib promotes KHSV and EBV lytic cycle by activating JNK and autophagy. *Sci. Rep.* 7: 13052.
 - 186 Matta, H. and Chaudhary, P.M. (2005). The proteasome inhibitor bortezomib (PS-341) inhibits growth and induces apoptosis in primary effusion lymphoma cells. *Cancer Biol. Ther.* 4: 77–82.
 - 187 Sarosiek, K.A., Cavallin, L.E., Bhatt, S. et al. (2010). Efficacy of bortezomib in a direct xenograft model of primary effusion lymphoma. *Proc. Natl. Acad. Sci. USA* 107: 13069–13074.
 - 188 Reid, E.G., Looney, D., Maldarelli, F. et al. (2018). Safety and efficacy of an oncolytic viral strategy using bortezomib with ICE/R in relapsed/refractory HIV-positive lymphomas. *Blood Adv.* 2: 3618–3626.
 - 189 Reid, E.G., Suazo, A., Lensing, S.Y. et al. (2019). Pilot trial AMC-063: safety and efficacy of bortezomib in AIDS-associated Kaposi sarcoma. *Clin. Cancer Res.* https://doi. org/10.1158/1078-0432.CCR-19-1044.
 - **190** Cheng, F., Pekkonen, P., Laurinavicius, S. et al. (2011). KSHV-initiated notch activation leads to membrane-type-1 matrix metalloproteinase-dependent lymphatic endothelial-to-mesenchymal transition. *Cell Host Microbe* 10: 577–590.
 - **191** Gasperini, P., Espigol-Frigole, G., McCormick, P.J. et al. (2012). Kaposi sarcoma herpesvirus promotes endothelial-to-mesenchymal transition through Notch-dependent signaling. *Cancer Res.* 72: 1157–1169.
 - **192** Wang, X., He, Z., Xia, T. et al. (2014). Latency-associated nuclear antigen of Kaposi sarcoma-associated herpesvirus promotes angiogenesis through targeting notch signaling effector Hey1. *Cancer Res.* 74: 2026–2037.
 - 193 Curry, C.L., Reed, L.L., Golde, T.E. et al. (2005). Gamma secretase inhibitor blocks Notch activation and induces apoptosis in Kaposi's sarcoma tumor cells. *Oncogene* 24: 6333–6344.

- **194** Wang, Y., Li, H., Tang, Q. et al. (2008). Kaposi's sarcoma-associated herpesvirus ori-Lytdependent DNA replication: involvement of host cellular factors. *J. Virol.* 82: 2867–2882.
- Gonzalez-Molleda, L., Wang, Y., and Yuan, Y. (2012). Potent antiviral activity of topoisomerase I and II inhibitors against Kaposi's sarcoma-associated herpesvirus. *Antimicrob. Agents Chemother.* 56: 893–902.
- Xu, B., Wang, L., Gonzalez-Molleda, L. et al. (2014). Antiviral activity of (+)-rutamarin against Kaposi's sarcoma-associated herpesvirus by inhibition of the catalytic activity of human topoisomerase II. *Antimicrob. Agents Chemother.* 58: 563–573.
- Haque, M., Davis, D.A., Wang, V. et al. (2003). Kaposi's sarcoma-associated herpesvirus (human herpesvirus 8) contains hypoxia response elements: relevance to lytic induction by hypoxia. *J. Virol.* 77: 6761–6768.
- Haque, M., Wang, V., Davis, D.A. et al. (2006). Genetic organization and hypoxic activation of the Kaposi's sarcoma-associated herpesvirus ORF34-37 gene cluster. *J. Virol.* 80: 7037–7051.
- Cai, Q., Lan, K., Verma, S.C. et al. (2006). Kaposi's sarcoma-associated herpesvirus latent protein LANA interacts with HIF-1 alpha to upregulate RTA expression during hypoxia: latency control under low oxygen conditions. *J. Virol.* 80: 7965–7975.
- Cai, Q., Murakami, M., Si, H., and Robertson, E.S. (2007). A potential alpha-helix motif in the amino terminus of LANA encoded by Kaposi's sarcoma-associated herpesvirus is critical for nuclear accumulation of HIF-1alpha in normoxia. *J. Virol.* 81: 10413–10423.
- Sodhi, A., Montaner, S., Patel, V. et al. (2000). The Kaposi's sarcoma-associated herpes virus G protein-coupled receptor up-regulates vascular endothelial growth factor expression and secretion through mitogen-activated protein kinase and p38 pathways acting on hypoxia-inducible factor 1alpha. *Cancer Res.* 60: 4873–4880.
- Shin, Y.C., Joo, C.H., Gack, M.U. et al. (2008). Kaposi's sarcoma-associated herpesvirus viral IFN regulatory factor 3 stabilizes hypoxia-inducible factor-1 alpha to induce vascular endothelial growth factor expression. *Cancer Res.* 68: 1751–1759.
- Carroll, P.A., Kenerson, H.L., Yeung, R.S., and Lagunoff, M. (2006). Latent Kaposi's sarcoma-associated herpesvirus infection of endothelial cells activates hypoxia-induced factors. *J. Virol.* 80: 10802–10812.
- Delgado, T., Carroll, P.A., Punjabi, A.S. et al. (2010). Induction of the Warburg effect by Kaposi's sarcoma herpesvirus is required for the maintenance of latently infected endothelial cells. *Proc. Natl. Acad. Sci. USA* 107: 10696–10701.
- Shrestha, P., Davis, D.A., Veeranna, R.P. et al. (2017). Hypoxia-inducible factor-1 alpha as a therapeutic target for primary effusion lymphoma. *PLoS Pathog.* 13: e1006628.
- He, M., Tan, B., Vasan, K. et al. (2017). SIRT1 and AMPK pathways are essential for the proliferation and survival of primary effusion lymphoma cells. *J. Pathol.* 242: 309–321.
- Zhong, C., Xu, M., Wang, Y. et al. (2017). An APE1 inhibitor reveals critical roles of the redox function of APE1 in KSHV replication and pathogenic phenotypes. *PLoS Pathog.* 13: e1006289.
- Dai, L., Qiao, J., Nguyen, D. et al. (2016). Role of heme oxygenase-1 in the pathogenesis and tumorigenicity of Kaposi's sarcoma-associated herpesvirus. *Oncotarget* 7: 10459–10471.
- Gruffaz, M., Yuan, H., Meng, W. et al. (2019). CRISPR-Cas9 screening of Kaposi's sarcoma-associated herpesvirus-transformed cells identifies XPO1 as a vulnerable target of cancer cells. *MBio* 10: e00866-19.

- 226 8 Kaposi's Sarcoma-associated Herpesvirus-Antiviral Treatment
 - **210** Chen, J., Dai, L., Goldstein, A. et al. (2019). Identification of new antiviral agents against Kaposi's sarcoma-associated herpesvirus (KSHV) by high-throughput drug screening reveals the role of histamine-related signaling in promoting viral lytic reactivation. *PLoS Pathog.* 15: e1008156.
 - **211** Hahn, A.S. and Desrosiers, R.C. (2014). Binding of the Kaposi's sarcoma-associated herpesvirus to the ephrin binding surface of the EphA2 receptor and its inhibition by a small molecule. *J. Virol.* 88: 8724–8734.
 - **212** Unal, A., Pray, T.R., Lagunoff, M. et al. (1997). The protease and the assembly protein of Kaposi's sarcoma-associated herpesvirus (human herpesvirus 8). *J. Virol.* 71: 7030–7038.
 - **213** Shimba, N., Nomura, A.M., Marnett, A.B., and Craik, C.S. (2004). Herpesvirus protease inhibition by dimer disruption. *J. Virol.* 78: 6657–6665.
 - **214** Lee, G.M., Shahian, T., Baharuddin, A. et al. (2011). Enzyme inhibition by allosteric capture of an inactive conformation. *J. Mol. Biol.* 411: 999–1016.
 - 215 Bani-Sadr, F., Fournier, S., and Molina, J.M. (2003). Relapse of Kaposi's sarcoma in HIV-infected patients switching from a protease inhibitor to a non-nucleoside reverse transcriptase inhibitor-based highly active antiretroviral therapy regimen. *AIDS* 17: 1580–1581.
 - **216** Leitch, H., Trudeau, M., and Routy, J.P. (2003). Effect of protease inhibitor-based highly active antiretroviral therapy on survival in HIV-associated advanced Kaposi's sarcoma patients treated with chemotherapy. *HIV Clin. Trials* 4: 107–114.
 - 217 Monini, P., Sgadari, C., Grosso, M.G. et al. (2009). Clinical course of classic Kaposi's sarcoma in HIV-negative patients treated with the HIV protease inhibitor indinavir. *AIDS* 23: 534–538.
 - **218** Guasparri, I., Keller, S.A., and Cesarman, E. (2004). KSHV vFLIP is essential for the survival of infected lymphoma cells. *J. Exp. Med.* 199: 993–1003.
 - **219** Godfrey, A., Anderson, J., Papanastasiou, A. et al. (2005). Inhibiting primary effusion lymphoma by lentiviral vectors encoding short hairpin RNA. *Blood* 105: 2510–2518.
 - **220** Zhang, Z., Chen, W., Sanders, M.K. et al. (2016). The K1 protein of Kaposi's sarcomaassociated herpesvirus augments viral lytic replication. *J. Virol.* 90: 7657–7666.
 - **221** Sandford, G., Choi, Y.B., and Nicholas, J. (2009). Role of ORF74-encoded viral G proteincoupled receptor in human herpesvirus 8 lytic replication. *J. Virol.* 83: 13009–13014.
 - **222** Weidner-Glunde, M., Mariggio, G., and Schulz, T.F. (2017). Kaposi's sarcoma-associated herpesvirus latency-associated nuclear antigen: replicating and shielding viral DNA during viral persistence. *J. Virol.* 91: e01083-16.
 - **223** Ballestas, M.E., Chatis, P.A., and Kaye, K.M. (1999). Efficient persistence of extrachromosomal KSHV DNA mediated by latency-associated nuclear antigen. *Science* 284: 641–644.
 - **224** Ballestas, M.E. and Kaye, K.M. (2001). Kaposi's sarcoma-associated herpesvirus latencyassociated nuclear antigen 1 mediates episome persistence through cis-acting terminal repeat (TR) sequence and specifically binds TR DNA. *J. Virol.* 75: 3250–3258.
 - **225** Garber, A.C., Shu, M.A., Hu, J., and Renne, R. (2001). DNA binding and modulation of gene expression by the latency-associated nuclear antigen of Kaposi's sarcoma-associated herpesvirus. *J. Virol.* 75: 7882–7892.
 - **226** Hellert, J., Weidner-Glunde, M., Krausze, J. et al. (2013). A structural basis for BRD2/4mediated host chromatin interaction and oligomer assembly of Kaposi

sarcoma-associated herpesvirus and murine gammaherpesvirus LANA proteins. *PLoS Pathog.* 9: e1003640.

- Hellert, J., Weidner-Glunde, M., Krausze, J. et al. (2015). The 3D structure of Kaposi sarcoma herpesvirus LANA C-terminal domain bound to DNA. *Proc. Natl. Acad. Sci. USA* 112: 6694–6699.
- Domsic, J.F., Chen, H.S., Lu, F. et al. (2013). Molecular basis for oligomeric-DNA binding and episome maintenance by KSHV LANA. *PLoS Pathog.* 9: e1003672.
- Barbera, A.J., Ballestas, M.E., and Kaye, K.M. (2004). The Kaposi's sarcoma-associated herpesvirus latency-associated nuclear antigen 1 N terminus is essential for chromosome association, DNA replication, and episome persistence. *J. Virol.* 78: 294–301.
- Kelley-Clarke, B., Ballestas, M.E., Komatsu, T., and Kaye, K.M. (2007). Kaposi's sarcoma herpesvirus C-terminal LANA concentrates at pericentromeric and peri-telomeric regions of a subset of mitotic chromosomes. *Virology* 357: 149–157.
- Curreli, F., Friedman-Kien, A.E., and Flore, O. (2005). Glycyrrhizic acid alters Kaposi sarcoma-associated herpesvirus latency, triggering p53-mediated apoptosis in transformed B lymphocytes. *J. Clin. Invest.* 115: 642–652.
- Chugh, P., Matta, H., Schamus, S. et al. (2005). Constitutive NF-kappaB activation, normal Fas-induced apoptosis, and increased incidence of lymphoma in human herpes virus 8 K13 transgenic mice. *Proc. Natl. Acad. Sci. USA* 102: 12885–12890.
- Keller, S.A., Schattner, E.J., and Cesarman, E. (2000). Inhibition of NF-kappaB induces apoptosis of KSHV-infected primary effusion lymphoma cells. *Blood* 96: 2537–2542.
- Keller, S.A., Hernandez-Hopkins, D., Vider, J. et al. (2006). NF-kappaB is essential for the progression of KSHV- and EBV-infected lymphomas in vivo. *Blood* 107: 3295–3302.
- Ballon, G., Chen, K., Perez, R. et al. (2011). Kaposi sarcoma herpesvirus (KSHV) vFLIP oncoprotein induces B cell transdifferentiation and tumorigenesis in mice. *J. Clin. Invest.* 121: 1141–1153.
- Ballon, G., Akar, G., and Cesarman, E. (2015). Systemic expression of Kaposi sarcoma herpesvirus (KSHV) Vflip in endothelial cells leads to a profound proinflammatory phenotype and myeloid lineage remodeling in vivo. *PLoS Pathog.* 11: e1004581.
- Bagneris, C., Ageichik, A.V., Cronin, N. et al. (2008). Crystal structure of a vFlip-IKKgamma complex: insights into viral activation of the IKK signalosome. *Mol. Cell* 30: 620–631.
Direct-Acting Antivirals for the Treatment of Chronic Hepatitis B Virus (HBV) Infection

Angelica Corcuera, Alastair Donald, and Andreas Urban

AiCuris Anti-Infective Cures AG, Wuppertal, Germany

9

9.1 Nucleos(t)ide Analog Reverse Transcriptase Inhibitors

Nucleoside/nucleotide analogs (NAs) inhibit the reverse transcriptase (RT) activity of the HBV polymerase protein (Pol). Since the approval of lamivudine (3TC) for HBV treatment in 1998, several other NAs have also been approved (Figure 9.1). With their high potency and high barrier to resistance, entecavir (ETV) and tenofovir (TFV) are considered the first-line treatment NAs, while the use of 3TC, telbivudine (LdT), emtricitabine (FTC), and adefovir (ADV) in monotherapy is no longer routinely recommended due to their relatively low barrier to resistance [6, 7]. Clevudine (CLV) was approved only in South Korea and the Philippines; however, the approval was withdrawn due to drug-related side effects [8]. Besifovir (BSV) is a novel acyclic nucleotide phosphonate with a similar chemical structure to ADV and TFV. It has a good clinical safety profile and the same potent antiviral activity as ETV and TFV. In 2017, its use was approved, but only in South Korea [9, 10].

In general, NAs show good bioavailability after oral dosing, although some of them must be administered as a prodrug. They have favorable pharmacokinetics with limited drugdrug interaction potential and low pill burden. It has been demonstrated that NAs are very effective in suppressing viral load to undetectable levels in a high number of patients. In addition, it could be demonstrated that NAs are able to block mother-to-child transmission which is the main transmission path for HBV in endemic areas [11].

The mode of action of NAs has been investigated in several studies. NAs interfere with the Pol in different ways (Table 9.1). Most of them preferentially act on one activity of the Pol. 3TC, FTC, ADV, and TFV block the reverse transcription of the negative (–) strand viral DNA from the pgRNA, while LdT and CLV block the positive (+) strand viral DNA synthesis. On the other hand, ETV blocks all three major activities of the Pol: base priming, reverse transcription, and synthesis of the positive strand viral DNA [2, 3, 12, 13]. Since BSV is also a guanosine analog, it is hypothesized that it could have a mode of action



Figure 9.1 Chemical structures of NAs approved for HBV therapy. Adefovir and tenofovir are acyclic analogs of the nucleotide adenosine monophosphate, differing only slightly at the acyclic moiety. Lamivudine and emtricitabine are both analogs of cytidine with an oxathiolane ring replacing deoxyribose. Lamivudine is an L-enantiomer derivative of cytidine, while emtricitabine is a fluoro-derivative of p-cytidine [1]. Telbivudine and clevudine are both L-enantiomeric analogs of thymidine [2, 3]. Entecavir is a deoxyguanosine derivative with an all-carbon pentacyclic ring replacing deoxyribose [4], while besifovir is an acyclic analog of guanosine monophosphate [5].

similar to ETV [5]. Aside from differences in mode of action, NAs also vary in potency, safety profile, and resistance profile. The most common drug resistance mutations appear in the YMDD motif within the active site of the RT domain of the Pol. An excellent overview of resistance profiles and the advantages and limitations of currently approved NAs for HBV treatment can be found in a recent review [14].

NAs provide stronger viral suppression compared to IFN therapy; however, they do not target cccDNA transcription, and so, result in relatively low rates of HBsAg loss and/or HBeAg seroconversion. Consequently, NAs given in monotherapy are not able to cure chronic hepatitis B infection [15]; however, the discovery of new NAs with higher viral suppression potency could theoretically be an option to prevent intracellular cccDNA recycling and to increase cure rates for NA-based treatment strategies.

Combination treatment with NAs and IFN also provides another theoretical possibility for a cure because as monotherapies, they both have different but complementary effects

Drug name and code	Classification	Mode of Action
Lamivudine (3TC)	L-cytidine analog	Acts on RT function; inhibition of (–) strand synthesis
Emtricitabine (FTC)	Fluorinated D-cytidine analog	Acts on RT function; inhibition of (-) strand synthesis
Telbivudine (LdT)	L-thymidine analog	Acts on DNA Pol function; inhibition of viral (+) strand synthesis
Clevudine (CLV)	L-thymidine analog	Preferentially acts on DNA Pol function but can also slightly interfere with priming and RT; inhibition of viral (+) strand synthesis
Adefovir (ADF)	Acyclic analog of adenosine monophosphate	Acts on RT function; inhibition of (-) strand synthesis
Tenofovir (TFV)	Acyclic analog of adenosine monophosphate	Acts on RT function; inhibition of (–) strand synthesis
Entecavir (ETV)	Deoxyguanosine analog	Acts on RT, DNA Pol and primase functions; inhibition of priming, reverse transcription and (+) strand synthesis
Besifovir (BSV)	Acyclic analog of a guanosine monophosphate	Hypothesized to be like ETV

Table 9.1Classification and mode of action of nucleos(t)ide analogs approved for the treatmentof chronic hepatitis B [1–5, 12, 13].

on the immune response and viral control of HBV. PEG-IFN boosts innate immunity, while NAs boost adaptive immunity [16]. Unfortunately, current combinations only offer small improvements in the rates of HBsAg loss, the best prognostic marker for HBV control. A recent open-label, active-controlled TDF and PEG-IFN combination therapy study with 740 CHB patients has shown that 120 weeks after the start of the therapy, the group treated with TDF and PEG-IFN for 48 weeks had almost three times higher rate of HBsAg loss (10.4%) compared to the group treated with PEG-IFN alone for 48 weeks (3.5%) or the group treated with TDF and PEG-IFN for 16 weeks followed by TDF alone for 32 weeks (3.5%) [17].

Aside from its RT function, the Pol also has RNaseH activity. The RNAseH destroys the pgRNA after it has been retrotranscribed into DNA. In vitro studies have shown that inhibiting the RNaseH function leads to truncation of the (-) strand DNA and blocks (+) strand DNA synthesis, resulting in extensive accumulation of RNA/DNA heteroduplexes within viral capsids. This prevents the formation of mature HBV genomes and the production of infectious viral particles [18]. To date, there is no HBV RNaseH inhibitor in clinical development. The reasons for this are mainly technical problems in establishing a screening cascade. Firstly, it is very difficult to produce the HBV RNaseH domain in its active form for use in biochemical assays. Secondly, it is not easy to establish cell-based assays with acceptable throughput and a predictive and discriminatory readout that can detect the effects of the screening compound, such as inhibition of (+) strand DNA synthesis and the accumulation of truncated (-) strand DNA and RNA/DNA heteroduplexes [19].

9.2 HBV Entry Inhibitors

Aside from the Pol, another interesting antiviral target is viral entry. In fact, one of the most clinically advanced non-NA therapies in development is Myrcludex B (MYR Pharma/Hepatera), which currently has ongoing Phase 2b [20] and Phase 3 [21] studies in patients with chronic hepatitis delta disease. Myrcludex B [22–25] is a synthetic myristoylated peptide that mimics the binding of HBV to its cellular receptor, sodium taurocholate cotransporting polypeptide (hNTCP), and stably blocks entry of HBV and the virusoid hepatitis delta virus (HDV) into hepatocytes [26, 27]. Myrcludex B, with a median inhibitory concentration of ~80 picomolar, was developed from a consensus sequence of residues 2–48 of the preS1 domain of the large HBV surface protein [28]. In particular, studies have shown that a well-conserved motif [residues 9-NPLGF(F/L)P-15] in this preS1 domain of human and primate hepadnaviruses is required for the binding of HBV to hNTCP during viral entry, and that the surrounding residues 2–8 and 18–48 increased its viral inhibition activity, while residues 49–78 decreased it [29, 30]. Recently, Donkers et al. [31] have also shown that although Myrcludex B binding to hNTCP is very strong, Myrcludex B can also transfer to newly synthesized hNTCP molecules, possibly extending its viral blocking efficacy.

Aside from Myrcludex B, other small molecules, including the calcineurin inhibitor cyclosporine A (CsA) [32], the angiotensin II receptor blocker irbesartan [33], the cholesterol absorption inhibitor ezetimibe [34], the HIV protease inhibitor ritonavir [35], vanitaracin A [36], and the immunosuppressant macrolide compound rapamycin, and its derivatives everolimus and temsirolimus [37] have also been shown to interact with the hNTCP receptor, thereby preventing the binding of the preS1 domain to the receptor and inhibiting viral entry. All these compounds, including Myrcludex B, however, also impair the sodium-dependent bile uptake function of hNTCP, which can lead to adverse effects [32–37]. Shimura et al. [38] have recently identified CsA derivatives that have sub-micromolar activity against HBV, do not have immunosuppressive activity, and do not impair sodium-dependent bile uptake. Their findings suggest that the anti-HBV activity of entry inhibitors could be functionally separated from bile acid transport, leading to better safety profiles for this class of inhibitors.

9.3 HBV Capsid Assembly Modulators (CAMs)

Another class of direct-acting antivirals with several compounds in clinical development are the CAMs. These are small molecules that act on the HBV core protein and capsid.

9.3.1 HBV Core Protein

HBV has a partially double-stranded, relaxed circular DNA genome (rcDNA) of approximately 3200 bp. It encodes four overlapping open reading frames (ORFs) [39] including the preC/C ORF which encodes the ~25 kDa pre-core protein (HBeAg) precursor and the ~21 kDa core protein (HBc, Cp) [40]. The latter is translated from the second start codon, which is 30 codons downstream from the pre-core start codon [40, 41].

The Cp is the building block of the icosahedral viral capsid; it protects and packages the viral genome. Cp has two domains: an α -helix-rich N-terminal domain composed of

residues 1-149 (Cp149, 16.8 kDa), and an intrinsically disordered arginine-rich C-terminal domain (CTD, 4.3 kDa) composed of the remaining 34-36 residues (variations occur between different genotype sequences) [41, 42]. Cp149 forms the outer shell of the capsid and is crucial for capsid assembly. Cp149 begins with an irregular N-terminal structure, followed by α -helices $\alpha 1$ (residues 13–17) and $\alpha 2$ (residues 27–43) that has a kink from residue 37. Its structure is dominated by a long α -helical hairpin composed of α 3 (residues 50–74) and α 4 (residues 79–109), with a kink between residues 89 and 92. It ends with α -helix α 5 (residues 113–126), an irregular proline loop (residues 128–136) and an extended strand (residues 137-142) (see Figure 9.1). The Cp149 monomer contains a hydrophobic core comprised of residues from $\alpha 1$, the loop connecting $\alpha 1$ and $\alpha 2$, $\alpha 4b$ (i.e. the part of $\alpha 4$ after the kink) and α 5, which stabilizes the monomer fold [43]. On the other hand, the CTD, located within the capsid, is important for encapsidation of the pre-genomic RNA (pgRNA) and its subsequent reverse transcription into the rcDNA genome, and for binding of the Cp to host proteins. Its function is mediated by four arginine-rich repeats punctuated with seven conserved serine residues and one threonine that can be phosphorylated [44]. In addition to its phosphorylation, the progressive dephosphorylation of the CTD as rcDNA synthesis proceeds may also contribute to the maturation signals necessary for the envelopment of the mature nucleocapsid [45, 46].

In general, capsids can assemble into two sizes: the majority (>95%) observed in natural infection and in vitro have a T = 4 symmetry (36 nm-diameter) composed of 120 dimers, while a small portion have a T = 3 symmetry (32 nm-diameter) composed of 90 dimers [41, 47–49]. The asymmetric unit of a T = 4 capsid has four quasi-equivalent subunits: A, B, C, and D, which form chemically identical, but structurally distinct AB and CD dimers which undergo conformational changes to reach an assembly-competent state. The intradimer interface is composed of a four-helix bundle corresponding to the highly immunogenic protruding spikes observed on the capsid [43, 44, 50, 51]. Different interactions between the AB and CD dimers form the 5-fold, 2-fold (or quasi-6-fold), 3-fold and quasi-3-fold icosahedral vertices. It is at these vertices that the capsids are fenestrated, allowing access of nucleotides and other molecules into and out of the capsid for rcDNA synthesis and CTD accessibility. The formation of the resulting interdimer interfaces is predominated by the burial of surfaces, which are approximately 75% hydrophobic, supporting an entropy-driven assembly. The resulting stable capsid is mainly held together by weak hydrophobic interactions at the interdimer contacts [43, 51, 54].

In infected cells, capsid assembly occurs in the cytoplasm and is initiated by interactions of the entire Cp with the ε signal on the pgRNA and the Pol [52]. Morphologically intact capsids can also be produced in a test tube by manipulating protein concentration, temperature, pH, and ionic strength. Capsid assembly begins with a slow nucleation step through the formation of a trimer of dimers, followed by rapid elongation by addition of one dimer at a time [41, 44, 53, 54]. Based on the property of spontaneous self-assembly, Zlotnick and his team were able to set up a target-based fluorescence quenching assay for the identification and characterization of both misdirectors and inhibitors of the HBV capsid assembly process. A mutant Cp150, in which all native cysteines dispensable for capsid assembly were mutated to alanines and an additional unique cysteine-reactive agents like BODIPY-FL maleimide. The resulting dimeric Cp150-BODIPY-FL is nearly

quantitatively labeled and highly fluorescent, but a significant reduction in fluorescence is observed upon assembly of these dimers into capsids [55].

9.3.2 First-Generation CAMs

So far, several classes of inhibitors of pgRNA packaging and HBV capsid assembly have been identified, either in cell-based or target-based high-throughput screening (HTS) campaigns or in rational drug design programs (see Figure 9.2). They act by accelerating nucleocapsid assembly, by selectively inhibiting or dysregulating pgRNA encapsidation, or by both [41, 44]. For this reason, these HBV inhibitors are referred to as capsid assembly modulators (CAMs), core protein allosteric modulators (CpAMs), or capsid assembly effectors (CAEs).

The first capsid inhibitior to be identified altogether, a CAM chemotype, was the heteroaryldihydropyrimidine (HAP), BAY 41-4109 [56]. It was discovered in a cell-based HTS campaign. BAY 41-4109 (also AIC429 see compound 1 in Figure 9.3), is a first-generation HAP compound. It has been shown to inhibit HBV DNA replication with an EC_{50} of 54 nM and to reduce the intracellular amount of capsids by reducing the half-life of the Cp in the HBV-expressing cell line HepG2.2.15 from >24 hours to 3 hours [56]. In addition, as it addresses a different target, it is active against NA-resistant HBV [56], and reduces viral DNA dose-dependently in the liver and in the plasma of HBV transgenic mice by preventing capsid formation [64]. Structural studies of this series of inhibitors revealed that they induce inappropriate capsid assembly at low concentrations, leading to degradation of capsid via the proteasome pathway [56], and, when in excess, promote a misdirected assembly reaction and decreased capsid stability [55, 65]. BAY 41-4109 and related compounds were originally invented at Bayer in the late 1990s. There is still ongoing chemical evolution of this scaffold, producing new entities like GLS4 [66, 67], HAP_ R10 [57], and compound 4 [68] (see compounds 2-4 in Figure 9.3), some of which are in clinical development (see Table 9.3). This enduring interest in HAPs is remarkable, and is based on the fact that among the CAM chemotypes discovered so far, this scaffold revealed a special and unique mode of action (see Section 9.3.4)

A few years before the HAP scaffold and its mode of action was published, the phenylpropenamide (PPA) derivative AT-61 was identified by Avid Therapeutics (followed up



Figure 9.2 HBV core protein assembly domain. (a) Cp149 monomer viewed from the intradimer interface. The monomer is composed of five helices, indicated as $\alpha 1 - \alpha 5$, which are connected by loops. The N- and C-terminal are indicated with the letters N and C, respectively. (b) Cp149 dimer is composed of two quasi-equivalent subunits with the majority of the contact points being at $\alpha 3$ from both subunits (PDB: 1QGT) [43, 50].



BAY 41-4109 Bayer/AiCuris EC₅₀ 54 nM (HepG2.2.15)



5 AT-130 Avid Therapeutics/Gilead EC₅₀ 130 nM (HepAD38)



9 Example 2.20, WO2019/166951 Novartis EC₅₀ <100 nM (HepG2.cl42)



13 NVR 3-778 Novira Therapeutics EC_{en} 340 nM (HepG2.2.15)



2 GLS4 HEC Pharm EC 12 nM (HepG2.2.15)



6 JNJ-6379, Example 102, WO2014/184350 Janssen Sciences EC₅₀ 60 nM (HepG2.2.15)

10

Example 373, WO2019/086141

AiCuris Anti-infective Cures EC₅₀ < 1000 nM (HepAD38)

14

N7-4

Chinese Academy of Sciences EC₅₀ 20 nM (HepAD38)

17 Phenyl piperidine-3-carboxamide



MeO.

со н



11

Example 2, WO2019/020070

Shanghai Hengrui Pharmaceuticals EC₅₀ 30 nM (HepG2.2.15)

15

PT, Example, 917 WO2016/109689

Novira Therapeutics EC₅₀ 20 nM (HepAD38)



Example 53c, WO2019/214610

Janssen Sciences EC₅₀ <5 nM (HepG2.2.15)

Me ОН N

8 Example 64, WO2017/156255 Aligos Therapeutics EC₅₀ <1 nM (HepAD38)



12 Example 2, WO2019/020070 Shanghai Hengrui Pharmaceuticals EC₅₀ 30 nM (HepG2.2.15)



16 DBT, Example 1588, WO2015/138895 Assembly Biosciences EC₅₀ 140 nM(HepAD38)





Tsinghua University Emory / Missouri University EC_{so} 20 nM (HepAD38) EC₅₀ 3600 nM (HepAD38) Figure 9.3 Structures of different CAMs. Antiviral activity data (EC₅₀s) were taken from the patents mentioned, except for the following references: BAY 41-4109 [56], GLS4 [42], HAP R10 [57], AT-130

[58], NVR 3-778 [59], NZ-4 [60], PT [61], DBT [61], phenyl piperidine-3-carboxamide [62], and HF9C6 [63], respectively.

later by Gilead) as a small molecule inhibitor of HBV replication in HBV-expressing cell lines [76]. Although the mode of action was not clear at the time of its discovery, it was demonstrated in 2007 [77] that AT-130 (see compound 5 in Figure 9.3), a related analog of AT-61 [58], selectively inhibits viral pgRNA packaging and that PPAs, like HAPs, are active against both wild-type and nucleos(t)ide analog (NA)-resistant strains of HBV [77-79].

9.3.3 Recent Progress in Novel CAMs

Beyond these two CAM chemotypes, several new CAM chemical scaffolds have meanwhile been developed, some of which are also in clinical trials. Advances in the medicinal chemistry of therapeutic agents for the treatment of chronic hepatitis B, including CAMs, have been described in two review articles [80, 81]. Another recent publication of significant note is the disclosure of the structure of Janssen's sulfamoyl-pyrrolamide clinical candidate JNJ-6379 [82, 83]. Closely related pyrrolo-oxalamide analogs have also been described by Gilead Sciences, and by Aligos Therapeutics/Emory University (compounds 7 and 8 in Figure 9.3, respectively) [84–86]. Compound 8 is believed to be Aligos Therapeutics' clinical candidate ALG-001075.

Indole-2-carboxamides such as compounds 9 and 10 (Figure 9.3) have also been disclosed as a further class of CAMs [87–89]. Interestingly, the indole-2-carboxamide unit acts as an isosteric replacement for the more frequently encountered phenylurea (e.g. compounds 11 [90] and 12 [91] in Figure 9.3) or arylamide substructures (e.g. compounds 7 and 8 in Figure 9.3). The aminothiazole hydrogen bond donor-acceptor pair of compound 10 is isosteric to the oxalamide unit of compounds such as compound 7, while compound 12 shows that further core variation is also tolerated.

CAMs bind in a pocket formed at the interface of a pair of Cp dimers (Figure 9.4). The binding pocket itself is predominantly comprised of nonpolar amino acids, with a hydrophobicity ratio of 0.58. Conserved ligand pharmacophore features are typically a hydrogen bond acceptor (paired with the side chain NH of W102, shown in sticks) and an aromatic ring (typically halogenated) filling the pocket formed by V124, R127 and T128 of one Cp monomer, and P25, D29, L30, T33, W102, I105, and S106 of a second Cp monomer. Many compounds also have an H-bond donor paired with the side chain of T128. An H-bonding interaction between ligand and S141 (side chain shown in sticks) either directly or mediated via a water molecule is also frequently observed.



Figure 9.4 Crystal structure of a HAP molecule binding at the Cp dimer-dimer pocket found at the C-D' interface (PDB: 5GMZ). Inset: structure of the HAP molecule. [92].

9.3.4 Modes of Action of CAMs

Because of the crucial function of the Cp and capsids in different stages of the HBV replication cycle, studies suggest that CAMs, which target both Cp and capsid, can combat HBV replication and persistence on different levels.

CAMs mainly prevent HBV replication in already established infections by directly interfering with pgRNA encapsidation by increasing rate and extent of assembly (i.e. mistimed assembly), resulting in the formation of empty capsids or aberrant structures [60, 65, 70, 93–95]. Based on size exclusion chromatography and electron microscopy analyses, this effect on capsid assembly can be categorized into two types. HAPs (see compounds 1–4 in Figure 9.3), the prototypic chemical scaffold having the eponymous "HAP-like" mode of action, induce the formation of aberrant structures [61, 71, 96, 97], leading to degradation of Cp with massive reduction in the halflife of the protein [56]. They transiently stabilize the Cp assembly-active state [50, 98], and increase the thermodynamic stability of the interdimer contact points [99], resulting in a change in the quaternary structure of the capsid [71, 100]. Recently, Kang et al. [101] showed that the FDA-approved antifungal drug Ciclopirox also acts on the HBV capsid and induces the formation of aberrant structures.

On the other hand, the formation of morphologically intact but empty capsids is referred to as the "non-HAP-like" mode of action, and is exhibited by compounds, such as PPAs [93], sulfamoylbenzamides (SBAs) and benzamides [70, 96], isothiafludine (NZ-4) [60], glyoxamide-pyrrolamides (GPAs) [61], pyrazolyl-thiazoles (PTs) [61], dibenzo-thiazepin-2-ones (DBTs) [61], phenylpiperidine-3-carboxamides [62], and HF9C6 [63] (see compounds 5–8, 13–18 in Figure 9.3). Katen et al. [93] have shown that the empty capsids were not a result of direct interference with pgRNA encapsidation, but rather a consequence of the faster assembly kinetics. In particular, they suggest that PPA could indiscriminately induce assembly nucleation regardless of the presence of pgRNA and Pol. However, this may not be the case for all these CAMs as Yang et al. [60, 102] have reported that NZ-4 may directly interfere with the interaction between pgRNA and the Cp, possibly at the C-terminal, arginine-rich nucleic acid binding domain of the Cp. As a consequence of the production of empty capsids by "non-HAP-like" CAMs, Lam et al. [59] have shown that SBA can reduce the secretion of DNA- and RNA-containing viral particles, while NAs prevent production of DNA-containing viral particles but increase the production of RNA-containing particles. However, when cells are treated with SBA and an NA, better viral suppression and fewer DNA- and RNA-containing viral particles are observed [59].

As discussed, aside from their biophysical effects on capsid assembly, HAPs also cause a strong proteasome-mediated reduction in both intracellular Cp and capsid amounts [56], whereas "non-HAP-like" CAMs, such as the SBAs, do not alter the absolute amounts of intracellular Cp and capsid particles [61, 71, 103]. In addition, treatment of HBV-producing cells with HAPs result in a time- and dose-dependent localization of punctate core antigen aggregates as visualized through immunofluorescence staining. At low concentrations (\leq 1.5-fold EC₅₀) and short treatment durations (\leq 3 days), core antigen aggregates are mainly found in the cytoplasm, with some located near the nucleus and enclosed by invaginations of the nuclear envelope [104, 105]. With higher concentrations (\geq 1.5-fold EC₅₀) and longer

treatment durations (\geq 4 days), the aggregates are mainly associated with the nucleus [61, 95, 106]. Huber et al. [106] and Lahlali et al. [95] have shown that the HAP-induced nucleus-associated core antigen aggregates are co-localized with the promyelocytic leukemia protein and 20S proteasomal subunit, which are sites of protein post-translational modification, activation, sequestration, and degradation. HAP-induced association of the Cp with these bodies may lead to different pleiotropic effects, such as induction of apoptosis or cellular senescence [106].

On the other hand, Corcuera et al. [61] and Huber et al. [63] have shown that the "non-HAP-like" CAMs induce the trapping and accumulation of the HBV core antigen signal in the cytoplasm, with virtually no apparent nuclear staining. Loss of the core antigen in the nucleus may affect maintenance of active covalently closed circular DNA (cccDNA) transcription since Cp binding to the minichromosome may alter nucleosome spacing [107], and in particular, affect histone acetylation and DNA methylation by interacting with CpG island 2 on the minichromosome [108]. Indeed, Lam et al. [59] have recently shown that treatment of primary human hepatocytes (PHHs) with SBA prior to infection resulted in inhibition of viral RNA transcription, including those for HBsAg and HBeAg, reducing their protein levels as well. Another study has shown that Cp in the nucleus may inhibit double-stranded RNA-mediated interferon response either by directly binding to the promoter regions of interferon-stimulated genes or recruitment of epigenome-modifying enzymes [109]. As a result, loss of Cp in the nucleus may remove this block on the interferon response.

Interestingly, CAMs can also act at an earlier stage of the HBV replication cycle, by inhibiting the establishment of the cccDNA pool and its maintenance via the recycling pathway. Several studies have illustrated that when CAMs were administered before in vitro infection, a reduction in cccDNA levels is observed, though this requires higher CAM concentrations and increased treatment duration compared to the primary CAM mechanism of action, i.e. interference with pgRNA encapsidation [96, 103, 110]. In particular, Guo et al. [110] have shown that for HAPs and SBAs, this effect on the cccDNA pool is possibly mediated by inducing the disassembly of nucleocapsids from de novo infection and the recycling pathway prior to delivery of the rcDNA genome to the nucleus. Their studies suggested that the intrinsic instability of mature nucleocapsids, resulting from the presence of the partially double-stranded rcDNA [111, 112], confers changes on the nucleocapsid structure which allow CAMs to bind to it and trigger its disassembly [110].

With this multi-pronged attack against HBV, CAMs have great potential to be one of the key players that can bring about HBV cure. The precise molecular (dynamic) basis for the differential effects on capsid morphology of the two CAM types remains unclear, as does the clinical relevance of this phenomenon for an eventual HBV cure [71].

9.3.5 Resistance Profile of CAMs

CAMs are generally active against the eight well-studied HBV genotypes (A-H) [59, 70, 72]. Analysis of the 10975 deposited Cp sequences in the Hepatitis B virus database (accessed 24 October 2019) [69] has shown that all the residues at the CAM binding pocket are generally well-conserved among the eight genotypes. In addition, CAMs are also active against NA-resistant HBV strains since they interfere with a different viral target [42, 56, 70, 79, 94].

Although CAMs are hypothesized to have a higher barrier to resistance than NAs [113, 114], studies of their resistance profile are still of great interest, not just in terms of resistance emergence, but also as a tool to better understand the interactions of different CAMs in the pocket and to guide the design of next-generation CAMs. Generally, in order to study drug-resistant virus mutants, sequential passage of the virus in cells in the presence of increasing compound concentrations is performed [115]; however, such a technique has not been developed successfully for HBV. Instead, transient or stable transfection of full-length HBV PCR amplicons or plasmids containing 1.1× to 2× HBV genome units into hepatoma cell lines (e.g. HepG2 or Huh7) is used. HBV replication is then initiated through synthesis of the pgRNA by the inherent HBV promoter (in plasmids with 1.3-2 genome units) or by a strong mammalian promoter, such as CMV (in plasmids with 1.1 genome units) [116]. Several groups have made use of these transient HBV replication systems to study a plethora of amino acid substitutions at the CAM-binding pocket, some of which are listed in Table 9.2 [59, 70-75]. Because of the indispensable role of the Cp and capsid in the replication cycle, all mutants studied have replication capacities similar to, or less than, the wild-type virus [71-75]. It should be noted, however, that the in vivo selection of mutations may not be identical to the in vitro results [115], particularly if quite an artificial system has to be used, as is the case for HBV.

Among the mutations, of particular interest are the substitutions at Cp residues 25 (in loop 2), 33 (in helix 2), and 124 (in helix 5) because of their varied effects on compound activity. The P25G substitution is highly resistant to all CAMs tested, while the P25A substitution is only moderately resistant to SBA and highly resistant to the HAPs. Interestingly, a change to serine results in a high resistance against HAPs, moderate resistance against SBA, and an increased susceptibility against JNJ-6379 [71, 74]. Zhou et al. [71] suggest that the alanine and serine substitutions could possibly change the conformation of loop 2, causing a disadvantageous interaction with the thiazole group of HAPs, which is absent from SBAs. On the other hand, the smaller glycine residue may create more space in the pocket, thereby weakening the van der Waals interaction between both types of CAMs tested and the Cp, resulting in reduced activity. Similarly, a change of residue T33 to one with a bulkier side chain, such as asparagine or glutamine is suggested to cause a steric hindrance in the pocket, resulting in the observed increased resistance to all CAMs tested; this effect is reduced when the residue is replaced with the slightly smaller serine [71].

Studies have shown that position 124 is crucial to capsid assembly as both pgRNA encapsidation and plus-strand DNA synthesis were impaired when valine was substituted with alanine, leucine, phenylalanine, or tryptophan [75, 117]. In addition, as this residue is located in the wall of the pocket [71], Tan et al. [117] have shown that increasing the hydrophobic surface area with V124X substitutions increases the association energy, and fills the pocket to different extents. Changing the valine to residues with a smaller side chain, such as alanine or glycine, results in resistance to SBAs and JNJ-6379, as the pocket is opened up, possibly weakening the interactions between the compound and the protein. Meanwhile, with this substitution, susceptibility to HAPs is maintained, possibly because of its slightly larger size and ability to fill the pocket [70, 74, 75]. On the other hand, a small increase in side chain volume with an isoleucine substitution results in a more favorable fit in the pocket and an increased susceptibility for HAPs and SBA [71]. Further increasing the size of the side chain with phenylalanine renders the mutant resistant to the slightly larger HAPs, but still susceptible to SBAs [70, 71]. Filling the pocket completely with a tryptophan

Table 9.2 Effect of amino acid substitutions in the CAM binding pocket on CAM activity.

		Susceptibility to CAMs ^a						
		HAP-like CAMs				Non-HAP-like (Non-HAP-like CAMs	
Amino acid Position substitution	Frequency in Cp sequences (Genotypes A–H) [69]	BAY 41-4109 [59,70-74]	GLS4 [75]	NVR-010-001- E2 ^b [73]	HAP_ R01 ^c [71]	JNJ-6379 [74]	NVR 3-778 [59, 71]	
F24L	0.082%	HR	-	-	-	S	-	
F24Y	1.084%	S	-	-	-	S	-	
P25A	0.009%	HR	-	-	HR	-	MR	
P25G	0.000%	HR	-	-	HR	-	HR	
P25S	0.055%	HR	-	-	HR	HS	MR	
D29G	0.146%	HR	-	-	-	MR	-	
L30F	1.276%	HR	-	-	-	HR	-	
T33N	0.018%	HR	-	-	HR	HR	HR	
T33P	0.009%	HR	-	-	-	HR	-	
T33Q	0.000%	HR	-	-	HR	-	HR	
T33S	1.257%	MR	-	-	S	MR	MR	
L37Q	0.027%	MR	-	-	-	HR	-	
Y38F	3.189%	S	-	-	-	S	-	
Y38H	1.175%	S	-	-	-	HS	-	
I105L	0.784%	HS	-	HS	-	HS	S	
I105T	0.656%	S	-	S	-	MR	HR	
I105V	1.130%	S	-	S	-	S	S	
S106T	0.073%	HS	-	-	-	MR	-	
	Amino acid substitution F24L F24Y P25A P25G D29G L30F T33N T33Q T33Q T33Q T33Q L37Q Y38H I105L I105V S106T	Amino acid substitution Frequency in Cp sequences (Genotypes A-H) [69] F24L 0.082% F24Y 1.084% P25A 0.009% P25G 0.000% P25S 0.055% D29G 0.146% L30F 1.276% T33P 0.009% T33Q 0.000% T33Q 0.000% T33S 1.257% L37Q 0.027% Y38F 3.189% Y38H 1.175% I105T 0.565% I105V 1.130% S106T 0.073%	Amino acid substitution Frequency in Cp sequences (Genotypes A-H) [69] BAY 41-4109 [59,70-74] F24L 0.082% HR F24Y 1.084% S P25A 0.009% HR P25G 0.000% HR D29G 0.146% HR L30F 1.276% HR T33N 0.018% HR T33Q 0.009% HR T33Q 0.027% MR L37Q 0.027% MR L37Q 0.27% S I105L 0.784% S I105Y 1.130% S I105Y 1.130% S	Susceptibility	Susceptibility-CAMs ³ Amino acid substitution Frequency in Cp sequences (Genotypes A-H) [69] Superior (Sep, 70-74) NR-010-001- (Sep, 70-74) F24L 0.082% BAY 41-4109 (Sep, 70-74) RLS4 [75] NVR-010-001- (Sep, 70-74) F24L 0.082% HR - - F24X 1.084% S - - F25A 0.009% HR - - F25G 0.000% HR - - F25G 0.055% HR - - F33N 0.146% HR - - F33N 0.018% HR - - T33Q 0.009% HR - - T33Q 0.027% MR - - <	Susceptibility-CAMS*Amino acid substitutionFrequency in Cp sequences (Genotypes A-H)[69]BAY 41-4109 (S9,70-74)NVR-010-01 (LS4[75)MAP_ R010-011F24L0.082%RIRF24L0.082%KIRF24X1.084%SF25A0.009%HRHRP25G0.000%HRHRP25G0.055%HRHRP25G0.146%HRHRP25G0.146%HRT30F0.146%HRT33P0.009%HRT33Q0.009%HRT33Q0.009%HRT33Q0.009%MRT33Q0.009%MRT33Q0.009%MRT33Q1.25%MRT33Q1.25%MRT33Q1.25%MRT33Q1.25%MRT33Q1.25%ST33Q1.25%MRT33Q1.25%MRT33Q1.25%S <tr< td=""><td>Susceptibilization Susceptibilization Amino acid substitution Frequency in Cp sequences (Genotypes A-H) [69] BAY 41-4109 [59,70-74] NR-010-001 E2⁶ [73] HAP_ RD² [71] Nu-HAP-like CA Paper Sequences (Genotypes A-H) [69] BAY 41-4109 [59,70-74] NR-010-001 E2⁶ [73] HAP_ RD² [71] Nu-HAP-like CA Paper Sequences (Genotypes A-H) [69] BAY 41-4109 [59,70-74] Nu-HAP-like CA Paper Sequences (Genotypes A-H) [69] Nu-HAP-like CA Paper Sequences (Genotypes A-H) [69] Nu-HAP-like CA Paper Sequences (F12] Nu-HAP-like CA Paper Sequences (F12] Nu-HAP-like CA Paper Sequences (F12]</br></br></br></br></br></td></tr<>	Susceptibilization Susceptibilization Amino acid substitution Frequency in Cp sequences (Genotypes A-H) [69] BAY 41-4109 [59,70-74] NR-010-001 E2 ⁶ [73] HAP_ RD ² [71] Nu-HAP-like CA Paper Sequences (Genotypes A-H) [69] BAY 41-4109 [59,70-74] NR-010-001 E2 ⁶ [73] HAP_ RD ² [71] Nu-HAP-like CA Paper Sequences (Genotypes A-H) [69] BAY 41-4109 [59,70-74] Nu-HAP-like CA Paper Sequences (Genotypes A-H) [69] Nu-HAP-like CA 	

Susceptibility to CAMs^a

HAP-like CAMs

Non-HAP-like CAMs

Position	Amino acid substitution	Frequency in Cp sequences (Genotypes A–H) [69]	BAY 41-4109 [59, 70-74]	GLS4 [75]	NVR-010-001- E2 ^b [73]	HAP_ R01 ^c [71]	JNJ-6379 [74]	NVR 3-778 [59,71]
109	T109A	0.064%	HS	-	-	-	HS	-
	T109I	1.367%	HR	HR	HR	-	HS	S
	T109M	0.738%	MR	S	MR	-	S	S
	T109S	0.155%	HS	HS	HS	-	S	MR
110	F110I	0.009%	HS	-	-	-	HR	HR
118	Y118F	0.383%	HR	-	MR	-	HR	HR
124	V124A	0.036%	S	HS	-	-	-	-
	V124F	0.009%	HR	-	-	HR	-	S
	V124G	0.064%	S	S	-	-	HR	-
	V124I	0.036%	HS	-	-	HS	-	HS
	V124W	0.219%	HR	-	-	-	-	-
125	W125F	0.000%	S	-	-	-	HS	-
127	R127H	0.064%	HR	-	-	-	MR	-
128	T128I	0.036%	HS	-	-	-	HR	-
132	Y132F	0.027%	HS	-	-	-	S	-

^a Susceptibility to CAMs based on fold-change (FC) in activity compared to wild-type virus. HS: highly susceptible (FC ≤ 0.5), S: susceptible (0.5 < FC < 2), MR: moderately resistant ($2 \le FC < 5$), HR: highly resistant (FC ≥ 5); : no data available ^b NVR-010-001-E2 is a methyl analog of GLS4. It has an EC₅₀ of 11 nM in HepG2.2.15 cells [73]. ^c HAP_R01 is an analog of HAP_R10, with the morpholine replaced with a 3,3-difluorpyrrolidine. It has an EC₅₀ of 3 nM in HepG2.2.15 cells [57].

242 9 Direct-Acting Antivirals for the Treatment of Chronic Hepatitis B Virus (HBV) Infection

residue results in a mutant with faster assembly kinetics and stronger association energy, mimicking CAM-induced assembly kinetics in the absence of CAMs [99]. Because of this steric block, V124W has been shown to be resistant to both "HAP-like" and "non-HAP-like" CAMs [70, 99, 100].

The differences in the cross-resistance profiles of the "HAP-like" and "non-HAP-like" CAMs, the absence of significant cross-resistance between CAMs and NAs in the transient replication assays [59, 71, 72, 74], and the observed additive and synergistic effects with other therapies (e.g. NAs [59], an RNAseH inhibitor [118], or a TLR7 agonist [119]) provide a good basis for the use of CAMs in combination therapy, hopefully leading to a cure. Potentially, a combination of HAP-like CAMs and CAMs with the other mode of action would result in an unsurmountable barrier to replication for HBV.

9.3.6 CAMs in Clinical Trials

As discussed above, the HBV Cp and its assembly is a promising drug target to achieve HBV cure, because it is involved in several key aspects of the viral replication cycle and it has no human homologs [56, 120], in contrast to modes of action involving e.g. polymerases or proteases. Consequently, several CAMs are already being tested in clinical trials (see Table 9.3). The SBA NVR 3-778 [59] (see Figure 9.1) was the first CAM to be evaluated in HBV-infected patients. In a 28-day trial, a mean reduction in serum levels of HBV DNA of 1.43 log10 IU/ml and 1.97 log₁₀ IU/ml was observed at a dose of 600 mg of NVR 3-778 given twice daily in monotherapy or in combination with pegIFN, respectively. NVR 3-778 was generally well-tolerated, and, unlike NAs, also had a clear effect on HBV RNA suppression, validating the idea of supplementing NA therapy with CAMs in order to improve viral control [121]. As of writing, no further updates have been provided regarding further development of this drug.

Compound code	Company	Phase of development	Mode of action
NVR 3-778	Novira Therapeutics/Janssen Pharmaceutica	Ph 2	Non-HAP like
$GLS4 (+RTV)^a$	HEC Pharm	Ph 2	HAP like
ABI-H0731	Assembly Biosciences	Ph 2	Non-HAP like
JNJ-(5613)6379	Janssen Pharmaceutica	Ph 2	Non-HAP like
RO7049389	Hoffmann-La Roche	Ph 2	HAP like
QL-007	Qilu Pharmaceutical	Ph 2	Not disclosed
AB-506	Arbutus Biopharma	Ph 1 (discontinued)	Non-HAP like
ABI-H2158	Assembly Biosciences	Ph 1	Non-HAP like
EDP-514	Enanta Pharmaceuticals	Ph 1	Not disclosed

 Table 9.3
 Capsid assembly modulators in clinical development.

a) Ritonavir (RTV) was originally discovered as an HIV protease inhibitor with antiretroviral activity. Now, it is widely used as a "PK booster" of other drugs, since it inhibits cytochrome P450-3A4 (CYP3A4).

GLS4 (see Figure 9.1), a second-generation CAM of the HAP chemotype, is also under clinical investigation. It had high antiviral activity in HBV-expressing cell lines, but showed a significant reduction in efficacy when tested in metabolically active, HBV-infected PHH [66, 67]. The reason for this activity drop is its low metabolic stability. Consequently, the efficacy of the combination of 120 mg of GLS4 and 100 mg of ritonavir, a widely used CYP3A4 inhibitor, is currently being evaluated in Phase II clinical trials [122]. Before, BAY 41-4109 (AIC429) had been tested in a Phase I single dose trial with good tolerability at single doses of 2.5–300 mg, without reaching the maximum tolerated dose at the highest dose administered. However, due to preclinical data, the compound was later given up.

Since the first proof of concept study with NVR 3-778, several so-called second-generation CAMs are currently under investigation in Phase II clinical studies. However, the clinical development for some of these pipeline drugs had to be stopped due to severe side effects as it has been reported for AB-506 more recently. Nevertheless, very potent third-generation CAMs, such as ABI-H2158 and EDP-514, have just entered clinical development [123] (see Table 9.3). In 2019, Assembly Biosciences presented interim results from three ABI-H0731 and NA Phase IIa combination studies. At treatment week 24 of the study involving HBeAg-positive, treatment-naïve patients, the combination group, which received 300 mg ABI-H0731 with 0.5 mg ETV q.d., had a more pronounced HBV DNA decline (-5.30 log₁₀ IU/ml) compared to the ETV monotherapy group $(-4.19 \log_{10} IU/ml)$ (p = 0.0452), which received only 0.5 mg ETV q.d. In addition, HBV pgRNA was reduced from baseline in the combination group by $2.34 \log_{10} IU/ml$, whereas in the ETV monotherapy group it was only reduced by 0.61 \log_{10} IU/ml (p < 0.001). This long-term (24 weeks) decline in HBV pgRNA is believed to reflect the reduction in cccDNA pools. Additionally, at treatment week 48, it was observed that the reductions from baseline in mean HBV DNA and pgRNA were 6.3 and 3.0 \log_{10} IU/ml, respectively, for patients treated with the ABI-H0731 + ETV combination. This second phase declines in pgRNA were strongly associated with reductions in viral antigens HBeAg and HBcrAg (surrogate markers for cccDNA) [124]. In the same year, Janssen Pharmaceuticals presented data from the first triple-combination trial, showing not only a robust HBV DNA and pgRNA decline to below the lower limit of quantification for most patients, but also an HBsAg reduction $\geq 1.0 \log_{10} IU/ml$ for all patients who received the cocktail of JNJ-3989 together with the RNAi drug JNJ-6379 and an NA [125].

These encouraging data give hope for the concept of CAMs to be part of a future combination regimen for HBV cure. Challenges in the development of more efficacious CAMs lie in the identification of more potent compounds with superior pharmacokinetic properties in order to facilitate lower clinical dosing, anticipating the eventual necessity for a multicomponent HBV cure regimen. Furthermore, there is also significant potential in targeting hepato-selective active uptake mechanisms (such as OATP1B1 and OATP1B3) to drive increased exposures in the liver and minimize toxicity and systemic side effects. Moreover, drug resistance in HBV therapy with non-NA direct acting antivirals is currently not well explored and could also be one driver for future CAM development. Further clinical studies with longer treatment durations will give more insights into the emergence of HBV drug resistance on CAMs.

9.4 Concluding Remarks

The field of targeted antiviral drug discovery for the treatment of chronic HBV infection is large and highly dynamic. After completing this manuscript in Spring 2020, improvements of existing drug programs, as well as the emergence of new direct-acting antiviral programs, have been published. The entry inhibitor Hepcludex formerly known as Myrcludex B (MYR GmbH, now part of Gilead) was conditionally approved in Europe for the treatment of chronic HDV. Several new CAM pipeline drugs have been announced, namely ABI-H3733(Phase I, Assembly BioScience), JNJ-64530440(Phase I, Janssen Pharmaceutica), ALG-000184 (Phase I, Aligos Therapeutics), and ZM-H1505R (Phase I, ZhiMeng Biopharma), respectively. For some of these new pipeline drugs, promising in vitro data showing sub-nanomolar EC₅₀ values in HBV-expressing cells have already been reported. For the advanced CAM drug programs, various triple-drug combination studies (e.g., CAM plus NA plus RNAi) have been started to prove whether CAM drugs can improve the cure rates. Unfortunately, a simpler two-drug combination Phase II trial, exploring whether meaningful sustained virologic response could be achieved after discontinuing therapy in virologically suppressed patients who had received the CAM drug ABI-H0731 (vebicorvir) and an NA for 12 to 18 months, failed as 39 out of 41 patients have relapsed (Press Release from Assembly Bioscience on Nov 5, 2020). The review article [126] will give a substantial update on the recent progress in the field of targeted anti-HBV drug discovery.

References

- 1 De Clercq, E. (1994). Antiviral activity spectrum and target of action of different classes of nucleoside analogues. *Nucleosides Nucleotides* 13 (6–7): 1271–1295.
- **2** Fung, J., Lai, C.-L., and Yuen, M.-F. (2006). Telbivudine: a new treatment option in the management of chronic hepatitis B. *Hep. B Annu.* 3 (1): 13–34.
- **3** Korba, B.E., Furman, P.A., and Otto, M.J. (2006). Clevudine: a potent inhibitor of hepatitis B virus in vitro and in vivo. *Expert Rev. Anti-Infect. Ther.* **4** (4): 549–561.
- **4** Bisacchi, G.S., Chao, S.T., Bachard, C. et al. (1997). BMS-200475, a novel carboxylic 2'-deoxyguanosine analog with potent and selective anti-hepatitis B virus activity in vitro. *Bioorg. Med. Chem. Lett.* 7 (2): 127–132.
- 5 Lai, C.L., Ahn, S.H., Lee, K.S. et al. (2014). Phase IIb multicentred randomised trial of besifovir (LB80380) versus entecavir in Asian patients with chronic hepatitis B. *Gut* 63 (6): 996–1004.
- **6** Terrault, N.A., Bzowej, N.H., Chang, K.M. et al. (2016). AASLD guidelines for treatment of chronic hepatitis B. *Hepatology* 63 (1): 261–283.
- **7** Lok, A.S., McMahon, B.J., Brown, R.S. Jr. et al. (2016). Antiviral therapy for chronic hepatitis B viral infection in adults: a systematic review and meta-analysis. *Hepatology* 63 (1): 284–306.
- 8 Park, J.Y., Kim, C.W., Bae, S.H. et al. (2016). Entecavir plus tenofovir combination therapy in patients with multidrug-resistant chronic hepatitis B: results of a multicentre, prospective study. *Liver Int.* 36 (8): 1108–1115.

- **9** Yuen, M.F., Ahn, S.H., Lee, K.S. et al. (2015). Two-year treatment outcome of chronic hepatitis B infection treated with besifovir vs. entecavir: results from a multicentre study. *J. Hepatol.* 62 (3): 526–532.
- **10** Ahn, S.H., Kim, W., Jung, Y.K. et al. (2019). Efficacy and safety of besifovir dipivoxil maleate compared with tenofovir disoproxil fumarate in treatment of chronic hepatitis B virus infection. *Clin. Gastroenterol. Hepatol.* 17 (9): 1850–1859 e1854.
- **11** Jia, F., Deng, F., Tong, S. et al. (2020). Efficacy of oral antiviral drugs to prevent mother-tochild transmission of hepatitis B virus: a network meta-analysis. *Hepatol. Int.* 14(3): 338–346.
- **12** WHO (2015). *Guidelines for the Prevention, Care and Treatment of Persons with Chronic Hepatitis B Infection.* Geneva: World Health Organization.
- Seifer, M., Hamatake, R.K., Colonno, R.J., and Standring, D.N. (1998). In vitro inhibition of hepdnavirus polymerase by the triphosphates of BMS-200475 and lobucavir. *Antimicrob. Agents Chemother*. 42 (12): 3200–3208.
- **14** Pierra Rouviere, C., Dousson, C.B., and Tavis, J.E. (2020). HBV replication inhibitors. *Antivir. Res.* 179: 104815.
- **15** Lok, A.S., Zoulim, F., Dusheiko, G., and Ghany, M.G. (2017). Hepatitis B cure: from discovery to regulatory approval. *Hepatology* 66 (4): 1296–1313.
- **16** Su, T.H. and Liu, C.J. (2017). Combination therapy for chronic hepatitis B: current updates and perspectives. *Gut Liver* 11 (5): 590–603.
- Ahn, S.H., Marcellin, P., Ma, X. et al. (2018). Hepatitis B surface antigen loss with tenofovir disoproxil fumarate plus peginterferon Alfa-2a: week 120 analysis. *Dig. Dis. Sci.* 63 (12): 3487–3497.
- 18 Tavis, J.E., Cheng, X., Hu, Y. et al. (2013). The hepatitis B virus ribonuclease H is sensitive to inhibitors of the human immunodeficiency virus ribonuclease H and integrase enzymes. *PLoS Pathog.* 9 (1): e1003125.
- **19** Tavis, J.E., Zoidis, G., Meyers, M.J., and Murelli, R.P. (2019). Chemical approaches to inhibiting the hepatitis B virus ribonuclease H. *ACS Infect. Dis.* 5 (5): 655–658.
- 20 ClinicalTrials.gov (25 Feb 2019). A Multicenter, Open-label, Randomized Phase 2b Clinical Study to Assess Efficacy and Safety of Bulevirtide in Combination With Pegylated Interferon Alfa-2a in Patients With Chronic Hepatitis Delta, NCT03852433. Bethesda, MD: National Library of Medicine (US).
- **21** ClinicalTrials.gov (25 Feb 2019). A Multicenter, Open-label, Randomized Phase 3 Clinical Study to Assess Efficacy and Safety of Bulevirtide in Patients With Chronic Hepatitis Delta, NCT03852719. Bethesda, MD: National Library of Medicine (US).
- ClinicalTrials.gov (14 Nov 2012). A Phase 1b/2a Randomized, Open-label Clinical Trial of Daily Myrcludex B Versus Entecavir in Patients With HBeAg Negative Chronic Hepatitis B, NCT02881008. Bethesda, MD: National Library of Medicine (US).
- ClinicalTrials.gov (13 Feb 2014). Randomized Open-label Substudy of Daily Myrcludex B Plus Pegylated Interferon-alpha-2a in Patients With HBeAg Negative Chronic Hepatitis B Co-infected With Hepatitis Delta, NCT02637999. Bethesda, MD: National Library of Medicine (US).
- **24** ClinicalTrials.gov, 16 Feb 2016. A Multicenter, Open-label, Randomized Clinical Study to Assess Efficacy and Safety of 3 Doses of Myrcludex B for 24 Weeks in Combination With

246 9 Direct-Acting Antivirals for the Treatment of Chronic Hepatitis B Virus (HBV) Infection

Tenofovir Compared to Tenofovir Alone to Suppress HBV Replication in Patients With Chronic Hepatitis D, NCT03546621. National Library of Medicine (US), Bethesda, MD.

- 25 ClinicalTrials.gov (Apr 2016). A Multicenter, Open-label, Randomised, Comparative, Parallel-Arm, Phase II Study to Assess Efficacy and Safety of Myrcludex B in Combination With Peginterferon Alfa-2a Versus Peginterferon Alfa-2a Alone in Patients With Chronic Viral Hepatitis B With Delta-agent, NCT02888106. Bethesda, MD: National Library of Medicine (US).
- 26 Bogomolov, P., Voronkova, N., Allweiss, L. et al. (2014). A proof-of-concept Phase 2a clinical trial with HBV/HDV entry inhibitor Myrcludex B. *Hepatology* 60 (6): 1279a–1280a.
- 27 Bogomolov, P., Alexandrov, A., Voronkova, N. et al. (2016). Treatment of chronic hepatitis D with the entry inhibitor myrcludex B: first results of a Phase Ib/IIa study. *J. Hepatol.* 65 (3): 490–498.
- **28** Urban, S., Bartenschlager, R., Kubitz, R., and Zoulim, F. (2014). Strategies to inhibit entry of HBV and HDV into hepatocytes. *Gastroenterology* 147 (1): 48–64.
- **29** Glebe, D., Urban, S., Knoop, E.V. et al. (2005). Mapping of the hepatitis B virus attachment site by use of infection-inhibiting preS1 lipopeptides and tupaia hepatocytes. *Gastroenterology* 129 (1): 234–245.
- **30** Gripon, P., Cannie, I., and Urban, S. (2005). Efficient inhibition of hepatitis B virus infection by acylated peptides derived from the large viral surface protein. *J. Virol.* 79 (3): 1613–1622.
- **31** Donkers, J.M., Appelman, M.D., and van de Graaf, S.F.J. (2019). Mechanistic insights into the inhibition of NTCP by myrcludex B. *JHEP Reports*.
- **32** Watashi, K., Sluder, A., Daito, T. et al. (2014). Cyclosporin A and its analogs inhibit hepatitis B virus entry into cultured hepatocytes through targeting a membrane transporter, sodium taurocholate cotransporting polypeptide (NTCP). *Hepatology* 59 (5): 1726–1737.
- **33** Ko, C., Park, W.J., Park, S. et al. (2015). The FDA-approved drug irbesartan inhibits HBV-infection in HepG2 cells stably expressing sodium taurocholate co-transporting polypeptide. *Antivir. Ther.* 20 (8): 835–842.
- **34** Lucifora, J., Esser, K., and Protzer, U. (2013). Ezetimibe blocks hepatitis B virus infection after virus uptake into hepatocytes. *Antivir. Res.* 97 (2): 195–197.
- **35** Blanchet, M., Sureau, C., and Labonte, P. (2014). Use of FDA approved therapeutics with hNTCP metabolic inhibitory properties to impair the HDV lifecycle. *Antivir. Res.* 106: 111–115.
- **36** Kaneko, M., Watashi, K., Kamisuki, S. et al. (2015). A novel tricyclic polyketide, vanitaracin A, specifically inhibits the entry of hepatitis B and D viruses by targeting sodium taurocholate cotransporting polypeptide. *J. Virol.* 89 (23): 11945–11953.
- Saso, W., Tsukuda, S., Ohashi, H. et al. (2018). A new strategy to identify hepatitis B virus entry inhibitors by AlphaScreen technology targeting the envelope-receptor interaction. *Biochem. Biophys. Res. Commun.* 501 (2): 374–379.
- **38** Shimura, S., Watashi, K., Fukano, K. et al. (2017). Cyclosporin derivatives inhibit hepatitis B virus entry without interfering with NTCP transporter activity. *J. Hepatol.* 66 (4): 685–692.

- **39** Miller, R.H., Kaneko, S., Chung, C.T. et al. (1989). Compact organization of the hepatitis B virus genome. *Hepatology* 9 (2): 322–327.
- **40** Walsh, R., Nuttall, S., Revill, P. et al. (2011). Targeting the hepatitis B virus precore antigen with a novel IgNAR single variable domain intrabody. *Virology* 411 (1): 132–141.
- **41** Zlotnick, A., Venkatakrishnan, B., Tan, Z. et al. (2015). Core protein: a pleiotropic keystone in the HBV lifecycle. *Antivir. Res.* 121: 82–93.
- **42** Wang, X.Y., Wei, Z.M., Wu, G.Y. et al. (2012). In vitro inhibition of HBV replication by a novel compound, GLS4, and its efficacy against adefovir-dipivoxil-resistant HBV mutations. *Antivir. Ther.* 17 (5): 793–803.
- **43** Wynne, S.A., Crowther, R.A., and Leslie, A.G. (1999). The crystal structure of the human hepatitis B virus capsid. *Mol. Cell* 3 (6): 771–780.
- **44** Venkatakrishnan, B. and Zlotnick, A. (2016). The structural biology of hepatitis B virus: form and function. *Annu. Rev. Virol.* 3 (1): 429–451.
- **45** Basagoudanavar, S.H., Perlman, D.H., and Hu, J. (2007). Regulation of hepadnavirus reverse transcription by dynamic nucleocapsid phosphorylation. *J. Virol.* 81 (4): 1641–1649.
- Perlman, D.H., Berg, E.A., O'Connor, P.B. et al. (2005). Reverse transcription-associated dephosphorylation of hepadnavirus nucleocapsids. *Proc. Natl. Acad. Sci. U. S. A.* 102 (25): 9020–9025.
- **47** Newman, M., Suk, F.M., Cajimat, M. et al. (2003). Stability and morphology comparisons of self-assembled virus-like particles from wild-type and mutant human hepatitis B virus capsid proteins. *J. Virol.* 77 (24): 12950–12960.
- **48** Zlotnick, A., Cheng, N., Conway, J.F. et al. (1996). Dimorphism of hepatitis B virus capsids is strongly influenced by the C-terminus of the capsid protein. *Biochemistry* 35 (23): 7412–7421.
- **49** Crowther, R.A., Kiselev, N.A., Böttcher, B. et al. (1994). Three-dimensional structure of hepatitis B virus core particles determined by electron cryomicroscopy. *Cell* 77 (6): 943–950.
- **50** Packianathan, C., Katen, S.P., Dann, C.E. 3rd, and Zlotnick, A. (2010). Conformational changes in the hepatitis B virus core protein are consistent with a role for allostery in virus assembly. *J. Virol.* 84 (3): 1607–1615.
- **51** Qazi, S., Schlicksup, C.J., Rittichier, J. et al. (2018). An assembly-activating site in the hepatitis B virus capsid protein can also trigger disassembly. *ACS Chem. Biol.* 13 (8): 2114–2120.
- 52 Bartenschlager, R. and Schaller, H. (1992). Hepadnaviral assembly is initiated by polymerase binding to the encapsidation signal in the viral RNA genome. *EMBO J.* 11 (9): 3413–3420.
- 53 Zlotnick, A., Johnson, J.M., Wingfield, P.W. et al. (1999). A theoretical model successfully identifies features of hepatitis B virus capsid assembly. *Biochemistry* 38 (44): 14644–14652.
- **54** Ceres, P. and Zlotnick, A. (2002). Weak protein-protein interactions are sufficient to drive assembly of hepatitis B virus capsids. *Biochemistry* 41 (39): 11525–11531.
- 55 Stray, S.J., Johnson, J.M., Kopek, B.G., and Zlotnick, A. (2006). An in vitro fluorescence screen to identify antivirals that disrupt hepatitis B virus capsid assembly. *Nat. Biotechnol.* 24 (3): 358–362.

- 248 9 Direct-Acting Antivirals for the Treatment of Chronic Hepatitis B Virus (HBV) Infection
 - **56** Deres, K., Schroder, C.H., Paessens, A. et al. (2003). Inhibition of hepatitis B virus replication by drug-induced depletion of nucleocapsids. *Science* 299 (5608): 893–896.
 - Qiu, Z., Lin, X., Zhang, W. et al. (2017). Discovery and pre-clinical characterization of third-generation 4-H heteroaryldihydropyrimidine (HAP) analogues as hepatitis B virus (HBV) capsid inhibitors. *J. Med. Chem.* 60 (8): 3352–3371.
 - **58** Perni, R.B., Conway, S.C., Ladner, S.K. et al. (2000). Phenylpropenamide derivatives as inhibitors of hepatitis B virus replication. *Bioorg. Med. Chem. Lett.* 10 (23): 2687–2690.
 - **59** Lam, A.M., Espiritu, C., Vogel, R. et al. (2019). Preclinical characterization of NVR 3-778, a first-in-class capsid assembly modulator against hepatitis B virus. *Antimicrob. Agents Chemother.* 63 (1): e01734-18.
 - **60** Yang, L., Shi, L.P., Chen, H.J. et al. (2014). Isothiafludine, a novel non-nucleoside compound, inhibits hepatitis B virus replication through blocking pregenomic RNA encapsidation. *Acta Pharmacol. Sin.* 35 (3): 410–418.
 - **61** Corcuera, A., Stolle, K., Hillmer, S. et al. (2018). Novel non-heteroarylpyrimidine (HAP) capsid assembly modifiers have a different mode of action from HAPs in vitro. *Antivir. Res.* 158: 135–142.
 - 62 Pei, Y., Wang, C., Ben, H. et al. (2019). Discovery of new hepatitis B virus capsid assembly modulators by an optimal high-throughput cell-based assay. ACS Infect. Dis. 5 (5): 778–787.
 - **63** Huber, A.D., Pineda, D.L., Liu, D. et al. (2019). Novel hepatitis B virus capsid-targeting antiviral that aggregates core particles and inhibits nuclear entry of viral cores. *ACS Infect. Dis.* 5 (5): 750–758.
 - **64** Weber, O., Schlemmer, K.H., Hartmann, E. et al. (2002). Inhibition of human hepatitis B virus (HBV) by a novel non-nucleosidic compound in a transgenic mouse model. *Antivir. Res.* 54 (2): 69–78.
 - **65** Stray, S.J., Bourne, C.R., Punna, S. et al. (2005). A heteroaryldihydropyrimidine activates and can misdirect hepatitis B virus capsid assembly. *Proc. Natl. Acad. Sci. U. S. A.* 102 (23): 8138–8143.
 - 66 Ren, Q., Liu, X., Luo, Z. et al. (2017). Discovery of hepatitis B virus capsid assembly inhibitors leading to a heteroaryldihydropyrimidine based clinical candidate (GLS4). *Bioorg. Med. Chem.* 25 (3): 1042–1056.
 - **67** Wu, G., Liu, B., Zhang, Y. et al. (2013). Preclinical characterization of GLS4, an inhibitor of hepatitis B virus core particle assembly. *Antimicrob. Agents Chemother.* 57 (11): 5344–5354.
 - 68 Jiang, Y., Wan, Z.-K., Cheung, W.S. et al. (2019). Dihydropyrimidine derivatives and uses thereof in the treatment of HBV infection or of HBV-induced diseases. WO 2019/214610 A1, filed 7 May 2019 and issued 14 November 2019.
 - **69** Hayer, J., Jadeau, F., Deleage, G. et al. (2013). HBVdb: a knowledge database for Hepatitis B Virus. *Nucleic Acids Res.* 41 (Database issue): D566–D570.
 - 70 Wu, S., Zhao, Q., Zhang, P. et al. (2017). Discovery and mechanistic study of benzamide derivatives that modulate hepatitis B virus capsid assembly. *J. Virol.* 91 (16): e00519–e00517.
 - **71** Zhou, Z., Hu, T., Zhou, X. et al. (2017). Heteroaryldihydropyrimidine (HAP) and sulfamoylbenzamide (SBA) inhibit hepatitis B virus replication by different molecular mechanisms. *Sci. Rep.* 7: 42374.

- 72 Berke, J.M., Tan, Y., Verbinnen, T. et al. (2017). Antiviral profiling of the capsid assembly modulator BAY41-4109 on full-length HBV genotype A-H clinical isolates and core site-directed mutants in vitro. *Antivir. Res.* 144: 205–215.
- 73 Klumpp, K., Lam, A.M., Lukacs, C. et al. (2015). High-resolution crystal structure of a hepatitis B virus replication inhibitor bound to the viral core protein. *Proc. Natl. Acad. Sci.* U. S. A. 112 (49): 15196–15201.
- 74 Verbinnen, T., Berke, J.M., Tan, Y. et al. (2018). Anti-HBV activity of capsid assembly modulators JNJ-56136379 (JNJ-6379) and BAY 41-4109 on an extensive panel of clinical isolates and HBV core mutants. Poster presented at the International HBV Meeting 2018, Taormina, Italy (3–6 Oct 2018).
- **75** Wang, J., Zhang, H., Zhang, Y. et al. (2017). Influences on viral replication and sensitivity to GLS4, a HAP compound, of naturally occurring T109/V124 mutations in hepatitis B virus core protein. *J. Med. Virol.* 89 (10): 1804–1810.
- **76** King, R.W., Ladner, S.K., Miller, T.J. et al. (1998). Inhibition of human hepatitis B virus replication by AT-61, a phenylpropenamide derivative, alone and in combination with (-) beta-L-2',3'-dideoxy-3'-thiacytidine. *Antimicrob. Agents Chemother.* 42 (12): 3179–3186.
- Feld, J.J., Colledge, D., Sozzi, V. et al. (2007). The phenylpropenamide derivative AT-130 blocks HBV replication at the level of viral RNA packaging. *Antivir. Res.* 76 (2): 168–177.
- **78** Delaney, W.E., Edwards, R., Colledge, D. et al. (2002). Phenylpropenamide derivatives AT-61 and AT-130 inhibit replication of wild-type and lamivudine-resistant strains of hepatitis B virus in vitro. *Antimicrob. Agents Chemother.* 46 (9): 3057–3060.
- **79** Billioud, G., Pichoud, C., Puerstinger, G. et al. (2011). The main hepatitis B virus (HBV) mutants resistant to nucleoside analogs are susceptible in vitro to non-nucleoside inhibitors of HBV replication. *Antivir. Res.* 92 (2): 271–276.
- **80** Pei, Y., Wang, C., Yan, S.F., and Liu, G. (2017). Past, current, and future developments of therapeutic agents for treatment of chronic hepatitis B virus infection. *J. Med. Chem.* 60 (15): 6461–6479.
- **81** Fanning, G.C., Zoulim, F., Hou, J., and Bertoletti, A. (2019). Therapeutic strategies for hepatitis B virus infection: towards a cure. *Nat. Rev. Drug Discov.* 18 (11): 827–844.
- 82 Lenz, O., Balmain, C.E., Snoeys, J. et al. (2019). Capsid assembly modulator dosing regimen. WO 2019/175657 A1, filed 13 March 2019 and issued 19 September 2019.
- **83** Vandyck, K., Haché, G.Y.P., Last, S.J. et al. (2014). Sulphamoylpyrrolamide derivatives and the use thereof as medicaments for the treatment of hepatitis B. WO 2014/184350 A1, filed 16 May 2014 and issued 20 November 2014.
- **84** Du, J., Kaplan, J.A., Kirschberg, T.A. et al. (2017). Substituted pyrrolizine compounds and uses thereof. WO 2018/039531 A1, filed 25 August 2017 and issued 1 March 2018.
- **85** Schinazi, R.F., Boucle, S., Amblard, F. et al. (2017). Elimination of hepatitis B virus with antiviral agents. WO 2017/156255 A1, filed 9 March 2017 and issued 14 September 2017.
- **86** Amblard, F., Boucle, S., Bassit, L. et al. (2020). Novel hepatitis B virus capsid assembly modulator induces potent antiviral responses in vitro and in humanized mice. *Antimicrob. Agents Chemother.* 64 (2): e01701–e01719.
- 87 Donald, A., Urban, A., Bonsmann, S. et al. (2018). Novel, highly active amino-thiazole substituted indole-2-carboxamides active against the hepatitis B virus (HBV). WO 2019/086141 A8, filed 2 November 2018 and issued 9 May 2019.

- 250 9 Direct-Acting Antivirals for the Treatment of Chronic Hepatitis B Virus (HBV) Infection
 - 88 Donald, A., Urban, A., Bonsmann, S. et al. (2018). Novel, highly active pyrazole-piperidine substituted indole-2-carboxamides active against the hepatitis B virus (HBV). WO 2019/086142 A1, filed 2 November 2018 and issued 9 May 2019.
 - Fu, J., Jain, R., Jin, X. et al. (2019). Indole-2-carbonyl compounds and their use for the treatment of hepatitis B. WO 2019/166951 A1, filed 26 February 2019 and issued 6 September 2019.
 - Li, X., He, W., Chen, Y. et al. (2018). Piperazine heteroaryl derivative, preparation method therefor and use of same in medicine. WO 2019/020070 A1, filed 26 July 2018 and issued 31 January 2019.
 - **91** Fu, J., Lindvall, M., Manning, J.R., and McEnroe, G. (2018). Novel dihydroisoxazole compounds and their use for the treatment of hepatitis B. WO 2019/097479 A1, filed 16 November 2018 and issued 23 May 2019.
 - Qiu, Z., Lin, X., Zhou, M. et al. (2016). Design and synthesis of orally bioavailable 4-methyl heteroaryldihydropyrimidine based hepatitis B virus (HBV) capsid inhibitors. *J. Med. Chem.* 59 (16): 7651–7666.
 - Katen, S.P., Chirapu, S.R., Finn, M.G., and Zlotnick, A. (2010). Trapping of hepatitis B virus capsid assembly intermediates by phenylpropenamide assembly accelerators. *ACS Chem. Biol.* 5 (12): 1125–1136.
 - Campagna, M.R., Liu, F., Mao, R. et al. (2013). Sulfamoylbenzamide derivatives inhibit the assembly of hepatitis B virus nucleocapsids. *J. Virol.* 87 (12): 6931–6942.
 - Lahlali, T., Berke, J.M., Vergauwen, K. et al. (2018). Novel potent capsid assembly modulators regulate multiple steps of the hepatitis B virus life cycle. *Antimicrob. Agents Chemother.* 62 (10): e00835-18.
 - **96** Berke, J.M., Dehertogh, P., Vergauwen, K. et al. (2017). Capsid assembly modulators have a dual mechanism of action in primary human hepatocytes infected with hepatitis B virus. *Antimicrob. Agents Chemother.* 61 (8): e00560–e00517.
 - Liu, C., Fan, G., Wang, Z. et al. (2017). Allosteric conformational changes of human HBV core protein transform its assembly. *Sci. Rep.* 7 (1): 1404.
 - Bourne, C.R., Finn, M.G., and Zlotnick, A. (2006). Global structural changes in hepatitis B virus capsids induced by the assembly effector HAP1. *J. Virol.* 80 (22): 11055–11061.
 - Tan, Z., Maguire, M.L., Loeb, D.D., and Zlotnick, A. (2013). Genetically altering the thermodynamics and kinetics of hepatitis B virus capsid assembly has profound effects on virus replication in cell culture. *J. Virol.* 87 (6): 3208–3216.
 - Katen, S.P., Tan, Z., Chirapu, S.R. et al. (2013). Assembly-directed antivirals differentially bind quasiequivalent pockets to modify hepatitis B virus capsid tertiary and quaternary structure. *Structure* 21 (8): 1406–1416.
 - Kang, J.A., Kim, S., Park, M. et al. (2019). Ciclopirox inhibits Hepatitis B Virus secretion by blocking capsid assembly. *Nat. Commun.* 10 (1): 2184.
 - Yang, L., Wang, Y.J., Chen, H.J. et al. (2016). Effect of a hepatitis B virus inhibitor, NZ-4, on capsid formation. *Antivir. Res.* 125: 25–33.
 - 103 Mani, N., Cole, A.G., Phelps, J.R. et al. (2018). Preclinical profile of AB-423, an inhibitor of hepatitis B virus pregenomic RNA encapsidation. *Antimicrob. Agents Chemother*. 62 (6): e00082-18.
 - Nair, S., Li, L., Francis, S. et al. (2018). Use of a fluorescent analogue of a HBV core protein-directed drug to interrogate an antiviral mechanism. *J. Am. Chem. Soc.* 140 (45): 15261–15269.

- Rat, V., Seigneuret, F., Burlaud-Gaillard, J. et al. (2019). BAY 41-4109-mediated aggregation of assembled and misassembled HBV capsids in cells revealed by electron microscopy. *Antivir. Res.* 169: 104557.
- Huber, A.D., Wolf, J.J., Liu, D. et al. (2018). The heteroaryldihydropyrimidine bay 38-7690 induces hepatitis B virus core protein aggregates associated with promyelocytic leukemia nuclear bodies in infected cells. *mSphere* 3 (2): e00131–e00118.
- Bock, C.T., Schwinn, S., Locarnini, S. et al. (2001). Structural organization of the hepatitis B virus minichromosome. *J. Mol. Biol.* 307 (1): 183–196.
- Guo, Y.H., Li, Y.N., Zhao, J.R. et al. (2011). HBc binds to the CpG islands of HBV cccDNA and promotes an epigenetic permissive state. *Epigenetics* 6 (6): 720–726.
- Gruffaz, M., Testoni, B., Luangsay, S. et al. (2013). 378 hepatitis B core (Hbc) protein is a key and very early negative regulator of the interferon response. *J. Hepatol.* 58: S155–S156.
- Guo, F., Zhao, Q., Sheraz, M. et al. (2017). HBV core protein allosteric modulators differentially alter cccDNA biosynthesis from de novo infection and intracellular amplification pathways. *PLoS Pathog.* 13 (9): e1006658.
- Cui, X., Ludgate, L., Ning, X., and Hu, J. (2013). Maturation-associated destabilization of hepatitis B virus nucleocapsid. *J. Virol.* 87 (21): 11494–11503.
- Dhason, M.S., Wang, J.C., Hagan, M.F., and Zlotnick, A. (2012). Differential assembly of Hepatitis B Virus core protein on single- and double-stranded nucleic acid suggest the dsDNA-filled core is spring-loaded. *Virology* 430 (1): 20–29.
- Liang, T.J., Block, T.M., McMahon, B.J. et al. (2015). Present and future therapies of hepatitis B: from discovery to cure. *Hepatology* 62 (6): 1893–1908.
- Nijampatnam, B. and Liotta, D.C. (2019). Recent advances in the development of HBV capsid assembly modulators. *Curr. Opin. Chem. Biol.* 50: 73–79.
- 115 Gao, Q., Gu, Z., Salomon, H. et al. (1994). Generation of multiple drug resistance by sequential in vitro passage of the human immunodeficiency virus type 1. Arch. Virol. 136 (1–2): 111–122.
- **116** Durantel, D. (2010). Fitness and infectivity of drug-resistant and cross-resistant hepatitis B virus mutants: why and how is it studied? *Antivir. Ther.* 15 (3 Pt B): 521–527.
- Tan, Z., Pionek, K., Unchwaniwala, N. et al. (2015). The interface between hepatitis B virus capsid proteins affects self-assembly, pregenomic RNA packaging, and reverse transcription. *J. Virol.* 89 (6): 3275–3284.
- Lomonosova, E., Zlotnick, A., and Tavis, J.E. (2017). Synergistic interactions between hepatitis B virus RNase H antagonists and other inhibitors. *Antimicrob. Agents Chemother.* 61 (3): e02441–e02416.
- Dai, L., Yu, Y., Gu, L. et al. (2018). Combination treatment of a TLR7 agonist RO7020531 and a capsid assembly modulator RO7049389 achieved sustainable viral load suppression and HBsAg loss in an AAV-HBV mouse model. *J. Hepatol.* 68: S17–S18.
- Lopatin, U. (2019). Drugs in the pipeline for HBV. *Clin. Liver Dis.* 23 (3): 535–555.
- 121 Yuen, M.F., Gane, E.J., Kim, D.J. et al. (2019). Antiviral activity, safety, and pharmacokinetics of capsid assembly modulator NVR 3-778 in patients with chronic HBV infection. *Gastroenterology* 156 (5): 1392–1403 e1397.
- 122 Zhao, N., Jia, B., Zhao, H. et al. (2019). A first-in-human trial of GLS4, a novel inhibitor of hepatitis B virus capsid assembly, following single- and multiple-ascending-oral-dose studies with or without ritonavir in healthy adult volunteers. *Antimicrob. Agents Chemother.* 64 (1): e01686–e01619.

- 252 9 Direct-Acting Antivirals for the Treatment of Chronic Hepatitis B Virus (HBV) Infection
 - **123** Cornberg, M., Lok, A.S., Terrault, N.A. et al. (2020). Guidance for design and endpoints of clinical trials in chronic hepatitis B report from the 2019 EASL-AASLD HBV treatment endpoints conference (double dagger). *J. Hepatol.* 72 (3): 539–557.
 - 124 Sulkowski, M.S., Agarwal, K., Fung, S. et al. (2019). Continued therapy with ABI-H0731
 + NrtI results in sequential reduction/loss of HBV DNA, HBV RNA, HBeAg, HBcrAg and HBsAg in HBeAg-positive patients. Poster presented at the American Association for the Study of Liver Diseases (AASLD) Annual Meeting, Boston, MA, USA (8–12 Nov 2019).
 - 125 Yuen, M.-F., Locarnini, S., Given, B. et al. (2019). First clinical experience with RNA interference-based triple combination therapy in chronic hepatitis B: JNJ-3989, JNJ-6379 and a nucleos(t)ide analogue. Poster presented at the American Association for the Study of Liver Diseases (AASLD) Annual Meeting, Boston, MA, USA (8–12 Nov 2019).
 - **126** Lee, H.W., Lee, J.S., and Ahn, S.H. (2021). Hepatitis B Virus Cure: Targets and Future Therapies. *Int. J. Mol. Sci.* 22(1): 213.

10

Hepatitis E Virus—Current Developments in Antiviral Strategies

Daniel Todt^{1,2}, Patrick Behrendt^{3,4,5}, and Eike Steinmann¹

¹ Department of Molecular and Medical Virology, Ruhr University Bochum, Bochum, Germany

³ Department for Gastroenterology, Hepatology and Endocrinology, Hannover Medical School, Hannover, Germany

⁴ Twincore-Center for Experimental and Clinical Infection Research, Hannover Medical School and Helmholtz Centre for

Infection Research, Hannover, Germany

⁵ German Centre for Infection Research, Hannover-, Braunschweig, Germany

10.1 Introduction

The first description of Hepatitis E Virus (HEV) as a novel agent responsible for enterically transmitted non-A, non-B hepatitis by Reyes and colleagues dates back to 1991 [1]. After its first recognition during an epidemic of hepatitis in the Kashmir valley, India, in 1978 [2], similar outbreaks in the 1980s were reported in Nepal [3], Burma [4], Pakistan [5], Mexico [6], and China [7]. Nevertheless, the earliest documented epidemic outbreak retrospectively identified to have been caused by fecal-orally transmitted HEV occurred in New Delhi, India, in 1955–1956 [8, 9]. Following serological and molecular studies, HEV is today considered to be globally distributed and to be the leading cause of enterically transmitted viral hepatitis worldwide. An estimated 20 million new infections occur each year resulting in about 3.4 million symptomatic cases of Hepatitis E and 70000 Hepatitis Erelated [10]. Initially, HEV was only considered to trigger partly large outbreaks in nondeveloped areas. However, improved detection methods and higher awareness have also revealed substantial infection events in developed areas [11]. In parts of Europe, a prevalence of anti-HEV-IgG up to 50% has been observed [12]. Being usually an uncomplicated and self-limiting infection, in individuals with underlying liver diseases, it induces progression to rapid liver failure. Most strikingly, and still not fully understood, HEV causes an acute symptomatology with death rates in up to 25% in pregnant women [13]. Furthermore, patients with hereditary or acquired immunodeficiency can become chronically infected [11]. Here, HEV shows similar escape mechanism as the Hepatitis C Virus (HCV). However, in contrast to HCV, treatments have not been well established yet. Standard of care for chronic infections involves either treatment with ribavirin or

² European Virus Bioinformatics Center (EVBC), 07743 Jena, Germany

254 10 Hepatitis E Virus-Current Developments in Antiviral Strategies

interferon, both having considerable side effects. Overall, despite the widespread distribution of this virus with approximately 3 million people affected worldwide HEV has been greatly neglected in the past. Direct-acting antivirals against HEV have not been described so far. However, for patients at risk to develop acute liver failure as well as chronically infected individuals, new antivirals need to be developed in the near future. In this chapter, we summarize the current knowledge about HEV and its therapy and provide an overview of the current developments in antiviral strategies.

10.2 Genetic Diversity and Molecular Virology of HEV

The human pathogenic HEV has recently been newly taxonomically assigned to the genus Orthohepevirus A in the family of Hepeviridae. The other members of this family being Orthohepevirus B (avian hepatitis E virus), Orthohepevirus C (rat hepatitis E virus), and Orthohepevirus D (bat hepatitis E virus) [14]. HEV infections in humans can be caused by five different genotypes (gt) which differ in their worldwide distribution, hosts, and route of transmission. Genotype 1 and 2 viruses are described mainly in developing countries and are transmitted via the fecal-oral route by waterborne outbreaks. They solely infect humans and show high epidemic potential in India, North Africa, and Asia, causing up to 120000 symptomatic cases in a single outbreak [15]. In contrast, HEV genotype 3 and genotype 4 infections are primarily diagnosed in industrialized nations. There have been many animals described, that serve as reservoir, including pigs [16, 17], deer and mongoose [18], wild boars [19], shellfish [20], rodents, bison, cattle, and dogs [16]. In 2014, a new genotype 7 was identified in dromedaries in the Middle East. HEV genotypes 3, 4, and 7 can cross the species barrier, as several reports indicate, where individuals became infected with HEV after eating infected raw meat from deer [21, 22], sausages containing pig liver [23, 24], or drinking camel milk [25]. Of special note, there have also been reports from Canada and Hong Kong on rat HEV, a member of the Orthohepevirus C genus that shares 50-60% nucleotide identity with the HEV A, to cause chronic infections in immunocompromised humans [26, 27].

HEV is an icosahedral-shaped virus with about 27–34 nm in diameter [28, 29]. The virion contains a ~7.2 kb long single-stranded RNA genome in positive orientation, which encodes for three open reading frames (ORF1-3) and is shaped like eukaryotic messenger RNA (mRNA) with a 5' 7-methylguanylate (7^mG) cap and a 3' poly(A) tail [29]. During the viral replication in the host cell, at least two RNA species are generated. In addition to the full-length RNA, also a subgenomic RNA of ~2.2 kb is transcribed, allowing expression of ORF2 and ORF3 [30, 31]. HEV is a quasi-enveloped virus. It can exist in a non-enveloped state, which is shed in the feces or it can be coated with a lipid-derived membrane, which circulates in the blood and might protect the virus from neutralizing antibodies [32]. ORF1 is the largest viral gene product and compromises a methyltransferase (MeT), a macro domain, a papain-like cysteine protease (PCP), a helicase (Hel), and an RNA-dependent RNA polymerase (RdRp), connected by a Y-domain and a hypervariable region (HVR) (Figure 10.1). The maturation of the ORF1 polyprotein and potential processing by the encoded protease are not fully understood and have been under discussion [33–38]. It has recently been suggested that the PCP domain, instead of having protease activity, rather exerts broad

de-ubiquitination activity preventing the proteosomal degradation of viral or selected host proteins [39]. The MeT, Hel, and RdRp have also been functionally analyzed, while the functions of the Y, the variable (V), and the macro domain remain uncertain [40]. An additional ORF4 has been identified in the coding sequence of ORF1 exclusively in HEV genotype 1 and is reported to increase the RdRp activity [41].

ORF2 encodes for the 660 amino acid (aa) virus capsid protein. Its transcription is driven by an intragenomic promoter in the junction region at the ORF1 3' end (Figure 10.1) [42]. It is relatively conserved among the viral genotypes and harbors a typical signal peptide sequence and three potential glycosylation sites [43]. This capsid protein plays a crucial role during the viral attachment process to the host cell and is the major target for neutralizing antibodies. The ORF3 protein is only 360 bp in length and is a multifunctional 13-kDa protein of 113 (genotype 3) or 114 aa (genotypes 1, 2, and 4), respectively (Figure 10.1). This small protein is implicated as part of quasi-enveloped virions and plays an important role during viral egress [44]. Palmitoylation of the ORF3 protein drives its membrane association [45]. Recently, for ORF3 an ion channel activity has been reported, which is critical for release of infectious particles [46].

With respect to the viral life cycle, many steps are only partially understood [47]. Similar to other viruses, heparan sulfate proteoglycans on the host cell surface are required for attachment [48]. Cellular surface receptors that trigger virus entry are still



Figure 10.1 Hepatitis E virus genome organization. Upon entering host cells, the viral ss(+)-RNA (blue) ORF1 is initially translated by the host's ribosomes to produce the nonstructural viral proteins methyltransferase (MeT), papain-like cysteine protease (PCP), helicase (Hel), and RNA-dependent RNA polymerase (RdRp). The functions of a Y-domain, X-domain, and hypervariable region (HVR) are not fully understood yet. The newly produced RdRp subsequently transcribes the +mRNA into a -RNA (pink). This -RNA is then used as a template to amplify new +RNA, full length (7.2 kb) as well as a 2.2 kb long subgenomic RNA which encodes for the ORF2 capsid proteins S (small), M (middle), and P (protruding) and for the ORF3. An intragenomic promoter (JR) controls subgenomic RNA transcription. *Source*: Based on [42].

256 10 Hepatitis E Virus-Current Developments in Antiviral Strategies

to be identified. Very recently, integrin α 3 was suggested as key molecule for cellular attachment and entry of naked HEV virions [49]. Following clathrin-dependent endocytosis [50] and cytoskeleton remodeling, the viral genome is uncoated and released, probably in a pH-independent mechanism [51]. Similar to all positively orientated single-stranded RNA viruses that do not bring any accessory proteins, it requires translation of the viral replicase by the host ribosome machinery. The HEV RdRp then produces complementary negative-strand RNA and subsequently synthesizes genomic positivestrand RNA, full length as well as subgenomic RNA, from this negative-strand RNA template in an alternating manner [52]. The subcellular site of RNA replication has not been identified yet. The ORF1 protein has been shown to be membrane-associated and to localize to an intermediate compartment between the endoplasmic reticulum and the Golgi, suggesting localization within the early secretory pathway [35]. For virus particle assembly and release, HEV hijacks the host cell's endosomal sorting complexes required for transport (ESCRT) machinery by the interaction of ORF3 with the tumor susceptibility gene 101 (TSG101) [53]. ORF2 is reported to bind to the 5' end of the HEV genome, implicating an encapsidation mechanism. It exists in several capsid forms, from which only a minority assembles into infectious viral particles [54]. After the release from hepatocytes, ORF3 plays an essential role in envelope formation. This envelope is stripped in the biliary tract, so that fecal HEV is found in high-density, non-enveloped structures [55].

10.3 Clinical Course of HEV Infections

The clinical course of HEV and Hepatitis A Virus (HAV) are in major parts indistinguishable. Many infected individuals may have subclinical symptoms and clinicians will only observe a seroconversion over time. In a large double-blinded vaccine trial from China, 56000 individuals were followed. Only 5% of the individuals of the control arm, who seroconverted developed symptoms of an acute HEV infection [56]. In these clinically apparent cases, overall mortality rates of 0.2-4% were observed [57]. It is estimated that HEV causes 70000 deaths per year [10], but this number could be largely underestimated due to assay variability. Of note, outcome may vary among infected individuals depending on patientspecific characteristics. Pregnant women are more likely to develop acute hepatitis (60 vs. 30% control group) [58] with higher mortality rates (up to 25%) [59] and increased numbers of stillbirths [10]. This might be explained by the different hormonal and immunological features during pregnancy. In particular, reduced expression of progesterone-receptor expression is associated with severe HEV-infection outcome. Of note, until now this has been only shown for genotype 1-infected pregnant women. Furthermore, higher viral load serves as an indicator for a severe outcome of the infection as well as increased maternal and fetal mortality [13].

Additionally, in individuals with pre-existing liver disease, HEV infection is associated with increased morbidity and mortality due to liver failure [60–63]. Also, prolonged alcohol

consumption serves as a risk factor for fulminant liver disease in the course of acute HEV infection [64] and intergenotypic differences have been described, causing more severe courses in genotype 1 compared to genotype 3-infected individuals [65]. In clinically apparent cases, the incubation period ranges from two to eight weeks, with a mean of 40 days [66]. The initial symptoms are typically unspecific and include flu-like symptoms like myalgia, arthralgia, weakness, loss of appetite, abdominal pain, and vomiting. This is usually accompanied by highly elevated liver enzymes (aminotransferases, alkaline phosphatase, γ -glutamyltransferase, and bilirubin) and the development of liver signs like jaundice, pruritus, uncolored stool, and darkened urine. Usually these symptoms are self-limiting and resolve within four to six weeks [67–70].

Of special note, despite HEV being a primary hepatotropic pathogen, there have been numerous extrahepatic manifestations described [71]. Besides renal, hematologic, and pancreatic manifestations, neurological symptoms are the most prominent, including neuralgic amyotrophy (brachial neuritis, Parsonage-Turner syndrome), and Guillain–Barré syndrome [72–79]. In addition to this, HEV RNA has been identified in the cerebrospinal fluid of infected patients [75, 76, 78, 80]. Whether these symptoms are caused by active replication of the virus in the respective tissues is still unknown, but *in vitro* data prove HEV to be capable of replicating in neuron-derived [81], as well as in placenta-derived cell lines [82].

Despite the different genotypes of HEV, only one serotype is described which has major implications for vaccine approaches as well as clearance of HEV-infection. With the onset of elevated transaminases, IgM against HEV can be detected [83]. Shortly thereafter, highly specific IgG is observed, which indicates control of infection and confers resistance to reinfection in immune-competent individuals for at least a certain time.

Besides these humoral responses, cellular immunity plays a critical role in virus control. Potent HEV-specific T-cell responses can be observed in individuals, who spontaneously or treatment-induced cleared HEV infection [84]. Additionally, these strong T-cell responses can be detected in patients, who previously cleared a subclinical HEV infection. In general, every circumstance that negatively influences important mechanisms of viral clearance can cause chronic infection, which is defined as persistent infection over a time-period of six months. Individuals, who are not able to clear the virus, display impaired T-cell responses [84]. Furthermore, at least in case of transplanted patients that were investigated, IgG could be detected although persistent infection took place. This might be due to low titers or reduced neutralizing potency of these antibodies. Theoretically, similar to what is known for HCV, the virus could potentially escape humoral responses due to its high genetic variability [85, 86]. Controversially, in chronically infected HEV patients, wellknown interferon-stimulated genes like IFIT1, IFI44L, RSAD2, EPSTI1, and ISG15 are over-expressed compared to healthy individuals [87]. Interestingly, this is again in line with findings in chronically HCV-infected patients, where it causes nonresponse to interferonbased treatment [88].

Chronic infection with HEV is solely described in the context of genotype 3. However, this observation could be biased as chronic infection is almost exclusively observed in areas, where occurrence of other genotypes is unlikely.

10.4 HEV Therapy

HEV-specific antiviral drugs are still lacking and treatment for patients with hepatitis E currently is only supportive. The current treatment algorithm for chronic infections by the European Association for the Study of the Liver (EASL) was published in 2018 [89]. First of all, the possibility of reduction of immunosuppressive medication needs to be evaluated [11]. Previous studies reported clearance rates of up to 25% employing this strategy [90]. However, the risk of rejection of the allograft is constituted by reduction of immunosuppression beyond a certain level [67, 91].

The treatment of choice for HEV infections is the off-label use of RBV. Its efficacy in reducing viral loads in patients has been evaluated in large studies for acute as well as chronic infections [65, 92, 93]. Also, in immunocompromised, SOT recipients or HIVinfected individuals, it can be safely applied and results in clearance of the viral RNA [94–96]. Furthermore, *in vitro* data suggest a moderate synergistic effect when combined with pegIFN- α [94]. However, because of its teratogenic characteristics, RBV cannot be administered to pregnant women and there are reports on RBV treatment failure in risk group patients. In the latter cases, RBV doses had to be reduced because of severe anemia and one patient died after experiencing a virological breakthrough associated with RBV dose reduction [65, 97]. Recently, investigations suggest that these treatment failures were possibly linked to the selection of a distinct HEV polymerase variant (G1634R) resulting in increased replication fitness [98, 99]. In a subsequent study, 63 SOT patients with chronic hepatitis E were screened for the existence of this variant. The authors found that the presence of the G1634R variant at RBV initiation does not lead to absolute RBV resistance and although its proportion was increased in patients whose treatment failed, the presence of the 1634R variant did not compromise the response to a second RBV treatment [100].

There are several modes of action proposed for RBV and its antiviral effect against RNA viruses [101, 102] and reviewed in the context of HEV [103, 104]. Among the indirect mechanisms, a T-cell-mediated effect is described, switching the balance of T helper cells from a T helper type 2 phenotype to a T helper type 1. RBV restores the T helper 1 phenotype needed for balanced expression and secretion of cytokines produced from type 1 and 2 T helper cells. A second indirect mode of action is the inhibition of the cellular inosine-5'monophosphate dehydrogenase (IMPDH). In mammalian cells, RBV is phosphorylated and RBV monophosphate (RMP) perfectly mimics inosine monophosphate (IMP) and thereby inhibits the synthesis of IMP to xanthosine monophosphate (XMP) by IMPDH. Consequently, no guanosine triphosphate (GTP), necessary for production of nascent (viral) RNA, is synthesized. This depletion of GTP pools has been proven to play a role in the anti-HEV effect of RBV in vitro [94]. Further mechanisms described for RBV are a direct influence on the expression if IFN stimulated genes (ISGs), the inhibition of (viral) MeTs, thereby interfering with the catalysis of the formation of 5' 7-methylguanosine cap structure, and the direct inhibition of viral polymerases [104]. Here, RBV triphosphate (RTP) is recognized by the RdRp leading to chain termination or preventing the binding of other nucleotides important for elongation [105]. In recent years, an additional mode of action for RBV was described, i.e. a mutagenic effect via incorporation into newly synthesized RNA genomes, leading to viral extinction. This effect has been observed in several RNA viral populations [106–110], as well as for HEV [99, 103, 104, 111].

10.5 Development of Novel Antivirals Against HEV

Both antiviral drugs—interferon as well as ribavirin—are not highly selectively targeting the viral life cycle. Therefore, both display considerable side effects, in particular flu-like symptoms, depressions, (interferon) or severe anemia (ribavirin). Also, treatment of transplanted patients with interferon might induce graft rejection and therefore should be used with caution. In particular, treatment of pregnant women is difficult, as both drugs are contraindicated under these circumstances. Therefore, novel antiviral against HEV are urgently needed [112].

The most advanced drug candidate in development as novel antiviral molecule against HEV is the highly potent HCV RdRp inhibitor sofosbuvir (Figure 10.2). Sofosbuvir is a nucleotide prodrug (Table 10.1) that acts as a chain terminator of the replicase after activation and incorporation into the viral genome [127]. Different studies reported an antiviral activity of sofosbuvir against HEV using different HEV replicons and human liver cell lines [113, 117, 128, 129]. Dao Thi and colleagues were the first to show that the drug-inhibited HEV replication in a low micmolar range and was additive with RBV [117]. However, the antiviral activity was 10-100-fold lower compared to HCV replicons and could not be confirmed by Wang et al. with GT1 and GT3 replicons in different cell lines [130]. The drug with or without RBV has been used to treat patients, who failed previous RBV or were ineligible for RBV therapy. However, the results of the published case reports are inconclusive and provide no clear evidence of efficiency [131]. While in one case successful treatment with sofosbuvir/RBV was reported [132], other studies showed only weak or moderate effects of sofosbuvir [133–137]. For these reasons, a multicenter study (SofE), an investigator-initiated, phase 2 pilot trial (NCT03282474), recently investigated the efficacy and safety of 400 mg sofosbuvir daily for 24 weeks in nine immunocompromised patients with chronic hepatitis E, who were ineligible or failed prior ribavirin therapy [138]. Sofosbuvir treatment showed only weak antiviral activities without curing the infection and HEV RNA decline was associated with ALT improvements [138]. These results suggest that sofosbuvir is not suited as monotherapy treatment and requires further investigations for possible combination treatments.

The reuse of an already existing compound is referred as drug repurposing [139]. The advantage of drug repurposing is that pharmacological information is already available and that *in vivo* testing has already been performed, although for another primary indication. Repurposing approaches in the field of HEV were reported for the compounds 2'-C-methylcytidine (2-CMC), NITD008, and GPCN114 (Figure 10.2; Table 10.1). The nucleoside analogue 2-CMC, which is also known as NM107, was originally developed against HCV but has a low oral bioavailability. However, the development of a prodrug, NM283, resulted in adverse toxic effect as the drug can serve as a substrate for the mitochondrial DNA polymerase [119]. An inhibitory effect of 2-CMC was demonstrated against HEV GT3 with a half maximum inhibitory concentration (IC₅₀) of $22 \,\mu$ M [117]. Qu et al. confirmed these anti-HEV effects and showed antagonistic effect when combined with RBV [118]. NITD008



Figure 10.2 Summary of the discussed anti-HEV compounds. The depicted molecules/extracts are classified according to the strategy that was used to identify them. So far, the antiviral activity against HEV of only four drugs (sofosbuvir, pegIFN- α , ribavirin, and silvestrol) was approved in experimental settings beyond *in vitro* cell culture systems. Source: [112]. CC BY 4.0.

[114–116] and showed a potency against HEV GT1 replicons with a half maximum effective concentration (EC_{50}) of $0.03 \mu M$ [113]. GPCN114 as non-nucleoside inhibitor with known activity against Picornaviruses was also identified by Netzler and colleagues to inhibit HEV GT1 replicons [113]. For both drugs, no clinical trials have been filed so far (Figure 10.2) [112].

Another molecule discussed as therapeutic agent for HEV is zinc (Figure 10.2; Table 10.1). Zinc treatment inhibited viral replication in the human hepatoma cell culture models of genotype-1 and genotype-3 strains in vitro at a concentration of $IC_{50} = 10 \,\mu\text{M}$ and $IC_{95} = 200 \,\mu\text{M}$ for ZN sulfate and ZN acetate [123]. These data suggested that the antiviral activity of zinc on HEV is likely mediated through its effect on the viral RdRp. However, the observed antiviral effect could be attributed to direct and indirect action(s) of zinc on multiple distinct virus/host targets/processes, as shown for other viruses [140]. A clinical study has not been conducted so far (Figure 10.2), but a case study evaluated the influence of zinc levels on the outcome of RBV treatment. Here, clearance of infection was not associated with higher zinc levels [141].

A recently reported high-throughput screening of a library containing over 1,000 FDAapproved drugs for anti-HEV activity identified deptropine, a histamine H1 receptor antagonist (Table 10.1) used in clinics to treat asthmatic symptoms (Figure 10.2) [122]. Although the exact mode of action remains to be elucidated, the anti-HEV activity of deptropine seems to involve the NF- κ B-RIPK1-caspase axis. Interestingly, the authors find a pronounced antagonistic effect of the identified drug with IFN α , while the effects combined with RBV add up [122]. No clinical data on deptropine in the context of HEV infections are published so far (Figure 10.2).

Besides tackling down the virus by direct-acting drugs, compounds interfering with the host cell biology needed for the obligate intracellular pathogens is another promising Table 10.1 Compounds tested for anti-HEV effect and discussed in this chapter.

Name	Structure	Function	CAS	References
GPCN114 2,2'-[(4-Chloro-1,2-phenylene)bis(oxy)] bis(5-nitrobenzonitrile)		Non-nucleoside inhibitor of the HEV RdRp		[113]
NITD008 7-(2-C-Ethynyl-\$-D-ribofuranosyl)-7H- pyrrolo[2,3-d]pyrimidin-4-amine	$\begin{array}{c} HO \\ HO \\ H \\ H$	Adenosine analogue	1044589-82-3	[114–116]
2'CMC 2'-C-methylcytidine		Nucleoside analogue	20724-73-6	[117, 118]
NM283 3'-O-valinyl ester of 2'-C-methylcytidine (dihydrochloride salt)	HO O O O HA O O HA O O HA O O HA O HA O	Nucleoside analog prodrug	640281-90-9	[119]

(Continued)

Name	Structure	Function	CAS	References
Sofosbuvir Isopropyl (2S)-2-{[(S)-{[(2R,3R,4R,5R)-5- (2,4-diaxo-3,4-dihydro-1(2H)- pyrimidinyl)-4-fluoro-3-hydroxy-4- methyltetrahydro-2-furanyl]methoxy} (phenoxy)phosphoryl]amino]propanoate		Nucleotide conjugate	1190307-88-0	[117]
pegIFNα		Triggers interferon-alpha receptor 1 and 2 (IFNAR1/2)	99210-65-8	[120, 121]
Ribavirin 1-(β-D-Ribofuranosyl)-1H-1,2,4-triazole-3- carboxamide		Nucleoside analogue	36791-04-5	[92, 93, 95]
Deptropine Dibenzheptropine		Antihistamine	604-51-3	[122]

Name	Structure	Function	CAS	References
Zinc	Zn ⁺⁺	Not fully understood		[123]
MPA Mycophenolic acid	HO J J J J J J J J J J J J J J J J J J J	Immuno-suppressant	24280-93-1	[124]
Silvestrol Methyl (1 <i>R</i> .2 <i>R</i> ,3 <i>S</i> ,3 <i>aR</i> ,8bS)-6- (((2 <i>S</i> ,3 <i>R</i> ,6 <i>R</i>)-6-(1 <i>k</i>)-1,2-dihydroxyethyl]-3- methoxy-1,4-dioxan-2-yl] oxy)-1,8b-dihydroxy-8-methoxy-3a-(4- methoxyphenyl)-3-phenyl-2,3,3a,8b- tetrahydro-1H-benzo[b]cyclopenta[d] furan-2-carboxylate	HO CHI COCH OF COCH	eIF4A inhibitor	697235-38-4	[125, 126]

approach [112]. Due to the lack of knowledge in the molecular virology of HEV and its possible host interaction partners, only a few targets with antiviral properties have been evaluated though [47]. They are summarized in the following:

Silvestrol is a structurally unique cyclopenta[b]benzofuran (Table 10.1) that can be isolated along with epi-silvestrol from the plant Aglaia foveolata and other Aglaia species belonging to the family of Meliacea [142]. It is described as a highly efficient, nontoxic, and specific inhibitor of the DEAD-box RNA helicase eIF4A [143], which is part of the eIF4F complex that drives cap-dependent translation initiation in eukaryotes [144]. The molecule has been analyzed mainly in the context of cancer treatments, but has been found in independent studies to exert an antiviral effect against HEV (Figure 10.2) [125, 126]. It was shown that silvestrol was able to reduce viral titers in A549 as well as HepG2 cells and block viral protein synthesis at concentrations of $IC_{50} = 2.9-6.7 \text{ nM}$, $IC_{90} = 27.3-64.7 \text{ nM}$ [125, 126]. This anti-HEV activity was further observed in vitro using laboratory and primary isolates to infect human liver cells and in vivo using humanized mice [126]. Of note, silvestrol was additionally effective against HEV harboring a fitness mutation [99], which renders the virus resistant to RBV in vivo [103]. At the same time, several other studies reported an antiviral effect of silvestrol against Ebola virus at concentrations of $IC_{50} = 96$ nM, Coronovirus at concentrations of $IC_{50} = 1.3-100 \text{ nM}$, $IC_{90} = 12-900 \text{ nM}$ depending on cell line and virus, and Zika virus at concentrations of 5-50 nM (no dose-response relationship) infections using *in vitro* cell culture model systems [145–147]. These results identified silvestrol as a novel natural compound blocking replication of several RNA viruses and may provide a basis for a chemical, biological, and preclinical development of silvestrol as broad-acting novel antiviral agent including HEV. As the antiviral target is a host factor, the emergence of resistance should be lower compared to viral targets; however, blocking essential cellular targets may also come with a risk of side effects.

The inosine-5'-monophosphate dehydrogenase (IMPDH) is another host target that was pursued as an antiviral approach. IMPDH is an essential enzyme in the purine nucleotide synthesis and is part of immunosuppressive regimens in organ transplant recipients. Next to RBV, mycophenolic acid (MPA) (Table 10.1), which targets an enzyme in the nucleotide synthesis pathway, was shown to inhibit HEV replication of GT3 replicons in Huh7 cells (Figure 10.2) [124]. This effect could be reverted by the addition of guanosine. However, these antiviral activities could not be confirmed *in vivo* and no additive inhibition of HEV RNA with RBV and MPA was observed in French patients (Figure 10.2) [90].

In summary, drug design and development for HEV infections are at their beginnings and have focused mainly on drug repurposing approaches with limited success [112]. A better understanding of the replication cycle of HEV is necessary to specifically develop direct-acting antivirals [47].

10.6 Prevention of Infection and Vaccination Strategies

HEV genotypes 1 and 2 are waterborne viruses. Therefore, similar to HAV, clean water facilities can efficiently prevent outbreaks of HEV. It is presumed that HEV, being a non-enveloped virus in the fecal–oral route of transmission, is relatively robust and stable upon environmental harms. The risk of foodborne HEV can be reduced significantly by cooking
meat. *In vitro* assays suggest that cooking meat for one minute at 70 °C leads to a 0.48 log reduction of infectivity and heating to 95 °C lead to pronounced inactivation of 3.67 log of infectivity [148–150]. Blood products are screened for HEV in some countries, but it is unclear whether common inactivation procedures cause elimination of infectious HEV-particles. While direct human-to human transmission of HEV is usually uncommon, patients should be advised to perform cleaning of used sanitary facilities. HAV is partially resistant to 80% ethanol-based disinfectants. Assuming similar environmental stability of HAV and HEV, disinfectants based on 80% ethanol only, should not be recommended for decontamination of HEV either [151]. Overall, intra-family or direct person-to person transmission seems to be seldom events, but have been described and suggested in a few cases [152]. Therefore, testing of partners/family-members should be considered if risk factors for development of severe infection are present in these individuals, in particular having an underlying liver disease, prolonged alcohol abuse, or being pregnant.

In early neutralization studies *in vitro*, infection of hepatoma cells (HepG2/C3A) could be efficiently inhibited with serum of vaccinated or previous infected rhesus monkeys. Of note, the neutralization was cross-reactive: sera of animals, which were infected with genotypes 1, 2, 3, and 4, potently inhibited infectivity of genotype 1 HEV-virus [153]. These findings give hope for the development of a potent, pangenotypic vaccine. In a large-scale vaccine campaign in China from 2011, the vaccine HEV 239 (Hecolin[®]) has been tested in over 50 000 individuals. Indeed, it displayed a strong potency by preventing acute HEV in 94–100% of the cases [56]. No major side effects have been documented and the vaccine got licensed in China in December 2011.

Nevertheless, it has not been approved in other countries yet and long-term prevention has not yet been demonstrated. This vaccine was evaluated by the NIH in a phase 1 clinical trial in US adults and the primary completion date was reached in August 2020 (NCT03827395, https://clinicaltrials.gov/). So far, results from this study have not been published yet. Additionally, the Norwegian Institute of Public Health is testing its efficacy in women with childbearing potential in Bangladesh (Phase 4) (NCT02759991, https://clinicaltrials.gov/). In a different vaccine study from Nepal, anti-HEV-IgG-titer of 120 Walter Reed units/ml (2–6 WHO units/ml) were protective, but lower levels could lead to risk of reinfection [154]. Although this vaccine completed phase 2 trials and showed potent inhibition of infection in 2000 tested individuals, further development was stopped and it has not been approved so far.

10.7 Conclusions

Hepatitis E is not affecting only developing nations, but is a globally existing health threat. Especially, several subgroups—pregnant women, persons with underlying liver diseases, immune-compromised patients—have a risk of high morbidity and mortality by infection with HEV and "standard" experimental treatment with interferon or ribavirin is often not eligible in this cohort due to various side effects.

Therefore, HEV should be excluded by differential diagnostics in any case of disease, associated with elevated liver enzymes. However, until now direct-acting antivirals have not been established, mainly due to lack of efficient in vitro systems. Recent achievements

in HEV cell culture systems have been made [155, 156] that hopefully facilitate the identification of new drug targets and novel antiviral compounds. The first high-throughput screening of compound libraries has been performed and may lead to attractive drugs. Additionally, two vaccine approaches have already proven good safety and potency in protection against HEV infection, although until now, there is no vaccine on the market, which can be applied globally and the duration of protection still needs to be evaluated. This would not only be beneficial and desirable for endemic areas. Patients in developed countries which display certain risk factors would also be candidates in need for a protective vaccine.

References

- **1** Reyes, G.R., Yarbough, P.O., Tam, A.W. et al. (1991). Hepatitis E virus (HEV): the novel agent responsible for enterically transmitted non-A, non-B hepatitis. *Gastroenterologia Japonica* 26 (Suppl 3): 142–147. https://doi.org/10.1007/bf02779285.
- **2** Khuroo, M.S. (2011). Discovery of hepatitis E: the epidemic non-A, non-B hepatitis 30 years down the memory lane. *Virus Research* 161 (1): 3–14. https://doi.org/10.1016/j. virusres.2011.02.007.
- Kane, M.A., Bradley, D.W., Shrestha, S.M. et al. (1984). Epidemic non-A, non-B hepatitis in Nepal. Recovery of a possible etiologic agent and transmission studies in marmosets. *JAMA* 252 (22): 3140–3145. https://doi.org/10.1001/jama.252.22.3140.
- **4** Hla, M., Myint, M.S., Tun, K. et al. (1985). A clinical and epidemiological study of an epidemic of non-A non-B hepatitis in Rangoon. *The American Journal of Tropical Medicine and Hygiene* 34 (6): 1183–1189. https://doi.org/10.4269/ajtmh.1985.34.1183.
- 5 de Cock, K.M., Bradley, D.W., Sandford, N.L. et al. (1987). Epidemic non-A, non-B hepatitis in patients from Pakistan. *Annals of Internal Medicine* 106 (2): 227–230. https://doi.org/1 0.7326/0003-4819-106-2-227.
- **6** Centers for Disease Control and Prevention (1987). Enterically transmitted non-A, non-B hepatitis—Mexico. *MMWR. Morbidity and Mortality Weekly Report* 36 (36): 597–602.
- 7 Zhuang, H. (1991). Advances in the research on non-A, non-B hepatitis in China.
 Zhonghua liu xing bing xue za zhi = Zhonghua liuxingbingxue zazhi 12 (6): 377–379.
- **8** Viswanathan, R. (2013). Infectious hepatitis in Delhi (1955–56): a critical studyepidemiology. 1957. *The National Medical Journal of India* 26 (6): 362–377.
- **9** Viswanathan, R. and Sidhu, A.S. (1957). Infectious hepatitis; clinical findings. *The Indian Journal of Medical Research* 45 (Suppl): 49–58.
- 10 Rein, D.B., Stevens, G.A., Theaker, J. et al. (2012). The global burden of hepatitis E virus genotypes 1 and 2 in 2005. *Hepatology (Baltimore, Md.)* 55 (4): 988–997. https://doi.org/10.1002/hep.25505.
- 11 Wedemeyer, H., Pischke, S., and Manns, M.P. (2012). Pathogenesis and treatment of hepatitis e virus infection. *Gastroenterology* 142 (6): 1388–1397.e1. https://doi. org/10.1053/j.gastro.2012.02.014.
- 12 Mansuy, J.-M., Bendall, R., Legrand-Abravanel, F. et al. (2011). Hepatitis E virus antibodies in blood donors, France. *Emerging Infectious Diseases* 17 (12): 2309–2312. https://doi. org/10.3201/eid1712.110371.

- 13 Bose, P.D., Das, B.C., Kumar, A. et al. (2011). High viral load and deregulation of the progesterone receptor signaling pathway: association with hepatitis E-related poor pregnancy outcome. *Journal of Hepatology* 54 (6): 1107–1113. https://doi.org/10.1016/j. jhep.2010.08.037.
- 14 Smith, D.B., Simmonds, P., Izopet, J. et al. (2016). Proposed reference sequences for hepatitis E virus subtypes. *Journal of General Virology* 97: 537–542. https://doi.org/10.1099/ jgv.0.000393.
- Khuroo, M.S. and Khuroo, M.S. (2016). Hepatitis E: an emerging global disease from discovery towards control and cure. *Journal of Viral Hepatitis* 23 (2): 68–79. https://doi.org/10.1111/jvh.12445.
- **16** Dong, C., Meng, J., Dai, X. et al. (2011). Restricted enzooticity of hepatitis E virus genotypes 1 to 4 in the United States. *Journal of Clinical Microbiology* 49 (12): 4164–4172. https://doi.org/10.1128/JCM.05481-11.
- 17 Goens, S.D. and Perdue, M.L. (2004). Hepatitis E viruses in humans and animals. *Animal Health Research Reviews* 5 (2): 145–156.
- **18** Meng, X.J., Dea, S., Engle, R.E. et al. (1999). Prevalence of antibodies to the hepatitis E virus in pigs from countries where hepatitis E is common or is rare in the human population. *Journal of Medical Virology* 59 (3): 297–302.
- 19 Kaci, S., Nöckler, K., and Johne, R. (2008). Detection of hepatitis E virus in archived German wild boar serum samples. *Veterinary Microbiology* 128 (3-4): 380–385. https://doi. org/10.1016/j.vetmic.2007.10.030.
- 20 Said, B., Ijaz, S., Kafatos, G. et al. (2009). Hepatitis E outbreak on cruise ship. *Emerging Infectious Diseases* 15 (11): 1738–1744. https://doi.org/10.3201/eid1511.091094.
- Tei, S., Kitajima, N., Takahashi, K., and Mishiro, S. (2003). Zoonotic transmission of hepatitis E virus from deer to human beings. *Lancet (London, England)* 362 (9381): 371–373. https://doi.org/10.1016/S0140-6736(03)14025-1.
- 22 Tomiyama, D., Inoue, E., Osawa, Y., and Okazaki, K. (2009). Serological evidence of infection with hepatitis E virus among wild Yezo-deer, Cervus nippon yesoensis, in Hokkaido, Japan. *Journal of Viral Hepatitis* 16 (7): 524–528. https://doi. org/10.1111/j.1365-2893.2009.01107.x.
- Colson, P., Borentain, P., Queyriaux, B. et al. (2010). Pig liver sausage as a source of hepatitis E virus transmission to humans. *The Journal of Infectious Diseases* 202 (6): 825–834. https://doi.org/10.1086/655898.
- 24 Wenzel, J.J., Preiss, J., Schemmerer, M. et al. (2011). Detection of hepatitis E virus (HEV) from porcine livers in Southeastern Germany and high sequence homology to human HEV isolates. *Journal of Clinical Virology: The Official Publication of the Pan American Society for Clinical Virology* 52 (1): 50–54. https://doi.org/10.1016/j.jcv.2011.06.006.
- 25 Lee, G.-H., Tan, B.-H., Teo, E.C.-Y. et al. (2016). Chronic infection with camelid hepatitis E virus in a liver transplant recipient who regularly consumes camel meat and milk. *Gastroenterology* 150 (2): 355–7.e3. https://doi.org/10.1053/j.gastro.2015.10.048.
- 26 Andonov, A., Robbins, M., Borlang, J. et al. (2019). Rat hepatitis E virus linked to severe acute hepatitis in an immunocompetent patient. *The Journal of Infectious Diseases* 220 (6): 951–955. https://doi.org/10.1093/infdis/jiz025.
- 27 Sridhar, S., Yip, C.C.Y., Wu, S. et al. (2018). Rat hepatitis E virus as cause of persistent hepatitis after liver transplant. *Emerging Infectious Diseases* 24 (12): 2241–2250. https://doi.org/10.3201/eid2412.180937.

- **268** 10 Hepatitis E Virus-Current Developments in Antiviral Strategies
 - 28 Balayan, M.S., Andjaparidze, A.G., Savinskaya, S.S. et al. (1983). Evidence for a virus in non-A, non-B hepatitis transmitted via the fecal-oral route. *Intervirology* 20 (1): 23–31. https://doi.org/10.1159/000149370.
 - 29 Emerson, S.U. and Purcell, R.H. (2003). Hepatitis E virus. *Reviews in Medical Virology* 13 (3): 145–154. https://doi.org/10.1002/rmv.384.
 - **30** Graff, J., Torian, U., Nguyen, H., and Emerson, S.U. (2006). A bicistronic subgenomic mRNA encodes both the ORF2 and ORF3 proteins of hepatitis E virus. *Journal of Virology* 80 (12): 5919–5926. https://doi.org/10.1128/JVI.00046-06.
 - **31** Ichiyama, K., Yamada, K., Tanaka, T. et al. (2009). Determination of the 5'-terminal sequence of subgenomic RNA of hepatitis E virus strains in cultured cells. *Archives of Virology* 154 (12): 1945–1951. https://doi.org/10.1007/s00705-009-0538-y.
 - 32 Chapuy-Regaud, S., Dubois, M., Plisson-Chastang, C. et al. (2017). Characterization of the lipid envelope of exosome encapsulated HEV particles protected from the immune response. *Biochimie* 141: 70–79. https://doi.org/10.1016/j.biochi.2017.05.003.
 - **33** Ansari, I.H., Nanda, S.K., Durgapal, H. et al. (2000). Cloning, sequencing, and expression of the hepatitis E virus (HEV) nonstructural open reading frame 1 (ORF1). *Journal of Medical Virology* 60 (3): 275–283.
 - 34 Paliwal, D., Panda, S.K., Kapur, N. et al. (2014). Hepatitis E virus (HEV) protease: a chymotrypsin-like enzyme that processes both non-structural (pORF1) and capsid (pORF2) protein. *The Journal of General Virology* 95 (Pt 8): 1689–1700. https://doi. org/10.1099/vir.0.066142-0.
 - 35 Perttilä, J., Spuul, P., and Ahola, T. (2013). Early secretory pathway localization and lack of processing for hepatitis E virus replication protein pORF1. *The Journal of General Virology* 94 (Pt 4): 807–816. https://doi.org/10.1099/vir.0.049577-0.
 - **36** Ropp, S.L., Tam, A.W., Beames, B. et al. (2000). Expression of the hepatitis E virus ORF1. *Archives of Virology* 145 (7): 1321–1337. https://doi.org/10.1007/s007050070093.
 - **37** Sehgal, D., Thomas, S., Chakraborty, M., and Jameel, S. (2006). Expression and processing of the hepatitis E virus ORF1 nonstructural polyprotein. *Virology Journal* 3: 38. https://doi.org/10.1186/1743-422X-3-38.
 - 38 Suppiah, S., Zhou, Y., and Frey, T.K. (2011). Lack of processing of the expressed ORF1 gene product of hepatitis E virus. *Virology Journal* 8: 245. https://doi. org/10.1186/1743-422X-8-245.
 - **39** Karpe, Y.A. and Lole, K.S. (2011). Deubiquitination activity associated with hepatitis E virus putative papain-like cysteine protease. *The Journal of General Virology* 92 (Pt 9): 2088–2092. https://doi.org/10.1099/vir.0.033738-0.
 - **40** Debing, Y., Moradpour, D., Neyts, J., and Gouttenoire, J. (2016a). Update on hepatitis E virology: implications for clinical practice. *Journal of Hepatology* 65 (1): 200–212. https://doi.org/10.1016/j.jhep.2016.02.045.
 - **41** Nair, V.P., Anang, S., Subramani, C. et al. (2016). Endoplasmic reticulum stress induced synthesis of a novel viral factor mediates efficient replication of genotype-1 hepatitis E virus. *PLoS Pathogens* 12 (4): e1005521. https://doi.org/10.1371/journal. ppat.1005521.
 - **42** Ding, Q., Nimgaonkar, I., Archer, N.F. et al. (2018). Identification of the intragenomic promoter controlling hepatitis E virus subgenomic RNA transcription. *mBio* 9 (3) https://doi.org/10.1128/mBio.00769-18.

- **43** Jameel, S., Zafrullah, M., Ozdener, M.H., and Panda, S.K. (1996). Expression in animal cells and characterization of the hepatitis E virus structural proteins. *Journal of Virology* 70 (1): 207–216.
- **44** Nagashima, S., Takahashi, M., Jirintai et al. (2011a). A PSAP motif in the ORF3 protein of hepatitis E virus is necessary for virion release from infected cells. *The Journal of General Virology* 92 (Pt 2): 269–278. https://doi.org/10.1099/vir.0.025791-0.
- **45** Gouttenoire, J., Pollán, A., Abrami, L. et al. (2018). Palmitoylation mediates membrane association of hepatitis E virus ORF3 protein and is required for infectious particle secretion. *PLoS Pathogens* 14 (12): e1007471. https://doi.org/10.1371/journal.ppat.1007471.
- 46 Ding, Q., Heller, B., Capuccino, J.M.V. et al. (2017). Hepatitis E virus ORF3 is a functional ion channel required for release of infectious particles. *Proceedings of the National Academy of Sciences of the United States of America* 114 (5): 1147–1152. https://doi. org/10.1073/pnas.1614955114.
- **47** Wißing, M.H., Brüggemann, Y., Steinmann, E. and Todt, D. (2021). Virus-host cell interplay during hepatitis E virus infection. *Trends in Microbiology* 29 (4): 309–319.
- **48** Kalia, M., Chandra, V., Rahman, S.A. et al. (2009). Heparan sulfate proteoglycans are required for cellular binding of the hepatitis E virus ORF2 capsid protein and for viral infection. *Journal of Virology* 83 (24): 12714–12724. https://doi.org/10.1128/JVI.00717-09.
- **49** Shiota, T., Li, T.-C., Nishimura, Y. et al. (2019). Integrin α3 is involved in non-enveloped hepatitis E virus infection. *Virology* 536: 119–124. https://doi.org/10.1016/j.virol.2019.07.025.
- Kapur, N., Thakral, D., Durgapal, H., and Panda, S.K. (2012). Hepatitis E virus enters liver cells through receptor-dependent clathrin-mediated endocytosis. *Journal of Viral Hepatitis* 19 (6): 436–448. https://doi.org/10.1111/j.1365-2893.2011.01559.x.
- 51 Holla, P., Ahmad, I., Ahmed, Z., and Jameel, S. (2015). Hepatitis E virus enters liver cells through a dynamin-2, clathrin and membrane cholesterol-dependent pathway. *Traffic (Copenhagen, Denmark)* 16 (4): 398–416. https://doi.org/10.1111/tra.12260.
- **52** Varma, S.P.K., Kumar, A., Kapur, N. et al. (2011). Hepatitis E virus replication involves alternating negative- and positive-sense RNA synthesis. *The Journal of General Virology* 92 (Pt 3): 572–581. https://doi.org/10.1099/vir.0.027714-0.
- **53** Nagashima, S., Takahashi, M., Jirintai, S. et al. (2011b). Tumour susceptibility gene 101 and the vacuolar protein sorting pathway are required for the release of hepatitis E virions. *The Journal of General Virology* 92 (Pt 12): 2838–2848. https://doi.org/10.1099/ vir.0.035378-0.
- **54** Montpellier, C., Wychowski, C., Sayed, I.M. et al. (2018). Hepatitis E virus lifecycle and identification of 3 forms of the ORF2 capsid protein. *Gastroenterology* 154 (1): 211–223.e8. https://doi.org/10.1053/j.gastro.2017.09.020.
- 55 Okamoto, H. (2013). Culture systems for hepatitis E virus. *Journal of Gastroenterology* 48 (2): 147–158. https://doi.org/10.1007/s00535-012-0682-0.
- 56 Zhu, F.-C., Zhang, J., Zhang, X.-F. et al. (2010). Efficacy and safety of a recombinant hepatitis E vaccine in healthy adults: a large-scale, randomised, double-blind placebocontrolled, phase 3 trial. *Lancet (London, England)* 376 (9744): 895–902. https://doi. org/10.1016/S0140-6736(10)61030-6.
- 57 Teshale, E.H., Hu, D.J., and Holmberg, S.D. (2010). The two faces of hepatitis E virus. *Clinical Infectious Diseases: An Official Publication of the Infectious Diseases Society of America* 51 (3): 328–334. https://doi.org/10.1086/653943.

- **270** 10 Hepatitis E Virus-Current Developments in Antiviral Strategies
 - 58 Bhatia, V., Singhal, A., Panda, S.K., and Acharya, S.K. (2008). A 20-year single-center experience with acute liver failure during pregnancy: is the prognosis really worse? *Hepatology (Baltimore, Md.)* 48 (5): 1577–1585. https://doi.org/10.1002/hep.22493.
 - **59** Patra, S., Kumar, A., Trivedi, S.S. et al. (2007). Maternal and fetal outcomes in pregnant women with acute hepatitis E virus infection. *Annals of Internal Medicine* 147 (1): 28–33. https://doi.org/10.7326/0003-4819-147-1-200707030-00005.
 - 60 Dalton, H.R., Hazeldine, S., Banks, M. et al. (2007). Locally acquired hepatitis E in chronic liver disease. *Lancet (London, England)* 369 (9569): 1260. https://doi.org/10.1016/S0140-6736(07)60595-9.
 - **61** Hamid, S.S., Atiq, M., Shehzad, F. et al. (2002). Hepatitis E virus superinfection in patients with chronic liver disease. *Hepatology (Baltimore, Md.)* 36 (2): 474–478. https://doi.org/10.1053/jhep.2002.34856.
 - **62** Kumar, M., Sharma, B.C., and Sarin, S.K. (2008). Hepatitis E virus as an etiology of acute exacerbation of previously unrecognized asymptomatic patients with hepatitis B virus-related chronic liver disease. *Journal of Gastroenterology and Hepatology* 23 (6): 883–887. https://doi.org/10.1111/j.1440-1746.2007.05243.x.
 - **63** Ramachandran, J., Eapen, C.E., Kang, G. et al. (2004). Hepatitis E superinfection produces severe decompensation in patients with chronic liver disease. *Journal of Gastroenterology and Hepatology* 19 (2): 134–138. https://doi.org/10.1111/j.1440-1746.2004.03188.x.
 - 64 Péron, J.M., Bureau, C., Poirson, H. et al. (2007). Fulminant liver failure from acute autochthonous hepatitis E in France: description of seven patients with acute hepatitis E and encephalopathy. *Journal of Viral Hepatitis* 14 (5): 298–303. https://doi. org/10.1111/j.1365-2893.2007.00858.x.
 - 65 Pischke, S., Hardtke, S., Bode, U. et al. (2013). Ribavirin treatment of acute and chronic hepatitis E: a single-centre experience. *Liver International: Official Journal of the International Association for the Study of the Liver* 33 (5): 722–726. https://doi.org/10.1111/liv.12114.
 - 66 Purcell, R.H. and Emerson, S.U. (2008). Hepatitis E: an emerging awareness of an old disease. *Journal of Hepatology* 48 (3): 494–503. https://doi.org/10.1016/j. jhep.2007.12.008.
 - **67** Behrendt, P., Steinmann, E., Manns, M.P., and Wedemeyer, H. (2014). The impact of hepatitis E in the liver transplant setting. *Journal of Hepatology* 61 (6): 1418–1429. https://doi.org/10.1016/j.jhep.2014.08.047.
 - **68** Hoofnagle, J.H., Nelson, K.E., and Purcell, R.H. (2012). Hepatitis E. *The New England Journal of Medicine* 367 (13): 1237–1244. https://doi.org/10.1056/NEJMra1204512.
 - **69** Kamar, N. and Pischke, S. (2019). Acute and persistent hepatitis E virus genotype 3 and 4 infection: clinical features, pathogenesis, and treatment. *Cold Spring Harbor Perspectives in Medicine* 9 (7) https://doi.org/10.1101/cshperspect.a031872.
 - **70** Scobie, L. and Dalton, H.R. (2013). Hepatitis E: source and route of infection, clinical manifestations and new developments. *Journal of Viral Hepatitis* 20 (1): 1–11. https://doi. org/10.1111/jvh.12024.
 - 71 Fousekis, F.S., Mitselos, I.V., and Christodoulou, D.K. (2019). Extrahepatic manifestations of hepatitis E virus: an overview. *Clinical and Molecular Hepatology*. https://doi. org/10.3350/cmh.2019.0082.

- **72** Belbezier, A., Deroux, A., Sarrot-Reynauld, F. et al. (2014). Myasthenia gravis associated with acute hepatitis E infection in immunocompetent woman. *Emerging Infectious Diseases* 20 (5): 908–910. https://doi.org/10.3201/eid2005.131551.
- 73 Dalton, H.R., Kamar, N., van Eijk, J.J.J. et al. (2016). Hepatitis E virus and neurological injury. *Nature Reviews. Neurology* 12 (2): 77–85. https://doi.org/10.1038/nrneurol.2015.234.
- 74 Del Bello, A., Arné-Bes, M.C., Lavayssière, L., and Kamar, N. (2012). Hepatitis E virusinduced severe myositis. *Journal of Hepatology* 57 (5): 1152–1153. https://doi.org/10.1016/j. jhep.2012.05.010.
- 75 Kamar, N., Izopet, J., Cintas, P. et al. (2010a). Hepatitis E virus-induced neurological symptoms in a kidney-transplant patient with chronic hepatitis. *American Journal of Transplantation: Official Journal of the American Society of Transplantation and the American Society of Transplant Surgeons* 10 (5): 1321–1324. https://doi.org/10.1111/j.1600-6143.2010.03068.x.
- 76 Kamar, N., Bendall, R.P., Peron, J.M. et al. (2011). Hepatitis E virus and neurologic disorders. *Emerging Infectious Diseases* 17 (2): 173–179. https://doi.org/10.3201/eid1702.100856.
- Pischke, S., Behrendt, P., Manns, M.P., and Wedemeyer, H. (2014). HEV-associated cryoglobulinaemia and extrahepatic manifestations of hepatitis E. *The Lancet Infectious Diseases* 14 (8): 678–679. https://doi.org/10.1016/S1473-3099(14)70823-0.
- **78** Silva, M., Wicki, B., Tsouni, P. et al. (2016). Hepatitis E virus infection as a direct cause of neuralgic amyotrophy. *Muscle & Nerve* 54 (2): 325–327. https://doi.org/10.1002/mus.25096.
- **79** van den Berg, B., van der Eijk, A.A., Pas, S.D. et al. (2014). Guillain–Barré syndrome associated with preceding hepatitis E virus infection. *Neurology* 82 (6): 491–497. https://doi.org/10.1212/WNL.00000000000111.
- **80** Shi, R., Soomro, M.H., She, R. et al. (2016). Evidence of hepatitis E virus breaking through the blood–brain barrier and replicating in the central nervous system. *Journal of Viral Hepatitis* 23 (11): 930–939. https://doi.org/10.1111/jvh.12557.
- **81** Drave, S.A., Debing, Y., Walter, S. et al. (2016). Extra-hepatic replication and infection of hepatitis E virus in neuronal-derived cells. *Journal of Viral Hepatitis* 23 (7): 512–521. https://doi.org/10.1111/jvh.12515.
- **82** Knegendorf, L., Drave, S.A., Thi, D. et al. (2018). Hepatitis E virus replication and interferon responses in human placental cells. *Hepatology Communications* 2 (2): 173–187. https://doi.org/10.1002/hep4.1138.
- **83** Khudyakov, Y. and Kamili, S. (2011). Serological diagnostics of hepatitis E virus infection. *Virus Research* 161 (1): 84–92. https://doi.org/10.1016/j.virusres.2011.06.006.
- 84 Suneetha, P.V., Pischke, S., Schlaphoff, V. et al. (2012). Hepatitis E virus (HEV)-specific T-cell responses are associated with control of HEV infection. *Hepatology (Baltimore, Md.)* 55 (3): 695–708. https://doi.org/10.1002/hep.24738.
- **85** Grandadam, M., Tebbal, S., Caron, M. et al. (2004). Evidence for hepatitis E virus quasispecies. *The Journal of General Virology* 85 (Pt 11): 3189–3194. https://doi.org/10.1099/vir.0.80248-0.
- **86** Lhomme, S., Abravanel, F., Dubois, M. et al. (2012). Hepatitis E virus quasispecies and the outcome of acute hepatitis E in solid-organ transplant patients. *Journal of Virology* 86 (18): 10006–10014. https://doi.org/10.1128/JVI.01003-12.

- **272** 10 Hepatitis E Virus-Current Developments in Antiviral Strategies
 - 87 Moal, V., Textoris, J., Ben Amara, A. et al. (2013). Chronic hepatitis E virus infection is specifically associated with an interferon-related transcriptional program. *The Journal of Infectious Diseases* 207 (1): 125–132. https://doi.org/10.1093/infdis/jis632.
 - 88 Nandakumar, R., Finsterbusch, K., Lipps, C. et al. (2013). Hepatitis C virus replication in mouse cells is restricted by IFN-dependent and -independent mechanisms. *Gastroenterology* 145 (6): 1414–23.e1. https://doi.org/10.1053/j.gastro.2013.08.037.
 - **89** EASL (2018). Clinical Practice Guidelines on hepatitis E virus infection. *Journal of Hepatology* 68 (6): 1256–1271.
 - **90** Kamar, N., Abravanel, F., Selves, J. et al. (2010c). Influence of immunosuppressive therapy on the natural history of genotype 3 hepatitis-E virus infection after organ transplantation. *Transplantation* 89 (3): 353–360. https://doi.org/10.1097/TP.0b013e3181c4096c.
 - **91** Pischke, S., Suneetha, P.V., Baechlein, C. et al. (2010). Hepatitis E virus infection as a cause of graft hepatitis in liver transplant recipients. *Liver Transplantation: Official Publication of the American Association for the Study of Liver Diseases and the International Liver Transplantation Society* 16 (1): 74–82. https://doi.org/10.1002/lt.21958.
 - **92** Kamar, N., Rostaing, L., Abravanel, F. et al. (2010d). Ribavirin therapy inhibits viral replication on patients with chronic hepatitis e virus infection. *Gastroenterology* 139 (5): 1612–1618. https://doi.org/10.1053/j.gastro.2010.08.002.
 - **93** Mallet, V., Nicand, E., Sultanik, P. et al. (2010). Brief communication: case reports of ribavirin treatment for chronic hepatitis E. *Annals of Internal Medicine* 153 (2): 85–89. https://doi.org/10.7326/0003-4819-153-2-201007200-00257.
 - **94** Debing, Y., Emerson, S.U., Wang, Y. et al. (2014a). Ribavirin inhibits in vitro hepatitis E virus replicatisson through depletion of cellular GTP pools and is moderately synergistic with alpha interferon. *Antimicrobial Agents and Chemotherapy* 58 (1): 267–273. https://doi.org/10.1128/AAC.01795-13.
 - 95 Kamar, N., Izopet, J., Tripon, S. et al. (2014). Ribavirin for chronic hepatitis E virus infection in transplant recipients. *The New England Journal of Medicine* 370 (12): 1111–1120. https://doi.org/10.1056/NEJMoa1215246.
 - **96** Robbins, A., Lambert, D., Ehrhard, F. et al. (2014). Severe acute hepatitis E in an HIV infected patient: successful treatment with ribavirin. *Journal of Clinical Virology: The Official Publication of the Pan American Society for Clinical Virology* 60 (4): 422–423. https://doi.org/10.1016/j.jcv.2014.05.003.
 - **97** Pischke, S., Stiefel, P., Franz, B. et al. (2012). Chronic hepatitis e in heart transplant recipients. *American Journal of Transplantation: Official Journal of the American Society of Transplantation and the American Society of Transplant Surgeons* 12 (11): 3128–3133. https://doi.org/10.1111/j.1600-6143.2012.04200.x.
 - **98** Debing, Y., Ramière, C., Dallmeier, K. et al. (2016b). Hepatitis E virus mutations associated with ribavirin treatment failure result in altered viral fitness and ribavirin sensitivity. *Journal of Hepatology* 65 (3): 499–508. https://doi.org/10.1016/j.jhep.2016.05.002.
 - **99** Todt, D., Gisa, A., Radonic, A. et al. (2016b). In vivo evidence for ribavirin-induced mutagenesis of the hepatitis E virus genome. *Gut* 65 (10): 1733–1743. https://doi. org/10.1136/gutjnl-2015-311000.
 - Lhomme, S., Kamar, N., Nicot, F. et al. (2015). Mutation in the hepatitis E virus polymerase and outcome of ribavirin therapy. *Antimicrobial Agents and Chemotherapy* 60 (3): 1608–1614. https://doi.org/10.1128/AAC.02496-15.

- **101** Graci, J.D. and Cameron, C.E. (2006). Mechanisms of action of ribavirin against distinct viruses. *Reviews in Medical Virology* 16 (1): 37–48. https://doi.org/10.1002/rmv.483.
- 102 Paeshuyse, J., Dallmeier, K., and Neyts, J. (2011). Ribavirin for the treatment of chronic hepatitis C virus infection: a review of the proposed mechanisms of action. *Current Opinion in Virology* 1 (6): 590–598. https://doi.org/10.1016/j.coviro.2011.10.030.
- **103** Todt, D., Meister, T.L., and Steinmann, E. (2018a). Hepatitis E virus treatment and ribavirin therapy: viral mechanisms of nonresponse. *Current Opinion in Virology* 32: 80–87. https://doi.org/10.1016/j.coviro.2018.10.001.
- **104** Todt, D., Walter, S., Brown, R.J.P., and Steinmann, E. (2016c). Mutagenic effects of ribavirin on hepatitis E virus-viral extinction versus selection of fitness-enhancing mutations. *Viruses* 8 (10) https://doi.org/10.3390/v8100283.
- 105 Thomas, E., Ghany, M.G., and Liang, T.J. (2012). The application and mechanism of action of ribavirin in therapy of hepatitis C. *Antiviral Chemistry & Chemotherapy* 23 (1): 1–12. https://doi.org/10.3851/IMP2125.
- **106** Contreras, A.M., Hiasa, Y., He, W. et al. (2002). Viral RNA mutations are region specific and increased by ribavirin in a full-length hepatitis C virus replication system. *Journal of Virology* 76 (17): 8505–8517. https://doi.org/10.1128/jvi.76.17.8505-8517.2002.
- 107 Lanford, R.E., Chavez, D., Guerra, B. et al. (2001). Ribavirin induces error-prone replication of GB virus B in primary tamarin hepatocytes. *Journal of Virology* 75 (17): 8074–8081. https://doi.org/10.1128/jvi.75.17.8074-8081.2001.
- **108** Severson, W.E., Schmaljohn, C.S., Javadian, A., and Jonsson, C.B. (2003). Ribavirin causes error catastrophe during Hantaan virus replication. *Journal of Virology* 77 (1): 481–488. https://doi.org/10.1128/jvi.77.1.481-488.2003.
- 109 Sierra, M., Airaksinen, A., González-López, C. et al. (2007). Foot-and-mouth disease virus mutant with decreased sensitivity to ribavirin: implications for error catastrophe. *Journal* of Virology 81 (4): 2012–2024. https://doi.org/10.1128/JVI.01606-06.
- **110** Vignuzzi, M., Stone, J.K., and Andino, R. (2005). Ribavirin and lethal mutagenesis of poliovirus: molecular mechanisms, resistance and biological implications. *Virus Research* 107 (2): 173–181. https://doi.org/10.1016/j.virusres.2004.11.007.
- **111** Debing, Y., Gisa, A., Dallmeier, K. et al. (2014b). A mutation in the hepatitis E virus RNA polymerase promotes its replication and associates with ribavirin treatment failure in organ transplant recipients. *Gastroenterology* 147 (5): 1008–11.e7. quiz e15-6. DOI: https://doi.org/10.1053/j.gastro.2014.08.040.
- **112** Kinast, V., Burkard, T.L., Todt, D., and Steinmann, E. (2019). Hepatitis E virus drug development. *Viruses* 11 (6) https://doi.org/10.3390/v11060485.
- **113** Netzler, N.E., Enosi Tuipulotu, D., Vasudevan, S.G. et al. (2019). Antiviral candidates for treating hepatitis E virus infection. *Antimicrobial Agents and Chemotherapy* 63 (6) https://doi.org/10.1128/AAC.00003-19.
- 114 Deng, Y.-Q., Zhang, N.-N., Li, C.-F. et al. (2016). Adenosine analog NITD008 is a potent inhibitor of Zika virus. Open Forum Infectious Diseases 3 (4): ofw175. https://doi. org/10.1093/ofid/ofw175.
- 115 Nelson, J., Roe, K., Orillo, B. et al. (2015). Combined treatment of adenosine nucleoside inhibitor NITD008 and histone deacetylase inhibitor vorinostat represents an immunotherapy strategy to ameliorate West Nile virus infection. *Antiviral Research* 122: 39–45. https://doi.org/10.1016/j.antiviral.2015.07.008.

- 274 10 Hepatitis E Virus-Current Developments in Antiviral Strategies
 - 116 Qing, J., Luo, R., Wang, Y. et al. (2016). Resistance analysis and characterization of NITD008 as an adenosine analog inhibitor against hepatitis C virus. *Antiviral Research* 126: 43–54. https://doi.org/10.1016/j.antiviral.2015.12.010.
 - 117 Dao Thi, V.L., Debing, Y., Wu, X. et al. (2016). Sofosbuvir inhibits hepatitis E virus replication in vitro and results in an additive effect when combined with ribavirin. *Gastroenterology* 150 (1): 82–85.e4. https://doi.org/10.1053/j.gastro.2015.09.011.
 - **118** Qu, C., Xu, L., Yin, Y. et al. (2017). Nucleoside analogue 2'-C-methylcytidine inhibits hepatitis E virus replication but antagonizes ribavirin. *Archives of Virology* 162 (10): 2989–2996. https://doi.org/10.1007/s00705-017-3444-8.
 - 119 Arnold, J.J., Sharma, S.D., Feng, J.Y. et al. (2012). Sensitivity of mitochondrial transcription and resistance of RNA polymerase II dependent nuclear transcription to antiviral ribonucleosides. *PLoS Pathogens* 8 (11): e1003030. https://doi.org/10.1371/journal.ppat.1003030.
 - 120 Haagsma, E.B., Riezebos-Brilman, A., van den Berg, A.P. et al. (2010). Treatment of chronic hepatitis E in liver transplant recipients with pegylated interferon alpha-2b. *Liver Transplantation: Official Publication of the American Association for the Study of Liver Diseases and the International Liver Transplantation Society* 16 (4): 474–477. https://doi. org/10.1002/lt.22014.
 - 121 Kamar, N., Abravanel, F., Garrouste, C. et al. (2010b). Three-month pegylated interferonalpha-2a therapy for chronic hepatitis E virus infection in a haemodialysis patient. *Nephrology, Dialysis, Transplantation: Official Publication of the European Dialysis and Transplant Association—European Renal Association* 25 (8): 2792–2795. https://doi. org/10.1093/ndt/gfq282.
 - **122** Qu, C., Li, Y., Li, Y. et al. (2019). FDA-drug screening identifies deptropine inhibiting hepatitis E virus involving the NF-κB-RIPK1-caspase axis. *Antiviral Research* 170: 104588. https://doi.org/10.1016/j.antiviral.2019.104588.
 - 123 Kaushik, N., Subramani, C., Anang, S. et al. (2017). Zinc salts block hepatitis E virus replication by inhibiting the activity of viral RNA-dependent RNA polymerase. *Journal of Virology* 91 (21) https://doi.org/10.1128/JVI.00754-17.
 - Wang, Y., Zhou, X., Debing, Y. et al. (2014). Calcineurin inhibitors stimulate and mycophenolic acid inhibits replication of hepatitis E virus. *Gastroenterology* 146 (7): 1775–1783. https://doi.org/10.1053/j.gastro.2014.02.036.
 - 125 Glitscher, M., Himmelsbach, K., Woytinek, K. et al. (2018). Inhibition of hepatitis E virus spread by the natural compound silvestrol. *Viruses* 10 (6) https://doi.org/10.3390/ v10060301.
 - 126 Todt, D., Moeller, N., Praditya, D. et al. (2018b). The natural compound silvestrol inhibits hepatitis E virus (HEV) replication in vitro and in vivo. *Antiviral Research* 157: 151–158. https://doi.org/10.1016/j.antiviral.2018.07.010.
 - Bhatia, H.K., Singh, H., Grewal, N., and Natt, N.K. (2014). Sofosbuvir: a novel treatment option for chronic hepatitis C infection. *Journal of Pharmacology & Pharmacotherapeutics* 5 (4): 278–284. https://doi.org/10.4103/0976-500X.142464.
 - 128 Li, T.-C., Bai, H., Yoshizaki, S. et al. (2019). Genotype 5 hepatitis E virus produced by a reverse genetics system has the potential for zoonotic infection. *Hepatology Communications* 3 (1): 160–172. https://doi.org/10.1002/hep4.1288.

- **129** Nishiyama, T., Kobayashi, T., Jirintai, S. et al. (2019). Screening of novel drugs for inhibiting hepatitis E virus replication. *Journal of Virological Methods* 270: 1–11. https://doi.org/10.1016/j.jviromet.2019.04.017.
- **130** Wang, W., Hakim, M.S., Nair, V.P. et al. (2016). Distinct antiviral potency of sofosbuvir against hepatitis C and E viruses. *Gastroenterology* 151 (6): 1251–1253. https://doi.org/10.1053/j.gastro.2016.09.061.
- Kamar, N. and Pan, Q. (2019). No clear evidence for an effect of sofosbuvir against hepatitis E virus in organ transplant patients. *Hepatology (Baltimore, Md.)* 69 (4): 1846–1847. https://doi.org/10.1002/hep.30546.
- 132 Drinane, M., Jing Wang, X., and Watt, K. (2019). Sofosbuvir and ribavirin eradication of refractory hepatitis E in an immunosuppressed kidney transplant recipient. *Hepatology* (*Baltimore, Md.*) 69 (5): 2297–2299. https://doi.org/10.1002/hep.30428.
- 133 Fraga, M., Gouttenoire, J., Sahli, R. et al. (2019). Sofosbuvir add-on to ribavirin for chronic hepatitis E in a cirrhotic liver transplant recipient: a case report. *BMC Gastroenterology* 19 (1): 76. https://doi.org/10.1186/s12876-019-0995-z.
- **134** Schulz, M., Papp, C.P., Bock, C.-T. et al. (2019). Combination therapy of sofosbuvir and ribavirin fails to clear chronic hepatitis E infection in a multivisceral transplanted patient. *Journal of Hepatology* 71 (1): 225–227. https://doi.org/10.1016/j.jhep.2019.03.029.
- Todesco, E., Demeret, S., Calin, R. et al. (2017). Chronic hepatitis E in HIV/HBV coinfected patient: lack of power of sofosbuvir-ribavirin. *AIDS (London, England)* 31 (9): 1346–1348. https://doi.org/10.1097/QAD.00000000001474.
- 136 Todesco, E., Mazzola, A., Akhavan, S. et al. (2018). Chronic hepatitis E in a heart transplant patient: sofosbuvir and ribavirin regimen not fully effective. *Antiviral Therapy* 23 (5): 463–465. https://doi.org/10.3851/IMP3227.
- 137 van der Valk, M., Zaaijer, H.L., Kater, A.P., and Schinkel, J. (2017). Sofosbuvir shows antiviral activity in a patient with chronic hepatitis E virus infection. *Journal of Hepatology* 66 (1): 242–243. https://doi.org/10.1016/j.jhep.2016.09.014.
- 138 Cornberg, M. (2019). Efficacy and safety of sofosbuvir monotherapy in patients with chronic hepatitis E-The HepNet SofE pilot study. *EASL 2019. European Association for the Study of the Liver*, 2019. https://www.journal-of-hepatology.eu/article/S0618-8278(19)30228-2/pdf (checked on 30 october 2019).
- **139** Pushpakom, S., Iorio, F., Eyers, P. et al. (2019). Drug repurposing: progress, challenges and recommendations. *Nature Review Drug Discovery* 18: 41–58. https://doi.org/10.1038/nrd.2018.168.
- **140** Kaushik, N., Anang, S., Ganti, K.P., and Surjit, M. (2018). Zinc: a potential antiviral against hepatitis E virus infection? *DNA and Cell Biology* 37 (7): 593–599. https://doi. org/10.1089/dna.2018.4175.
- Marion, O., Abravanel, F., Izopet, J., and Kamar, N. (2019). Failure to respond to ribavirin despite elevated intra-erythrocyte zinc level in transplant-patients with chronic hepatitis E virus infection. *Transplant Infectious Disease: An Official Journal of the Transplantation Society* 21 (2): e13050. https://doi.org/10.1111/tid.13050.
- 142 Kim, S., Hwang, B.Y., Su, B.-N. et al. (2007). Silvestrol, a potential anticancer rocaglate derivative from Aglaia foveolata, induces apoptosis in LNCaP cells through the mitochondrial/apoptosome pathway without activation of executioner caspase-3 or -7. *Anticancer Research* 27 (4B): 2175–2183.

- **276** 10 Hepatitis E Virus-Current Developments in Antiviral Strategies
 - **143** Bordeleau, M.-E., Robert, F., Gerard, B. et al. (2008). Therapeutic suppression of translation initiation modulates chemosensitivity in a mouse lymphoma model. *The Journal of Clinical Investigation* 118 (7): 2651–2660. https://doi.org/10.1172/JCI34753.
 - **144** Silvera, D., Formenti, S.C., and Schneider, R.J. (2010). Translational control in cancer. *Nature Reviews. Cancer* 10 (4): 254–266. https://doi.org/10.1038/nrc2824.
 - **145** Biedenkopf, N., Lange-Grünweller, K., Schulte, F.W. et al. (2017). The natural compound silvestrol is a potent inhibitor of Ebola virus replication. *Antiviral Research* 137: 76–81. https://doi.org/10.1016/j.antiviral.2016.11.011.
 - **146** Elgner, F., Sabino, C., Basic, M. et al. (2018). Inhibition of Zika virus replication by silvestrol. *Viruses* 10 (4) https://doi.org/10.3390/v10040149.
 - **147** Müller, C., Schulte, F.W., Lange-Grünweller, K. et al. (2018). Broad-spectrum antiviral activity of the eIF4A inhibitor silvestrol against corona- and picornaviruses. *Antiviral Research* 150: 123–129. https://doi.org/10.1016/j.antiviral.2017.12.010.
 - Emerson, S.U., Arankalle, V.A., and Purcell, R.H. (2005). Thermal stability of hepatitis E virus. *The Journal of Infectious Diseases* 192 (5): 930–933. https://doi.org/10.1086/432488.
 - 149 Johne, R., Trojnar, E., Filter, M., and Hofmann, J. (2016). Thermal stability of hepatitis E virus as estimated by a cell culture method. *Applied and Environmental Microbiology* 82 (14): 4225–4231. https://doi.org/10.1128/AEM.00951-16.
 - **150** Schielke, A., Filter, M., Appel, B., and Johne, R. (2011). Thermal stability of hepatitis E virus assessed by a molecular biological approach. *Virology Journal* 8: 487. https://doi.org/10.1186/1743-422X-8-487.
 - 151 Terpstra, F.G., van den Blink, A.E., Bos, L.M. et al. (2007). Resistance of surface-dried virus to common disinfection procedures. *The Journal of Hospital Infection* 66 (4): 332–338. https://doi.org/10.1016/j.jhin.2007.05.005.
 - Mansuy, J.-M., Huynh, A., Abravanel, F. et al. (2009). Molecular evidence of patient-topatient transmission of hepatitis E virus in a hematology ward. *Clinical Infectious Diseases: An Official Publication of the Infectious Diseases Society of America* 48 (3): 373–374. https://doi.org/10.1086/595893.
 - **153** Emerson, S.U., Clemente-Casares, P., Moiduddin, N. et al. (2006). Putative neutralization epitopes and broad cross-genotype neutralization of hepatitis E virus confirmed by a quantitative cell-culture assay. *The Journal of General Virology* 87 (Pt 3): 697–704. https://doi.org/10.1099/vir.0.81545-0.
 - 154 Shrestha, M.P., Scott, R.M.N., Joshi, D.M. et al. (2007). Safety and efficacy of a recombinant hepatitis E vaccine. *The New England Journal of Medicine* 356 (9): 895–903. https://doi.org/10.1056/NEJMoa061847.
 - **155** Meister, T.L., Bruening, J., Todt, D., and Steinmann, E. (2019). Cell culture systems for the study of hepatitis E virus. *Antivsssiral Research* 163: 34–49. https://doi.org/10.1016/j. antiviral.2019.01.007.
 - **156** Todt, D., Friesland, M., Moeller, N. et al.(2020). Robust hepatitis E virus infection and transcriptional response in human hepatocytes. *Proceedings of the National Academy of Sciences* 117(3): 1731–1741.

11

Antiviral Therapy of Adenovirus Infections

Peter Lischka

AiCuris Anti-infective Cures AG, Wuppertal, Germany

11.1 Human Adenovirus

Human adenoviruses (HAdVs) are ubiquitous, non-enveloped, linear double-stranded DNA viruses first isolated over 65 years ago from human adenoid tissue [1]. Since then, more than 80 different serotypes have been identified, classified into seven subgroups (A–G) primarily on the basis of genomic sequence homology, oncogenicity in rodents, and the ability to agglutinate erythrocytes of different species [2, 3]. Virus transmission in the general population occurs through close contact with infected bodily fluids, aerosol droplets, water, contaminated fomites, or medical instruments. In contrast to enveloped viruses, HAdVs remain infectious outside of the body for extended periods of time (e.g. up to three weeks on surfaces) and are highly resistant to physical and chemical reagents; these properties render the virus highly contagious [4].

The infectious virus particles have a diameter of 70–100 nm and are composed of two major structural elements: (i) the nucleoprotein core that contains the genomic DNA (~36 kb, encoding ~35 genes) and (ii) the outer icosahedral capsid which is largely made up of the major virus protein, hexon. The virus capsid contains 12 projecting spikes at the vertices of the icosahedra consisting of thin fibers attached to the penton base [3, 5, 6]. These spikes, together with the hypervariable regions of the capsid hexons, mediate attachment (and ultimate entry) to one of several specific cellular surface receptors such as CAR (coxsackievirus and adenovirus receptor), CD46, sialic acid, etc., and thus confer tissue specificity (Figure 11.2) [8, 9]. Interestingly, cellular receptors for HAdV appear to vary between distinct virus subgroups which might contribute to the varying tissue tropism and the different disease associations observed for distinct HAdV serotypes [4, 5, 10, 11].

In fact, HAdVs are common human pathogens responsible for a wide range of human diseases including respiratory tract infections, gastrointestinal tract disorders, genitourinary diseases, or ocular infections (Table 11.1). While HAdV infections mostly cause mild and self-limiting diseases in immunocompetent adults and children, there are at least two

	Adenovirus serotypes associated with disease (sorted in subgroups)							
Syndrome	A ^a	В	c	D	E	F	G	
Gastroenteritis	12 ^b , 18, 31					40, 41	52	
Hemmorrhagic cysitis		11, 21						
Hepatitis			1, 2, 5					
Upper respiratory disease		3, 7, 21	1, 2, 5, 6					
Lower respiratory disease		3, 7, 14, 21	1, 2		4			
Pertussis syndrome			5					
Pharyngoconjunctival fever		3, 7, 14	2		4			
Follicular conjunctivitis		3, 7, 11			4			
Epidemic keratokonjunctivitis				8, 37, 53, 54, 56, 64				

 Table 11.1
 Human diseases caused by adenoviruses.

a) Adenovirus subgroup.

b) Adenovirus serotype.



Figure 11.1 Chemical structures of antiviral drugs used for therapy of HAdV infections. (a) Drugs currently marketed for an unrelated indication and used "off label" for systemic treatment of HAdV infections in HSCT patients. (b) Novel antiviral drug in clinical development for systemic HAdV treatment. (c) Povidone-iodine, an extended release formulation of this disinfectant is currently in clinical development for the topical treatment of acute HAdV conjunctivitis.

common manifestations of major clinical concern: (i) severe ocular infections, i.e. epidemic keratoconjunctivitis (EKC) and (ii) infections of immunocompromised individuals, in particular transplant recipients undergoing an hematopoietic stem cell transplantation (HSCT). A HAdV infection in the latter patient population is often associated with serious and even life-threatening disease, whereas a severe HAdV eye infection carries the risk of potential long-term consequences for visual acuity [3, 4].

Despite the significant clinical impact of HAdV, there are currently no approved drugs available to combat these infections. A live oral vaccine against serotypes 4 and 7 was developed exclusively for the US military to limit HAdV infections in recruits [12]. However, since this vaccine is not available to the general population, there remains no vaccine available for general use in children or adults in any part of the world [5]. Given this, current antiviral therapies are entirely dependent upon suboptimal "off-label" therapeutic options or palliative care. Hence, there is a high unmet medical need for potent and specific antiadenovirus drugs, particularly for severely immunocompromised HSCT patients and patients suffering from severe eye infections like EKC.

11.2 Adenovirus in Human Stem Cell Transplantation

11.2.1 Incidence, Transmission, and Clinical Manifestation

Worldwide, more than 140000 solid organ- and 90000 bone marrow transplants are performed annually and viral infections continue to be a major cause of morbidity and mortality following both solid organ transplantation (SOT) and hematopoietic stem cell transplantation (HSCT) due to a combination of direct cytopathic effects of replication and indirect, host-dependent immunopathological mechanisms. Besides herpesviruses like human cytomegalovirus (HCMV; see also chapter 6), herpes simplex virus (HSV; see also chapter 5), varicella zoster virus (VZV), and Epstein Barr virus (EBV), human polyomaviruses (e.g. BK-Virus), and in particular HAdV infections have increasingly been recognized as significant causes of transplant-related morbidity and mortality in the context of HSCT [13, 14]. Clinical manifestations following a HAdV infection in HSCT recipients include respiratory tract disease, gastroenteritis, hepatitis, cystitis, and multi-organ disseminated disease which can ultimately cause lethal organ damage.

In the setting of pediatric allogeneic HSCT, an overall incidence of HAdV infection from 6 to 42% has been observed, with a mortality rate of up to 80% for disseminated disease. The estimated incidence range in adult allogeneic HSCT is significantly lower (3–15%) but the clinical manifestations can be equally severe [4, 15–18]. Why HAdV is so much more problematic in pediatric HSCT patients than in adults is not fully understood but may be attributable to the permanent circulation of the virus among children, the higher persistence rate of the virus in early childhood, and/or the site of virus reactivation (see below) [15, 16].

The vast majority of HAdV serotypes associated with infection and disease in transplant recipients belong to subgroup C (e.g. C1, C2, C5), though certain serotypes of subgroup A and B have also been observed in these patients (Table 11.1). Coinfections with different



Figure 11.2 Adenovirus replication cycle. (a) Attachment, entry, uncoating: infection starts with binding of fiber to its specific cellular receptor and binding of penton base to $\alpha\beta$ -integrins followed by endocytosis. The virion is released into the cytoplasm via acidification of endosomes and travels to the nuclear pore while undergoing a stepwise uncoating process. After final dismantling, viral DNA is delivered into the nucleus. (b) Gene expression and genome replication: the HAdV replication cycle is divided in two stages, early and late phase. During the early phase, mainly nonstructural regulatory proteins are expressed which function to (i) activate additional virus genes, (ii) modulate expression of host genes necessary for DNA synthesis, (iii) avoid premature death of infected cells, and (iv) initiate virus genome replication. DNA replication is followed by the transcription of the late virus genes encoding mainly structural proteins and proteins necessary for particle maturation. (c) Assembly, maturation, release: newly synthesized structural components are imported into the nucleus and assemble into new, immature viral particles. This process is accompanied by packaging of progeny viral DNA. After particles have undergone a maturation process, infectious virus is released from the cell as a result of virally induced cell lysis. Proteins studied as potential antiviral targets in the different steps of the HAdV replication cycle are indicated (for details, see text). Source: Modified from [7].

serotypes are common and have been frequently reported [4, 15, 19]. It should be noted that all disease manifestations listed in Table 11.1 are seen in transplant recipients. However, transplant recipients are at risk to develop a disseminated infection originating from one of these local infections.

Although nosocomial outbreaks of adenovirus infections in transplant centers have been reported [20–22], it is believed that donor stem cell product-associated virus transmission and/or reactivation of latent virus are the major sources of early infection in HSCT patients. This hypothesis is underpinned by the following facts and observations: (i) following a primary infection in early childhood, HAdV can persist in a latent state in a variety of tissues of the infected host (tonsils, adenoids, lung, and intestines) from which the virus can reactivate upon severe immunosuppression (ii) a HAdV infection/disease in heavily immunosuppressed HSCT patients stereotypically tends to occur within the first few months after transplantation while patients are still subject to strict protective isolation, and (iii) HAdV strains detected in patients after HSCT are usually identical to those identified prior to transplantation [4, 23].

Interestingly, although virus reactivation could generally occur at different sites and tissues, a clinically significant HAdV replication preceding an invasive infection or disseminated disease in pediatric HSCT recipients almost always starts in the GI tract. It goes without saying that this recent finding has major implications for diagnosis, monitoring, and therapy of a HAdV infection at least in the pediatric allogeneic HSCT setting. Unfortunately, there are insufficient data available on adult HSCT to date [4, 15, 24].

11.2.2 Current Therapy Strategies and Antiviral Agents

HAdV infections in immunocompromised individuals arising from reactivation of latent viruses are difficult to treat and sometimes overwhelm the patient and result in death. In principle, there are three strategies to limit the impact of HAdV in patients undergoing HSCT: prophylaxis, pre-emptive therapy, and therapeutic treatment. In prophylaxis, the drug is given to all patients from the time of transplantation onward, irrespective of the risk for HAdV infection or reactivation. Thus, prophylaxis follows the rationale of preventing virus replication during the period of greatest immunosuppression after HSCT and prior to immune reconstitution. In pre-emptive therapy, patients are monitored by means of laboratory tests for active infection and are treated once a certain viral threshold has been detected. The goal of this intervention strategy is to prevent the progression of an active but asymptomatic infection toward disseminated disease. The treatment approach finally provides antiviral therapy only to symptomatic patients with overt adenovirus disease.

Unfortunately, recent data suggest that all drugs that are currently available for pharmacologic intervention against HAdV infections (i) have no or only limited efficacy when started as treatments for active, symptomatic HAdV disease and (ii) suffer from inadequate overall efficacy and/or are associated with toxicities when applied in a prophylactic regimen. As a consequence, major guidelines do not recommend either strategy with the currently available antivirals, thus leaving clinicians with only pre-emptive therapy as the mainstay for prevention of HAdV-associated morbidity and mortality [4, 15, 23, 25]. However, once better-tolerated and highly potent drugs become available, treatment

guidelines will most likely switch to universal prophylaxis, since this strategy is less dependent on the availability and cost issues of diagnostic procedures required for pre-emptive therapy and might even be associated with health and survival benefits for the transplant patient similar to those recently observed from prophylactic therapy of HCMV infections in HSCT with letermovir (see chapter 6, [26–28].

In recent years, *in vitro* activity against HAdV was reported for many different drugs and molecules; however, as outlined in Section 11.1, there is still no specifically approved antiviral therapy available for HAdV infections in the transplant setting [13, 15, 18]. Among those compounds with substantial antiviral activity in cell culture, case reports and case series describing clinical use in the immunocompromised host are available for only three drugs: ribavirin, (Val)ganciclovir, and cidofovir (Figure 11.1a) [4, 15, 23, 24].

11.2.2.1 Ribavirin

Ribavirin (Figure 11.1a) is a purine nucleoside analogue with broad-spectrum in vitro activity against many RNA and DNA viruses. Accordingly, the drug was approved for the treatment of chronic hepatitis C virus in combination with peginterferon alpha (CopegusTM; Rebetol[™]) or as a treatment for respiratory syncytial virus in infants and young children (Virazole[™]). Although different mechanisms of action have been proposed for this drug, including inhibition of viral polymerases, inhibition of viral RNA capping, or interference with the GTP synthesis pathway, the exact mechanism explaining its anti-HAdV activity remains unknown [29]. In vitro, anti-HAdV activity of ribavirin appears to be serotypedependent with highest antiviral activity seen against subgroup C viruses [30, 31]. However, depending on the assay used, the cells and the readouts as well as on the virus genotypes, EC_{50} values vary widely from 0.5 to 34 μ M. Clinical data for ribavirin are sparse and inconsistent, with some case reports suggesting a therapeutic benefit for some patients while others do not see an antiviral response on ribavirin treatment [32–34]. Consequently, ribavirin is not generally recommended for treatment of HAdV infections but can be considered in cases with serotype C infections, especially in patients with decreased renal function (see cidofovir side effects).

11.2.2.2 (Val)Ganciclovir

Another more specific antiviral drug that was assessed for efficacy against HAdV is the approved anti-CMV drug ganciclovir (GCV; Cymeven[®], Roche) and its oral prodrug valganciclovir (VGCV; Valcyte[®]; Roche) (Figure 11.1a) [35]. GCV/VGCV is a deoxyguanosine analogue that requires specific intracellular phosphorylation for full activation. In a first step, GCV is selectively monophosphorylated in infected cells by virus-encoded protein kinases like the HSV thymidine kinase or the HCMV protein kinase pUL97. Subsequently, GCV-P is converted to a triphosphate by cellular enzymes. The active form of GCV preferentially inhibits viral DNA polymerases and is also incorporated into progeny viral DNA which drastically slows down chain elongation [29]. However, there is also off-target activity in the form of cellular polymerase inhibition. Neutropenia, anemia, thrombocytopenia, and a putative long-term reproductive toxicity are the most common serious side effects associated with GCV treatment [36]. Importantly, adenoviruses, in contrast to herpesviruses, lack a viral kinase gene resulting in inefficient activation of GCV in HAdV-infected cells. Accordingly, GCV has demonstrated only a modest anti-HAdV efficacy *in vitro* (EC50 values in the upper two digit μ M range) and the rare reports on systemic (V)GCV administration in the clinics do not support its use as an anti-HAdV drug in the HSCT setting [35, 37].

11.2.2.3 Cidofovir

Most evidence for the *in vivo* efficacy of antiviral HAdV therapy in the pre-emptive setting is available for cidofovir (CDV; Vistide®, Gilead, Figure 11.1a), a drug that was initially approved for the treatment of HCMV retinitis [29]. CDV is a phosphonomethoxy analogue of cytosine that has demonstrated broad-spectrum activity against double-stranded DNA viruses including herpesviruses and adenoviruses. In contrast to GCV, CDV does not require initial modification by a viral enzyme, since its conversion to a triphosphate is catalyzed by cellular enzymes. The active form of CDV acts as a competitive inhibitor of the viral DNA polymerase and causes premature chain termination during viral DNA synthesis [36, 38]. CDV is sufficiently active against all HAdV subtypes (EC50 values ranging from 0.5 to 62μ M) and is characterized by a long intracellular half-life allowing infrequent administration of the drug [13]. Given, that a pre-emptive CDV therapy of HAdV-infected HSCT patients is associated with clinical improvement and a survival benefit, this drug is the current standard of care (SOC) for controlling HAdV infections and preventing disseminated HAdV disease in immunocompromised patients undergoing an allogeneic stem cell transplantation [15, 23, 25, 39]. Although encouraging outcomes are seen with CDV therapy, the widespread clinical use of CDV is limited since the drug is slowly absorbed, poorly bioavailable, myelosuppressive, and importantly, causes severe nephrotoxicity [18, 29, 40].

In summary, current antiviral therapy of HAdV infections in HSCT is suboptimal and available drugs suffer from low efficacy and/or problems with associated toxicities which are particularly detrimental after HSCT. Accordingly, there is a high unmet medical need for new, safe, and effective anti-HAdV agents that can be used in pre-emptive but hopefully also prophylactic strategies in HSCT.

11.2.3 Novel Antiviral Approaches in Clinical Development

11.2.3.1 Brincidofovir

As outlined under Section 11.2.2.3, the current standard drug for therapy of HAdV-related disease in HSCT is the broad-spectrum antiviral CDV, but its use is limited due to poor oral bioavailability and inherent nephrotoxicity. Attempts to overcome these drawbacks led to the synthesis of the CDV derivative brincidofovir (BCV; CMX001; hexadecyloxypropyl-CDV; Cidofovir-HDP) (Figure 11.1b; Table 11.2). BCV is a lipid ester prodrug of CDV which is highly efficiently delivered into target cells. Within cells, the lipid side chain of BCV is cleaved off by phospholipases and CDV is released (see Section 11.2.2.3) [41, 42]. Initial studies evaluating the antiviral activity of BCV have demonstrated an increased *in vitro* potency relative to CDV against a wide range of double-stranded DNA viruses including cytomegaloviruses (~400× higher efficacy) and adenoviruses (~65× higher efficacy) presumably due to the higher intracellular concentrations of the active drug CDV [43]. Subsequent preclinical and early clinical studies indicated that the lipid conjugation of CDV also results (i) in improved oral bioavailability of the drug and (ii) in a favorable

Product	Sponsor	Phase	Indication	Delivery	Status
CMX001 (Brincidofovir)	Chimerix	Phase 2/3 (NCT03749317 ^a , NCT02087306)	Adenovirus infections in pediatric HSCT recipients	Oral delivery (tablet and pediatric liquid suspension)	Terminated due to low enrollment rate
SyB V-1901 (Brincidofovir)	SymBio	Phase 2a (NCT04706923)	Adult and pediatric subjects with adenovirus viremia	Intravenous infusion	Not yet recruiting
OKG-0301 (Ranpirnase/ Onconase)	Okogen	Phase 2 (NCT03856645)	Acute adenoviral conjunctivitis	Ophthalmic drops	Recruiting
IVIEW-1201 (Povidone iodine 1%)	IVIEW Therapeutics	Phase 2 (NCT03749317)	Acute adenoviral conjunctivitis	Ophthalmic sustained release gel	Recruiting

Table 11.2	HAdV antiviral	agents	in clinical	develo	pment

a) ClinicalTrials.gov identifier.

safety profile with little to no evidence of associated nephrotoxicity [18, 44]. These findings paved the way for BCV to move into full clinical development. Initially, *Chimerix Inc.* (Durham, North Carolina, USA) sought to develop BCV as an oral therapeutic for the prophylaxis of CMV infections in immunocompromised HSCT recipients and for the potential treatment of smallpox infections in case of a bioterrorism attack. However, development in the CMV indication was discontinued based on negative results of the phase 3 SUPPRESS trial [45].

Clinical development for HAdV was started after a series of case studies reported favorable clinical outcomes following BCV administration in immunocompromised patients with disseminated HAdV infection [43, 46, 47]. Several clinical trials investigating the efficacy of BCV in patients with HAdV were initiated by Chimerix Inc., including a phase 3 open-label study in HSCT (AdVise Trial; NCT02087306). Although rapid declines in HAdV-DNA levels were seen in the majority of treated subjects, no meaningful difference in overall survival was observed between BCV-treated patients and historic controls [48]. One reason for this could have been that ultimately the identification of a valid comparator cohort for use as historic controls had not been possible [48]. Based on this outcome and due to the observation that diarrhea was the main dose-limiting adverse effect associated with prolonged BCV administration, a phase 2 study was initiated to assess safety, overall tolerability, and antiviral activity of a "short-course" BCV therapy for the treatment of HAdV infections in pediatric HSCT recipients (AdAPT Trial, NCT03339401). Since previous studies have shown that clearance of HAdV from the blood of patients has a positive impact on overall survival after HSCT, the primary endpoint chosen for this study was a comparison of the viral burden (average HAdV DNA levels) in blood over 16 weeks in subjects treated with short-course oral BCV versus those who receive local SOC. However, in the second half of 2019, all active trials evaluating the use of BCV in humans were halted due to low patient recruitment (Table 11.2). At the same time, *Chimerix Inc.* announced the out-licensing of BCV to *SymBio Pharmaceuticals Limited* for all human indications excluding smallpox (http://www.chimerix.com). Given this, the future of BCV as novel HAdV therapy is currently unclear though a new phase 2a dose ranging study study in adult and pediatric subjects with HAdV viremia was recently innitiated by SymBio (NCT04706923; Table 11.2). [18, 24, 27].

11.3 Ocular Adenovirus Infections

11.3.1 Incidence, Transmission, and Clinical Manifestation

Acute infectious conjunctivitis is an extremely common condition that, in principle, can be caused by either bacterial or viral pathogens. However, it is estimated that more than 50% of all infectious conjunctivitis cases have a viral etiology and that up to 90% of these cases are caused by HAdV, making this virus the single-most common cause of ocular infections worldwide. Accordingly, HAdV presents a serious public health risk with around 25 million HAdV-associated conjunctivitis cases per year worldwide including ~6 million cases in the United States [8, 49, 50]. Virus outbreaks occur globally on a regular basis but are more frequently seen in densely populated countries or regions like Southeast Asia. Accordingly, Japan alone has an incidence of more than one million cases of ocular HAdV infections per year [51, 52].

Ocular HAdV infections affect all ages and socioeconomic classes with no specific gender affinity [53]. Virus transmission occurs primarily through respiratory droplets or direct contact with ocular secretions, e.g. via finger-to-eye infection. Importantly, patients start to shed infectious virus several days prior to the onset of symptoms and remain infectious during the entire course of the acute phase of the disease (~two to three weeks) thus giving the virus time to spread among people in close proximity, e.g. family members, coworkers, etc. Due to the highly contagious nature of HAdV in connection with its extreme extracellular stability on common fomites like towels, doorknobs, soap, eyeglasses, etc., epidemic HAdV outbreaks are frequently seen in closed settings with high population densities such as schools, day care institutions, hospitals, and the military and—unfortunately—are also spread in ophthalmologic units via patients, healthcare workers, or contaminated medical instruments [9].

Manifestations of ocular HAdV infections can be categorized into three major clinical syndromes: follicular conjunctivitis (FC), pharyngeal conjunctival fever (PCF), and EKC. The first two conditions (FC and PCF) are relatively mild, self-limiting infections of the conjunctiva that last for 3–7 days and importantly do not result in long-term sequelae. FC symptoms include bulbar conjunctival injection, chemosis, and eyelid edema. PCF appears similar, though in addition to the ocular manifestations, PCF is associated with cold-like symptoms including fever, sore throat, and rhinitis. The most common HAdV serotypes causing FC and PCF are members of the subgroups B (e.g. B3, B7, B11, B14) and E (e.g. E4) (Table 11.1) [3, 8, 54].

EKC, however, is a more frequent and far more serious condition that, as indicated by its name, involves both the conjunctiva and the cornea. EKC is caused largely by a subset

of subgroup D viruses namely D8, 37, 53, 54, 56, and 64 (previously described as HAdV-D19a) (Table 11.1) [3, 53]. A correlation between receptor usage and cell tropism of EKCcausing HAdVs has been suggested, given that almost all EKC-causing HAdVs are capable of using sialic acid glycans on the surface of corneal epithelial cells for primary attachment, preceding (co-)receptor-mediated entry into the cornea [9, 10, 55]. EKC typically starts unilaterally, but due to the contagious nature of the virus, both of the patient's eves usually become involved in the course of the disease. Clinical signs of EKC appear after an incubation period of 5-12 days and patients present with red eye, ocular irritation, foreign body sensation, photophobia, watery discharge, FC, edema, and pain. Corneal involvement, which is a hallmark of EKC, typically starts three to four days after the onset of symptoms with diffuse epithelial keratitis accompanied by preauricular lymphadenopathy followed by the formation of subepithelial corneal infiltrates (SEIs) or "nummuli" [3, 8, 53]. SEIs are punctate multifocal areas of epithelial opacity that form in the corneal stroma due to the infiltration of various immune cells. This leukocyte infiltration is a response to cytokines secreted from productively HAdV-infected keratocytes located in the corneal stroma [8–10, 55]. Approximately 30–50% of patients with EKC will develop stromal infiltrates that cause photophobia and disturbances in vision. Although the acute phase of EKC is self-limiting and usually lasts up to three weeks, corneal opacities due to SEIs may persist for months to years after the initial infection has resolved. Moreover, a chronic subepithelial infiltration of the corneal stroma by leucocytes can damage the superficial stroma of the eye and may lead to corneal scarring causing permanent vision loss and photophobia. EKC may be further complicated by persistent dry eye syndrome requiring long-term treatment [8, 9, 54, 56].

11.3.2 Current Therapy Strategies and Antiviral Agents

Although EKC (without knowing the underlying pathogen) was initially clinically described in Austria more than 100 years ago, to date there is still no specific approved antiviral therapy to alleviate the clinical symptoms, shorten the course of infection, prevent the formation of corneal opacities, or to block virus spread [8, 57]. Given this unmet medical need, several drugs were evaluated in the past for their potential clinical efficacy as topical treatments for EKC, albeit with moderate success.

11.3.2.1 Corticosteroids

The first example is topical corticosteroids that were prescribed frequently during the acute phase of the infection to reduce inflammation-related discomfort and pain and to prevent the formation of SEIs. Although steroid treatment can provide transient symptomatic relief, its use remains controversial since steroid treatment (i) could enhance HAdV replication, (ii) prolongs HAdV shedding and disease, and (iii) may cause a rebound increase in SEIs upon cessation of drug [8, 53, 58]. These findings, in addition to the known steroid-associated side effects (e.g. glaucoma, cataract formation) restrict the use of steroid treatment to complicated cases with severe keratitis, photophobia, and SEIs, significantly impairing visual acuity. Nonsteroidal anti-inflammatory agents do not increase HAdV replication but have been shown to be ineffective against SEIs [54].

11.3.2.2 Povidone-iodine and SHP640

Concerns about prolonged viral shedding upon steroid usage in EKC patients lead to the idea of combining a potent topical corticosteroid with a suitable antiviral agent in order to treat both the inflammatory and the infectious components of EKC [59]. SHP649 (FST-100, TAK-640; Shire, Foresight Biotherpeutics, Takeda) is a combination of 0.1% dexamethasone with 0.6% povidone-iodine (PVP-I; Figure 11.1c). PVP-I is a powerful disinfectant routinely used in ocular and general surgery. It is highly effective against viruses, bacteria, fungi, and other parasites, immune to the development of bacterial/viral resistance and does not cause irritation on skin [8, 60]. Several concentrations of PVP-I were tested as monotherapy for potential clinical use in EKC, albeit with inconclusive outcomes [60, 61]. A combination of PVP-I and dexamethasone, however, should have the potential to treat the viral component of EKC as well as immune-related sequelae such as SEIs. Moreover, since PVP-I basically kills bacteria and viruses, this medication could potentially be used to treat patients suffering from any kind of infectious conjunctivitis without the need for identification of the underlying pathogen [59]. However, concerns regarding the potential efficacy of PVP-I in EKC have also been raised given that PVP-I (i) does not accumulate in ocular tissue and thus has a very short residence time on the ocular surface, (ii) is not active against intracellular adenoviral particles, and (iii) has only a reduced virucidal activity against EKC-causing HAdVs of subgroup D [60-63]. Following the acquisition of *Foresight Biotherapeutics*, *Shire* (a wholly owned subsidiary of Takeda) was developing SHP640 for the potential treatment of adenoviral and bacterial conjunctivitis. Results of a phase 2 proof-of-concept trial were recently published comparing SHP640 against PVP-I and vehicle in patients with HAdV conjunctivitis [64]. Patients treated with SHP640 showed significantly better outcomes in terms of clinical resolution and virus eradication than vehicle, but only a trend toward clinical significance for SHP640 versus PVP-I alone. Moreover, SHP640 treatment had no effect on SEI formation or on virus spread from the primary infected eye to the fellow eye [64]. Based on these initial efficacy data, SHP640 advanced to phase 3 clinical trials. However, in May 2019, two ongoing phase 3 studies in adenoviral conjunctivitis (NCT02998541, NCT02998554) were terminated by the sponsor, suggesting that the program has been discontinued by Takeda.

11.3.2.3 Nucleoside Analogues

Additional treatment avenues have been pursued beyond these, but none have led to approval of a drug for ocular HAdV infections [8, 53, 60]. Among others, these approaches included ophthalmologic formulations of the classic antiviral nucleoside analogues GCV (Zirgan, *Bausch & Lomb*) and CDV (*Gilead, Bausch & Lomb*) (for details about the drugs, see Section 11.2.2; Figure 11.1a). GCV demonstrated only limited efficacy against ocular HAdV serotypes *in vitro* (EC50 values in the upper two digit μ M range) and produced variable results when tested as an ophthalmic formulation in ocular adenoviral animal models or in small clinical trials [8, 60, 65]. Data from large-scale controlled clinical trials are not available but the controversial efficacy results of GCV for treating HAdV conjunctivitis might be explained by insufficient GCV activation in certain cell types due to a lack of a viral kinase in HAdV (see Section 11.2.2.2). In contrast, topical CDV has demonstrated significant antiviral activity both *in vitro* and in animal models and has

successfully completed a small randomized controlled phase 3 trial [66]. In this trial, the administration of 1% CDV eye drops significantly reduced the frequency of severe corneal opacities. Unfortunately, administration of 1% CDV was associated with substantial local toxicities and a lower, less toxic dose (0.2%) was clinically ineffective [66, 67]. Consequently, the clinical development of topical CDV for ocular HAdV infections was discontinued [53, 65].

11.3.2.4 Palliative Treatment

Since there is little evidence for a beneficial effect on ocular HAdV infections from any of the agents discussed above (see Sections 11.3.2.1–11.3.2.3), current treatment strategies still rely entirely upon supportive therapy directed toward limiting the severity of symptoms such as cold compresses and artificial tears. However, palliative approaches do not (i) prevent the spread of disease, (ii) interfere with virus replication/shedding, or (iii) reduce the risk of long-term sequelae or vision loss, nor the spread in the community. Accordingly, there remains an unmet need for a specific, safe, and effective therapy or prophylaxis that is well tolerated as an antiviral for the treatment of EKC.

11.3.3 Novel Antiviral Approaches in Clinical Development

In spite of the increasing need for a novel HAdV antiviral, only two agents are currently in active clinical development for ocular HAdV infections. Interestingly, neither agent is a classical direct acting antiviral (DAA); one drug is an ophthalmic formulation of a repurposed biological anticancer compound, and the other agent is a novel formulation of the broad range antiseptic PVP-I (see Section 11.3.2.2).

11.3.3.1 OKG-0301

OKG-0301 (Ranpirnase, Onconase; *Okogen* under the license of *Tamir Biotechnology*, Table 11.2) is an ophthalmic formulation of the amphibian endoribonuclease ranpirnase that previously advanced under the name Onconase as an intravenous formulation to phase 3 clinical trials in oncology [68]. *Okogen* licensed ranpirnase for ocular use from *Tamir Biotechnology* after preclinical studies found that ranpirnase demonstrates a broad-spectrum antiviral activity in cell culture, including HAdV-infected cells. OKG-0301 is thought to preferentially enter virally infected cells and to inhibit viral replication via degradation of tRNAs resulting in the inhibition of protein synthesis [69]. Further studies showed that OKG-0301 significantly reduced viral titers in adenovirus infected eyes in addition to shortening the duration of viral shedding in a clinically predictive rabbit ocular model [70]. In January 2019, a multisite, randomized, placebo-controlled, double-blinded phase 2 proof-of-concept trial (RUBY) was initiated in patients with acute adenoviral conjunctivitis in Australia. The goal of the study is to evaluate the safety and efficacy of two different doses of OKG-0301 versus placebo (NCT03856645). By February 2021 the study is still ongoing and recruiting patients.

11.3.3.2 iVIEW-1201

The second agent currently in clinical development for the potential treatment of adenoviral conjunctivitis is iVIEW-1201 (*iVIEW Therapeutics*), a long-acting, extended-release, gel

formulation of PVP-I that retains the broad *in vitro* antimicrobial and antiviral properties of "classical" PVP-I preparations (http://www.iviewtherapeutics.com). Given its long-acting sustained-release formulation, iVIEW-1201 addresses one major difficulty in topical treatment of ocular diseases namely, provision and maintenance of an optimal ocular drug concentration over an extended period of time. By prolonging the corneal retention time of PVP-I, it is hoped that the ocular bioavailability of iVIEW-1201 is improved leading to an enhanced virucidal activity on the surface of the infected eye [71]. However, despite this apparent improvement, it remains to be seen whether topically administrated iVIEW-1201 will significantly reduce titers of EKC-causing HAdV serotypes and, even more importantly, whether it will limit the development of corneal infiltrates given that the drug does not permeate into infected corneal cells. iVIEW obtained an approval for clinical phase 2 trials in patients with HAdV conjunctivitis in China, India, and the United States (http://www.iviewtherapeutics.com). With a first patient treated December 2019 in India, this trial should address the question of efficacy of iVIEW-1201 in the intended opthalmologic use (NCT03749317). By February 2021 no interim results were reported and the trial continues to recruite patients.

11.4 Drug Targets for Direct Acting Antivirals

As outlined above (Sections 11.2 and 11.3), current treatment options for combating HAdV infections are either ineffective and/or cause severe toxicities. Despite the clear medical need, HAdV drug discovery has received limited attention in recent years and only very few candidate compounds are currently in clinical development for HAdV-associated diseases. To overcome this problem, additional research is needed to discover new antiviral drugs that are potent, safe, and well tolerated and ideally possess a new mode of action. In theory, HAdV is a good target for direct acting antivirals as its DNA genome encodes a series of proteins that (i) are highly conserved among disease-related serotypes, (ii) are dissimilar to- or even lack a human analogue, and most importantly (iii) fulfill essential functions at multiple stages in the virus life cycle [7]. However, in reality, the number of viral proteins that have been exploited as potential anti-HAdV targets in the past is very low. Even so, those proteins already studied extensively as potential anti-HAdV targets cover at least three different steps of the HAdV replication cycle: (i) virus attachment/entry, (ii) viral genome replication, and (iii) particle maturation (Figure 11.2).

11.4.1 Virus Attachment and Entry

Blocking the attachment of viral pathogens to its specific cellular receptors is, in principle, an attractive approach for antiviral drug discovery since potential drugs interfere with the very first step in the viral replication cycle and thus prevent cell penetration and the subsequent expression of potentially toxic viral genes. Several agents inhibiting HAdV attachment and/ or entry *in vitro* have been identified ([7], Figure 11.2a) and at least one, APD-209 (*Adenovir Pharma*), has been evaluated as a topical drug in clinical trials for EKC. APD-209 is based on small soluble multivalent sialic acids (SAs), that prevent EKC causing HAdV serotypes from binding to its receptor on ocular cells: SA-containing glycans [10, 72]. Based on favorable preclinical safety and efficacy data, a phase 2 proof-of-concept trial was initiated in EKC

patients (NCT0197743). Unfortunately, ADP-209 failed to demonstrate statistical significance in this study leading to discontinuation of the program. Currently, it is unclear whether the negative outcome of this trial is due to the specific properties of ADP-209 (e.g. insufficient efficacy, strain coverage, drug stability, frequency of dosing, etc.) or whether the overall concept of combating HAdV eye infections with molecules that act solely on the surface of an infected eye is inadequate. Potential limitations of topically applied entry inhibitors are (i) no penetration of the drug into infected cells, (ii) no activity against intracellular virus, (iii) no accumulation in infected tissue, and (iv) rapid clearing from the ocular surface, in particular by lacrimal flow and eye blinking. Since these limitations also apply to topically administered antiseptic antivirals such as iVIEW-1201 (see Section 11.3.3.2), the outcome of the ongoing iVIEW-1201 trial will shed more light on this conceptual question.

11.4.2 Virus Genome Replication

Not least due to the success of the anti-herpesviral drugs acyclovir, ganciclovir, and cidofovir (see Section 11.2.2), targeting the viral DNA-polymerase with e.g. classical nucleoside analogues was a popular antiviral approach to interfere with viral genome replication (Figure 11.2b). HAdVs encode a DNA-polymerase that is structurally divergent from host polymerases but utilizes the same intracellular pool of nucleotide substrates. This is important since in contrast to herpesviruses, HAdVs lack a viral "nucleotide kinase" and thus, the principle of activating nucleoside drugs only in infected cells by phosphorylation cannot be applied to HAdV. Accordingly, the main challenge for the design of novel nucleoside inhibitors is to identify molecules that preferentially inhibit the viral polymerase over the host's polymerases since this is a prerequisite for preventing off-target effects and to ensure that a triphosphate is generated with high efficacy by the cell or by offering a nucleotide. An alternative approach would be the design of drugs addressing unique sites on the HAdV polymerase, distinct from those targeted by the nucleoside analogues. Successful antiviral agents following the latter principle are exemplified by the anti-HIV non-nucleoside reverse transcriptase inhibitors (NNRTIs) [73]. Today, many publications have reported on various molecules targeting the HAdV DNA-polymerase, including nucleoside and nucleotide analogues [35, 65, 74]. However, with the exception of the drugs discussed earlier (GCV, CDV, and BCV), all remain to be evaluated in clinical trials. Interestingly, the majority of these molecules originated from drug repurposing approaches performed by academic research groups and not from industry-driven drug discovery programs specifically designed to identify HAdV-polymerase inhibitors which again demonstrates the recent lack of apparent interest in anti-HAdV drug discovery. Given that a natural consequence of drug repurposing is an intrinsic lack of target specificity, the chance that molecules identified by these approaches will be advanced to clinical development is low.

11.4.3 Particle Maturation

A key enzyme in HAdV maturation and in the generation of infectious virus particles is the virus-encoded cysteine protease AVP (adenain). AVP plays an essential role in several steps of the virus life cycle including uncoating of the incoming virus particle and proteolytic maturation of newly assembled, noninfectious particles giving rise to mature, infectious

progeny virus (Figure 11.2c) [75, 76]. Viral proteases are proven targets of efficacious antiviral therapies, including human immunodeficiency virus (HIV) and hepatitis C virus (HCV) [77]. Accordingly, specific inhibition of the essential adenovirus protease AVP may also represent an effective treatment strategy for HAdV infections [78]. Proteases constitute a large family of enzymes, frequently with binding pockets which are difficult to target, e.g. because of their topology or lipophilic nature. Given that many naturally occurring protease inhibitors are proteins, a recent two-pronged Hit-discovery campaign identified a tetrapeptide nitrile and a pyrimidine nitrile as two independent starting points for structure-guided medical chemistry [79]. A subsequent Hit-to-Lead approach resulted in potent AVP inhibitors with *in vitro* IC₅₀ values in the picomolar range. Unfortunately, despite high solubility and good cell permeability, the exceptional anti-AVP potency of these inhibitors could not be translated into adequate antiviral activity in cell culture infection experiments and the program was discontinued [80]. In conclusion, despite the identification of highly potent and specific inhibitors of AVP, the final validation of AVP as a valid anti-HAdV drug target thus remains elusive.

The approaches discussed in this review describe the key antiviral strategies currently being pursued. However, there are a multitude of additional highly specific viral processes presenting opportunities for the development of novel antiviral targets and strategies (extensively reviewed in [7]), but these will require more intensive and concerted effort if they are to yield effective novel treatments. Ideally, a multipronged approach addressing several targets should be pursued to allow suppression of the virus at different steps in the replication cycle. This would offer the added benefit of counteracting the potential emergence of drug-resistant virus strains, an issue of particular importance given the lack of alternative treatment options.

11.5 Conclusion

HAdV infections remain a significant cause of transplant-related morbidity and mortality in the context of pediatric HSCT and ocular HAdV infections represent a serious public health burden due to their rapid spread and the potential severity of ocular symptoms. To date, there is no systemic or topical treatment approved in the United States or the EU. Current antiviral therapies of HAdV infections are inadequate and rely solely on suboptimal "off-label" therapeutic options or palliative care. Despite this high unmet medical need, only very few candidate compounds are currently in clinical development for HAdVassociated diseases. Accordingly, additional research is urgently needed to find specific, safe, and effective antiviral agents for systemic use in immunocompromised individuals and/or for the topical or systemic treatment of adenovirus ocular infections.

References

1 Rowe, W.P., Huebner, R.J., Gilmore, L.K. et al. (1953). Isolation of a cytopathogenic agent from human adenoids undergoing spontaneous degeneration in tissue culture. *Proc. Soc. Exp. Biol. Med.* 84: 570–573.

- 2 Dhingra, A., Hage, E., Ganzenmueller, T. et al. (2019). Molecular evolution of human adenovirus (HAdV) species C. *Sci. Rep.* 9: 1039.
- **3** Ghebremedhin, B. (2014). Human adenovirus: viral pathogen with increasing importance. *Eur J Microbiol Immunol (Bp)* 4: 26–33.
- **4** Lion, T. (2014). Adenovirus infections in immunocompetent and immunocompromised patients. *Clin. Microbiol. Rev.* 27: 441–462.
- **5** Waye, M. and Sing, C. (2010). Anti-viral drugs for human adenoviruses. *Pharmaceuticals* 3 (10): 3343–3354.
- **6** Russell, W.C. (2009). Adenoviruses: update on structure and function. *J. Gen. Virol.* 90: 1–20.
- 7 Martinez-Aguado, P., Serna-Gallego, A., Marrugal-Lorenzo, J.A. et al. (2015).
 Antiadenovirus drug discovery: potential targets and evaluation methodologies. *Drug Discov. Today* 20: 1235–1242.
- **8** Chigbu, D.I. and Labib, B.A. (2018). Pathogenesis and management of adenoviral keratoconjunctivitis. *Infect. Drug Resist.* 11: 981–993.
- **9** Jonas, R.A., Ung, L., Rajaiya, J., and Chodosh, J. (2019). Mystery eye: human adenovirus and the enigma of epidemic keratoconjunctivitis. *Prog. Retin. Eye Res.* https://doi.org/10.1016/j.preteyeres.2019.100826:100826.
- **10** Chandra, N., Frangsmyr, L., and Arnberg, N. (2019). Decoy receptor interactions as novel drug targets against EKC-causing human adenovirus. *Viruses* 11 (3): 242.
- **11** Arnberg, N. (2012). Adenovirus receptors: implications for targeting of viral vectors. *Trends Pharmacol. Sci.* 33: 442–448.
- 12 Radin, J.M., Hawksworth, A.W., Blair, P.J. et al. (2014). Dramatic decline of respiratory illness among US military recruits after the renewed use of adenovirus vaccines. *Clin. Infect. Dis.* 59: 962–968.
- 13 Chemaly, R.F., Hill, J.A., Voigt, S., and Peggs, K.S. (2019). In vitro comparison of currently available and investigational antiviral agents against pathogenic human double-stranded DNA viruses: a systematic literature review. *Antivir. Res.* 163: 50–58.
- **14** Hill, J.A., Mayer, B.T., Xie, H. et al. (2017). The cumulative burden of double-stranded DNA virus detection after allogeneic HCT is associated with increased mortality. *Blood* 129: 2316–2325.
- 15 Hiwarkar, P., Kosulin, K., Cesaro, S. et al. (2018). Management of adenovirus infection in patients after haematopoietic stem cell transplantation: state-of-the-art and real-life current approach: a position statement on behalf of the Infectious Diseases Working Party of the European Society of Blood and Marrow Transplantation. *Rev. Med. Virol.* 28: e1980.
- 16 Sedlacek, P., Petterson, T., Robin, M. et al. (2019). Incidence of adenovirus infection in hematopoietic stem cell transplantation recipients: findings from the AdVance Study. *Biol. Blood Marrow Transplant.* 25: 810–818.
- **17** Schaar, K., Roger, C., Pozzuto, T. et al. (2016). Biological antivirals for treatment of adenovirus infections. *Antivir. Ther.* 21: 559–566.
- **18** Wold, W.S.M., Tollefson, A.E., Ying, B. et al. (2019). Drug development against human adenoviruses and its advancement by Syrian hamster models. *FEMS Microbiol. Rev.* 43: 380–388.
- 19 Gray, G.C., McCarthy, T., Lebeck, M.G. et al. (2007). Genotype prevalence and risk factors for severe clinical adenovirus infection, United States 2004–2006. *Clin. Infect. Dis.* 45: 1120–1131.

- **20** Yolken, R.H., Bishop, C.A., Townsend, T.R. et al. (1982). Infectious gastroenteritis in bone-marrow-transplant recipients. *N. Engl. J. Med.* 306: 1010–1012.
- **21** Mattner, F., Sykora, K.W., Meissner, B., and Heim, A. (2008). An adenovirus type F41 outbreak in a pediatric bone marrow transplant unit: analysis of clinical impact and preventive strategies. *Pediatr. Infect. Dis. J.* 27: 419–424.
- Leruez-Ville, M., Chardin-Ouachee, M., Neven, B. et al. (2006). Description of an adenovirus A31 outbreak in a paediatric haematology unit. *Bone Marrow Transplant*. 38: 23–28.
- **23** Lindemans, C.A., Leen, A.M., and Boelens, J.J. (2010). How I treat adenovirus in hematopoietic stem cell transplant recipients. *Blood* 116: 5476–5485.
- 24 Dailey Garnes, N.J.M., Ragoonanan, D., and Aboulhosn, A. (2019). Adenovirus infection and disease in recipients of hematopoietic cell transplantation. *Curr. Opin. Infect. Dis.* 32: 591–600.
- **25** Matthes-Martin, S., Feuchtinger, T., Shaw, P.J. et al. (2012). European guidelines for diagnosis and treatment of adenovirus infection in leukemia and stem cell transplantation: summary of ECIL-4 (2011). *Transpl. Infect. Dis.* 14: 555–563.
- **26** Chemaly, R.F., Ullmann, A.J., Stoelben, S. et al. (2014). Letermovir for cytomegalovirus prophylaxis in hematopoietic-cell transplantation. *N. Engl. J. Med.* 370: 1781–1789.
- 27 Lopez, S.M.C., Michaels, M.G., and Green, M. (2018). Adenovirus infection in pediatric transplant recipients: are effective antiviral agents coming our way? *Curr. Opin. Organ Transplant.* 23: 395–399.
- 28 Ljungman, P., Schmitt, M., Marty, F.M. et al. (2019). A mortality analysis of letermovir prophylaxis for cytomegalovirus (CMV) in CMV-seropositive recipients of allogeneic hematopoietic-cell transplantation. *Clin. Infect. Dis.* https://doi.org/10.1093/cid/ciz490.
- **29** De Clercq, E. and Li, G. (2016). Approved antiviral drugs over the past 50 years. *Clin. Microbiol. Rev.* 29: 695–747.
- **30** Morfin, F., Dupuis-Girod, S., Mundweiler, S. et al. (2005). In vitro susceptibility of adenovirus to antiviral drugs is species-dependent. *Antivir. Ther.* 10: 225–229.
- Morfin, F., Dupuis-Girod, S., Frobert, E. et al. (2009). Differential susceptibility of adenovirus clinical isolates to cidofovir and ribavirin is not related to species alone. *Antivir. Ther.* 14: 55–61.
- **32** Abe, S., Miyamura, K., Oba, T. et al. (2003). Oral ribavirin for severe adenovirus infection after allogeneic marrow transplantation. *Bone Marrow Transplant.* 32: 1107–1108.
- Homma, M., Inoue, Y., Hasegawa, Y. et al. (2008). Blood ribavirin concentration in high-dose ribavirin for adenovirus-induced haemorrhagic cystitis – a case report. *J. Clin. Pharm. Ther.* 33: 75–78.
- 34 Lankester, A.C., Heemskerk, B., Claas, E.C. et al. (2004). Effect of ribavirin on the plasma viral DNA load in patients with disseminating adenovirus infection. *Clin. Infect. Dis.* 38: 1521–1525.
- **35** Naesens, L., Lenaerts, L., Andrei, G. et al. (2005). Antiadenovirus activities of several classes of nucleoside and nucleotide analogues. *Antimicrob. Agents Chemother.* 49: 1010–1016.
- **36** De Clercq, E. and Holy, A. (2005). Acyclic nucleoside phosphonates: a key class of antiviral drugs. *Nat. Rev. Drug Discov.* 4: 928–940.
- **37** Chen, F.E., Liang, R.H., Lo, J.Y. et al. (1997). Treatment of adenovirus-associated haemorrhagic cystitis with ganciclovir. *Bone Marrow Transplant*. 20: 997–999.
- **38** Chamberlain, J.M., Sortino, K., Sethna, P. et al. (2019). Cidofovir diphosphate inhibits adenovirus 5 DNA polymerase via both nonobligate chain termination and direct

inhibition, and polymerase mutations confer cidofovir resistance on intact virus. *Antimicrob. Agents Chemother.* 63 (1): e01925-18.

- **39** Neofytos, D., Ojha, A., Mookerjee, B. et al. (2007). Treatment of adenovirus disease in stem cell transplant recipients with cidofovir. *Biol. Blood Marrow Transplant*. 13: 74–81.
- **40** Lischka, P. and Zimmermann, H. (2008). Antiviral strategies to combat cytomegalovirus infections in transplant recipients. *Curr. Opin. Pharmacol.* 8: 541–548.
- **41** Beadle, J.R., Hartline, C., Aldern, K.A. et al. (2002). Alkoxyalkyl esters of cidofovir and cyclic cidofovir exhibit multiple-log enhancement of antiviral activity against cytomegalovirus and herpesvirus replication in vitro. *Antimicrob. Agents Chemother.* 46: 2381–2386.
- **42** Hostetler, K.Y. (2009). Alkoxyalkyl prodrugs of acyclic nucleoside phosphonates enhance oral antiviral activity and reduce toxicity: current state of the art. *Antivir. Res.* 82: A84–A98.
- **43** Florescu, D.F. and Keck, M.A. (2014). Development of CMX001 (Brincidofovir) for the treatment of serious diseases or conditions caused by dsDNA viruses. *Expert Rev. Anti-Infect. Ther.* 12: 1171–1178.
- Tippin, T.K., Morrison, M.E., Brundage, T.M., and Mommeja-Marin, H. (2016).
 Brincidofovir is not a substrate for the human organic anion transporter 1: a mechanistic explanation for the lack of nephrotoxicity observed in clinical studies. *Ther. Drug Monit.* 38: 777–786.
- **45** Marty, F.M., Winston, D.J., Chemaly, R.F. et al. (2019). A randomized, double-blind, placebo-controlled phase 3 trial of oral brincidofovir for cytomegalovirus prophylaxis in allogeneic hematopoietic cell transplantation. *Biol. Blood Marrow Transplant.* 25: 369–381.
- **46** Paolino, K., Sande, J., Perez, E. et al. (2011). Eradication of disseminated adenovirus infection in a pediatric hematopoietic stem cell transplantation recipient using the novel antiviral agent CMX001. *J. Clin. Virol.* 50: 167–170.
- **47** Florescu, D.F., Pergam, S.A., Neely, M.N. et al. (2012). Safety and efficacy of CMX001 as salvage therapy for severe adenovirus infections in immunocompromised patients. *Biol. Blood Marrow Transplant.* 18: 731–738.
- 48 Prasas, V.K. (2017). Treatment of adenovirus infection in allogeneic hematopoietic cell transplant patients with brincidofovir: final 36 week results from the advise trial. Abstr BMT Tandem Meetings, Orlando, FL, 22–26 February 2017.
- **49** O, Brien, T.P., Jeng, B.H., McDonald, M., and Raizman, M.B. (2009). Acute conjunctivitis: truth and misconceptions. *Curr. Med. Res. Opin.* 25: 1953–1961.
- For Pennington, M.R., Saha, A., Painter, D.F. et al. (2019). Disparate entry of adenoviruses dictates differential innate immune responses on the ocular surface. *Microorganisms* 7 (9): 351.
- 51 Kaneko, H., Suzutani, T., Aoki, K. et al. (2011). Epidemiological and virological features of epidemic keratoconjunctivitis due to new human adenovirus type 54 in Japan. *Br. J. Ophthalmol.* 95: 32–36.
- **52** Aoki, K. and Tagawa, Y. (2002). A twenty-one year surveillance of adenoviral conjunctivitis in Sapporo, Japan. *Int. Ophthalmol. Clin.* 42: 49–54.
- 53 Omari, A.A. and Mian, S.I. (2018). Adenoviral keratitis: a review of the epidemiology, pathophysiology, clinical features, diagnosis, and management. *Curr. Opin. Ophthalmol.* 29: 365–372.

- 54 Garcia-Zalisnak, D., Rapuano, C., Sheppard, J.D., Davis, A.R. (2018). Adenovirus ocular infections: prevalence, pathology, pitfalls, and practical pointers. *Eye Contact Lens* 44:1–7.
- 55 Chandra, N., Frangsmyr, L., Imhof, S. et al. (2019). Sialic acid-containing glycans as cellular receptors for ocular human adenoviruses: implications for tropism and treatment. *Viruses* 11 (5): 395.
- 56 Arici, C., Sultan, P., Mergen, B., et al. (2019). The impact of bilateral subepithelial corneal infiltrates on tear film after epidemic keratoconjunctivitis. *Eye Contact Lens* 44 (5): 281–284. https://doi.org/10.1097/ICL.00000000000678.
- 57 Fuchs, E. (1889). Keratitis punctata superficialis. Wien. Klin. Wochenschr. 44: 837–842.
- **58** Romanowski, E.G., Yates, K.A., and Gordon, Y.J. (2002). Topical corticosteroids of limited potency promote adenovirus replication in the Ad5/NZW rabbit ocular model. *Cornea* 21: 289–291.
- 59 Pelletier, J.S., Stewart, K., Trattler, W. et al. (2009). A combination povidone-iodine 0.4%/ dexamethasone 0.1% ophthalmic suspension in the treatment of adenoviral conjunctivitis. *Adv. Ther.* 26: 776–783.
- **60** Romanowski, E.G. (2014). Is there an anti-adenoviral drug on the horizon? *Expert Rev. Ophthalmol.* 8 (5): 427–435.
- 61 Yates, K.A., Shanks, R.M.Q., Kowalski, R.P., et al. (2018). The in vitro evaluation of povidone-iodine against multiple ocular adenoviral types. *J Ocul Pharmacol Ther* 35 (2): 132–136. https://doi.org/10.1089/jop.2018.0122.
- 62 Monnerat, N., Bossart, W., and Thiel, M.A. (2006). Povidone-iodine for treatment of adenoviral conjunctivitis: an in vitro study. *Klin. Monatsbl. Augenheilkd.* 223: 349–352.
- **63** Jhanji, V., Chan, T.C., Li, E.Y. et al. (2015). Adenoviral keratoconjunctivitis. *Surv. Ophthalmol.* 60: 435–443.
- 64 Pepose, J.S., Ahuja, A., Liu, W., et al. (2018). Randomized, controlled, phase 2 trial of povidone-iodine/dexamethasone ophthalmic suspension for treatment of adenoviral conjunctivitis. *Am J Ophthalmol* 194: 7–15. https://doi.org/10.1016/j.ajo.2018.05.012.
- **65** Kinchington, P.R., Romanowski, E.G., and Jerold, G.Y. (2005). Prospects for adenovirus antivirals. *J. Antimicrob. Chemother.* 55: 424–429.
- **66** Hillenkamp, J., Reinhard, T., Ross, R.S. et al. (2002). The effects of cidofovir 1% with and without cyclosporin a 1% as a topical treatment of acute adenoviral keratoconjunctivitis: a controlled clinical pilot study. *Ophthalmology* 109: 845–850.
- **67** Hillenkamp, J., Reinhard, T., Ross, R.S. et al. (2001). Topical treatment of acute adenoviral keratoconjunctivitis with 0.2% cidofovir and 1% cyclosporine: a controlled clinical pilot study. *Arch. Ophthalmol.* 119: 1487–1491.
- **68** Porta, C., Paglino, C., and Mutti, L. (2008). Ranpirnase and its potential for the treatment of unresectable malignant mesothelioma. *Biologics* 2: 601–609.
- **69** Squiquera, L., Taxman, D.J., Brendle, S.A. et al. (2017). Ranpirnase eradicates human papillomavirus in cultured cells and heals anogenital warts in a Phase I study. *Antivir. Ther.* 22: 247–255.
- 70 Romanowski, E.G., Yates, K., Shanks, R.M., Romanowski, J.E., and Kowalski, R.P. (2017).
 OKG-0301, a novel ribonuclease, demonstrates antiviral activity against adenovirus in the Ad5/NZW rabbit ocular model. Abstr ARVO 2017, Baltimore.

- **296** *11 Antiviral Therapy of Adenovirus Infections*
 - **71** Gower, N.J.D., Barry, R.J., Edmunds, M.R. et al. (2016). Drug discovery in ophthalmology: past success, present challenges, and future opportunities. *BMC Ophthalmol.* 16: 11.
 - **72** Caraballo, R., Saleeb, M., Bauer, J. et al. (2015). Triazole linker-based trivalent sialic acid inhibitors of adenovirus type 37 infection of human corneal epithelial cells. *Org. Biomol. Chem.* 13: 9194–9205.
 - **73** Wang, Y., De Clercq, E., and Li, G. (2019). Current and emerging non-nucleoside reverse transcriptase inhibitors (NNRTIs) for HIV-1 treatment. *Expert Opin. Drug Metab. Toxicol.* 15: 813–829.
 - 74 Sanchez-Cespedes, J., Martinez-Aguado, P., Vega-Holm, M. et al. (2016). New 4-Acyl-1phenylaminocarbonyl-2-phenylpiperazine derivatives as potential inhibitors of adenovirus infection. Synthesis, biological evaluation, and structure-activity relationships. *J. Med. Chem.* 59: 5432–5448.
 - **75** Greber, U.F. (1998). Virus assembly and disassembly: the adenovirus cysteine protease as a trigger factor. *Rev. Med. Virol.* 8: 213–222.
 - **76** Mangel, W.F. and San, M.C. (2014). Structure, function and dynamics in adenovirus maturation. *Viruses* 6: 4536–4570.
 - 77 Agbowuro, A.A., Huston, W.M., Gamble, A.B., and Tyndall, J.D.A. (2018). Proteases and protease inhibitors in infectious diseases. *Med. Res. Rev.* 38: 1295–1331.
 - 78 McGrath, W.J., Graziano, V., Zabrocka, K., and Mangel, W.F. (2013). First generation inhibitors of the adenovirus proteinase. *FEBS Lett.* https://doi.org/10.1016/j. febslet.2013.05.033.
 - **79** Mac Sweeney, A., Grosche, P., Ellis, D. et al. (2014). Discovery and structure-based optimization of adenain inhibitors. *ACS Med. Chem. Lett.* 5: 937–941.
 - **80** Grosche, P., Sirockin, F., Mac Sweeney, A. et al. (2015). Structure-based design and optimization of potent inhibitors of the adenoviral protease. *Bioorg. Med. Chem. Lett.* 25: 438–443.

ssDNA-Viruses: Human Parvovirus Infection

Susanne Modrow

12

Institute of Medical Microbiology and Hygiene, University of Regensburg, Regensburg, Germany

12.1 Introduction

Single-stranded DNA-viruses are classified into the families of *Parvoviridae*, *Circoviridae*, and *Anelloviridae*. All these viruses are characterized as small, non-enveloped particles with genomes of 1.7–2.1 kilo bases (1000 bases) (kb) (*Circoviridae*), 2.0–3.9 kb (*Anelloviridae*), and 4.0–6.0 kb (*Parvoviridae*) [1–3]. With respect to anelloviruses, various species of Torque-teno-, Torque-teno-midi- and Torque-teno-mini-viruses (TTV, TTMDV, TTMV) have been identified to persist in vertebrates including humans, but diseases have been associated neither with acute infection nor with persistence [4–6]. Circoviruses are well-known pathogens of livestock and animals: human-associated circoviral DNA-sequences could be amplified from various samples and excretions from patients as well as from healthy humans [7]. With respect to parvoviruses, several species are well-known risk factors for fetal health both in livestock and pets, e.g. porcine parvovirus, canine minute virus, and feline panleukopenia virus. Two parvoviral species, human parvovirus B19 (B19V) and human bocavirus (HBoV), are recognized as human pathogens and will be discussed in this chapter.

12.2 Classification

The family of *Parvoviridae* comprises viruses characterized by small (lat. *parvus* = small), non-enveloped particles with a diameter of 20–28 nm containing a linear single-stranded DNA (ssDNA) molecule of about 5000–6000 nucleotides. Parvovirus B19 (primate erythroparvovirus 1, B19V) and human bocavirus (primate bocaparvovirus HBoV 1, 2) occur within the subfamily *Parvovirinae*, genera *Erythroparvo-* and *Bocaparvovirus*, respectively. During the past years, several other parvovirus have been isolated from humans, including parvovirus 4 (PARV4), human bufavirus (BuV), cutavirus (CutaV), and tusavirus

298 12 ssDNA-Viruses: Human Parvovirus Infection

(TusaV) [8, 9]. Until now, their clinical significance and association with human diseases remain unclear. Furthermore, most humans are infected with adeno-associated virus (AAV), members of the genus Dependoparvovirus, without developing symptoms. Whereas AAV-replication is dependent on concurrent infection of the cells by adeno- or herpesviruses, all other human parvoviruses are not dependent on helper-viruses and replicate autonomously. Parvovirus B19 displays a preference to infect erythroid precursor cells; the tropism of human bocavirus is targeted to cells of the respiratory and/or gastrointestinal tract [10–12].

12.3 Molecular Biology

Parvoviral particles (Figure 12.1) are very stable. The non-enveloped icosahedral capsids consist of 60 capsomers formed by two (B19V: VP1, VP2) or three [HBoV: viral protein 1, viral protein 2, viral protein 3 (VP1, VP2, VP3)] structural proteins [8]. Approximately 90–95% of the particles consist of the major capsid protein VP2 (B19V: 58 kDa, HBoV: 60 kDa) [13]. Minor capsid proteins VP1 and VP3 are incorporated into the structure in non-stochiometrical relations [14, 15]. VP2s are identical to the carboxyterminal regions of VP1. Aminoterminal extensions of 227 and 129 amino acids represent the so-called "VP1-unique region" of VP1-proteins of B19V (83 kDa) and HBoV (74 kDa), respectively. VP3 (64 kDa) is present as a third component in HBoV-capsids displaying an aminoterminal extension of 39 residues as compared to VP2. Phospholipase A2-like activities are associated with the VP1-unique region [16, 17]. This enzyme activity is necessary for parvoviral infectivity and responsible for particle release from endosomes following receptor-mediated uptake. It may



Figure 12.1 Schematic structure and composition of a parvovirus particle. *Source*: Modrow, Falke, Truyen, Schatzl: Molekulare Virologie, 3.Aufl. © 2010, Springer Nature.

further contribute to inflammatory processes mediated by the production of leukotrienes and prostaglandins, the natural products of phospholipase A2 (PLA2)-enzymatic activity.

B19V- and HBoV-genomes span over 5596 and 5543 nucleotides of single-stranded DNA, respectively. Sequencing of several isolates of both parvoviral species led to the identification of distinct genotypes with sequence divergence of approximately 11–14% [18–20]. The internal coding sequences of about 4800 (B19V) and 5200 (HBoV) bases are flanked by terminal repeats that are necessary for genome replication [8, 10]. The coding regions contain two large open reading frames (ORFs), whose expression is controlled by only one promoter element at the 3'end of the genome. Viral transcripts are processed by alternative splicing to mRNAs and used for the synthesis of the respective viral proteins [21, 22]. The ORF adjacent to the promoter encodes the nonstructural proteins while capsid proteins are encoded by the ORF localized further downstream (Figure 12.2). Nonstructural proteins 1 (NS1, B19V: 77kDa; HBoV: 100kDa) are multifunctional and possess site-specific DNA-binding, endonuclease, and helicase/ATPase activities, which might be used for the development of specific antiviral drugs. The NS1 of B19V has been shown to transactivate the viral and various cellular promoters [23–26]. Its cytotoxicity is related to cell cycle arrest and apoptosis involving the caspase 3-pathway [27–30]. As similar NS1-activities are known



Figure 12.2 Genome, transcription and translation map of parvovirus B19. The line at the top of the figure represents the parvovirus B19 genome with its ITR-elements at the ends and the location of the p6-promoter. The lines underneath represent the various transcripts which are used for the translation of the respective viral proteins indicated by bars. Exon sequences are indicated by thick lines, the introns which are removed by RNA-splicing by thin lines. Polyadenylation sites are represented by the jagged symbols. *Source*: Modrow, Falke, Truyen, Schatzl: Molekulare Virologie, 3.Aufl. © 2010, Springer Nature.

for other parvoviruses, analogous functions are assumedly to be associated with NS1 of HBoV. The B19V-genome encodes at least two additional small nonstructural polypeptides of 11 and 7.5 kDa. During HBoV-infection, some smaller NS1-variants (NS2, NS3, and NS4) are produced by alternative RNA-splicing sharing the carboxyterminal domains of NS1. The function of these small nonstructural proteins is poorly understood. HBoV encodes a further nonstructural protein NP1 (25 kDa) with functions in genome replication and RNA-processing [31, 32].

12.4 Parvovirus Replication

A common feature of all parvoviruses is the dependence for infection and replication on diving cells. In contrast to other DNA-viruses, polyoma-, papilloma-, adeno-, and herpesviruses parvoviruses do not encode factors that induce the entry of the host cells into the S-phase. At least in part, this feature may explain the tropism of parvovirus to infect special cell types with distinct states of cellular differentiation. However, the analysis of the replication cycle of both B19V and HBoV is hampered by the lack of cell culture systems that allow efficient and reproducible virus propagation in vitro.

Following airborne transmission via droplets, first cycles of both B19V- and HBoVreplication are presumed to occur in the nasopharyngeal lymphoid or epithelial tissues resulting in viremia. B19V subsequently infects erythroid precursors [burst-forming units – erythroid (BFU-E), colony forming units – erythroid (CFU-E), and erythroblasts] in the bone marrow by first binding to blood group antigen P (globoside, Gb4) as cellular receptor [33–35]. Receptor-mediated endocytosis is mediated by binding to a co-receptor, α 5 β 1 integrin, in high affinity conformation [36] and/or to the autoantigen Ku80 [37].

The cellular receptor used for HBoV entry has not yet been identified. From the nasopharyngeal tissue, HBoV is transferred to the tracheobronchial (HBoV1, 3) or intestinal epithelial cells (HBoV2, 4) that become infected and support efficient replication [8]. These first events determine the different cell and organ tropism of B19V and HBoV. During the next steps, PLA2-like activity as part of the VP1-unique region catalyzes the release of the viral capsid from the endosome, followed by the transport into the nucleus, where the ssDNA genome is assumed to be completed to a dsDNA molecule by cellular DNApolymerases using the 3'end of the genome as primer. During the following steps, the transcription of the viral genome is catalyzed by cellular RNA-polymerase II. The primary transcripts are spliced, thereby generating mRNA-species that are transported into the cytoplasma and translated by ribosomes. The respective viral nonstructural and structural proteins are retransported into the nucleus. NS1 is active as transcriptional activator of the viral promoter. The endonuclease and helicase activities of NS1 are involved in the replication of the genome. Viral DNA-synthesis is performed and catalyzed by cellular DNA-polymerases δ and α [38]. In the nucleus, newly generated ssDNA-genomes assemble with VP1, VP2, and—in case of HBoV—VP3 into viral particles that are released by NS1-mediated apoptosis.
12.5 Diseases Associated with Parvovirus Infection

12.5.1 Parvovirus B19

The virus was first discovered in 1975 as a cause of systemic infections of adults who either were asymptomatic or had mild nonspecific symptoms such as headache, pyrexia, malaise, fatigue, and myalgia [39-41]. In 1983, B19V was identified as the etiological agent of erythema infectiosum (fifth disease) [42, 43]. B19V-infection is ubiquitous and frequent in both children and adults. In Europe, about 70% of adults (age 25-40 years) and more than 80% of the elderly (age > 80 years) are seropositive, although these values may vary slightly by country [44–48]. In general, B19V is transmitted by respiratory aerosol spread or by hand-to-mouth contact from individuals with acute infection [49]. In immunocompetent individuals, acute B19V-infection is asymptomatic or may induce a flu-like disease. Viremia is extremely high before the onset of detectable immune responses and up to 10¹³ particles and/or virus genomes may be present per milliliter blood [50]. Similar amounts may be detected in respiratory secretions [51] and other body fluids (S. Modrow, personal communication). Due to its tropism for erythroid precursor cells, B19V infects and destroys erythroid precursors (BFU-E, CFU-E, and erythroblasts) in the bone marrow, thereby causing transient anemia. Initiating immune responses and B19V-specific antibodies are thought to be responsible for development of rash and arthritis. Whereas arthritis is generally transient in children, symptoms may last for several weeks and months in adults. Neutralizing antibodies directed against epitopes present in the VP1-unique region and against particulate structures of VP2 persist life-long [8, 52, 53].

Besides generally mild leucocytopenia or thrombocytopenia, severe hematologic sequelae presenting as pure red cell aplasia or pancytopenia, hepatitis, myocarditis, myositis, acute lung injury, meningoencephalitis, and neurological disease may occur in rare cases [54–59]. In some patients, acute B19V-infection may induce or trigger autoimmune disorders ranging from mild arthralgias, Hashimoto thyroiditis, to severe necrotizing vasculitis [60, 61]. Furthermore, B19V-infection may be associated with a wide spectrum of additional diseases:

After acute infection and elimination of B19V from peripheral blood, low amounts of viral DNA $(10^2-10^3 \text{ geq}/10^6 \text{ cells})$ can be detected in various tissues of healthy adults, e.g. myocardium, skin, bone marrow, tonsils, and synovia, probably life-long [62–68]. Since similar concentrations of viral DNA are present in myocardial tissues from healthy individuals and patients suffering from dilated cardiomyopathy, the causal association between B19V-DNA detection and heart disease has been discussed, but appears rather questionable [69]. Whether this B19V-DNA latency in tissue may be a source for virus reactivation in immunocompetent individuals is currently unknown.

Severe disease manifestations, however, are associated with special situations or immunological and genetic features of individual patients. In pregnancy, B19V may be transmitted to the unborn child. Depending on the stage of gestation and fetal development, B19V-infection may cause spontaneous abortion, miscarriage, nonimmune hydrops fetalis, and fetal death. Due to the lack of erythroid precursor cells that are only present in fetal blood after weeks 10–12 of gestation, B19V-infection in very early pregnancy is not associated with fetal anemia. However, the risk of B19V-associated fetal anemia resulting in nonimmune hydrops fetalis is particularly high during the second trimester, between weeks of gestation 11–23 when the fetus is growing rapidly and the fetal red cell mass increases 30-fold [70–73]. Fetal death has been largely confined to maternal B19V-infection within the first 20 weeks of gestation [74–79]. Another, but rare manifestation is fetal myocarditis. After birth, some of these children need heart transplantation [80, 81]. Epidemiological data in developed countries indicate that 200–400 of 10000 pregnant women may suffer from acute B19V-infection, potentially endangering fetal life and causing 2–3 cases of fetal death per 10000 live births [82–85].

Besides pregnant women, patients with shortened half-live of red blood cells are at high risk to establish a transient aplastic crisis (TAC). The development of this severe and life-threatening disease may occur in acutely B19V-infected patients with underlying hereditary or acquired hematologic disorders. This problem is a well-known complication in patients with sickle-cell anemia. Due to chronic hematolytic anemia, these patients display increased red blood cell destruction, which is considerably worsened by the viral infection. Although less frequently, similar complications are observed in B19V-infected individuals with hereditary somatocytosis, Fanconi-anemia, thalassemia, hemoglobin C disease, red cell enzyme deficiencies, iron deficiency anemia, immune hemolytic anemia, and in patients suffering from high blood loss due to trauma or surgery [8, 86–93].

In immunocompromised B19V-infected patients, viremia due to ongoing viral replication may persist over long time periods and cause continuous or recurrent symptoms [91–94]. In the absence of antiviral immune response, symptoms induced by immunocomplex formation (rash, arthritis) do not occur. Due to their inability to produce neutralizing antibodies, immunocompromised individuals are particularly at risk to establish persistent B19V-infection, resulting in pure red-cell aplasia and severe chronic anemia [59, 95–97]. In transplant recipients, the virus may be either transmitted by contact to acutely infected individuals or via the donor organ from seropositive individuals. Whereas donor-transmitted B19V-infection is associated with transient, low-level viremia in seropositive transplant recipients, persistent high-level viremia may be observed when initially seronegative recipients, e.g. children, are infected during the phase of intensive immune suppression [98, 99].

12.5.2 HBoV

HBoV1-infection is ubiquitous and widespread, especially in children. At an age of six years, more than 95% of children are seropositive. HBoV1 has been detected in nasopharyngeal samples of patients with upper and lower respiratory tract disease, frequently in combination with other viral or bacterial pathogens [8, 100, 101]. As HBoV-DNA could also be amplified from healthy individuals, asymptomatic infection seems to be common, but some severe courses of severe airway constriction, e.g. pneumothorax and pneumomediastinum, have been reported, mostly in immunosuppressed patients [102, 103]. Following HBoV1-infection, viral genomes are detected in nasopharyngeal tissue, possibly life-long [104]. Genomes of HBoV2 have been detected preferentially in stool of patients with diarrhea, frequently combined with noro- or rotavirus [105]. As HBoV2 can also be found in stool of asymptomatic children, the causal association with disease is unclear. HBoV3 and HBoV4 infections have been diagnosed rarely, therefore the association with distinct symptoms remains unclear [105, 106]. In general, the infection is mild and self-limited, so there is no antiviral therapy available. In case of severe HBoV-infection in immunocompromised patients, immunoglobulin therapy might be considered,

12.6 Antiviral Chemotherapy of Parvovirus B19-infection

In vitro, some inhibitory effects on B19V-replication have been shown by hydroxyurea, cidofovir, brincidofovir, and coumarin derivatives, but the results were not convincing [107–111]. Since parvoviruses do not encode respective DNA-polymerases, the antiviral effects of these compounds might be associated with inhibitory effects on cellular processes and impairment of cell growth and viability. Similarly, unspecific inhibitory effects have to be discussed for telbivudine. Telbivudine is known as nucleotide inhibitor of the hepatitis B virus reverse transcriptase, an enzyme not encoded by parvoviruses. In vitro, telbivudine did not display direct effects on B19V-replication or gene expression, but may inhibit virus-induced apoptosis [112, 113]. Another approach intended to address the NS1-endonuclease activity by various flavonoid molecules similarly showed only limited effects [114].

In general, acute B19V-infection is mild and self-limited. In these uncomplicated cases, antiviral therapy is not necessary. The situation is different in immunosuppressed patients and in individuals with underlying disorders in red blood cell differentiation, which develop severe aplastic anemia. Similarly, in acutely infected pregnant women, measures to control virus propagation are not available. Respective antiparvoviral chemotherapy would be helpful to avoid transuterine B19V-transmission and infection of the fetus. The development of antiparvoviral chemotherapy would also be desirable for treatment of fetal anemia and hydrops fetalis. At present, only supportive, symptomatic, or unspecific treatment is recommended, such as erythrocyte donation and immunoglobulin therapy. The development of antiviral chemotherapy is hampered by the fact that good viral targets are scarce. Due to the restricted information encoded by parvoviral ssDNA-genomes, almost all steps of viral gene expression and genome replication are dependent on cellular enzymes. The only targets may represent the viral phospholipase A2-like activity of the VP1-unique regions and endonuclease and helicase associated with the NS1-protein. Since these viral enzymes display catalytic activities comparable to cellular functions, severe side effects have to be expected by antivirals addressing VP1and NS1-proteins and good counterscreens with the cellular enzymes will be necessary to achieve a sufficient therapeutic window. Furthermore, testing of respective compounds is impeded as neither animal models or reliable cell culture systems are available for B19V-infection.

12.7 Therapeutic Options and Recommendations

12.7.1 Acute B19V-infection Associated with Transient Aplastic Crisis (TAC)

TAC in patients with underlying hereditary or acquired hematologic is a severe and potentially life-threatening disease that has to be treated by erythrocyte transfusion.

12.7.2 Prolonged B19V-symptoms in Immunocompetents

In patients with prolonged medium-/high-level viremia, the donation of immunoglobulins may be considered. Some benefit has been shown in patients with persistent arthritis and arthropathy [115–118].

12.7.3 Fetal Disease

Intrauterine transfusion of erythrocytes is the treatment of choice in fetal anemia usually diagnosed by peak systolic velocity (PSV)-middle cerebral arterial (MCA) (PSV-MCA). The actual hemoglobin level, reticulo- and thrombocyte counts can only be measured invasively by puncture of the umbilical vein, i.e. when it is punctured for transfusion. A recent report on the management of B19V-infection in pregnancy describes pretransfusion hemoglobin concentrations ranging between 1.6 and 7.4g/dl [119]. It may be concluded that hemoglobin values below 8 g/dl require intrauterine erythrocyte transfusion [120]. In most cases, a single erythrocyte transfusion of some milliliters is sufficient to resolve anemia and hydrops. The proportion of fetuses with severe hydrops that survive after intrauterine transfusion is 83–85% [118, 119, 121]. The outcome of the treated fetuses/children is uneventful, especially regarding neurological development [122, 123]. The intravenous administration of immunoglobulin is not recommended for treatment of acute B19V-infection in pregnancy. There is only one report which describes resolution of fetal symptoms after immunoglobulin administered to the pregnant woman [124].

12.7.4 Immunocompromised Patients

High-dose intravenous immunoglobulin therapy is recommended in immunocompromised patients and transplant recipients with chronic anemia and pure red cell aplasia. Actually, 0.4g of immunoglobulins are used per kilogram body weight and day over a period of five days [94, 97, 125–128]. Following treatment, symptoms may recur at intervals and repeated application of immunoglobulin therapy may be needed.

References

- 1 Shulman, L.M. and Davidson, I. (2017). Viruses with circular single-stranded DNA genomes are everywhere! *Annu. Rev. Virol.* 4 (1): 159–180.
- 2 Breitbart, M., Delwart, E., Rosario, K. et al. (2017). ICTV virus taxonomy profile: circoviridae. *J. Gen. Virol.* 98: 1997–1998.
- **3** Cotmore, S.F., Agbandje-McKenna, M., Canuti, M. et al. (2019). ICTV virus taxonomy profile: parvoviridae. *J. Gen. Virol.* 100: 367–368.
- **4** Maggi, F. and Bendinelli, M. (2010). Human anelloviruses and the central nervous system. *Rev. Med. Virol.* 20 (6): 392–407.
- 5 Spandole, S., Cimponeriu, D., Berca, L.M., and Mihăescu, G. (2015). Human anelloviruses: an update of molecular, epidemiological and clinical aspects. *Arch. Virol.* 160 (4): 893–908.

- **6** Focosi, D., Antonelli, G., Pistello, M., and Maggi, F. (2016). Torquetenovirus: the human virome from bench to bedside. *Clin. Microbiol. Infect.* 22 (7): 589–593.
- **7** Sauvage, V., Gomez, J., Barray, A. et al. (2018). High prevalence of cyclovirus Vietnam (CyCV-VN) in plasma samples from Madagascan healthy blood donors. *Infect. Genet. Evol.* 66: 9–12.
- **8** Qiu, J., Söderlund-Venermo, M., and Young, N.S. (2017). Human parvoviruses. *Clin. Microbiol. Rev.* 30 (1): 43–113.
- **9** Brown, K.E. (2010). The expanding range of parvoviruses which infect humans. *Rev. Med. Virol.* 20 (4): 231–244.
- **10** Guido, M., Tumolo, M.R., Verri, T. et al. (2016). Human bocavirus: current knowledge and future challenges. *World J. Gastroenterol.* 22 (39): 8684–8697.
- **11** Deng, X., Li, Y., and Qiu, J. (2014). Human bocavirus 1 infects commercially available primary human airway epithelium cultures productively. *J. Virol. Methods* 195: 112–119.
- **12** Deng, X., Yan, Z., Cheng, F. et al. (2016). Replication of an autonomous human parvovirus in non-dividing human airway epithelium is facilitated through the DNA damage and repair pathways. *PLoS Pathog.* 12 (1): e1005399.
- **13** Kaufmann, B., Simpson, A.A., and Rossmann, M.G. (2004). The structure of human parvovirus B19. *Proc. Natl. Acad. Sci. USA* 101: 11628–11633.
- **14** Kawase, M., Momoeda, M., Young, N.S., and Kajigaya, S. (1995). Most of the VP1-unique region of parvovirus B19 is on the capsid surfache. *Virology* 211: 359–366.
- Kaufmann, B., Chipman, P.R., Kostyuchenko, V.A. et al. (2009). Visualization of the externalized VP2 N termini of infectious human parvovirus B19. *J. Virol.* 82: 7306–7312.
- **16** Dorsch, S., Liebisch, G., Kaufmann, B. et al. (2002). The VP1-unique region of parvovirus B19 and its constituent phospholipase A2-like activity. *J. Virol.* 76: 2014–2018.
- **17** Zadori, Z., Szelei, J., Lacoste, M.C. et al. (2001). A viral phospholipase A2 is required for parvovirus infectivity. *Dev. Cell* 1: 291–302.
- **18** Nguyen, Q.T., Sifer, C., Schneider, V. et al. (1999). Novel human erythrovirus associated with transient aplastic anemia. *J. Clin. Microbiol.* 37: 2483–2487.
- **19** Hokynar, K., Söderlund-Venermo, M., Pesonen, M. et al. (2002). A new parvovirus genotype persistent in human skin. *Virology* 302: 224–228.
- **20** Servant, A., Laperche, S., Lallemand, F. et al. (2002). Genetic diversity within human erythroviruses: identification of three genotypes. *J. Virol.* 76: 9124–9134.
- **21** Ozawa, K., Ayub, J., Hao, Y.S. et al. (1987). Novel transcription map for the B19 (human) pathogenic parvovirus. *J. Virol.* 61: 2395–2406.
- **22** Shimomura, S., Wong, S., Brown, K.E. et al. (1993). Early and late gene expression in UT-7 cells infected with B19 parvovirus. *Virology* 194: 149–156.
- **23** Raab, U., Bauer, B., Gigler, A. et al. (2001). Cellular transcription factors that interact with the p6 promoter elements of parvovirus B19. *J. Gen. Virol.* 82: 1473–1480.
- **24** Raab, U., Beckenlehner, K., Lowin, T. et al. (2002). NS1-protein of parvovirus B19 interacts directly with DNA sequences of the p6-promoter and with cellular transcription factors Sp1/Sp3. *Virology* 293: 86–93.
- **25** Vassias, I., Hazan, U., Michel, Y. et al. (1998). Regulation of human B19 parvovirus promoter expression by hGABP (E4TF1) transcription factor. *J. Biol. Chem.* 273: 8287–8293.

306 12 ssDNA-Viruses: Human Parvovirus Infection

- **26** Fu, Y., Ishii, K.K., Munakata, Y. et al. (2002). Regulation of tumor necrosis factor alpha promoter by human parvovirus B19 NS1 through activation of AP-1 and AP-2. *J. Virol.* 76: 5395–5403.
- 27 Moffatt, S., Tanaka, N., Tada, K. et al. (1996). A cytotoxic nonstructural protein, NS1, of human parvovirus B19 induced activation of interleukin 6-gene expression. *J. Virol.* 70: 8485–8491.
- **28** Moffatt, S., Yaegashi, N., Tada, K. et al. (1998). Human parvovirus B19 nonstructural (NS1) protein induces apoptosis in erythroid lineage cells. *J. Virol.* 72: 3018–3028.
- **29** Sol, N., Le Junter, J., Vassias, I. et al. (1999). Possible interaction between the NS-1 protein and tumor necrosis factor alpha pathways in erythroid cell apoptosis induced by human parvovirus B19. *J. Virol.* 73: 8762–8770.
- **30** Morita, E., Nakashima, A., Asao, H. et al. (2003). Human parvovirus B19 nonstructural protein (NS1) induces cell cycle arrest at G(1) Phase. *J. Virol.* 77: 2915–2921.
- **31** Dong, Y., Fasina, O.O., and Pintel, D.J. (2018). The human bocavirus 1 NP1 protein is a multifunctional regulator of viral RNA processing. *J. Virol.* 92 (22): e01187–e01118.
- **32** Zou, W., Cheng, F., Shen, W. et al. (2016). Nonstructural protein NP1of human bocavirus 1 plays a critical role in the expression of viral capsid proteins. *J. Virol.* 90 (9): 4658–4669.
- **33** Brown, K.E., Anderson, S.M., and Young, N.S. (1993). Erythrocyte P antigen: cellular receptor for B19 parvovirus. *Science* 262: 114–117.
- **34** Anderson, M.J., Higgins, P.G., Davis, L.R. et al. (1985). Experimental parvovirual infection in humans. *J. Infect. Dis.* 152: 257–265.
- **35** Nasir, W., Nilsson, J., Olofsson, S. et al. (2014). Parvovirus B19 VLP recognizes globoside in supported lipid bilayers. *Virology* 456-457: 364–369.
- **36** Weigel-Kelley, K.A., Yoder, M.C., and Srivastava, A. (2003). Alpha5beta1 integrin as a cellular coreceptor for human parvovirus B19: requirement for functional activation of beta1 integrin for viral entry. *Blood* 102: 3927–3933.
- **37** Munakata, Y., Saito-Ito, T., Kumura-Ishii, K. et al. (2005). Ku80 autoantigen as a cellular coreceptor for human parvovirus B19 infection. *Blood* 106: 3449–3456.
- **38** Zou, W., Wang, Z., Xiong, M. et al. (2018). Human parvovirus B19 utilizes cellular DNA replication machinery for viral DNA replication. *J. Virol.* 92 (5): e01881–e01817.
- **39** Cossart, Y.E., Cant, B., Field, A.M., and Widdows, D. (1975). Parvovirus-like particles in human sera. *Lancet* 1: 72–73.
- **40** Paver, W.K. and Clarke, S.K.R. (1976). Comparison of human fecal and serum parvo-like viruses. *J. Clin. Microbiol.* 4: 67–70.
- **41** Shneerson, J.M., Mortimer, P.P., and Vandervelde, E.M. (1980). Febrile illness due to a parvovirus. *Br. Med. J.* 280: 1580.
- **42** Anderson, M., Jones, S.E., Fisher-Hoch, S.P. et al. (1983). The human parvovirus, the cause of erythema infectiosum (fifth disease)? *Lancet* i: 1378.
- **43** Anderson, M., Lewis, E., Kidd, I.M. et al. (1984). An outbreak of erythema infectiosum associated with human parvovirus infection. *J. Hyg. (London)* 93: 85–93.
- **44** Heegaard, E.D. and Brown, K.E. (2002). Human parvovirus B19. *Clin. Microbiol. Rev.* 15: 485–505.
- **45** Cohen, B.J. and Buckley, M.M. (1988). The prevalence of antibody to human parvovirus B19 in England and Wales. *J. Med. Microbiol.*: 25151–25153.

- **46** Kelly, H.A., Siebert, D., Hammond, R. et al. (2000). The age-specific prevalence of human parvovirus immunity in Victoria, Australia compared with other parts of the world. *Epidemiol. Infect.* 124: 449–457.
- **47** Tsujimura, M., Matsushita, K., Shiraki, H. et al. (1995). Human parvovirus B19 infection in blood donors. *Vox Sang.* 69: 206–212.
- **48** Röhrer, C., Gärtner, B., Sauerbrei, A. et al. (2008). Seroprevalence of parvovirus B19 in the German population. *Epidemiol. Infect.* 136: 1564–1575.
- **49** Chorba, T., Coccia, P., Holman, R.C. et al. (1996). The role of parvovirus B19 in aplastic crisis and erythema infectiosum. *J. Infect. Dis.* 154: 383–393.
- **50** Bredl, S., Plentz, A., Wenzel, J.J. et al. (2011). False-negative serology in patients with acute parvovirus B19 infection. *J. Clin. Virol.* 51: 115–120.
- **51** Modrow, S. and Gärtner, B. (2006). Parvovirus B19 infection in pregnancy. *Deutsches Ärtzeblatt.* 43: A2869–A2876.
- 52 Gigler, A., Dorsch, S., Hemauer, A. et al. (1999). Generation of neutralizing human monoclonal antibodies against parvovirus B19 proteins. *J. Virol.* 73 (3): 1974–1979.
- **53** Sun, Y., Klose, T., Liu, Y. et al. (2019). Structure of parvovirus B19 decorated by Fabs from a human antibody. *J. Virol.* 93 (9): e01732–e01718.
- **54** Bültmann, B.D., Klingel, K., Sotlar, K. et al. (2003). Parvovirus B19: a pathogen responsible for more than hematologic disorders. *Virchows Arch.* 442: 8–17.
- 55 Langnas, A.N., Markin, R.S., Cattral, M.S., and Naides, S.J. (1995). Parvovirus B19 as a possible causative agent of fulminant liver failure and associated aplastic anaemia. *Hepatology* 22: 1661–1665.
- 56 Rogers, B.B., Rogers, Z.R., and Timmons, C.F. (1996). Polymerase chain reaction amplification of archival material for parvovirus B19 in children with transient erythroblastopenia of childhood. *Pediatr. Pathol. Lab. Med.* 16: 471–478.
- **57** Bousvaros, A., Sundel, R., Thorne, G.M. et al. (1998). Parvovirus B19-associated interstitial lung disease, hepatitis, and myositis. *Pediatr. Pulmonol.* 26: 365–369.
- **58** Barah, F., Vallely, P.J., Chiswick, M.L. et al. (2001). Human parvovirus B19 infection associated with acute meningoencephalitis. *Lancet* 358: 729–730.
- 59 Kerr, J.R. and Modrow, S. (2006). Human and primate parvovirus infections and associated disease. In: *Parvoviruses* (eds. J.R. Kerr, S.F. Cotmore, M.E. Bloom, et al.), 385–416. London: Arnold Publishers, Hodder.
- **60** Lehmann, H.W., von Landenberg, P., and Modrow, S. (2003). Parvovirus B19 infection and autoimmune disease. *Autoimmun. Rev.* 2: 218–223.
- **61** Lehmann, H.W., Lutterbüse, N., Plentz, A. et al. (2008). Association of parvovirus B19 infection and Hashimoto's thyroiditis in children. *Viral Immunol.* 21: 379–383.
- **62** Norja, P., Hokynar, K., Aaltonen, L.M. et al. (2006). Bioportfolio: lifelong persistence of variant and prototypic erythrovirus DNA genomes in human tissue. *Proc. Natl. Acad. Sci. USA* 103: 7450–7453.
- **63** Kuethe, F., Lindner, J., Matschke, K. et al. (2009). Prevalence of parvovirus B19 and human bocavirus DNA in the heart of patients with no evidence of dilated cardiomyopathy or myocarditis. *Clin. Infect. Dis.* 49: 1660–1666.
- **64** Cassinotti, P., Burtonboy, G., Fopp, M., and Siegl, G. (1997). Evidence for persistence of human parvovirus B19 DNA in bone marrow. *J. Med. Virol.* 53: 229–232.

- 308 12 ssDNA-Viruses: Human Parvovirus Infection
 - 65 Lotze, U., Egerer, R., Tresselt, C. et al. (2004). Frequent detection of parvovirus B19 genome in the myocardium of adult patients with idiopathic dilated cardiomyopathy. *Med. Microbiol. Immunol. (Berl)* 193: 75–82.
 - **66** Söderlund, M., von Essen, R., Haapasaari, J. et al. (1997). Persistence of parvovirus B19 DNA in synovial membranes of young patients with and without chronic arthropathy. *Lancet* 349: 1063–1065.
 - **67** Söderlund-Venermo, M., Hokynar, K., Nieminen, J. et al. (2002). Persistence of human parvovirus B19 in human tissues. *Pathol. Biol. (Paris)* 50: 307–316.
 - **68** Kuethe, F., Sigusch, H.H., Hilbig, K. et al. (2007). Detection of viral genome in the myocardium: lack of prognostic and functional relevance in patients with acute dilated cardiomyopathy. *Am. Heart J.* 153: 850–858.
 - **69** Rigopoulos, A.G., Klutt, B., Matiakis, M. et al. (2019). Systematic review of PCR proof of parvovirus B19 genomes in endomyocardial biopsies of patients presenting with myocarditis or dilated cardiomyopathy. *Viruses* 11 (6): E566.
 - **70** Morey, A.L., Ferguson, D.J., and Fleming, K.A. (1993). Ultrastructural features of fetal erythroid precursors infected with parvovirus B19 in vitro: evidence of cell death by apoptosis. *J. Pathol.* 169: 213–220.
 - **71** Yaegashi, N., Niinuma, T., Chisaka, H. et al. (1998). The incidence of, and factors leading to, parvovirus B19-related hydrops fetalis following maternal infection; report of 10 cases and meta-analysis. *J. Infect.* 37: 28–35.
 - **72** Yaegashi, N., Niinuma Chisaka, H. et al. (1999). Serologic study of human parvovirus B19 infection in pregnancy in Japan. *J. Infect.* 38: 30–35.
 - **73** Gray, E.S., Davidson, R.J., and Anand, A. (1987). Human parvovirus and fetal anaemia. *Lancet* i: 1144.
 - 74 Miller, E., Fairley, C.K., Bohen, B.J., and Seng, C. (1998). Immediate and long term outcome of human parvovirus B19 infection in pregnancy. *Br. J. Obstet. Gynecol.* 105: 174–178.
 - **75** Chisaka, H., Ito, K., Niikura, H. et al. (2006). Clinical manifestations and outcomes of parvovirus B19 infection during pregnancy in Japan. *Tohoku J. Exp. Med.* 209: 277–283.
 - 76 Enders, M., Weidner, A., Zoellner, I. et al. (2004). Fetal morbidity and mortality after acute human parvovirus B19 infection in pregnancy: prospective evaluation of 1018 cases. *Prenat. Diagn.* 24: 513–518.
 - **77** Nyman, M., Tolfvenstam, T., Petersson, K. et al. (2002). Detection of human parvovirus B19 infection in first-trimester fetal loss. *Obstet. Gynecol.* 99: 795–798.
 - 78 Enders, M., Klingel, K., Weidner, A. et al. (2010). Risk of fetal hydrops and non-hydropic late intrauterine fetal death after gestational parvovirus B19 infection. *J. Clin. Virol.* 49: 163–168.
 - **79** Riipinen, A., Väisänen, E., Nuutila, M. et al. (2008). Parvovirus B19 infection in fetal deaths. *Clin. Infect. Dis.* 47: 1519–1525.
 - **80** Nigro, G., Bastianon, V., Colloridi, V. et al. (2000). Human parvovirus B19 infection in infancy associated with acute and chronic lymphocytic myocarditis and high cytokine levels: report of 3 cases and review. *Clin. Infect. Dis.* 31: 65–69.
 - 81 von Kaisenberg, C.S., Bender, G., Scheewe, J. et al. (2001). A case of fetal parvovirus B19 myocarditis, terminal cardiac heart failure, and perinatal heart transplantation. *Fetal Diagn. Ther.* 16: 427–432.

- **82** Brown, T., Anand, A., Ritchie, L.D. et al. (1984). Intrauterine parvovirus infection associated with hydrops fetalis. *Lancet* ii: 1033–1034.
- **83** Anand, A., Gray, E.S., Brown, T. et al. (1987). Human parvovirus infection in pregnancy and hydrops fetalis. *N. Engl. J. Med.* 316: 183–186.
- **84** Vyse, A.J., Andrews, N.J., Hesketh, L.M., and Pebody, R. (2007). The burden of parvovirus B19 infection in women of childbearing age in England and Wales. *Epidemiol. Infect.* 135: 1354–1362.
- **85** de Jong, E.P., Walther, F.J., Kroes, A.C., and Oepkes, D. (2011). Parvovirus B19 infection in pregnancy: new insights and management. *Prenat. Diagn.* 31: 419–425.
- 86 Pattison, J.R. (1987). B19 virus—a pathogenic human parvovirus. Blood Rev. 1 (1): 58-64.
- 87 Conrad, M.E., Studdard, H., and Anderson, L.J. (1988). Aplastic crisis in sickle cell disorders: bone marrow necrosis and human parvovirus infection. *Am J Med Sci* 295 (3): 212–215.
- **88** Mabin, D.C. and Chowdhury, V. (1990). Aplastic crisis caused by human parvovirus in two patients with hereditary stomatocytosis. *Br. J. Haematol.* 76 (1): 153–154.
- **89** Cubel, R.C., Valadão, M.C., Pereira, W.V. et al. (1992). Aplastic crisis due to human parvovirus B19 infection in hereditary hemolytic anaemia. *Rev. Inst. Med. Trop. Sao Paulo* 34 (5): 479–482.
- **90** Servey, J.T., Reamy, B.V., and Hodge, J. (2007). Clinical presentations of parvovirus B19 infection. *Am. Fam. Physician* 75 (3): 373–376.
- **91** Slavov, S.N., Kashima, S., Pinto, A.C., and Covas, D.T. (2011). Human parvovirus B19: general considerations and impact on patients with sickle-cell disease and thalassemia and on blood transfusions. *FEMS Immunol. Med. Microbiol.* 62 (3): 247–262.
- **92** Tsitsikas, D.A., Gallinella, G., Patel, S. et al. (2014). Bone marrow necrosis and fat embolism syndrome in sickle cell disease: increased susceptibility of patients with non-SS genotypes and a possible association with human parvovirus B19 infection. *Blood Rev.* 28 (1): 23–30.
- **93** Kerr, J.R. (2015). A review of blood diseases and cytopenias associated with human parvovirus B19 infection. *Rev. Med. Virol.* 25 (4): 224–240.
- **94** Liefeldt, L., Plentz, A., Klempa, B. et al. (2005). Recurrent high level parvovirus B19/ genotype 2 viremia in a renal transplant recipient analyzed by real-time PCR for simultaneous detection of genotypes 1 to 3. *J. Med. Virol.* 75: 161–169.
- **95** Kurtzman, G.J., Ozawa, K., Cohen, B. et al. (1987). Chronic bone marrow failure due to persistent B19 parvovirus infection. *N. Engl. J. Med.* 317: 287–294.
- **96** Pont, J., Puchhammer-Stöckl, E., Chott, A. et al. (1992). Recurrent granulocytic aplasia as clinical presentation of a persistent parvovirus B19 infection. *Br. J. Haematol.* 80: 160–165.
- **97** Plentz, A., Hahn, J., Holler, E. et al. (2004). Long-term parvovirus B19 viraemia associated with pure red cell aplasia after allogeneic bone marrow transplantation. *J. Clin. Virol.* 31: 16–19.
- **98** Plentz, A., Würdinger, M., Kudlich, M., and Modrow, S. (2013). Low-level DNAemia of parvovirus B19 (genotypes 1–3) in adult transplant recipients is not associated with anaemia. *J. Clin. Virol.* 58 (2): 443–448.
- **99** Würdinger, M., Modrow, S., and Plentz, A. (2017). Impact of parvovirus B19 Viremia in liver transplanted children on anemia: a retrospective study. *Viruses* 9 (6): E149.

- 310 12 ssDNA-Viruses: Human Parvovirus Infection
 - **100** Lindner, J. and Modrow, S. (2008). Human bocavirus—a novel parvovirus to infect humans. *Intervirology* 51 (2): 116–122.
 - **101** Karalar, L., Lindner, J., Schimanski, S. et al. (2010). Prevalence and clinical aspects of human bocavirus infection in children. *Clin. Microbiol. Infect.* 16 (6): 633–639.
 - 102 Uršič, T., Jevšnik, M., Zigon, N. et al. (2012). Human bocavirus and other respiratory viral infections in a 2-year cohort of hospitalized children. *J. Med. Virol.* 84 (1): 99–108.
 - 103 Sadeghi, M., Kantola, K., Finnegan, D.P. et al. (2013). Possible involvement of human bocavirus 1 in the death of a middle-aged immunosuppressed patient. *J. Clin. Microbiol.* 51 (10): 3461–3463.
 - **104** Falcone, V., Ridder, G.J., Panning, M. et al. (2011). Human bocavirus DNA in paranasal sinus mucosa. *Emerg. Infect. Dis.* 17 (8): 1564–1565.
 - **105** Jartti, T., Hedman, K., Jartti, L. et al. (2012). Human bocavirus-the first 5 years. *Rev. Med. Virol.* 22 (1): 46–64.
 - **106** Broccolo, F., Falcone, V., Esposito, S., and Toniolo, A. (2015). Human bocaviruses: possible etiologic role in respiratory infection. *J. Clin. Virol.* 2: 75–81.
 - 107 Bonvicini, F., Bua, G., Manaresi, E., and Gallinella, G. (2016). Enhanced inhibition of parvovirus B19 replication by cidofovir in extendedly exposed erythroid progenitor cells. *Virus Res.* 220: 47–51.
 - **108** Bonvicini, F., Bua, G., Conti, I. et al. (2017). Hydroxyurea inhibits parvovirus B19 replication in erythroid progenitor cells. *Biochem. Pharmacol.* 136: 32–39.
 - **109** Bua, G., Conti, I., Manaresi, E. et al. (2019). Antiviral activity of brincidofovir on parvovirus B19. *Antivir. Res.* 162: 22–29.
 - **110** Conti, I., Morigi, R., Locatelli, A. et al. (2019). Synthesis of 3-(imidazo[2,1-b]thiazol-6yl)-2H-chromen-2-one derivatives and study of their antiviral activity against parvovirus B19. *Molecules* 24 (6): E1037.
 - Manaresi, E. and Gallinella, G. (2019). Advances in the development of antiviral strategies against parvovirus B19. *Viruses* 11 (7): 659. https://doi.org/10.3390/v11070659. PMID: 31323869; PMCID: PMC6669595.
 - **112** Zobel, T., Bock, C.T., Kühl, U. et al. (2019). Telbivudine reduces parvovirus B19-induced apoptosis in circulating angiogenic cells. *Viruses* 11 (3): E227.
 - **113** Van Linthout, S., Elsanhoury, A., Klein, O. et al. (2018). Telbivudine in chronic lymphocytic myocarditis and human parvovirus B19 transcriptional activity. *ESC Heart Fail.* 5 (5): 818–829.
 - Xu, P., Ganaie, S.S., Wang, X. et al. (2019). Endonuclease activity Inhibition of the NS1 Protein of Parvovirus B19 as a Novel Target for Antiviral Drug Development. *Antimicrob. Agents Chemother.* 63 (3): e01879–e01818.
 - 115 Stahl, H.D., Pfeiffer, R., and Emmrich, F. (2000). Intravenous treatment with immunoglobulins may improve chronic undifferentiated mono- and oligoarthritis. *Clin. Exp. Rheumatol.* 18 (4): 515–517.
 - 116 Lehmann, H.W., Plentz, A., Von Landenberg, P. et al. (2004). Intravenous immunoglobulin treatment of four patients with juvenile polyarticular arthritis associated with persistent parvovirus B19 infection and antiphospholipid antibodies. *Arthritis Res Ther.* 6 (1): R1–R6.

- Ogawa, E., Otaguro, S., Murata, M. et al. (2008). Intravenous immunoglobulin therapy for severe arthritis associated with human parvovirus B19 infection. *J. Infect. Chemother.* 14 (5): 377–382.
- **118** Bozzola, E., Krzysztofiak, A., and Cortis, E. (2010). Neurological impairment and arthritis in an immunocompetent child with human parvovirus B19 chronic infection. *Infez Med.* 18 (3): 187–190.
- **119** Simms, R.A., Liebling, R.E., Patel, R.R. et al. (2009). Management and outcome of pregnancies with parvovirus B19 infection over seven years in a tertiary fetal medicine unit. *Fetal Diagn. Ther.* 25: 373–378.
- **120** Plentz, A. and Modrow, S. (2011). Diagnosis, management and possibilities to prevent parvovirus B19 infection in pregnancy. *Future Virol.* 6: 1435–1450.
- 121 Nyman, M., Skjöldebrand-Sparre, L., and Broliden, K. (2005). Non-hydropic intrauterine fetal death more than 5 months after primary parvovirus B19 infection. *J. Perinat. Med.* 33: 176–178.
- **122** Rodis, J.F., Borgida, A.F., Wilson, M. et al. (1998). Management of parvovirus infection in pregnancy and outcomes of hydrops: a survey of members of the Society of Perinatal Obstetricians. *Am. J. Obstet. Gynecol.* 179: 985–988.
- 123 Dembinski, J., Haverkamp, F., Maara, H. et al. (2002). Neurodevelopmental outcome after intrauterine red cell transfusion for parvovirus B19-induced fetal hydrops. *BJOG* 109: 1232–1234.
- **124** Selbing, A., Josefsson, A., Dahle, L.O., and Lindgren, R. Parvovirus B19 infection during pregnancy treated with high-dose intravenous gammaglobulin. *Lancet* 345: 660–661.
- **125** Crabol, Y., Terrier, B., Rozenberg, F. et al. Intravenous immunoglobulin therapy for pure red cell aplasia related to human parvovirus B19 infection: a retrospective study of 10 patients and review of the literature. *Clin. Infect. Dis.* 56 (7): 968–977.
- 126 Liefeldt, L., Buhl, M., Schweickert, B. et al. (2002). Eradication of parvovirus B19 infection after renal transplantation requires reduction of immunosuppression and high-dose immunoglobulin therapy. *Nephrol. Dial. Transplant. off. Publ. Eur. Dial. Transpl. Assoc. Eur. Renal Assoc.* 17: 1840–1842.
- **127** Waldman, M. and Kopp, J.B. (2007). Parvovirus-B19-associated complications in renal transplant recipients. *Nat. Clin. Pract. Nephrol.* 3 (10): 540–550.
- **128** Modrow, S. and Huzly, D. Labordiagnostik schwangerschaftsrelevanter Virusinfektionen. https://www.awmf.org/leitlinien/detail/ll/093-002.html (accessed 31 May 2019).

Jana Van Dycke¹, Arno Cuvry¹, Johan Neyts^{1,2}, and Joana Rocha-Pereira¹

 ¹ KU Leuven – Department of Microbiology, Immunology and Transplantation, Rega Institute, Laboratory of Virology and Chemotherapy, Leuven, Belgium
² Global Virus Network (GVN), Centers of Excellence, Baltimore, MD, USA

13.1 Introduction

13.1.1 Disease Burden and Pathogenesis

Human norovirus (HuNoV), previously known as Norwalk virus, was first discovered after an outbreak of gastroenteritis in Norwalk, Ohio [1], and was the first viral agent shown to cause gastroenteritis. HuNoV is the most common cause of viral gastroenteritis, causing annually ~700 million infections and resulting in ~200 000 deaths worldwide [2]. Norovirus gastroenteritis affects all age groups, with symptomatic infections being characterized by an acute onset of nausea, abdominal pain, vomiting, and non-bloody diarrhea. The incubation period is 24–48 hours, with the symptoms lasting 2–3 days in healthy adults. A norovirus infection can be particularly severe for young children, the elderly, and immunocompromised, lasting weeks to months and with higher risk of hospitalization than for the general population. Diarrhea caused by various agents is still one of the most important causes of childhood mortality in low- and middle-income countries; one million young children die annually due to diarrhea before they reach the age of 5 [3-5]. Around 70% of these cases have a viral etiology [6]. Besides HuNoV, other human viruses cause diarrhea, mostly in young children. These include human rotaviruses, human enteric adenoviruses (Adenoviridae, dsDNA), human sapovirus (HuSaV, also belonging to the Caliciviridae, with (+)ssRNA) and human astroviruses (HAstVs, Astroviridae, (+)ssRNA). In young children, rota- and norovirus are the two most important gastro enteric viruses. Noroviruses cause an estimate of 200 000 deaths yearly and are currently the most important pathogen causing severe childhood diarrhea in countries where vaccination against rotavirus is routinely implemented [7, 8]. Among the elderly, norovirus accounts for the majority of gastroenteritis-associated hospitalizations [9, 10]. In patients with an immunocompromised state, norovirus gastroenteritis can become chronic and may persist for months to years [11-14]. This can result in dramatic weight loss due to prolonged diarrhea,

malnutrition, and dehydration [15]; all of these will contribute to an altered intestinal mucosal barrier, increased morbidity, and will highly likely worsen the underlying condition [16]. Reduction of immunosuppressive therapy is, when feasible, the strategy of choice in transplant patients.

Norovirus outbreaks are very common due to a plethora of reasons. The virus particles remain infectious for up to two weeks on surfaces, are resistant to many disinfectants, heating up to 60 °C and freezing. Moreover, infected persons can shed the virus for 20–40 days in their feces (also in asymptomatic infections) and it is estimated that only ~18 particles are necessary to cause an infection [17–20]. Norovirus is acquired via the fecal–oral route, most often via the consumption of contaminated food or water, or person-to-person contact due to the aerosolized particles from vomit or stool. Transmission occurs mostly in communal, semi-closed environments [21–23]. Long-term care facilities are the most common setting for norovirus outbreaks, followed by restaurants, schools, hospitals, and cruise ships [21]. Noroviruses are a major culprit for the closure of hospital wards. Every winter in the United Kingdom, an average of up to 113000 beds are unavailable due to gastroenteritis, resulting in a final cost in the United Kingdom of £6.9–£10.0 million when including staff absence costs due to illness [24, 25]. The closure of hospital wards, may be cost-effective if done efficient and particularly if targeted to high-throughput units [25].

Currently there is no approved antiviral available to treat and/or prevent a norovirus infection. The current treatment is merely supportive through rehydration and supplementation of electrolytes. In some patients, approved drugs have been used off-label to treat norovirus infections with varied success, such as ribavirin [26], favipiravir [27], and nitazoxanide [28]. The strategy for the development of a HuNoV antiviral could go in two different directions. Firstly, an antiviral could be developed to treat ongoing acute and chronic infections. This could have a tremendous impact on the morbidity and mortality of populations at risk for a prolonged and severe disease (young children, the elderly, and immunocompromised), while also reducing the shedding of the virus and therefore reducing the risk of further transmission. The latter has been shown to be successful in the treatment of HIV, as antiretroviral drugs can reduce the concentration of virus in the blood and genital secretions of the person with HIV and therefore reduce transmission of the virus. Second, a prophylactic approach should be considered as HuNoV outbreaks are very impactful and antivirals could also play a major role in the prevention and control of outbreaks that occur in hospitals or cruise ships, for example. One could also consider household prophylaxis to prevent the virus to spread from a child to the parents and siblings, as the probability of this happening is high. Furthermore, HuNoV outbreaks mostly occur during winter. Since nursing homes are the most common setting and comprise a high-risk population, strategies for seasonal prophylactic use of an antiviral could be designed. Antivirals as prophylaxis are successfully being used against other human viruses such as influenza A virus [29]. In addition, it has been put forth that antiviral strategies should not per se target merely a single virus, but also yield options for syndrome-based treatment, including the development of broad-spectrum antivirals. This would be particularly relevant for acute viral gastroenteritis, as the clinical symptoms are similar for every viral agent and treatment should be started as early as possible. As multiple viruses can cause diarrhea, an antiviral that could target multiple viral agents of diarrhea, by focusing on highly conserved viral proteins such as the RNA-dependent RNA polymerase (RdRp), could be the way forward [30, 31].

13.1.2 Viral Genome

The *Caliciviridae* are a family of small (27-40 nm), non-enveloped viruses with a linear (+)ssRNA genome and an icosahedral capsid, containing 11 different genera. Noro- and Sapovirus are the only genera that comprise human viruses. The norovirus genome is organized into three open reading frames (ORF1-3). The ORF1 of norovirus encodes the six nonstructural (NS) proteins in the following order: the p48/N-terminal protein (or NS1/2), the NTPase (NS3), the p22 (NS4), the VPg (NS5), the viral protease (Pro, NS6), and the viral RNA-dependent RNA polymerase (RdRp, NS7). ORF2 and ORF3 encode for the major and minor structural capsid proteins VP1 and VP2, respectively (Figure 13.1). In the mouse norovirus (MNV), a fourth ORF is described, this is encoded by the subgenomic RNA, in an alternative reading frame overlapping with the VP1 coding region [32]. The ORF4 is translated during virus infection, encoding for virulence factor 1 (VF1) which localizes mostly to the mitochondria and has a role in infection and virulence. Due to an increase in reports of new noroviruses, the classification was recently updated [33]. The genus Norovirus is now organized in 10 genogroups (GI-GX), which can be further divided into 60 P-types based on the sequence of the RdRp. The addition of the RdRp sequence in the classification is due to the high recombination rate in the junction of ORF 1 and 2 [34]. Norovirus from the genogroups GI, GII, GIV, GVIII, and GIX contain strains that can infect humans, GIII comprises bovine and ovine strains and GV holds the mouse and rat noroviruses. GIV, GVI, and GVII include canine strains and GX contains bat strains [33]. There are no reports of zoonotic transmission, although this could potentially occur since antibodies against HuNoV have been found in swine's and HuNoV can infect gnotobiotic piglets and calves [35, 36]. The GII.4 noroviruses are responsible for the vast majority of outbreaks in humans (80%) and are linked to the highest mortality and hospitalization rates over the last years [37]. Since 2014, the GII.17 has also emerged in eastern Asia and has spread globally [38, 39].

13.1.3 Replication Cycle

The current knowledge on HuNoV replication still derives partly from studies with related caliciviruses and is based on the analogy with other (+)ssRNA viruses. HuNoV replication occurs in the cytoplasm of the cell. In order to attach to the cell surface, the P2 subdomain of the VP1 capsid protein interacts with the histo-blood group antigens (HBGAs), heparan sulphate or sialic acid [40–43]. HBGAs are carbohydrates that contain saccharide moieties that are expressed on red blood cells and mucosal epithelia. They can also be found as free oligosaccharides in saliva, blood, and intestinal contents [44]. The different HBGAs are



Figure 13.1 The norovirus genome. Source: Created with www.biorender.com.

formed through sequential addition of a monosaccharide to the terminal disaccharide of a precursor glycan: this involves multiple fucosyltransferases (FUT). Susceptibility to HuNoV depends on FUT2 and FUT3; these are responsible for the biosynthesis of secretor and Lewis antigens, respectively. Individuals who lack both alleles of the FUT2 gene (nonsecretors) are less susceptible to GII.4 HuNoV infection [45]. Although HBGAs are considered the main attachment factor for HuNoV, there are HuNoV strains that do not interact with any of the available synthetic HBGAs [46-48]. This suggests that it is possible that another (still unidentified) protein receptor or additional cofactors may be required [48–51]. For HuNoV GII.3, it was shown that bile acid was necessary for replication in the human intestinal enteroids (HIE) [52]. It was recently reported that bile acids promote replication by enhanced endosomal uptake, endosomal acidification and subsequent activity of endosomal/lysosomal enzyme acid sphingomyelinase (ASM), and enhanced ceramide levels on the apical membrane [53]. The role of bile acid for other HuNoV genotypes remains to be investigated. In contrast to HuNoV, the entry process of MNV is better understood. MNV entry is clathrin/caveolin-independent and mediated by dynamin II and cholesterol [54]. Entry is pH-independent, which is expected for an enteric virus; acidic pH does not cause any conformational changes in the capsid [55]. Proteinaceous receptors (CD34, CD300lf, CD300ld) were found to modulate and facilitate MNV entry and infection [56, 57]. But CD300lf was shown not to be a receptor for HuNoV [58]. Although the process of virus uncoating is not known, recent work showed that the minor capsid protein of feline calicivirus (FCV) forms a pore in the capsid upon receptor engagement, hypothetically playing an important role in viral genome release [59]. Once the viral genome is released into the cytoplasm, translation starts by the recruitment of the cell translation machinery through interactions with the VPg. Translation of the ORF1 results in a large polyprotein which is cleaved co- and post-translationally by the viral NS6 protease. This results in the release of the NS proteins, ready for replication complex formation. The replication complex is formed by recruitment of cellular membranes to the perinuclear region of the cell, through interactions with NS1/2 and NS4 [60-62]. After the replication complex is fully assembled, the negative RNA strand (antigenomic RNA) is synthesized from the mRNA template. This antigenomic RNA is then used as a template for production of new genomic and subgenomic RNA (ORF2/3) by the viral RdRp. Due to the large amount of VP1 particles required for the capsid formation, the VP1 and VP2 proteins are translated from the subgenomic RNA as this is present at higher levels than the genomic RNA. Finally, the replicated genomes are translated (within the replication complex) or packaged into the capsid for virion assembly and exit. It was recently shown that noroviruses can exit in a non-lytic manner, as viral clusters enclosed within vesicles [63]. These vesicles provide a protective environment and significantly contribute to viral infectivity and transmission.

13.1.4 In Vitro Models

A first HuNoV replication model using B cells was described in 2014 [64]. However, the replication observed is modest and not readily reproducible [65]. Still, it was remarkable to discover that B cells are a target cell of norovirus (further demonstrated in [66]) and that the use of unfiltered feces as inoculum rendered a higher yield of replication, highlighting the role of the gut microbiota in enhancing HuNoV infectivity (see Section 13.2). In 2016,

an ex vivo model using stem cell-derived HIEs was developed [52]. HuNoV was shown to replicate in non-transformed HIE monolayer cultures, which contain multiple cell types such as paneth cells, goblet cells, and enterocytes. Both HuNoV GII.4 and HuNoV GII.3 were able to replicate, although addition of bile to the culture was necessary to achieve HuNoV GII.3 replication [52]. As the use of HIEs in virology becomes more standardized and further optimization is carried out, this system offers an important opportunity to study HuNoV replication and inhibition thereof. In both models, antiviral testing has been attempted using 2'-C-methylcytidine, an experimental broad-spectrum polymerase inhibitor [67-69]. Before these models were available, the only possible approach to study the HuNoV RNA replication were HuNoV GI.1 replicon-bearing cell lines, in which HuNoV GI-I NS proteins are expressed [70, 71]. Since the structural proteins are not expressed in these HuNoV GI.1 replicon cell lines, early (attachment, entry, uncoating) or late events (assembly, genome packaging, exit) of the virus life cycle cannot be studied. However, the study of the activity (and inhibition) of all the replicative enzymes/NS proteins, replication complex formation, and replication of the viral genome are possible. Since models that support HuNoV replication have only been developed/reported recently, most antiviral studies have used MNV as a cultivable alternative. MNV is similar to its human counterparts in terms of the fundamental mechanisms of replication, genetic similarity, and environmental stability [72-74]. MNV has a tropism for macrophages, B cells, and dendritic cells and can be grown routinely in the murine macrophage cell line RAW 264.7 and the murine microglial BV-2 cell line [72].

13.1.5 In Vivo Models

13.1.5.1 Human Norovirus

Multiple larger animal models were used to test HuNoV such as chimpanzees, gnotobiotic pigs, and calves [35, 36, 75]. After intravenous (iv) infection in chimpanzees, no diarrhea was observed, but the infection induced a week-lasting shedding in stool and a serum antibody response. Virus was detected in the intestinal and liver biopsies of the chimpanzees. In gnotobiotic pigs and calves, diarrhea and mild lesions in the intestine were observed after an oral infection with filtered feces from a HuNoV GII.4-infected patient. Such large animal models are valuable but are economically and practically not useful in large-scale antiviral studies.

In 2013, the first mouse model of HuNoV infection was reported, using Rag^{-/-}yc^{-/-} BALB/c mice [76]. Infection was successful after simultaneous intraperitoneal (ip) injection and oral administration of a human stool suspension containing HuNoV; however, the oral route was not sufficient to cause infection [77]. Furthermore, the infected mice did not develop clinical symptoms and the virus was cleared by day three post infection (pi). Viral antigens were detected in the intestines (stomach, ileum, caecum, jejunum, etc.). In 2019, our group established zebrafish larvae (*Danio rerio*) as a new small animal model to study replication of clinically relevant HuNoV strains [78]. After injecting a HuNoV-containing (HuNoV GI or GII) human stool suspension into the yolk of three-day-old zebrafish larvae, the virus replicates to high titers in the intestine and in cells of the hematopoietic lineage. A virus-specific innate immune response was observed after inoculation of the virus and importantly, viral replication could be reduced upon an antiviral treatment. This model

serves as an excellent platform to aid in the search for norovirus antivirals, as zebrafish larvae are small, easy to breed, have a low-cost maintenance, rapid life cycle, and are optically transparent. Moreover, the zebrafish larvae fit in 96- and 384-well plates, compounds can be simply added to their swimming water and very little quantities of compound are needed for testing, in contrast to what is needed for studies in mice [79].

13.1.5.2 The Murine Norovirus as a Surrogate for In Vivo Studies

There are around 30 known strains of MNV, the majority of which cause a persistent asymptomatic infection in immunocompetent mice. When infected with the MNV-1.CW3, innate immune-deficient mice [STAT1^{-/-}, AG129 (*deficient* in the interferon alpha/beta and gamma receptor)] can develop an acute infection with severe diarrhea and weight loss, and almost 100% mortality in four to nine days after infection [80]. In these immune-deficient mice, MNV largely recapitulates the pathogenesis of the HuNoV in humans namely: infection via the oral route, resulting in diarrhea, gastric bloating, and high levels of viral RNA detected in feces. Infection of mice with the MNV-1.CW3 strain allows to study the antiviral effect of small-molecule inhibitors on an acute infection and the virus transmission [81, 82]. Infection of mice with the MNV-CR6 strain results in an asymptomatic persistent infection; it persists asymptomatically in both wild-type and innate immune-deficient mice [83, 84]. The MNV-CR6 strain has been used to study the antiviral effect of small-molecule inhibitors on a persistent infection [85].

13.2 Antiviral Targets

Multiple antiviral targets can be considered in the many steps in the replication cycle of norovirus; these are highlighted in Figure 13.2.

13.2.1 Binding and Entry

Since the binding of HuNoV to HBGAs is the better characterized initial step of viral entry, targeting the HBGAs could be an interesting strategy to develop antivirals that prevent the start of viral replication [86, 87]. By X-ray crystallography and saturation transfer difference nuclear magnetic resonance (STD NMR), citrate was shown to compete with HBGA to bind the norovirus capsid protruding domain [88]. The latter suggests that citrate (and chemicals that mimic certain carbohydrates) could be used as anti-norovirus therapy and that the P2 subdomain of the VP1 capsid protein could be targeted as this is the region that interacts with the HBGAs.

After binding, the viral genome must be released from the capsid and enter the cytoplasm where replication occurs. For HuNoV GII.3, it was shown that bile acid promotes entry into jejunal HIEs [53]. However, more information on entry, release, and pH-dependence of the HuNoV is still lacking. The VP2 could also be considered an antiviral target as it was shown (for FCV) to form a pore in the capsid upon receptor engagement and it was hypothesized that this pore could allow the release of the genome, through the endosomal membrane, into the cytoplasm [59]. When targeting the viral capsid protein, one has to



Figure 13.2 Targeting the various steps of the replication of human norovirus. *Source*: Created with www.biorender.com.

take into account that the capsid region is highly variable and RNA viruses have a high mutation rate with mutations arising mostly within the capsid region. Therefore, the challenges of rapid occurrence of drug-resistant virus variants should be considered. For example, resistance developed rapidly against pleconaril, which is a capsid-binding antiviral for human rhinoviruses (HRV) [89]. Thus, the use of a capsid inhibitor as a single antiviral therapy would not be advised, although such capsid binders may have potential in combination with antiviral molecules targeting other steps in the viral replication.

13.2.2 Nonstructural Proteins

Out of the six NS proteins, the protease and the RdRp are the proteins with a better-characterized function and structure. These are also very similar to the picornavirus homologs; therefore, the knowledge gained from picornaviruses could aid in the search for HuNoV antivirals. Below we describe each NS protein including the protease and RdRp and how they could serve as a potential target for antiviral therapy.

13.2.2.1 NS1/2 (p48)

The NS1/2 has no significant sequence similarity to any viral or cellular proteins, and is the least understood protein of the norovirus genome. The latter could imply that this protein has a role that is unique for noroviruses. Thus, it could be a very interesting target for specific anti-norovirus drug development. The protein consists of two parts: an N-terminal and a C-terminal part. When studying different caliciviruses, the NS1/2 protein has been detected in different organelles, but overall it is shown that this protein encodes for mechanisms to impede normal trafficking within secretory pathways in a way that promotes replication via membrane reorganization [61, 90-93]. The N-terminal part (NS1) is a secretory protein which is required to overcome epithelial host defense barriers mediated by IFN- λ [94]. Moreover, RNAseq analysis showed that the NS1/2 protein can affect the immune system (chemokine, cytokine, and Toll-like receptor signaling) and intracellular pathways (NFκB, MAPK, PI3K-Akt signaling) in murine monocytes [95]. The C-terminal part contains a transmembrane domain and is more conserved within the Caliciviridae family and contains a highly conserved transmembrane domain. This domain potentially acts as an anchor involved in intracellular membrane rearrangements; this conserved region could potentially be targeted [96]. When using the Tulane virus (a rhesus Calicivirus), it was shown that the virus uses Ca^{2+} signaling during infection and that NS1/2 has a viroporin activity that disrupts Ca²⁺ homeostasis [97]. Since viroporins are viral proteins that modify cellular membranes, targeting this protein could result in the inhibition of virus release from infected cells.

13.2.2.2 NS3 (NTPase/Helicase)

The NS3 has high sequence similarities with the NTPase/helicases of picornaviruses (2C protein) and is classified within the superfamily 3 of RNA helicases [98, 99]. The NTPase domain catalyzes the hydrolysis of nucleoside triphosphates, using the released energy (NTP) to unwind the viral nucleic acids (helicase function) [100]. It was recently confirmed that the NS3 of norovirus indeed has both the NTPase and helicase activity [101]. This study also showed that NS3 stimulates the NS7/RdRp-mediated RNA synthesis *in vitro*,

suggesting that NS3 plays an important role in norovirus RNA replication [101]. Moreover, the authors showed that NS3 could be a relevant antiviral target because guanidinium chloride (GuHCl, Figure 13.3a) can inhibit the RNA helicase activity of NS3 *in vitro* and can inhibit the replication of the HuNoV GI.1 replicon [101]. The NS3 of HuNoV GII.4 was detected on different membrane compartments of the secretory pathway and was closely associated to intracellular lipid storage compartments [93].

13.2.2.3 NS4 (p22)

The NS4 has some sequence similarity to the 3A protein in *Picornaviridae*, which inhibits the protein trafficking from the ER to the Golgi. A similar role has been described for the norovirus NS4 [62, 98, 102] that is identified as a key factor in inducing membrane alterations [93]. Targeting this mechanism for the inhibition of the normal ER transport or the formation of the membranous viral replication complex could be an attractive target for future antiviral therapies.

13.2.2.4 NS5 (VPg)

The VPg is linked to the 5' end of the genome and acts as a cap substitute, serving as a primer and recruits translation initiation factors from the host cell [103]. The initiation of antigenomic RNA synthesis by the RdRp is dependent upon uridylylation of VPg. Blocking the uridylylation step or the interaction between the uridylylated VPg and the poly(A) tail of the viral genome could be a good antiviral strategy [104]. Norovirus translation is dependent on the interaction of VPg with the eIF4F complex (eukaryotic initiation factor



Figure 13.3 The chemical structure of (a) Guanidinium chloride, (b) Hippuristanol, (c) Rupintrivir. *Source:* Created with ChemDraw.

4F) [105]. The HuNoV GI.1 VPg interacts with the eIF4E (cap binding protein) [106, 107], while FCV and MNV interact with the complex via another component, namely eIF4A (RNA helicase) [108]. The eIF4A inhibitor, hippuristanol (Figure 13.3b), was able to block the *in vitro* translation of both FCV and MNV [108]. However, this compound has a cytotoxic effect as it can also inhibit cellular protein synthesis. The VPg structures of FCV and MNV are available, revealing that VPg is involved in multiple protein–protein interactions [109]. There is no crystal structure available to study the VPg–RdRp interaction; however, this interaction should be further investigated as a potential antiviral target. While targeting the VPg would likely yield a strong inhibition of norovirus replication due to its critical role, the challenge is to do so by directly disrupting a virus-specific interaction and not by interfering with normal cellular processes that could result in toxic side effects.

13.2.2.5 NS6 (Protease)

The NS6 is a cysteine protease with high structural and sequence similarities with the protease of picornaviruses (3C protein) and coronaviruses [110, 111]. The protease cleaves the NS polyprotein into the single NS proteins. This enzyme is one of the few HuNoV proteins for which the structural information is available, so there is a good basis for in silico drug design [112, 113]. An enzymatic assay to study the HuNoV protease using fluorescence resonance energy transfer (FRET) is also available [114]. It has been put forth that the role of the norovirus protease extends to the prevention of eliciting an immune response upon infection by cleavage of the mitochondrial antiviral-signaling protein (MAVS) [61]. Protease-targeting antivirals that were developed against other RNA viruses have been studied for potential anti-norovirus activity, for example, rupintrivir (AG-7088, Figure 13.3c).

Rupintrivir is a protease inhibitor designed for the treatment of HRV. After successful treatment of volunteers that were experimentally induced with a rhinovirus infection, a second phase II trial was conducted in which rupintrivir was administered intranasally to patients with a natural HRV infection within 36 hours of the appearance of the first symptoms. In the latter study it was shown that rupintrivir was not able to efficiently inhibit natural HRV infections, and further development was stopped. Even the orally available analogue could not reduce disease severity or viral load in naturally infected HRV patients sufficiently [115–117]. However, this compound was shown to have broad-spectrum antiviral activity against, for example, other picornaviruses, coronaviruses, and caliciviruses [118]. Rupintrivir inhibited the HuNoV GI.1 replicon with an EC₅₀ of <1.5 μ M [71, 119].

13.2.2.6 NS7 (RNA-Dependent RNA Polymerase)

The RdRp is a critical enzyme of viral replication. It is responsible for the synthesis of the genomic, subgenomic, and antigenomic norovirus RNA. The RdRp targeting compounds are divided into the nucleoside and non-nucleoside analogues.

13.2.2.6.1 Nucleoside Analogues Nucleoside analogues with antiviral activity (such as the anti-herpes drug acyclovir or the anti-HCV drug sofosbuvir) mimic as their 5'-O-triphosphate metabolite an dATP, dCTP, dGTP, or dTTP when it concerns DNA viruses or either ATP, GTP, CTP, or UTP when it concerns RNA viruses. Nucleoside analogues are

incorporated by the viral polymerases into nascent nucleic acid chains, preventing the incorporation of the next nucleotide, which results in the formation of an incomplete and nonfunctional RNA strand. In addition, they may increase the error frequency of the RdRP. The active site of the RdRp is highly conserved among viral families. This implies that nucleoside analogues are highly likely to have a broad-spectrum activity [31]. Chimerix reported that CMX521 (Figure 13.4a) (a purine nucleoside analogue) inhibits norovirus replication in mice. This is the first norovirus inhibitor that progressed to phase 1 clinical studies. Further development has been halted because of a poor pharmacokinetic profile [120]. A number of other nucleoside analogues that were developed against other (+)ssRNA viruses have been studied for their potential anti-norovirus activity, such as 2'-*C*-methylcytidine (2CMC, Figure 13.4b), 7-deaza-2'-*C*-methyladenosine (7DMA Figure 13.4c), ribavirin (Figure 13.4d), and T-705 (Favipiravir, Figure 13.4e).

2CMC is a cytidine analogue that was initially developed (as its prodrug valopicitabine) as an inhibitor of hepatitis C virus (HCV) replication, but further development was stopped due to gastrointestinal side effects [121, 122]. 2CMC is the most active and most thoroughly studied anti-norovirus molecule *in vitro* and *in vivo* and is therefore considered as the



Figure 13.4 The chemical structure of (a) CMX521, (b) 2⁻C-Methylcytidine, (c) 7-Deaza-2⁻C-methyladenosine, (d) Ribavirin, (e) Favipiravir. (*Source*: Created with ChemDraw.

benchmark compound in antiviral studies against HuNoV [67, 69, 81, 82]. *In vitro*, 2CMC inhibits MNV replication, the HuNoV GI.1 replicon as well as HuNoV GII.4 replication in infected B cells [69, 81] at an EC₅₀ of ~5 μ M or ~0.3 μ M, respectively. 2CMC rescued MNV-1. CW3-infected AG129 mice from virus-induced death upon prophylactic or postexposure treatment. Moreover, viral transmission was blocked when AG129 mice (sentinels) were treated prophylactically [81, 82, 123]. In mice persistently infected with MNV.CR6, 2CMC reduced viral shedding to undetectable levels [85]. 2CMC could also significantly reduce HuNoV GII.4 viral loads in Rag^{-/-}yc^{-/-} BALB/c mice and HuNoV GII.6 viral loads in zebrafish larvae by ~1 log₁₀ or 2.4 log₁₀, respectively [67, 78].

7DMA is an adenosine analogue that was also initially developed as an inhibitor of HCV replication, (it was shown to reduce the HCV loads in infected chimpanzees [124, 125]). 7DMA inhibits MNV replication with an EC_{50} of ~14µM, with no toxic side effect in the RAW cells up to 100µM, the highest concentration tested and inhibits the HuNoV GI.1 replicon with an $EC_{50} \sim 3 \mu M$ [30].

Ribavirin is a guanosine analogue with broad-spectrum activity against both RNA and DNA viruses. It is approved to treat chronic HCV infections (in combination with pegylated interferon) [126], and also to treat respiratory syncytial virus (RSV) infections in young children [127, 128]. It is also active against several other viruses such as Lassa fever virus, hepatitis E virus, and hantaviruses [129–133]. By co-crystallization it was shown, as expected, that the 5'-o-triphosphate of ribavirin binds to the active site of the MNV RdRp [134]. Ribavirin has an inhibitory effect against MNV *in vitro* and inhibits the HuNoV GI.1 replicon with an EC₅₀ of ~40 μ M [135]. However, for norovirus it was shown that ribavirin likely exerts its antiviral effect by depletion of intracellular GTP pools and not via a direct interaction with the RdRp [135, 136].

T-705 is a pyrazine analogue; it needs to be phosphoribosylated intracellularly to become active. The active form is recognized as a substrate by RdRp and inhibits the RNA polymerase activity. Its antiviral effect is attenuated by the addition of purine nucleic acids, indicating the viral RdRp mistakenly recognizes favipiravir as a purine analogue [137]. Favipiravir (T-705) has broad-spectrum activity against multiple RNA viruses such as several flaviviruses, hantaviruses, arenaviruses, and Ebola virus [138, 139]. T-705 is approved in Japan to treat influenza infections [140]. In vitro, T-705 exerts some antiviral activity against MNV and the HuNoV GI.1 replicon, with an EC₅₀ of ~250 μ M and an EC₅₀ of ~21 μ M, respectively [141, 142]. T-705 treatment of MNV-infected mice showed variable efficacy in the reduction of viral loads [85, 143]. T-705 treatment of an immune-compromised patient with a chronic norovirus infection was reported to reduce some clinical symptoms. Sequencing revealed that the patient was infected with different noroviruses of the same monophyletic clade and that upon treatment, there was a selection for a distinct viral variant [27]. Overall, the RdRp is one of the most important NS proteins for many viruses. However, we still lack important information on the progress of norovirus RNA synthesis [144] and compared with picornaviruses, the norovirus RdRp is much understudied. A better understanding of the exact mechanism of action of this viral protein would aid in the development of HuNoV antivirals.

13.2.2.6.2 Non-Nucleoside Analogues Non-nucleoside analogues will inhibit the RdRp by an allosteric mechanism that may result in a more specific antiviral effect as compared

with (certain) nucleoside analogues [145]. There are three known binding sites for nonnucleoside analogues in the norovirus RdRp [146–148]. In silico approaches and RdRp enzyme activity assays allowed for high-throughput screenings to find new non-nucleoside RdRp inhibitors [149]. Recently, compound **54** (Figure 13.5a) was selected as the best candidate in a large in silico screen with an IC₅₀ of 5.6μ M (HuNoV RdRp) and an IC₅₀ of 12.1μ M (MNV RdRp) [150]. Other molecules such as NAF2 (Figure 13.5b), suramin (Figure 13.5c), PPNDS (Figure 13.5d), and NF023 (Figure 13.5e) have also been described as inhibitors of the RdRp [146–148, 151, 152]. However, due to their poor cell permeability, the *in vitro* activity could not be confirmed or the efficacy is much lower in cell culture [146, 149]. For example, suramin needed liposomes to be delivered before it could inhibit MNV *in vitro* [153]. These molecules can be considered as tool compounds which cannot be further developed; however, they can be used as a starting point to find new molecules that can bind the same pockets. To progress in the development of non-nucleoside inhibitors for norovirus, more research is needed to prevent off-target effects, better bioavailability, and cell permeability.

13.2.3 Host Factors

Antivirals may also be directed at host factors that are essential for the viral replication cycle. Targeting host factors could result in higher barrier to resistance, but these could also affect the host and result in adverse effects [154]. An interesting antiviral approach is to stimulate the immune response of the host. Interferons (IFN) play an important role in the antiviral mechanisms and in the innate immune response. Interferon has been used in the past for the treatment of infections with HBV and HCV. A number of studies have shown the impact of interferon type I and II on norovirus infections [72, 80, 135, 155-157]. Also, type III interferon plays an important role in blocking norovirus replication. Interferon λ was shown to protect mice against MNV challenge; therefore, IFN- λ may possibly be explored for the prophylaxis of norovirus infections [158-160]. Toll-like receptor (TLR) 7 agonists that stimulate interferon production were shown to block MNV replication, these include resiquimod (R-848, Figure 13.6a), Vesatolimod (GS-9620, Figure 13.6b), Gardiquimod (Figure 13.6c), and imiquimod (R-837, Figure 13.6d) [161]. Furthermore, R-848 also reduced replication of the HuNoV GI.1 replicon by 50%; however, at the very high concentration of $100 \,\mu M$ [161]. The TLR4 agonist, poly-y-glutamic acid (y-PGA, Figure 13.6e) inhibits MNV replication very efficiently in vitro ($EC_{90} < 100 \text{ nM}$) and in vivo when administered orally (50 mg/kg) [162]. As TLR agonists have been used for many years as vaccine adjuvants, they may thus possibly be repurposed for the treatment of norovirus infections. Another host factor that interacts and plays a role in norovirus replication and can therefore be targeted is the heat shock protein 90 (Hsp90), a chaperone protein that assists in the maturation of multiple proteins [163]. The inhibition of Hsp90 activity by 17-dimethylaminoethylamino-17-demethoxygeldanamycin (17-DMAG, Figure 13.6f) resulted in the inhibition of MNV replication in BV-2 cells and reduced MNV replication in the distal ileum of MNV-1.CW3-infected BALB/c mice by ~1 log₁₀ [163] at a dose of 30 mg/kg. Recently, using a combination of a CRISPR screen and a proteomic analysis of the viral translation and replication complexes, the core stress granule protein G3BP1 was identified as a host factor essential for norovirus infection and may thus be a potential antiviral target [164].



(e)

Figure 13.5 The chemical structure of (a) Compound 54, (b) NAF2, (c) Suramin, (d) PPNDS, (e) NF023. *Source:* Created with ChemDraw.





(a)





(d)

(c)



(e)



Figure 13.6 The chemical structure of (a) Resiquimod (R-848), (b) Vesatolimod (GS-9620), (c) Gardiquimod, (d) Imiquimod (R-837), (e) Poly-γ-glutamic acid (γ-PGA), (f) 17-dimethylaminoethylamino-17-demethoxygeldanamycin (17-DMAG), (g) Nitazoxanide. *Source:* Created with ChemDraw.

13.2.4 Antibodies

Monoclonal antibodies (mAbs) are frequently being used to treat cancer and inflammatory diseases, and have shown (some) promise in the treatment for HIV, HCV, and Ebola virus [165–167]. The application of antibodies as prophylaxis or as therapy for HuNoV may be useful for vulnerable individuals. The use of orally administered human immunoglobulins or specific immunoglobulin (Ig)A purified from sera collected from GI.1-infected participants has also been shown to reduce disease symptoms [168–171]. The latter results have been from limited individual cases and more studies are required to determine if this treatment results in a specific immune response or it merely helps the boosting of the immune system and therefore clearing the virus. Treatment with antibodies retrieved from human rotavirus immunized chicken eggs or immunized bovine colostrum resulted in modest antiviral activity against human rotavirus in young children [172, 173]. mAbs against HuNoV GI.1 from human and chimpanzee B cells were shown to block carbohydrate binding by HuNoV GI.1 virus-like particles (VLPs) in a neutralization assay [174]. Two mAbs (D8 and D7) mixed with HuNoV GI.1 were able to neutralize the virus and therefore prevented infection in chimpanzees [174]. The binding spectrum of the antibodies was limited to the HuNoV GI.1, so the immune response is genotype-specific. The latter points out that much more research is needed and that antibodies need to be able to recognize diverse HuNoV genotypes in order to be used successfully as an antiviral strategy for HuNoV infections. Llama-derived nanobodies are fully functional antibodies that only consist of heavy chains and are able to neutralize HuNoV [175, 176]. Nanobody 85 has shown binding affinity with multiple norovirus VLPs (GI.11, GII.1, GII.2, GII.4, GII.12, GII.17). As nanobody 85 has shown to be less HuNoV genotype-specific than mAbs, these could have a higher potential to be used as an antiviral strategy [177, 178]. However, these studies are performed only using VLPs, structural analysis, and surrogate assays for HuNoV GI.1 neutralization. The recent positive developments in model systems for HuNoV will allow us to better study and understand the use of mAbs and nanobodies as antiviral strategies.

13.2.5 Gut Microbiota

Norovirus, being an enteric virus, naturally encounters another import player in human health and disease: the gut microbiota. Recent studies have shown that norovirus, but also poliovirus, mouse mammary tumor virus (MMTV), rotavirus, and reovirus T3SA+ can bind commensal gut bacteria and use this interaction to their advantage [179]. For example, incubation of poliovirus in the presence of bacterial LPS stabilizes the virus capsid making it more resistant against heat and disinfectants. Moreover, preincubation of poliovirus with LPS resulted in a 200 times more efficient binding to its receptor in HeLa cells [180, 181]. A GII.4 HuNoV-positive stool sample used to infect human B cells resulted in a hundredfold increase in viral genomes after five days, while a much less efficient replication was observed when the fecal sample was first filtered [64]. However, HuNoV replication was restored in a dose-dependent way when the commensal bacteria *Enterobacter cloacae*, which expresses H-type HBGA on its outer surface, was added. The importance of the HBGA-like structures on the bacteria was confirmed when the addition of synthetic H-type HBGA resulted in a similar effect [64]. HuNoV GII.4 is not the only genotype and

E. cloacae not the only bacteria that show these norovirus-bacteria interactions. A strong binding was observed between HuNoV GI.7, HuNoV GII.6, and HuNoV GII.3 VLPs and the Enterobacter sp SENG-6 (closely related to E. cloacae) [182]. Another study showed that HuNoV GII.4 and HuNoV GI.6 could efficiently bind other common gut bacteria such as Enterococcus faecium, Lactobacillus gasseri, Bacteroides thetaiotamicron, and Lactobacillus plantarum, which all express HBGAs [183]. For MNV, a similar relationship with the gut microbiota is observed: an antibiotic pretreatment of two weeks to deplete the gut microbiota of mice prevented the establishment of a persistent infection by MNV.CR6 virus. This effect was reverted when the gut microbiota of the mice was restored with fecal microbiota transplantation (FMT) [184]. In gnotobiotic pigs that received a FMT, with a human fecal sample, an enhanced replication of HuNoV GII.4 was observed [185]. This intricate relationship between norovirus and the gut microbiota could be the next target for new antiviral drugs or therapeutic strategies. For example, preventing the binding between the viral capsid and the HBGA-like structures (although their exact role during HuNoV infection is still unknown) is still an unexplored route to treat or prevent HuNoV infections. Structural analysis suggests that HBGA-blocking IgA antibodies can prevent the attachment of norovirus by sterically masking the HBGA binding site on the P-domain of the virus [186]. Alternatively to steric hindrance, HBGA-blocking antibodies could also work through directly competing with HBGA binding site or by inducing conformational changes in the P-domain and thus allosterically disrupting the binding site [186].

Another option is to cleave the terminal sugar moiety of the glycans using an enzyme. This could be used both against the glycans on the outer surface or on the epithelial cells, depending on which structures are more crucial during HuNoV infection. This concept led to the development of compound DAS181 (fludase, Ansun Biopharma) [187]. This drug, currently in phase 3 clinical trials (NCT03808922), is a sialidase that cleaves sialic acid and so prevents the entry of para-influenza viruses, which use sialic acid as their receptor, into respiratory epithelial cells [188]. A similar drug could be developed for HuNoV, potentially both by targeting fucose [189] and sialic acids [41], which have been found to interact with the HuNoV capsid.

Alternatively, there is evidence that bacteria in the form of probiotics could be used to help fight off norovirus infections. Supplementation of retinoic acid (the metabolite of vitamin A) during MNV-1 infection in mice reduced MNV persistence while a simultaneous upregulation of the levels of *Lactobacillus* sp., a known probiotic, was seen in the gut microbiome [190]. In gnotobiotic pigs, by pre-colonization with a cocktail formulation of *Lactobacillus rhamnosus GG* and *Escherichia coli*, both bacteria that can bind norovirus, viral shedding was completely inhibited during GII.4 infection and stronger antiviral T-cell responses were observed [191]. Similar symptom-reducing effects like reduced viral shedding, diarrhea, or viral replication were also seen in other viral diseases such as rotavirus (supplementation with *Bifidobacterium lactis*) and influenza virus (supplementation with *Lactococcus lactis*) [192].

Furthermore, the gut microbiota also shapes the antiviral immune response by maintaining intestinal epithelial cell integrity [193] and regulating Treg and Th17 cells [194, 195]. Interestingly, the persistent MNV.CR6 infection observed in WT C57BL/6 mice is suggested to be facilitated by commensal bacteria skewing the antiviral IFN- λ response, as gut microbiota-depleted WT C57BL/6 mice were no longer susceptible to a persistent MNV.

CR6 infection. On the other hand, microbiota-depleted $Ifnlr1^{-/-}$ mice, $Stat1^{-/-}$ mice, and Irf3^{-/-} mice (all lacking important genes involved in IFN- λ signaling) were still susceptible to MNV.CR6 [184]. Also, commensal gut bacteria in mice were shown to prevent acute MNV-1 infection in the proximal small intestine but also enhanced infection in the distal parts of the gut (caecum, colon). This prevention of infection in the proximal gut was lost in Ifnlr1^{-/-} mice, showing that type III interferon (interferon class to which IFN- λ belongs) is involved [196]. Interestingly, mice treated with clindamycin to deplete their bile acid composition and mice treated with antibiotics to deplete their gut microbiota both showed a similar increase of viral titers in the proximal small intestine, suggesting that the bile acids metabolized by the gut bacteria are responsible for priming the IFN response in the proximal intestine [196]. This theory was strengthened when after colonizing the antibiotics-treated mice with Clostridium scindens, a known producer of secondary bile acids by 7α -dehydroxylation and oxidation, the viral inhibition was restored [196]. Moreover, IFN- λ was able to cure WT C57BL/6 mice of their persistent MNV.CR6 infection [158] and transmission of MNV-1.CW3 from mice to mice could be prevented by injection $(10 \,\mu g \text{ in the left})$ and right tibialis anterior muscles) of an IFN- λ expressing plasmid in sentinel mice that were co-housed with MNV-infected untreated mice (see also Section 13.3) [159]. There is no evidence yet that a similar skewing is happening during HuNoV infection, but using the newly developed models, this could be elucidated in the near future and allows us to target this process during antiviral treatment.

13.2.6 Unknown Targets

There are compounds that exhibit anti-norovirus activity but for which the mechanism of action is unknown, for example, nitazoxanide (Figure 13.6g) and zinc.

Nitazoxanide is an antiprotozoal drug that has a broad-spectrum activity against parasites (*Giardia*, *Cryptosporidium parvum*), but also against multiple bacteria and viruses, including norovirus and FCV [197, 198]. No antiviral effect in MNV-infected cells has been observed [30, 199]; however, nitazoxanide reduced viral RNA levels in the HuNoV GI.1 replicon system at a concentration of $5\mu g/ml$ [199]. Despite the fact that the exact mechanism of action is still lacking, nitazoxanide is one of the few molecules that has gone to clinical trials for the treatment of HuNoV infection [200]; however, the antiviral effect of nitazoxanide in norovirus-infected patients is controversial [28, 200, 201]. In clinical trials and patient records, significant reduction of the duration of symptoms was noted [200, 201]; however, many cases in which nitazoxanide was unsuccessful have been reported as well [28]. Moreover, in most reports, the focus is merely on the duration of symptoms and no reduction in viral RNA data is stated [202–204]. Although several clinical trials have been set-up to test the use of nitazoxanide for HuNoV infections, there is no consensus on the beneficial results. These contradictory results ask for caution when prescribing nitazoxanide for the treatment of HuNoV infections.

Zinc is a vital mineral that is ubiquitous within cells and has multiple important biological functions. Zinc is normally absorbed in the small intestine from the consumption of various foods, such as animal products, shellfish, seeds, and nuts. The WHO recommends zinc supplementation in young children, as it reduces the duration and severity of diarrhea [205–209]. However, the exact mechanism of action is still unclear [210, 211].

In a study using rat enterocytes from the ileum, it was shown that zinc inhibits cAMPstimulated Cl secretion by selectively inhibiting a cAMP-activated basolateral potassium channel [212]. Because zinc has multiple functions in the human body, it is highly likely that there are many mechanisms of action that all contribute to a better clearance of pathogens. Zinc is not pathogen-specific and it has also shown as to reduce the susceptibility and duration of acute lower respiratory tract infections [205, 213, 214]. Many studies suggest that zinc supplementation during acute diarrhea in young children could have a significant effect; however, also a positive effect on the growth of *Campylobacter jejuni* has been observed upon zinc supplementation [215]. Therefore, this treatment strategy needs to be studied in more detail and appropriate fine-tuning based on the specific causative pathogen is needed.

13.3 Vaccine Development

The approval of a HuNoV vaccine could have a tremendous impact on the large extent of HuNoV outbreaks and HuNoV-related deaths. For example, the introduction of two human rotavirus vaccines, Rotarix (GSK Biologicals) and Rotateq (Merck & CO), resulted in a significant reduction of rotavirus-related childhood deaths. The numbers declined globally from almost 600 000 in 2000 to 215 000 in 2013 [216, 217]. Until today, there is no vaccine on the market to prevent HuNoV infections, but significant efforts have been made and are ongoing. One of the vaccine candidates is an oral vaccine based on a nonreplicating adenovirus vector expressing the HuNoV GI.1 VP1 gene, containing a doublestranded RNA adjuvant (Vaxart). Phase I clinical trials have been successfully completed in which the vaccine showed to be immunogenic with no reports of any safety concerns [218]. Currently, the most advanced vaccine candidate is a GI.1/GII.4 bivalent VLP-based vaccine which is administered intramuscular (Takeda); it has finished phase II of clinical development [219, 220]. However, after the last clinical trial, no significant reduction of illness was reported after a rechallenge with a GII.4 norovirus [221]. Additional studies are ongoing to evaluate the long-term immunogenicity after a single vaccine administration (NCT03039790) and results from a phase IIb field efficacy study in recruits of the US military (NCT02669121) are pending. The vaccine studies are promising, but we still need to understand more about the immune response during and after a HuNoV infection. Information is lacking due to the fact that potential neutralizing antibodies could not be studied until very recently as there were no HuNoV cultivation systems. From volunteer studies in the 1970s, we know that after a norovirus infection a rather shortterm (<five years) immunity develops; however, without cross-protection against other norovirus strains [222-224]. This means that HuNoV infections occur repeatedly during a lifetime.

As seen with the rotavirus vaccines, some factors can condition vaccine effectiveness. First, there is a great diversity in HuNoV genotypes with continuous antigenic evolution and a new pandemic strain emerges every two to three years. Just a few genotypes are covered by the currently studied vaccine candidates [225, 226]. Especially after the emergence of the HuNoV GII.17, it needs to be considered if a GII.4 vaccine would provide sufficient coverage for the future. Second, the immune response upon vaccination of young children

in developing countries can be influenced by malnutrition, immune status, and co-pathogens. Third, the risk of intussusception in young children upon vaccination must be taken into account, although if rotavirus vaccination causes intussusception in young children is still under discussion [227]. The latest clinical trial using the HuNoV GI.1/GII.4 bivalent VLP-based vaccine reported only one child with intussusception upon vaccination (NCT02153112). Taking all of these into consideration, the development of a HuNoV vaccine is very challenging and could likely require annual vaccination [228].

13.4 Conclusion and Perspectives

In this chapter, we reviewed and suggested potential targets and antiviral strategies for HuNoV infections. Despite the fact that it has been more than 50 years since HuNoV was first discovered, no antivirals are available. However, in the last five years, important progress in HuNoV *in vitro* and *in vivo* systems has been made, which will contribute significantly to a better understanding of the HuNoV biology and possible targets for antiviral strategies.

Theoretically, many targets could be explored; however, we believe that the main focus should be toward essential viral replication enzymes, such as the protease and the RdRp. The active sites within these enzymes are conserved across multiple diarrhea-causing viruses, potentially allowing for broad-spectrum antivirals. Targeting viral-specific enzymes will likely also reduce cellular toxicity. Moving toward the development of a pan-diarrheal antiviral, using a combination therapy of antivirals that act on different critical steps in the viral life cycle is likely the way forward.

References

- Kapikian, A.Z., Wyatt, R.G., Dolin, R. et al. (1972). Visualization by immune electron microscopy of a 27-nm particle associated with acute infectious nonbacterial gastroenteritis. *J. Virol.* 10 (5): 1075–1081.
- **2** CDC. About norovirus. http://www.cdc.gov/norovirus/about/overview.html (accessed August 2019).
- **3** Black, R.E., Cousens, S., Johnson, H.L. et al. (2010). Global, regional, and national causes of child mortality in 2008: a systematic analysis. *Lancet* 375 (9730): 1969–1987.
- **4** Liu, L., Johnson, H.L., Cousens, S. et al. (2012). Global, regional, and national causes of child mortality: an updated systematic analysis for 2010 with time trends since 2000. *Lancet* 379 (9832): 2151–2161.
- 5 United Nations Inter-Agency Group for Child Mortality Estimation (UN IGME) (2017). Levels and trends in chilhood mortality - report 2017:40.
- 6 Elliott, E.J. (2007). Acute gastroenteritis in children. BMJ 334 (7583): 35-40.
- 7 Lopman, B.A., Steele, D., Kirkwood, C.D., and Parashar, U.D. (2016). The vast and varied global burden of norovirus: prospects for prevention and control. *PLoS Med.* 13 (4): e1001999.
- **8** Pires, S.M., Fischer-Walker, C.L., Lanata, C.F. et al. (2015). Aetiology-specific estimates of the global and regional incidence and mortality of diarrhoeal diseases commonly transmitted through food. *PLoS One* 10 (12): e0142927.

- **9** Trivedi, T.K., DeSalvo, T., Lee, L. et al. (2012). Hospitalizations and mortality associated with norovirus outbreaks in nursing homes, 2009-2010. *JAMA* 308 (16): 1668–1675.
- 10 Hall, A.J., Curns, A.T., McDonald, L.C. et al. (2012). The roles of Clostridium difficile and norovirus among gastroenteritis-associated deaths in the United States, 1999-2007. *Clin. Infect. Dis.* 55 (2): 216–223.
- **11** Bok, K. and Green, K.Y. (2012). Norovirus gastroenteritis in immunocompromised patients. *N. Engl. J. Med.* 367 (22): 2126–2132.
- 12 Bok, K., Prevots, D.R., Binder, A.M. et al. (2016). Epidemiology of Norovirus Infection Among Immunocompromised Patients at a Tertiary Care Research Hospital, 2010–2013. *Open Forum Infect. Dis.* 3 (3): ofw169.
- 13 Ludwig, A., Adams, O., Laws, H.J. et al. (2008). Quantitative detection of norovirus excretion in pediatric patients with cancer and prolonged gastroenteritis and shedding of norovirus. *J. Med. Virol.* 80 (8): 1461–1467.
- **14** Woodward, J., Gkrania-Klotsas, E., and Kumararatne, D. (2017). Chronic norovirus infection and common variable immunodeficiency. *Clin. Exp. Immunol.* 188 (3): 363–370.
- 15 Angarone, M.P., Sheahan, A., and Kamboj, M. (2016). Norovirus in transplantation. Curr. Infect. Dis. Rep. 18 (6): 17.
- 16 Schwartz, S., Vergoulidou, M., Schreier, E. et al. (2011). Norovirus gastroenteritis causes severe and lethal complications after chemotherapy and hematopoietic stem cell transplantation. *Blood* 117 (22): 5850–5856.
- 17 Atmar, R.L., Opekun, A.R., Gilger, M.A. et al. (2008). Norwalk virus shedding after experimental human infection. *Emerg. Infect. Dis.* 14 (10): 1553–1557.
- **18** Lopman, B., Gastañaduy, P., Park, G.W. et al. (2012). Environmental transmission of norovirus gastroenteritis. *Curr. Opin. Virol.* 2 (1): 96–102.
- **19** Siebenga, J.J., Beersma, M.F., Vennema, H. et al. (2008). High prevalence of prolonged norovirus shedding and illness among hospitalized patients: a model for in vivo molecular evolution. *J. Infect. Dis.* 198 (7): 994–1001.
- **20** Teunis, P.F., Moe, C.L., Liu, P. et al. (2008). Norwalk virus: how infectious is it? *J. Med. Virol.* 80 (8): 1468–1476.
- **21** Vega, E., Barclay, L., Gregoricus, N. et al. (2014). Genotypic and epidemiologic trends of norovirus outbreaks in the United States, 2009 to 2013. *J. Clin. Microbiol.* 52 (1): 147–155.
- **22** Kroneman, A., Verhoef, L., Harris, J. et al. (2008). Analysis of integrated virological and epidemiological reports of norovirus outbreaks collected within the foodborne viruses in Europe network from 1 July 2001 to 30 June 2006. *J. Clin. Microbiol.* 46 (9): 2959–2965.
- **23** Inns, T., Clough, H.E., Harris, J.P. et al. (2019). Estimating the burden of care home gastroenteritis outbreaks in England, 2014-2016. *BMC Infect. Dis.* 19 (1): 12.
- 24 Sandmann, F.G., Jit, M., Robotham, J.V., and Deeny, S.R. (2017). Burden, duration and costs of hospital bed closures due to acute gastroenteritis in England per winter, 2010/11-2015/16. *J. Hosp. Infect.* 97 (1): 79–85.
- 25 Sadique, Z., Lopman, B., Cooper, B.S., and Edmunds, W.J. (2016). Cost-effectiveness of ward closure to control outbreaks of norovirus infection in United Kingdom National Health Service Hospitals. *J. Infect. Dis.* 213 (Suppl 1): S19–S26.
- 26 Woodward, J.M., Gkrania-Klotsas, E., Cordero-Ng, A.Y. et al. (2015). The role of chronic norovirus infection in the enteropathy associated with common variable immunodeficiency. *Am. J. Gastroenterol.* 110 (2): 320–327.

- 334 13 Antiviral Targets and Strategies to Treat and Prevent Human Norovirus Infections
 - 27 Ruis, C., Brown, L.K., Roy, S. et al. (2018). Mutagenesis in norovirus in response to favipiravir treatment. N. Engl. J. Med. 379 (22): 2173–2176.
 - 28 Siddiq, D.M., Koo, H.L., Adachi, J.A., and Viola, G.M. (2011). Norovirus gastroenteritis successfully treated with nitazoxanide. *J. Inf. Secur.* 63 (5): 394–397.
 - Welliver, R., Monto, A.S., Carewicz, O. et al. (2001). Effectiveness of oseltamivir in preventing influenza in household contacts: a randomized controlled trial. *JAMA* 285 (6): 748–754.
 - **30** Van Dycke, J., Arnoldi, F., Papa, G. et al. (2018). A single nucleoside viral polymerase inhibitor against norovirus, rotavirus, and sapovirus-induced diarrhea. *J. Infect. Dis.* 218 (11): 1753–1758.
 - 31 Bassetto, M., Van Dycke, J., Neyts, J. et al. (2019). Targeting the viral polymerase of diarrhea-causing viruses as a strategy to develop a single broad-spectrum antiviral therapy. *Viruses* 11 (2).
 - **32** McFadden, N., Bailey, D., Carrara, G. et al. (2011). Norovirus regulation of the innate immune response and apoptosis occurs via the product of the alternative open reading frame 4. *PLoS Pathog.* 7 (12): e1002413.
 - **33** Chhabra, P., de Graaf, M., Parra, G.I. et al. (2019). Updated classification of norovirus genogroups and genotypes. *J. Gen. Virol.* 10: 1393–1406.
 - **34** Kroneman, A., Vega, E., Vennema, H. et al. (2013). Proposal for a unified norovirus nomenclature and genotyping. *Arch. Virol.* 158 (10): 2059–2068.
 - **35** Cheetham, S., Souza, M., Meulia, T. et al. (2006). Pathogenesis of a genogroup II human norovirus in gnotobiotic pigs. *J. Virol.* 80 (21): 10372–10381.
 - 36 Souza, M., Azevedo, M.S., Jung, K. et al. (2008). Pathogenesis and immune responses in gnotobiotic calves after infection with the genogroup II.4-HS66 strain of human norovirus. *J. Virol.* 82 (4): 1777–1786.
 - 37 Desai, R., Hembree, C.D., Handel, A. et al. (2012). Severe outcomes are associated with genogroup 2 genotype 4 norovirus outbreaks: a systematic literature review. *Clin. Infect. Dis.* 55 (2): 189–193.
 - **38** de Graaf, M., van Beek, J., Vennema, H. et al. (2015). Emergence of a novel GII.17 norovirus End of the GII.4 era? *Euro Surveill*. 20 (26): 21178.
 - **39** Singh, B.K., Koromyslova, A., Hefele, L. et al. (2015). Structural evolution of the emerging 2014-2015 GII.17 noroviruses. *J. Virol.* 90 (5): 2710–2715.
 - **40** Tamura, M., Natori, K., Kobayashi, M. et al. (2004). Genogroup II noroviruses efficiently bind to heparan sulfate proteoglycan associated with the cellular membrane. *J. Virol.* 78 (8): 3817–3826.
 - **41** Rydell, G.E., Nilsson, J., Rodriguez-Diaz, J. et al. (2009). Human noroviruses recognize sialyl Lewis x neoglycoprotein. *Glycobiology* 19 (3): 309–320.
 - **42** Hutson, A.M., Atmar, R.L., Graham, D.Y., and Estes, M.K. (2002). Norwalk virus infection and disease is associated with ABO histo-blood group type. *J. Infect. Dis.* 185 (9): 1335–1337.
 - **43** Taube, S., Perry, J.W., Yetming, K. et al. (2009). Ganglioside-linked terminal sialic acid moieties on murine macrophages function as attachment receptors for murine noroviruses. *J. Virol.* 83 (9): 4092–4101.
 - **44** Tan, M. and Jiang, X. (2005). Norovirus and its histo-blood group antigen receptors: an answer to a historical puzzle. *Trends Microbiol.* 13 (6): 285–293.

- **45** Lindesmith, L., Moe, C., Marionneau, S. et al. (2003). Human susceptibility and resistance to Norwalk virus infection. *Nat. Med.* 9 (5): 548–553.
- **46** Huang, P., Farkas, T., Zhong, W. et al. (2005). Norovirus and histo-blood group antigens: demonstration of a wide spectrum of strain specificities and classification of two major binding groups among multiple binding patterns. *J. Virol.* 79 (11): 6714–6722.
- **47** Shirato, H., Ogawa, S., Ito, H. et al. (2008). Noroviruses distinguish between type 1 and type 2 histo-blood group antigens for binding. *J. Virol.* 82 (21): 10756–10767.
- **48** Almand, E.A., Moore, M.D., and Jaykus, L.A. (2017). Norovirus binding to ligands beyond Histo-blood group antigens. *Front. Microbiol.* 8: 2549.
- **49** Shirato, H. (2011). Norovirus and histo-blood group antigens. *Jpn. J. Infect. Dis.* 64 (2): 95–103.
- **50** Makino, A., Shimojima, M., Miyazawa, T. et al. (2006). Junctional adhesion molecule 1 is a functional receptor for feline calicivirus. *J. Virol.* 80 (9): 4482–4490.
- 51 Farkas, T., Yang, K., Le Pendu, J. et al. (2019). The coxsackievirus and adenovirus receptor, a required host factor for recovirus infection, is a putative enteric calicivirus receptor. *J. Virol.* 93 (22): e00869-19.
- **52** Ettayebi, K., Crawford, S.E., Murakami, K. et al. (2016). Replication of human noroviruses in stem cell-derived human enteroids. *Science* 353 (6306): 1387–1393.
- 53 Murakami, K.T.V., Karandikar, U., Lin, S. et al. (2019). Bile acids and ceramide are critical to allow GII.3 human norovirus entry into human intestinal enteroids. In the 7th International Calicivirus Conference; Sydney 2019. https://doi.org/10.1073/ pnas.1910138117.
- 54 Gerondopoulos, A., Jackson, T., Monaghan, P. et al. (2010). Murine norovirus-1 cell entry is mediated through a non-clathrin-, non-caveolae-, dynamin- and cholesterol-dependent pathway. J. Gen. Virol. 91 (Pt 6): 1428–1438.
- **55** Perry, J.W., Taube, S., and Wobus, C.E. (2009). Murine norovirus-1 entry into permissive macrophages and dendritic cells is pH-independent. *Virus Res.* 143 (1): 125–129.
- 56 Bragazzi Cunha, J. and Wobus, C.E. (2016). Select membrane proteins modulate MNV-1 infection of macrophages and dendritic cells in a cell type-specific manner. *Virus Res.* 222: 64–70.
- 57 Haga, K., Fujimoto, A., Takai-Todaka, R. et al. (2016). Functional receptor molecules CD300lf and CD300ld within the CD300 family enable murine noroviruses to infect cells. *Proc. Natl. Acad. Sci. U. S. A.* 113 (41): E6248–e55.
- **58** Graziano, V., Walker, F.C., Kennedy, E.A. et al. (2019). CD300lf is the primary physiologic receptor of murine norovirus but not human norovirus. *bioRxiv* 16: 859025.
- **59** Conley, M.J., McElwee, M., Azmi, L. et al. (2019). Calicivirus VP2 forms a portal-like assembly following receptor engagement. *Nature* 565 (7739): 377–381.
- **60** Denison, M.R. (2008). Seeking membranes: positive-strand RNA virus replication complexes. *PLoS Biol.* 6 (10): e270.
- **61** Hyde, J.L. and Mackenzie, J.M. (2010). Subcellular localization of the MNV-1 ORF1 proteins and their potential roles in the formation of the MNV-1 replication complex. *Virology* 406 (1): 138–148.
- **62** Sharp, T.M., Guix, S., Katayama, K. et al. (2010). Inhibition of cellular protein secretion by Norwalk virus nonstructural protein p22 requires a mimic of an endoplasmic reticulum export signal. *PLoS One* 5 (10): e13130.

- **63** Santiana, M., Ghosh, S., Ho, B.A. et al. (2018). Vesicle-Cloaked Virus Clusters Are Optimal Units for Inter-organismal Viral Transmission. *Cell Host Microbe* 24 (2): 208–20.e8.
- **64** Jones, M.K., Watanabe, M., Zhu, S. et al. (2014). Enteric bacteria promote human and mouse norovirus infection of B cells. *Science* 346 (6210): 755–759.
- 65 Jones, M.K., Grau, K.R., Costantini, V. et al. (2015). Human norovirus culture in B cells. *Nat. Protoc.* 10 (12): 1939–1947.
- 66 Grau, K.R., Roth, A.N., Zhu, S. et al. (2017). The major targets of acute norovirus infection are immune cells in the gut-associated lymphoid tissue. *Nat. Microbiol.* 2 (12): 1586–1591.
- **67** Kolawole, A.O., Rocha-Pereira, J., Elftman, M.D. et al. (2016). Inhibition of human norovirus by a viral polymerase inhibitor in the B cell culture system and in the mouse model. *Antivir. Res.* 132: 46–49.
- 68 Hosmillo, M., Chaudhry, Y., Nayak, K. et al. (2019). Norovirus replication in human intestinal epithelial cells is restricted by the interferon-induced JAK/STAT signalling pathway and RNA Polymerase II mediated transcriptional responses. *bioRxiv* 11: 731802.
- **69** Rocha-Pereira, J., Jochmans, D., Dallmeier, K. et al. (2012). Inhibition of norovirus replication by the nucleoside analogue 2'-C-methylcytidine. *Biochem. Biophys. Res. Commun.* 427 (4): 796–800.
- **70** Chang, K.O., Sosnovtsev, S.V., Belliot, G. et al. (2006). Stable expression of a Norwalk virus RNA replicon in a human hepatoma cell line. *Virology* 353 (2): 463–473.
- 71 Kitano, M., Hosmillo, M., Emmott, E. et al. (2018). Selection and characterization of rupintrivir-resistant Norwalk virus replicon cells in vitro. *Antimicrob. Agents Chemother*. 62 (5): e00201-18.
- **72** Wobus, C.E., Karst, S.M., Thackray, L.B. et al. (2004). Replication of norovirus in cell culture reveals a tropism for dendritic cells and macrophages. *PLoS Biol.* 2 (12): e432.
- **73** Hirneisen, K.A. and Kniel, K.E. (2013). Comparing human norovirus surrogates: murine norovirus and Tulane virus. *J. Food Prot.* 76 (1): 139–143.
- **74** Bae, J. and Schwab, K.J. (2008). Evaluation of murine norovirus, feline calicivirus, poliovirus, and MS2 as surrogates for human norovirus in a model of viral persistence in surface water and groundwater. *Appl. Environ. Microbiol.* 74 (2): 477–484.
- 75 Bok, K., Parra, G.I., Mitra, T. et al. (2011). Chimpanzees as an animal model for human norovirus infection and vaccine development. *Proc. Natl. Acad. Sci. U. S. A.* 108 (1): 325–330.
- **76** Taube, S., Kolawole, A.O., Hohne, M. et al. (2013). A mouse model for human norovirus. *MBio* 4 (4): e00450–e00413.
- Kolawole, A.O., Gonzalez-Hernandez, M.B., Turula, H. et al. (2016). Oral norovirus infection is blocked in mice lacking Peyer's patches and mature M cells. *J. Virol.* 90 (3): 1499–1506.
- **78** Van Dycke, J., Ny, A., Conceição-Neto, N. et al. (2019). A robust human norovirus replication model in zebrafish larvae. *PLoS Pathog.* 15 (9): e1008009.
- **79** Cully, M. (2019). Zebrafish earn their drug discovery stripes. *Nat. Rev. Drug Discov.* 18 (11): 811–813.
- **80** Karst, S.M., Wobus, C.E., Lay, M. et al. (2003). STAT1-dependent innate immunity to a Norwalk-like virus. *Science* 299 (5612): 1575–1578.
- Rocha-Pereira, J., Jochmans, D., Debing, Y. et al. (2013). The viral polymerase inhibitor 2'-C-methylcytidine inhibits Norwalk virus replication and protects against norovirus-induced diarrhea and mortality in a mouse model. *J. Virol.* 87 (21): 11798–11805.
- Rocha-Pereira, J., Jochmans, D., and Neyts, J. (2015). Prophylactic treatment with the nucleoside analogue 2'-C-methylcytidine completely prevents transmission of norovirus. *J. Antimicrob. Chemother.* 70 (1): 190–197.
- Strong, D.W., Thackray, L.B., Smith, T.J., and Virgin, H.W. (2012). Protruding domain of capsid protein is necessary and sufficient to determine murine norovirus replication and pathogenesis in vivo. *J. Virol.* 86 (6): 2950–2958.
- Nice, T.J., Strong, D.W., McCune, B.T. et al. (2013). A single-amino-acid change in murine norovirus NS1/2 is sufficient for colonic tropism and persistence. *J. Virol.* 87 (1): 327–334.
- Rocha-Pereira, J., Van Dycke, J., and Neyts, J. (2015). Treatment with a nucleoside polymerase inhibitor reduces shedding of murine norovirus in stool to undetectable levels without emergence of drug-resistant variants. *Antimicrob. Agents Chemother.* 60 (3): 1907–1911.
- Ali, E.S., Rajapaksha, H., Carr, J.M., and Petrovsky, N. (2016). Norovirus drug candidates that inhibit viral capsid attachment to human histo-blood group antigens. *Antivir. Res.* 133: 14–22.
- Zhang, X.F., Tan, M., Chhabra, M. et al. (2013). Inhibition of histo-blood group antigen binding as a novel strategy to block norovirus infections. *PLoS One* 8 (7): e69379.
- Hansman, G.S., Shahzad-Ul-Hussan, S., McLellan, J.S. et al. (2012). Structural basis for norovirus inhibition and fucose mimicry by citrate. *J. Virol.* 86 (1): 284–292.
- 89 Ledford, R.M., Collett, M.S., and Pevear, D.C. (2005). Insights into the genetic basis for natural phenotypic resistance of human rhinoviruses to pleconaril. *Antivir. Res.* 68 (3): 135–138.
- Bailey, D., Kaiser, W.J., Hollinshead, M. et al. (2010). Feline calicivirus p32, p39 and p30 proteins localize to the endoplasmic reticulum to initiate replication complex formation. *J. Gen. Virol.* 91 (Pt 3): 739–749.
- **91** Fernandez-Vega, V., Sosnovtsev, S.V., Belliot, G. et al. (2004). Norwalk virus N-terminal nonstructural protein is associated with disassembly of the Golgi complex in transfected cells. *J. Virol.* 78 (9): 4827–4837.
- **92** Ettayebi, K. and Hardy, M.E. (2003). Norwalk virus nonstructural protein p48 forms a complex with the SNARE regulator VAP-A and prevents cell surface expression of vesicular stomatitis virus G protein. *J. Virol.* 77 (21): 11790–11797.
- Doerflinger, S.Y., Cortese, M., Romero-Brey, I. et al. (2017). Membrane alterations induced by nonstructural proteins of human norovirus. *PLoS Pathog.* 13 (10): e1006705.
- Lee, S., Liu, H., Wilen, C.B. et al. (2019). A secreted viral nonstructural protein determines intestinal norovirus pathogenesis. *Cell Host Microbe* 25 (6): 845–57.e5.
- Lateef, Z., Gimenez, G., Baker, E.S., and Ward, V.K. (2017). Transcriptomic analysis of human norovirus NS1-2 protein highlights a multifunctional role in murine monocytes. *BMC Genomics* 18 (1): 39.
- Baker, E.S., Luckner, S.R., Krause, K.L. et al. (2012). Inherent structural disorder and dimerisation of murine norovirus NS1-2 protein. *PLoS One* 7 (2): e30534.
- **97** Strtak, A.C., Perry, J.L., Sharp, M.N. et al. (2019). Recovirus NS1-2 has viroporin activity that induces aberrant cellular calcium signaling to facilitate virus replication. *mSphere* 4 (5): e00506-19.

- **338** 13 Antiviral Targets and Strategies to Treat and Prevent Human Norovirus Infections
 - **98** Hardy, M.E. (2005). Norovirus protein structure and function. *FEMS Microbiol. Lett.* 253 (1): 1–8.
 - **99** Pfister, T. and Wimmer, E. (2001). Polypeptide p41 of a Norwalk-like virus is a nucleic acid-independent nucleoside triphosphatase. *J. Virol.* 75 (4): 1611–1619.
 - **100** Kwong, A.D., Rao, B.G., and Jeang, K.T. (2005). Viral and cellular RNA helicases as antiviral targets. *Nat. Rev. Drug Discov.* 4 (10): 845–853.
 - **101** Li, T.F., Hosmillo, M., Schwanke, H. et al. (2018). Human norovirus NS3 has RNA helicase and chaperoning activities. *J. Virol.* 92 (5): e01606-17.
 - **102** Wessels, E., Duijsings, D., Lanke, K.H. et al. (2006). Effects of picornavirus 3A proteins on protein transport and GBF1-dependent COP-I recruitment. *J. Virol.* 80 (23): 11852–11860.
 - **103** Rohayem, J., Robel, I., Jager, K. et al. (2006). Protein-primed and de novo initiation of RNA synthesis by norovirus 3Dpol. *J. Virol.* 80 (14): 7060–7069.
 - **104** Rohayem, J., Bergmann, M., Gebhardt, J. et al. (2010). Antiviral strategies to control calicivirus infections. *Antivir. Res.* 87 (2): 162–178.
 - **105** Daughenbaugh, K.F., Fraser, C.S., Hershey, J.W., and Hardy, M.E. (2003). The genomelinked protein VPg of the Norwalk virus binds eIF3, suggesting its role in translation initiation complex recruitment. *EMBO J.* 22 (11): 2852–2859.
 - **106** Daughenbaugh, K.F., Wobus, C.E., and Hardy, M.E. (2006). VPg of murine norovirus binds translation initiation factors in infected cells. *Virol. J.* 3: 33.
 - **107** Goodfellow, I., Chaudhry, Y., Gioldasi, I. et al. (2005). Calicivirus translation initiation requires an interaction between VPg and eIF 4 E. *EMBO Rep.* 6 (10): 968–972.
 - **108** Chaudhry, Y., Nayak, A., Bordeleau, M.E. et al. (2006). Caliciviruses differ in their functional requirements for eIF4F components. *J. Biol. Chem.* 281 (35): 25315–25325.
 - 109 Leen, E.N., Kwok, K.Y., Birtley, J.R. et al. (2013). Structures of the compact helical core domains of feline calicivirus and murine norovirus VPg proteins. J. Virol. 87 (10): 5318–5330.
 - **110** Nakamura, K., Someya, Y., Kumasaka, T. et al. (2005). A norovirus protease structure provides insights into active and substrate binding site integrity. *J. Virol.* 79 (21): 13685–13693.
 - **111** Zeitler, C.E., Estes, M.K., and Venkataram Prasad, B.V. (2006). X-ray crystallographic structure of the Norwalk virus protease at 1.5-A resolution. *J. Virol.* 80 (10): 5050–5058.
 - **112** Galasiti Kankanamalage, A.C., Kim, Y., Weerawarna, P.M. et al. (2015). Structure-guided design and optimization of dipeptidyl inhibitors of norovirus 3CL protease. Structure-activity relationships and biochemical, X-ray crystallographic, cell-based, and in vivo studies. *J. Med. Chem.* 58 (7): 3144–3155.
 - **113** Chang, K.O., Kim, Y., Lovell, S. et al. (2019). Antiviral drug discovery: norovirus proteases and development of inhibitors. *Viruses* 11 (2): 197.
 - **114** Chang, K.O., Takahashi, D., Prakash, O., and Kim, Y. (2012). Characterization and inhibition of norovirus proteases of genogroups I and II using a fluorescence resonance energy transfer assay. *Virology* 423 (2): 125–133.
 - 115 Binford, S.L., Maldonado, F., Brothers, M.A. et al. (2005). Conservation of amino acids in human rhinovirus 3C protease correlates with broad-spectrum antiviral activity of rupintrivir, a novel human rhinovirus 3C protease inhibitor. *Antimicrob. Agents Chemother.* 49 (2): 619–626.
 - 116 Patick, A.K., Binford, S.L., Brothers, M.A. et al. (1999). In vitro antiviral activity of AG7088, a potent inhibitor of human rhinovirus 3C protease. *Antimicrob. Agents Chemother.* 43 (10): 2444–2450.

- 117 Patick, A.K. (2006). Rhinovirus chemotherapy. Antivir. Res. 71 (2–3): 391–396.
- **118** Kim, Y., Lovell, S., Tiew, K.C. et al. (2012). Broad-spectrum antivirals against 3C or 3C-like proteases of picornaviruses, noroviruses, and coronaviruses. *J. Virol.* 86 (21): 11754–11762.
- **119** Rocha-Pereira, J., Nascimento, M.S., Ma, Q. et al. (2014). The enterovirus protease inhibitor rupintrivir exerts cross-genotypic anti-norovirus activity and clears cells from the norovirus replicon. *Antimicrob. Agents Chemother.* 58 (8): 4675–4681.
- **120** Lanier RS, Kolawole, A., Hosmillo, M. et al. (2018). CMX521: a Nucleoside with Pan-Genotypic Activity against Norovirus. 31st International Conference on Antiviral Research; Porto, Portugal 2018.
- 121 Afdhal, N., O'Brien, C., Godofsky, E. et al. (2006). Valopicitabine (NM283), alone or with peg-interferon, compared to peg-interferon/ribavirin (pegifn/rbv) retreatment in patients with HCV-1 interferon and prior non-response to pegifn/rbv on-year results. *J. Hepatol.* 46: S5.
- **122** Gardelli, C., Attenni, B., Donghi, M. et al. (2009). Phosphoramidate prodrugs of 2'-Cmethylcytidine for therapy of hepatitis C virus infection. *J. Med. Chem.* 52 (17): 5394–5407.
- **123** Rocha-Pereira, J., Kolawole, A.O., Verbeken, E. et al. (2016). Post-exposure antiviral treatment of norovirus infections effectively protects against diarrhea and reduces virus shedding in the stool in a mortality mouse model. *Antivir. Res.* 132: 76–84.
- **124** Carroll, S.S., Ludmerer, S., Handt, L. et al. (2009). Robust antiviral efficacy upon administration of a nucleoside analog to hepatitis C virus-infected chimpanzees. *Antimicrob. Agents Chemother.* 53 (3): 926–934.
- **125** Olsen, D.B., Davies, M.E., Handt, L. et al. (2011). Sustained viral response in a hepatitis C virus-infected chimpanzee via a combination of direct-acting antiviral agents. *Antimicrob. Agents Chemother.* 55 (2): 937–939.
- **126** Afdhal, N., O'Brien, C., Godofsky, E. et al. (2006). 39 Valopicitabine (NM283), alone or with peginterferon, compared to peg interferon/ribavirin (pegifn/RBV) re-treatment in hepatitis patients with prior non-response to PEGIFN/RBV: week 24 results. *J. Hepatol.* 44: S19.
- **127** listed Na (1986). Ribavirin aerosol approved for severe cases of RSV in infants and young children. *FDA Drug Bull*.
- Hall, C.B., McBride, J.T., Walsh, E.E. et al. (1983). Aerosolized ribavirin treatment of infants with respiratory syncytial viral infection. A randomized double-blind study. *N. Engl. J. Med.* 308 (24): 1443–1447.
- **129** Bausch, D.G., Hadi, C.M., Khan, S.H., and Lertora, J.J. (2010). Review of the literature and proposed guidelines for the use of oral ribavirin as postexposure prophylaxis for Lassa fever. *Clin. Infect. Dis.* 51 (12): 1435–1441.
- **130** McCormick, J.B., King, I.J., Webb, P.A. et al. (1986). Lassa fever. Effective therapy with ribavirin. *N. Engl. J. Med.* 314 (1): 20–26.
- **131** Debing, Y., Emerson, S.U., Wang, Y. et al. (2014). Ribavirin inhibits in vitro hepatitis E virus replication through depletion of cellular GTP pools and is moderately synergistic with alpha interferon. *Antimicrob. Agents Chemother.* 58 (1): 267–273.
- 132 Shrestha, A., Gupta, B.P., and Lama, T.K. (2017). Current treatment of acute and chronic hepatitis E virus infection: role of antivirals. *Euroasian J. Hepato-gastroenterol.* 7 (1): 73–77.

- 340 13 Antiviral Targets and Strategies to Treat and Prevent Human Norovirus Infections
 - **133** Moreli, M.L., Marques-Silva, A.C., Pimentel, V.A., and da Costa, V.G. (2014). Effectiveness of the ribavirin in treatment of hantavirus infections in the Americas and Eurasia: a meta-analysis. *Virus Dis.* 25 (3): 385–389.
 - Alam, I., Lee, J.H., Cho, K.J. et al. (2012). Crystal structures of murine norovirus-1 RNA-dependent RNA polymerase in complex with 2-thiouridine or ribavirin. *Virology* 426 (2): 143–151.
 - **135** Chang, K.O. and George, D.W. (2007). Interferons and ribavirin effectively inhibit Norwalk virus replication in replicon-bearing cells. *J. Virol.* 81 (22): 12111–12118.
 - **136** Leyssen, P., Balzarini, J., De Clercq, E., and Neyts, J. (2005). The predominant mechanism by which ribavirin exerts its antiviral activity in vitro against flaviviruses and paramyxoviruses is mediated by inhibition of IMP dehydrogenase. *J. Virol.* 79 (3): 1943–1947.
 - **137** Furuta, Y., Takahashi, K., Kuno-Maekawa, M. et al. (2005). Mechanism of action of T-705 against influenza virus. *Antimicrob. Agents Chemother.* 49 (3): 981–986.
 - **138** Furuta, Y., Gowen, B.B., Takahashi, K. et al. (2013). Favipiravir (T-705), a novel viral RNA polymerase inhibitor. *Antivir. Res.* 100 (2): 446–454.
 - **139** Oestereich, L., Ludtke, A., Wurr, S. et al. (2014). Successful treatment of advanced Ebola virus infection with T-705 (favipiravir) in a small animal model. *Antivir. Res.* 105: 17–21.
 - **140** Kiso, M., Takahashi, K., Sakai-Tagawa, Y. et al. (2010). T-705 (favipiravir) activity against lethal H5N1 influenza A viruses. *Proc. Natl. Acad. Sci. U. S. A.* 107 (2): 882–887.
 - **141** Rocha-Pereira, J., Jochmans, D., Dallmeier, K. et al. (2012). Favipiravir (T-705) inhibits in vitro norovirus replication. *Biochem. Biophys. Res. Commun.* 424 (4): 777–780.
 - **142** Jin, Z., Tucker, K., Lin, X. et al. (2015). Biochemical evaluation of the inhibition properties of favipiravir and 2'-C-methyl-cytidine triphosphates against human and mouse norovirus RNA polymerases. *Antimicrob. Agents Chemother.* 59 (12): 7504–7516.
 - **143** Arias, A., Thorne, L., and Goodfellow, I. (2014). Favipiravir elicits antiviral mutagenesis during virus replication in vivo. *eLife* 3: e03679.
 - 144 Smertina, E., Urakova, N., Strive, T., and Frese, M. (2019). Calicivirus RNA-dependent RNA polymerases: evolution, structure, protein dynamics, and function. *Front. Microbiol.* 10: 1280.
 - **145** Deval, J., Jin, Z., Chuang, Y.C., and Kao, C.C. (2017). Structure(s), function(s), and inhibition of the RNA-dependent RNA polymerase of noroviruses. *Virus Res.* 234: 21–33.
 - **146** Mastrangelo, E., Pezzullo, M., Tarantino, D. et al. (2012). Structure-based inhibition of norovirus RNA-dependent RNA polymerases. *J. Mol. Biol.* 419 (3–4): 198–210.
 - **147** Tarantino, D., Pezzullo, M., Mastrangelo, E. et al. (2014). Naphthalene-sulfonate inhibitors of human norovirus RNA-dependent RNA-polymerase. *Antivir. Res.* 102: 23–28.
 - **148** Netzler, N.E., Enosi Tuipulotu, D., Eltahla, A.A. et al. (2017). Broad-spectrum nonnucleoside inhibitors for caliciviruses. *Antivir. Res.* 146: 65–75.
 - 149 Eltahla, A.A., Lim, K.L., Eden, J.S. et al. (2014). Nonnucleoside inhibitors of norovirus RNA polymerase: scaffolds for rational drug design. *Antimicrob. Agents Chemother.* 58 (6): 3115–3123.
 - **150** Ferla, S., Netzler, N.E., Ferla, S. et al. (2018). In silico screening for human norovirus antivirals reveals a novel non-nucleoside inhibitor of the viral polymerase. *Sci. Rep.* 8 (1): 4129.
 - 151 Croci, R., Tarantino, D., Milani, M. et al. (2014). PPNDS inhibits murine norovirus RNA-dependent RNA-polymerase mimicking two RNA stacking bases. *FEBS Lett.* 588 (9): 1720–1725.

- 152 Croci, R., Pezzullo, M., Tarantino, D. et al. (2014). Structural bases of norovirus RNA dependent RNA polymerase inhibition by novel suramin-related compounds. *PLoS One* 9 (3): e91765.
- **153** Mastrangelo, E., Mazzitelli, S., Fabbri, J. et al. (2014). Delivery of suramin as an antiviral agent through liposomal systems. *ChemMedChem* 9 (5): 933–939.
- **154** Kaufmann, S.H.E., Dorhoi, A., Hotchkiss, R.S., and Bartenschlager, R. (2018). Hostdirected therapies for bacterial and viral infections. *Nat. Rev. Drug Discov.* 17 (1): 35–56.
- 155 Maloney, N.S., Thackray, L.B., Goel, G. et al. (2012). Essential cell-autonomous role for interferon (IFN) regulatory factor 1 in IFN-gamma-mediated inhibition of norovirus replication in macrophages. J. Virol. 86 (23): 12655–12664.
- **156** Thackray, L.B., Duan, E., Lazear, H.M. et al. (2012). Critical role for interferon regulatory factor 3 (IRF-3) and IRF-7 in type I interferon-mediated control of murine norovirus replication. *J. Virol.* 86 (24): 13515–13523.
- **157** Changotra, H., Jia, Y., Moore, T.N. et al. (2009). Type I and type II interferons inhibit the translation of murine norovirus proteins. *J. Virol.* 83 (11): 5683–5692.
- **158** Nice, T.J., Baldridge, M.T., McCune, B.T. et al. (2015). Interferon-lambda cures persistent murine norovirus infection in the absence of adaptive immunity. *Science* 347 (6219): 269–273.
- **159** Rocha-Pereira, J., Jacobs, S., Noppen, S. et al. (2018). Interferon lambda (IFN-lambda) efficiently blocks norovirus transmission in a mouse model. *Antivir. Res.* 149: 7–15.
- **160** Arthur, S.E., Sorgeloos, F., Hosmillo, M., and Goodfellow, I. (2019). Epigenetic suppression of interferon lambda receptor expression leads to enhanced HuNoV replication in vitro. *bioRxiv* 10: 523282.
- **161** Enosi Tuipulotu, D., Netzler, N.E., Lun, J.H. et al. (2018). TLR7 agonists display potent antiviral effects against norovirus infection via innate stimulation. *Antimicrob. Agents Chemother.* 62 (5): e02417-17.
- **162** Lee, W., Kim, M., Lee, S.H. et al. (2018). Prophylactic efficacy of orally administered Bacillus poly-gamma-glutamic acid, a non-LPS TLR4 ligand, against norovirus infection in mice. *Sci. Rep.* 8 (1): 8667.
- **163** Vashist, S., Urena, L., Gonzalez-Hernandez, M.B. et al. (2015). Molecular chaperone Hsp90 is a therapeutic target for noroviruses. *J. Virol.* 89 (12): 6352–6363.
- **164** Hosmillo, M., Lu, J., McAllaster, M.R. et al. (2019). Noroviruses subvert the core stress granule component G3BP1 to promote viral VPg-dependent translation. *eLife* 8: e46681.
- **165** Qiu, X., Wong, G., Audet, J. et al. (2014). Reversion of advanced Ebola virus disease in nonhuman primates with ZMapp. *Nature* 514 (7520): 47–53.
- 166 Barouch, D.H., Whitney, J.B., Moldt, B. et al. (2013). Therapeutic efficacy of potent neutralizing HIV-1-specific monoclonal antibodies in SHIV-infected rhesus monkeys. *Nature* 503 (7475): 224–228.
- **167** de Jong, Y.P., Dorner, M., Mommersteeg, M.C. et al. (2014). Broadly neutralizing antibodies abrogate established hepatitis C virus infection. *Sci. Transl. Med.* 6 (254): 254ra129.
- **168** Lindesmith, L.C., Beltramello, M., Swanstrom, J. et al. (2015). Serum Immunoglobulin A Cross-Strain Blockade of Human Noroviruses. *Open Forum Infect. Dis.* 2 (3): ofv084.
- **169** Gairard-Dory, A.C., Degot, T., Hirschi, S. et al. (2014). Clinical usefulness of oral immunoglobulins in lung transplant recipients with norovirus gastroenteritis: a case series. *Transplant. Proc.* 46 (10): 3603–3605.

- 342 13 Antiviral Targets and Strategies to Treat and Prevent Human Norovirus Infections
 - **170** Nussbaum, E.Z., Azar, M.M., Cohen, E. et al. (2020). Orally administered human immunoglobulin therapy for norovirus enteritis in solid organ transplant recipients: a case series at a single academic transplant center. *Clin. Infect. Dis.* 71 (8): e206–e209.
 - **171** Florescu, D.F., Hill, L.A., McCartan, M.A., and Grant, W. (2008). Two cases of Norwalk virus enteritis following small bowel transplantation treated with oral human serum immunoglobulin. *Pediatr. Transplant.* 12 (3): 372–375.
 - **172** Sarker, S.A., Casswall, T.H., Juneja, L.R. et al. (2001). Randomized, placebo-controlled, clinical trial of hyperimmunized chicken egg yolk immunoglobulin in children with rotavirus diarrhea. *J. Pediatr. Gastroenterol. Nutr.* 32 (1): 19–25.
 - 173 Sarker, S.A., Casswall, T.H., Mahalanabis, D. et al. (1998). Successful treatment of rotavirus diarrhea in children with immunoglobulin from immunized bovine colostrum. *Pediatr. Infect. Dis. J.* 17 (12): 1149–1154.
 - **174** Chen, Z., Sosnovtsev, S.V., Bok, K. et al. (2013). Development of Norwalk virus-specific monoclonal antibodies with therapeutic potential for the treatment of Norwalk virus gastroenteritis. *J. Virol.* 87 (17): 9547–9557.
 - **175** Garaicoechea, L., Aguilar, A., Parra, G.I. et al. (2015). Llama nanoantibodies with therapeutic potential against human norovirus diarrhea. *PLoS One* 10 (8): e0133665.
 - **176** Ruoff, K., Kilic, T., Devant, J. et al. (2019). Structural basis of nanobodies targeting the prototype norovirus. *J. Virol.* 93 (6): e02005-18.
 - 177 Koromyslova, A.D. and Hansman, G.S. (2017). Nanobodies targeting norovirus capsid reveal functional epitopes and potential mechanisms of neutralization. *PLoS Pathog.* 13 (11): e1006636.
 - **178** Koromyslova, A.D. and Hansman, G.S. (2015). Nanobody binding to a conserved epitope promotes norovirus particle disassembly. *J. Virol.* 89 (5): 2718–2730.
 - **179** Almand, E.A., Moore, M.D., and Jaykus, L.A. (2017). Virus-bacteria interactions: an emerging topic in human infection. *Viruses* 9 (3).
 - 180 Robinson, C.M., Jesudhasan, P.R., and Pfeiffer, J.K. (2014). Bacterial lipopolysaccharide binding enhances virion stability and promotes environmental fitness of an enteric virus. *Cell Host Microbe* 15 (1): 36–46.
 - **181** Kuss, S.K., Best, G.T., Etheredge, C.A. et al. (2011). Intestinal microbiota promote enteric virus replication and systemic pathogenesis. *Science* 334 (6053): 249–252.
 - **182** Miura, T., Sano, D., Suenaga, A. et al. (2013). Histo-blood group antigen-like substances of human enteric bacteria as specific adsorbents for human noroviruses. *J. Virol.* 87 (17): 9441–9451.
 - **183** Almand, E.A., Moore, M.D., Outlaw, J., and Jaykus, L.A. (2017). Human norovirus binding to select bacteria representative of the human gut microbiota. *PLoS One* 12 (3): e0173124.
 - 184 Baldridge, M.T., Nice, T.J., McCune, B.T. et al. (2015). Commensal microbes and interferon-lambda determine persistence of enteric murine norovirus infection. *Science* 347 (6219): 266–269.
 - **185** Lei, S., Twitchell, E.L., Ramesh, A.K. et al. (2019). Enhanced GII.4 human norovirus infection in gnotobiotic pigs transplanted with a human gut microbiota. *J. Gen. Virol.* 100 (11): 1530–1540.
 - **186** Shanker, S., Czako, R., Sapparapu, G. et al. (2016). Structural basis for norovirus neutralization by an HBGA blocking human IgA antibody. *Proc. Natl. Acad. Sci. U. S. A.* 113 (40): E5830–e7.

- 187 Malakhov, M.P., Aschenbrenner, L.M., Smee, D.F. et al. (2006). Sialidase fusion protein as a novel broad-spectrum inhibitor of influenza virus infection. *Antimicrob. Agents Chemother.* 50 (4): 1470–1479.
- **188** Koszalka, P., Tilmanis, D., and Hurt, A.C. (2017). Influenza antivirals currently in late-phase clinical trial. *Influenza Other Respir. Viruses* 11 (3): 240–246.
- **189** Koromyslova, A.D., Leuthold, M.M., Bowler, M.W., and Hansman, G.S. (2015). The sweet quartet: binding of fucose to the norovirus capsid. *Virology* 483: 203–208.
- **190** Lee, H. and Ko, G. (2016). Antiviral effect of vitamin A on norovirus infection via modulation of the gut microbiome. *Sci. Rep.* 6: 25835.
- 191 Lei, S., Ramesh, A., Twitchell, E. et al. (2016). High protective efficacy of probiotics and Rice bran against human norovirus infection and diarrhea in Gnotobiotic pigs. *Front. Microbiol.* 7: 1699.
- **192** Kanauchi, O., Andoh, A., AbuBakar, S., and Yamamoto, N. (2018). Probiotics and paraprobiotics in viral infection: clinical application and effects on the innate and acquired immune systems. *Curr. Pharm. Des.* 24 (6): 710–717.
- 193 Rakoff-Nahoum, S., Paglino, J., Eslami-Varzaneh, F. et al. (2004). Recognition of commensal microflora by toll-like receptors is required for intestinal homeostasis. *Cell* 118 (2): 229–241.
- 194 Gaboriau-Routhiau, V., Rakotobe, S., Lecuyer, E. et al. (2009). The key role of segmented filamentous bacteria in the coordinated maturation of gut helper T cell responses. *Immunity* 31 (4): 677–689.
- **195** Ivanov, I.I., Atarashi, K., Manel, N. et al. (2009). Induction of intestinal Th17 cells by segmented filamentous bacteria. *Cell* 139 (3): 485–498.
- **196** Grau, K.R., Zhu, S., Peterson, S.T. et al. (2019). The intestinal regionalization of acute norovirus infection is regulated by the microbiota via bile acid-mediated priming of type III interferon. *Nat. Microbiol.* 5 (1): 84–92.
- **197** Rossignol, J.F. (2014). Nitazoxanide: a first-in-class broad-spectrum antiviral agent. *Antivir. Res.* 110: 94–103.
- **198** Fumian, T.M., Tuipulotu, D.E., Netzler, N.E. et al. (2018). Potential therapeutic agents for feline calicivirus infection. *Viruses* 10 (8).
- **199** Dang, W., Yin, Y., Peppelenbosch, M.P., and Pan, Q. (2017). Opposing effects of nitazoxanide on murine and human norovirus. *J. Infect. Dis.* 216 (6): 780–782.
- **200** Rossignol, J.F. and El-Gohary, Y.M. (2006). Nitazoxanide in the treatment of viral gastroenteritis: a randomized double-blind placebo-controlled clinical trial. *Aliment. Pharmacol. Ther.* 24 (10): 1423–1430.
- 201 Kempf, B., Edgar, J.D., Mc Caughey, C., and Devlin, L.A. (2017). Nitazoxanide is an ineffective treatment of chronic norovirus in patients with X-linked agammaglobulinemia and may yield false-negative polymerase chain reaction findings in stool specimens. *J. Infect. Dis.* 215 (3): 486–487.
- **202** Mahapatro, S., Mahilary, N., Satapathy, A.K., and Das, R.R. (2017). Nitazoxanide in acute rotavirus diarrhea: a randomized control trial from a developing country. *J. Trop. Med.* 2017: 7942515.
- **203** Rossignol, J.F., Abu-Zekry, M., Hussein, A., and Santoro, M.G. (2006). Effect of nitazoxanide for treatment of severe rotavirus diarrhoea: randomised double-blind placebo-controlled trial. *Lancet* 368 (9530): 124–129.

- 344 13 Antiviral Targets and Strategies to Treat and Prevent Human Norovirus Infections
 - Teran, C.G., Teran-Escalera, C.N., and Villarroel, P. (2009). Nitazoxanide vs. probiotics for the treatment of acute rotavirus diarrhea in children: a randomized, single-blind, controlled trial in Bolivian children. *Int. J. Infect. Dis.* 13 (4): 518–523.
 - Patel, A.B., Dibley, M.J., Mamtani, M. et al. (2010). Influence of zinc supplementation in acute diarrhea differs by the isolated organism. *Int. J. Pediatr.* 2010: 671587.
 - Bhutta, Z.A., Bird, S.M., Black, R.E. et al. (2000). Therapeutic effects of oral zinc in acute and persistent diarrhea in children in developing countries: pooled analysis of randomized controlled trials. *Am. J. Clin. Nutr.* 72 (6): 1516–1522.
 - Bhatnagar, S., Bahl, R., Sharma, P.K. et al. (2004). Zinc with oral rehydration therapy reduces stool output and duration of diarrhea in hospitalized children: a randomized controlled trial. *J. Pediatr. Gastroenterol. Nutr.* 38 (1): 34–40.
 - Sachdev, H.P., Mittal, N.K., Mittal, S.K., and Yadav, H.S. (1988). A controlled trial on utility of oral zinc supplementation in acute dehydrating diarrhea in infants. *J. Pediatr. Gastroenterol. Nutr.* 7 (6): 877–881.
 - Sazawal, S., Black, R.E., Bhan, M.K. et al. (1995). Zinc supplementation in young children with acute diarrhea in India. *N. Engl. J. Med.* 333 (13): 839–844.
 - Berni Canani, R., Buccigrossi, V., and Passariello, A. (2011). Mechanisms of action of zinc in acute diarrhea. *Curr. Opin. Gastroenterol.* 27 (1): 8–12.
 - Hoque, K.M. and Binder, H.J. (2006). Zinc in the treatment of acute diarrhea: current status and assessment. *Gastroenterology* 130 (7): 2201–2205.
 - Hoque, K.M., Rajendran, V.M., and Binder, H.J. (2005). Zinc inhibits cAMP-stimulated Cl secretion via basolateral K-channel blockade in rat ileum. *Am. J. Physiol. Gastrointest. Liver Physiol.* 288 (5): G956–G963.
 - World Health Organization (2017). Zinc supplementation in the management of diarrhoea e-Library of Evidence for Nutrition Actions (eLENA). http://www.who.int/elena/titles/zinc_diarrhoea/en (accessed August 2019).
 - Brown, K.H., Peerson, J.M., Baker, S.K., and Hess, S.Y. (2009). Preventive zinc supplementation among infants, preschoolers, and older prepubertal children. *Food Nutr. Bull.* 30 (1 Suppl): S12–S40.
 - Gielda, L.M. and DiRita, V.J. (2012). Zinc competition among the intestinal microbiota. *MBio* 3 (4): e00171–e00112.
 - Parashar, U.D., Hummelman, E.G., Bresee, J.S. et al. (2003). Global illness and deaths caused by rotavirus disease in children. *Emerg. Infect. Dis.* 9 (5): 565–572.
 - 217 Tate, J.E., Burton, A.H., Boschi-Pinto, C., and Parashar, U.D. (2016). Global, regional, and national estimates of rotavirus mortality in children <5 years of age, 2000-2013. *Clin. Infect. Dis.* 62 (Suppl 2): S96–s105.
 - Kim, L., Liebowitz, D., Lin, K. et al. (2018). Safety and immunogenicity of an oral tablet norovirus vaccine, a phase I randomized, placebo-controlled trial. *JCI Insight* 3 (13): e121077.
 - Treanor, J.J., Atmar, R.L., Frey, S.E. et al. (2014). A novel intramuscular bivalent norovirus virus-like particle vaccine candidate reactogenicity, safety, and immunogenicity in a phase 1 trial in healthy adults. *J. Infect. Dis.* 210 (11): 1763–1771.
 - Leroux-Roels, G., Cramer, J.P., Mendelman, P.M. et al. (2018). Safety and immunogenicity of different formulations of norovirus vaccine candidate in healthy adults: a randomized, controlled, double-blind clinical trial. *J. Infect. Dis.* 217 (4): 597–607.

- Bernstein, D.I., Atmar, R.L., Lyon, G.M. et al. (2015). Norovirus vaccine against experimental human GII.4 virus illness: a challenge study in healthy adults. *J. Infect. Dis.* 211 (6): 870–878.
- Lindesmith, L., Moe, C., Lependu, J. et al. (2005). Cellular and humoral immunity following Snow Mountain virus challenge. *J. Virol.* 79 (5): 2900–2909.
- Simmons, K., Gambhir, M., Leon, J., and Lopman, B. (2013). Duration of immunity to norovirus gastroenteritis. *Emerg. Infect. Dis.* 19 (8): 1260–1267.
- Donaldson, E.F., Lindesmith, L.C., Lobue, A.D., and Baric, R.S. (2010). Viral shapeshifting: norovirus evasion of the human immune system. *Nat. Rev. Microbiol.* 8 (3): 231–241.
- van Beek, J., de Graaf, M., Al-Hello, H. et al. (2018). Molecular surveillance of norovirus, 2005-16: an epidemiological analysis of data collected from the NoroNet network. *Lancet Infect. Dis.* 18 (5): 545–553.
- Tohma, K., Lepore, C.J., Gao, Y. et al. (2019). Population genomics of GII.4 noroviruses reveal complex diversification and new antigenic sites involved in the emergence of pandemic strains. *MBio* 10 (5): e02202-19.
- Clark, A.D., Hasso-Agopsowicz, M., Kraus, M.W. et al. (2019). Update on the global epidemiology of intussusception: a systematic review of incidence rates, age distributions and case-fatality ratios among children aged <5 years, before the introduction of rotavirus vaccination. *Int. J. Epidemiol.*
- Cortes-Penfield, N.W., Ramani, S., Estes, M.K., and Atmar, R.L. (2017). Prospects and challenges in the development of a norovirus vaccine. *Clin. Ther.* 39 (8): 1537–1549.

14

Antiviral Strategies Against (Non-polio) Picornaviruses

Rana Abdelnabi¹, Jim Baggen¹, Carmen Mirabelli², and Johan Neyts¹

¹ Department of Microbiology, Immunology and Transplantation, Rega Institute for Medical Research, Laboratory of Virology and Chemotherapy, KU Leuven, Leuven, Belgium

² Department of Microbiology and Immunology, University of Michigan, Ann Arbor, MI, USA

14.1 Classification and Clinical Impact

Enteroviruses (EV) are non-enveloped positive single-stranded RNA [(+)ssRNA] viruses belonging to the family of the Picornaviridae, and are mainly transmitted by the fecal-oral route [1]. The human EV are subdivided into different species according to molecular classification (Table 14.1). EV of group A (EV-A) include Coxsackievirus group A (CVA) and the main causative agents of hand, foot, and mouth disease CVA6, CVA16, and EV-A71 [2]. Recently, EV-A71 has been reported to cause serious epidemics in Asia that are accompanied by severe neurological complications similar to poliomyelitis [3]. EV-B contains six serotypes of Coxsackievirus B (CVB1-6) viruses as well as echoviruses and can cause various pathologies ranging from mild illness to myocarditis, encephalitis, aseptic meningitis, and pancreatitis [4]. In particular, CVB3 is considered to be a major cause of viral myocarditis that can be fatal for neonates. Echoviruses can also cause severe infections in infants that may lead to sepsis [4]. EV-C includes the nearly eradicated polioviruses (PV) (etiological agent of paralytic poliomyelitis) and Coxsackievirus responsible for hemorrhagic conjunctivitis, such as CV-A24. Lastly, enterovirus D comprises the re-emergent EV-D68, recently responsible for epidemics of respiratory diseases and polio-like paralysis [5]. The rhinoviruses (hRV) A-C also belong to the genus of the human EV. Rhinoviruses are the causative agents of the common cold and are the most important factors in the exacerbations of asthma and chronic obstructive pulmonary disease (COPD) (Table 14.1) [6].

14.2 The Enterovirus Replication Cycle

14.2.1 Virion Structure

EV are non-enveloped viruses with a single-stranded positive-sense RNA genome that is contained within a \sim 30 nm capsid. Although the external features of the capsid can differ

Enterovirus species	Турез	Clinical manifestations (non- exhaustive list)
Enterovirus A	Coxsackievirus A [1-8, 10, 12, 14, 16], Enterovirus A [71, 76, 89, 90, 114, 119-121]	 Hand, foot, and mouth disease (HFMD) Herpangina Encephalitis
Enterovirus B	Coxsackievirus B 1–6, Coxsackievirus A9, Echovirus [1-7, 9, 11-21, 24-27, 29-33], Enterovirus B [6, 69, 73-75, 77–84, 86-88, 93, 97, 98, 100, 101, 106, 107]	Viral myopericarditisAseptic meningitisEncephalitis
Enterovirus C	Poliovirus 1-3 Coxsackievirus A [1, 11, 13, 17, 19-22, 24], Enterovirus C [95, 96, 99, 102, 104, 105, 109, 113, 116-118]	 Paralytic disease Hemorrhagic conjunctivitis (Coxsackievirus A24)
Enterovirus D	Enterovirus D [68, 70, 94, 111]	Acute respiratory infectionsAcute flaccid paralysisHemorrhagic conjunctivitis (EV-D70)
Rhinovirus A Rhinovirus B	~83 types ~32 types	 Common cold Exacerbations of asthma and chronic obstructive pulmonary
Rhinovirus C	~oo types	disease (COPD)

 Table 14.1
 Classification and clinical manifestations of human enteroviruses [4, 7].

Sources: From D. Lugo, P. Krogstad, Enteroviruses in the early 21st century: new manifestations and challenges, *Curr. Opin. Pediatr.* 28 (2016) 107–113 and O.S. Nikonov, E.S. Chernykh, M.B. Garber, E.Y. Nikonova, Enteroviruses: classification, diseases they cause, and approaches to development of antiviral drugs, *Biochemist* 82 (2017) 1615–1631.

between enterovirus types, their virions have overall a similar architecture. The capsid has an icosahedral symmetry with a pseudo T = 3 arrangement and is composed of 60 identical protomers, each protomer consisting of the capsid proteins VP1, VP2, VP3, and VP4 (Figure 14.1a). VP1, VP2, and VP3 constitute the outer surface, whereas the small myristylated protein VP4 resides at the interior of the capsid. The virion has several surfaceexposed loops that differ strongly between serotypes and determine their antigenicity. Most EV have a circular depression around the five-fold symmetry axis that is termed the *canyon* and serves as the binding site for many enterovirus receptors [8]. At the floor of the canyon, there is a hydrophobic cavity (the *pocket*) that contains a lipid molecule, the *pocket factor*, which regulates virion stability.

14.2.2 Genome Structure

The 5'-end of the viral single-stranded RNA genome is covalently linked to the peptide VPg (3B), which serves as a primer for RNA replication. The genome consists of a highly structured 5' untranslated region (UTR), a single open reading frame encoding the viral



(b)

Figure 14.1 Enterovirus structure. (a) Schematic structure of an enterovirus particle, which is composed of 60 protomers that contain each of the surface proteins VP1, VP2, VP3, and the internal protein VP4. Shown are the asymmetric unit (outlined in blue), the canyon (green), and the different symmetry axes (red). (b) Particle types observed during in vitro uncoating. Sedimentation coefficients are shown between brackets. Uncoating cues (receptor binding or low pH) induce structural changes in the native virion, including particle expansion and release of VP4. The resulting intermediate particle is the A-particle (altered particle). Further stimuli induce release of the RNA genome, resulting in the empty particle. *Source*: From [9]. © 2018, Springer Nature.

polyprotein, and a short 3' UTR that is terminated by a poly(A) tail. RNA structures in the 5' UTR form an internal ribosome entry site (IRES), which recruits the translation machinery and is required to initiate polyprotein translation.

14.2.3 Replication Cycle Stages

14.2.3.1 Entry

The life cycle of an enterovirus starts with the attachment to a host cell via receptor binding (Figure 14.2). Some enterovirus serotypes require a single receptor for

infection, whereas other serotypes need to engage multiple receptors [9]. Attachment receptors facilitate binding to the cell surface and may promote virus internalization, whereas uncoating receptors also induce structural changes in the virion that promote genome release. Receptor binding induces endocytic uptake of the virus, after which cellular cues (receptor binding or a low endosomal pH) promote virus uncoating. The uncoating process usually begins with release of the pocket factor, followed by structural changes that result in particle expansion [10], exposure of the VP1 N-terminus [11], and release of VP4. *In vitro*, these events lead to the formation of an uncoating intermediate termed A-particle (altered particle) [12] (Figure 14.1b). Finally, the viral genome exits the virion and enters the cytoplasm via a proteinaceaous transmembrane pore [13].

14.2.3.2 Translation and Replication

After cytoplasmic delivery, the genome is translated into a polyprotein, which is proteolytically processed by viral proteases (2A^{pro}, 3C^{pro}, and 3CD^{pro}) into capsid proteins (VP0, VP1, and VP3), replication proteins (2A-C and 3A-D), and several functional cleavage intermediates. The proteases 2A^{pro} and 3C^{pro} also cleave several host proteins in order to suppress the cellular antiviral responses and to promote viral protein expression via host shut-off, the arrest of transcription and translation of host mRNA. Genome replication takes place at the surface of virus-induced tubulovesicular membrane structures termed replication organelles, which are mainly organized by the viral proteins 3A and 2BC together with several host proteins, such as ACBD3, PI4KB, and OSBP [14-16] (Figure 14.2). These factors enrich the membranes in cholesterol and phosphatidylinositol-4-phosphate lipids, which are required for efficient replication. The viral genome is replicated by the viral RNA-dependent RNA polymerase 3D^{pol}, aided by the viral helicase 2C. First, a negative-strand copy of the incoming positive-strand genome is synthesized, yielding a double-stranded replication intermediate. Subsequently, the negative-strand RNA is replicated to form new positive-strand RNA genomes, which can either serve as templates for additional rounds of translation/replication or may be packaged into new virions [9].

14.2.3.3 Assembly and Release

The encapsidation of newly synthesized enterovirus genomes into progeny virions is a stepwise process that is thought to be closely associated with the replication process [13]. The formation of new virions begins with the self-assembly of protomers (consisting of VP0, VP1, and VP3) into pentamers (Figures 14.1 and 14.2). These pentamers assemble around a nascent viral genome to form an immature particle, the provirion. Finally, the genomic RNA catalyzes the cleavage of VP0 into VP2 and VP4, which yields the mature virus particle [17]. The release of progeny virions from an infected cell was long thought to occur exclusively via cell lysis. However, it was shown more recently that enterovirus spread occurs from cell to cell in a non-lytic manner, via extracellular vesicles that contain multiple virions [18, 19]. These may be formed through the engulfment of virions by double-membrane autophagosomes, which subsequently fuse with the plasma membrane to release single-membrane vesicles (Figure 14.2).



Figure 14.2 The enterovirus life cycle and strategies for its inhibition. After receptor binding, a virion is internalized into the cell via endocytosis and delivers its positive(+)-strand RNA genome across the endosomal membrane into the cytoplasm. The genome is covalently bound to viral protein VPg (3B), which serves as a primer for replication. The genome is translated into a single polyprotein, which is subsequently cleaved into replication proteins (2A-2C and 3A-3D) and capsid proteins (VP0, VP1, and VP3). Genome replication takes place on virus-induced replication organelles, in which a favorable lipid environment is created by viral proteins 2BC and 3A, together with the host factors acyl-CoA-binding domain-containing protein (ACBD3), phosphatidylinositol 4-kinase- β (PI4KB) (which synthesizes phosphatidylinositol-4-phosphate (PI4P)), and oxysterol-binding protein (OSBP) (which recruits cholesterol). The first step of replication is the synthesis of a negative(-)-strand RNA by the viral RNA-dependent RNA polymerase 3Dpol. This strand serves as the template for the synthesis of new (+) strand RNA molecules, which can either enter another round of replication or be incorporated into progeny virions. The capsid proteins can self-assemble into protomers and pentamers, after which the pentamers assemble around a nascent genomic RNA to form the provirion. Finally, the genomic RNA catalyzes the cleavage of VP0 into VP4 and VP2 to yield the mature, infectious virion. Mature virions are released from the host cell via cell lysis or via non-lytic release in extracellular vesicles. Source: From [9]. © 2018, Springer Nature.

14.3 Prevention

For the human EV, approved vaccines are available only for PV and EV-A71 (two EV-A71 vaccines were approved in China) [20]. There are two types of poliovirus vaccines. The inactivated poliovirus vaccine (IPV) was developed by Jonas Salk in the 1950s and it is

currently used in most developed countries because of its safety. The live-attenuated oral polio vaccine (OPV) was developed by Albert Sabin and was first licensed in the United States in 1963 [21]. OPV has been widely used because of its lower cost, higher ability to induce mucosal immunity than IPV, and ease of administration. However, the use of OPV is associated with the risks of vaccine-associated paralytic poliomyelitis (VAPP) and development of mutants with the neurovirulence and transmission characteristics of the wild-type virus. These are either vaccine-derived polioviruses (VDPV) or recombinants in the nonstructural region of the genome with co-circulating members of enterovirus of species-C [21]. In 1988, the WHO launched the Global Polio Eradication Initiative (GPEI). These efforts reduced the number of PV cases by >99% [22]. However, in the last years, wild-type PVs are still circulating in parts of sub-Saharan Africa, the horn of Africa, Pakistan, and Afghanistan, whereas many VDPV strains are reported throughout the central African region [22,23]. Some immunodeficient individuals also excrete PV, including VDPV. Efficient and safe antiviral drugs are needed to clear the excretion of the virus in such individuals. Both EV71 vaccines are formalin-inactivated and based on the C4 genotype virus strain [20].

14.4 Antiviral Strategies Against Enteroviruses

Currently, there is no approved antiviral drug for the treatment or prophylaxis of enterovirus infections. Here, we will review the potential targets for the development of antivirals against EV and provide examples for molecules that act on these targets (Figures 14.3 and 14.4).

14.4.1 Directly Acting Antivirals

14.4.1.1 Early-Stage Inhibitors

The first enterovirus inhibitors in clinical development were the WIN compounds (developed by Sterling Winthrop). These molecules are capsid binders that interact with the hydrophobic pocket underneath the canyon floor, which results in capsid rigidity. As a consequence, these molecules prevent interaction of the virion with its receptor and/or the resulting destabilization of the capsid (A-particle formation) and uncoating [24, 25]. Among these compounds, pleconaril (WIN63843) showed efficient and broad-spectrum activity against EV, Figure 14.3 [26]. Although the compound resulted in some antiviral effects in clinical trials [27, 28], its application as a drug for common cold treatment was rejected by the FDA in 2002 because of safety reasons [29]. Other capsid binders (Figure 14.3), with a similar mechanism of action, include pirodavir (by Janssen Pharmaceutica) [30], BTA-798 (vapendavir, by Biota Pharmaceuticals) [31], ca603 [32], and pocapavir (V-073) [33]. Although BTA-798 successfully passed Phase 1 and the first stage of Phase 2 clinical trials, the compound failed to prove effective *versus* placebo in the second stage of the Phase 2 trial for the treatment of rhinovirus infections in patients with asthma [34].

In addition, two selective poliovirus entry inhibitors were reported (Figure 14.3), i.e. pocapavir (V-073) [33] and H1PVAT [35], with poor or no antiviral activity against other EV. The efficacy of pocapavir has been evaluated in healthy volunteers challenged with the



Figure 14.3 Structural formulae of molecules targeting the entry stages of the enterovirus replication cycle. Source: Based on [26].



Figure 14.4 Structural formulae of molecules targeting the post-entry stages of the enterovirus replication cycle. (a) Molecules targeting the enterovirus 2C protein. (b) Molecules targeting the enterovirus proteases. (c) Molecules targeting the enterovirus RdRp. (d) Molecules targeting host factors.

monovalent oral PV type 1 vaccine [36]. The drug was well tolerated and markedly enhanced the clearance of virus but drug-resistant variants emerged rapidly in the treated volunteers [36]. Unfortunately, most of the developed capsid binders share the same resistance profile where the virus rapidly acquires resistance mutations in the capsid, mainly in the pocket region, that reduce the binding of the compounds. In addition, hRV-C (associated with severe respiratory infections and childhood asthma exacerbations) are completely resistant to these classical capsid binders because the hydrophobic pocket is collapsed in the particle [37].

A class of tryptophan dendrimers (MADAL compounds) has been reported by our team (Figure 14.3), recently. It potently inhibits the entry of different EV-A71 clinical isolates (IC₅₀ values in the nM range) [38]. Unlike classical capsid binders, these compounds bind to the five-fold axis of the viral capsid, thereby preventing attachment of the virus to its cellular receptors PSGL1 and heparan sulfate [38]. Since MADAL compounds have high molecular weights (that may result in low bioavailability *in vivo*), medicinal chemistry efforts are currently ongoing to simplify and reduce the backbone of this series without affecting the antiviral activity [38]. The anti-trypanosomiasis drug suramin was reported as a selective inhibitor of EV-A entry and attachment through interaction with the positively charged region surrounding the five-fold axis of the capsid [39]. Interestingly there is no cross-resistance between suramin and MADAL compounds [38].

Recently, we identified a new druggable pocket in the entero-/rhinovirus particle [40]. The newly discovered pocket is formed by viral proteins VP1 and VP3 and appears to be present in most, if not all entero- and rhinoviruses, including the pleconaril-resistant RV-C species [40]. The compounds targeting this pocket (i.e. compound 17 and its analogues, Figure 14.3) exert activity against CVBs (IC₅₀ of 0.7μ M against CVB3) and other EV of groups C and D and even against rhinoviruses A and B [40]. Interestingly, these molecules are not cross-resistant with the classical capsid binders. Binding of the inhibitors to the pocket is believed to stabilize the conformation of a key region of the virion and hence to prevent rearrangements that allow the transition to the A-particle. Therefore, this new pocket can be considered as a promising target for the design of broad-spectrum antivirals against different EV [40].

14.4.1.2 Viral Protease Inhibitors

The protease activity of EV is encoded by the 2A and 3C proteins. Both proteins are chymotrypsin-like cysteine proteases and are necessary for proteolytic cleavage of the viral polyprotein into mature and functional proteins [41]. The cleavage site for the 2A protease is located at the VP1-2A junction whereas the 3C protease cleaves the viral polyprotein at all the other junctions [41].

The 3C protease is an attractive target for drug development due to its highly conserved catalytic domain among EV and its limited similarity to host cell proteases [42]. The most potent and extensively studied 3C inhibitor is rupintrivir (Figure 14.4).

Rupintrivir is a Michael-acceptor peptidomimetic that binds irreversibly to the 3C protease leading to formation of a stable covalent complex [43]. The compound exerts potent and broad-spectrum activity against EV and was able to reduce symptoms and viral load in rhinovirus human challenge trials [44, 45]. However, the molecule failed to result in a significant reduction in disease severity and viral load in naturally infected patients [46].

AG7404 (also named compound 1, Figure 14.4) is a rupintrivir analogue with improved oral bioavailability [47]. AG7404 elicits potent antiviral activity against rhino- and enteroviruses (mean IC₅₀ values of 50 and 75 nm, respectively) and was also well tolerated in a phase 1-ascending, single-dose study but was not further developed because of marginal pharmacokinetics [46]. A novel 3C inhibitor was reported (SG85) with broad-spectrum anti-rhinovirus activity (mean IC₅₀ of $0.04 \,\mu$ M) that is not cross-resistant with rupintrivir, Figure 14.4 [48, 49]. Structural modeling and the mutations detected in the drug-resistant variants suggested that a hydrogen bond is formed between the backbone amide of the P2 residue of the inhibitor and the side chain oxygen of the serine residue at position 127 of the RHV14 3C protease [49]. In addition, the natural flavonoid quercetin (Figure 14.4) has been reported to inhibit the *in vitro* replication of EV-A71 (mean IC₅₀ value of 10 μ M) by blocking the purified EV-A71 3C protease [50]. Molecular modeling suggests that the compound blocks the substrate-binding pocket of the enzyme [50].

The HCV protease inhibitor telaprevir (Figure 14.4) was recently shown to inhibit EV-D68 replication in RD cells with an IC_{50} below 1 μ M [41]. The molecule was found to be an irreversible inhibitor of the EV-D68 2A protease [41]. Docking of telaprevir in the EV-D68 2Apro model revealed that the α -ketoamide of telaprevir is in close proximity to the catalytic C107 residue of the enzyme, suggesting a covalent complex formation. These results also highlight the importance of the enterovirus 2A protease as a potential drug target and the promise for potentially repurposing of HCV protease inhibitors.

14.4.1.3 2C Targeting Molecules

The enterovirus 2C protein is a multifunctional protein, which is highly conserved among EV. 2C possesses ATPase, GTPase, and helicase activity [51] and has also been reported to be involved in RNA binding, in the interaction with several viral and host proteins, morphogenesis, and uncoating [42]. Guanidine hydrochloride (GuaHCl, Figure 14.4) is a 2C targeting compound with antiviral activity against several EV including PV and Coxsackieviruses [52]. The IC50 of Guanidine HCL is 320-390 µM against Polio and ~2 mM against Coxsackieviruses. GuaHCl inhibits the 2C ATPase activity and its interaction with host cell membranes [53]. The compound also inhibits the initiation of negative-strand RNA synthesis during the poliovirus replication cycle [54]. Other reported 2C-targeting molecules include HBB, MRL-1237 [55], and TBZE-029 [56], which share a benzimidazole scaffold, Figure 14.4a. However, TBZE-029 does not inhibit the 2C ATPase activity [56]. The FDA-approved drugs dibucaine (local anesthetic), pirlindole (antidepressant), and zuclopenthixol (antidepressant) (Figure 14.4a) were identified as in vitro CVB3 replication inhibitors [57]. The molecules also elicit antiviral activity against other EV and proved cross-resistant with TBZE-029 and GuHCl, suggesting that 2C is the potential target for these compounds [57]. Similarly, the serotonin reuptake inhibitor fluoxetine (Prozac, Figure 14.4) inhibits enterovirus replication in vitro (IC₅₀ values $0.4-8\,\mu$ M) and has been proposed to target the viral 2C protein as well [58].

14.4.1.4 RNA-dependent RNA Polymerase (RdRp) "(3D) Inhibitors"

3D polymerase inhibitors can be classified into nucleoside and non-nucleoside analogues. The nucleoside analogues bind to the active site of the polymerase causing chain termination and/or increasing the error rate when incorporated into the growing RNA chain. On the other hand, non-nucleoside analogues bind to allosteric sites of the enzyme, which

inhibit conformational changes necessary for the efficiency of the polymerase catalytic activity.

The polymerase targeting drug favipiravir (Figure 14.4c) exerts, although weak, activity against some EV such as CVB3 [59] and EV-A71 [60]. Favipiravir-resistant EV-A71 variants were selected that carry a serine to asparagine mutation at position 121 in the fingers subdomain of the viral RdRp [60]. In addition, the conserved lysine residue (at position 159) in the F1 RdRp motif (which has also been reported to be linked to resistance of the chikungunya virus to favipiravir) was shown to be involved in the antiviral activity of favipiravir against CVB3 [59]. Favipiravir appears also to increase the mutation rates during CVB3 replication [59]. Amiloride (anti-hypertensive drug) and its derivatives were shown to inhibit the replication of rhinovirus 2 (IC₅₀ values 7–120 μ M) and CVB3 (IC₅₀ values 2–60 μ M) [61] by targeting the viral polymerase and by exerting a mutagenic effect on viral RNA replication [62]. GPC-N114 (Figure 14.4c) is a non-nucleoside inhibitor that exerts broad-spectrum activity against EV (IC₅₀ values 0.1–1.7 μ M) by targeting the RNA-binding channel of the viral polymerase [63]. The broad-spectrum activity of GPC-N114 makes it a promising RdRp inhibitor for EV.

14.4.2 Host-targeting Antivirals

14.4.2.1 Inhibitors of Lipid Processing

During enterovirus replication, the viral 3A protein recruits phospatidylinositol-4-kinase III β (PI4KIII β) to secretory organelle membranes, which generates a phosphatidylinositol 4-phosphate (PI4P) lipids-enriched membrane microenvironment (Figure 14.2). Consequently, PI4P recruits the cellular oxysterol binding protein (OSBP) into these organelles that results in a cholesterol-rich environment which is favorable for viral RNA replication [15]. Molecules that target the PI4KIII β -PI4P-OSBP pathway were shown to inhibit enterovirus replication. These molecules include PI4KIII β inhibitors (Figure 14.4d) such as enviroxime [64], GW5074 (IC₅₀ values 2–6 μ M) [65] and T-00127-HEV1 (IC₅₀ of 0.8 μ M against PV pseudovirus) [66] as well as the OSBP inhibitors (Figure 14.4d), e.g. OSW-1 (IC₅₀ values 2.4–9.4 nM) [67], itraconazole (IC₅₀ values 0.3–1.6 μ M) [68], and TTP-8307 (IC₅₀ values 0.27–5 μ M) [69].

14.4.2.2 Assembly Inhibitors

Glutathione (GSH) is an essential host factor for stabilization of EV during viral morphogenesis [70, 71]. Molecules that rapidly deplete the intracellular GSH levels such as TP219 (inhibits CVA16 and CVB3 replication with IC₅₀ values of 2.7 and 7 μ M, respectively) [70] (Figure 14.4d), or which block the GSH biosynthesis, e.g. buthionine sulphoximine (BSO, Figure 14.4d), can inhibit enterovirus replication [71, 72]. Intracellular vesicular transport is involved in several processes during the viral replication cycle including virus assembly and release. Retro-2^{cycl} (Figure 14.4d) and Retro-2.1 are modulators of vesicular trafficking through intracellular redistribution of Syntaxins, which is a family of proteins involved in intracellular vesicle transport [73]. Both compounds inhibited the *in vitro* replication of EV-A71 by blocking the vesicle transport steps required for progeny virus release (IC₅₀ values of 12.5 and 0.05 μ M, respectively) [73]. Dosing of Retro-2^{cycl} (10 mg/kg) to EV-A71infected mice resulted in 90% protection from a lethal challenge [73]. Geldanamycin

(Figure 14.4d) is an inhibitor of HSP-90 (heat shock protein 90) that was shown to interfere with the correct folding and maturation of enterovirus capsid proteins [74]. The compound inhibited poliovirus replication in HeLa cells with an IC_{50} of 0.11 μ M [74].

14.4.3 Monoclonal Antibodies

Two monoclonal antibodies (mAbs) against EV-A71 (named A9 and D6) were obtained by immunizing mice with a mixture of mature and pro-mature virus particles. Both antibodies potently inhibit EV-A71 replication in cell culture [75]. Mechanistic studies and cryo-EM structures of these mAbs bound with EV-A71 revealed that A9 and D6 block the interaction between the virus and its cellular receptor SCARB2 [75]. In addition, these mAbs, especially A9, resulted in destabilization of the viral capsid structure [75]. In another study, a bispecific mAb against both EV-A71 and CVA16 (named Bs(scFv)4-IgG-1) was developed and was shown to result in 100% protection against lethal disease in mice infected with either EV-A71, CVA16, or both when dosed (10 or $3 \mu g/g$) at 24 hours postinfection [76].

14.4.4 Alternative Strategies

14.4.4.1 Direct Targeting of the RNA Genome

RNA viruses, such as enterovirus, are purely replicating in the cytoplasm. Blockage of their viral cycle via genome degradation could in theory constitute a strategy to block infection at early stages. Two main approaches have been or are being explored to that end: RNA interference (RNAi) and CRISPR-Cas13. Whereas the RNAi strategy exploits a physiological mammalian response and cellular machinery (via Dicer/RISC), the CRISPR-Cas13 requires adaptation of a physiological bacterial response to the mammalian host. siRNA approaches have been developed for several members of the enterovirus genus, mostly EV-A71, EV-D68, and CV-B3 [77–79]. On the other hand, a proof-of-concept study for the CRISPR-Cas13 strategy has been published in the context of other RNA virus infection but not yet for EV [80]. The main limitations of these strategies are the need for pan-enterovirus activity as well as the issue of delivery. Both siRNA and CRISPR strategies rely on RNA sequences or guides that are complementary to the target. Given the molecular diversity of enterovirus, the design of pan-enterovirus RNA sequences/guides that will provide a good coverage is challenging. In addition, these strategies need to be coupled with innovative delivery system to limit off-target effects or systemic toxicity.

14.4.4.2 Dual Target Therapeutics

Although enteroviruses are mostly known as viruses inducing acute infections, there is an increasing body of evidence that associates enterovirus with chronic inflammatory syndromes. Rhinovirus association with asthma and COPD is probably the most relevant example. The development of therapeutic approaches that simultaneously relieve the symptoms of the chronic syndromes being also active against the viral infection would be the most ideal scenario [6]. Interferon signaling could be an important target to alleviate symptoms but also to prime the cells and subsequently prevent viral infection. As an example, azithromycin is a macrolide antibiotic used for the treatment of COPD since it shows a

reduction in exacerbation frequency together with an improvement in quality of life of patients with COPD. Azithromycin transiently induced an increase of IFN β , IFN λ , and RIG-I expression in bronchial epithelial cells that had been infected with rhinovirus. As a consequence, viral infection and release were reduced [81].

14.5 Conclusions

Recent re-emergence of some non-polio EV and the severe complications associated with their infections indicate the urgent need for potent antivirals against these viruses. Despite the diversity of EV, the development of potent and broad-spectrum rhino-/enterovirus inhibitors should be very well feasible. Potent directly acting antivirals (DAA) are available for the treatment of infections with herpesviruses, HIV, HBV, HCV, and influenza viruses [82]. Most of these drugs target the viral polymerases and/or reverse transcriptase and viral proteases, but also factors such as the HCV NS5A protein, the HIV-co-receptors, and the influenza neuraminidase. Recently, a drug targeting a viral terminase has been approved for treatment of the human cytomegalovirus [83] and capsid binders [84] are presently intensively being researched for hepatitis B [84]. Excellent targets for the development of (broad-spectrum) rhino-/enterovirus inhibitors include the viral 3C, and possibly also the 2A, protease, the 2C ATPase/helicase, and the 3D RdRp. Capsid binders have since long been studied. For various reasons (which may also include the rapid development of resistance to such inhibitors and the fact that hRV-C have a collapsed pocket and is thus not sensitive to capsid binders) none has yet been approved. Recent discoveries in rhino- and enteroviruses biology may also provide new potential targets for drug discovery. For example, the recently discovered VP1-VP3 interprotomer druggable pocket in the capsid of rhino-/enteroviruses [40]. Finally, an important aspect to be studied is whether combination therapy of potent entero-/rhinovirus inhibitors may be needed to avoid/prevent the development of drug-resistance, a strategy which was very successful for HIV and HCV.

References

- **1** Muehlenbachs, A., Bhatnagar, J., and Zaki, S.R. (2015). Tissue tropism, pathology and pathogenesis of enterovirus infection. *J. Pathol.* 235: 217–228. https://doi.org/10.1002/path.4438.
- **2** Bian, L., Wang, Y., Yao, X. et al. (2015). Coxsackievirus A6: a new emerging pathogen causing hand, foot and mouth disease outbreaks worldwide. *Expert Rev. Anti-Infect. Ther.* 13: 1061–1071. https://doi.org/10.1586/14787210.2015.1058156.
- **3** Chong, P., Liu, C.C., Chow, Y.H. et al. (2015). Review of enterovirus 71 vaccines. *Clin. Infect. Dis.* 60: 797–803. https://doi.org/10.1093/cid/ciu852.
- **4** Lugo, D. and Krogstad, P. (2016). Enteroviruses in the early 21st century: new manifestations and challenges. *Curr. Opin. Pediatr.* 28: 107–113. https://doi.org/10.1097/MOP.00000000000303.
- **5** Maisels, C.K. (2016). The emergence of enterovirus-D68. *Microbiol. Spectr.*: 46–64. https://doi.org/10.1128/microbiolspec.EI10-0018-2016.Correspondence.

- **6** Mirabelli, C., Scheers, E., and Neyts, J. (2017). Novel therapeutic approaches to simultaneously target rhinovirus infection and asthma/COPD pathogenesis. *F1000 Res.* 6: 1860. https://doi.org/10.12688/f1000research.11978.1.
- 7 Nikonov, O.S., Chernykh, E.S., Garber, M.B., and Nikonova, E.Y. (2017). Enteroviruses: classification, diseases they cause, and approaches to development of antiviral drugs. *Biochemist* 82: 1615–1631. https://doi.org/10.1134/s0006297917130041.
- 8 Rossmann, M.G., He, Y., and Kuhn, R.J. (2002). Picornavirus-receptor interactions. *Trends Microbiol.* 10: 324–331. https://doi.org/10.1016/S0966-842X(02)02383-1.
- **9** Baggen, J., Thibaut, H.J., and Van Kuppeveld, F.J.M. (2018). The life cycle of non-polio enteroviruses and how to target it. *Nat. Rev. Microbiol.* https://doi.org/10.1038/ s41579-018-0005-4.
- **10** Belnap, D.M., Filman, D.J., Trus, B.L. et al. (2000). Molecular tectonic model of virus structural transitions: the putative cell entry states of poliovirus. *J. Virol.* 74: 1342–1354. https://doi.org/10.1128/JVI.74.3.1342-1354.2000.
- Fricks, C.E. and Hogle, J.M. (1990). Cell-induced conformational change in poliovirus: externalization of the amino terminus of VP1 is responsible for liposome binding. *J. Virol.* 64: 1934–1945.
- 12 Crowell, R.L. and Philipson, L. (1971). Specific alterations of coxsackievirus B3 eluted from HeLa cells. J. Virol. 8: 509–515.
- Nugent, C.I., Johnson, K.L., Sarnow, P., and Kirkegaard, K. (1999). Functional coupling between replication and packaging of poliovirus replicon RNA. *J. Virol.* 73: 427–435. https://doi.org/10.1128/JVI.03209-12.
- **14** Lyoo, H., van der Schaar, H.M., Dorobantu, C.M. et al. (2019). ACBD3 is an essential pan-enterovirus host factor that mediates the interaction between viral 3A protein and cellular protein PI4KB. *MBio* 10: 1–15. https://doi.org/10.1128/mBio.02742-18.
- 15 Hsu, N.Y., Ilnytska, O., Belov, G. et al. (2010). Viral reorganization of the secretory pathway generates distinct organelles for RNA replication. *Cell* 141: 799–811. https://doi. org/10.1016/j.cell.2010.03.050.
- 16 Roulin, P.S., Lötzerich, M., Torta, F. et al. (2014). Rhinovirus uses a phosphatidylinositol 4-phosphate/cholesterol counter-current for the formation of replication compartments at the ER-Golgi interface. *Cell Host Microbe* 16: 677–690. https://doi.org/10.1016/j. chom.2014.10.003.
- 17 Basavappa, R., Syed, R., Flore, O. et al. (1994). Role and mechanism of the maturation cleavage of VP0 in poliovirus assembly: structure of the empty capsid assembly intermediate at 2.9 A resolution. *Protein Sci.* 3: 1651–1669. https://doi.org/10.1002/ pro.5560031005.
- 18 Robinson, S.M., Tsueng, G., Sin, J. et al. (2014). Exits the host cell in shed microvesicles displaying autophagosomal markers. *PLoS Pathog.* 10: 1–21. https://doi.org/10.1371/ journal.ppat.1004045.
- **19** Chen, Y.H., Du, W., Hagemeijer, M.C. et al. (2015). Phosphatidylserine vesicles enable efficient en bloc transmission of enteroviruses. *Cell* 160: 619–630. https://doi.org/10.1016/j. cell.2015.01.032.
- 20 Reed, Z. and Cardosa, M.J. (2016). Status of research and development of vaccines for enterovirus 71. *Vaccine* 34: 2967–2970. https://doi.org/10.1016/j.vaccine.2016.02.077.

- **21** Orenstein, W.A., T.C.O.I. Committee on Infectious Diseases, Sutter, R. et al. (2015). Eradicating polio: how the world's pediatricians can help stop this crippling illness forever. *Pediatrics* 135: 196–202. https://doi.org/10.1542/peds.2014-3163.
- Moturi, E.K., Porter, K.A., Wassilak, S.G.F. et al. (2014). Progress toward polio eradication worldwide, 2013–2014. *MMWR Morb. Mortal. Wkly Rep.* 63: 468–472. https://doi.org/10.1093/infdis/jit808.
- 23 Eichner, M. and Brockmann, S.O. (2013). Polio emergence in Syria and Israel endangers Europe. *Lancet* 382: 1777. https://doi.org/10.1016/S0140-6736(13)62220-5.
- Rossmann, M.G. (1989). The structure of antiviral agents that inhibit uncoating when complexed with viral capsides. *Antivir. Res.* 11: 3–13. https://doi.org/10.1016/0166-3542(89)90016-8.
- 25 Pevear, D.C., Fancher, M.J., Felock, P.J. et al. (1989). Conformational change in the floor of the human rhinovirus canyon blocks absorption to {H}e{L}a cell receptors. *J. Virol.* 63: 2002–2007.
- **26** Pevear, D.C., Tull, T.M., Seipel, M.E., and Groarke, J.M. (1999). Activity of pleconaril against enteroviruses. *Antimicrob. Agents Chemother.* 43: 2109–2115.
- 27 Rotbart, H.A. and Webster, A.D. (2001). Treatment of potentially life-threatening enterovirus infections with pleconaril. *Clin. Infect. Dis.* 32: 228–235. https://doi.org/10.1007/s11908-001-0073-9.
- 28 Hayden, F.G., Herrington, D.T., Coats, T.L. et al. (2003). Efficacy and safety of oral pleconaril for treatment of colds due to picornaviruses in adults: results of 2 double-blind, randomized, placebo-controlled trials. *Clin. Infect. Dis.* 36: 1523–1532. https://doi.org/10.1086/375069.
- 29 Senior, K. (2002). Newsdesk FDA panel rejects common cold treatment. *Lancet Infect. Dis.* 2: 264. https://doi.org/10.1016/S1473-3099(02)00277-3.
- Andries, K., Dewindt, B., Snoeks, J. et al. (1992). In vitro activity of pirodavir (R 77975), a substituted phenoxy-pyridazinamine with broad-spectrum antipicornaviral activity. *Antimicrob. Agents Chemother.* 36: 100–107. https://doi.org/10.1128/AAC.36.1.100.
- **31** Watson, K.G., Brown, R.N., Cameron, R. et al. (2003). An orally bioavailable oxime ether capsid binder with potent activity against human rhinovirus. *J. Med. Chem.* 46: 3181–3184. https://doi.org/10.1021/jm0202876.
- **32** Lacroix, C., Laconi, S., Angius, F. et al. (2015). In vitro characterisation of a pleconaril/ pirodavir-like compound with potent activity against rhinoviruses. *Virol. J.* 12: 106. https:// doi.org/10.1186/s12985-015-0330-4.
- 33 Oberste, M.S., Moore, D., Anderson, B. et al. (2009). In vitro antiviral activity of V-073 against polioviruses. *Antimicrob. Agents Chemother*. 53: 4501–4503. https://doi.org/10.1128/AAC.00671-09.
- 34 Egorova, A., Ekins, S., Schmidtke, M., and Makarov, V. (2019). Back to the future: advances in development of broad-spectrum capsid-binding inhibitors of enteroviruses. *Eur. J. Med. Chem.* 178: 606–622. https://doi.org/10.1016/j.ejmech.2019.06.008.
- 35 A. Tijsma, H.J. Thibaut, S. a H. Spieser, A. De Palma, M. Koukni, E. Rhoden, S. Oberste, G. Pürstinger, A. Volny-Luraghi, J. Martin, A. Marchand, P. Chaltin, J. Neyts, P. Leyssen, H1PVAT is a novel and potent early-stage inhibitor of poliovirus replication that targets VP1, *Antivir. Res.* 110 (2014) 1–9. doi:https://doi.org/10.1016/j.antiviral.2014.07.003.

- 36 Collett, M.S., Hincks, J.R., Benschop, K. et al. (2017). Antiviral activity of pocapavir in a randomized, blinded, placebo-controlled human oral poliovirus vaccine challenge model. *J. Infect. Dis.* 215: 335–343. https://doi.org/10.1093/infdis/jiw542.
- 37 Liu, Y., Hill, M.G., Klose, T. et al. (2016). Atomic structure of a rhinovirus C, a virus species linked to severe childhood asthma. *Proc. Natl. Acad. Sci.* 113: 8997–9002. https://doi.org/10.1073/pnas.1606595113.
- 38 Sun, L., Lee, H., Thibaut, H.J. et al. (2019). Viral engagement with host (co-)receptors blocked by a novel class of tryptophan dendrimers that targets the 5-fold-axis of the enterovirus-A71 capsid. *PLoS Pathog.* 15: e1007760. https://doi.org/10.1371/journal. ppat.1007760.
- **39** Ren, P., Zheng, Y., Wang, W. et al. (2017). Suramin interacts with the positively charged region surrounding the 5-fold axis of the EV-A71 capsid and inhibits multiple enterovirus A. *Sci. Rep.* 7 https://doi.org/10.1038/srep42902.
- **40** Abdelnabi, J.R., Geraets, J.A., Ma, Y. et al. (2019). A novel druggable interprotomer pocket in the capsid of rhino- and enteroviruses. *PLoS Biol* 71: 1–17.
- **41** Musharrafieh, R., Ma, C., Zhang, J. et al. (2019). Validating enterovirus D68-2A pro as an antiviral drug target and the discovery of telaprevir as a potent D68-2A pro inhibitor. *J. Virol.* 93 https://doi.org/10.1128/jvi.02221-18.
- **42** Thibaut, H.J., De Palma, A.M., and Neyts, J. (2012). Combating enterovirus replication: state-of-the-art on antiviral research. *Biochem. Pharmacol.* 83: 185–192. https://doi.org/10.1016/j.bcp.2011.08.016.
- 43 Matthews, D.A., Dragovich, P.S., Webber, S.E. et al. (1999). Structure-assisted design of mechanism-based irreversible inhibitors of human rhinovirus 3C protease with potent antiviral activity against multiple rhinovirus serotypes. *Proc. Natl. Acad. Sci. USA* 96: 11000–11007. https://doi.org/10.1073/pnas.96.20.11000.
- 44 Hayden, F.G., Turner, R.B., Gwaltney, J.M. et al. (2003). Phase II, randomized, doubleblind, placebo-controlled studies of ruprintrivir nasal spray 2-percent suspension for prevention and treatment of experimentally induced rhinovirus colds in healthy volunteers. *Antimicrob. Agents Chemother.* 47: 3907–3916. https://doi.org/10.1128/ AAC.47.12.3907-3916.2003.
- Hsyu, P.H., Pithavala, Y.K., Gersten, M. et al. (2002). Pharmacokinetics and safety of an antirhinoviral agent, ruprintrivir, in healthy volunteers. *Antimicrob. Agents Chemother*. 46: 392–397. https://doi.org/10.1128/AAC.46.2.392-397.2002.
- 46 Patick, A.K., Brothers, M.A., Maldonado, F. et al. (2005). In vitro antiviral activity and single-dose pharmacokinetics in humans of a novel, orally bioavailable inhibitor of human rhinovirus 3C protease. *Antimicrob. Agents Chemother.* 49: 2267–2275. https://doi.org/10.1128/AAC.49.6.2267-2275.2005.
- 47 P.S. Dragovich, T.J. Prins, R. Zhou, T.O. Johnson, Y. Hua, H.T. Luu, S.K. Sakata, E.L. Brown, F.C. Maldonado, T. Tuntland, C.A. Lee, S.A. Fuhrman, L.S. Zalman, A.K. Patick, D.A. Matthews, E.Y. Wu, M. Guo, B.C. Borer, N.K. Nayyar, T. Moran, L. Chen, P.A. Rejto, P.W. Rose, M.C. Guzman, E.Z. Dovalsantos, S. Lee, K. McGee, M. Mohajeri, A. Liese, J. Tao, M.B. Kosa, B. Liu, M.R. Batugo, J.P.R. Gleeson, Z.P. Wu, J. Liu, J.W. Meador, R.A. Ferre, Structure-based design, synthesis, and biological evaluation of irreversible human rhinovirus 3C protease inhibitors. 8. Pharmacological optimization of orally bioavailable 2-pyridone-containing

peptidomimetics, J. Med. Chem. 46 (2003) 4572–4585. doi:https://doi.org/10.1021/jm030166l.

- 48 Tijsma, A., Franco, D., Tucker, S. et al. (2014). The capsid binder vapendavir and the novel protease inhibitor SG85 inhibit enterovirus 71 replication. *Antimicrob. Agents Chemother*. 58: 6990–6992. https://doi.org/10.1128/AAC.03328-14.
- **49** Lacroix, C., George, S., Leyssen, P. et al. (2015). The enterovirus 3C protease inhibitor SG85 efficiently blocks rhinovirus replication and is not cross-resistant with rupintrivir. *Antimicrob. Agents Chemother.* 59: 5814–5818. https://doi.org/10.1128/AAC.00534-15.
- **50** Yao, C., Xi, C., Hu, K. et al. (2018). Inhibition of enterovirus 71 replication and viral 3C protease by quercetin. *Virol. J.* 15 https://doi.org/10.1186/s12985-018-1023-6.
- 51 Guan, H., Tian, J., Qin, B. et al. (2017). Crystal structure of 2C helicase from enterovirus 71. *Sci. Adv.* 3 https://doi.org/10.1126/sciadv.1602573.
- 52 De Palma, A.M., Vliegen, I., De Clercq, E., and Neyts, J. (2008). Selective inhibitors of picornavirus replication. *Med. Res. Rev.* 28: 823–884. https://doi.org/10.1002/med.20125.
- 53 Pfister, T. and Wimmer, E. (1999). Characterization of the nucleoside triphosphatase activity of poliovirus protein 2C reveals a mechanism by which guanidine inhibits poliovirus replication. *J. Biol. Chem.* 274: 6992–7001. https://doi.org/10.1074/ jbc.274.11.6992.
- 54 Barton, D.J. and Flanegan, J.B. (1997). Synchronous replication of poliovirus RNA: initiation of negative-strand RNA synthesis requires the guanidine-inhibited activity of protein 2C. J. Virol. 71: 8482–8489. http://www.pubmedcentral.nih.gov/articlerender.fcgi?a rtid=192311&tool=pmcentrez&rendertype=abstract.
- 55 Shimizu, H., Agoh, M., Agoh, Y. et al. (2000). Mutations in the 2C region of poliovirus responsible for altered sensitivity to benzimidazole derivatives. *J. Virol.* 74: 4146–4154. https://doi.org/10.1128/JVI.74.9.4146-4154.2000.
- 56 De Palma, A.M., Heggermont, W., Lanke, K. et al. (2008). The thiazolobenzimidazole TBZE-029 inhibits enterovirus replication by targeting a short region immediately downstream from motif C in the nonstructural protein 2C. J. Virol. 82: 4720–4730. https:// doi.org/10.1128/JVI.01338-07.
- 57 Ulferts, R., De Boer, S.M., Van Der Linden, L. et al. (2016). Screening of a library of FDA-approved drugs identifies several enterovirus replication inhibitors that target viral protein 2C. *Antimicrob. Agents Chemother.* 60: 2627–2638. https://doi.org/10.1128/ AAC.02182-15.
- 58 Bauer, L., Manganaro, R., Zonsics, B. et al. (2019). Fluoxetine Inhibits enterovirus replication by targeting the viral 2C protein in a stereospecific manner. *ACS Infect. Dis.* 5: 1609–1623. https://doi.org/10.1021/acsinfecdis.9b00179.
- **59** Abdelnabi, R., de Morais, A.T.S., Leyssen, P. et al. (2017). Understanding the mechanism of the broad-spectrum antiviral activity of favipiravir (T-705): key role of the F1 motif of the viral polymerase. *J. Virol.* 91: e00487–e00417. https://doi.org/10.1128/JVI.00487-17.
- **60** Wang, Y., Li, G., Yuan, S. et al. (2016). In vitro assessment of combinations of enterovirus inhibitors against enterovirus 71. *Antimicrob. Agents Chemother*. 60: 5357–5367. https://doi.org/10.1128/AAC.01073-16.
- 61 Harrison, D.N., Gazina, E.V., Purcell, D.F. et al. (2008). Amiloride derivatives inhibit coxsackievirus B3 RNA replication. *J. Virol.* 82: 1465–1473. https://doi.org/10.1128/JVI.01374-07.

- **62** Levi, L.I., Gnädig, N.F., Beaucourt, S. et al. (2010). Fidelity variants of RNA dependent RNA polymerases uncover an indirect mutagenic activity of amiloride compounds. *PLoS Pathog.* 6: 1–14. https://doi.org/10.1371/journal.ppat.1001163.
- 63 van der Linden, L., Vives-Adrián, L., Selisko, B. et al. (2015). The RNA template channel of the RNA-dependent RNA polymerase as a target for development of antiviral therapy of multiple genera within a virus family. *PLoS Pathog.* 11: 1–23. https://doi.org/10.1371/ journal.ppat.1004733.
- **64** Heinz, B.A. and Vance, L.M. (1995). The antiviral compound enviroxime targets the 3A coding region of rhinovirus and poliovirus. *J. Virol.* 69: 4189–4197. http://www.pubmedcentral.nih.gov/articlerender.fcgi?artid=189156&tool=pmcentrez&rendertype =abstract.
- **65** Arita, M., Wakita, T., and Shimizu, H. (2009). Cellular kinase inhibitors that suppress enterovirus replication have a conserved target in viral protein 3A similar to that of enviroxime. *J. Gen. Virol.* 90: 1869–1879. https://doi.org/10.1099/vir.0.012096-0.
- **66** Arita, M., Kojima, H., Nagano, T. et al. (2011). Phosphatidylinositol 4-kinase III beta is a target of enviroxime-like compounds for antipoliovirus activity. *J. Virol.* 85: 2364–2372. https://doi.org/10.1128/JVI.02249-10.
- **67** Albulescu, L., Strating, J.R.P.M., Thibaut, H.J. et al. (2015). Broad-range inhibition of enterovirus replication by OSW-1, a natural compound targeting OSBP. *Antivir. Res.* 117: 110–114. https://doi.org/10.1016/j.antiviral.2015.02.013.
- 68 J.R.P.M. Strating, L. van der Linden, L. Albulescu, J. Bigay, M. Arita, L. Delang, P. Leyssen, H.M. van der Schaar, K.H.W. Lanke, H.J. Thibaut, R. Ulferts, G. Drin, N. Schlinck, R.W. Wubbolts, N. Sever, S.A. Head, J.O. Liu, P.A. Beachy, M.A. DeMatteis, M.D. Shair, V.M. Olkkonen, J. Neyts, F.J.M. van Kuppeveld, Itraconazole inhibits enterovirus replication by targeting the oxysterol-binding protein, *Cell Rep.* 10 (2015) 600–615. doi:https://doi.org/10.1016/j.celrep.2014.12.054.
- 69 L. Albulescu, J. Bigay, B. Biswas, M. Weber-Boyvat, C.M. Dorobantu, L. Delang, H.M. van der Schaar, Y.S. Jung, J. Neyts, V.M. Olkkonen, F.J.M. van Kuppeveld, J.R.P.M. Strating, Uncovering oxysterol-binding protein (OSBP) as a target of the anti-enteroviral compound TTP-8307, *Antivir. Res.* 140 (2017) 37–44. doi:https://doi.org/10.1016/j.antiviral.2017.01.008.
- **70** Thibaut, H.J., van der Linden, L., Jiang, P. et al. (2014). Binding of glutathione to enterovirus capsids is essential for virion morphogenesis. *PLoS Pathog.* 10 https://doi. org/10.1371/journal.ppat.1004039.
- 71 Ma, H.C., Liu, Y., Wang, C. et al. (2014). An interaction between glutathione and the capsid is required for the morphogenesis of C-cluster enteroviruses. *PLoS Pathog.* 10 https://doi. org/10.1371/journal.ppat.1004052.
- 72 Mikami, T., Satoh, N., Hatayama, I., and Nakane, A. (2004). Buthionine sulfoximine inhibits cytopathic effect and apoptosis induced by infection with human echovirus 9. *Arch. Virol.* 149: 1117–1128. https://doi.org/10.1007/s00705-003-0283-6.
- 73 Dai, W., Wu, Y., Bi, J. et al. (2017). Antiviral effects of retro-2cycl and retro-2.1 against enterovirus 71 in vitro and in vivo. *Antivir. Res.* 144: 311–321. https://doi.org/10.1016/j. antiviral.2017.07.001.
- **74** Geller, R., Vignuzzi, M., Andino, R., and Frydman, J. (2007). Evolutionary constraints on chaperone-mediated folding provide an antiviral approach refractory to development of drug resistance. *Genes Dev.* 21: 195–205. https://doi.org/10.1101/gad.1505307.

- **75** Zhu, L., Xu, K., Wang, N. et al. (2018). Neutralization mechanisms of two highly potent antibodies against human enterovirus 71. *MBio* 9 https://doi.org/10.1128/mBio.01013-18.
- **76** Zhou, B., Xu, L., Zhu, R. et al. (2019). A bispecific broadly neutralizing antibody against enterovirus 71 and coxsackievirus A16 with therapeutic potential. *Antivir. Res.* 161: 28–35. https://doi.org/10.1016/j.antiviral.2018.11.001.
- **77** Nygårdas, M., Vuorinen, T., Aalto, A.P. et al. (2009). Inhibition of coxsackievirus B3 and related enteroviruses by antiviral short interfering RNA pools produced using φ6 RNA-dependent RNA polymerase. *J. Gen. Virol.* https://doi.org/10.1099/vir.0.011338-0.
- 78 Klaiber, N., McVoy, M.A., and Zhao, W. (2019). Susceptibility of enterovirus-D68 to RNAi-mediated antiviral knockdown. *Antivir. Res.* https://doi.org/10.1016/j. antiviral.2019.104565.
- **79** Liu, H., Qin, Y., Kong, Z. et al. (2016). SiRNA rargeting the 2Apro genomic region prevents enterovirus 71 replication in vitro. *PLoS ONE* https://doi.org/10.1371/journal. pone.0149470.
- Freije, C.A., Myhrvold, C., Boehm, C.K. et al. (2019). Programmable inhibition and detection of RNA viruses using Cas13. *Mol. Cell* https://doi.org/10.1016/J. MOLCEL.2019.09.013.
- **81** Menzel, M., Akbarshahi, H., Bjermer, L., and Uller, L. (2016). Azithromycin induces anti-viral effects in cultured bronchial epithelial cells from COPD patients. *Sci. Rep.* 6 https://doi.org/10.1038/srep28698.
- B2 De Clercq, E. and Li, G. (2016). Approved antiviral drugs over the past 50 years. *Clin. Microbiol. Rev.* 29: 695–747. https://doi.org/10.1128/CMR.00102-15.
- **83** Lischka, P., Hewlett, G., Wunberg, T. et al. (2010). In vitro and in vivo activities of the novel anticytomegalovirus compound AIC246. *Antimicrob. Agents Chemother*. 54: 1290–1297. https://doi.org/10.1128/AAC.01596-09.
- **84** Deres, K., Schröder, C.H., Paessens, A. et al. (2003). Inhibition of hepatitis B virus replication by drug-induced depletion of nucleocapsids. *Science* (80-.) 299: 893–896. https://doi.org/10.1126/science.1077215.

Novel Antiviral Strategies Against Emerging Arbovirus Infections

Yannick Brüggemann¹, Carina Elsner², and Stephanie Pfaender^{1,3,4}

¹ Department for Molecular and Medical Virology, Faculty of Medicine, Ruhr University Bochum, Bochum, Germany

² Institute of Virology, University Hospital Essen, Essen, Germany

³ Institute of Virology and Immunology IVI, Bern, Switzerland

⁴ Department of Infectious Diseases and Pathobiology, Vetsuisse Faculty, University of Bern, Bern, Switzerland

15.1 Introduction

Arthropod-borne viruses (arboviruses) are distributed worldwide and over 500 species grouped into 8 virus families have been described so far [1]. Several of these viruses constitute a significant global health burden as they can cause a variety of symptoms in humans upon infection [2]. Generally, arboviruses require a minimum of two hosts, a vertebrate and an arthropod host, for successful proliferation and dissemination [3]. In most cases, humans are considered as dead-end or incidental hosts since they usually do not contribute to the transmission cycle by developing viremia which would be high enough to infect arthropod hosts [4].

Even though arboviruses are distributed globally, their majority can be found in tropical areas where they can be transmitted permanently by cold-blooded arthropods [5]. However, several factors facilitate the (re-) emergence of these viruses toward previously naïve populations. An increase in human population densities, combined with urbanization and globalization as well as other demographic and societal changes have significantly influenced transmission dynamics of arboviral diseases [6]. Consequently, exposure frequencies of humans, domestic animals, and livestock to mosquitos have increased and combined with international travel and globalization accelerate the geographic redistribution and outbreak of more frequent and larger arbovirus epidemics. During recent centuries, five human epidemic mosquito-borne arboviruses namely Zika virus (ZIKV), yellow fever virus (YFV), dengue virus (DENV), West Nile virus (WNV), and chikungunya virus (CHIKV) have emerged in both hemispheres [7]. The first four viruses are grouped into the genus of flaviviruses of the family of Flaviviridae, which also includes other clinically relevant members (e.g. Japanese encephalitis virus [JEV], tick-borne encephalitis virus [TBEV], and Usutu virus [USUV]) [8]. In contrast, CHIKV is classified as an alphavirus belonging to the family Togaviridae [9]. All are transmitted in zoonotic cycles involving nonhuman primates (i.e. prosimans, monkeys, apes), and in case of WNV vertebrate hosts, and arboreal mosquitoes. They have entered human-to-human cycles involving *Aedes* and *Culex* spp. mosquito transmission.

15.1.1 Zika Virus

ZIKV infection is the most recent example of a (re-)emerging arbovirus and was declared as a Public Health Emergency of International Concern by the World Health Organization in 2016. ZIKV was first isolated in 1947 from a sentinel monkey during studies of enzootic yellow fever in the Zika forest in Uganda, East Africa [10]. The first potential infections of human individuals were reported during the 1950s in Africa and later in Asia and in the Pacific regions. The early perception ZKIV infection was that of a relatively mild, self-limiting illness, occasionally accompanied by fever, red eyes, joint pain, headache, and a maculopapular rash. Hence due to its mild clinical symptoms, ZIKV infections were rarely investigated and further potentially misdiagnosed due to serological presentation and/or clinical signs resembling mild forms of chikungunya or dengue infection [11]. Therefore, the virus circulated silently for several decades between Africa and Southeast Asia without causing any detectable epidemics and with only 13 reported cases of natural human infections, until in 2007 the first known epidemic outbreak of ZIKV on Yap Island, Micronesia, was reported [11, 12]. Since then several outbreaks of ZIKV spawned large regions of the South Pacific and in 2015/16, large outbreaks occurred in the Americas. Importantly, during the recent epidemics, ZIKV infections were unexpectedly linked to birth defects like severe neurological complications such as microcephaly, brain malformations, and other birth defects of the fetuses of infected pregnant women [13]. Furthermore, in adults, ZIKV infections were found to be associated with the autoimmune disorder Guillain-Barré syndrome [14]. As of September 2019, a total of 86 countries have reported evidence of mosquito-transmitted ZIKV infections.

15.1.2 Yellow Fever Virus

YFV and its associated human disease yellow fever have been known in Africa for centuries. The virus circulates in an enzootic cycle between nonhuman primates and Aedes mosquitos in Sub-Saharan rain forests [15]. Most YFV infections are asymptomatic and self-limiting within one week . Patients experience fever, headache, chills, muscle pain, and nausea. However, about 10-25% of patients further develop hemorrhagic fever, a pansystemic viral sepsis with viremia, fever, kidney, and liver damage and with a case-fatality rate of 20-50% in symptomatic cases [16]. Historical records implied that YFV and its mosquito vector were introduced into the Western hemisphere via the slave trade from the fifteenth century onward [17]. Changes in the transmission pattern from a forest or jungle cycle to an urban cycle are characterized by a rapid human to mosquito to human transmission [18]. The virus has been known to cause major epidemics in the Americas and in Africa from the seventeenth century to present times. Within Africa, YFV epidemics arise irregularly and have caused major outbreaks in the past (e.g. in Gambia in 1978-1979, Nigeria in 1969, and Ethiopia in 1962) [19]. In the Western hemisphere, major outbreaks occurred as early as 1737 in coastal Virginia, a major outbreak in Philadelphia with 5000 deaths in the summer of 1793, and during the next few years repeated outbreaks in New York City and north to Boston [20]. Furthermore, several outbreaks were identified in European ports. In December 2015, a yellow fever epidemic broke out in Angola and the Republic of Congo and quickly spread within densely populated urban areas. International travel facilitated disease spread to Kenya, the Democratic Republic of the Congo, and China [19]. Consequently, in 2016, the WHO declared the current yellow fever epidemic as a global threat [21]. Several YFV outbreaks are continuously being reported within Africa and 32 African countries are being considered at risk of yellow fever, including a total population of 610 million people, among which more than 219 million live in urban settings. The virus is by now considered to be endemic in in tropical Africa, ten South- and Central American countries, and in several Caribbean islands [21]. Although an effective vaccine has been available since the 1930s [15], effective implementation is frequently hampered during outbreaks in resource-limited countries due to limited vaccine supply.

15.1.2.1 Dengue Virus

Dengue viruses are one of the geographically most widespread arboviruses. It is estimated that 390 million people are being infected annually, of which 96 million show a disease manifestation [22]. There are four antigenically distinct DENV serotypes (DENV 1-4) that display comparable epidemiology and cause similar diseases in humans [23]. In contrast to other arboviruses, DENVs are independent of a primitive enzootic forest cycle and have fully adapted to the human host, resulting in direct circulation of the virus between humans and their arboviral Aedes vector [24]. It is speculated that DENV has emerged during the fifteenth through nineteenth century in Africa, before it was globally transmitted during the eighteenth and nineteenth century due to the expansion of commercial shipping [25]. For the first time isolated in the 1940s, the virus has by then already spread globally and is considered a major public health problem in most tropical countries [26]. Asian countries are mostly affected and account for 75% of the dengue disease burden, followed by Latin America and African countries [22]. Infections result in a wide variety of clinical manifestations ranging from mild febrile illness to severe and fatal disease [27]. The initial phase can be accompanied by high temperature, headache, vomiting, myalgia, and joint pain, as well as mild hemorrhagic manifestations [23]. Critical phases can include increased vascular permeability (systemic vascular leak syndrome) and shock, with case-fatality rates in patients with dengue shock syndrome as high as 44% [23, 27]. In general, infections with DENV can have different outcomes. A primary infection can provide lifelong protection, however only against the infecting serotype. Secondary infections after infection with a different serotype as well as infections of infants born to dengue-immune mothers often result in more severe cases with hemorrhagic fever/dengue shock syndrome [28]. In late 2015, the world's first dengue vaccine by Sanofi Pasteur, CYD-TDV or Dengvaxia, was licensed and was approved 2019 by the FDA [29, 30]. However, given the increased severity of secondary DENV infections, the vaccine is not approved for use in individuals not previously infected by any DENV serotype or for whom this information is unknown [30].

15.1.3 West Nile Virus

WNV is second to DENV based on the extent of its global distribution [7]. WNV has a broad vector and host tropism and infects next to a variety of *Culex* spp. mosquitos also a broad range of vertebrates including humans, birds, horses, and pigs, with wild birds being

370 15 Novel Antiviral Strategies Against Emerging Arbovirus Infections

suspected to be the optimal hosts for viral amplification [31]. WNV causes disease in humans with a wide variety of presentations. In 75% of cases, the infection remains asymptomatic; however, it can also lead to fever and malaise as well as to neuroinvasive disease cases (less than 1%) with a case-fatality rate of approximately 10% [32]. The virus was originally isolated in Uganda in 1937 and since then has been shown to be widespread in the Middle East and India. It caused infrequent outbreaks in Israel, Egypt, France, and South Africa [33]. Until the early 1990s, human outbreaks were mainly associated with mild febrile illnesses; however, this changed dramatically with the emergence of new viral strains with likely African origin and increased human disease incidence in parts of Russia and Southern and Eastern Europe and with large outbreaks of increased clinical severity occurring in several countries [32]. In 1999, WNV expanded to the Western hemisphere resulting in increasing reports of deaths and encephalitis in New York. It spread within three years to most parts of the United States and the neighboring countries of Canada and Mexico [34]. As of today, WNV is found in Africa, Europe, Asia, North America, Australia, and the Middle East and is now considered the most important causative agent of viral encephalitis worldwide [33].

15.1.4 Chikungunya Virus

The arthropod-borne CHIKV (CHIKV) is an alphavirus grouped into the family of Togaviridae and is the causative agent of the so-called Chikungunya fever (CHIKF). CHIKV was first isolated in 1953 in Tanzania and re-emerged in an unprecedented outbreak between 2005 and 2006 in the Indian Ocean islands [35-37]. In October 2013, the first known autochthonous CHIKV cases in the Western hemisphere have been reported on the island of Saint Martin [38]. Since then, several autochthonous cases were reported from numerous other Caribbean islands leading to the assumption that virus spread to other island countries and expansion into mainland areas of South, Central, and North America are inevitable [38, 39]. Initially rated as a nonfatal disease, CHIKV infection developed as a global health burden. The acute phase of an infection is characterized by symptoms of high fever, severe joint pain, and rash, which improve after 7-10 days. Some patients suffer from severe outcomes by developing neurological complications including encephalopathy and Guillain-Barré syndrome (reviewed in [40, 41]). Moreover, approximately 50% of infected patients develop chronic chikungunya arthritis (CCA) persisting for month or years [42-44]. In some individuals, mostly neonates, the elderly and patients with underlying medical comorbidities, CHIKV infections can lead to death [45, 46]. CHIKV has been classified into three different genotypes: East-Central-South-African (ECSA), West African, and Asian. The virus is mainly transmitted via two species of infected mosquitoes, namely Aedes aegypti and Aedes albopictus. Furthermore, maternal-to-child transmission with neonates acquiring the infection during birth has been reported [47, 48]. Until the massive outbreak and viral spread in 2005–2006, CHIKV has been primarily found to be transmitted by Aedes aegypti. The adaption to the vector Aedes albopictus is the result of a single mutation A226V within the E1 glycoprotein [49]. In addition, co-circulation of different CHIKV genotypes enhances the risk of coinfections and recombination. Of note, multiple coinfections with DENV, ZIKV, and CHIKV have also been reported [50]. For example, a simultaneous transmission of CHIKV and ZIKV as result of a single bite of a mosquito is conceivable [51]. These scenarios pose additional challenges to the identification of potent antivirals.

15.2 Intervention Strategies

So far, no specific antiviral therapeutics to prevent or treat arbovirus infection have been licensed. In most cases, infected patients are treated symptomatically using antipyretics and nonsteroidal anti-inflammatory drugs (NSAIDs). Therefore, currently the best strategy to prevent arbovirus infection and viral spread remains vector control and avoiding mosquito bites. However, extensive global efforts to understand the biology and pathogenesis of various (re-)emerging arboviruses are carried out to identify potential molecular targets for antiviral compounds. Furthermore, by screening large compound libraries or repurposing of clinically approved drugs, great efforts have been made to identify drug candidates directed against viral proteins (DAA; direct-acting antivirals) and against host targets (host-targeting antivirals). Potential broad-spectrum antiviral activity has been observed for various compounds during target-specific or phenotypic *in vitro* and *in vivo* assays (Table 15.1).

15.3 Genome Organization and Viral Replication Cycle

All flaviviruses share a similar genome organization: one long open reading frame (ORF) which is translated as a polyprotein and cleaved by virus-encoded and host proteases into three structural proteins (the capsid [C], precursor membrane [prM], and envelope glycoprotein [E]) that form the viral particle and seven nonstructural (NS) proteins (NS1, NS2A, NS2B, NS3, NS4A, NS4B, and NS5) that fulfill several roles during viral replication (Figure 15.1) [56]. Both E and prM form the surface structure of the virion. Whereas the viral E protein mediates virus adsorption, internalization, and fusion with the host cell membranes, prM acts as a chaperone to facilitate conformational folding of E [57]. The surface structural protein E is the primary viral protein against which neutralizing antibodies are produced [58]. The capsid protein mediates assembly and packaging of the viral RNA genome to form the viral nucleocapsid. Furthermore, it has a crucial role during viral particle assembly where it associates with other proteins on lipid droplets and the endoplasmic reticulum (ER) [59]. The nonstructural proteins coordinate viral replication, assembly, proteolysis, maturation, and host immunity regulation [60]. NS1 has several roles; the intracellular dimer form contributes to genome replication and virion maturation, whereas the secreted hexamer plays a role in immune evasion [60]. The NS2 region encodes two proteins, NS2A and NS2B [61]. NS2A is involved in the proteolytic processing of the C terminus of NS1, which forms a part of the replication complex and modulates host antiviral responses, while NS2B acts as a cofactor to the N-terminal region of NS3 and its associated protease activity [61]. Together they are responsible for the cleavage and post-translational modification of the viral polyprotein [62]. Next to its protease function, the NS3 C-terminal domain possesses RNA helicase,



Figure 15.1 Schematic representation of the flavivirus (left) and CHIKV genome organization (right).

Family	Genus	Virus(es)	Vaccine	Antivirals	Discovery of virus
Flaviviridae	Flavivirus	Yellow fever virus	Yes	No	1927 [52]
		Dengue viruses	Yes	No	1960s [53]
		West Nile virus	No	No	1937 [54]
		Zika viruses	No	No	1947 [55]
Togaviridae	Alphavirus	Chikungunya virus	No	No	1953 [36]

Table 15.1	Antiviral drugs and	l vaccines for	emerging	arboviruses.

nucleoside, and RNA triphosphatase activities and contributes to viral RNA replication and virus particle formation. In addition, NS2B-NS3 is involved in the assembly of the flavivirus replication complex and modulates viral pathogenesis and the host immune response [63]. The NS4A protein associates with the virus replication complex and induces ER membrane rearrangements and autophagy to promote viral replication [64–67], while the NS4B protein modulates the host innate immune response and the helicase activity of NS3 [68–70]. The NS5 protein is composed of an N-terminal methyltransferase (MTase) domain and a C-terminal RNA-dependent-RNA polymerase (RdRp) domain [67].

In contrast, the genome of CHIKV contains two ORFs which encode for five structural proteins (including the glycoproteins E1, E2, and E3 as well as 6k and capsid protein) and four nonstructural proteins (nsPs1–4) which mediate genome replication, RNA capping, polyprotein cleavage, and other functions required for viral replication (Figure 15.1) [71].

The replication cycle of flaviviruses along with the most relevant sites of pharmacological intervention are schematically outlined in Figure 15.2a. In short, following binding to a host cell through interaction with specific attachment factors, viral particles are internalized by clathrin-mediated endocytosis. Endosomal acidification triggers fusion of viral and endosomal membranes and thereby releases the viral nucleocapsid into the cytosol [72]. The viral genome is then translated resulting in the synthesis of a polyprotein that is coand post-translationally cleaved by cellular and viral proteases [56, 73].

Flaviviruses subsequently initiate the formation of a replication complex on modified ER membranes which mediates the replication of the genomic RNA (+) template into RNA (-), serving as a template for new RNA (+) synthesis. Viral assembly then begins through budding of immature virions containing a newly synthesized RNA genome into the ER lumen. Various protease and pH-dependent maturation steps in the ER and Golgi complex including different post-translational modifications generate mature infectious particles, which are released through the secretory pathway [56, 74–76].

In contrast, upon release into the cytosol, the CHIKV genome is only partially translated into a polyprotein, which consists of the four nsP1–4 [74]. NsPs subsequently assemble and modify the plasma membrane to form viral replication compartments (spherules) to protect viral RNA intermediates from degradation and recognition by cellular pattern-recognition receptors [71]. As infection progresses, spherules are internalized to the surfaces of modified endo- and lysosomal compartments to form so-called cytopathic vesicles (CPV) [77]. Subsequent RNA translation produces the structural polyprotein (C-E3-E2-6K-E1) and autoproteolysis releases free capsid protein (C) into the cytoplasm which assembles with the viral RNA genome to form nucleocapsids. The remaining polyprotein is then




Figure 15.2 Schematic representation of the replication cycle and polyprotein organization for flaviviruses (a) and CHIKV (b) and potential sites of pharmacological intervention. C, capsid protein; E, envelope glycoprotein; NS (Flavivirus) or nsPs (CHIKV), nonstructural protein; prM, pre-membrane glycoprotein; RdRp, RNA-dependent-RNA polymerase; MTase, methyltransferase.

directed to the ER where it undergoes cleavage into individual proteins. The E2/E1 proteins are further post-translationally modified and transit through the secretory system to the plasma membrane where they assemble with previously formed nucleocapsids to release mature progeny viruses [71, 78] [Figure 15.2b].

15.4 Targets For Antiviral Therapy

A viral target protein should ideally combine two features: it should be essential for the viral replication cycle and have a low frequency of "tolerated" (nonlethal, but resistance-conferring) mutations [79]. Direct-acting antivirals (DAAs) often facilitate the emergence of resistance in RNA viruses, as the extraordinarily high mutation rates paired with short generation times promotes rapid generation of viral variants, which escape treatment [80]. RNA viruses thus do not exist as a homogenous clonal population within an infected host, but rather diversify into a heterogenous population (a so-called "quasispecies"), which hovers around a most fit "master sequence" [81]. In contrast, host-targeting antivirals are less likely to result in the selection of resistant viruses as host factors are genetically more stable than viral factors [79]. Nevertheless, essential structural or functional features within viral proteins also often have a high genetic barrier to mutations and thus show a high degree of evolutionary conservation. A high degree of structural conservation across different viruses can further indicate targets with potential for broad-spectrum relevance and further enables to estimate the likelihood of mutations conferring drug resistance [79].

Several of the enzymatic motifs involved in protein and RNA processing such as the catalytic motifs of the viral protease (NS3 flavivirus; nsP2 CHIKV), the ATP- and RNAbinding regions of the RNA helicase, and the substrate or metal recognition motifs of the RdRp (flavivirus: NS5, CHIKV nsP4) are highly conserved among different RNA viruses [82-84]. Both NS5 and NS3 proteins of different flaviviruses, including HCV, have thus emerged as key targets for antiviral drug development. The pangenotypic prodrug sofosbuvir is an example of an efficient direct-acting antiviral drug that interferes with the NS5B polymerase of HCV [85]. On the other hand, the nonstructural proteins NS2 and NS4 display remarkably high interspecies variability and therefore are more prone to drug resistance-conferring mutations and further provide fewer promising targets for antiviral drugs with broad activity. Notably, the treatment of chronic hepatitis C virus infection has been revolutionized by the development of several DAAs, including nonstructural proteins 3/4A (NS3/4A) protease inhibitors (PIs), NS5B nucleoside polymerase inhibitors (NPIs), NS5B non-nucleoside polymerase inhibitors (NNPIs), and NS5A inhibitors [86]. Table 15.2 provides an overview of selected DAAs and their current stage of development, mostly repurposed drugs from e.g. HCV.

15.5 Direct-acting Antivirals (DAAs)

See Table 15.2.

15.6 RNA-dependent RNA Polymerase Inhibitors

The RNA-dependent RNA polymerase (RdRp) (flavivirus: NS5, CHIKV nsP4) is the most conserved gene among the members of the *Flaviviridae* family [114]. Since human cells lack the RdRp domain, this viral class of enzymes provides a highly attractive target for the

Table 15.2Selected direct-acting antivirals for different arboviral viral proteinsand the respective viruses/biological systems which have been used to evaluatethe respective drugs.

Target	Compound	Virus	Biological system	Comment
RdRp	7DMA	ZIKVV WNV DENV	Cell culture (EC ₅₀ 0.6 μ M), mice [79, 87] Cell culture (EC ₅₀ 0.3 μ M), mice [88] Cell culture (EC ₅₀ 15 μ M) [89], mice [79]	Nucleoside analogue originally developed as a treatment for HCV but failed during clinical trials [90].
	BCX4430	ZIKV DENV YFV WNV JEV CHIKV	Cell culture $(EC_{50} \sim 5 \mu g/ml)$ [91], mice [91] Cell culture $(EC_{50} 32.8 \mu M)$ [92] Cell culture $(EC_{50} 14.1 \mu M)$ [92], Hamster [93] Cell culture $(EC_{50} 1.5 \mu M)$ [94] Cell culture $(EC_{50} 43.6 \mu M)$ [92] Cell culture $(EC_{50} > 100 \mu M)$ [92]	Nucleoside analogue with potential broad-spectrum activity. A phase 1 trial of clinical safety and pharmacokinetics has been successfully completed [79].
	Sofosbuvir	ZIKV DENV YFV CHIKV	$ \begin{array}{l} \mbox{Cell culture (EC_{50} 1-5 \mu M),} \\ \mbox{mice [95-98]} \\ \mbox{Cell culture (EC_{50} 1.4 \mu M) [99, 100]} \\ \mbox{Cell culture (EC_{50} 4.2 \mu M), mice [101]} \\ \mbox{Cell culture (EC_{50} 17 \mu M) [102],} \\ \mbox{mice [102]} \end{array} $	FDA-approved nucleoside analogue inhibitor for the treatment of chronic HCV [103, 104].
Protease	Temoporfin	ZIKV	Cell culture (EC ₅₀ 0.02 µM), mice [105]	Photosensitizer drug approved by the European Union for the treatment of squamous cell carcinoma of the head and neck.
	Erythrosin B	ZIKV DENV YFV WNV	Cell culture ($EC_{50} 0.6 \mu M$) [106] Cell culture ($EC_{50} 1.2 \mu M$) [106] Cell culture ($EC_{50} 0.4 \mu M$) [105] Cell culture ($EC_{50} 0.7 \mu M$) [106]	FDA-approved food additive.
NS4B	Lycorine	WNV DENV YFV	Cell culture (EC ₅₀ 0.2μ M) [107] Cell culture (1.2μ M) [107] Cell culture (1.2μ M) [107]	Natural alkaloid found in various <i>Amaryllidaceae</i> species.
MTase	4-HPR	ZIKV DENV	Cell culture (EC ₅₀ .2.3 μM) [108], mice [109] Cell culture (EC ₅₀ ~1 μM), mice [110]	Potential chemotherapeutic agent with different phase 1/2 clinical trials for breast cancer treatment completed [111].
NS2B	NSC135618	DENV	Cell culture (EC ₅₀ 0.81 μ M)	[134]
		ZIKV	Cell culture (EC ₅₀ 1 μ M)	[134]
		WNV	Cell culture (EC ₅₀ 1.27 μ M)	[134]
		YFV	Cell culture (EC ₅₀ 0.28 μ M)	[134]

(Continued)

Target	Compound	Virus	Biological system	Comment
NS3 helicase	ST-610	DENV	Cell culture (EC ₅₀ 0.272 µM)	[135]
			AG129 mice	[135]
Nonspecific	Ivermectin	DENV JEV YFV CHIKV	Cell culture ($EC_{50} > 1 \mu M$) [112] Cell culture ($EC_{50} 0.3 \mu M$) [112] Cell culture ($EC_{50} 3.1 \mu M$) [112] Cell culture ($EC_{50} 0.6 \mu M$) [113]	FDA-approved antiparasitic agent.
	Chloroquine	CHIKV	Cell culture [158]	Phase 3 clinical trial (ClinicalTrials.gov Identifier: NCT00391313) showed no difference in infection outcome.
		ZIKV	Cell culture (EC ₅₀ 9.82–14.2 µM) [157]	

Table 15.2	(Continued)
------------	-------------

development of broad-spectrum antivirals. RdRp inhibitors can be classified into two groups: nucleoside analogues and non-nucleoside inhibitors (NAI and NNI, respectively) [115]. Nucleoside analogues are among the largest class of small molecule-based viral inhibitors. Current efforts to identify novel compounds targeting the RdRp of different emerging arboviruses have generated encouraging results in cell culture and animal model systems. Whereas NNI directly bind to the RdRp to inhibit RNA synthesis, NAIs are substrate analogues which either act as chain terminators upon incorporation into the nascent RNA chain or promote accumulation of mutations in viral genomes beyond an error threshold that eventually results in the loss of viral infectivity [116–119]. Numerous nucleoside inhibitors based on the structure of the four natural nucleotides have been designed and evaluated regarding their antiviral activity in vitro [115]. 7-Deaza-2'-C-methyladenosine (7DMA) was originally developed as a treatment for HCV, but despite promising results in animal studies, it ultimately failed in clinical trials [90], potentially due to mitochondrial toxicity [96, 120]. Subsequent studies further showed broad-spectrum antiviral activity against ZIKV, WNV, DENV in vitro and in vivo [79]. Since 7DMA could potentially be applied during short-term therapy of acute arboviral diseases, it may still represent a promising antiviral candidate to date [96]. The adenosine analogue BCX4430, originally also intended as a treatment for HCV and Filovirus infections was shown to exert broad-spectrum antiviral effectiveness in vitro against mosquito-transmitted viruses such as YFV, DENV-2, JEV, CHIKV, WNV, and ZIKV [92-94]. An antiviral activity and a favorable pharmacokinetic profile against YFV and ZIKV were further observed during in vivo animal studies and a phase 1 clinical trial (ClinicalTrials.gov identifier: NCT02319772) has been completed [91, 93, 121].

The conversion of NAI into a biologically active triphosphate by cellular kinases is an essential and often limiting step for the function of all NAI but mostly not assessed during initial biochemical assays. The efficiency of this step can vary greatly among individual host cellular types due to differences in expression/activity levels of nucleoside kinases and other components involved in nucleoside metabolism and transport, and therefore result in cell-type-dependent antiviral activity [115]. Using a monophosphate prodrug approach based on the introduction of the phosphorylated group into the 5' nucleoside position, this limitation can be partially bypassed and has been successfully used during the development of the phosphoramidate prodrug sofosbuvir [122]. Sofosbuvir is a commercially available, FDA-approved RdRp inhibitor for the treatment of Hepatitis C [103, 104]. Within cells, the prodrug is converted to the pharmacological active compound GS-461203 (2'-deoxy-2'- α -fluoro- β -Cmethyluridine-5'-triphosphate), which acts as a defective substrate of the RdRp to inhibit viral RNA synthesis [123]. Beyond HCV, the antiviral activity of sofosbuvir was also confirmed during cell-based assays for different emerging arboviruses, including ZIKV, DENV, YFV, and CHIKV [95, 99, 100, 102, 124, 125]. Antiviral activity against ZIKV varied greatly among cell lines with different origin [96]. This cell-line-dependent antiviral activity of sofosbuvir was later shown to correlate with the intracellular concentration of its active triphosphate, implying a link to cell-specific metabolism [126]. Additionally, sequence analysis of ZIKV isolated from sofosbuvir-treated cells further revealed an increase of mutations in the viral genome, suggesting a secondary mode of action by increasing error-prone replication [127]. Oral treatment with sofosbuvir was shown to reduce ZIKV-induced mortality in various mouse model systems and further reduce ZIKV levels from 60 to 90% in different anatomical compartments [95-98]. Likewise, sofosbuvir treatment of YFV- or CHIKV-infected mice reduced virus-induced mortality [101, 102]. Due to its safety profile in humans and antiviral effects in vitro and in mice, sofosbuvir may represent a novel therapeutic option for the treatment of different emerging arboviruses. However, since sofosbuvir was designed to target HCV within liver cells, it remains to be determined if sufficiently high exposure levels are also reached within the respective viral replicating sites to achieve antiviral effects in humans. Overall, the development of specific nucleoside inhibitors is often complicated due to efficacy and toxicity issues and in the case of NAI repurposed from HCV/HIV treatment, differences in the cellular tropism.

15.7 Protease/Helicase Inhibitors

Following the RdRp, the NS3 protein (nsP2 for CHIKV) is the second-most conserved viral protein among the family of *Flaviviridae* [74, 128]. NS3/nsP2 perform a variety of essential functions within the viral life cycles rendering them attractive drug targets. The N-terminal (C-terminal for CHIKV) protease domain mediates proteolytic processing of the viral polyprotein, while the C-terminal (N-terminal for CHIKV) domain possesses ATPase/helicase activity to unwind the double-stranded RNA (dsRNA) intermediates [78, 128]. Considering its structural conservation and multifunctional roles during viral replication, targeting the protease/helicase is widely considered as an attractive strategy for the development of broad-spectrum antivirals [129].

Targeting viral proteases has been successfully applied for the treatment of HCV and HIV with various peptidic and pseudopeptidic compounds currently in clinical use [130], see also the article by Stephan in this book. However, examination of selected HIV/HCV PIs against DENV and CHIKV revealed only weak antiviral activity and a lowered selectivity index (SI = CC_{50}/EC_{50}) for the individual drugs [131]. Nonetheless, given the encouraging results from HCV and HIV, and due to straightforward enzymatic assays and availability of structural information, various potential inhibitors of the NS3 protease domain have been

evaluated during both high-throughput screens (HTS) and structure-based drug design approaches [79]. Unfortunately, so far, most attempts to develop potent flavivirus PIs which focused on the NS3 active site had limited success, potentially complicated by two features: First, due to a rather shallow substrate binding site, potential inhibitors likely rely on only a few binding sides, creating a low barrier of resistance-conferring mutations as observed for HCV PIs [132]. Second, the active site displays a strong preference for substrates with polybasic recognition sites. However, the incorporation of basic or polar functional groups in potential inhibitors frequently limits bioavailability and therefore antiviral efficiency in vivo [79]. Hence, inhibitors derived from substrate-mimicking peptides often display highly potent activities during initial in vitro studies but may display poor in vivo efficacy due to poor penetration of charged peptides across cellular membranes [133, 134]. Notably, the flavivirus NS3 protease relies on NS2B as a cofactor to perform its function. Targeting the NS3-NS2B interaction thus provides a different strategy to inhibit flavivirus NS3 protease function [129, 135]. Unlike the flat and featureless active site, the NS3 pockets containing the NS2B interaction residues are deep and hydrophobic [136]. A HTS assay to identify inhibitors of the NS2B-NS3 interaction revealed three compounds (temoporfin, niclosamide, and nitazoxanide) with broad-spectrum anti-flavivirus activities (Table 15.2). The most potent drug, temoporfin, not only inhibited ZIKV infection in human placental and neural progenitor cells, but also prevented ZIKV-induced viremia and mortality in mouse models [105]. Likewise, erythrosin B (EB), an FDA-approved food additive was shown to interfere with the NS3-NS2B interaction and to significantly reduce the titers of DENV2, ZIKV, YFV, JEV, and WNV in vitro [106]. Due to its excellent safety profile, EB might represent a promising candidate for future pharmacological studies. Furthermore, the previously described structure of the NS2B/NS3 protease complex opens the possibility to perform "structure-guided" virtual screens to identify potential small molecule inhibitors [137, 138]. For example, a virtual screen of a library from the National Cancer Institute Developmental Therapeutics Program (NCI DTP) to identify inhibitors of the viral protease cofactor NS2B revealed the compound NSC135618 as a potent inhibitor of several flaviviruses including DENV, ZIKV, WNV, and YFV [139]. The development of compounds targeting the NS3-NS2B interaction thus seems to provide a promising approach without suffering from the previously discussed drawbacks of active-site inhibitors.

So far, only limited progress has been made toward the development of specific inhibitors against the flavivirus helicase and currently no helicase inhibitors have been approved for clinical trials or usage. The benzoxazole analogue, ST-610, was reported to display low cyto-toxicity and inhibit viral replication of DENV-1–4 and YFV (but not WNV or JEV) *in vitro* and to lower viral load in a mouse model upon DENV infection [140]. Further compounds such as the pyrrolone derivative (compound 25) and suramin were found to lower WNV, DENV, and CHIKV replication in cell culture [141–143]. Nonetheless, the development of specific inhibitors is frequently complicated by the need to achieve selectivity against host enzymes with similar enzymatic functions.

15.8 Envelope Protein Inhibitors

Viral envelope proteins mediate virus entry to initiate an infection. Unlike viral proteases and polymerases, envelope proteins lack a conserved active site providing a major barrier in the development of effective antivirals. The flavivirus envelope protein is composed of three ectodomains; a conserved pocket (β OG pocket) located between domains I and II mediates conformational changes in the E protein necessary for membrane fusion and viral entry [144]. The β OG pocket has therefore been suggested as potential target to develop broad-spectrum antivirals against multiple, mosquito-borne flaviviruses [145]. In agreement with this, different pyrimidine and cyanohydrazone inhibitors were recently shown to efficiently block ZIKV, WNV, and JEV replication *in vitro* and inhibit DENV *in vivo* [146–148]. A virtual docking screening approach further identified different potential compounds, including NITD-448, compound 6 (with a quinazoline nucleus), P02, D02, D04, D05, A5, and 1662G07 [149]. Some of these compounds such as A5 were further tested in cell culture and showed inhibition of WNV, DENV, and YFV [117]. Likewise, various potential inhibitory compounds including phenothiazine and bafilomycin for CHIKV were identified *in silico* and could be further evaluated during future studies [150, 151].

15.9 Capsid Protein Inhibitors

Compared to other viral proteins, little efforts have been carried out to identify potential capsid protein inhibitors. A single small molecule inhibitor, ST-148, has been found to inhibit DENV replication in multiple cell types and to reduce viral load in a mouse model [140]. On the other hand, ST-148 did not affect JEV replication and displayed only weak activity against YFV. Targeting a hydrophobic pocket of the CP of CHIKV using picolinic acid blocked its interaction with the E2 protein and inhibited CHIKV replication *in vitro* [152].

15.10 NS4B Inhibitors

The flavivirus NS4B protein forms an integral membrane protein which localizes to intracellular membranes and functions as an essential component of the viral replication complex. Since the NS4B proteins of flaviviruses are orthologous genes (genes encoded at the same position of the viral ORF) but not homologous (evolutionarily related) and therefore frequently differ in their structure and/or function in the viral life cycle inhibitory effects of specific compounds often remain limited to a single virus [68, 99]. The natural alkaloid lycorine (Table 15.2) potently inhibited viral replication of WNV, DENV, and YFV in cell culture and so far is the only identified compound with potential broad-spectrum antiviral activity [153].

15.11 Methyltransferase Inhibitors

Besides its function as RdRp, the N-terminal domain of the NS5 protein of flaviviruses and the nsP1 protein of CHIKV mediate capping of the 5' end of newly synthesized viral RNAs, which is critical for viral genome replication. Structural conservation between the core domains of various MTases harbors the potential to identify inhibitors with broad-spectrum antiviral activity, but also complicates the design of virus-selective inhibitors without affecting host MTases [154]. Various inhibitors of viral MTases for CHIKV and flaviviruses

have been identified by cell-based assays, virtual screening, and structure-based design [117, 155–158]. Despite the identification of specific viral MTase inhibitors during structure-based design approaches, the respective compounds such as the natural nucleoside sinefungin frequently lack efficient membrane permeability and therefore robust antiviral activity [117, 159, 160]. The synthetic retinoid, N-(4-hydroxyphenyl)-retinamide (4-HPR, Table 15.2), has been extensively examined in humans for treatment or prevention of a variety of cancers [111]. Recent studies further reported antiviral activity of 4-HPR against DENV and ZIKV in cell culture and different mouse models [108–110, 161]. Although the exact inhibitory mechanism remains unclear, 4-HPR has been proposed as a candidate for drug repurposing due to its demonstrated activity against multiple flaviviruses and its well-established safety and tolerability in humans [109].

15.12 Inhibitors with Nonspecific Action

Various compounds with antiviral activity against different arboviruses but unknown mechanism have further been identified during different phenotypic assays. Repurposing strategies are a particularly attractive option due to the established pharmacokinetics and safety profiles in humans. This strategy further holds the potential to yield "emergency" antivirals, for which a higher incidence of side effects and limited broad-spectrum activity can be tolerated [79]. Ivermectin (Table 15.2), a broadly used antihelminthic drug, was shown to selectively inhibit the replication of JEV, DENV, CHIKV, and YFV in cell culture [112, 113]. Given that ivermectin has been used for the treatment of a variety of parasitic diseases for over two decades, assessing its potential for the treatment of arboviruses infections in clinical trials maybe feasible. Chloroquine is used as an antimalarial drug since 1934 and is suggested to inhibit viral entry or assembly and has further been shown to exert antiviral effects against various flaviviruses as wells as CHIKV (Table 15.2) [162-165]. Of note, chloroquine has been tested in the phase 3 clinical trial "CuraChik" (ClinicalTrials.gov Identifier: NCT00391313) in patients with clinical chikungunya disease diagnosed within less than 48 hours; however, treatment showed no significant difference in outcome regarding the duration of febrile arthralgia or decrease in viral load. Consequently, administration of chloroquine to treat acute CHIKV infection was abandoned [166]. Nevertheless, considering the extensive clinical experience with this drug in the context of malaria, chloroquine may be an attractive candidate for prophylactic use. An overview for more information about potential candidates can be obtained from the following reviews [79, 165, 167–169].

15.13 Host Targeting Antivirals

Targeting of host factors to combat viral infections can provide an attractive antiviral strategy, given that host cell processes are often employed by multiple viruses and are less prone to the emergence of drug-resistant variants [170]. However, it is important to consider that cellular proteins do function in a complex network of interactions which could facilitate potential off-target effects and toxicity. Nevertheless, it has been shown previously that interference with host cell molecules or pathways that are required for viral replication can be a promising approach [171].

15.14 Host Cell Nucleoside Biosynthesis Inhibitors

The availability of nucleosides is one of the main requirements for virus replication. Host enzymes that are involved in host cell nucleoside biosynthesis such as inosine monophosphate dehydrogenase (IMPDH) and dihydroorotate dehydrogenase (DHODH) offer attractive targets for a broad-spectrum antiviral therapy. One of the most prominent drugs interfering with viral RNA synthesis is the guanosine analogue ribavirin (RBV, Table 15.3). Next to its direct inhibition of IMPDH, it is believed that RBV exerts in addition immunomodulatory effects as well as direct effects on the virus replication, including interference with RNA capping, polymerase inhibition, and lethal mutagenesis [216]. RBV has been used in combination therapies against HCV [217] and it has been shown to act as an antiviral in vitro against various flaviviruses as well as CHIKV; however, antiviral efficiency was often dependent on the cell line [175, 218, 219]. Despite its potent antiviral activity in vitro and in vivo, clinical studies often failed to show a potent antiviral effect against flaviviruses [175, 176, 220], indicating that it is not a suitable flaviviral therapeutic. There are only few studies testing the efficacy of RBV in vivo against CHIKV; however, some studies suggest that in mouse models, a combination therapy of RBV together with non-steroidal anti-inflammatory drugs or doxycycline can reduce CHIKV replication and attenuate its infectivity [177, 178]. In addition, RBV was tested in an observational study with 20 CHIKV-infected patients. Seven out of ten patients treated with RBV for seven days completely recovered in the monitored time period of four weeks follow-up [179]. Due to the lack of a placebo control group and the limited number of patients, this study gives a hint for the potential of RBV to treat CHIKV infection and further research should be conducted to confirm its efficacy.

In addition to RBV, several other IMPDH inhibitors have been developed, including 5-ethynyl-1- β -d-ribofuranosylimidazole-4-carboxamide (EICAR; 5-ethynylribavirin, Table 15.3) or the non-nucleoside mycophenolic acid, which both showed increased antiviral activity against selected flaviviruses [79]. However, whereas the former resulted in higher cytotoxicity [221], the latter exhibited immunosuppressive activity [222], rendering both not suitable for clinical implementation. Other IMPHD inhibitors that have been tested against various emerging viruses include merimepodib and azathioprine, the former showing an antiviral activity against several viruses including ZIKV and CHIKV [174] and the latter against ZIKV *in vitro* [172] (Table 15.3).

Next to IMPDH inhibitors, interference with the pyrimidine biosynthesis via inhibition of the DHODH has been exploited. Brequinar has been shown to be potently antiviral against selected flaviviruses *in vitro* (Table 15.3); however, due to its low therapeutic index, it was not approved for clinical use [223]. Other DHOD inhibitors include the indole derivative compounds A3 and 2-(4-benzyl-3-ethoxy-5-methyl-1H-pyrazol-1-yl)pyrimidine, of which the compound A3 showed antiviral activity against certain flaviviruses *in vitro* [180, 224]. Further studies will be needed to evaluate their potential as broad-spectrum antiviral compounds against emerging arboviruses *in vivo*.

Despite their potential as broad-acting antiviral compounds, nucleoside biosynthesis inhibitors often have severe limitations due to their often narrow therapeutic window as well as a potential for immunosuppressive effects which render them not suitable for treatment with expected coinfections, pregnancy, or extended (prophylactic) dosage regimens. Furthermore, resistance-associated mutations can occur due to various mechanisms requiring combination regimens with other antiviral compounds [79].

15.15 Host Cell Lipid Biosynthesis Inhibitors

Several studies have implicated a crucial role of the host cell lipid metabolism for flavivirus infection by an altered lipid homeostasis as well expansive membrane rearrangements within the infected cell [225]. Key processes that might be influenced by differential lipid biosynthesis could involve virion biogenesis, membrane remodeling, genome replication, as well as autophagy and apoptosis processes. Several host enzymes offer potential as antiviral targets as they regulate different aspects of the lipid metabolism which also benefits viral replication (extensively reviewed in [226]).

The sterol regulatory element-binding proteins (SREBP) have been analyzed as targets for antiviral intervention. Different SREBP inhibitors, namely nordihydroguaiaretic acid (NDGA), its derivate M_4N , as well as PF-429242 and fatostatin (Table 15.3) have been successfully used to reduce ZIKV, DENV, or WNV infection *in vitro* [181–183]. Another class of inhibitors targets activators of the adenosine monophosphate-activated protein kinase (AMPK), which constitutes one of the main cellular energy sensors responsible for glycolysis and lipid metabolism. Specific inhibitors of these activators (Table 15.3), e.g. the small molecule compound PF-06409577 as well as the licensed drug metformin or 5-aminoimidazole-4-carboxamide ribonucleotide (AICAR), have been shown to inhibit ZIKV, DENV, and WNV *in vitro* [184–186].

One of the key enzymes of the fatty acid metabolism, acetyl-CoA carboxylase (ACC), offers another target for antiviral interference. Several compounds that target this enzyme have been shown to affect replication of ZIKV, DENV, and WNV [187, 227]. Of note, one of these compounds, PF-05175157 (Table 15.3), has already been evaluated in clinical trials in healthy volunteers (ClinicalTrials.gov Identifier: NCT01433380) and for the treatment of diabetes mellitus (ClinicalTrials.gov Identifier: NCT0179263). PF-05175157 has further been tested against WNV in a mouse model, which resulted in a reduction of viral loads in serum and kidney [187]. Additionally, therapeutic intervention with the fatty acid synthase (FASN) has been shown to inhibit DNEV and WNV replication [228–230].

Other potential approaches could target the sphingolipid metabolism which has been suggested to play a functional role in flavivirus infection [231]. Indeed, treatment with an inhibitor of neutral sphingomyelinase-2, GW4869, was able to inhibit infection with WNV, USUV, and ZIKV in cell lines as well as primary human fetal astrocytes [232–234]. Of note, the same compound seems to have the opposite effect on alphavirus infection, as Sindbis virus infectivity was increased after GW4869 treatment [233].

Given the crucial role of host cholesterol during flavivirus infection [235], direct interference with the cholesterol metabolism offers another promising antiviral approach (Table 15.3). Statins (e.g. lovastatin, fluvastatin), a class of cholesterol biosynthesis inhibitors that are a safe and widely used class of drug for the treatment of hypercholesterolemia, have been shown to efficiently inhibit WNV [194, 236] and DNEV replication [191, 192, 237]. However, treatment with lovastatin in a randomized, double-blind, placebo-controlled trial found no evidence of a beneficial effect on any of the clinical manifestations or on dengue viremia [193]. Furthermore, lovastatin increased replication of Semliki Forest Virus in an *in vitro* model, another alphavirus demonstrating the virus-specificity of these inhibitors [194]. In contrast, intracellular cholesterol transport inhibitors, e.g. treatment with the FDA-approved antidepressant drug imipramine, have been shown to successfully reduce not only flavivirus infectivity, but also CHIKV in cell culture [195], demonstrating the

Inhibitor class	Target ^a	Compound ^b	Biological system	Viruses	
Nucleoside biosynthesis inhibitors	IMPDH	Ribavirin EICAR Mycophenolic acid Merimepodib Azathioprine	Cell culture Mice Clinical trial	ZIKV [172–174], DENV [175], WNV [176], CHIKV [174, 177–179]	
	DHODH	A3 (indole derivate)	Cell culture	DENV [180], WNV [180]	
Lipid biosynthesis inhibitors	SREBP	NDGA M₄N PF-429242 Fatostatin	Cell culture	ZIKV [181], DENV [182, 183], WNV [181]	
	АМРК	PF-06409577 Metformin AICAR	Cell culture	ZIKV [184, 185], DENV [185, 186], WNV [185]	
	ACC	PF-05175157	Cell culture Mice Clinical trial	ZIKV [187], DENV [187], WNV [181, 187]	
	Cholesterol metabolism	Fluvastatin Lovastatin Imipramine Cholesterol derivates	Cell culture Mice	ZIKV [188–190], DENV [191–193], WNV [194], CHIKV [195]	
	Sphingomyelinase-2	GW4869	Cell culture	WNV [228], ZIKV [227, 229]	
Kinase inhibitors	NAK family of serine/threonine kinases	Sunitinib Erlotinib	Cell culture Mice	ZIKV [196], DENV [196], WNV [196], CHIKV [196]	
	Src family kinases	Dasatinib Sarcatinib	Cell culture	DENV [197–199], WNV [200], CHIKV [17]	
	CDK	PHA-690509	Cell culture	ZIKV [201, 202]	
	Receptor tyrosine kinases (c-Abl)	Imatinib, GNF-2	Cell culture	DENV [140, 242, 243]	
	mTORC1/2	Dasatinib, Torin 1	Cell culture	CHIKV [244]	
Protein metabolism inhibitors	α-Glucosidase	Iminosugars	Cell culture Mice Clinical trial	DENV [203-206]	

 Table 15.3
 Overview of different classes of host-targeting antivirals.

(Continued)

Inhibitor class	Target ^a	Compound ^b	Biological system	Viruses
	α-glucosidase	Castanospermine, Celgosivir	Cell culture, mice, Clinical trial	DENV [199, 200]
	UPP	Bortezomib	Cell culture Mice	DENV [209], ZIKV [172, 210]
	Cyclophilins (CyPs)	Cyclosporine A (CsA)	Cell culture	WNV [207], DENV [207], YFV [207, 208], JEV [254]
Endocytosis and membrane fusion inhibitors	Membrane fusion	Arbidol	Cell culture	ZIKV [211, 212], WNV [211], CHIKV [213]
	Endosomal acidification	Chloroquine Ammonium chloride Bafilomycin A1 Niclosamide	Cell culture	DENV [214], ZIKV [95], CHIKV [215]

Table 1	5.3	(Continued)
---------	-----	-------------

Summarized are tested compounds for different targets, the biological system used, and the viruses.

a) IMPDH, inosine monophosphate dehydrogenase; DHODH, dihydroorotate dehydrogenase; SREBP, sterol regulatory element-binding protein; AMPK, AMP-activated protein kinase; ACC, acetyl-CoA-carboxylase; CDK, cyclin-dependent kinases; UPP, ubiquitin-proteasome pathway.

 EICAR, 5-ethynyl-1-beta-D-ribofuranoysylimidazole-4; NDGA, nordihydroguaiaretic acid; M4N, tetra-o-methyl nordihydroguaiaretic acid; AICAR, 5-aminoimidazole-4-carboxamide-1-β-D-ribofuranoside.

potential of such drug candidates for a broad anti-arboviral treatment. Finally, cholesterol derivatives including 25-hydroxycholesterol (25-HC) and 7-ketocholesterol (7-KC) have shown promising antiviral properties against various viruses, including ZIKV *in vitro* and in a mouse model [188–190].

Importantly, even though most of the studies mentioned above are still in the initial stages of experimental validation and have mainly been performed *in vitro*, several of the compounds target key processes that are involved in a variety of human diseases, including obesity, metabolic syndrome, type II diabetes, and cancer. In consequence, some of the drugs mentioned above have already undergone diverse phases of clinical trials or are licensed for use in humans, facilitating drug repurposing [238–242].

15.16 Host Kinase Inhibitors

Phosphorylation of viral proteins mediated by cellular kinases is an important post-translational modification to regulate and ensure viral replication. Therefore, known host kinase inhibitors are potential candidates to be used as broad-spectrum antivirals in the future [243]. For example, two approved anticancer drugs, sunitinib and erlotinib (Table 15.3), have been tested for their antiviral activity against flaviviruses as well as the alphavirus CHIKV [196]. These drugs inhibit the host kinase adaptor protein 2 (AP2)-associated protein kinase 1 (AAK1) and cyclin G-associated kinase (GAK), both kinases belonging to the Num-associated kinases (NAK) family of serine/threonine kinases. Since these kinases are involved in clathrin-mediated trafficking, they might affect intracellular viral trafficking and thus lower viral infectivity [244, 245]. Indeed, antiviral activity has been shown in vitro against DENV, WNV, ZIKV, and CHIKV at micromolar concentrations (EC50 values between 0.51 and $6.28\,\mu$ M). Furthermore, the combination of both drugs was beneficial in reducing viremia and the overall outcome of DENV infection in a mouse model at a dose of 30 mg/kg [196]. Another host kinase family known to play a role during viral infections are the Src family kinases (SFKs) [246]. DENV replication has been found to depend on SFKs such as Fyn, Lyn, c-Src, and CSK [197-199], whereas for WNV, the kinase c-Yes seems to be important [200]. Particularly for DENV infection, two approved drugs, namely dasatinib and sarcatinib (Table 15.3), inhibited viral assembly in an in vitro study (EC₉₀ values 12.2 and 4.7 µM, respectively) [199]. In addition, both inhibitors target the cellular Abelson tyrosine kinase (c-Abl), which belongs to the Abl family of nonreceptor kinases, a kinase involved in the replication of many viruses including DENV. In cell culture assays at a concentration of 15μ M, the approved anticancer drug imatinib showed anti-DENV activity and GNF-2, an allosteric inhibitor of c-Abl, developed as anticancer drug was even more effective against DENV infection in this study [145, 247, 248]. Recently, the effect of SFK inhibitors has been also tested against alphaviruses including CHIKV infection in vitro. Both dasatinib and Torin 1, an inhibitor of mTORC1/2, successfully blocked CHIKV replication during translation of structural proteins at concentrations of 100 nM [249]. Cyclindependent kinases (CDKs) are broadly used by many viruses to manipulate the host cell cycle [250]. Since flaviviruses do not express any CDKs, CDK inhibitors seem to impair host functions essential for viral replication. For example, the compound PHA-690509 (Table 15.3) was shown to inhibit ZIKV infection (IC₅₀ = $0.9 \,\mu$ M) by analyzing NS1 expression in vitro [201, 202]. Overall, host kinase inhibitors may be promising in their capability to be used as antivirals. There is a broad potency in repurposing already approved host kinase inhibitors and the effort to balance effective antiviral activity with reduced cell toxicity to speed up the evaluation for clinical use should be made.

15.17 Protein Metabolism Inhibitors

Viruses hijack the host protein metabolism for a successful replication and translation of viral proteins. Key components include α -glucosidase, cyclophilins, and the ubiquitin-proteasome pathway (UPP). α -Glucosidase catalyzes the removal of glucose units from N-linked oligosaccharides and has been shown to participate in the folding and maturation of flaviviral glycoproteins [251]. Indeed, several α -glucosidase inhibitors have been shown to act as antivirals against several viruses, including flaviviruses [252]. Importantly, they have been shown to display a high genetic resistance barrier toward the emergence of escape mutations in a mouse model during DENV infection [203]. Among the most promising inhibitors are iminosugars and their derivates (Table 15.3).

The iminosugar castanospermine and its derivate celgosivir have been shown to efficiently inhibit DENV at a low micromolar IC₅₀ *in vitro* and *in vivo* at doses of 10–250 mg/ kg and 7.5 or 75 mg/kg, respectively (reviewed in [204]). In a randomized, double-blind, placebo-controlled trial (CELADEN; ClinicalTrials.gov Identifier: NCT01619969), celgosivir led to a mean virological log reduction (VLR = -0.22) of DENV from baseline for days 2, 3, and 4 compared to the placebo group; however, the difference was not statistically significant [205], and the study failed in reaching its primary endpoints of lowering viremia or fever [206]. A second phase 2 clinical trial with a revised dosing regimen is currently ongoing (ClinicalTrials.gov identifier: NCT02569827). One of the main disadvantages of iminosugars are the generally high doses, which are required and which can be associated with a high toxicity. Furthermore, such compounds seem to have a weak activity during the postinfection period. Whether these issues can be overcome at all by derivatization or combination with other antivirals will need to be shown in further studies. In addition, they might be suitable candidates for an early or even prophylactic treatment [79].

Cyclophilins (CyPs) facilitate proper protein folding and have been shown to play a crucial role in the replication cycle of various viruses [253]. Cyclosporine A (CsA), an 11-amino-acid cyclic peptide inhibitor of CyPs, has been shown to act as antiviral against WNV (20μ M), DENV (8μ M), YFV (8μ M), and JEV ($\sim 5\mu$ M) *in vitro* [207, 208, 254]. However, no antiviral activity of CsA against CHIKV could be observed [255]. Besides, CsA other CyPs inhibitors might offer potential as anti-flaviviral compounds. Alisporivir, a cyclophilin inhibitor with pan-genotypic anti-HCV activity and a high barrier to viral resistance, has been shown to be effective as IFN-free option for the treatment of HCV in phase 2 (ClinicalTrials.gov Identifier: NCT01215643) [256]. *In vitro* studies did show an antiviral activity against selected tick-borne flaviviruses, including tick-borne encephalitis virus [TBEV] at low μ M concentrations (0.5–10 μ M); however, no effect on WNV replication [257]. Nevertheless, it is important to keep in mind that the dependency of CypA during virus replication can be cell-type specific, thereby resulting in inconclusive results *in vitro* which should be validated *in vivo*.

The host UPP has been implicated as an important host factor during positive-stranded RNA viral infections [258]. It is speculated that viruses from several families are able to reprogram the UPP and thereby alter the cellular environment toward viral replication, whereas inhibition of the UPP can interfere with viral propagation. Importantly, several drugs that inhibit the function of the proteasome, a major player of the UPP, have been licensed for therapeutic use. Bortezomib (Table 15.3), a licensed proteasome inhibitor to treat multiple myeloma and mantle cell lymphoma, has been shown to reduce DENV titers of all four serotypes at low nanomolar drug concentrations in primary monocytes [209]. In addition, in vitro assays revealed bortezomib as an antiviral drug against ZIKV [172] and against alphaviruses [259]. Despite its moderate toxicity in vitro [172], it has been shown to reduce viral load and signs of pathology in ZIKV-infected mice [210]. Finally, the drug has been shown to reduce JEV-induced lethality in mice, to alleviate suffering in JEV-infected mice and reduce the damage in brains caused by JEV infection [260]. Other inhibitors of the proteasome, including MG132 and USP14, have shown to have an antiviral activity against flaviviruses and alphaviruses in vitro [261-265], highlighting proteasome inhibitors as potential anti-arboviral compounds.

15.18 Endocytosis and Membrane Fusion Inhibitors

Flaviviruses and CHIKV enter host cells through clathrin-mediated endocytosis [266-269]. Conformational changes in viral envelope proteins triggered during endosome acidification mediate membrane fusion and subsequent release of the viral genome from the capsid [71, 270, 271]. Each individual step during the virus entry/membrane fusion process is critical to successfully establish an infection and thus represents a potential druggable target. One example is arbidol (ARB, also known as umifenovir, Table 15.3), a broad-spectrum antiviral compound and clinically approved in Russia and China for prophylaxis and treatment of influenza virus infections [272, 273]. It intercalates with membrane lipids to prevent fusion events between virus particles and host membranes and thereby blocks virus entry into host cells [274]. In vitro studies revealed antiviral activities against ZIKV (IC₅₀ 11–15 μ M), CHIKV (IC₅₀ ~25 μ M), and WNV $(EC_{50} \sim 18 \mu M)$, indicating that ARB may represent a potential candidate for treatment and prophylaxis of emerging arbovirus infections with a well-established safety profile [211–213]. Various lysosomotropic agents, such as chloroquine, ammonium chloride, bafilomycin A1, and niclosamide (Table 15.3), have been shown to exert direct antiviral effects on several emerging arbovirus infections in vitro [96, 214, 215, 275], implying that the blockade of endosomal acidification during endocytosis can be an effective strategy to block viral entry. However, although inhibitors perturbing cellular processes such as endocytosis and endosomal acidification have been valuable tools to study various aspects of the viral life cycle, translating these findings into clinical use remains challenging due to frequently observed significant cytotoxic effects. Nonetheless, results from genome wide screens may identify novel targets in the endosomal pathway without deleterious effects on the host cell [268, 276].

15.19 Conclusion and Future Perspectives

Arbovirus infections constitute a global health burden and have a long history of infecting humans. They result in regular pandemics as well as in an increasing frequency of autochthonous transmission. However, they are mostly considered as neglected tropical diseases. That means they are outside the purview of the Global Fund and its related programs which mostly concentrate on HIV, tuberculosis (TB), and malaria [277]. Consequently, only two candidate vaccines for DENV and YFV have been approved, while no specific antiviral therapeutics to prevent or treat arbovirus infections have been licensed yet. Notwithstanding, several approaches have been employed in recent years, including testing of drugs with known antiviral activity against other viruses or screening of compound libraries comprising hundreds to thousands of bioactive compounds (see Table 15.4 for selected compounds). In this context, repurposing of drugs that have already gone through several steps of drug approval by regulatory agencies may offer potential as fast intervention strategies in cases of virus (re-)emergence. In addition, novel approaches including structure-based drug design, computational approaches using docking algorithms and virtual screening, as well as traditional techniques of HTS with combinatorial chemistry can increase the efficiency and speed of drug discovery. Along this line, future research should

Table 15.4 Chemical names and formula of selected potential antiviral drugs.

Name	Chemical name	Formula	CAS number
7DMA	7-Deaza-2'-C-methyladenosine	$\mathrm{C}_{12}\mathrm{H}_{16}\mathrm{N}_{4}\mathrm{O}_{4}$	443642-29-3
BCX4430	(2S, 3S, 4R, 5R) - 2 - (4 - amino - 5H - pyrrolo[3, 2 - d] pyrimidin - 7 - yl) - 5 - (hydroxymethyl) - 3, 4 - pyrrolidinediol - 2, 3 - y - y - y - y - y - y - y - y - y -	$C_{11}H_{15}N_5O_3$	222631-44-9
Sofosbuvir	$\label{eq:loss_start} Isopropyl-(2S)-2-(\{[(2R,3R,4R,5R)-5-(2,4-dioxopyrimidin-1-y])-4-fluor-3-hydroxy-4-methyl-tetrahydrofuran-2-yl]methoxy-phonoxy-phosphoryl]amino)propanoat$	$C_{22}H_{29}FN_{3}O_{9}P$	1190307-88-0
Temoporfin	3,3',3'',3'''-(2,3-Dihydroporphyrin-5,10,15,20-tetrayl)tetraphenol	$C_{44}H_{32}N_4O_4$	122341-38-2
Erythrosin B	Dinatrium-2-(2,4,5,7-tetraiodo-6-oxido-3-oxo-xanthen-9-yl)benzoate	$\mathrm{C_{20}H_6I_4Na_2O_5}$	16423-68-0
Lycorine	1,2,4,5,12b,12c-Hexahydro-7H-[1,3]dioxolo[4,5-j]pyrrolo[3,2,1-de]phenanthridine-1,2-diol	$\mathrm{C_{16}H_{17}NO_4}$	476-28-8
4-HPR	15-[(4-Hydroxyphenyl)amino]retinal	C ₂₆ H ₃₃ NO ₂	65646-68-6
Ivermectin	22,23-Dihydroavermectin B1	$C_{48}H_{74}O_{14}$	70161-11-4
Ribavirin	1-[(2R,3R,4R,5R)-3,4-Dihydroxy-5-hydroxymethyl-oxolan-2-yl]-1,2,4-triazol-3-carboxamid	$\mathrm{C_8H_{12}N_4O_5}$	36791-04-5
EICAR	5-Ethynyl-1-beta-D-ribofuranosylimidazole-4-carboxamide	$C_{11}H_{16}N_3O_{14}P_3\\$	214827-88-0
Mycophenolic acid	6-(4-Hydroxy-6-methoxy-7-methyl-3-oxo-1H-2-benzofuran-5-yl)-4-methylhex-4-enoic acid	$C_{17}H_{20}O_{6}$	24280-93-1
Merimepodib	[(3S)-Oxolan-3-yl] N-[[3-[[3-methoxy-4-(1,3-oxazol-5-yl)phenyl]carbamoylamino]phenyl]methyl] carbamate	$C_{23}H_{24}N_4O_6$	198821-22-6
Azathioprine	6-(3-Methyl-5-nitroimidazol-4-yl)sulfanyl-7H-purine	$C_9H_7N_7O_2S$	446-86-6
NDGA	4-[4-(3,4-Dihydroxyphenyl)-2,3-dimethylbutyl]benzene-1,2-diol	$C_{18}H_{22}O_4$	500-38-9
M4N	4-[(2S,3R)-4-(3,4-Dimethoxyphenyl)-2,3-dimethylbutyl]-1,2-dimethoxybenzene	$C_{22}H_{30}O_4$	24150-24-1
PF-429242	$\label{eq:linear} 4-(Diethylaminomethyl)-N-[2-(2-methoxyphenyl)ethyl]-N-[(3R)-pyrrolidin-3-yl] benzamide$	C25H35N3O2	947303-87-9
Fatostatin	4-(4-Methylphenyl)-2-(2-propylpyridin-4-yl)-1,3-thiazole	$\mathrm{C}_{18}\mathrm{H}_{18}\mathrm{N}_{2}\mathrm{S}$	298197-04-3
PF-06409577	6-Chloro-5-[4-(1-hydroxycyclobutyl)phenyl]-1H-indole-3-carboxylic acid	C ₁₉ H ₁₆ ClNO ₃	1467057-23-3
Metformin	3-(Diaminomethylidene)-1,1-dimethylguanidine	$\mathrm{C_4H_{11}N_5}$	657-24-9
AICAR	5-Amino-1-[(2R,3R,4S,5R)-3,4-dihydroxy-5-(hydroxymethyl)oxolan-2-yl]imidazole-4-carboxamide	$\mathrm{C_9H_{14}N_4O_5}$	2627-69-2

Name	Chemical name	Formula	CAS number
PF-05175157	l'-(2-Methyl-3H-benzimidazole-5-carbonyl)-1-propan-2-ylspiro[4,6-dihydroindazole-5,4'-piperidine]-7-one	$C_{23}H_{27}N_5O_2$	1301214-47-0
Fluvastatin	(E,3R,5S)-7-[3-(4-Fluorophenyl)-1-propan-2-ylindol-2-yl]-3,5-dihydroxyhept-6-enoic acid	$C_{24}H_{26}FNO_4$	93957-54-1
Lovastatin	[(15,3R,75,85,8aR)-8-[2-[(2R,4R)-4-Hydroxy-6-oxooxan-2-yl]ethyl]-3,7-dimethyl-1,2,3,7,8,8a- hexahydronaphthalen-1-yl] (2S)-2-methylbutanoate	$C_{24}H_{36}O_{5}\\$	75330-75-5
Imipramine	3-(5,6-Dihydrobenzo[b][1]benzazepin-11-yl)-N,N-dimethylpropan-1-amine	$C_{19}H_{24}N_2$	50-49-7
Sunitinib	N-[2-(Diethylamino)ethyl]-5-[(Z)-(5-fluoro-2-oxo-1H-indol-3-ylidene) methyl]-2,4-dimethyl1-1H-pyrrole-3-carboxamide	$C_{22}H_{27}FN_4O_2$	557795-19-4
Erlotinib	N-(3-Ethynylphenyl)-6,7-bis(2-methoxyethoxy)quinazolin-4-amine	C22H23N3O4	183321-74-6
Dasatinib	N-(2-Chloro-6-methylphenyl)-2-[[6-[4-(2-hydroxyethyl)piperazin-1-yl]-2-methylpyrimidin-4-yl] amino]-1,3-thiazole-5-carboxamide	C22H26CIN7O2S	302962-49-8
Sarcatinib	N-(5-Chloro-1,3-benzodioxol-4-yl)-7-[2-(4-methylpiperazin-1-yl)ethoxy]-5-(oxan-4-yloxy) quinazolin-4-amine	$C_{27}H_{32}ClN_5O_5\\$	379231-04-6
PHA-690509	(2S)-2-(4-Acetamidophenyl)-N-(5-propan-2-yl-1,3-thiazol-2-yl)propanamide	$C_{17}H_{21}N_3O_2S$	492445-28-0
Cyclosporine A	(35,65,95,12R,155,185,215,245,305,3 3\$)-30-Ethyl-33-[(C; I, R,2R)-1-hydroxy-2-methylhex-4-enyl]-1,4,7,10,12,15,19,25,28-nonamethyl- 6,9,18,24-tetrakis(2-methylpropyl)-3,21-di(propan-2-yl)-1,4,7,10,13,16,19,22,25,28,31- undecazacyclottririacontane-2,5,8,11,14,17,20,23,26,29,32-undecone	$C_{62}H_{111}N_{11}O_{12}$	59865-13-3
Bortezomib	[(1R)-3-Methyl-1-[[(2S)-3-phenyl-2-(pyrazine-2-carbonylamino)propanoyl]amino]butyl]boronic acid	$C_{19}H_{25}BN_4O_4$	179324-69-7
Arbidol	Ethyl 6-bromo-4-[(dimethylamino)methyl]-5-hydroxy-1-methyl-2-(phenylsulfanylmethyl) indole-3-carboxylate	$\mathrm{C}_{22}\mathrm{H}_{25}\mathrm{BrN}_{2}\mathrm{O}_{3}\mathrm{S}$	131707-25-0
Chloroquine	4-N-(7-Chloroquinolin-4-yl)-1-N,1-N-diethylpentane-1,4-diamine	$C_{18}H_{26}ClN_3$	56598-66-4
Ammonium chloride	Ammonium chloride	NH4Cl	12125-02-9
Bafilomycin A1	(3Z,5E,7R,8S,9S,11E,13E,15S,16R)-16-[(2S,3R,4S)-4-[(2R,4R,5S,6R)-2,4-Dihydroxy-5-methyl-6-propan-2-yloxan-2-yl]-3-hydroxypentan-2-yl]-8-hydroxy-3,15-dimethoxy-5,7,9,11-tetramethyl-1-oxacyclohexadeca-3,5,11,13-tetraen-2-one	$C_{35}H_{58}O_9$	88899-55-2
Niclosamide	5-Chloro-N-(2-chloro-4-nitrophenyl)-2-hydroxybenzamide	$\mathrm{C_{13}H_8Cl_2N_2O_4}$	50-65-7

focus on understanding the replication mechanisms and identifying virus-host interactions to facilitate the different novel approaches regarding drug design. In order to develop successful therapeutic approaches to treat emerging arbovirus infections, a combination approach utilizing antivirals and host response-directed countermeasures could be employed. So far, most of the antiviral candidates have been evaluated *in vitro* only, some of them have also been tested in animal models. However, only few candidates have advanced into clinical trials (i.e. BCX4430, chloroquine, PF-05175157, celgosivir, alisporivir), highlighting the need for further research and efforts toward developing an effective treatment of these neglected diseases.

References

- **1** Karabatsos, N. (1978). Supplement to international catalogue of arboviruses including certain other viruses of vertebrates. *The American Journal of Tropical Medicine and Hygiene* 27: 372–440; https://doi.org/10.4269/ajtmh.1978.27.372.
- **2** Kolawole, O.M., Seriki, A.A., Irekeola, A.A., and Ogah, J.I. (2018). The neglect and fast spread of some arboviruses: a note for healthcare providers in Nigeria. *Diseases* 6; https://doi.org/10.3390/diseases6040099.
- 3 Kuno, G. and Chang, G.-J.J. (2005). Biological transmission of arboviruses: reexamination of and new insights into components, mechanisms, and unique traits as well as their evolutionary trends. *Clinical Microbiology Reviews* 18: 608–637; https://doi.org/10.1128/CMR.18.4.608-637.2005.
- **4** Kenney, J.L. and Brault, A.C. (2014). The role of environmental, virological and vector interactions in dictating biological transmission of arthropod-borne viruses by mosquitoes. *Advances in Virus Research* 89: 39–83; https://doi.org/10.1016/B978-0-12-800172-1.00002-1.
- **5** Gubler, D.J. (1996). The global resurgence of arboviral diseases. *Transactions of the Royal Society of Tropical Medicine and Hygiene* 90: 449–451; https://doi.org/10.1016/ s0035-9203(96)90286-2.
- **6** Gubler, D.J. (2002). The global emergence/resurgence of arboviral diseases as public health problems. *Archives of Medical Research* 33: 330–342.
- 7 Gould, E., Pettersson, J., Higgs, S. et al. (2017). Emerging arboviruses: why today? *One Health (Amsterdam, Netherlands)* 4: 1–13; https://doi.org/10.1016/j.onehlt.2017.06.001.
- **8** Barrows, N.J. et al. (2018). Biochemistry and molecular biology of flaviviruses. *Chemical Reviews* 118: 4448–4482; https://doi.org/10.1021/acs.chemrev.7b00719.
- **9** Schwartz, O. and Albert, M.L. (2010). Biology and pathogenesis of chikungunya virus. Nature reviews. *Microbiology* 8: 491–500; https://doi.org/10.1038/nrmicro2368.
- 10 Dick, G.W.A. (1952). Zika virus. II. Pathogenicity and physical properties. *Transactions of the Royal Society of Tropical Medicine and Hygiene* 46: 521–534; https://doi.org/10.1016/0035-9203(52)90043-6.
- **11** Baud, D., Gubler, D.J., Schaub, B. et al. (2017). An update on Zika virus infection. *Lancet* (*London, England*) 390: 2099–2109; https://doi.org/10.1016/S0140-6736(17)31450-2.
- 12 Liu, Z.-Y., Shi, W.-F., and Qin, C.-F. (2019). The evolution of Zika virus from Asia to the Americas. Nature reviews. *Microbiology* 17: 131–139; https://doi.org/10.1038/ s41579-018-0134-9.

- **13** Pierson, T.C. and Diamond, M.S. (2018). The emergence of Zika virus and its new clinical syndromes. *Nature* 560: 573–581; https://doi.org/10.1038/s41586-018-0446-y.
- 14 Barbi, L., Coelho, A.V.C., Alencar, L.C.A.D., and Crovella, S. (2018). Prevalence of Guillain–Barré syndrome among Zika virus infected cases: a systematic review and meta-analysis. *The Brazilian Journal of Infectious Diseases: An Official Publication of the Brazilian Society of Infectious Diseases* 22: 137–141; https://doi.org/10.1016/j. bjid.2018.02.005.
- 15 Monath, T.P. (2019). The Arboviruses: Epidemiology and Ecology. CRC Press.
- 16 Wasserman, S., Tambyah, P.A., and Lim, P.L. (2016). Yellow fever cases in Asia: primed for an epidemic. *International Journal of Infectious Diseases: IJID: Official Publication of the International Society for Infectious Diseases* 48: 98–103; https://doi.org/10.1016/j. ijid.2016.04.025.
- Bryant, J.E., Holmes, E.C., and Barrett, A.D.T. (2007). Out of Africa: a molecular perspective on the introduction of yellow fever virus into the Americas. *PLoS Pathogens* 3: e75; https://doi.org/10.1371/journal.ppat.0030075.
- **18** Monath, T.P. (2001). Yellow fever: an update. *The Lancet Infectious Diseases* 1: 11–20; https://doi.org/10.1016/S1473-3099(01)00016-0.
- 19 Wilder-Smith, A. and Monath, T.P. (2017). Responding to the threat of urban yellow fever outbreaks. *The Lancet Infectious Diseases* 17: 248–250; https://doi.org/10.1016/S1473-3099(16)30588-6.
- **20** Eckert, J. (1993). In the days of the epidemic: the 1793 yellow fever outbreak in Philadelphia as seen by physicians. *Transactions & Studies of the College of Physicians of Philadelphia* 15: 31–38.
- 21 World Health Organization. WHO|Yellow fever: a current threat. HYPERLINK "https:// urldefense.com/v3/__https://www.who.int/csr/disease/yellowfev/impact1/en/__;!!N11eV2 iwtfs!8ou4E8NA0_0ys_3DaWdRjq72djj0OltUpOL28K5aOgvkINrcxe0oKZ6hHOU\$" https://www.who.int/csr/disease/yellowfev/impact1/en/ (accessed 5 November 2019).
- **22** Bhatt, S. et al. (2013). The global distribution and burden of dengue. *Nature* 496: 504–507; https://doi.org/10.1038/nature12060.
- 23 Simmons, C.P., Farrar, J.J., van Nguyen, V.C., and Wills, B. (2012). Dengue. *The New England Journal of Medicine* 366: 1423–1432; https://doi.org/10.1056/NEJMra1110265.
- **24** Carrington, L.B. and Simmons, C.P. (2014). Human to mosquito transmission of dengue viruses. *Frontiers in Immunology* 5: 290; https://doi.org/10.3389/fimmu.2014.00290.
- **25** Gubler, D.J. (2006). Dengue/dengue haemorrhagic fever: history and current status. *Novartis Foundation Symposium* 277: 3–16; discussion 16–22, 71–3, 251–3.
- **26** Messina, J.P. et al. (2014). Global spread of dengue virus types: mapping the 70 year history. *Trends in Microbiology* 22: 138–146; https://doi.org/10.1016/j.tim.2013.12.011.
- **27** Rigau-Pérez, J.G. et al. (1998). Dengue and dengue haemorrhagic fever. *Lancet (London, England)* 352: 971–977; https://doi.org/10.1016/s0140-6736(97)12483-7.
- **28** Grange, L. et al. (2014). Epidemiological risk factors associated with high global frequency of inapparent dengue virus infections. *Frontiers in Immunology* 5: 280; https://doi.org/10.3389/fimmu.2014.00280.
- **29** Wilder-Smith, A., Vannice, K.S., Hombach, J. et al. (2016). Population perspectives and World Health Organization recommendations for CYD-TDV dengue vaccine. *The Journal of Infectious Diseases* 214: 1796–1799; https://doi.org/10.1093/infdis/jiw341.

- **392** 15 Novel Antiviral Strategies Against Emerging Arbovirus Infections
 - **30** FDA (2019). First FDA-approved vaccine for the prevention of dengue disease in endemic regions.https://www.fda.gov/news-events/press-announcements/first-fda-approved-vaccine-prevention-dengue-disease-endemic-regions (accessed 5 November 2019).
 - **31** Clark, M.B. and Schaefer, T.J. (2019). StatPearls. *West Nile Virus* (Treasure Island (FL)). (accessed November 2019).
 - **32** Petersen, L.R., Brault, A.C., and Nasci, R.S. (2013). West Nile virus: review of the literature. *JAMA* 310: 308–315; https://doi.org/10.1001/jama.2013.8042.
 - 33 Chancey, C., Grinev, A., Volkova, E., and Rios, M. (2015). The global ecology and epidemiology of West Nile virus. *BioMed Research International* 2015: 376230; https://doi. org/10.1155/2015/376230.
 - **34** Petersen, L.R. and Hayes, E.B. (2008). West Nile virus in the Americas. *The Medical Clinics* of North America 92: 1307–1322, ix; https://doi.org/10.1016/j.mcna.2008.07.004.
 - 35 Staples, J.E., Breiman, R.F., and Powers, A.M. (2009). Chikungunya fever: an epidemiological review of a re-emerging infectious disease. *Clinical Infectious Diseases: An Official Publication of the Infectious Diseases Society of America* 49: 942–948; https://doi.org/10.1086/605496.
 - 36 Robinson, M.C. (1955). An epidemic of virus disease in Southern Province, Tanganyika Territory, in 1952–53. I. Clinical features. *Transactions of the Royal Society of Tropical Medicine and Hygiene* 49: 28–32; https://doi.org/10.1016/0035-9203(55)90080-8.
 - Higgs, S. (2006). The 2005–2006 Chikungunya epidemic in the Indian Ocean. Vector Norne and Zoonotic Diseases (Larchmont, N.Y.) 6: 115–116; https://doi.org/10.1089/vbz.2006.6.115.
 - **38** Leparc-Goffart, I., Nougairede, A., Cassadou, S. et al. (2014). Chikungunya in the Americas. *Lancet (London, England)* 383: 514; https://doi.org/10.1016/S0140-6736(14)60185-9.
 - **39** Nasci, R.S. (2014). Movement of chikungunya virus into the Western hemisphere. *Emerging Infectious Diseases* 20: 1394–1395; https://doi.org/10.3201/eid2008.140333.
 - **40** Pinheiro, T.J., Guimarães, L.F., Silva, M.T.T., and Soares, C.N. (2016). Neurological manifestations of Chikungunya and Zika infections. *Arquivos de Neuro-Psiquiatria* 74: 937–943; https://doi.org/10.1590/0004-282X20160138.
 - **41** Das, T. et al. (2010). Chikungunya fever: CNS infection and pathologies of a re-emerging arbovirus. *Progress in Neurobiology* 91: 121–129; https://doi.org/10.1016/j. pneurobio.2009.12.006.
 - **42** Amaral, J.K., Bilsborrow, J.B., and Schoen, R.T. (2019). Brief report: the disability of chronic chikungunya arthritis. *Clinical Rheumatology* 38: 2011–2014; https://doi.org/10.1007/s10067-019-04529-x.
 - **43** Borgherini, G. et al. (2008). Persistent arthralgia associated with chikungunya virus: a study of 88 adult patients on reunion island. *Clinical Infectious Diseases: An Official Publication of the Infectious Diseases Society of America* 47: 469–475; https://doi.org/10.1086/590003.
 - **44** Sugiura, S. and Ohno, R. (1990). T cell blastogenesis by mitogens. *Nihon Rinsho. Japanese Journal of Clinical Medicine* 48 (Suppl): 723–725.
 - **45** Tandale, B.V. et al. (2009). Systemic involvements and fatalities during chikungunya epidemic in India, 2006. *Journal of Clinical Virology: The Official Publication of the Pan American Society for Clinical Virology* 46: 145–149; https://doi.org/10.1016/j. jcv.2009.06.027.

- 46 Economopoulou, A. et al. (2009). Atypical chikungunya virus infections: clinical manifestations, mortality and risk factors for severe disease during the 2005–2006 outbreak on Réunion. *Epidemiology and Infection* 137: 534–541; https://doi.org/10.1017/ S0950268808001167.
- **47** Contopoulos-Ioannidis, D., Newman-Lindsay, S., Chow, C., and LaBeaud, A.D. (2018). Mother-to-child transmission of chikungunya virus: a systematic review and metaanalysis. *PLoS Neglected Tropical Diseases* 12: e0006510; https://doi.org/10.1371/journal. pntd.0006510.
- **48** Ramful, D. et al. (2007). Mother-to-child transmission of chikungunya virus infection. *The Pediatric Infectious Disease Journal* 26: 811–815; https://doi.org/10.1097/ INF.0b013e3180616d4f.
- **49** Tsetsarkin, K.A., Vanlandingham, D.L., McGee, C.E., and Higgs, S. (2007). A single mutation in chikungunya virus affects vector specificity and epidemic potential. *PLoS Pathogens* 3: e201; https://doi.org/10.1371/journal.ppat.0030201.
- **50** Villamil-Gómez, W.E., González-Camargo, O., Rodriguez-Ayubi, J. et al. (2016). Dengue, chikungunya and zika co-infection in a patient from Colombia. *Journal of Infection and Public Health* 9: 684–686; https://doi.org/10.1016/j.jiph.2015.12.002.
- 51 Göertz, G.P., Vogels, C.B.F., Geertsema, C. et al. (2017). Mosquito co-infection with zika and chikungunya virus allows simultaneous transmission without affecting vector competence of Aedes aegypti. *PLoS Neglected Tropical Diseases* 11: e0005654; https://doi. org/10.1371/journal.pntd.0005654.
- 52 Stokes, A., Bauer, J.H., and Hudson, N.P. (2001). The transmission of yellow fever to Macacus rhesus. 1928. *Reviews in Medical Virology* 11: 141–148; https://doi.org/10.1002/ rmv.311.
- 53 Halstead, S.B. and Cohen, S.N. (2015). Dengue hemorrhagic fever at 60 years: early evolution of concepts of causation and treatment. *Microbiology and Molecular Biology Reviews: MMBR* 79: 281–291; https://doi.org/10.1128/MMBR.00009-15.
- 54 Smithburn, K.C., Hughes, T.P., Burke, A.W., and Paul, J.H.A. (1940). Neurotropic virus isolated from the blood of a native of Uganda 1. *The American Journal of Tropical Medicine and Hygiene* s1–20: 471–492; https://doi.org/10.4269/ajtmh.1940.s1-20.471.
- 55 Dick, G.W.A., Kitchen, S.F., and Haddow, A.J. (1952). Zika virus. I. Isolations and serological specificity. *Transactions of the Royal Society of Tropical Medicine and Hygiene* 46: 509–520; https://doi.org/10.1016/0035-9203(52)90042-4.
- 56 Slon Campos, J.L., Mongkolsapaya, J., and Screaton, G.R. (2018). The immune response against flaviviruses. *Nature Immunology* 19: 1189–1198; https://doi.org/10.1038/ s41590-018-0210-3.
- 57 Roby, J.A., Setoh, Y.X., Hall, R.A., and Khromykh, A.A. (2015). Post-translational regulation and modifications of flavivirus structural proteins. *The Journal of General Virology* 96: 1551–1569; https://doi.org/10.1099/vir.0.000097.
- **58** Zhang, X. et al. (2017). Structures and functions of the envelope glycoprotein in flavivirus infections. *Viruses* 9; https://doi.org/10.3390/v9110338.
- **59** Samsa, M.M. et al. (2009). Dengue virus capsid protein usurps lipid droplets for viral particle formation. *PLoS Pathogens* 5: e1000632; https://doi.org/10.1371/journal.ppat.1000632.
- **60** Rastogi, M., Sharma, N., and Singh, S.K. (2016). Flavivirus NS1: a multifaceted enigmatic viral protein. *Virology Journal* 13: 131; https://doi.org/10.1186/s12985-016-0590-7.

- **394** 15 Novel Antiviral Strategies Against Emerging Arbovirus Infections
 - **61** Gopala Reddy, S.B., Chin, W.-X., and Shivananju, N.S. (2018). Dengue virus NS2 and NS4: minor proteins, mammoth roles. *Biochemical Pharmacology* 154: 54–63; https://doi.org/10.1016/j.bcp.2018.04.008.
 - **62** Lei, J. et al. (2016). Crystal structure of Zika virus NS2B-NS3 protease in complex with a boronate inhibitor. *Science (New York, N.Y.)* 353: 503–505; https://doi.org/10.1126/science. aag2419.
 - **63** Luo, D., Vasudevan, S.G., and Lescar, J. (2015). The flavivirus NS2B-NS3 protease-helicase as a target for antiviral drug development. *Antiviral Research* 118: 148–158; https://doi.org/10.1016/j.antiviral.2015.03.014.
 - **64** McLean, J.E., Wudzinska, A., Datan, E. et al. (2011). Flavivirus NS4A-induced autophagy protects cells against death and enhances virus replication. *The Journal of Biological Chemistry* 286: 22147–22159; https://doi.org/10.1074/jbc.M110.192500.
 - 65 Roosendaal, J., Westaway, E.G., Khromykh, A., and Mackenzie, J.M. (2006). Regulated cleavages at the West Nile virus NS4A-2K-NS4B junctions play a major role in rearranging cytoplasmic membranes and Golgi trafficking of the NS4A protein. *Journal of Virology* 80: 4623–4632; https://doi.org/10.1128/JVI.80.9.4623-4632.2006.
 - 66 Miller, S., Kastner, S., Krijnse-Locker, J. et al. (2007). The non-structural protein 4A of dengue virus is an integral membrane protein inducing membrane alterations in a 2K-regulated manner. *The Journal of Biological Chemistry* 282: 8873–8882; https://doi.org/10.1074/jbc.M609919200.
 - **67** Apte-Sengupta, S., Sirohi, D., and Kuhn, R.J. (2014). Coupling of replication and assembly in flaviviruses. *Current Opinion in Virology* 9: 134–142; https://doi.org/10.1016/j. coviro.2014.09.020.
 - **68** Zmurko, J., Neyts, J., and Dallmeier, K. (2015). Flaviviral NS4b, chameleon and jack-in-thebox roles in viral replication and pathogenesis, and a molecular target for antiviral intervention. *Reviews in Medical Virology* 25: 205–223; https://doi.org/10.1002/rmv.1835.
 - **69** Umareddy, I., Chao, A., Sampath, A. et al. (2006). Dengue virus NS4B interacts with NS3 and dissociates it from single-stranded RNA. *The Journal of General Virology* 87: 2605–2614; https://doi.org/10.1099/vir.0.81844-0.
 - 70 Muñoz-Jordán, J.L. et al. (2005). Inhibition of alpha/beta interferon signaling by the NS4B protein of flaviviruses. *Journal of Virology* 79: 8004–8013; https://doi.org/10.1128/JVI.79.13.8004-8013.2005.
 - 71 Silva, L.A. and Dermody, T.S. (2017). Chikungunya virus: epidemiology, replication, disease mechanisms, and prospective intervention strategies. *The Journal of Clinical Investigation* 127: 737–749; https://doi.org/10.1172/JCI84417.
 - 72 Bressanelli, S. et al. (2004). Structure of a flavivirus envelope glycoprotein in its low-pHinduced membrane fusion conformation. *The EMBO Journal* 23: 728–738; https://doi. org/10.1038/sj.emboj.7600064.
 - **73** Yost, S.A. and Marcotrigiano, J. (2013). Viral precursor polyproteins: keys of regulation from replication to maturation. *Current Opinion in Virology* 3: 137–142; https://doi.org/ 10.1016/j.coviro.2013.03.009.
 - 74 Solignat, M., Gay, B., Higgs, S. et al. (2009). Replication cycle of chikungunya: a re-emerging arbovirus. *Virology* 393: 183–197; https://doi.org/10.1016/j.virol.2009.07.024.
 - 75 Hasan, S.S., Sevvana, M., Kuhn, R.J., and Rossmann, M.G. (2018). Structural biology of Zika virus and other flaviviruses. *Nature Structural & Molecular Biology* 25: 13–20; https:// doi.org/10.1038/s41594-017-0010-8.

- 76 Fernandez-Garcia, M.-D., Mazzon, M., Jacobs, M., and Amara, A. (2009). Pathogenesis of flavivirus infections: using and abusing the host cell. *Cell Host & Microbe* 5: 318–328; https://doi.org/10.1016/j.chom.2009.04.001.
- 77 Spuul, P., Balistreri, G., Kääriäinen, L., and Ahola, T. (2010). Phosphatidylinositol 3-kinase-, actin-, and microtubule-dependent transport of Semliki Forest Virus replication complexes from the plasma membrane to modified lysosomes. *Journal of Virology* 84: 7543–7557; https://doi.org/10.1128/JVI.00477-10.
- **78** Wong, K.Z. and Chu, J.J.H. (2018). The interplay of viral and host factors in chikungunya virus infection: targets for antiviral strategies. *Viruses* 10; https://doi.org/10.3390/v10060294.
- 79 Boldescu, V., Behnam, M.A.M., Vasilakis, N., and Klein, C.D. (2017). Broad-spectrum agents for flaviviral infections: dengue, Zika and beyond. *Nature Reviews Drug Discovery* 16: 565–586; https://doi.org/10.1038/nrd.2017.33.
- **80** Sanjuán, R. and Domingo-Calap, P. (2016). Mechanisms of viral mutation. *Cellular and Molecular Life Sciences: CMLS* 73: 4433–4448; https://doi.org/10.1007/s00018-016-2299-6.
- 81 Domingo, E., Sheldon, J., and Perales, C. (2012). Viral quasispecies evolution. *Microbiology and Molecular Biology Reviews: MMBR* 76: 159–216; https://doi.org/10.1128/ MMBR.05023-11.
- **82** Götte, B., Liu, L., and McInerney, G.M. (2018). The enigmatic alphavirus non-structural protein 3 (nsP3) revealing its secrets at last. *Viruses* 10; https://doi.org/10.3390/v10030105.
- **83** Mathur, K. et al. (2016). Analysis of chikungunya virus proteins reveals that non-structural proteins nsP2 and nsP3 exhibit RNA interference (RNAi) suppressor activity. *Scientific Reports* 6: 38065; https://doi.org/10.1038/srep38065.
- 84 Papageorgiou, L., Loukatou, S., Sofia, K. et al. (2016). An updated evolutionary study of *Flaviviridae* NS3 helicase and NS5 RNA-dependent RNA polymerase reveals novel invariable motifs as potential pharmacological targets. *Molecular BioSystems* 12: 2080–2093; https://doi.org/10.1039/c5mb00706b.
- 85 Weisberg, I.S. and Jacobson, I.M. (2017). A pangenotypic, single tablet regimen of sofosbuvir/velpatasvir for the treatment of chronic hepatitis C infection. *Expert Opinion on Pharmacotherapy* 18 (5): 535–543. https://doi.org/10.1080/14656566.2017.1282459.
- 86 Vermehren, J., Park, J.S., Jacobson, I.M., and Zeuzem, S. (2018). Challenges and perspectives of direct antivirals for the treatment of hepatitis C virus infection. *Journal of Hepatology* 69: 1178–1187; https://doi.org/10.1016/j.jhep.2018.07.002.
- 87 Zmurko, J. et al. (2016). The viral polymerase inhibitor 7-deaza-2'-C-methyladenosine is a potent inhibitor of in vitro Zika virus replication and delays disease progression in a robust mouse infection model. *PLoS Neglected Tropical Diseases* 10: e0004695; https://doi.org/10.1371/journal.pntd.0004695.
- **88** Eyer, L. et al. (2019). Viral RNA-dependent RNA polymerase inhibitor 7-deaza-2'-Cmethyladenosine prevents death in a mouse model of West Nile Virus infection. *Antimicrobial Agents and Chemotherapy* 63; https://doi.org/10.1128/AAC.02093-18.
- 89 Schul, W., Liu, W., Xu, H.-Y. et al. (2007). A dengue fever viremia model in mice shows reduction in viral replication and suppression of the inflammatory response after treatment with antiviral drugs. *The Journal of Infectious Diseases* 195: 665–674; https://doi.org/10.1086/511310.
- **90** Olsen, D.B. et al. (2004). A 7-deaza-adenosine analog is a potent and selective inhibitor of hepatitis C virus replication with excellent pharmacokinetic properties. *Antimicrobial Agents and Chemotherapy* 48: 3944–3953; https://doi.org/10.1128/AAC.48.10.3944-3953.2004.

- **396** 15 Novel Antiviral Strategies Against Emerging Arbovirus Infections
 - **91** Julander, J.G. et al. (2017). Efficacy of the broad-spectrum antiviral compound BCX4430 against Zika virus in cell culture and in a mouse model. *Antiviral Research* 137: 14–22; https://doi.org/10.1016/j.antiviral.2016.11.003.
 - **92** Warren, T.K. et al. (2014). Protection against filovirus diseases by a novel broad-spectrum nucleoside analogue BCX4430. *Nature* 508: 402–405; https://doi.org/10.1038/ nature13027.
 - **93** Julander, J.G. et al. (2014). BCX4430, a novel nucleoside analog, effectively treats yellow fever in a Hamster model. *Antimicrobial Agents and Chemotherapy* 58: 6607–6614; https://doi.org/10.1128/AAC.03368-14.
 - **94** Eyer, L. et al. (2017). Antiviral activity of the adenosine analogue BCX4430 against West Nile virus and tick-borne flaviviruses. *Antiviral Research* 142: 63–67; https://doi.org/10.1016/j.antiviral.2017.03.012.
 - 95 Bullard-Feibelman, K.M. et al. (2017). The FDA-approved drug sofosbuvir inhibits Zika virus infection. *Antiviral Research* 137: 134–140; https://doi.org/10.1016/j.antiviral. 2016.11.023.
 - **96** Baz, M. and Boivin, G. (2019). Antiviral agents in development for Zika virus infections. *Pharmaceuticals (Basel, Switzerland)* 12; https://doi.org/10.3390/ph12030101.
 - **97** Ferreira, A.C. et al. (2017). Sofosbuvir protects Zika virus-infected mice from mortality, preventing short- and long-term sequelae. *Scientific Reports* 7: 9409; https://doi.org/10.1038/s41598-017-09797-8.
 - **98** Lazear, H.M. et al. (2016). A mouse model of Zika virus pathogenesis. *Cell Host & Microbe* 19: 720–730; https://doi.org/10.1016/j.chom.2016.03.010.
 - **99** Xu, H.-T. et al. (2017). Evaluation of Sofosbuvir (β-d-2'-deoxy-2'-α-fluoro-2'-β-C-methyluridine) as an inhibitor of dengue virus replication<sup. *Scientific Reports* 7: 6345; https://doi.org/10.1038/s41598-017-06612-2.
 - **100** Gan, C.S., Lim, S.K., Chee, C.F. et al. (2018). Sofosbuvir as treatment against dengue? *Chemical Biology & Drug Design* 91: 448–455; https://doi.org/10.1111/cbdd.13091.
 - **101** de Freitas, C.S. et al. (2019). Yellow fever virus is susceptible to sofosbuvir both in vitro and in vivo. *PLoS Neglected Tropical Diseases* 13: e0007072; https://doi.org/10.1371/journal.pntd.0007072.
 - 102 Ferreira, A.C. et al. (2019). Beyond members of the *Flaviviridae* family, Sofosbuvir also inhibits chikungunya virus replication. *Antimicrobial Agents and Chemotherapy* 63; https://doi.org/10.1128/AAC.01389-18.
 - **103** Keating, G.M. (2014). Sofosbuvir: a review of its use in patients with chronic hepatitis C. *Drugs* 74: 1127–1146; https://doi.org/10.1007/s40265-014-0247-z.
 - Pearlman, B.L. and Hinds, A.E. (2018). Review article: novel antivirals for hepatitis
 C-sofosbuvir/velpatasvir/voxilaprevir, glecaprevir/pibrentasvir. *Alimentary Pharmacology* & *Therapeutics* 48: 914–923; https://doi.org/10.1111/apt.14977.
 - 105 Li, Z. et al. (2017). Existing drugs as broad-spectrum and potent inhibitors for Zika virus by targeting NS2B-NS3 interaction. *Cell Research* 27: 1046–1064; https://doi.org/10.1038/ cr.2017.88.
 - 106 Li, Z. et al. (2018). Erythrosin B is a potent and broad-spectrum orthosteric inhibitor of the flavivirus NS2B-NS3 protease. *Antiviral Research* 150: 217–225; https://doi. org/10.1016/j.antiviral.2017.12.018.

- 107 Zou, G. et al. (2009). A single-amino acid substitution in West Nile virus 2K peptide between NS4A and NS4B confers resistance to lycorine, a flavivirus inhibitor. *Virology* 384: 242–252; https://doi.org/10.1016/j.virol.2008.11.003.
- 108 Wang, C., Yang, S.N.Y., Smith, K. et al. (2017). Nuclear import inhibitor *N*-(4-hydroxyphenyl) retinamide targets Zika virus (ZIKV) nonstructural protein 5 to inhibit ZIKV infection. *Biochemical and Biophysical Research Communications* 493: 1555–1559; https://doi.org/10.1016/j.bbrc.2017.10.016.
- **109** Pitts, J.D., Li, P.-C., de Wispelaere, M., and Yang, P.L. (2017). Antiviral activity of *N*-(4-hydroxyphenyl) retinamide (4-HPR) against Zika virus. *Antiviral Research* 147: 124–130; https://doi.org/10.1016/j.antiviral.2017.10.014.
- **110** Fraser, J.E. et al. (2014). A nuclear transport inhibitor that modulates the unfolded protein response and provides in vivo protection against lethal dengue virus infection. *The Journal of Infectious Diseases* 210: 1780–1791; https://doi.org/10.1093/infdis/jiu319.
- 111 Moore, M.M. et al. (2010). A phase II study of fenretinide in patients with hormone refractory prostate cancer: a trial of the Cancer Therapeutics Research Group. *Cancer Chemotherapy and Pharmacology* 66: 845–850; https://doi.org/10.1007/ s00280-009-1228-x.
- **112** Mastrangelo, E. et al. (2012). Ivermectin is a potent inhibitor of flavivirus replication specifically targeting NS3 helicase activity: new prospects for an old drug. *The Journal of Antimicrobial Chemotherapy* 67: 1884–1894; https://doi.org/10.1093/jac/dks147.
- **113** Varghese, F.S. et al. (2016). Discovery of berberine, abamectin and ivermectin as antivirals against chikungunya and other alphaviruses. *Antiviral Research* 126: 117–124; https://doi.org/10.1016/j.antiviral.2015.12.012.
- Yap, T.L. et al. (2007). Crystal structure of the dengue virus RNA-dependent RNA polymerase catalytic domain at 1.85-angstrom resolution. *Journal of Virology* 81: 4753–4765; https://doi.org/10.1128/JVI.02283-06.
- Eyer, L., Nencka, R., de Clercq, E. et al. (2018). Nucleoside analogs as a rich source of antiviral agents active against arthropod-borne flaviviruses. *Antiviral Chemistry & Chemotherapy* 26: 1–28. HYPERLINK "https://urldefense.com/v3/__https://doi.org/10.1177/2040206618761299__;!!N11eV2iwtfs!44bOVUwOZMoG0wiisCrWXPtzTzJNQR7 GdmOP7IIES80WE4fuM0bVby3052Q\$" https://doi.org/10.1177/2040206618761299.
- **116** Domingo, E. and Holland, J.J. (1997). RNA virus mutations and fitness for survival. *Annual Review of Microbiology* 51: 151–178; https://doi.org/10.1146/annurev.micro.51.1.151.
- **117** García, L.L., Padilla, L., and Castaño, J.C. (2017). Inhibitors compounds of the flavivirus replication process. *Virology Journal* 14: 95; https://doi.org/10.1186/s12985-017-0761-1.
- **118** Holmes, E.C. (2003). Error thresholds and the constraints to RNA virus evolution. *Trends in Microbiology* 11: 543–546; https://doi.org/10.1016/j.tim.2003.10.006.
- **119** Todt, D., Walter, S., Brown, R.J.P., and Steinmann, E. (2016). Mutagenic effects of ribavirin on hepatitis E virus-viral extinction versus selection of fitness-enhancing mutations. *Viruses* 8; https://doi.org/10.3390/v8100283.
- Arnold, J.J. et al. (2012). Sensitivity of mitochondrial transcription and resistance of RNA polymerase II dependent nuclear transcription to antiviral ribonucleosides. *PLoS Pathogens* 8: e1003030; https://doi.org/10.1371/journal.ppat.1003030.

- **398** 15 Novel Antiviral Strategies Against Emerging Arbovirus Infections
 - **121** Taylor, R. et al. (2016). BCX4430 a broad-spectrum antiviral adenosine nucleoside analog under development for the treatment of Ebola virus disease. *Journal of Infection and Public Health* 9: 220–226; https://doi.org/10.1016/j.jiph.2016.04.002.
 - **122** Sofia, M.J. et al. (2010). Discovery of a β-d-2'-deoxy-2'-α-fluoro-2'-β-C-methyluridine nucleotide prodrug (PSI-7977) for the treatment of hepatitis C virus. *Journal of Medicinal Chemistry* 53: 7202–7218; https://doi.org/10.1021/jm100863x.
 - 123 Fung, A. et al. (2014). Efficiency of incorporation and chain termination determines the inhibition potency of 2'-modified nucleotide analogs against hepatitis C virus polymerase. *Antimicrobial Agents and Chemotherapy* 58: 3636–3645; https://doi.org/10.1128/ AAC.02666-14.
 - 124 Mendes, É.A. et al. (2019). Sofosbuvir inhibits yellow fever virus in vitro and in patients with acute liver failure. *Annals of Hepatology*; https://doi.org/10.1016/j. aohep.2019.09.001.
 - **125** Dousson, C.B. (2018). Current and future use of nucleo(s)tide prodrugs in the treatment of hepatitis C virus infection. *Antiviral Chemistry & Chemotherapy* 26: 2040206618756430; https://doi.org/10.1177/2040206618756430.
 - **126** Mumtaz, N. et al. (2017). Cell-line dependent antiviral activity of sofosbuvir against Zika virus. *Antiviral Research* 146: 161–163; https://doi.org/10.1016/j.antiviral.2017.09.004.
 - **127** Sacramento, C.Q. et al. (2017). The clinically approved antiviral drug sofosbuvir inhibits Zika virus replication. *Scientific Reports* 7: 40920; https://doi.org/10.1038/srep40920.
 - **128** Bollati, M. et al. (2010). Structure and functionality in flavivirus NS-proteins: perspectives for drug design. *Antiviral Research* 87: 125–148; https://doi.org/10.1016/j. antiviral.2009.11.009.
 - **129** Brecher, M., Zhang, J., and Li, H. (2013). The flavivirus protease as a target for drug discovery. *Virologica Sinica* 28: 326–336; https://doi.org/10.1007/s12250-013-3390-x.
 - **130** Clercq, d., de, E., and Li, G. (2016). Approved antiviral drugs over the past 50 years. *Clinical Microbiology Reviews* 29: 695–747; https://doi.org/10.1128/CMR.00102-15.
 - **131** Bhakat, S. et al. (2015). Reaching beyond HIV/HCV: nelfinavir as a potential starting point for broad-spectrum protease inhibitors against dengue and chikungunya virus. *RSC Advances* 5: 85938–85949; https://doi.org/10.1039/C5RA14469H.
 - **132** Wyles, D.L. (2013). Antiviral resistance and the future landscape of hepatitis C virus infection therapy. *The Journal of Infectious Diseases* 207 (Suppl 1): S33–S39; https://doi.org/10.1093/infdis/jis761.
 - Yang, C.-C. et al. (2011). Novel dengue virus-specific NS2B/NS3 protease inhibitor, BP2109, discovered by a high-throughput screening assay. *Antimicrobial Agents and Chemotherapy* 55: 229–238; https://doi.org/10.1128/AAC.00855-10.
 - 134 da Silva-Júnior, E.F. and Araújo-Júnior, J.X.d. (2019). Peptide derivatives as inhibitors of NS2B-NS3 protease from Dengue, West Nile, and Zika flaviviruses. *Bioorganic & Medicinal Chemistry* 27: 3963–3978; https://doi.org/10.1016/j.bmc.2019.07.038.
 - 135 Falgout, B., Pethel, M., Zhang, Y.M., and Lai, C.J. (1991). Both nonstructural proteins NS2B and NS3 are required for the proteolytic processing of dengue virus nonstructural proteins. *Journal of Virology* 65: 2467–2475.
 - Pant, A. et al. (2017). Proteases in mosquito borne diseases: new avenues in drug development. *Current Topics in Medicinal Chemistry* 17: 2221–2232; https://doi.org/10.217 4/1568026617666170130122231.

- Pambudi, S. et al. (2013). A small compound targeting the interaction between nonstructural proteins 2B and 3 inhibits dengue virus replication. *Biochemical and Biophysical Research Communications* 440: 393–398; https://doi.org/10.1016/j. bbrc.2013.09.078.
- 138 Erbel, P. et al. (2006). Structural basis for the activation of flaviviral NS3 proteases from dengue and West Nile virus. *Nature Structural & Molecular Biology* 13: 372–373; https:// doi.org/10.1038/nsmb1073.
- 139 Brecher, M. et al. (2017). A conformational switch high-throughput screening assay and allosteric inhibition of the flavivirus NS2B-NS3 protease. *PLoS Pathogens* 13: e1006411; https://doi.org/10.1371/journal.ppat.1006411.
- **140** Byrd, C.M. et al. (2013). A novel benzoxazole inhibitor of dengue virus replication that targets the NS3 helicase. *Antimicrobial Agents and Chemotherapy* 57: 1902–1912; https://doi.org/10.1128/AAC.02251-12.
- **141** Sweeney, N.L. et al. (2015). Benzothiazole and pyrrolone flavivirus inhibitors targeting the viral helicase. *ACS Infectious Diseases* 1: 140–148; https://doi.org/10.1021/id5000458.
- **142** Ho, Y.-J. et al. (2015). Suramin inhibits chikungunya virus entry and transmission. *PloS ONE* 10: e0133511; https://doi.org/10.1371/journal.pone.0133511.
- 143 Basavannacharya, C. and Vasudevan, S.G. (2014). Suramin inhibits helicase activity of NS3 protein of dengue virus in a fluorescence-based high throughput assay format. *Biochemical and Biophysical Research Communications* 453: 539–544; https://doi.org/10.1016/j.bbrc.2014.09.113.
- **144** Modis, Y., Ogata, S., Clements, D., and Harrison, S.C. (2003). A ligand-binding pocket in the dengue virus envelope glycoprotein. *Proceedings of the National Academy of Sciences of the United States of America* 100: 6986–6991; https://doi.org/10.1073/pnas.0832193100.
- **145** Clark, M.J. et al. (2016). GNF-2 inhibits dengue virus by targeting Abl kinases and the viral E protein. *Cell Chemical Biology* 23: 443–452; https://doi.org/10.1016/j.chembiol. 2016.03.010.
- **146** de Wispelaere, M. et al. (2018). Inhibition of flaviviruses by targeting a conserved pocket on the viral envelope protein. *Cell Chemical Biology* 25: 1006–1016.e8; https://doi. org/10.1016/j.chembiol.2018.05.011.
- **147** Wang, Q.-Y. et al. (2015). Discovery of dengue virus NS4B inhibitors. *Journal of Virology* 89: 8233–8244; https://doi.org/10.1128/JVI.00855-15.
- **148** Li, P.-C. et al. (2019). Small molecules targeting the flavivirus E protein with broadspectrum activity and antiviral efficacy in vivo. *ACS Infectious Diseases* 5: 460–472; https://doi.org/10.1021/acsinfecdis.8b00322.
- 149 Kampmann, T. et al. (2009). In silico screening of small molecule libraries using the dengue virus envelope E protein has identified compounds with antiviral activity against multiple flaviviruses. *Antiviral Research* 84: 234–241; https://doi.org/10.1016/j. antiviral.2009.09.007.
- 150 Agarwal, G. et al. (2019). Virtual screening of inhibitors against envelope glycoprotein of chikungunya virus: a drug repositioning approach. *Bioinformation* 15: 439–447; https://doi.org/10.6026/97320630015439.
- **151** Deeba, F. et al. (2017). Potential entry inhibitors of the envelope protein (E2) of chikungunya virus: in silico structural modeling, docking and molecular dynamic studies. *Virus Disease* 28: 39–49; https://doi.org/10.1007/s13337-016-0356-2.

- **400** 15 Novel Antiviral Strategies Against Emerging Arbovirus Infections
 - **152** Sharma, R. et al. (2016). Inhibition of chikungunya virus by picolinate that targets viral capsid protein. *Virology* 498: 265–276; https://doi.org/10.1016/j.virol.2016.08.029.
 - 153 Xie, X., Zou, J., Wang, Q.-Y., and Shi, P.-Y. (2015). Targeting dengue virus NS4B protein for drug discovery. *Antiviral Research* 118: 39–45; https://doi.org/10.1016/j. antiviral.2015.03.007.
 - **154** Lim, S.P. et al. (2013). Ten years of dengue drug discovery: progress and prospects. *Antiviral Research* 100: 500–519; https://doi.org/10.1016/j.antiviral.2013.09.013.
 - 155 Feibelman, K.M., Fuller, B.P., Li, L. et al. (2018). Identification of small molecule inhibitors of the chikungunya virus nsP1 RNA capping enzyme. *Antiviral Research* 154: 124–131; https://doi.org/10.1016/j.antiviral.2018.03.013.
 - **156** Brecher, M. et al. (2015). Identification and characterization of novel broad-spectrum inhibitors of the flavivirus methyltransferase. *ACS Infectious Diseases* 1: 340–349; https://doi.org/10.1021/acsinfecdis.5b00070.
 - 157 Brecher, M. et al. (2015). Novel broad spectrum inhibitors targeting the flavivirus methyltransferase. *PloS ONE* 10: e0130062; https://doi.org/10.1371/journal. pone.0130062.
 - **158** Mudgal, R., Mahajan, S., and Tomar, S. (2019). Inhibition of chikungunya virus by an adenosine analog targeting the SAM-dependent nsP1 methyltransferase. *FEBS Letters*; https://doi.org/10.1002/1873-3468.13642.
 - **159** Zou, J. and Shi, P.-Y. (2019). Strategies for Zika drug discovery. *Current Opinion in Virology* 35: 19–26; https://doi.org/10.1016/j.coviro.2019.01.005.
 - 160 Kaur, R., Mudgal, R., Narwal, M., and Tomar, S. (2018). Development of an ELISA assay for screening inhibitors against divalent metal ion dependent alphavirus capping enzyme. *Virus Research* 256: 209–218; https://doi.org/10.1016/j.virusres.2018.06.013.
 - **161** Carocci, M. et al. (2015). The bioactive lipid 4-hydroxyphenyl retinamide inhibits flavivirus replication. *Antimicrobial Agents and Chemotherapy* 59: 85–95; https://doi.org/10.1128/AAC.04177-14.
 - **162** Delvecchio, R. et al. (2016). Chloroquine, an endocytosis blocking agent, inhibits Zika virus infection in different cell models. *Viruses* 8; https://doi.org/10.3390/v8120322.
 - 163 Khan, M., Santhosh, S.R., Tiwari, M. et al. (2010). Assessment of in vitro prophylactic and therapeutic efficacy of chloroquine against Chikungunya virus in vero cells. *Journal of Medical Virology* 82: 817–824; https://doi.org/10.1002/jmv.21663.
 - **164** Savarino, A., Boelaert, J.R., Cassone, A. et al. (2003). Effects of chloroquine on viral infections: an old drug against today's diseases. *The Lancet Infectious Diseases* 3: 722–727.
 - **165** Subudhi, B.B., Chattopadhyay, S., Mishra, P., and Kumar, A. (2018). Current strategies for inhibition of chikungunya infection. *Viruses* 10; https://doi.org/10.3390/v10050235.
 - 166 Lamballerie, X.d. et al. (2008). On chikungunya acute infection and chloroquine treatment. *Vector Borne and Zoonotic Diseases (Larchmont, N.Y.)* 8: 837–839; https://doi. org/10.1089/vbz.2008.0049.
 - **167** Mercorelli, B., Palù, G., and Loregian, A. (2018). Drug repurposing for viral infectious diseases: how far are we? *Trends in Microbiology* 26: 865–876; https://doi.org/10.1016/j. tim.2018.04.004.
 - 168 Zakaria, M.K., Carletti, T., and Marcello, A. (2018). Cellular targets for the treatment of flavivirus infections. *Frontiers in Cellular and Infection Microbiology* 8: 398; https://doi. org/10.3389/fcimb.2018.00398.

- **169** Botta, L., Rivara, M., Zuliani, V., and Radi, M. (2018). Drug repurposing approaches to fight dengue virus infection and related diseases. *Frontiers in Bioscience (Landmark Edition)* 23: 997–1019; https://doi.org/10.2741/4630.
- **170** Bekerman, E. and Einav, S. (2015). Infectious disease. Combating emerging viral threats. *Science (New York, N.Y.)* 348: 282–283; https://doi.org/10.1126/science.aaa3778.
- **171** Schor, S. and Einav, S. (2018). Combating intracellular pathogens with repurposed host-targeted drugs. *ACS Infectious Diseases* 4: 88–92; https://doi.org/10.1021/acsinfecdis.7b00268.
- **172** Barrows, N.J. et al. (2016). A screen of FDA-approved drugs for inhibitors of Zika virus infection. *Cell Host & Microbe* 20: 259–270; https://doi.org/10.1016/j.chom.2016.07.004.
- **173** Kamiyama, N. et al. (2017). Ribavirin inhibits Zika virus (ZIKV) replication in vitro and suppresses viremia in ZIKV-infected STAT1-deficient mice. *Antiviral Research* 146: 1–11; https://doi.org/10.1016/j.antiviral.2017.08.007.
- 174 Tong, X. et al. (2018). Merimepodib, an IMPDH inhibitor, suppresses replication of Zika virus and other emerging viral pathogens. *Antiviral Research* 149: 34–40; https://doi.org/10.1016/j.antiviral.2017.11.004.
- **175** Pires de Mello, C.P., Drusano, G.L., Rodriquez, J.L. et al. (2018). Antiviral effects of clinically-relevant interferon-α and ribavirin regimens against dengue virus in the hollow fiber infection model (HFIM). *Viruses* 10; https://doi.org/10.3390/v10060317.
- **176** Chowers, M.Y. et al. (2001). Clinical characteristics of the West Nile fever outbreak, Israel, 2000. *Emerging Infectious Diseases* 7: 675–678; https://doi.org/10.3201/eid0704.010414.
- 177 Rothan, H.A. et al. (2016). Mefenamic acid in combination with ribavirin shows significant effects in reducing chikungunya virus infection in vitro and in vivo. *Antiviral Research* 127: 50–56; https://doi.org/10.1016/j.antiviral.2016.01.006.
- **178** Rothan, H.A. et al. (2015). A combination of doxycycline and ribavirin alleviated chikungunya infection. *PloS ONE* 10: e0126360; https://doi.org/10.1371/journal. pone.0126360.
- **179** Ravichandran, R. and Manian, M. (2008). Ribavirin therapy for chikungunya arthritis. *Journal of Infection in Developing Countries* 2: 140–142.
- Hoffmann, H.-H., Kunz, A., Simon, V.A. et al. (2011). Broad-spectrum antiviral that interferes with de novo pyrimidine biosynthesis. *Proceedings of the National Academy of Sciences of the United States of America* 108: 5777–5782; https://doi.org/10.1073/ pnas.1101143108.
- 181 Merino-Ramos, T., Jiménez de Oya, N., Saiz, J.-C., and Martín-Acebes, M.A. (2017). Antiviral activity of nordihydroguaiaretic acid and its derivative tetra-O-methyl nordihydroguaiaretic acid against West Nile Virus and Zika virus. *Antimicrobial Agents* and Chemotherapy 61; https://doi.org/10.1128/AAC.00376-17.
- 182 Soto-Acosta, R., Bautista-Carbajal, P., Syed, G.H. et al. (2014). Nordihydroguaiaretic acid (NDGA) inhibits replication and viral morphogenesis of dengue virus. *Antiviral Research* 109: 132–140; https://doi.org/10.1016/j.antiviral.2014.07.002.
- 183 Uchida, L. et al. (2016). Suppressive effects of the site 1 protease (S1P) inhibitor, PF-429242, on dengue virus propagation. *Viruses* 8; https://doi.org/10.3390/v8020046.
- 184 Cheng, F., Ramos da Silva, S., Huang, I.-C. et al. Suppression of Zika virus infection and replication in endothelial cells and astrocytes by PKA inhibitor PKI 14-22. *Journal of Virology* 92 https://doi.org/10.1128/JVI.02019-17 (2018).

- **402** 15 Novel Antiviral Strategies Against Emerging Arbovirus Infections
 - 185 Jiménez de Oya, N., Blázquez, A.-B., Casas, J. et al. (2018). Direct activation of adenosine monophosphate-activated protein kinase (AMPK) by PF-06409577 inhibits flavivirus infection through modification of host cell lipid metabolism. *Antimicrobial Agents and Chemotherapy* 62; https://doi.org/10.1128/AAC.00360-18.
 - 186 Soto-Acosta, R., Bautista-Carbajal, P., Cervantes-Salazar, M. et al. (2017). DENV up-regulates the HMG-CoA reductase activity through the impairment of AMPK phosphorylation: a potential antiviral target. *PLoS Pathogens* 13: e1006257; https://doi. org/10.1371/journal.ppat.1006257.
 - 187 Jiménez de Oya, N. et al. (2019). Targeting host metabolism by inhibition of acetyl-coenzyme A carboxylase reduces flavivirus infection in mouse models. *Emerging Microbes & Infections* 8: 624–636; https://doi.org/10.1080/22221751.2019.1604084.
 - **188** Li, C. et al. (2017). 25-Hydroxycholesterol protects host against Zika virus infection and its associated microcephaly in a mouse model. *Immunity* 46: 446–456; https://doi.org/10.1016/j.immuni.2017.02.012.
 - 189 Willard, K.A., Elling, C.L., Stice, S.L., and Brindley, M.A. (2018). The oxysterol
 7-ketocholesterol reduces Zika virus titers in vero cells and human neurons. *Viruses* 11; https://doi.org/10.3390/v11010020.
 - **190** Tricarico, P.M., Caracciolo, I., Gratton, R. et al. (2019). 25-Hydroxycholesterol reduces inflammation, viral load and cell death in ZIKV-infected U-87 MG glial cell line. *Inflammopharmacology* 27: 621–625; https://doi.org/10.1007/s10787-018-0517-6.
 - **191** Martinez-Gutierrez, M., Correa-Londoño, L.A., Castellanos, J.E. et al. (2014). Lovastatin delays infection and increases survival rates in AG129 mice infected with dengue virus serotype 2. *PloS ONE* 9: e87412; https://doi.org/10.1371/journal.pone.0087412.
 - **192** Rothwell, C. et al. (2009). Cholesterol biosynthesis modulation regulates dengue viral replication. *Virology* 389: 8–19; https://doi.org/10.1016/j.virol.2009.03.025.
 - 193 Whitehorn, J. et al. (2016). Lovastatin for the treatment of adult patients with dengue: a randomized, double-blind, placebo-controlled trial. *Clinical Infectious Diseases: An Official Publication of the Infectious Diseases Society of America* 62: 468–476; https://doi.org/10.1093/cid/civ949.
 - **194** Mackenzie, J.M., Khromykh, A.A., and Parton, R.G. (2007). Cholesterol manipulation by West Nile virus perturbs the cellular immune response. *Cell Host & Microbe* 2: 229–239; https://doi.org/10.1016/j.chom.2007.09.003.
 - 195 Wichit, S. et al. (2017). Imipramine inhibits chikungunya virus replication in human skin fibroblasts through interference with intracellular cholesterol trafficking. *Scientific Reports* 7: 3145; https://doi.org/10.1038/s41598-017-03316-5.
 - 196 Bekerman, E. et al. (2017). Anticancer kinase inhibitors impair intracellular viral trafficking and exert broad-spectrum antiviral effects. *The Journal of Clinical Investigation* 127: 1338–1352; https://doi.org/10.1172/JCI89857.
 - **197** Chu, J.J.H. and Yang, P.L. (2007). c-Src protein kinase inhibitors block assembly and maturation of dengue virus. *Proceedings of the National Academy of Sciences of the United States of America* 104: 3520–3525; https://doi.org/10.1073/pnas.0611681104.
 - **198** Kumar, R., Agrawal, T., Khan, N.A. et al. (2016). Identification and characterization of the role of c-terminal Src kinase in dengue virus replication. *Scientific Reports* 6: 30490; https://doi.org/10.1038/srep30490.

- 199 Wispelaere, M. de, LaCroix, A. J. & Yang, P. L. The small molecules AZD0530 and dasatinib inhibit dengue virus RNA replication via Fyn kinase. *Journal of Virology* 87, 7367–7381; https://doi.org/10.1128/JVI.00632-13 (2013).
- 200 Hirsch, A.J. et al. (2005). The Src family kinase c-Yes is required for maturation of West Nile virus particles. *Journal of Virology* 79: 11943–11951; https://doi.org/10.1128/ JVI.79.18.11943-11951.2005.
- **201** Xu, M. et al. (2016). Identification of small-molecule inhibitors of Zika virus infection and induced neural cell death via a drug repurposing screen. *Nature Medicine* 22: 1101–1107; https://doi.org/10.1038/nm.4184.
- **202** Abrams, R.P.M., Solis, J., and Nath, A. (2017). Therapeutic approaches for Zika virus infection of the nervous system. *Neurotherapeutics: The Journal of the American Society for Experimental NeuroTherapeutics* 14: 1027–1048; https://doi.org/10.1007/s13311-017-0575-2.
- **203** Plummer, E. et al. (2015). Dengue virus evolution under a host-targeted antiviral. *Journal of Virology* 89: 5592–5601; https://doi.org/10.1128/JVI.00028-15.
- 204 Watanabe, S., Low, J.G.-H., and Vasudevan, S.G. (2018). Preclinical antiviral testing for dengue virus infection in mouse models and its association with clinical studies. ACS Infectious Diseases 4: 1048–1057; https://doi.org/10.1021/acsinfecdis.8b00054.
- **205** Low, J.G. et al. (2014). Efficacy and safety of celgosivir in patients with dengue fever (CELADEN): a phase 1b, randomised, double-blind, placebo-controlled, proof-of-concept trial. *The Lancet Infectious Diseases* 14: 706–715; https://doi.org/10.1016/S1473-3099(14)70730-3.
- 206 Sung, C. et al. (2016). Extended evaluation of virological, immunological and pharmacokinetic endpoints of CELADEN: a randomized, placebo-controlled trial of celgosivir in dengue fever patients. *PLoS Neglected Tropical Diseases* 10: e0004851; https:// doi.org/10.1371/journal.pntd.0004851.
- **207** Qing, M. et al. (2009). Cyclosporine inhibits flavivirus replication through blocking the interaction between host cyclophilins and viral NS5 protein. *Antimicrobial Agents and Chemotherapy* 53: 3226–3235; https://doi.org/10.1128/AAC.00189-09.
- **208** Vidotto, A. et al. (2017). Systems biology reveals NS4B-cyclophilin A interaction: a new target to inhibit YFV replication. *Journal of Proteome Research* 16: 1542–1555; https://doi.org/10.1021/acs.jproteome.6b00933.
- **209** Choy, M.M. et al. (2015). Proteasome inhibition suppresses dengue virus egress in antibody dependent infection. *PLoS Neglected Tropical Diseases* 9: e0004058; https://doi.org/10.1371/journal.pntd.0004058.
- **210** Xin, Q.-L. et al. (2017). Quantitative proteomic analysis of mosquito C6/36 cells reveals host proteins involved in Zika virus infection. *Journal of Virology* 91; https://doi.org/10.1128/JVI.00554-17.
- **211** Haviernik, J. et al. (2018). Arbidol (umifenovir): a broad-spectrum antiviral drug that inhibits medically important arthropod-borne flaviviruses. *Viruses* 10; https://doi.org/10.3390/v10040184.
- **212** Fink, S.L. et al. (2018). The antiviral drug arbidol inhibits Zika virus. *Scientific Reports* 8: 8989; https://doi.org/10.1038/s41598-018-27224-4.
- **213** Delogu, I. et al. (2011). In vitro antiviral activity of arbidol against chikungunya virus and characteristics of a selected resistant mutant. *Antiviral Research* 90: 99–107; https://doi.org/10.1016/j.antiviral.2011.03.182.

- **404** 15 Novel Antiviral Strategies Against Emerging Arbovirus Infections
 - **214** Jung, E. et al. (2019). Neutralization of acidic intracellular vesicles by niclosamide inhibits multiple steps of the dengue virus life cycle in vitro. *Scientific Reports* 9: 8682; https://doi.org/10.1038/s41598-019-45095-1.
 - **215** Nuckols, J.T. et al. (2014). pH-dependent entry of chikungunya virus fusion into mosquito cells. *Virology Journal* 11: 215; https://doi.org/10.1186/s12985-014-0215-y.
 - **216** Graci, J.D. and Cameron, C.E. (2006). Mechanisms of action of ribavirin against distinct viruses. *Reviews in Medical Virology* 16: 37–48; https://doi.org/10.1002/rmv.483.
 - **217** Paeshuyse, J., Dallmeier, K., and Neyts, J. (2011). Ribavirin for the treatment of chronic hepatitis C virus infection: a review of the proposed mechanisms of action. *Current Opinion in Virology* 1: 590–598; https://doi.org/10.1016/j.coviro.2011.10.030.
 - 218 Crance, J.M., Scaramozzino, N., Jouan, A., and Garin, D. (2003). Interferon, ribavirin, 6-azauridine and glycyrrhizin: antiviral compounds active against pathogenic flaviviruses. *Antiviral Research* 58: 73–79; https://doi.org/10.1016/s0166-3542 (02)00185-7.
 - 219 Franco, E.J., Rodriquez, J.L., Pomeroy, J.J. et al. (2018). The effectiveness of antiviral agents with broad-spectrum activity against chikungunya virus varies between host cell lines. *Antiviral Chemistry & Chemotherapy* 26: 1-7. HYPERLINK "https://urldefense.com/v3/__https://doi__; !!N11eV2iwtfs!44bOVUwOZMoGOwiisCrWXPtzTzJNQR7GdmOP7IIES8OWE4fuM0bV5U PXG38\$" https://doi.org/10.1177/2040206618807580.
 - **220** Kumar, R. et al. (2009). Randomized, controlled trial of oral ribavirin for Japanese encephalitis in children in Uttar Pradesh, India. *Clinical Infectious Diseases: An Official Publication of the Infectious Diseases Society of America* 48: 400–406; https://doi.org/10.1086/596309.
 - 221 Minakawa, N. and Matsuda, A. (1999). Mechanism-based design of inosine
 5-monophosphate dehydrogenase inhibitors: synthesis and biological activities of
 5-ethynyl-1-beta-d-ribofuranosylimidazole-4-carboxamide (EICAR) and its derivatives.
 Current Medicinal Chemistry 6: 615–628.
 - **222** Bentley, R. (2000). Mycophenolic acid: a one hundred year odyssey from antibiotic to immunosuppressant. *Chemical Reviews* 100: 3801–3826; https://doi.org/10.1021/cr990097b.
 - 223 First, M.R. (1997). An update on new immunosuppressive drugs undergoing preclinical and clinical trials: potential applications in organ transplantation. *American Journal of Kidney Diseases: The Official Journal of the National Kidney Foundation* 29: 303–317; https://doi.org/10.1016/s0272-6386(97)90046-9.
 - 224 Munier-Lehmann, H. et al. (2015). Original 2-(3-alkoxy-1H-pyrazol-1-yl)pyrimidine derivatives as inhibitors of human dihydroorotate dehydrogenase (DHODH). *Journal of Medicinal Chemistry* 58: 860–877; https://doi.org/10.1021/jm501446r.
 - **225** Jordan, T.X. and Randall, G. (2016). Flavivirus modulation of cellular metabolism. *Current Opinion in Virology* 19: 7–10; https://doi.org/10.1016/j.coviro.2016.05.007.
 - **226** Martín-Acebes, M.A., Jiménez de Oya, N., and Saiz, J.-C. (2019). Lipid metabolism as a source of druggable targets for antiviral discovery against Zika and other flaviviruses. *Pharmaceuticals (Basel, Switzerland)* 12; https://doi.org/10.3390/ph12020097.
 - 227 Merino-Ramos, T. et al. (2016). Modification of the host cell lipid metabolism induced by hypolipidemic drugs targeting the acetyl coenzyme A carboxylase impairs West Nile Virus replication. *Antimicrobial Agents and Chemotherapy* 60: 307–315; https://doi.org/10.1128/AAC.01578-15.

- **228** Martín-Acebes, M.A., Blázquez, A.-B., Jiménez de Oya, N. et al. (2011). West Nile virus replication requires fatty acid synthesis but is independent on phosphatidylinositol-4-phosphate lipids. *PloS ONE* 6: e24970; https://doi.org/10.1371/journal.pone.0024970.
- 229 Heaton, N.S. et al. (2010). Dengue virus nonstructural protein 3 redistributes fatty acid synthase to sites of viral replication and increases cellular fatty acid synthesis. *Proceedings* of the National Academy of Sciences of the United States of America 107: 17345–17350; https://doi.org/10.1073/pnas.1010811107.
- **230** Poh, M.K. et al. (2012). U18666A, an intra-cellular cholesterol transport inhibitor, inhibits dengue virus entry and replication. *Antiviral Research* 93: 191–198; https://doi.org/10.1016/j.antiviral.2011.11.014.
- **231** Leier, H.C., Messer, W.B., and Tafesse, F.G. (2018). Lipids and pathogenic flaviviruses: an intimate union. *PLoS Pathogens* 14: e1006952; https://doi.org/10.1371/journal. ppat.1006952.
- Zhou, W., Woodson, M., Sherman, M.B. et al. (2019). Exosomes mediate Zika virus transmission through SMPD3 neutral Sphingomyelinase in cortical neurons. *Emerging Microbes & Infections* 8: 307–326; https://doi.org/10.1080/22221751.201 9.1578188.
- **233** Martín-Acebes, M.A. et al. (2014). The composition of West Nile virus lipid envelope unveils a role of sphingolipid metabolism in flavivirus biogenesis. *Journal of Virology* 88: 12041–12054; https://doi.org/10.1128/JVI.02061-14.
- **234** Huang, Y. et al. (2018). Zika virus propagation and release in human fetal astrocytes can be suppressed by neutral sphingomyelinase-2 inhibitor GW4869. *Cell Discovery* 4: 19; https://doi.org/10.1038/s41421-018-0017-2.
- 235 Osuna-Ramos, J.F., Reyes-Ruiz, J.M., and Del Ángel, R.M. (2018). The role of host cholesterol during flavivirus infection. *Frontiers in Cellular and Infection Microbiology* 8: 388; https://doi.org/10.3389/fcimb.2018.00388.
- **236** Bryan-Marrugo, O.L. et al. (2016). The anti-dengue virus properties of statins may be associated with alterations in the cellular antiviral profile expression. *Molecular Medicine Reports* 14: 2155–2163; https://doi.org/10.3892/mmr.2016.5519.
- 237 Martínez-Gutierrez, M., Castellanos, J.E., and Gallego-Gómez, J.C. (2011). Statins reduce dengue virus production via decreased virion assembly. *Intervirology* 54: 202–216; https:// doi.org/10.1159/000321892.
- 238 Shanbhogue, P. and Hannun, Y.A. (2018). Exploring the therapeutic landscape of sphingomyelinases. *Handbook of Experimental Pharmacology*; https://doi. org/10.1007/164_2018_179.
- **239** Soyal, S.M., Nofziger, C., Dossena, S. et al. (2015). Targeting SREBPs for treatment of the metabolic syndrome. *Trends in Pharmacological Sciences* 36: 406–416; https://doi.org/10.1016/j.tips.2015.04.010.
- **240** Hardie, D.G., Ross, F.A., and Hawley, S.A. (2012). AMP-activated protein kinase: a target for drugs both ancient and modern. *Chemistry & Biology* 19: 1222–1236; https://doi.org/10.1016/j.chembiol.2012.08.019.
- **241** Hardie, D.G. (2018). Keeping the home fires burning: AMP-activated protein kinase. *Journal of the Royal Society Interface* 15; https://doi.org/10.1098/rsif.2017.0774.
- **242** Cheng, X., Li, J., and Guo, D. (2018). SCAP/SREBPs are central players in lipid metabolism and novel metabolic targets in cancer therapy. *Current Topics in Medicinal Chemistry* 18: 484–493; https://doi.org/10.2174/1568026618666180523104541.

- **406** 15 Novel Antiviral Strategies Against Emerging Arbovirus Infections
 - **243** Schor, S. and Einav, S. (2018). Repurposing of kinase inhibitors as broad-spectrum antiviral drugs. *DNA and Cell Biology* 37: 63–69; https://doi.org/10.1089/dna.2017.4033.
 - 244 Ricotta, D., Conner, S. D., Schmid, S. L., Figura, K. von & Honing, S. Phosphorylation of the AP2 mu subunit by AAK1 mediates high affinity binding to membrane protein sorting signals. *The Journal of Cell Biology* 156, 791–795; https://doi.org/10.1083/ jcb.200111068 (2002).
 - **245** Umeda, A., Meyerholz, A., and Ungewickell, E. (2000). Identification of the universal cofactor (auxilin 2) in clathrin coat dissociation. *European Journal of Cell Biology* 79: 336–342; https://doi.org/10.1078/S0171-9335(04)70037-0.
 - 246 Pagano, M.A., Tibaldi, E., Palù, G., and Brunati, A.M. (2013). Viral proteins and Src family kinases: mechanisms of pathogenicity from a "liaison dangereuse". *World Journal* of Virology 2: 71–78; https://doi.org/10.5501/wjv.v2.i2.71.
 - **247** Adrián, F.J. et al. (2006). Allosteric inhibitors of Bcr-abl-dependent cell proliferation. *Nature Chemical Biology* 2: 95–102; https://doi.org/10.1038/nchembio760.
 - **248** Rossari, F., Minutolo, F., and Orciuolo, E. (2018). Past, present, and future of Bcr-Abl inhibitors: from chemical development to clinical efficacy. *Journal of Hematology & Oncology* 11: 84; https://doi.org/10.1186/s13045-018-0624-2.
 - **249** Broeckel, R. et al. (2019). Src family kinase inhibitors block translation of alphavirus subgenomic mRNAs. *Antimicrobial Agents and Chemotherapy* 63; https://doi.org/10.1128/AAC.02325-18.
 - **250** Schang, L.M. (2003). The cell cycle, cyclin-dependent kinases, and viral infections: new horizons and unexpected connections. *Progress in Cell Cycle Research* 5: 103–124.
 - 251 Courageot, M. P., Frenkiel, M. P., Dos Santos, C. D., Deubel, V. & Desprès, P. Alphaglucosidase inhibitors reduce dengue virus production by affecting the initial steps of virion morphogenesis in the endoplasmic reticulum. *Journal of Virology* 74, 564–572; https://doi.org/10.1128/jvi.74.1.564-572.2000 (2000).
 - 252 Pérez-García, L.A., Martínez-Duncker, I., and Mora Montes, H.M. (2017). The endoplasmic reticulum alpha-glycosidases as potential targets for virus control. *Current Protein & Peptide Science* 18: 1090–1097; https://doi.org/10.217 4/1389203717666160813161729.
 - **253** Frausto, S.D., Lee, E., and Tang, H. (2013). Cyclophilins as modulators of viral replication. *Viruses* 5: 1684–1701; https://doi.org/10.3390/v5071684.
 - **254** Kambara, H. et al. (2011). Involvement of cyclophilin B in the replication of Japanese encephalitis virus. *Virology* 412: 211–219; https://doi.org/10.1016/j.virol.2011.01.011.
 - **255** Scholte, F.E.M. et al. (2013). Characterization of synthetic chikungunya viruses based on the consensus sequence of recent E1-226V isolates. *PloS ONE* 8: e71047; https://doi.org/10.1371/journal.pone.0071047.
 - 256 Pawlotsky, J.-M. et al. (2015). Alisporivir plus ribavirin, interferon free or in combination with pegylated interferon, for hepatitis C virus genotype 2 or 3 infection. *Hepatology* (*Baltimore, Md.*) 62: 1013–1023; https://doi.org/10.1002/hep.27960.
 - **257** Chiramel, A.I. et al. (2016). Alisporivir has limited antiviral effects against Ebola virus strains Makona and Mayinga. *The Journal of Infectious Diseases* 214: S355–S359; https://doi.org/10.1093/infdis/jiw241.
 - **258** Choi, A.G., Wong, J., Marchant, D., and Luo, H. (2013). The ubiquitin-proteasome system in positive-strand RNA virus infection. *Reviews in Medical Virology* 23: 85–96; https://doi.org/10.1002/rmv.1725.

- **259** Amaya, M. et al. (2015). The ubiquitin proteasome system plays a role in venezuelan equine encephalitis virus infection. *PloS ONE* 10: e0124792; https://doi.org/10.1371/journal.pone.0124792.
- **260** Lv, B.-M. et al. (2018). Drug repurposing for Japanese encephalitis virus infection by systems biology methods. *Molecules (Basel, Switzerland)* 23; https://doi.org/10.3390/molecules23123346.
- **261** Wang, S. et al. (2016). The ubiquitin-proteasome system is essential for the productive entry of Japanese encephalitis virus. *Virology* 498: 116–127; https://doi.org/10.1016/j. virol.2016.08.013.
- **262** Nag, D.K. and Finley, D. (2012). A small-molecule inhibitor of deubiquitinating enzyme USP14 inhibits dengue virus replication. *Virus Research* 165: 103–106; https://doi.org/10.1016/j.virusres.2012.01.009.
- **263** Llamas-González, Y.Y., Campos, D., Pascale, J.M. et al. (2019). A functional ubiquitinproteasome system is required for efficient replication of New World Mayaro and Una Alphaviruses. *Viruses* 11; https://doi.org/10.3390/v11040370.
- **264** Gilfoy, F., Fayzulin, R., and Mason, P.W. (2009). West Nile virus genome amplification requires the functional activities of the proteasome. *Virology* 385: 74–84; https://doi.org/10.1016/j.virol.2008.11.034.
- **265** Fernandez-Garcia, M.-D. et al. (2011). Appraising the roles of CBLL1 and the ubiquitin/ proteasome system for flavivirus entry and replication. *Journal of Virology* 85: 2980–2989; https://doi.org/10.1128/JVI.02483-10.
- 266 Bernard, E. et al. (2010). Endocytosis of chikungunya virus into mammalian cells: role of clathrin and early endosomal compartments. *PloS ONE* 5: e11479; https://doi. org/10.1371/journal.pone.0011479.
- **267** Chu, J.J.H. and Ng, M.L. (2004). Infectious entry of West Nile virus occurs through a clathrin-mediated endocytic pathway. *Journal of Virology* 78: 10543–10555; https://doi.org/10.1128/JVI.78.19.10543-10555.2004.
- **268** Pierson, T.C. and Kielian, M. (2013). Flaviviruses: braking the entering. *Current Opinion in Virology* 3: 3–12; https://doi.org/10.1016/j.coviro.2012.12.001.
- **269** van der Schaar, H.M. et al. (2008). Dissecting the cell entry pathway of dengue virus by single-particle tracking in living cells. *PLoS Pathogens* 4: e1000244; https://doi.org/10.1371/journal.ppat.1000244.
- **270** Heinz, F.X. and Allison, S.L. (2000). Structures and mechanisms in flavivirus fusion. *Advances in Virus Research* 55: 231–269; https://doi.org/10.1016/s0065-3527(00)55005-2.
- **271** Modis, Y., Ogata, S., Clements, D., and Harrison, S.C. (2004). Structure of the dengue virus envelope protein after membrane fusion. *Nature* 427: 313–319; https://doi.org/10.1038/nature02165.
- Boriskin, Y.S., Leneva, I.A., Pécheur, E.-I., and Polyak, S.J. (2008). Arbidol: a broad-spectrum antiviral compound that blocks viral fusion. *Current Medicinal Chemistry* 15: 997–1005; https://doi.org/10.2174/092986708784049658.
- Pécheur, E.-I. et al. (2016). The synthetic antiviral drug arbidol inhibits globally prevalent pathogenic viruses. *Journal of Virology* 90: 3086–3092; https://doi.org/10.1128/JVI.02077-15.
- **274** Villalaín, J. (2010). Membranotropic effects of arbidol, a broad anti-viral molecule, on phospholipid model membranes. *The Journal of Physical Chemistry B* 114: 8544–8554; https://doi.org/10.1021/jp102619w.

- **408** | 15 Novel Antiviral Strategies Against Emerging Arbovirus Infections
 - 275 Paton, N.I. and Aboulhab, J. (2005). Hydroxychloroquine, hydroxyurea and didanosine as initial therapy for HIV-infected patients with low viral load: safety, efficacy and resistance profile after 144 weeks. *HIV Medicine* 6: 13–20; https://doi.org/10.1111/ j.1468-1293.2005.00259.x.
 - **276** Han, J. et al. (2018). Genome-wide CRISPR/Cas9 screen identifies host factors essential for influenza virus replication. *Cell Reports* 23: 596–607; https://doi.org/10.1016/j. celrep.2018.03.045.
 - **277** LaBeaud, A.D. (2008). Why arboviruses can be neglected tropical diseases. *PLoS Neglected Tropical Diseases* 2: e247; https://doi.org/10.1371/journal.pntd.0000247.
 - 278 Weisberg IS, Jacobson IM. A pangenotypic, single tablet regimen of sofosbuvir/ velpatasvir for the treatment of chronic hepatitis C infection. Expert Opin Pharmacother. 2017 Apr;18(5):535-543. doi: 10.1080/14656566.2017.1282459 Epub 2017 Mar 24. PMID: 28092171.
Lucie Fénéant, Bianca Bodmer, Thomas C. Mettenleiter, Allison Groseth, and Thomas Hoenen

Institute of Molecular Virology and Cell Biology, Friedrich-Loeffler-Institut, Greifswald, Germany

16.1 Introduction

Members of several virus families are classified as Biosafety Level 4 (BSL4) pathogens based on their ability to cause life-threatening human disease with a significant risk of spread, thus posing a high risk both to the individual and the community. While the list of viruses included in this category varies somewhat by country (often taking into account whether a particular pathogen is endemic to that region), members of four virus families are almost universally included: filoviruses, henipaviruses, arenaviruses, and bunyaviruses.

The exotic nature of BSL4 viruses and the frequently fatal outcome of these diseases, combined with the sporadic occurrence of outbreaks and limited case numbers, set them apart from other infectious diseases, and poses special challenges for the development of treatment or prophylactic strategies. Indeed, clinical trials are very difficult to conduct in many instances due to the lack of a well-developed medical infrastructure in the affected areas and/or limited patient numbers. Further, there is often only a very limited commercial interest in the development of treatments for these diseases since they affect relatively small numbers of patients, often in some of the world's poorest countries. Consequently, the perceived importance of therapeutic development for these agents has traditionally been limited and, therefore, treatment options are sparse. Even fewer have been approved by any kind of regulatory body.

More recently, however, the development of novel treatments and vaccines for many of these agents has been recognized as international priority, for instance, in WHO R&D Roadmaps [1–4]. This has led to a drastic increase in early-stage research for such therapies; however, few have so far progressed to testing in highly stringent and predictive animal models, usually nonhuman primates (NHPs), or even to human clinical testing. Nonetheless, there have clearly been successes toward the treatment of BSL4 virus infections in these more advanced phases of testing, and it is these strategies that are summarized here.

* Lucie Fénéant and Bianca Bodmer contributed equally to this chapter.

New Drug Development for Known and Emerging Viruses, First Edition. Edited by Helga Rübsamen-Schaeff and Helmut Buschmann. © 2022 WILEY-VCH GmbH. Published 2022 by WILEY-VCH GmbH.

16

16.2 Filoviruses

16.2.1 Virus and Disease Overview

Filoviruses are negative-sense, single-stranded RNA viruses of the family *Filoviridae*, which includes six genera. Those known to be relevant for human disease belong to the *Ebolavirus* and *Marburgvirus* genera, and cause viral hemorrhagic fever (VHF) outbreaks. Consequently, filoviruses are classified as BSL4 agents.

The first member of the filovirus family, Marburg virus (MARV, genus Marburgvirus), was identified in 1967 as the causative agent of a VHF outbreak primarily affecting the city of Marburg, Germany, and specifically individuals who had contact to tissues derived from African green monkeys [5, 6]. Of the 32 cases recognized during this outbreak, 7 died, resulting in a case fatality rate (CFR) of 23% [7]. Members of a second filovirus genus, Ebolavirus, emerged in 1976 concurrently in South Sudan (formerly Sudan) and the Democratic Republic of Congo (formerly Zaire), resulting in the infection of 284 (CFR 53%) and 318 (CFR 88%) individuals, respectively. Since then, outbreaks of filovirus disease (FVD) of variable extent have occurred sporadically throughout Western and Central Africa. The most devastating outbreak was caused by Ebola virus (EBOV) and lasted from late 2013 until 2016. It mainly affected the West African countries of Guinea, Sierra Leone, and Liberia, resulting in 28652 confirmed cases, of which 11325 patients died (CFR 40%) [8]. Together with another large outbreak in DRC from 2018 to 2020 with almost 3 500 cases [9] that defied control efforts for two years despite public health intervention by the global community (including vaccination), these episodes highlight the previously neglected potential for filoviruses to cause large and even multinational outbreaks.

Whereas both marburgviruses, MARV and Ravn virus (RAVV), cause FVD [10], of the six ebolaviruses known to date, only four are associated with human disease (EBOV, Sudan virus (SUDV), Bundibugyo virus (BDBV), and Taï Forest virus (TAFV)). In contrast, Reston virus (RESTV) appears to cause asymptomatic infections in humans [8, 11], while for several novel filoviruses, which were recently identified in diverse geographical regions [12–14], the pathogenic potential remains unknown. This raises the possibility of future emergence of filovirus-induced diseases, also outside known filovirus endemic regions.

FVD is a zoonotic infection, but the exact transmission route remains unclear in many cases. Spillover into the human population is believed to occur from infected animals through exposure to body fluids, i.e. blood or excreta (for instance, as a result of bites, during handling of carcasses, or through environmental contamination), with further spread from human-to-human via blood and other body fluids. Both MARV and RAVV have been isolated from fruit bats (*Rousettus aegypticus*), which seem to be a key reservoir for these viruses. While no ebolavirus has yet been isolated from bats (reviewed in [15]), based on serological analyses they are also proposed to be the reservoir for members of this genus.

Symptoms of FVD infection begin suddenly after an incubation period of 3–13 days and are initially unspecific [16–19]. During the first 10 days, fever is the most frequently reported symptom, followed by fatigue, headache, myalgia, and renal dysfunction. In some

cases, a rash also develops early during infection. Gastrointestinal symptoms such as diarrhea, nausea, and vomiting lead to fluid loss and further weaken the patient. On days 7–12 after onset of symptoms, patients either show signs of recovery or progress to exhibit symptoms of shock including diminished consciousness and coma [16]. Hemorrhages and overt bleeding are not universally observed, even among fatal cases, but if they do occur, they typically affect the gastrointestinal tract [16–18]. Higher viral loads in the blood of patients correlate with a poor prognosis, and an influence of patient age on outcome has also been reported [20–22]. Additionally, respiratory symptoms such as dyspnea, tachypnea, and hiccups are indicative of a poor prognosis [20]. Sequelae of FVD include arthralgia, temporary hair loss, and ocular complications such as blurred vision, conjunctivitis, uveitis, or vision loss [23]. Data on neurological sequelae suggest increased rates of fatigue, insomnia, and depression in FVD survivors; however, these could also reflect the psychological trauma and effects of stigmatization by the community [24].

16.2.2 Antiviral Strategies

In response to the dramatic West African EBOV outbreak, tremendous effort has been devoted recently to the development of antiviral approaches against filovirus infection. However, to date, most of them have not yet shown efficacy in highly predictive models of filovirus infection (i.e. NHPs) and/or advanced to clinical testing. Nonetheless, there are a few different strategies that have shown promise. Currently, the approaches that have advanced the furthest are immunotherapies using specific monoclonal antibodies and postexposure vaccination with a recombinant vesicular stomatitis virus (rVSV) encoding the EBOV glycoprotein GP (rVSV-EBOV, also known as Ervebo, rVSV Δ G-EBOV GP, or rVSV-ZEBOV). However, small molecule, antisense, and host-directed therapies have also shown promise in NHPs and could potentially fill important treatment gaps.

16.2.2.1 Immunotherapies

Antibodies can counter filovirus infection in two ways: (i) through direct virus neutralization by targeting domains of GP necessary for entry, or (ii) through non-neutralizing antibodies that direct the immune-mediated clearance of infected cells via antibody-dependent cellular cytotoxicity and complement-dependent cytotoxicity [25]. During the West African EBOV outbreak, antibody-containing convalescent sera were tested for treatment in a nonrandomized historically controlled clinical trial [26]. No serious adverse reactions were detected; however, no benefit for patient survival was observed either. Later analysis of antibody titers in these sera revealed no significant correlation between the amount of IgG or neutralizing antibody (nAb) titers and survival [27].

Monoclonal antibodies (or mixtures of several monoclonal antibodies) have also been developed as potential treatments of FVD, and these show more clinical success than convalescent sera. The three antibody preparations that are the most advanced in terms of testing are ZMapp, REGN-EB3, and the monoclonal antibody (mAb) mAb114. ZMapp is a humanized antibody cocktail based on antibodies originally isolated from mice immunized with Venezuelan equine encephalitis virus replicons encoding EBOV GP or with a rVSV expressing EBOV GP [28, 29]. It consists of two nAbs binding the EBOV GP

base (2G4 and 4G7) and one non-neutralizing antibody targeting the glycan cap of EBOV GP (13C6) [30]. This antibody cocktail was tested in a randomized, controlled clinical phase 1/2 study [31]. While this study suggested a benefit of treatment with ZMapp, it failed the prespecified statistical threshold, possibly as a result of lower than anticipated enrollment numbers due to the end of the outbreak. The second antibody cocktail, REGN-EB3 (also known as Regeneron), also contains three antibodies (REGN 3470, 3471, and 3479), which were isolated from immunized mice encoding fully human antibody variable region gene segments [32]. REGN 3479 and REGN 3470 show neutralizing activity by targeting the fusion loop and the head region of the GP subunit GP1, respectively, whereas REGN 3471 is non-neutralizing and binds the outer glycan cap of this protein. Tolerability and safety of this cocktail was shown during a randomized phase 1 clinical trial [33]. Finally, mAb114 is a single nAb isolated from an FVD survivor [34] that binds the glycan cap and the GP1 core, thereby preventing binding of GP to its receptor, Niemann-Pick C1, after endosomal cleavage of the glycan cap [35]. Safety and tolerability of mAb114 were also confirmed during an open-label phase 1 study [36]. Each of these three mAb preparations, in addition to the nucleoside analogue remdesivir (described in more detail in Section 16.2.2.3), have recently been tested for treatment of FVD in a clinical trial (registration number NCT03719586) in the Democratic Republic of Congo, where a prolonged FVD outbreak occured from 2018 to 2020. Data from this trial showed that patients receiving either REGN-EB3 or mAb114 had a higher chance of survival compared to the other treatments, making them the only treatment options to date for which significant effectivity against FVD has been demonstrated in a clinical trial [37]. Based on this success, both compounds have recently obtained U.S. Food and Drug Administration approval.

16.2.2.2 Postexposure Prophylaxis

While primarily used for preventive purposes, the rVSV-EBOV vaccine also has potential for use in postexposure prophylaxis (PEP). It is based on an attenuated, replication competent vesicular stomatitis virus, which encodes the EBOV GP instead of the VSV glycoprotein (G). Vaccination results in viral protein expression and virus replication; however, virus particles express the EBOV GP on their surface instead of the VSV G and facilitate the induction of humoral immune responses directed against EBOV GP, as shown in NHPs [38]. Several clinical trials have shown safety and efficacy of the vaccine, which recently obtained approval by both the European Medicines Agency as well as the U.S. Food and Drug Administration, and it has been used for ring vaccinations with more than 250 000 recipients to date [9] (see [39] for an overview of EBOV vaccine candidates in clinical trials). As a PEP treatment, rVSV-EBOV vaccination was analyzed in NHPs 20-30 minutes after challenge with an uniformly lethal dose of EBOV [40]. While 50% of animals survived, all showed signs of illness by day 6, and the severity of these signs correlated with death. When NHPs were treated with rVSV-encoding MARV GP instead of VSV G (rVSV-MARV) 20-30 minutes after lethal infection with MARV, all animals survived [41, 42]. Initiation of treatment with rVSV-MARV 24 hours after infection still resulted in the survival of five out of six animals, while treatment after 48 hours saved two out of six animals [42]. In 2009, following a potential EBOV infection due to an accidental needle stick injury during an animal experiment in a

BSL4 facility, a virologist received rVSV-EBOV 48 hours after the accident as PEP [43]. Similarly, in 2014 and 2015, five people involved in the West African EBOV outbreak, including nurses and physicians, were treated with rVSV-EBOV between 24 and 72 hours after potential EBOV exposure [44]. None of the patients developed signs or symptoms of FVD, and no laboratory evidence for EBOV infection was detectable. Indeed, while four out of the six patients showed seroconversion to EBOV GP, which is encoded by the rVSV-EBOV vaccine, no evidence for antibodies directed against other EBOV proteins could be found [43, 44], suggesting not only that rVSV-EBOV can function as PEP in humans, but may induce sterile immunity (provided that these individuals were, indeed, infected).

16.2.2.3 Small Molecules

Where their mechanism of action is known, most small molecules (Table 16.1) that have been tested in NHPs or in clinical trials against filovirus infection target either entry of the virus or viral RNA synthesis. For example, amiodarone, which is approved for anti-arrhythmic therapy, also inhibits filovirus entry, with IC_{50} values of 16 μ M in Vero E6 cells and 7 μ M in human monocyte-derived macrophages, by targeting virus entry in the late endosomal compartment. As such, it was administered to 65 patients during the West African outbreak following compassionate use protocols [48, 63] based solely on in vitro data [47]. Unfortunately, in patients the treatment did not show any beneficial effect, and indeed an in vivo study in guinea pigs conducted afterward also showed no protection after challenge with EBOV in this animal model [46].

A more frequently pursued approach involves the use of nucleoside analogues such as BCX4430, favipiravir (also known as T-705) and remdesivir (also known as GS-5734), which are incorporated into the viral RNA as purine analogues during replication, thereby leading to mismatch mutations and/or early chain termination (Table 16.1). Such strategies would be particularly promising as these drugs show broad efficacy against several RNA viruses. While BCX4430 (an adenosine analogue) has been tested in a phase 1 clinical study confirming its safety and tolerability following intramuscular application [51], its efficacy for the treatment of FVD remains to be evaluated in phase 2 and 3 clinical trials. The purine analogue favipiravir shows antiviral activity against several RNA viruses in vitro and in various animal models [64]. It is currently licensed for the treatment of Influenza virus infections in Japan and is in clinical trials in the United States for the same indication [65]. With respect to its ability to treat EBOV infection in humans, two retrospective analyses of compassionate use in Guinea [58] and Sierra Leone [60] showed no significant improvement in CFR, although in one study prolonged survival times and reduced viral loads were noted. A historically controlled, non-randomized, single-arm proof-of-concept study in Guinea did not show any clinical benefit as well, possibly due to low target serum concentrations of favipiravir, which might have had an influence on the outcome [59, 66]. Thus, further studies remain necessary to evaluate the potential benefits of favipiravir against FVD. A third nucleotide analogue, i.e. the adenosine analogue remdesivir (Table 16.1), showed reduced viral titers and survival benefit when delivered intravenously in NHPs at two or three days post infection with EBOV [61]. The efficiency of remdesivir in humans was tested in a randomized clinical trial (also discussed in Section 16.2.2.1); however, its effect was

Table 16.1 Overview of small molecules tested against filovirus infection.

Name	Structure	Mechanism of action	In vitro effect	In vivo effect in NHPs	Clinical regimen
Amiodarone	Contraction of the second seco	 Inhibitor of viral fusion activity in the late endosomal compartment [45] 	 EBOV: IC₅₀ of 15.9 μM in Vero cells [46] IC₅₀ of 5.5 μM in Huh7 cells [46] IC₅₀ of 6.6 μM in human monocyte- derived macrophages [46] IC₅₀ of 0.4 μM in EAhy cells [47] 	Not determined	EBOV: • Compassionate use in 65 patients during the West African EBOV outbreak, details regarding regimen not available [48]
BCX4430		 Polymerase inhibitor Mutagenic agent 	EBOV: 	EBOV: • 16 or 25 mg/kg twice daily [50] • 100 mg/kg twice as loading dose, then 25 mg/kg twice daily [51]	 EBOV: Up to 10 mg/kg daily for seven days [51]



(Continued)

Table 16.1 (Continued)

Name	Structure	Mechanism of action	In vitro effect	In vivo effect in NHPs	Clinical regimen
Favipiravir (T-705)	$\bigcup_{\mu \in \mathcal{F}_{\mu}} \bigcup_{\mu \in \mathcal{F}_{\mu}} \operatorname{NH}_{2}$	 Polymerase inhibitor Mutagenic agent 	 EBOV: CL₂₅ of 30.9 μM in HEK 293T cells [54] CL₂₅ of 67.0 μM in Vero E6 cells [55] MARV: IC₅₀ of 43.3 μM in Vero cells [50] 	 EBOV: 400 mg/kg as loading dose on day 3, 200 mg/kg on days 2–10 [56] 250 mg/kg twice or 125 mg/kg twice as loading dose on day 0, then 150 or 75 mg/kg twice daily on days 1–13 [56] 200 mg/kg twice as loading dose on day 2, then 100 mg/kg twice daily until day 12 [57] 150 or 250 mg/kg twice as loading dose on day 0, then 150 mg/kg twice daily until day 12 [57] MARV: 200 mg/kg twice as loading dose on day 0, then 150 mg/kg twice daily on days 1–13 [56] 	EBOV: 6 0000 mg as loading dose, then 2400 mg daily until day 9 [58, 59] 8 000 mg twice as loading dose, then 600 mg twice daily [60]

Name	Structure	Mechanism of action	In vitro effect	In vivo effect in NHPs	Clinical regimen
Remdesivir (GS-5734)	$\begin{array}{c} \begin{array}{c} & & & \\ & & $	 Polymerase inhibitor Delayed chain termination 	EBOV: • IC ₅₉ of 0.1 μM in HeLa cells [61] • IC ₅₉ of 0.003-0.1 μM in Huh-7 cells [61, 62] • IC ₅₉ of 0.1 μM in HFF-1 cells [61] • IC ₅₉ of 0.1 μM in HMVEC-TERT cells [61] • IC ₅₉ of 0.1 μM in human monocyte derived macrophages [61] MARV: • IC ₅₉ of 0.01-0.02 μM in Huh-7 cells [62]	 BOV: 3 mg/kg on days 0–11 or days 2–13 after infection [61] 10 mg/kg as loading dose on day 2 or 3, then 3 mg/kg for 11 days [61] 10 mg/kg on days 3–14 after infection [61] 	EBOV: • 200 mg as loading dose, then 100 mg daily for 9–12 days [37]

limited compared to two other treatment options under investigation, i.e. the antibodies mAB114 and REGN-EB3 [37].

Finally, brincidofovir (Table 16.1), which was originally developed as an antiviral drug against double-stranded DNA viruses, also shows efficacy against EBOV in vitro at concentrations of $0.3-8\,\mu$ M (IC₅₀ values in HeLa and Vero E6 cells, respectively), although the mechanism is unknown [52]. Based on these data, a single-arm phase 2 clinical trial was started in January 2015 in Liberia but was stopped after the enrolment of only four patients (all of whom died of FVD), due to the decision of the manufacturer to stop development of brincidofovir for treatment of filovirus infections [53].

16.2.2.4 Antisense Therapy

To date, two nucleic acid-based antisense therapies have shown efficacy in NHPs [67–70]. The first one uses uncharged DNA analogues based on phosphorodiamidate morpholino oligomers (PMOs), which block protein translation of complementary mRNAs by steric hindrance. The tested PMOs were directed against the EBOV viral proteins VP35 or VP24 (AVI-6002) or MARV VP24 and the MARV nucleoprotein (NP; AVI-6003) [67, 68]. Tolerability of AVI-6002 and AVI-6003 were analyzed in phase 1 single-ascending-dose studies, which showed safety and tolerability of the tested products in humans, independent of the target sequences used [71]. However, despite these initially promising results, development has been discontinued by the manufacturer.

The second strategy is based on antisense therapy using small interfering RNAs (siR-NAs), which bind and direct degradation of complementary RNA sequences, thereby allowing silencing of gene expression [72]. Here, a combination of siRNAs targeting the EBOV proteins L, VP24, and VP35 and prepared in a lipid nanoparticle formulation was analyzed in a historically controlled, single-arm phase 2 clinical trial. However, the trial was terminated early after interim analyses showed a lack of patient benefit with respect to survival [73]. After the trial was concluded, the manufacturer discontinued development of filovirus-specific siRNA products.

16.2.2.5 Host-directed Therapies

To date, only one strategy aimed at modulating the host response has reached advanced stages of testing for treatment of FVD. It is based on the observation that filovirus infection of macrophages leads to secretion of high levels of an interferon (IFN)- α subset, whereas only low levels of IFN- β secretion can be observed [74]. To balance this dysregulation, IFN β -1a treatment was tested in a historically controlled, single-arm proof-of-concept trial with nine patients. A possible survival benefit was suggested, but due to the low patient numbers, no firm conclusions could be drawn [75].

16.3 Henipaviruses

Henipaviruses are part of the family *Paramyxoviridae*, and as such are negative-sense single-stranded RNA viruses. Two species within the genus *Henipavirus*, Hendra virus (HeV) and Nipah virus (NiV), are classified as BSL4 agents due to their ability to cause severe and frequently fatal human disease.

16.3.1 Disease Overview

HeV was first identified in 1994 in Hendra (a suburb of Brisbane), Australia, when 14 out of 21 horses suffering from a severe respiratory illness, as well as one out of two patients with close contact to the infected horses, succumbed to infection [76]. Indeed, transmission for all human cases reported to date, of which four out of seven have been fatal (CFR 57%), occurred through close contact with infected horses [77]. The natural reservoirs of HeV are flying foxes of the genus *Pteropus*, from which HeV has been successfully isolated [78]. Since 2015, a HeV vaccine for use in horses containing a soluble version of HeV G is available in Australia, thereby reducing the risk of human spillover [79]. Symptoms of HeV disease begin after an incubation period of 7 to 21 days, and in 6 of the 7 reported cases symptoms were initially unspecific, including fever, myalgia, and respiratory disease, while the seventh case developed encephalitis without any initial unspecific symptoms [80]. In severe cases, encephalitis is common, with three patients (of which only one recovered) developing acute encephalitis and one patient developing a fatal encephalitis 13 months following initial recovery from the infection. To date, there is no evidence for human-to-human transmission of HeV [81].

In 1998, NiV, a second human pathogenic *Henipavirus*, emerged in Malaysia [82]. Here, transmission could be traced back to infected pigs that probably fed on fruit contaminated by infected *Pteropus* bats [83, 84], which were identified as the natural reservoir host for NiV. The outbreak in Malaysia was eventually contained in 1999, but during this time 265 human cases were reported, of which 105 died (CFR 40%). Interestingly, however, 8% of infections were reported to be asymptomatic [85]. The outbreak also spread to Singapore through the import of infected pigs, resulting in illness in 11 slaughterhouse workers [86]. While neither of these countries has reported additional NiV cases since this time, Bangladesh and India have faced repeated outbreaks with CFRs of approximately 75% [87]. Other than transmission from infected pigs, the consumption of raw date palm sap contaminated by NiV shedding bats is believed to be a major route of transmission to the human population. Human-to-human transmission of NiV also occurs through shedding of virus in urine and respiratory secretions [88, 89]. Symptoms of NiV infection start after an incubation period of between four days to two months with unspecific symptoms similar to those observed for HeV infection, including fever, headache, unconsciousness, weakness, and respiratory difficulties, and in severe cases infection can also result in encephalitis [90, 91]. Similar to HeV, encephalitis in NiV-infected patients can also develop months to years after the initial infection. Furthermore, other neurologic sequelae have been described, e.g. personality changes and deficits in attention [90, 92].

16.3.2 Antiviral Strategies

16.3.2.1 Small Molecules

To date, several purine analogues (Table 16.2) have been tested for the treatment of HeV or NiV in clinical trials and/or shown promise in NHP studies. Already in 2008 and 2009, ribavirin was used to treat three individuals with confirmed HeV infection, which had close contact with infected horses [80, 95]. Of these patients, two died. While this suggests a possible lack of efficacy, it has also been suggested that the ribavirin serum

concentration was too low for efficient reduction of viral replication, when compared to in vitro efficacy data where an inhibitory effect was observable at concentrations of $49 \,\mu$ M and above [95, 100]. In contrast, NHP data showed a prolonged survival time in HeVinfected animals when ribavirin was given, although no benefit was observed with respect to CFR [97]. Ribavirin was also tested for NiV treatment in an open-label study during the Malaysian outbreak. In this study, 140 patients received ribavirin (compared to 54 patients who did not), resulting in a 36% reduction in mortality and a tendency toward a reduction in neurological sequelae in survivors [98]. Similarly, aciclovir (Table 16.2) was given empirically to nine patients with encephalitis during a NiV outbreak in Singapore. Eight of these patients survived [93]. Finally, remdesivir (Table 16.2) has recently been shown to protect NHPs from lethal NiV challenge [94]; however, clinical data in humans are still lacking.

16.3.2.2 Immunotherapies

Another promising approach for the treatment of HeV and NiV infection is based on mAb therapy. A mAb with cross-reactivity to both HeV and NiV (m102) was produced through recombinant antibody technology [101]. Modification of m102 through light-chain shuf-fling and heavy-chain variable domain random mutagenesis enhanced binding and potency further leading to mAb m102.4 [102]. m102.4 binds the site on the Henipavirus G protein responsible for interaction with its entry receptors Ephrin-B2 and Ephrin-B3. In lethal NHP models of HeV and NiV infection, all animals treated with m102.4 survived, even if treatment was initiated as late as five days after infection [103]. Furthermore, 10 human patients have received m102.4 after high-risk exposure to HeV infection, and one due to a potential NiV infection, none of which subsequently showed any signs of disease [104]. Thus, while data remain limited, m102.4 appears to be a promising treatment option, despite the need for larger and controlled clinical trials.

16.4 Arenaviruses

Arenaviruses are bi-segmented single-stranded negative-sense RNA viruses that compose the *Arenaviridae* family. Arenaviruses that infect mammals are divided into two serocomplexes correlating with their endemicity: the Old World Arenaviruses, found for the most part in Africa, and the New World Arenaviruses, which are mainly restricted to South America. Both groups include a number of viruses for which infection results in severe human disease, as well as others with little or no ability to cause human disease.

16.4.1 Old World Arenaviruses

Old World Arenaviruses mostly originate from Africa, with the notable exceptions of Lymphocytic Choriomeningitis Virus (LCMV), which has been found to circulate across the globe, as well as Dandenong virus, which was responsible for a small cluster of transplant-related deaths in Australia. The latter is believed to have been acquired by the donor during travel in rural areas of former Yugoslavia [105]. Among the Old World Arenaviruses, significant disease is known to be caused in humans following natural infection by LCMV,

Table 16.2 Overview of small molecules tested	l against henipavirus infection.
---	----------------------------------

Name	Structure	Mechanism of action	In vitro effect	In vivo effect in NHPs	Clinical regimen
Aciclovir	HN - N-OH HO - O - N - N	Unknown for henipaviruses	Not determined	Not determined	NiV: • Empirical therapy with aciclovir IV in nine patients, details regarding regimen not available [93]
Remdesivir (GS-5734)	HAN HON ON HON CHS	 Polymerase inhibitor Delayed chain termination 	HeV: • IC ₅₀ of 0.06 μM in HeLa cells [62] NiV: • IC ₅₀ of 0.03–0.05 μM in HeLa cells [62]	NiV: • 10 mg/kg daily for 12 days [94]	HeV: • 30 mg/kg loading dose, then 15 mg/kg four times a day for four days, then 8 mg/ kg three times a day [95]

Name	Structure	Mechanism of action	In vitro effect	In vivo effect in NHPs	Clinical regimen
Ribavirin		 Polymerase inhibitor Mutagenic agent Increases antiviral immune response Increases cell survival 	NiV: I Cs ₀ of 1.3 μ M in A549 cells [96] I Cs ₀ of 0.7 μ M in BAEC cells [96] I Cs ₀ of 1.8 μ M in BitK21 cells [96] I Cs ₀ of 1.0 μ M in BitK21 cells [96] I Cs ₀ of 1.0 μ M in ECV304 cells [96] I Cs ₀ of 1.0 μ M in HECV304 cells [96] I Cs ₀ of 0.0 μ M in HeL2 cells [96] I Cs ₀ of 0.9 μ M in HeL2 cells [96] I Cs ₀ of 0.9 μ M in HeL2 cells [96] I Cs ₀ of 0.9 μ M in M17 cells [96] I Cs ₀ of 1.1 μ M in MMEC cells [96] I Cs ₀ of 1.2 μ M in Vero cells [96]	HeV: Loading dose of 50 mg/kg, 10 mg/kg three times a day for 14 days starting 1 day before infection, 12 hours or 2 days after infection [97]	 NiV: 2 g loading dose, 1.2 g three times a day on days 2-4, then 1.2 g twice a day on days 5 and 6, then 0.6 g twice a day for additional 1-4 days [98] 30 mg/kg loading dose, then 16 mg/kg four times a day for four days, then 8 mg/kg three times a day for four days, then 8 mg/kg three times a day for four days, then 3 mg/kg three times a day for four days, then 3 g four times a day for six days [99]

Lassa virus (LASV) and Lujo virus (LUJV), with the last two causing VHF and being classified as BSL4 agents. Interestingly, however, LCMV has also been reported to cause severe illness, including features similar to that of VHF in patients who become infected through organ transplantation [106].

16.4.1.1 Disease Overview

LASV is the etiological agent of Lassa fever disease (LFD), which was first recognized in 1969 in Nigeria, following the infection of three missionary nurses, of whom two died [107]. An estimated 100000–300000 humans are infected each year, mainly in Nigeria, Guinea, Sierra Leone, Benin, and Liberia, resulting in about 5000 fatalities annually [108]. Further, the seroprevalence for LASV can be very high, ranging from 14 to 58% in highly endemic areas of Liberia and Nigeria, respectively [109, 110]. Based on such data it appears that up to 80% of LASV infections are in fact asymptomatic, with an overall fatality rate of 1–2%, although this may reach 15–20% in hospitalized patients [108, 111].

LASV is a zoonotic pathogen and is typically transmitted from its reservoir host, the multimammate mouse (*Mastomys natalensis*), through urine and feces. LASV can also be transmitted from human-to-human via direct contact with body fluids, albeit with limited efficiency. Nosocomial transmission is, however, a significant concern [112]. After a 7- to 21-day incubation period, patients develop a nonspecific febrile illness and may further develop joint pain, diarrhea, vomiting, abdominal pain, and headaches. In the most severe cases, patients can show facial edema and bleeding, as well as pleural and pericardial effusions, the latter being associated with a poor prognosis. Death usually occurs within 14 days after the onset of symptoms due to multiple organ failure and terminal shock (reviewed in [113]). While most infections are nonfatal, one common and severe consequence is acute sensoneural deafness, which affects 29% of patients [114].

More recently, a novel and apparently highly pathogenic Old World Arenavirus named LUJV was isolated after a nosocomial outbreak in South Africa in 2008. The index patient was a tour operator evacuated from Zambia who presented symptoms very similar to severe LFD. Four of the five infected patients died, resulting in an apparent CFR of 80%. This remains the only occurrence of LUJV described to date. Attempts to identify the reservoir host of LUJV based on sampling of rodent species in Zambia have so far been unsuccessful [115, 116].

16.4.1.2 Antiviral Strategies

16.4.1.2.1 Small Molecules

There is currently no specific approved antiviral treatment available for either LASV or LUJV. The current standard of care for LFD is limited to supportive care and an off-label use of ribavirin (Table 16.3), generally as a 10-day course administered intravenously [125, 132]. Two regimen are currently used: the first regimen involves a loading dose of 33 mg/kg followed by 16 mg/kg every six hours for four days and then 8 mg/kg every eight hours for six days [132]. The second regimen starts with a loading dose of 100 mg/kg divided in two doses, followed by 25 mg/kg daily for six days and then 12.5 mg/kg daily for three days [125]. The mechanism of action of ribavirin against LASV is unknown, but may involve the induction of error catastrophe during viral replication, as has been suggested for LCMV [133]. Early work, already in the late 1970s and early 1980, in LASV-infected NHPs showed

Table 16.3 Overview of small molecules tested against infection with BSL4 arenaviruses.

Name	Structure	Mechanism of action	In vitro effect	In vivo effect in NHPs	Clinical regimen
Ribavirin		 Polymerase inhibitor Mutagenic agent Increases antiviral immune response Increases cell survival 	LASV: • IC ₅₀ or ED ₅₀ from 26.5 to 82 μM in Vero cells [117, 118] JUNV: • IC ₅₀ of 4.5 to 13 μM in Vero cells [117] MACV: • ED ₅₀ of 131 μM in Vero cells [119]	LASV: • 50mg/kg loading dose, 10mg/kg every eight hours [120, 121] JUNV: • 60mg/kg loading dose, 15 mg/kg daily [122] • 15mg/kg every 12 hours [123] • 75mg/kg loading dose, 25 mg/kg every 12 hours [123] MACV: • 10 or 20 mg/kg every 12 hours [124]	 LASV: 33 mg/kg loading dose, 16 mg/kg every six hours for four days, then 8 mg/kg every eight hours for six days IV [111] 100 mg/kg loading dose, 25 mg/kg daily for six days (hen 12.5 mg/kg daily for six days, then 12.5 mg/kg daily for six days IV [125] LUJV: The only surviving patient received 2 g loading dose orally, then 1 g every six hours orally until IV was available, then 20 mg/kg IV every six hours[126] JUNV: 34 mg/kg loading dose, 17 mg/kg every six hours for four days, then 8 mg/kg every eight hours for six days [127] SABV: 30 mg/kg loading dose, 15 mg/kg every six hours for four days, then 7.5 mg/kg every eight hours for six days [128]
Favipiravir	OH O	 Polymerase 	LASV:	LASV:	LASV:
(T-705)	N NH2	inhibitor • Mutagenic agent	 IC₅₀ from 10.8 to 70.7 µM in Vero cells [129] IC₅₀ of 29.3 µM in Vero cells [118] 	• 300 mg/kg daily [130]	 16 mg/kg ribavirin IV every 6 hours, additionally 2g favipiravir orally as loading does followed by 1g every 12 hours [131] 1g ribavirin every six hours, additionally 2g favipiravir orally as loading dose followed by 1.2g every 12 hours [131]

a beneficial effect of ribavirin treatment when treated with a 50 mg/kg loading dose followed by 10 mg/kg every eight hours [120, 121, 124]. Additionally, a number of case reports and retrospective cohort studies showed a beneficial effect of ribavirin in hospitalized patients [134–139]. However, it is important to recognize that many retrospective and prospective cohort studies evaluating the efficacy of ribavirin against LFD present a heavy risk of bias. This can arise, for instance, due to a lack of transparency and randomization regarding patient assignment to "no treatment" versus "treatment" arms (which may lead to inclusion of severely ill patients that experience rapid death after hospitalization, thereby preventing the start of ribavirin treatment) or the lack of appropriate control groups [140]. Indeed, a recent meta-analysis indicates the possibility of ribavirin being more detrimental than beneficial in patients presenting with a mild infection [140]. Doubts regarding the efficiency of ribavirin were also raised by an early clinical trial performed by the US Centers for Disease Control and Prevention in Sierra Leone that found little evidence of efficacy of ribavirin, but an increased risk of death for patients with aspartate transaminase (AST) levels below 150 IU/l [141]. Additional concerns have been raised by the fact that orally delivered ribavirin may not reach concentrations shown to inhibit LASV (26.5-82 µM in vitro IC_{50} [117], and that some in vivo studies have failed to show an effect of ribavirin on viremia [118, 129]. These observations clearly need to be taken into consideration when reassessing recommended guidelines for treating LFD. With regard to LUJV, among the two of the five patients who received ribavirin, one subsequently died. While it is interesting to note that the patient who survived was the only one to receive intravenous ribavirin [126], these extremely limited case numbers currently preclude evaluation of the efficacy of ribavirin against LUJV.

More recently, favipiravir (Table 16.2) has been shown to be effective against LASV infection in vitro with an IC_{50} of 29.3 μ M [118], and in guinea pigs and NHPs at a dosage of 300 mg/kg [129, 130]. Further, the possibility of combination therapies between favipiravir and ribavirin are being pursued with promising results in a lethal mouse model [118]. Moreover, this combination was also used in the field for two patients infected from the same index case [131]. Both patients survived and showed a decrease in viremia after administration of combination treatment. However, as the authors emphasize, there remains a lack of evidence to definitively attribute this decrease in viremia to (co-)treatment with favipiravir, and further clinical studies are clearly required to assess its efficacy against human LASV infection.

16.4.1.2.2 Immunotherapies

Early in vivo work in NHPs showed the potential of a combination of plasma from convalescent animals that survived LASV infection together with ribavirin to increase survival when administered up to seven days postinfection [142]. Further, nAb concentration in the immune plasma used for treatment was found to be crucial for treatment efficacy [143–145]. However, the utility of this approach is hampered by the limited cross-reactivity of sera against different LASV strains [143, 145], which are genetically highly diverse [146]. Additionally, nAb titers are low in convalescent plasma from LASV-infected patients. Perhaps as a result, a prospective cohort study failed to see a significant benefit of human convalescent plasma in patients when administered as monotherapy [132], while mixed results have been obtained from other clinical studies and case reports. These include the

successful treatment of a laboratory-acquired infection [147], variable results when immune plasma was administered early after onset of symptoms [148, 149], or even lack of any beneficial effect [150]. As an alternative to convalescent immune plasma, the use of human mAbs appears to be a viable option. In a recent study performed in NHPs, B cells isolated from convalescent donors were used to produce human mAbs. Treatment with a cocktail of these human mAbs led to 100% survival when administered immediately postchallenge [151]. However, while this suggests a possible utility for treatment of known accidental exposures (i.e. in a hospital or laboratory setting), further studies are clearly also still needed to assess the efficacy of these mAbs when given at a more clinically relevant stage of the infection.

16.4.2 New World Arenaviruses

New World Arenaviruses are divided into three clades A, B, and C, with all of those known to cause VHF belonging to Clade B. These include Junín virus (JUNV), Machupo virus (MACV), Guanarito virus (GTOV), Sabiá virus (SABV), and Chapare virus (CHPV), which are the etiological agents for Argentine, Bolivian, Venezuelan, Brazilian, and Chapare hemorrhagic fever, respectively. They are typically classified as BSL4 agents, although in some countries work with JUNV is possible under enhanced BSL-3 conditions and/or in restricted access areas by vaccinated personnel [152–154]. While other New World Arenaviruses are also able to infect humans, and can in some cases cause disease, they are normally nonfatal and infection is primarily laboratory acquired or identified by serological investigations. However, Whitewater Arroyo virus (a recombinant clade A/B virus) was suggested to be responsible for three fatal infections in California in 1999/2000 [155].

16.4.2.1 Disease Overview

Among the highly pathogenic New World Arenaviruses associated with VHF, JUNV infection is by far the most prevalent. Outbreaks in the pampas areas of Argentina have been recorded since the 1950s, with the virus being isolated in 1958. Worryingly, the endemic area has been growing constantly, correlating with the intensification of agricultural practices in this area, and likely resulting in increased contact between human populations and the rodent reservoir host Calomys musculus (reviewed in [156]). The incidence of Argentine hemorrhagic fever was up to 3500 cases per year prior to the introduction of the Candid#1 live-attenuated vaccine, which became available in the early 1990s, and with a CFR that ranged from 15 to 30% in untreated patients [156]. Similarly, in 1959, an outbreak of VHF in the rural area of Beni Department in Bolivia led to the identification of the closely related MACV (reviewed in [157]). Since then, sporadic outbreaks have been recorded in the same area, which corresponds to the region where a specific lineage of Calomys callosus, the reservoir host, is found [157]. While the number of cases is limited, the average CFR ranges from 25 to 35%, and a recent increase in disease incidence has been observed [157]. Three decades later, in 1989, GTOV was recognized and isolated in Venezuela, where several hundred cases have been documented to date [158]. The virus reservoir is the cotton rat Sigmodon alstoni [159], which thrives in rural areas of the country. Also here, infection seems to be seasonally associated with agricultural work [160].

While the aforementioned New World Arenaviruses are highly endemic and human infections occur on a regular basis, SABV and CHPV infections are considerably scarcer. SABV was first reported in 1990 in Brazil [161], and only three subsequent cases have been documented, two of which were laboratory-acquired infections (reviewed in [162]). Both naturally acquired infections were fatal. The reservoir for this virus has so far not been identified despite an attempt to survey rodents in the area [163]. Finally, CHPV was identified in Bolivia in 2008, after several cases of VHF were reported in a single province (Chapare province, Cochabamba Department), with one fatal case [164]. In June 2019, more than 15 years after the first outbreak, five new cases including three deaths were reported [165]. The reservoir host for this virus is not known.

Zoonotic transmission of New World Arenaviruses generally occurs via rodent urine and feces. Based on the greater case numbers, disease progression has been best described for JUNV. Here, as for other VHF diseases, early symptoms are rather unspecific, with patients showing signs of febrile illness after a 6- to 14-day incubation period. In severe cases, which represent 30% of patients, the second phase of the disease is characterized by signs of hemorrhage in about half of the cases (bleeding from mucosa, hematemesis, and petechia), and/or neurological symptoms (encephalopathy, delirium, tremor coma, and seizures). About eight days after onset of symptoms, surviving patients enter the convalescent phase, which can last for up to three months, and in which they display general weakness, hair loss, and tachycardia [166–168]. A similar disease progression is also observed for other VHFs caused by New World Arenaviruses [160, 161, 164, 169].

16.4.2.2 Antiviral Strategies

JUNV is the only New World Arenavirus, and indeed currently the only BSL4 virus for which both an effective human vaccine [170–172] and a highly effective specific treatment (based on plasma from convalescent patients) [173, 174] are available and are being used routinely. However, decreasing case numbers (due to the success of the vaccination program) threaten the long-term viability of convalescent plasma banking, while funding limitations make continued vaccine availability uncertain. Further, the lack of similar vaccine and treatment options for other New World Arenaviruses emphasizes the continued need to develop additional treatment options.

16.4.2.2.1 Immunotherapies

JUNV is the only arenavirus for which treatment with plasma from convalescent patients has been developed into a clearly defined therapeutic regimen with demonstrated clinical benefit. An initial double-blind randomized study in the late 1970s already showed that such treatment was able to decrease mortality from 16.5 to 1.1% [174]. A prospective cohort study further confirmed these results and showed a dose-dependent response of patients to the treatment [173]. In this study, all patients treated with a dose of nAbs over 3000 therapeutic units (TU)/kg survived and survival gradually decreased with lower doses [173]. The currently defined standard of care for JUNV infection involves plasma infusions corresponding to at least 3500 TU/kg of neutralizing antibody, an approach that is highly effective if started within eight days after onset of symptoms [167]. While this treatment approach is highly efficient at controlling the acute infection and preventing death, a significant complication is the occurrence of a nonfatal late neurological syndrome in 10% of

treated patients following initial convalescence [174]. The mechanism underlying the development of this syndrome remains unknown; however, similar observations have also been made in various animal models [175–177].

The relatively low number of cases for other New World Arenaviruses coupled with their often sporadic occurrence has so far limited the possibility to evaluate the efficacy of similar treatments. However, in principle, such an approach is feasible since passive immunotherapy using plasma from human donors previously infected with MACV decreased mortality in NHPs when given early after onset of symptoms, albeit with the occurrence of late neurological syndrome [178]. Although such treatments in human patients have so far shown only limited success [179, 180], an immune plasma bank was created in Bolivia in 2010, in collaboration with the Pan American Health Organization (PAHO) and World Health Organization (WHO), to face future outbreaks. Unfortunately, even if such treatments were proven to be effective for other New World Arenaviruses, there would likely be significant difficulties in obtaining sufficient amounts of high-titer plasma. Nonetheless, the possibility to use plasma from patients that have recovered from infection with a different New World Arenavirus has been suggested due to the existence of cross-neutralizing antibodies. However, it remains unclear whether such sera would have sufficiently potent cross-neutralizing activity to be clinically effective. Alternatively, research aiming at the identification of mAbs able to efficiently cross-neutralize multiple species of New World Arenaviruses is a promising approach, and some preliminary studies in this direction have already been performed [181, 182].

16.4.2.2.2 Small Molecules

Based on its use in the treatment of the related LASV, ribavirin (Table 16.3) was one of the first treatment options investigated for New World Arenaviruses. It was successful in controlling infection with JUNV in vitro at IC_{50} s ranging from 5 to $13 \mu M$ [183] as well as in vivo in both rats and NHPs at dosages above 15 mg/kg daily [122, 123, 184]. More recently, however, more mixed results were obtained in guinea pigs, with ribavirin increasing survival time, but without always improving CFR [185]. Similarly, a prospective study performed on six patients treated with ribavirin between 9 and 11 days after onset of symptoms using a loading dose of 34 mg/kg followed by 17 mg/kg every six hours for four days and then 8 mg/kg every eight hours for six days showed an increased survival time but no impact on CFR. However, this may have been due to the very late onset of treatment [127] due to ethical limitations regarding the use of alternative therapies during the window within which the standard of care (i.e. plasma therapy) is known to be highly effective. Ribavirin is currently not recommended for use in JUNV-infected patients, but rather the use of immunotherapy is advised. However, it is considered a viable treatment option for VHFs caused by other New World Arenaviruses, or in cases where immunotherapy is not accessible (i.e. imported infections), even if data examining the efficacy of its use against those specific pathogens are scarce [186]. In this regard, one in vivo study has shown that ribavirin was able to induce a dose-dependent decrease of viremia in NHPs infected with MACV at dosages above 10 mg/kg every 12 hours [124]. Further, intravenous ribavirin was used experimentally in two of three suspected cases of MACV infection. Both patients who received ribavirin survived, but laboratory confirmation of MACV infection was only conducted for one of them [187]. Intravenous ribavirin was also successfully used as PEP in a case of occupational exposure to SABV with a 30 mg/kg loading dose followed by 15 mg/kg every six hours for four days and then 7.5 mg/kg every eight hours for six days [128]. Overall, while these data are clearly too limited to draw strong conclusions, they at least suggest a possible benefit of ribavirin treatment. Consistent with this, during the June 2019 outbreak of CHPV, PAHO and WHO emphasized that there are currently no clinical trials demonstrating ribavirin efficacy against this virus and advised caution regarding its use [165].

16.5 Bunyaviruses

Bunyaviruses are tri-segmented single-stranded negative-sense RNA viruses of the order *Bunyavirales*. Those that are recognized as presenting a significant risk to human health are members of the families *Nairoviridae* (genus *Orthonairovirus*), *Phenuiviridae* (genus *Phlebovirus*), *Peribunyaviridae* (genus *Orthobunyavirus*), and *Hantaviridae* (genus *Orthohantavirus*). While many bunyaviruses can cause human disease of varying severity, including Rift Valley Fever virus, La Crosse virus, Hantaan virus, and Oropouche virus, only Crimean Congo Hemorrhagic Fever Virus (CCHFV) is widely classified as a BSL4 agent.

16.5.1 Disease Overview

CCHFV was first recognized in 1944 after an outbreak in the Crimean peninsula, and an identical virus was isolated from a patient in the former Belgian Congo in 1956. However, it was only more than a decade later that the identity of both viruses was recognized. CCHFV is endemic to a vast region spanning Europe, Asia, and Africa, where it circulates in a variety of wild and domestic mammals, including cattle, goat, and sheep. Transmission of CCHFV is primarily facilitated by ticks of the genus Hyalomma. However, infection of humans also occurs as a result of direct contact with blood from infected animals (primarily agricultural animals), and via human-to-human transmission through contact with body fluids. Indeed, the seroprevalence in farm animals can be as high as 80% in endemic areas [188], while in humans in these areas, it is on average around 4.5%, with a higher risk in specific populations such as farmers, butchers, and slaughterhouse workers [188]. While most infections are believed to be asymptomatic, in symptomatic cases the disease starts with a nonspecific febrile illness after an incubation time ranging from one to seven days. Three to five days after onset of symptoms, severe cases may show signs of hemorrhage. High viremia and a lack of virus-specific IgG production, as well as clinical signs including ecchymoses and gingival bleeding, are associated with a poor prognosis [189, 190]. Reported CFRs in outbreaks vary widely from 3 to 80% and high CFRs are usually reported in small outbreaks, suggesting a bias in case detection (reviewed in [191]). There also appear to be strain-specific differences in disease severity [192]. In fatal cases, death usually occurs between 5 and 14 days after the onset of symptoms due to shock and multi-organ failure. The convalescent phase in CCHF survivors can last up to a year and is characterized by general weakness, tachycardia, and sometimes other symptoms like memory loss [193].

16.5.2 Antiviral Strategies

As for most BSL4 agents, there is no specific treatment for CCHFV. An inactivated vaccine produced in neonatal mouse brain was developed in the former Soviet Union and is still in use in Bulgaria [194, 195]. However, due to safety concerns and limitations in the scale of production, as well as a lack of data regarding its efficacy against most strains of CCHFV, it appears unlikely that its use will be extended. Thus, the development of effective antiviral strategies is clearly needed. Unfortunately, the lack of adequate animal models has for decades impaired the preclinical evaluation of treatments against this disease (reviewed in [196]). Indeed, it was only in 2018 that a cynomolgus macaque model that recapitulates CCHFV infection in humans was first reported [197], However, issues with reproducibility, possibly due to the outbred nature of the animals [198], indicate that there are still significant challenges that have to be overcome in this area.

16.5.2.1 Immunotherapies

Immunotherapy has only been used in a limited number of CCHFV-infected patients. A study described the benefit of treating severe cases of CCHFV with an intravenous preparation of immunoglobulin from the plasma of donors vaccinated with the CCHFV Bulgarian vaccine. All patients recovered, but there was no control group [199, 200]. Two other studies used sera from convalescent patients to treat CCHFV during outbreaks in South Africa and Dubai [201, 202]. In the South African outbreak, all patients treated by passive immunotherapy survived and the two patients who died did not receive immune serum due to delayed diagnosis [201]. During the outbreak in Dubai, only one of the seven patients received immunotherapy, and that individual subsequently survived the infection [202]. Similarly, hyperimmune globulin against CCHFV was prepared from pooled plasma of multiple CCHFV survivors and used to treat patients with viral loads above 10⁸ copies/ml. Almost 90% survival was observed, while historically without treatment the CFR in such cases is close to 90% [203]. However, while certainly encouraging, none of these studies have included a proper control arm or had a significant number of patients based on which to draw firm conclusions. The WHO is currently advocating additional studies on the development of mAbs for CCHF treatment [4], and while to date only a few such studies have been conducted, they show encouraging results both in vitro and in animal models [204–206].

16.5.2.2 Small Molecules

Currently, CCHFV treatment relies mostly on supportive care, together with the use of ribavirin, which, though controversial, is generally recommended. In vitro data have shown an inhibitory effect of ribavirin (Table 16.4) on CCHFV with IC_{50} s ranging from 11.5 to 56.6 μ M, as have in vivo studies in neonatal mice using dosages above 50 mg/kg [207–209, 227]. Deep sequencing analysis in a single patient after treatment suggests that ribavirin has a mutagenic effect on CCHFV, thereby driving the virus into error catastrophe [222]. Numerous retrospective cohort studies [210–213, 215, 219–221, 228, 229], case-control studies [218], and case reports [216, 217, 222–224, 230, 231] on the efficacy of ribavirin have been conducted and most of them have suggested a beneficial effect. However, the only randomized clinical trial performed to date failed to show a beneficial effect of ribavirin [214]. Additionally, three meta-analyses concluded that there are neither beneficial nor

Table 16.4 Overview of small molecules tested against infection with CCHEV.

Name	Structure	Mechanism of action	In vitro effect	In vivo effect	Clinical regimen
Ribavirin		 Polymerase inhibitor Mutagenic agent Increases antiviral immune response Increases cell survival 	 IC₅₀ or ED₅₀ of 11.5-65.5 μM in Vero cells [207, 208] 	 50 or 100 mg/kg, single or daily dose for four days in neonatal mice [209] 	 30 mg/kg initial loading dose, then 15 mg/kg every six hours for four days, then 7.5 mg/kg every eight hours for six days orally [210-214] 4 g daily for four days, then 2.4 g daily for six days [215-217] 17 mg/kg loading dose, then 17 mg/kg loading dose, then 17 mg/kg every six hours for four days, then 8 mg/kg every eight hours for six days IV [218] 4 g daily for six days [219] 2 g loading dose, then 1 g every six hours for four days, then 500 mg every eight hours for six days [220, 221] 1 g every six hours for one day, then 1 g every six hours for one day IV, then 500 mg every eight hor four days IV [222] 2 g loading dose, 4g daily for four days, then 2 g daily for four days, then 2 g daily for six days [223, 224]

Favipiravir (T-705) $ \begin{array}{c} & \text{Polymerase inhibitor} \\ & \text{Mutagenic agent} \end{array} $ • Polymerase inhibitor • Mutagenic agent • $IC_{50} \text{ of } 1.03 \mu\text{M in} \\ & \text{Huh7 cells [225]} \\ & IC_{50} \text{ of } 7 \mu\text{M in Vero cells [207]} \end{array} $ • $IS \text{mg/kg, 30 or 300 \text{mg/}} \\ & \text{g daily in IFNAR}^{-/-} \\ & \text{mice [55]} \\ & \text{of ravipiravir + 100 \text{mg/kg}} \\ & \text{of ravipiravir + 100 \text{mg/kg}} \\ & \text{of ravipiravir + 100 \text{mg/kg}} \\ & \text{of ravipiravir - 100 \text{mg/kg}} \\ & \text{of ravipiravir - 100 \text{mg/kg}} \\ & \text{of ravipiravir - 100 \text{mg/kg}} \\ & \text{mice [55]} \\ & \text{of ravipiravir - 100 \text{mg/kg}} \\ & \text{of ravipiravir - 100 \text{mg/kg}} \\ & \text{mice [55]} \\ & \text{of ravipiravir - 100 \text{mg/kg}} \\ & \text{mice [256]} \\ \end{array} $	Name	Structure	Mechanism of action	In vitro effect	In vivo effect	Clinical regimen
	Favipiravir (T-705)	OH OH NH2	 Polymerase inhibitor Mutagenic agent 	$ \begin{tabular}{lllllllllllllllllllllllllllllllllll$	 15 mg/kg, 30 or 300 mg/ kg daily in IFNAR^{-/-} mice [55] 7.5 or 30 mg/kg daily of favipiravir + 100 mg/kg of ribavirin in IFNAR^{-/-} mice [55] 300 mg/kg daily in IFNAR^{-/-} mice [226] 	Not determined

detrimental effects of ribavirin treatment [232–234]. This is consistent with the results of the largest cohort study to date, which showed no beneficial effect of ribavirin in patients with a mild infection, and a potential detrimental effect in patients with a severe infection displaying impaired renal and liver functions [212]. Additionally, two of the meta-analyses evaluated the benefit of early versus late administration of ribavirin (i.e. before or after five days after onset of symptoms), and both concluded that there is no beneficial effect of an early administration of the drug [233, 234]. One of these meta-analyses featured a particularly thorough analysis of both included and excluded studies [234], noting a number of limitations in existing studies. In particular, many studies use historical records as their control arm, resulting in the risk that patients received a significantly different quality of supportive care. Also, the dosage and route of administration of ribavirin, while in the same range, varies between available studies. Thus, a proper clinical trial, in the context of good supportive care, remains necessary to fully evaluate the efficacy of ribavirin for the treatment of CCHFV.

Similar to other VHF, the use of favipiravir (Table 16.4) could also be beneficial. To date, it has been shown to exert antiviral effects alone or in combination with ribavirin in vitro, with an IC_{50} between 1 and 7μ M, and in vivo against multiple strains of CCHFV in preliminary studies at dosages above 15 mg/kg daily, or as low as 7.5 mg/kg if used in combination with 100 mg/kg ribavirin daily [207, 225, 226]. Further, a recent study has shown decreased viremia in NHPs treated with a 300 mg/kg dosage of favipiravir starting 24 hours post infection [235], and future clinical trials with this molecule against CCHFV are on the WHO priority list [4]. A major concern in the treatment of CCHFV is the finding that the virus can persist in treated animals and cause a delayed fatal infection [226], highlighting the possible need for combination therapy against CCHFV, as well as a close follow-up of patients during and after treatment.

16.6 Considerations for the Development of Treatment Strategies Against Viral Hemorrhagic Fever Viruses

The ability to conduct high-quality trials for BSL4 agents is often hampered by the epidemic nature of most of these diseases, coupled with limited numbers of patients and/or poor infrastructure in affected areas [236]. Combined with the lack of economic incentives for commercial development of such treatments, this has been a major hurdle to the development and testing of therapies. Further, even where such therapies exist, they may face significant logistical problems with respect to production and/or availability. This advocates for multiple treatment options against a given pathogen in order to protect against changes in the availability of a single treatment or shortages during outbreaks, thereby ensuring sustainable management of these diseases. Indeed, this also would help to ensure that products are available that are suitable for specific applications, for instance, in patients with severe vomiting where oral treatment may be difficult, or in locations where the medical infrastructure needed for antibody delivery may be unavailable, an issue that can be particularly important in the treatment of these exotic infections. A multipronged treatment approach would also increase the chances to have effective treatment options available in the event of newly emerging virus species/strains, which-depending on their makeup—may not be adequately managed with all approaches.

Despite the fact that the major limitation in the treatment of infections with BSL4 pathogens is the lack of available therapies with demonstrated efficacy, the testing of potentially novel treatments for viruses where a standard of care (whether truly effective or not) has already been established presents special challenges. This is most notable for infection with JUNV, where the testing of antiviral compounds has so far been restricted to patients already outside the treatment window for convalescent plasma therapy. Thus, while these treatment attempts have been relatively unsuccessful [127], it is clear that treatment of these patients represents a major challenge, and that such approaches might be much more successful if initiated earlier. In particular, it would appear that testing of novel therapies in combination with the existing standard of care might offer a way forward. Indeed, this approach is being pursued for treatment of LASV (where there is conflicting evidence about the efficacy of the standard therapy alone) with apparent success at the level of small animal model testing [118].

At the same time, however, the high CFRs associated with many of these infections make it very tempting to use potential treatments even if only limited information regarding their efficiency in animal models is available. An example is the development of countermeasures for filovirus infection, which was highly promoted during the West African Ebola outbreak in 2013–2016. In this case, soon after the outbreak was declared a public health emergency of international concern, the WHO decided that it was ethically acceptable to start clinical trials for EBOV countermeasures showing promising results in laboratory and animal models, but for which safety and efficacy had not yet been tested in humans [237]. However, most of the treatments failed to show efficacy, and for some of these studies considerable ethical concerns have been raised [238]. This experience clearly demonstrates the need for sufficient in vivo data using predictive animal models before conducting clinical trials in humans, even in outbreak scenarios. Further, careful study planning, although challenging, remains critical to obtain high-quality data to support the establishment of well-supported treatment guidelines that will truly be effective.

Finally, with respect to the use of treatments for which only limited data may be available, the research communities working with BSL4 agents present an additional dilemma. For most of these agents no vaccines or treatment options have been licensed (and even where this has been achieved, there remain major hurdles regarding their availability in practise), posing the difficult question what to do in the event of a laboratory-acquired infection. While in specific cases access to experimental therapies has been made available in the past, the management of such cases is currently left to be dealt with on an emergency case-by-case basis with little or no formal planning or coordination. While there are considerable legal obstacles that may prevent the import and stockpiling of experimental medications for use under such circumstances, the consequences of not doing so are potentially dramatic and raise significant ethical issues per se. Additionally, there are logistical issues that surround acquiring and coordinating the access to these medications, and in some cases also maintenance of the expertise needed to deliver them correctly.

It is important to note that the same issues arise in relation to our ability to treat imported infections with these agents, something that is increasingly likely in a highly connected and mobile world. To address this problem, conditions have to be generated that facilitate rapid and reliable access to both licensed therapies (where they exist) as well as the use of experimental treatment options, and particularly to allow stockpiling of the necessary therapeutics or PEP for use in emergencies in countries with BSL4 laboratories.

Nonetheless, while there are many challenges inherent in the development and testing of therapeutic options for BSL4 viruses, past and recent successes exist and demonstrate that treating these diseases is clearly possible. What is critically needed, and unfortunately often lacking, is sustained political interest and continuous support at the national and international level. On the other hand, increased public awareness of the potential for even these relatively rare and exotic viruses to cause large outbreaks, including the potential for significant spread to non-endemic areas, may be a reason for cautious optimism regarding the development and availability of treatment options with proven efficacy against BSL4 agents in the future.

References

- 1 WHO (2018a). Ebola/Marburg research and development (R&D) roadmap. https://www. who.int/blueprint/priority-diseases/key-action/Ebola-Marburg_Draft_Roadmap_ publiccomment_MAY2018.pdf (accessed on 1 December 2019).
- **2** WHO (2018b). Lassa fever research and development (R&D) roadmap. https://www.who. int/blueprint/priority-diseases/key-action/LassaFever_Draft_Roadmap_publiccomment_ MAY2018.pdf (accessed on 1 December 2019).
- **3** WHO (2018c). Nipah research and development (R&D) roadmap. https://www.who.int/ blueprint/priority-diseases/key-action/Nipah_Draft_Roadmap_publiccomment_MAY2018. pdf (accessed on 1 December 2019).
- **4** WHO (2018d). Roadmap for research and product development against Crimean-Congo haemorrhagic fever (CCHF). https://www.who.int/blueprint/priority-diseases/key-action/ cchf-draft-r-and-d-roadmap.pdf (accessed on 1 December 2019).
- 5 Siegert, R., Shu, H.L., Slenczka, W. et al. (1967). On the etiology of an unknown human infection originating from monkeys. *Dtsch. Med. Wochenschr.* 92 (51): 2341–2343.
- **6** Siegert, R., Shu, H.L., Slenczka, W. et al. (1968). The aetiology of an unknown human infection transmitted by monkeys (preliminary communication). *Ger. Med. Mon.* 13 (1): 1–2.
- 7 CDC (2014). Known cases and outbreaks of Marburg hemorrhagic fever, in chronological order. https://www.cdc.gov/vhf/marburg/outbreaks/chronology.html (accessed on 1 December 2019).
- **8** CDC (2019). 40 years of Ebola virus disease around the world 2019. https://www.cdc.gov/ vhf/ebola/history/chronology.html (accessed on 1 December 2019).
- **9** WHO (2019). Ebola Virus Disease Democratic Republic of Congo: External Situation Report 68/2019. https://www.who.int/publications-detail/ebola-virus-disease-democratic-republic-of-congo-external-situation-report-68-2019 (accessed on 1 December 2019).
- **10** Kuhn, J.H., Adachi, T., Adhikari, N.K.J. et al. (2019). New filovirus disease classification and nomenclature. *Nat. Rev. Microbiol.* 17 (5): 261–263.
- **11** Miranda, M.E., White, M.E., Dayrit, M.M. et al. (1991). Seroepidemiological study of filovirus related to Ebola in the Philippines. *Lancet* 337 (8738): 425–426.
- **12** Negredo, A., Palacios, G., Vazquez-Moron, S. et al. (2011). Discovery of an ebolavirus-like filovirus in europe. *PLoS Pathog.* 7 (10): e1002304.

- **436** 16 Current Therapies for Biosafety Level 4 Pathogens
 - **13** Kemenesi, G., Kurucz, K., Dallos, B. et al. (2018). Re-emergence of Lloviu virus in Miniopterus schreibersii bats, Hungary, 2016. *Emerg. Microbes Infect.* 7 (1): 66.
 - 14 Shi, M., Lin, X.D., Chen, X. et al. (2018). The evolutionary history of vertebrate RNA viruses. *Nature* 556 (7700): 197–202.
 - **15** Emanuel, J., Marzi, A., and Feldmann, H. (2018). Filoviruses: ecology, molecular biology, and evolution. *Adv. Virus Res.* 100: 189–221.
 - 16 Chertow, D.S., Kleine, C., Edwards, J.K. et al. (2014). Ebola virus disease in West Africa – clinical manifestations and management. N. Engl. J. Med. 371 (22): 2054–2057.
 - 17 Kortepeter, M.G., Bausch, D.G., and Bray, M. (2011). Basic clinical and laboratory features of filoviral hemorrhagic fever. J. Infect. Dis. 204 (Suppl 3): S810–S816.
 - **18** Lever, R.A. and Whitty, C.J. (2016). Ebola virus disease: emergence, outbreak and future directions. *Br. Med. Bull.* 117 (1): 95–106.
 - **19** Rojek, A., Horby, P., and Dunning, J. (2017). Insights from clinical research completed during the west Africa Ebola virus disease epidemic. *Lancet Infect. Dis.* 17 (9): e280–e292.
 - **20** Zhang, X., Rong, Y., Sun, L. et al. (2015). Prognostic analysis of patients with Ebola virus disease. *PLoS Negl. Trop. Dis.* 9 (9): e0004113.
 - **21** Schieffelin, J.S., Shaffer, J.G., Goba, A. et al. (2014). Clinical illness and outcomes in patients with Ebola in Sierra Leone. *N. Engl. J. Med.* 371 (22): 2092–2100.
 - 22 Fitzpatrick, G., Vogt, F., Moi Gbabai, O.B. et al. (2015). The contribution of Ebola viral load at admission and other patient characteristics to mortality in a Medecins Sans Frontieres Ebola Case Management Centre, Kailahun, Sierra Leone, June–October 2014. J. Infect. Dis. 212 (11): 1752–1758.
 - **23** Vetter, P., Kaiser, L., Schibler, M. et al. (2016). Sequelae of Ebola virus disease: the emergency within the emergency. *Lancet Infect. Dis.* 16 (6): e82–e91.
 - 24 Lotsch, F., Schnyder, J., Goorhuis, A., and Grobusch, M.P. (2017). Neuropsychological long-term sequelae of Ebola virus disease survivors a systematic review. *Travel Med. Infect. Dis.* 18: 18–23.
 - 25 Saphire, E.O., Schendel, S.L., Gunn, B.M. et al. (2018). Antibody-mediated protection against Ebola virus. *Nat. Immunol.* 19 (11): 1169–1178.
 - **26** van Griensven, J., Edwards, T., de Lamballerie, X. et al. (2016). Evaluation of convalescent plasma for Ebola virus disease in Guinea. *N. Engl. J. Med.* 374 (1): 33–42.
 - 27 van Griensven, J., Edwards, T., Baize, S., and Ebola-Tx, Consortium. (2016). Efficacy of convalescent plasma in relation to dose of Ebola virus antibodies. *N. Engl. J. Med.* 375 (23): 2307–2309.
 - **28** Qiu, X., Alimonti, J.B., Melito, P.L. et al. (2011). Characterization of Zaire ebolavirus glycoprotein-specific monoclonal antibodies. *Clin. Immunol.* 141 (2): 218–227.
 - **29** Wilson, J.A., Hevey, M., Bakken, R. et al. (2000). Epitopes involved in antibody-mediated protection from Ebola virus. *Science* 287 (5458): 1664–1666.
 - **30** Davidson, E., Bryan, C., Fong, R.H. et al. (2015). Mechanism of binding to Ebola virus glycoprotein by the ZMapp, ZMAb, and MB-003 cocktail antibodies. *J. Virol.* 89 (21): 10982–10992.
 - 31 Davey, R.T. Jr., Dodd, L., Proschan, M.A. et al. (2016). A randomized, controlled trial of ZMapp for Ebola virus infection. *N. Engl. J. Med.* 375 (15): 1448–1456.
 - Pascal, K.E., Dudgeon, D., Trefry, J.C. et al. (2018). Development of clinical-stage human monoclonal antibodies that treat advanced Ebola virus disease in nonhuman primates. *J. Infect. Dis.* 218 (suppl 5): S612–S626.

- **33** Sivapalasingam, S., Kamal, M., Slim, R. et al. (2018). Safety, pharmacokinetics, and immunogenicity of a co-formulated cocktail of three human monoclonal antibodies targeting Ebola virus glycoprotein in healthy adults: a randomised, first-in-human phase 1 study. *Lancet Infect. Dis.* 18 (8): 884–893.
- **34** Corti, D., Misasi, J., Mulangu, S. et al. (2016). Protective monotherapy against lethal Ebola virus infection by a potently neutralizing antibody. *Science* 351 (6279): 1339–1342.
- Misasi, J., Gilman, M.S., Kanekiyo, M. et al. (2016). Structural and molecular basis for Ebola virus neutralization by protective human antibodies. *Science* 351 (6279): 1343–1346.
- **36** Gaudinski, M.R., Coates, E.E., Novik, L. et al. (2019). Safety, tolerability, pharmacokinetics, and immunogenicity of the therapeutic monoclonal antibody mAb114 targeting Ebola virus glycoprotein (VRC 608): an open-label phase 1 study. *Lancet* 393 (10174): 889–898.
- 37 Mulangu, S., Dodd, L.E., Davey, R.T. Jr. et al. (2019). A randomized, controlled trial of Ebola virus disease therapeutics. *N. Engl. J. Med.* 381 (24): 2293–2303.
- 38 Jones, S.M., Feldmann, H., Stroher, U. et al. (2005). Live attenuated recombinant vaccine protects nonhuman primates against Ebola and Marburg viruses. *Nat. Med.* 11 (7): 786–790.
- **39** Marzi, A. and Mire, C.E. (2019). Current Ebola virus vaccine progress. *BioDrugs* 33 (1): 9–14.
- **40** Feldmann, H., Jones, S.M., Daddario-DiCaprio, K.M. et al. (2007). Effective post-exposure treatment of Ebola infection. *PLoS Pathog.* 3 (1): e2.
- **41** Daddario-DiCaprio, K.M., Geisbert, T.W., Stroher, U. et al. (2006). Postexposure protection against Marburg haemorrhagic fever with recombinant vesicular stomatitis virus vectors in non-human primates: an efficacy assessment. *Lancet* 367 (9520): 1399–1404.
- **42** Geisbert, T.W., Hensley, L.E., Geisbert, J.B. et al. (2010). Postexposure treatment of Marburg virus infection. *Emerg. Infect. Dis.* 16 (7): 1119–1122.
- Gunther, S., Feldmann, H., Geisbert, T.W. et al. (2011). Management of accidental exposure to Ebola virus in the biosafety level 4 laboratory, Hamburg, Germany. *J. Infect. Dis.* 204 (Suppl 3): S785–S790.
- **44** Wong, K.K., Davey, R.T. Jr., Hewlett, A.L. et al. (2016). Use of postexposure prophylaxis after occupational exposure to Zaire Ebolavirus. *Clin. Infect. Dis.* 63 (3): 376–379.
- **45** Salata, C., Baritussio, A., Munegato, D. et al. (2015). Amiodarone and metabolite MDEA inhibit Ebola virus infection by interfering with the viral entry process. *Pathog. Dis.* 73 (5).
- **46** Dyall, J., Johnson, J.C., Hart, B.J. et al. (2018). In vitro and in vivo activity of amiodarone against Ebola virus. *J. Infect. Dis.* 218 (suppl_5): S592–S596.
- **47** Gehring, G., Rohrmann, K., Atenchong, N. et al. (2014). The clinically approved drugs amiodarone, dronedarone and verapamil inhibit filovirus cell entry. *J. Antimicrob. Chemother.* 69 (8): 2123–2131.
- **48** Turone, F. (2014). Doctors trial amiodarone for Ebola in Sierra Leone. *BMJ* 349: g7198.
- **49** Warren, T.K., Wells, J., Panchal, R.G. et al. (2014). Protection against filovirus diseases by a novel broad-spectrum nucleoside analogue BCX4430. *Nature* 508 (7496): 402–405.
- **50** Madelain, V., Nguyen, T.H., Olivo, A. et al. (2016). Ebola virus infection: review of the pharmacokinetic and pharmacodynamic properties of drugs considered for testing in human efficacy trials. *Clin. Pharmacokinet.* 55 (8): 907–923.
- **51** Warren, T., MacLennan, S., Mathis, A. et al. (2017). Efficacy of galidesivir against Ebola virus disease in Rhesus monkeys. *Open Forum Infect. Dis.* 4 (suppl 1): S302–S.

- **52** McMullan, L.K., Flint, M., Dyall, J. et al. (2016). The lipid moiety of brincidofovir is required for in vitro antiviral activity against Ebola virus. *Antivir. Res.* 125: 71–78.
- **53** Dunning, J., Kennedy, S.B., Antierens, A. et al. (2016). Experimental treatment of Ebola virus disease with brincidofovir. *PLoS ONE* 11 (9): e0162199.
- **54** McCarthy, S.D., Majchrzak-Kita, B., Racine, T. et al. (2016). A rapid screening assay identifies monotherapy with interferon- β and combination therapies with nucleoside analogs as effective inhibitors of Ebola virus. *PLoS Negl. Trop. Dis.* 10 (1): e0004364.
- **55** Oestereich, L., Ludtke, A., Wurr, S. et al. (2014). Successful treatment of advanced Ebola virus infection with T-705 (favipiravir) in a small animal model. *Antivir. Res.* 105: 17–21.
- **56** Bixler, S.L., Bocan, T.M., Wells, J. et al. (2018). Efficacy of favipiravir (T-705) in nonhuman primates infected with Ebola virus or Marburg virus. *Antivir. Res.* 151: 97–104.
- **57** Guedj, J., Piorkowski, G., Jacquot, F. et al. (2018). Antiviral efficacy of favipiravir against Ebola virus: a translational study in cynomolgus macaques. *PLoS Med.* 15 (3): e1002535.
- **58** Kerber, R., Lorenz, E., Duraffour, S. et al. (2019). Laboratory findings, compassionate use of favipiravir, and outcome in patients with Ebola virus disease, Guinea, 2015 a retrospective observational study. *J. Infect. Dis.* 220 (2): 195–202.
- **59** Sissoko, D., Laouenan, C., Folkesson, E. et al. (2016). Experimental treatment with favipiravir for Ebola virus disease (the JIKI trial): a historically controlled, single-arm proof-of-concept trial in Guinea. *PLoS Med.* 13 (3): e1001967.
- 60 Bai, C.Q., Mu, J.S., Kargbo, D. et al. (2016). Clinical and virological characteristics of Ebola virus disease patients treated with favipiravir (T-705)-Sierra Leone, 2014. *Clin. Infect. Dis.* 63 (10): 1288–1294.
- **61** Warren, T.K., Jordan, R., Lo, M.K. et al. (2016). Therapeutic efficacy of the small molecule GS-5734 against Ebola virus in rhesus monkeys. *Nature* 531 (7594): 381–385.
- **62** Lo, M.K., Jordan, R., Arvey, A. et al. (2017). GS-5734 and its parent nucleoside analog inhibit filo-, pneumo-, and paramyxoviruses. *Sci. Rep.* 7: 43395.
- **63** Salata, C., Munegato, D., Martelli, F. et al. (2018). Amiodarone affects Ebola virus binding and entry into target cells. *N. Microbiol.* 41 (2): 162–164.
- **64** Delang, L., Abdelnabi, R., and Neyts, J. (2018). Favipiravir as a potential countermeasure against neglected and emerging RNA viruses. *Antivir. Res.* 153: 85–94.
- **65** Principi, N., Camilloni, B., Alunno, A. et al. (2019). Drugs for Influenza treatment: is there significant news? *Front. Med. (Lausanne)* 6: 109.
- 66 Nguyen, T.H., Guedj, J., Anglaret, X. et al. (2017). Favipiravir pharmacokinetics in Ebola-Infected patients of the JIKI trial reveals concentrations lower than targeted. *PLoS Negl. Trop. Dis.* 11 (2): e0005389.
- **67** Warren, T.K., Whitehouse, C.A., Wells, J. et al. (2015). A single phosphorodiamidate morpholino oligomer targeting VP24 protects rhesus monkeys against lethal Ebola virus infection. *MBio* 6 (1): e02344-14.
- **68** Iversen, P.L., Warren, T.K., Wells, J.B. et al. (2012). Discovery and early development of AVI-7537 and AVI-7288 for the treatment of Ebola virus and Marburg virus infections. *Viruses* 4 (11): 2806–2830.
- **69** Geisbert, T.W., Lee, A.C., Robbins, M. et al. (2010). Postexposure protection of non-human primates against a lethal Ebola virus challenge with RNA interference: a proof-of-concept study. *Lancet* 375 (9729): 1896–1905.

- **70** Thi, E.P., Lee, A.C., Geisbert, J.B. et al. (2016). Rescue of non-human primates from advanced Sudan ebolavirus infection with lipid encapsulated siRNA. *Nat. Microbiol.* 1 (10): 16142.
- 71 Heald, A.E., Iversen, P.L., Saoud, J.B. et al. (2014). Safety and pharmacokinetic profiles of phosphorodiamidate morpholino oligomers with activity against Ebola virus and Marburg virus: results of two single-ascending-dose studies. *Antimicrob. Agents Chemother.* 58 (11): 6639–6647.
- 72 Davidson, B.L. and McCray, P.B. Jr. (2011). Current prospects for RNA interference-based therapies. *Nat. Rev. Genet.* 12 (5): 329–340.
- **73** Dunning, J., Sahr, F., Rojek, A. et al. (2016). Experimental treatment of Ebola virus disease with TKM-130803: a single-arm phase 2 clinical trial. *PLoS Med.* 13 (4): e1001997.
- **74** Villinger, F., Rollin, P.E., Brar, S.S. et al. (1999). Markedly elevated levels of interferon (IFN)-gamma, IFN-alpha, interleukin (IL)-2, IL-10, and tumor necrosis factor-alpha associated with fatal Ebola virus infection. *J. Infect. Dis.* 179 (Suppl 1): S188–S191.
- 75 Konde, M.K., Baker, D.P., Traore, F.A. et al. (2017). Interferon beta-1a for the treatment of Ebola virus disease: a historically controlled, single-arm proof-of-concept trial. *PLoS ONE* 12 (2): e0169255.
- **76** Murray, K., Selleck, P., Hooper, P. et al. (1995). A morbillivirus that caused fatal disease in horses and humans. *Science* 268 (5207): 94–97.
- 77 Field, H.E. (2016). Hendra virus ecology and transmission. Curr. Opin. Virol. 16: 120–125.
- **78** Halpin, K., Young, P.L., Field, H.E., and Mackenzie, J.S. (2000). Isolation of Hendra virus from pteropid bats: a natural reservoir of Hendra virus. *J Gen Virol*. 81 (Pt 8): 1927–1932.
- **79** Tan, R., Hodge, A., Klein, R. et al. (2018). Virus-neutralising antibody responses in horses following vaccination with Equivac(R) HeV: a field study. *Aust. Vet. J.* 96 (5): 161–166.
- **80** Mahalingam, S., Herrero, L.J., Playford, E.G. et al. (2012). Hendra virus: an emerging paramyxovirus in Australia. *Lancet Infect. Dis.* 12 (10): 799–807.
- **81** Weatherman, S., Feldmann, H., and de Wit, E. (2018). Transmission of henipaviruses. *Curr. Opin. Virol.* 28: 7–11.
- 82 CDC (1999). Outbreak of Hendra-like virus—Malaysia and Singapore, 1998–1999. MMWR Morb. Mortal. Wkly Rep. 48 (13): 265–269.
- **83** Yob, J.M., Field, H., Rashdi, A.M. et al. (2001). Nipah virus infection in bats (order Chiroptera) in peninsular Malaysia. *Emerg. Infect. Dis.* 7 (3): 439–441.
- 84 Ang, B.S.P., Lim, T.C.C., and Wang, L. (2018). Nipah virus infection. J. Clin. Microbiol. 56 (6).
- **85** Tan, K.-S., Tan, C.-T., and Goh, K.-J. (1999). Epidemiological aspects of Nipah virus infection. *Neurol. J. Southeast Asia* 4: 77–81.
- 86 Chew, M.H., Arguin, P.M., Shay, D.K. et al. (2000). Risk factors for Nipah virus infection among abattoir workers in Singapore. J. Infect. Dis. 181 (5): 1760–1763.
- **87** Thakur, N. and Bailey, D. (2019). Advances in diagnostics, vaccines and therapeutics for Nipah virus. *Microbes Infect.* 21 (7): 278–286.
- 88 Luby, S.P., Hossain, M.J., Gurley, E.S. et al. (2009). Recurrent zoonotic transmission of Nipah virus into humans, Bangladesh, 2001–2007. *Emerg. Infect. Dis.* 15 (8): 1229–1235.
- 89 Chua, K.B., Lam, S.K., Goh, K.J. et al. (2001). The presence of Nipah virus in respiratory secretions and urine of patients during an outbreak of Nipah virus encephalitis in Malaysia. *J Infect.* 42 (1): 40–43.

- **90** Goh, K.J., Tan, C.T., Chew, N.K. et al. (2000). Clinical features of Nipah virus encephalitis among pig farmers in Malaysia. *N. Engl. J. Med.* 342 (17): 1229–1235.
- **91** Hossain, M.J., Gurley, E.S., Montgomery, J.M. et al. (2008). Clinical presentation of nipah virus infection in Bangladesh. *Clin. Infect. Dis.* 46 (7): 977–984.
- **92** Banerjee, S., Gupta, N., Kodan, P. et al. (2019). Nipah virus disease: a rare and intractable disease. *Intractable Rare Dis Res.* 8 (1): 1–8.
- **93** Paton, N.I., Leo, Y.S., Zaki, S.R. et al. (1999). Outbreak of Nipah-virus infection among abattoir workers in Singapore. *Lancet* 354 (9186): 1253–1256.
- **94** Lo, M.K., Feldmann, F., Gary, J.M. et al. (2019). Remdesivir (GS-5734) protects African green monkeys from Nipah virus challenge. *Sci. Transl. Med.* 11 (494).
- **95** Playford, E.G., McCall, B., Smith, G. et al. (2010). Human Hendra virus encephalitis associated with equine outbreak, Australia, 2008. *Emerg. Infect. Dis.* 16 (2): 219–223.
- **96** Aljofan, M., Saubern, S., Meyer, A.G. et al. (2009). Characteristics of Nipah virus and Hendra virus replication in different cell lines and their suitability for antiviral screening. *Virus Res.* 142 (1–2): 92–99.
- **97** Rockx, B., Bossart, K.N., Feldmann, F. et al. (2010). A novel model of lethal Hendra virus infection in African green monkeys and the effectiveness of ribavirin treatment. *J. Virol.* 84 (19): 9831–9839.
- **98** Chong, H.T., Kamarulzaman, A., Tan, C.T. et al. (2001). Treatment of acute Nipah encephalitis with ribavirin. *Ann. Neurol.* 49 (6): 810–813.
- **99** Chandni, R., Renjith, T.P., Fazal, A. et al. (2020). Clinical manifestations of Nipah virus-infected patients who presented to the emergency department during an outbreak in Kerala State in India, May 2018. *Clin. Infect. Dis.* 71 (1): 152–157.
- **100** Wright, P.J., Crameri, G., and Eaton, B.T. (2005). RNA synthesis during infection by Hendra virus: an examination by quantitative real-time PCR of RNA accumulation, the effect of ribavirin and the attenuation of transcription. *Arch. Virol.* 150 (3): 521–532.
- **101** Zhu, Z., Dimitrov, A.S., Bossart, K.N. et al. (2006). Potent neutralization of Hendra and Nipah viruses by human monoclonal antibodies. *J. Virol.* 80 (2): 891–899.
- **102** Zhu, Z., Bossart, K.N., Bishop, K.A. et al. (2008). Exceptionally potent cross-reactive neutralization of Nipah and Hendra viruses by a human monoclonal antibody. *J. Infect. Dis.* 197 (6): 846–853.
- **103** Geisbert, T.W., Mire, C.E., Geisbert, J.B. et al. (2014). Therapeutic treatment of Nipah virus infection in nonhuman primates with a neutralizing human monoclonal antibody. *Sci. Transl. Med.* 6 (242): 242ra82.
- **104** Broder, C.C., Weir, D.L., and Reid, P.A. (2016). Hendra virus and Nipah virus animal vaccines. *Vaccine* 34 (30): 3525–3534.
- **105** Palacios, G., Druce, J., Du, L. et al. (2008). A new arenavirus in a cluster of fatal transplant-associated diseases. *N. Engl. J. Med.* 358 (10): 991–998.
- Anesi, J.A. and Silveira, F.P. (2019). Practice ASTIDCo. Arenaviruses and West Nile Virus in solid organ transplant recipients: guidelines from the American Society of Transplantation Infectious Diseases Community of Practice. *Clin. Transpl.* 33 (9): e13576.
- 107 Frame, J.D., Baldwin, J.M. Jr., Gocke, D.J., and Troup, J.M. (1970). Lassa fever, a new virus disease of man from West Africa. I. Clinical description and pathological findings. *Am. J. Trop. Med. Hyg.* 19 (4): 670–676.

- **108** McCormick, J.B., Webb, P.A., Krebs, J.W. et al. (1987). A prospective study of the epidemiology and ecology of Lassa fever. *J. Infect. Dis.* 155 (3): 437–444.
- 109 Tobin, E., Asogun, D., Akpede, N. et al. (2015). Lassa fever in Nigeria: insights into seroprevalence and risk factors in rural Edo State: a pilot study. *J. Med. Trop.* 17 (2): 51–55.
- Yalley-Ogunro, J.E., Frame, J.D., and Hanson, A.P. (1984). Endemic Lassa fever in Liberia.
 VI. Village serological surveys for evidence of Lassa virus activity in Lofa County, Liberia.
 Trans. R. Soc. Trop. Med. Hyg. 78 (6): 764–770.
- **111** McCormick, J.B., King, I.J., Webb, P.A. et al. (1987). A case–control study of the clinical diagnosis and course of Lassa fever. *J. Infect. Dis.* 155 (3): 445–455.
- **112** Fisher-Hoch, S.P., Tomori, O., Nasidi, A. et al. (1995). Review of cases of nosocomial Lassa fever in Nigeria: the high price of poor medical practice. *BMJ* 311 (7009): 857–859.
- **113** Richmond, J.K. and Baglole, D.J. (2003). Lassa fever: epidemiology, clinical features, and social consequences. *BMJ* 327 (7426): 1271–1275.
- 114 Mateer, E.J., Huang, C., Shehu, N.Y., and Paessler, S. (2018). Lassa fever-induced sensorineural hearing loss: a neglected public health and social burden. *PLoS Negl. Trop. Dis.* 12 (2): e0006187.
- **115** Ishii, A., Thomas, Y., Moonga, L. et al. (2011). Novel arenavirus, Zambia. *Emerg. Infect. Dis.* 17 (10): 1921–1924.
- **116** Ishii, A., Thomas, Y., Moonga, L. et al. (2012). Molecular surveillance and phylogenetic analysis of Old World arenaviruses in Zambia. *J. Gen. Virol.* 93 (Pt 10): 2247–2251.
- **117** Bausch, D.G., Hadi, C.M., Khan, S.H., and Lertora, J.J. (2010). Review of the literature and proposed guidelines for the use of oral ribavirin as postexposure prophylaxis for Lassa fever. *Clin. Infect. Dis.* 51 (12): 1435–1441.
- **118** Oestereich, L., Rieger, T., Ludtke, A. et al. (2016). Efficacy of favipiravir alone and in combination with ribavirin in a lethal, immunocompetent mouse model of Lassa fever. *J. Infect. Dis.* 213 (6): 934–938.
- **119** Huggins, J.W. (1989). Prospects for treatment of viral hemorrhagic fevers with ribavirin, a broad-spectrum antiviral drug. *Rev. Infect. Dis.* 11 (Suppl 4): S750–S761.
- **120** Jahrling, P.B., Hesse, R.A., Eddy, G.A. et al. (1980). Lassa virus infection of rhesus monkeys: pathogenesis and treatment with ribavirin. *J. Infect. Dis.* 141 (5): 580–589.
- **121** Stephen, E.L. and Jahrling, P.B. (1979). Experimental Lassa fever virus infection successfully treated with ribavirin. *Lancet* 1 (8110): 268–269.
- **122** McKee, K.T. Jr., Huggins, J.W., Trahan, C.J., and Mahlandt, B.G. (1988). Ribavirin prophylaxis and therapy for experimental argentine hemorrhagic fever. *Antimicrob. Agents Chemother.* 32 (9): 1304–1309.
- **123** Weissenbacher, M.C., Calello, M.A., Merani, M.S. et al. (1986). Therapeutic effect of the antiviral agent ribavirin in Junin virus infection of primates. *J. Med. Virol.* 20 (3): 261–267.
- **124** Stephen, E.L., Jones, D.E., Peters, C.J. et al. (1979). Ribavirin Treatment of Toga-, Arenaand Bunyavirus Infections in Subhuman Primates and Other Laboratory Animal Species. Defense Technical Information Center: Fort Belvoir, VA.
- 125 Todd W, Lockwood D, Sundar S. Viral haemorrhagic fevers. Davidson Principles and Practice of Medicine. 20th ed. Boon N, Colledge N, Walker B, Hunter J, editors. Philadelphia: Elsevier; 2006.

- 442 16 Current Therapies for Biosafety Level 4 Pathogens
 - **126** Sewlall, N.H., Richards, G., Duse, A. et al. (2014). Clinical features and patient management of Lujo hemorrhagic fever. *PLoS Negl. Trop. Dis.* 8 (11): e3233.
 - **127** Enria, D.A., Briggiler, A.M., Levis, S. et al. (1987). Tolerance and antiviral effect of ribavirin in patients with Argentine hemorrhagic fever. *Antivir. Res.* 7 (6): 353–359.
 - **128** Barry, M., Russi, M., Armstrong, L. et al. (1995). Brief report: treatment of a laboratoryacquired Sabia virus infection. *N. Engl. J. Med.* 333 (5): 294–296.
 - **129** Safronetz, D., Rosenke, K., Westover, J.B. et al. (2015). The broad-spectrum antiviral favipiravir protects guinea pigs from lethal Lassa virus infection post-disease onset. *Sci. Rep.* 5: 14775.
 - **130** Rosenke, K., Feldmann, H., Westover, J.B. et al. (2018). Use of favipiravir to treat Lassa virus infection in Macaques. *Emerg. Infect. Dis.* 24 (9): 1696–1699.
 - **131** Raabe, V.N., Kann, G., Ribner, B.S. et al. (2017). Favipiravir and ribavirin treatment of epidemiologically linked cases of Lassa fever. *Clin. Infect. Dis.* 65 (5): 855–859.
 - **132** McCormick, J.B., King, I.J., Webb, P.A. et al. (1986). Lassa fever. Effective therapy with ribavirin. *N. Engl. J. Med.* 314 (1): 20–26.
 - **133** Moreno, H., Gallego, I., Sevilla, N. et al. (2011). Ribavirin can be mutagenic for arenaviruses. *J. Virol.* 85 (14): 7246–7255.
 - **134** Buba, M.I., Dalhat, M.M., Nguku, P.M. et al. (2018). Mortality among confirmed Lassa fever cases during the 2015–2016 outbreak in Nigeria. *Am. J. Publ. Health* 108 (2): 262–264.
 - **135** Asogun, D.A., Adomeh, D.I., Ehimuan, J. et al. (2012). Molecular diagnostics for lassa fever at Irrua specialist teaching hospital, Nigeria: lessons learnt from two years of laboratory operation. *PLoS Negl. Trop. Dis.* 6 (9): e1839.
 - **136** Shaffer, J.G., Grant, D.S., Schieffelin, J.S. et al. (2014). Lassa fever in post-conflict sierra leone. *PLoS Negl. Trop. Dis.* 8 (3): e2748.
 - **137** Ajayi, N.A., Nwigwe, C.G., Azuogu, B.N. et al. (2013). Containing a Lassa fever epidemic in a resource-limited setting: outbreak description and lessons learned from Abakaliki, Nigeria (January–March 2012). *Int. J. Infect. Dis.* 17 (11): e1011–e1016.
 - **138** Kyei, N.N., Abilba, M.M., Kwawu, F.K. et al. (2015). Imported Lassa fever: a report of 2 cases in Ghana. *BMC Infect. Dis.* 15: 217.
 - 139 Dahmane, A., van Griensven, J., Van Herp, M. et al. (2014). Constraints in the diagnosis and treatment of Lassa Fever and the effect on mortality in hospitalized children and women with obstetric conditions in a rural district hospital in Sierra Leone. *Trans. R. Soc. Trop. Med. Hyg.* 108 (3): 126–132.
 - **140** Eberhardt, K.A., Mischlinger, J., Jordan, S. et al. (2019). Ribavirin for the treatment of Lassa fever: a systematic review and meta-analysis. *Int. J. Infect. Dis.* 87: 15–20.
 - **141** Bodycombe, D. and Davis, H. (1992). Final Report Analysis of a Clinical Trial Ribavirin and the Treatment of Lassa Fever. *Report*, 7 *February 1992, SUBJECT: IND 16666 Ribavirin (Virazole) (Serial No. 011).*
 - 142 Jahrling, P.B., Peters, C.J., and Stephen, E.L. (1984). Enhanced treatment of Lassa fever by immune plasma combined with ribavirin in cynomolgus monkeys. J. Infect. Dis. 149 (3): 420–427.
 - **143** Jahrling, P.B. (1983). Protection of Lassa virus-infected guinea pigs with Lassa-immune plasma of guinea pig, primate, and human origin. *J. Med. Virol.* 12 (2): 93–102.
 - 144 Jahrling, P.B., Frame, J.D., Rhoderick, J.B., and Monson, M.H. (1985). Endemic Lassa fever in Liberia. IV. Selection of optimally effective plasma for treatment by passive immunization. *Trans. R. Soc. Trop. Med. Hyg.* 79 (3): 380–384.

- 145 Jahrling, P.B. and Peters, C.J. (1984). Passive antibody therapy of Lassa fever in cynomolgus monkeys: importance of neutralizing antibody and Lassa virus strain. *Infect. Immun.* 44 (2): 528–533.
- **146** Andersen, K.G., Shapiro, B.J., Matranga, C.B. et al. (2015). Clinical sequencing uncovers origins and evolution of Lassa virus. *Cell* 162 (4): 738–750.
- **147** Leifer, E., Gocke, D.J., and Bourne, H. (1970). Lassa fever, a new virus disease of man from West Africa. II. Report of a laboratory-acquired infection treated with plasma from a person recently recovered from the disease. *Am. J. Trop. Med. Hyg.* 19 (4): 677–679.
- **148** Frame, J.D., Verbrugge, G.P., Gill, R.G., and Pinneo, L. (1984). The use of Lassa fever convalescent plasma in Nigeria. *Trans. R. Soc. Trop. Med. Hyg.* 78 (3): 319–324.
- 149 Clayton, A.J. (1977). Lassa immune serum. Bull. World Health Organ. 55 (4): 435-439.
- **150** White, H.A. (1972). Lassa fever. A study of 23 hospital cases. *Trans. R. Soc. Trop. Med. Hyg.* 66 (3): 390–401.
- **151** Mire, C.E., Cross, R.W., Geisbert, J.B. et al. (2017). Human-monoclonal-antibody therapy protects nonhuman primates against advanced Lassa fever. *Nat. Med.* 23 (10): 1146–1149.
- **152** ABSA (2019). Risk group database. https://my.absa.org/Riskgroups (accessed on 1 December 2019).
- **153** CDC, NIH. Biosafety in Microbiological and Biomedical Laboratories. 5th ed. Washington, DC: U.S. Department of Health and Human Services; 2009.
- **154** (2006). Se inaugura hoy el mayor laboratorio de bioseguridad de América latina. *La Nacion.*
- **155** CDC (2000). Fatal illnesses associated with a new world arenavirus California, 1999–2000. *MMWR Morb. Mortal. Wkly Rep.* 49 (31): 709–711.
- 156 Enria, D.A., Briggiler, A.M., and Feuillade, M.R. (1998). An overview of the epidemiological, ecological and preventive hallmarks of Argentine haemorrhagic fever (Junin virus). *Bull. de l'Institut Pasteur.* 96 (2): 103–114.
- **157** Patterson, M., Grant, A., and Paessler, S. (2014). Epidemiology and pathogenesis of Bolivian hemorrhagic fever. *Curr. Opin. Virol.* 5: 82–90.
- 158 Tesh, R.B., Jahrling, P.B., Salas, R., and Shope, R.E. (1994). Description of Guanarito virus (Arenaviridae: Arenavirus), the etiologic agent of Venezuelan hemorrhagic fever. *Am. J. Trop. Med. Hyg.* 50 (4): 452–459.
- **159** Tesh, R.B., Wilson, M.L., Salas, R. et al. (1993). Field studies on the epidemiology of Venezuelan hemorrhagic fever: implication of the cotton rat Sigmodon alstoni as the probable rodent reservoir. *Am. J. Trop. Med. Hyg.* 49 (2): 227–235.
- **160** de Manzione, N., Salas, R.A., Paredes, H. et al. (1998). Venezuelan hemorrhagic fever: clinical and epidemiological studies of 165 cases. *Clin. Infect. Dis.* 26 (2): 308–313.
- **161** Lisieux, T., Coimbra, M., Nassar, E.S. et al. (1994). New arenavirus isolated in Brazil. *Lancet* 343 (8894): 391–392.
- **162** Ellwanger, J.H. and Chies, J.A. (2017). Keeping track of hidden dangers—the short history of the Sabia virus. *Rev. Soc. Bras. Med. Trop.* 50 (1): 3–8.
- **163** Bisordi, I., Levis, S., Maeda, A.Y. et al. (2015). Pinhal virus, a new arenavirus isolated from Calomys tener in Brazil. *Vector Borne Zoonotic Dis.* 15 (11): 694–700.

- 444 16 Current Therapies for Biosafety Level 4 Pathogens
 - **164** Delgado, S., Erickson, B.R., Agudo, R. et al. (2008). Chapare virus, a newly discovered arenavirus isolated from a fatal hemorrhagic fever case in Bolivia. *PLoS Pathog.* 4 (4): e1000047.
 - 165 PAHO (2019). Epidemiological alert hemorrhagic fever due to arenavirus in Bolivia. https://www.paho.org/hq/index.php?option=com_docman&view=download&category_ slug=bolivian-hemorrhagic-fever-2221&alias=49428-18-july-2019-hemorrhagic-fever-dueto-arenavirus-in-bolivia-epidemiological-alert-1&Itemid=270&lang=en. (accessed on 1 December 2019).
 - **166** Maiztegui, J.I. (1975). Clinical and epidemiological patterns of Argentine haemorrhagic fever. *Bull. World Health Organ.* 52 (4–6): 567–575.
 - 167 Enria, D.A., Briggiler, A.M., and Sanchez, Z. (2008). Treatment of Argentine hemorrhagic fever. Antivir. Res. 78 (1): 132–139.
 - **168** Harrison, L.H., Halsey, N.A., McKee, K.T. Jr. et al. (1999). Clinical case definitions for Argentine hemorrhagic fever. *Clin. Infect. Dis.* 28 (5): 1091–1094.
 - **169** Mackenzie, R.B., Beye, H.K., Valverde, L., and Garron, H. (1964). Epidemic hemorrhagic fever in Bolivia. I. A preliminary report of the epidemiologic and clinical findings in a new epidemic area in South America. *Am. J. Trop. Med. Hyg.* 13: 620–625.
 - 170 McKee, K.T. Jr., Oro, J.G., Kuehne, A.I. et al. (1993). Safety and immunogenicity of a live-attenuated Junin (Argentine hemorrhagic fever) vaccine in rhesus macaques. *Am. J. Trop. Med. Hyg.* 48 (3): 403–411.
 - Maiztegui, J.I., McKee, K.T. Jr., Barrera Oro, J.G. et al. (1998). Protective efficacy of a live attenuated vaccine against Argentine hemorrhagic fever. *J Infect Dis.* 177 (2): 277–283.
 - **172** Ambrosio, A.M., Riera, L.M., Saavedra Mdel, C., and Sabattini, M.S. (2006). Immune response to vaccination against Argentine hemorrhagic Fever in an area where different arenaviruses coexist. *Viral. Immunol.* 19 (2): 196–201.
 - **173** Enria, D.A., Briggiler, A.M., Fernandez, N.J. et al. (1984). Importance of dose of neutralising antibodies in treatment of Argentine haemorrhagic fever with immune plasma. *Lancet* 2 (8397): 255–256.
 - **174** Maiztegui, J.I., Fernandez, N.J., and de Damilano, A.J. (1979). Efficacy of immune plasma in treatment of Argentine haemorrhagic fever and association between treatment and a late neurological syndrome. *Lancet* 2 (8154): 1216–1217.
 - 175 Kenyon, R.H., Green, D.E., Eddy, G.A., and Peters, C.J. (1986). Treatment of Junin virus-infected guinea pigs with immune serum: development of late neurological disease. *J. Med. Virol.* 20 (3): 207–218.
 - **176** Avila, M.M., Samoilovich, S.R., Laguens, R.P. et al. (1987). Protection of Junin virusinfected marmosets by passive administration of immune serum: association with late neurologic signs. *J. Med. Virol.* 21 (1): 67–74.
 - **177** Weissenbacher, M.C., Avila, M.M., Calello, M.A. et al. (1986). Effect of ribavirin and immune serum on Junin virus-infected primates. *Med. Microbiol. Immunol.* 175 (2–3): 183–186.
 - **178** Eddy, G.A., Wagner, F.S., Scott, S.K., and Mahlandt, B.J. (1975). Protection of monkeys against Machupo virus by the passive administration of Bolivian haemorrhagic fever immunoglobulin (human origin). *Bull. World Health Organ.* 52 (4–6): 723–727.
 - **179** Stinebaugh, B.J., Schloeder, F.X., Johnson, K.M. et al. (1966). Bolivian hemorrhagic fever. A report of four cases. *Am. J. Med.* 40 (2): 217–230.
- **180** Aguilar, P.V., Camargo, W., Vargas, J. et al. (2009). Reemergence of Bolivian hemorrhagic fever, 2007–2008. *Emerg. Infect. Dis.* 15 (9): 1526–1528.
- **181** Zeitlin, L., Geisbert, J.B., Deer, D.J. et al. (2016). Monoclonal antibody therapy for Junin virus infection. *Proc. Natl. Acad. Sci. USA* 113 (16): 4458–4463.
- 182 Clark, L.E., Mahmutovic, S., Raymond, D.D. et al. (2018). Vaccine-elicited receptorbinding site antibodies neutralize two New World hemorrhagic fever arenaviruses. *Nat. Commun.* 9 (1): 1884.
- **183** Rodriguez, M., McCormick, J.B., and Weissenbacher, M.C. (1986). Antiviral effect of ribavirin on Junin virus replication in vitro. *Rev. Argent. Microbiol.* 18 (2): 69–74.
- **184** Remesar, M.C., Blejer, J.L., Weissenbacher, M.C., and Nejamkis, M.R. (1988). Ribavirin effect on experimental Junin virus-induced encephalitis. *J. Med. Virol.* 26 (1): 79–84.
- **185** Salazar, M., Yun, N.E., Poussard, A.L. et al. (2012). Effect of ribavirin on Junin virus infection in guinea pigs. *Zoonoses Publ. Health* 59 (4): 278–285.
- **186** Veliziotis, I., Roman, A., Martiny, D. et al. (2020). Clinical management of argentine hemorrhagic fever using Ribavirin and Favipiravir, Belgium. *Emerg. Infect. Dis.* 26 (7): 1562–1566.
- **187** Kilgore, P.E., Ksiazek, T.G., Rollin, P.E. et al. (1997). Treatment of Bolivian hemorrhagic fever with intravenous ribavirin. *Clin. Infect. Dis.* 24 (4): 718–722.
- **188** Nasirian, H. (2019). Crimean-Congo hemorrhagic fever (CCHF) seroprevalence: a systematic review and meta-analysis. *Acta Trop.* 196: 102–120.
- 189 Ozturk, B., Tutuncu, E., Kuscu, F. et al. (2012). Evaluation of factors predictive of the prognosis in Crimean-Congo hemorrhagic fever: new suggestions. *Int. J. Infect. Dis.* 16 (2): e89–e93.
- Cevik, M.A., Erbay, A., Bodur, H. et al. (2008). Clinical and laboratory features of Crimean-Congo hemorrhagic fever: predictors of fatality. *Int. J. Infect. Dis.* 12 (4): 374–379.
- **191** Bente, D.A., Forrester, N.L., Watts, D.M. et al. (2013). Crimean-Congo hemorrhagic fever: history, epidemiology, pathogenesis, clinical syndrome and genetic diversity. *Antivir. Res.* 100 (1): 159–189.
- **192** Midilli, K., Gargili, A., Ergonul, O. et al. (2009). The first clinical case due to AP92 like strain of Crimean-Congo hemorrhagic fever virus and a field survey. *BMC Infect. Dis.* 9: 90.
- **193** Metanat, M., Sharifi Mood, B., and Masoud, S. (2014). Crimean-Congo hemorrhagic fever: a reemerging disease. *Arch. Clin. Infect. Dis.* 9 (1).
- **194** Papa, A., Papadimitriou, E., and Christova, I. (2011). The Bulgarian vaccine Crimean-Congo haemorrhagic fever virus strain. *Scand. J. Infect. Dis.* 43 (3): 225–229.
- **195** Mousavi-Jazi, M., Karlberg, H., Papa, A. et al. (2012). Healthy individuals' immune response to the Bulgarian Crimean-Congo hemorrhagic fever virus vaccine. *Vaccine* 30 (44): 6225–6229.
- **196** Garrison, A.R., Smith, D.R., and Golden, J.W. (2019). Animal models for Crimean-Congo hemorrhagic fever human disease. *Viruses* 11 (7).
- **197** Haddock, E., Feldmann, F., Hawman, D.W. et al. (2018). A cynomolgus macaque model for Crimean-Congo haemorrhagic fever. *Nat. Microbiol.* 3 (5): 556–562.

- 446 16 Current Therapies for Biosafety Level 4 Pathogens
 - Cross, R.W., Prasad, A.N., Borisevich, V. et al. (2020). Crimean-Congo hemorrhagic fever virus strains Hoti and Afghanistan cause viremia and mild clinical disease in cynomolgus monkeys. *PLoS Negl. Trop. Dis.* 14 (8): e0008637.
 - Vassilenko, S.M., Vassilev, T.L., Bozadjiev, L.G. et al. (1990). Specific intravenous immunoglobulin for Crimean-Congo haemorrhagic fever. *Lancet* 335 (8692): 791–792.
 - Vassilev, T., Valchev, V., Kazarov, G. et al. (1991). A reference preparation for human immunoglobulin against Crimean/Congo hemorrhagic fever. *Biologicals* 19 (1): 57.
 - van Eeden, P.J., van Eeden, S.F., Joubert, J.R. et al. (1985). A nosocomial outbreak of Crimean-Congo haemorrhagic fever at Tygerberg Hospital. Part II. Management of patients. *S. Afr. Med. J.* 68 (10): 718–721.
 - Suleiman, M.N., Muscat-Baron, J.M., Harries, J.R. et al. (1980). Congo/Crimean haemorrhagic fever in Dubai. An outbreak at the Rashid Hospital. *Lancet* 2 (8201): 939–941.
 - Kubar, A., Haciomeroglu, M., Ozkul, A. et al. (2011). Prompt administration of Crimean-Congo hemorrhagic fever (CCHF) virus hyperimmunoglobulin in patients diagnosed with CCHF and viral load monitorization by reverse transcriptase-PCR. *Jpn. J. Infect. Dis.* 64 (5): 439–443.
 - 204 Bertolotti-Ciarlet, A., Smith, J., Strecker, K. et al. (2005). Cellular localization and antigenic characterization of Crimean-Congo hemorrhagic fever virus glycoproteins. *J. Virol.* 79 (10): 6152–6161.
 - Golden, J.W., Shoemaker, C.J., Lindquist, M.E. et al. (2019). GP38-targeting monoclonal antibodies protect adult mice against lethal Crimean-Congo hemorrhagic fever virus infection. *Sci. Adv.* 5 (7): eaaw9535.
 - Zivcec, M., Guerrero, L.I.W., Albarino, C.G. et al. (2017). Identification of broadly neutralizing monoclonal antibodies against Crimean-Congo hemorrhagic fever virus. *Antivir. Res.* 146: 112–120.
 - Oestereich, L., Rieger, T., Neumann, M. et al. (2014). Evaluation of antiviral efficacy of ribavirin, arbidol, and T-705 (favipiravir) in a mouse model for Crimean-Congo hemorrhagic fever. *PLoS Negl. Trop. Dis.* 8 (5): e2804.
 - Rusnak, J.M. (2011). Experience with ribavirin for treatment and postexposure prophylaxis of hemorrhagic fever viruses: crimean Congo hemorrhagic fever, Lassa fever, and hantaviruses. *Appl. Biosaf.* 16 (2): 67–87.
 - Tignor, G.H. and Hanham, C.A. (1993). Ribavirin efficacy in an in vivo model of Crimean-Congo hemorrhagic fever virus (CCHF) infection. *Antivir. Res.* 22 (4): 309–325.
 - Mardani, M., Jahromi, M.K., Naieni, K.H., and Zeinali, M. (2003). The efficacy of oral ribavirin in the treatment of Crimean-Congo hemorrhagic fever in Iran. *Clin. Infect. Dis.* 36 (12): 1613–1618.
 - Alavi-Naini, R., Moghtaderi, A., Koohpayeh, H.R. et al. (2006). Crimean-Congo hemorrhagic fever in Southeast of Iran. *J. Infect.* 52 (5): 378–382.
 - 212 Elaldi, N., Bodur, H., Ascioglu, S. et al. (2009). Efficacy of oral ribavirin treatment in Crimean-Congo haemorrhagic fever: a quasi-experimental study from Turkey. *J. Infect.* 58 (3): 238–244.
 - Sharifi-Mood, B., Alavi-Naini, R., Metanat, M., and Rakhshani, F. (2006). Ribavirin: an effective drug for treatment of children with Crimean-Congo hemorrhagic fever: a seven-years experience. *Pak. J. Biol. Sci.* 9 (8): 1598–1600.

- 214 Koksal, I., Yilmaz, G., Aksoy, F. et al. (2010). The efficacy of ribavirin in the treatment of Crimean-Congo hemorrhagic fever in Eastern Black Sea region in Turkey. *J. Clin. Virol.* 47 (1): 65–68.
- 215 Bodur, H., Erbay, A., Akinci, E. et al. (2011). Effect of oral ribavirin treatment on the viral load and disease progression in Crimean-Congo hemorrhagic fever. *Int. J. Infect. Dis.* 15 (1): e44–e47.
- **216** Ergonul, O., Celikbas, A., Dokuzoguz, B. et al. (2004). Characteristics of patients with Crimean-Congo hemorrhagic fever in a recent outbreak in Turkey and impact of oral ribavirin therapy. *Clin. Infect. Dis.* 39 (2): 284–287.
- **217** Fisher-Hoch, S.P., Khan, J.A., Rehman, S. et al. (1995). Crimean Congo-haemorrhagic fever treated with oral ribavirin. *Lancet* 346 (8973): 472–475.
- **218** Cevik, M.A., Elaldi, N., Akinci, E. et al. (2008). A preliminary study to evaluate the effect of intravenous ribavirin treatment on survival rates in Crimean-Congo hemorrhagic fever. *J. Infect.* 57 (4): 350–351.
- **219** Dokuzoguz, B., Celikbas, A.K., Gok, S.E. et al. (2013). Severity scoring index for Crimean-Congo hemorrhagic fever and the impact of ribavirin and corticosteroids on fatality. *Clin. Infect. Dis.* 57 (9): 1270–1274.
- **220** Izadi, S. and Salehi, M. (2009). Evaluation of the efficacy of ribavirin therapy on survival of Crimean-Congo hemorrhagic fever patients: a case–control study. *Jpn. J. Infect. Dis.* 62 (1): 11–15.
- **221** Ozkurt, Z., Kiki, I., Erol, S. et al. (2006). Crimean-Congo hemorrhagic fever in Eastern Turkey: clinical features, risk factors and efficacy of ribavirin therapy. *J. Infect.* 52 (3): 207–215.
- 222 Espy, N., Perez-Sautu, U., Ramirez de Arellano, E. et al. (2018). Ribavirin had demonstrable effects on the Crimean-Congo hemorrhagic fever virus (CCHFV) population and load in a patient with CCHF infection. J. Infect. Dis. 217 (12): 1952–1956.
- **223** Athar, M.N., Baqai, H.Z., Ahmad, M. et al. (2003). Short report: Crimean-Congo hemorrhagic fever outbreak in Rawalpindi, Pakistan, February 2002. *Am. J. Trop. Med. Hyg.* 69 (3): 284–287.
- **224** Bangash, S.A. and Khan, E.A. (2003). Treatment and prophylaxis with ribavirin for Crimean-Congo hemorrhagic fever is it effective? *J. Pak. Med. Assoc.* 53 (1): 39–41.
- **225** Welch, S.R., Scholte, F.E.M., Flint, M. et al. (2017). Identification of 2'-deoxy-2'fluorocytidine as a potent inhibitor of Crimean-Congo hemorrhagic fever virus replication using a recombinant fluorescent reporter virus. *Antivir. Res.* 147: 91–99.
- **226** Hawman, D.W., Haddock, E., Meade-White, K. et al. (2018). Favipiravir (T-705) but not ribavirin is effective against two distinct strains of Crimean-Congo hemorrhagic fever virus in mice. *Antivir. Res.* 157: 18–26.
- Watts, D.M., Ussery, M.A., Nash, D., and Peters, C.J. (1989). Inhibition of Crimean-Congo hemorrhagic fever viral infectivity yields in vitro by ribavirin. *Am. J. Trop. Med. Hyg.* 41 (5): 581–585.
- **228** Sharifi-Mood, B., Metanat, M., Ghorbani-Vaghei, A. et al. (2009). The outcome of patients with Crimean-Congo hemorrhagic fever in Zahedan, Southeast of Iran: a comparative study. *Arch. Iran Med.* 12 (2): 151–153.

- 448 16 Current Therapies for Biosafety Level 4 Pathogens
 - **229** Tasdelen Fisgin, N., Ergonul, O., Doganci, L., and Tulek, N. (2009). The role of ribavirin in the therapy of Crimean-Congo hemorrhagic fever: early use is promising. *Eur. J. Clin. Microbiol. Infect. Dis.* 28 (8): 929–933.
 - **230** Jabbari, A., Besharat, S., Abbasi, A. et al. (2006). Crimean-Congo hemorrhagic fever: case series from a medical center in Golestan province, Northeast of Iran (2004). *Ind. J. Med. Sci.* 60 (8): 327–329.
 - **231** Smego, R.A. Jr., Sarwari, A.R., and Siddiqui, A.R. (2004). Crimean-Congo hemorrhagic fever: prevention and control limitations in a resource-poor country. *Clin. Infect. Dis.* 38 (12): 1731–1735.
 - **232** Ascioglu, S., Leblebicioglu, H., Vahaboglu, H., and Chan, K.A. (2011). Ribavirin for patients with Crimean-Congo haemorrhagic fever: a systematic review and meta-analysis. *J. Antimicrob. Chemother.* 66 (6): 1215–1222.
 - **233** Soares-Weiser, K., Thomas, S., Thomson, G., and Garner, P. (2010). Ribavirin for Crimean-Congo hemorrhagic fever: systematic review and meta-analysis. *BMC Infect. Dis.* 10: 207.
 - **234** Johnson, S., Henschke, N., Maayan, N. et al. (2018). Ribavirin for treating Crimean Congo haemorrhagic fever. *Cochrane Database Syst. Rev.* 6: CD012713.
 - 235 Hartman, D.W., Haddock, E., Meade-White, K. et al. (2020). Efficacy of favipiravir (T-705) against Crimean-Congo hemorrhagic fever virus infection in cynomolgus macaques. *Antiviral Res.* 181: 104858.
 - 236 Carazo Perez, S., Folkesson, E., Anglaret, X. et al. (2017). Challenges in preparing and implementing a clinical trial at field level in an Ebola emergency: a case study in Guinea, West Africa. *PLoS Negl. Trop. Dis.* 11 (6): e0005545.
 - WHO (2014). Ethical considerations for use of unregistered interventions for Ebola viral disease. https://apps.who.int/iris/bitstream/handle/10665/130997/WHO_HIS_KER_%20 GHE_14.1_eng.pdf;jsessionid=6D96515B09CCC0345BC3C66333AD3C29?sequence=1 (accessed on 1 December 2019).
 - **238** Richardson, T., Johnston, A.M., and Draper, H. (2017). A systematic review of Ebola treatment trials to assess the extent to which they adhere to ethical guidelines. *PLoS ONE* 12 (1): e0168975.

Rolando Cannalire¹, Enzo Tramontano², and Vincenzo Summa¹

¹ Department of Pharmacy, University of Napoli "Federico II", Napoli, Italy
 ² Department of Life and Environmental Sciences, University of Cagliari, Cagliari, Italy

17.1 Overview on Coronavirus (CoV)

17.1.1 CoV Epidemiology

Coronaviruses (CoVs) are pleomorphic, large, enveloped, positive-sense single-stranded RNA ((+)ssRNA) viruses belonging to the order *Nidovirales*, family *Coronaviridae*, subfamily *Coronavirinae*, classified into four genera: α -, β -, γ -, and δ -CoVs. CoVs are globally distributed in a large number of species including bats, camels, pigs, bovines, felines, poultry, mice, dogs, rabbits, whales, humans, and many others. The α - and β -CoVs infect only mammals, while γ - and δ -CoVs infect birds, even if some of them may also infect mammals [1]. The ~30 kb RNA genome of CoV contains at least six open reading frames (ORFs, see later for details), with an overall 54% identity at the whole genome level. However, each virus lineage can be distinguished by its ORF structure.

The 5' genomic region encodes the first ORFs, making about two-thirds of the whole genome length and scoding 16 nonstructural proteins (nsps) (nsp1–16), except γ -*CoVs* that lacks nsp1, and is mostly conserved with a 58% nsps identity. Differently, the 3' genomic region varies among genera, and even from strain to strain within a viral group, with a 43% identity on the structural protein-coding region [2, 3]. CoVs encode four main structural proteins: the spike surface glycoprotein (S) protein, the nucleocapsid (N) protein, the membrane (M) protein, and the envelope (E) protein. While most CoVs require all of them for a mature virion, some do not, indicating that some structural proteins might be dispensable, and their functions can be taken over by additional proteins [4]. Most often, the CoV envelope contains three to four viral proteins. The most abundant structural proteins are the S and M proteins, while the E protein constitutes a minor but critical structural component. In addition, in some CoVs, a hemagglutinin esterase is also found. The last structural protein, N, binds the viral RNA genome to form the nucleoprotein.

Viral entry is initiated by the specific engagement of CoVs S proteins with the corresponding cell host surface receptors, a key step to establish infection. In recent years, a number of

studies have proposed that changes in S proteins can foster the expansion to CoV from one species to another [5, 6] and hence the wide range of animal hosts has been proposed to be mainly due to both high mutation rates and recombination events of CoVs [7, 8]. In particular, cross-species transmission has been proposed to take place between different CoVs infecting a common host. In this respect, it is worth to note that CoVs infecting single animal species belongs to different genera. As an example, the six CoVs infecting pigs belong to α -CoVs (porcine epidemic diarrhea virus, PEDV; porcine respiratory coronavirus, PRCV; acute diarrhea syndrome virus, SADS-CoV; porcine transmissible gastroenteritis coronavirus, TGEV), β -CoVs (porcine hemagglutinating encephalomyelitis virus, PHEV), and δ -CoVs (porcine δ -CoV, PDCoV). A similar picture is also present in circulating human CoVs that can also be divided into low and high pathogenic viruses. Among the former are the α -CoVs HCoV-229E, HCoV-NL63, and the β -CoVs HCoV-OC43 and HCoV-HKU1 that generally cause mild upper respiratory illness and collectively are associated with 10-30% of common cold cases [9]. Among the latter are the highly similar Severe Acute Respiratory Syndrome CoV (SARS-CoV-1) and SARS-CoV-2, and the Middle East respiratory syndrome CoV (MERS-CoV) that caused three outbreaks in 2002 (SARS-CoV-1), in 2011 (MERS-CoV), and in 2019 (SARS-CoV-2). While the 2002 epidemic has been declared to be over in 2003 [10], the 2011 is still ongoing [11] and has involved >25 countries so far, and the 2019 is currently determining the worse pandemic event of this century. As in the case of other CoVs and also many human viruses causing burdening diseases, existing sometimes since centuries, such as measles virus, influenza virus, smallpox virus, or more recently dengue virus, Ebola virus, HIV, Chikungunya virus, Zika virus, and others, also these highly pathogenic CoVs originated by animal-to-human host-switching. In particular, it is likely that SARS-CoV-1 originated from bats through sequential recombination of bat CoV (SARS-CoVs) and that masked palm civets (Paguma larvata) were intermediate hosts [2]. MERS-CoV is also believed to have originated in bats, and dromedary camels were suggested to play roles as intermediate hosts [11]. In the case of SARS-CoV-2, also bats have been suggested as likely reservoir hosts [12] and pangolins have been suggested as possible hosts in the emergence of the SARS-CoV-2 [13], even if the scientific debate seems to be not concluded yet.

17.1.2 HCoV Replication

In the present book chapter, we present the CoV replication cycle taking into account mostly the etiological agent of the current pandemic, SARS-CoV-2. The considered steps are mostly shared among the three human highly pathogenic viruses that, otherwise, differ for their accessory proteins that probably modulate virus–host interactions processes such as in cell proliferation, innate immune evasion, programmed cell death, and others. While these viral functions could be indeed exploited as drug targets, they are not the specific focus of the present chapter.

SARS-CoV-2 has a single-stranded, 5'-capped, (+)ssRNA genome, and has been annotated to contain 14 ORFs, coding for 27 proteins (Figure 17.1a) [14]. The first two ORFs (ORF1a and -b) encompassing roughly the 5'-terminal two-thirds of the viral genome, encode for the 16 viral nsps. In particular, the ORF1ab encodes the pp1ab polyprotein that contains 15 nsps (nsp1–nsp10 and nsp12–nsp16), while the ORF1a encodes the pp1a polyprotein that contains 10 nsps (nsp1–nsp10). The ORFs at the 3'-terminus of the genome encode for four



Figure 17.1 (a) Genomic structure of SARS-CoV-2 and comparison with respect to SARS-CoV-1 and MERS-CoV. ORFs encoding for polyproteins (pp) including the various nonstructural (ns) elements are represented in blue, while genes encoding for accessory proteins are colored in gray. For SARS-CoV-2, the various structural proteins are represented in a schematic virion structure as well. Proteins that differ from SARS-CoV-2 in SARS-CoV-1 and MERS-CoV are indicated in red. Nucleotide positions are referred to the representative genomic sequences for SARS-CoV-2 (NC_045512), SARS-CoV-1 (NC_004718), and MERS-CoV (NC_019843). (b) Schematic representation of CoV life cycle.

structural proteins, namely the S surface glycoprotein, the small E protein, the M protein, and the N protein, and eight accessory proteins (3a, 3b, p6, 7a, 7b, 8b, 9b, and orf14).

The SARS-CoV-2 replication cycle starts with the interaction of the viral S protein with the human angiotensin-converting enzyme 2 (ACE2), found in lung, heart, kidneys, and

intestine cell membranes (Figure 17.1b) [15]. The S protein is a large glycosylated transmembrane protein forming homotrimers and, in addition to playing a crucial role in mediating virus attachment to the host cell receptor, is also the main antigen exposed at the surface of the virion, acting as a major inducer for the host immune response. The S protein, mostly extending outside the virion, belongs to the class I viral fusion proteins and it can be divided into a S1 subunit, critical for receptor recognition, and the S2 subunit, important for membrane fusion. In fact, following ACE-2 binding, S is cleaved into the two S1 and S2 subunits by a cellular protease triggering an efficient entry into the cytoplasm by endocytosis. In the acidic endosome, S is further cleaved by lysosomal proteases for exposing the fusion peptide (FP), leading to fusion of the virus envelope with the endosome membrane, and ending in the viral RNA release into the cytosol. Different from SARS-CoV-1 and SARS-CoV-2, the receptor for MERS is the dipeptidyl peptidase 4 (DPP4) also known as CD26, a multifunctional type-II transmembrane glycoprotein endowed with exopeptidase activity that preferentially cleaves dipeptides from hormones and chemokines at a site following a proline amino acid residue, important for controlling their bioactivity [11]. Of note, it has been shown that CoVs can enter cells also by membrane fusion [16], and which of the two pathways (endocytosis or membrane fusion) is most relevant needs to be clarified yet. Once entered into the cytoplasm, the 5'-capped, (+)ssRNA genome is recognized by the cellular translation machinery leading to the first viral polyproteins production that are subsequently processed into individual proteins (Figure 17.1b). Of note, the shift between the shorter ORF1a and the longer ORF1b is due to a -1 ribosomal frameshift in the overlapping region between ORF1a and -1b just upstream the stop codon, hence enabling the production of the larger pp1ab polyprotein. The frameshifting process occurs with a ~20-50% efficiency and is triggered by a slippery sequence, UUUAAAC, followed by an RNA pseudoknot structure [17]. The pp1a and pp1ab polyproteins are cleaved into single proteins by two viral proteases, a papain-like protein (PL^{pro}) encoded by nsp3, and a chymotrypsin-like cysteine protease (3CL^{pro}) encoded by nsp5 [18]. While 3CL^{pro} catalyzes the proteolytic cleavage of all nsps downstream of nsp4, and is thus referred to as the main viral protease, PL^{pro}, in addition to being responsible for the few other cleavage events, has also additional roles in the evasion of the immune system. Liberation of the single nsps allows formation of the replication complex, whose main element is the RNA-dependent RNA polymerase (RdRp). It is encoded by the nsp12 gene, and copies the (+)ssRNA into (-)ssRNA that, in turn, serves as template for more (+)ssRNA genomic copies and subgenomic mRNAs. The replication complex is also formed by processivity factors (nsp7-8), a helicase (nsp13), single-strand binding proteins (nsp9), a proofreading exonuclease (nsp14), other cofactors (e.g. nsp10), and capping enzymes (e.g. nsp16). Such multifactorial replication complex is not common among (+)ssRNA viruses and has also the peculiarity of a proofreading system leading to high-fidelity viral replication. In fact, due to the big size of SARS-CoV-2, avoiding the accumulation of mutations is essential for the viral population, since it has been shown that 15 kb is the living threshold for viral genomes without proofreading and that bigger genomes require proofreading to avoid a high mutational burden leading the viral population into catastrophic events [19]. Indeed, the similar SARS-CoV-1 mutation rates (10^{-6}) are some order of magnitude lower than the ones of most RNA viruses [20]. The SARS-CoV-2 replication, as the one of other CoVs and (+)RNA viruses, has been proposed to take place in double membrane vesicles to which the replication complex is anchored and that spatially

divide the site of viral RNA replication from intrinsic immune sensors and from downstream virion assembly taking place in the endoplasmic reticulum–Golgi intermediate compartment (ERGIC). Of note, nsps with integral transmembrane domains such as nsp3, nsp4, and nsp6 are involved in membrane manipulation allowing vesicle formation, probably by introducing curvatures in the ER membranes [21]. While the copies of the (+)ssRNA genome are produced by the replication complex, the structural glycoproteins (M, E, and S) incorporated into the envelope are translated into the ER and accumulate in a ERGIC budding site different from the one producing viral genomes. The M protein is important for incorporating essential viral components into the new virions during morphogenesis. The E protein participates in viral maturation and forms ion channels in the viral membrane whose role is still unclear, while the S protein is decorating the external portion of the virion in view of the next viral replication cycle. The last structural protein, the N protein, has the role of associating the viral genome and M and hence directs genome packaging into new viral particles. Finally, progeny virions in vesicles are transported to the cell surface and released to the outside environments through the exocytic pathway or cell lysis.

17.1.3 Human Corona Virus (HCoV) Diseases Natural History

HCoVs are known to cause a large proportion of minor upper respiratory tract infections; however, SARS-CoV-1, MERS, and SARS-CoV-2 cause severe diseases that can have a fatality of up to 10% for SARS-CoV-1 and 36% for MERS [10, 22, 23]. Both SARS-CoV-1 and MERS had a limited number of cases as compared with SARS-CoV-2, and so their pathogenesis is poorly understood. Both viruses were reported to have around 5 days of incubation and then 95% of patients developed symptoms within 13 days of exposure. Common early symptoms were fever, coughing, myalgia, and headache, then lung disease takes place leading to abnormal chest X rays in 60–100% SARS-CoV-1 and 90–100% MERS patients. Immunopathogenic events seem to contribute to disease progression into acute respiratory distress syndrome (ARDS) that is associated with high upregulation of pro-inflammatory cytokines production.

The current pandemic events have put in place enormous efforts in understanding the SARS-CoV-2-induced infection events, and the close examination of the COVID-19 disease progression in patients has identified three phases of the disease (Table 17.1). There is general agreement to consider an individualized treatment approach for patients based on their symptoms corresponding to these stages of infection. The three phases have a different type of biological interaction with the virus.

The first phase (phase 1) starts through the contact with the infected droplet from individual to individual. SARS-CoV-2 binds to epithelial cells presenting the ACE2 receptor and begins to replicate (Table 17.1). The virus diffuses locally activating a well-coordinated innate immune response that is the first-line physiological response to infection. This phase can cause an asymptomatic stage, in which the infected individual is positive to RT-PCR tests, can spread the virus but does not clinically manifest disease symptoms, or it can cause mild symptoms that are generally confused with a common cold or flu such as cough, fatigue, headache, sore throat, runny nose, diarrhea, and others.

The second phase (phase 2) occurs during a persistent replication of the virus in the upper and lower airway and at the pulmonary level. In this phase, the immune system can be severely affected by the infection leading to primarily respiratory symptoms such as

Phase	Virus location	Host response	Symptoms
1	Early infection in epithelial cells of upper airway	Innate immunity	Asymptomatic or mild symptoms similar to common cold or flu (cough, fatigue, headache, sore throat, runny nose, diarrhea, and others)
2	Persistent replication at the pulmonary level	Immune system severely affected	Respiratory symptoms (shortness of breath, persistent cough, up to low oxygen levels) and formation of blood clots
3	Gas exchange units of the lungs	Hyperactivation of the immune system with hyperinflammation	Serious respiratory symptoms, alveolar damage, multiorgan injury

Table 17.1	Phases	of COVID-19	disease.
10010 1/11	i nases	01 00 110 17	ansease.

shortness of breath, persistent cough, up to low oxygen levels. Another severe complication in this phase can be the formation of blood clots.

The third phase (phase 3) corresponds to hyperactivation of the immune system and hyperinflammatory progression. The virus reaches the gas exchange units of the lungs leading to diffuse alveolar damage and at the same time induces dysregulation of immune response. The hyperinflammatory profile consistent with a cytokine storm, in which the body attaches its own tissue, is robustly associated with COVID-19 severity and is also accompanied by injuries to other organs. In fact, COVID-19 infection has been reported to impact on the cardiovascular system, the renal system, the hepatic and gastrointestinal system, and to lead to neurological complications. A number of COVID-19 sequelae have been also reported and are currently under scrutiny.

The three phases clearly overlap and, to have an effective therapy, it is very important to recognize each stage and tailor the general treatment toward a specifically personalized treatment.

Looking at the physiopathology of the disease, the antiviral drugs should be more efficacious in the phase 1 and 2 of the disease. This also opens up the possibility to use potent selective and safe antiviral for prophylaxis when an individual gets in contact with a positive patient and still does not know if it has been infected. The scenario clearly indicates that there is space for antiviral drugs for prevention and treatment in the presence of vaccines and therapeutic antibodies that will be developed.

In the following sections, the antiviral approaches exploited to fight CoV infection, mainly SARS-CoVs and to a lesser extent MERS-CoV, are described focusing on licensed and investigational drugs. Moreover, the most promising and characterized viral targets, such as the 3CL^{pro}, the nsp12 RdRp, and the S fusion protein, and the S/ACE2 interaction as the viral target to inhibit entry, and the inhibitors, reported so far, are analyzed.

17.2 Licensed and Clinical Investigational Drugs Against CoVs

The 3CL^{pro} and the nsp12 RdRp are the best characterized SARS-CoV targets and show the highest degree of conservation across CoVs, fostering the identification of broad-spectrum inhibitors. Indeed, the most advanced antiviral agents approved for COVID-19 or in clinical development are inhibitors of RdRp and of 3CL^{pro}.

17.2.1 Licensed Drugs Against CoVs

The first and currently only approved drug against a CoV infection and licensed for COVID-19 treatment is Remdesivir (GS-5734), developed by Gilead Sciences (Figure 17.2). Remdesivir received FDA and EUA [28] approval following the encouraging results from the National Institute of Allergy and Diseases (NIAID), Gilead clinical trials, and from the compassionate use programs [29–31]. The fast approval was the consequence of a drug repurposing approach applied for the SARS-CoV-2 pandemia. Remdesivir is the Sp isomer ester monophosphoramidate prodrug of 1'-cyano-substituted adenine C-ribonucleoside analogue known as GS-441524 [32–34]. Originally identified within an antiviral research program against HCV and respiratory syncytial virus (RSV), Remdesivir showed a broad antiviral activity against other RNA viruses, including Ebola virus and SARS-CoV [32, 35, 36]. Although Remdesivir performed well in preclinical studies, it did not meet efficacy endpoints in a randomized trial conducted during an Ebola outbreak [37].

Remdesivir has shown [38, 39] potent antiviral activity against SARS-CoV-1 and MERS-CoV ($EC_{50} = 0.069$ and 0.074μ M, respectively) in primary human airway epithelial cell (HAE) cultures [38]. More recently, Remdesivir also showed antiviral activity against SARS-CoV-2 infections, with a potency varying from the nM to the μ M range depending on cell lines and evaluation method (Figure 17.2) [24–27].

Remdesivir treatment significantly ameliorated loss of pulmonary function in female C57Bl/6 carboxyl esterase 1c deficient (Ces1c-/-) mice administered (25 mg/kg sc) 1 day postinfection by SARS-CoV-2 and BID thereafter. The transgenic mouse model better recapitulates the DMPK profile of Remdesivir in humans, that lack Ces1c, a mouse serum esterase dramatically reducing the half-life of Remdesivir [38]. Remdesivir in vivo efficacy against SARS-CoV-2 was also observed in rhesus macaques (12 h p.i with 10 mg/kg and once-daily for 6 days with 5 mg/kg) [40] reducing virus titers, clinical disease, and damage to the lungs. The treatment resulted in reduced virus replication in the lower respiratory tract, but not in the upper tract, suggesting the need to explore drug delivery strategies to improve Remdesivir distribution.

Regarding the use of Remdesivir in humans, the optimal duration of treatment is still under evaluation in clinical trials. Both 5-day and 10-day treatment durations are suggested, based on the severity of the disease, with 200 mg iv on the first day, followed by 100 mg on each consecutive day. The approval was granted on the basis of the available data from the Gilead's SIMPLE trials (NCT04292899), the Adaptive COVID-19 Treatment Trial (ACTT), sponsored by NIAID (NCT04280705), in patients with severe manifestations of COVID-19, and from the compassionate use program.



Figure 17.2 Remdesivir and its antiviral activity against SARS-CoVs and MERS-CoV. *Source*: ^aFrom Ref. [24], ^bfrom Ref. [25], ^cfrom Ref. [26], and ^dfrom Ref. [27].

The two SIMPLE trials evaluated the safety and efficacy of 5-day and 10-day duration of Remdesivir in hospitalized patients with severe (first study) and moderate (second study) COVID-19, showing similar results for both regimes [29]. Thus, priority should be given to a five-day regimen for patients at the early stages of severe disease [30]. During the trial, the side effects were mild and, however, the absence of a placebo control group and the open-label design of this study did not permit an overall assessment of the benefit of Remdesivir. Another limitation is the lack of SARS-CoV-2 viral-load data. The recently published results of the second SIMPLE trial study showed that among patients with moderate COVID-19, those randomized to a five-day course of Remdesivir plus standard of care had higher odds of having an improvement in clinical status compared with those randomized to standard of care alone [31].

In contrast to Gilead's trials, interim results from another large trial, the WHO's SOLIDARITY, seem to indicate that none of the four repurposed antiviral drugs tested including Remdesivir have a significant effect in reducing COVID-19 mortality, the need for ventilators, or the duration of hospitalization [41]. However, this was an open-label trial as well.

The ACTT trial is a randomized, double-blind, placebo-controlled multicenter trial conducted in around 100 sites globally, evaluating the time to recovery of hospitalized adults diagnosed with COVID-19 up to day 29. Remdesivir was superior to placebo in shortening the time to recovery and lowering mortality even though the difference was not significant [31].

Further data will be needed, including the full statistical analysis of the entire population involved in the ACTT trial, in order to fully understand the optimal use of Remdesivir [42]. Moreover, there is room for further studies involving potential combination therapies with other antivirals and anti-inflammatory agents in appropriate regimens. In this regard, the ACTT3 trial (NCT04492475) will evaluate the combination of IFN- β 1a and Remdesivir compared with Remdesivir alone.

Significant efforts have been made to find an alternative route of administration. Clinical trials (GS-US-553-9018 and NCT04539262) have been initiated to evaluate an inhaled nebulized formulation of Remdesivir, which can be particularly suited for early-stage COVID-19 patients since the upper respiratory tract is the initial site of SARS-CoV-2 infection and spread.

The interaction of Remdesivir with the SARS-CoV-2 RdRp has been structurally characterized and the details are described in Section 17.3.1.

17.2.2 Clinical Investigational Drugs Against CoVs

Along investigational drugs targeting the CoV RdRp, the isobutyrate ester derivative Molnupiravir (EIDD-2801), the orally bioavailable prodrug of the ribonucleoside β -D- N^4 -hydroxycytidine analog **1** (NHC, EIDD-1931) (Figure 17.3), represents a promising candidate which has advanced into clinical trials.

Compound **1** is a broad-spectrum antiviral agent with in vitro activity against multiple unrelated viruses [43–47] (including CoVs). In particular, **1** was able to inhibit SARS-CoV-2 ($EC_{50} = 0.3$ and $0.08 \,\mu$ M in Vero and Calu3 cells, respectively), SARS-CoV-1 ($EC_{50} = 0.14 \,\mu$ M in HAE cells), and MERS-CoV (average $EC_{50} = 0.15 \,\mu$ M in Calu3 cells) with low toxicity (CC_{50} S>10 μ M), showing also remarkable potency against a model CoV mouse hepatitis virus (MHV), bearing resistance mutations to Remdesivir [43]. DMPK profiling for **1** in rodents demonstrated dose-dependent oral bioavailability (56, 43, and 36%, $T_{1/2}$ of 5.2, 3.2, and 2.7 hours following oral administration of 50, 150, and 500 mg/kg, respectively).



Figure 17.3 Chemical structures of the prodrug Molnupiravir and parent compound **1** with its biological activity.

Conversely, its oral bioavailability in cynomolgus macaques was limited, thus prompting the development of the 5'-isobutyrate ester prodrug Molnupiravir, which demonstrated good oral bioavailability in nonhuman primates and quick release of free parent drug 1 [45]. In prophylactic in vivo efficacy studies, Molnupiravir has shown a significant reduction of the virus titer and ameliorated the pulmonary function of mice infected with SARS- and MERS-CoVs (500 mg/kg p.o. administration 2 hours before infection and every 12 hours) [43]. The decrease of MERS-CoV yields both in vitro and in vivo were correlated with an increase of the frequency of mutation in the viral RNA, highlighting a putative mechanism of lethal mutagenesis. The interesting preclinical results prompted the FDA and the UK Medicines and Healthcare Products Regulatory Agency to allow for the human safety testing of Molnupiravir. Phase I trial recently has been successfully completed (NCT04392219) and currently un number of additional Phase II/III studies are ongoing. In particular, two Phase IIa randomized, double-blind, placebo-controlled trials were activated to evaluate the antiviral efficacy of Molnupiravir in adults diagnosed with COVID-19 (NCT04405570, NCT04405739). In the late October 2020, two Phase II/III studies have started to evaluate the efficacy, safety, and PK of in non-hospitalized (NCT04575597) and hospitalized (NCT04575584) adult participants with COVID-19, and results are expected for the end of 2021. To this aim, the Drug Innovation Ventures at Emory (DRIVE) LLC, which developed the drug, partnered with the biotech company Ridgeback Biotherapeutics LP and with Merck, to advance Molnupiravir through clinical development and to optimize the drug's availability during the pandemic.

Galidesivir (BCX4430) is a *C*-adenosine analogue acting as a non-obligate RNA chain terminator, initially investigated as antifilovirus agent by BioCryst Pharmaceuticals (Figure 17.4) [48]. Galidesivir was shown to possess weak antiviral activities against several RNA viruses, including SARS-CoV-1 (EC₅₀ = 57.7 μ M, measured by neutral red uptake assay) and MERS-CoV (EC₅₀ = 68.4 μ M assessed by high-content image analysis) [48], whereas it does not



Figure 17.4 Structure and biological activity of Galidesivir.

inhibit SARS-CoV-2 up to 100 μ M [25]. Notably, Galidesivir can be administered by both parenteral (im and ip) and oral routes [49]. In mouse, rat (both treated with 2 mg/kg), guinea pig (50 mg/kg), and cynomolgus macaque (20 mg/kg), Galidesivir DMPK is characterized by rapid clearance from the plasma with a $T_{1/2} < 5$ min. Conversely, the $T_{1/2}$ of the active triphosphate form in the liver in rats (administered 30 mg/kg) is substantially longer at 6.2 hours [48]. Despite the apparently weak anti-CoV activity, in April 2020, the company BioCryst Pharmaceuticals, Inc. started a randomized, double-blind, placebo-controlled Phase 1 trial to evaluate the PK, safety, and antiviral effects of Galidesivir (IV administered) vs. placebo in hospitalized adult subjects with either Yellow Fever or COVID-19 in Brazil (NCT03891420).

Other classes of antivirals that exhibit single agent efficacy or that are complementary to RdRp inhibitors for use in combination regimens are essential to meet the unmet medical need of a cure against for COVID-19 and potentially emerging CoVs. PF-07304814 is the phosphate ester prodrug of PF-00835231, a peptidomimetic reversible covalent inhibitor of CoVs 3CL^{pro}, which entered into a Phase Ib clinical trial. Following the SARS outbreak in 2002-2003, researchers at Pfizer identified SARS-CoV-1 3CL^{pro} covalent reversible/irreversible inhibitors [50, 51] having hydroxymethylketone-based warheads, but the project was discontinued thanks to the control of the first SARS pandemic. Following the pandemic SARS-CoV-2, PF-00835231 was rescued from the shelf and has been identified as an SARS-CoV-2 3CL^{pro} inhibitor exerting very potent activity against a wide panel of 3CL^{pro} from α -, β -, and γ -CoVs, with K_i/IC_{50} ranging from the low-nM to pM range ($K_i = 0.27$ nM and $IC_{50} = 6.9 nM$ against SARS-CoV-2 $3CL^{pro}$; $K_i = 4 nM$ against SARS-CoV-1 $3CL^{pro}$), whereas inactive on HIV, HCV, and a panel of human proteases (Figure 17.5), proving its selectivity for CoVs [52, 53]. The efficacy of PF-00835231 against SARS-CoV-1 and SARS-CoV-2 was evaluated in Vero cells, showing EC50 values of 4.8 and 39.7 µM, respectively. However, PF-00835231 has been shown to be a substrate of P-gp highly expressed in Vero cells. In fact, in the presence of a P-gp inhibitor, the antiviral activity of PF-00835231 was significantly increased, with EC50 values of 0.23 and 0.76 µM against SARS-CoV-1 and SARS-CoV-2,



PF-07304814 R = PO₃H₂

Figure 17.5 SARS-CoVs 3CLpro inhibitors: PF-00835231 with biological activity and its phosphate iv prodrug PF-07304814; oral candidate PF-07321332; ${}^{a}EC_{50}$ value, in brackets data in the presence of P-gp inhibitor.

coupled to low toxicity (CC_{50} > 100 µM). Interestingly, PF-00835231 exhibits additive/synergistic effect in combination with Remdesivir against SARS-CoV-2 in cell-based assays [53]. Compound PF-00835231 has been co-crystallized with both 3CL pro of SARS CoV-1 and CoV-2 (PDB Codes 6XHL and 6XHM) [52], showing – as expected – almost identical binding modes. Structural details of 3CL^{pro}/PF-00835231 interaction are discussed in Section 17.3.2 (Figure 17.14a). PF-00835231 has been extensively characterized for its PK properties following iv administration to rats (2mg/kg), dogs, and monkeys (1mg/kg) resulting in moderate plasma clearance (rat, 27, dog, 18, and monkey, 29 ml/min/kg), short half-life (rat, 0.7, dog, 1.5 and monkey, 1.2 hours), low volumes of distribution (rat, 0.75, dog, 1.1, and monkey, 1.41/ kg), and good AUC (rat, 1250, dog, 932, and monkey, 583 ng.h/ml) [53]. Unfortunately, PF-00835231 shows very low oral bioavailability in rats and monkeys (<2%). Further preclinical characterization demonstrated that PF-00835231 did not inhibit CYP450 enzymes and membrane transporters, thus indicating a low risk to cause drug-drug interactions. A prodrug strategy has been applied to PF-00835231 in order to improve the aqueous solubility and its phosphate ester PF-07304814 has been reported [53]. After iv administration, PF-07304814 rapidly released 68, 81, and 76% active form PF-00835231 in rats, dogs, and monkeys, respectively. The safety profiles of PF-07304814 and PF-00835231 were individually evaluated in vitro and in vivo. In vitro, the compounds were negative in the AMES test, did not induce micronuclei formation, did not inhibit hERG, and had no effect in human blood hemocompatibility tests. Following continuous iv infusion (24 hours) of PF-07304814 in rats, no toxic signs were observed until 1000 mg/kg. On the basis of these promising preclinical results, on 15 September 2020, Pfizer announced the initiation of a double-blind, placebo-controlled Phase Ib clinical trial (NCT04535167) to evaluate the safety, tolerability, and pharmacokinetics of PF-07304814 [54]. Moreover, Pfizer presented its oral COVID-19 antiviral clinical candidate PF-07321332 (Figure 17.5) on April 6th at ACS Spring 2021 Meeting. The compound, which is currently in Phase 1 clinical trials (NCT04756531) in healthy patients, is the first orally administered inhibitor of SARS-CoV-2 3CLpro in the clinic. The compound is structurally related to PF-00835231, which was modified on order to retain the excellent enzymatic inhibition and antiviral activity and gaining oral bioavailability. Importantly, the P2 residue was modified replacing the leucine with a less polar bicycloproline moiety. The indole was substituted with a tertbutylglycine with a trifluoroacetyl capping group and a nitrile was inserted as warhead instead of the hydroxyketone moiety. Unfortunately, data on biological activity and on PK properties of PF-07321332 have not been published yet.

17.3 Medicinal Chemistry Approaches Toward the Identification of New Drugs

Different strategies targeting both viral and host factors essential for the CoVs replication have been pursued [55, 56]. The most promising are described in the following paragraphs. The 3CL^{pro} and the nsp12 RdRp are the best characterized SARS-CoV targets and show the highest degree of conservation across CoVs, fostering the identification of broad-spectrum inhibitors [57, 58]. Besides targeting intracellular steps of viral replication, virus entry into host cells represents another way to prevent/treat viral infections. The main target for this approach are the CoV S fusion protein and the S/ACE2 interaction as viral factors to inhibit entry [16, 59].

To date, the most promising and investigated approach to fight SARS-CoVs infection relies on targeting these three proteins. The analysis of the targets and the inhibitors discovered until now are reported below.

17.3.1 SARS-CoVs RdRp: Structure, Function, and Inhibition

The RdRp encoded by the nsp12 gene catalyzes the formation of a phosphodiester bond between NTPs in a primer-dependent manner and requires the formation of a complex with nsp7 and nsp8 cofactors to acquire processivity [60]. In addition, nsp8 exerts RNA primase activity enabling *de novo* RNA synthesis [61]. The structural and functional features of the RdRps among the CoV family are highly conserved with the RdRps of SARS-CoVs sharing 96% amino acid identity [43, 62].

Structures of SARS-CoV-2 full-length RdRp, in complex with nsp7 and nsp8, were determined by cryo-EM (PDB ID: 6M71 [63], 7BV1, [64] 7BW4 [65]) showing the typical righthand architecture (Figure 17.6), constituted by the finger, palm, and thumb subdomains [66]. The nsp12 RdRp domain is connected through an interface to the N-terminal region possessing nucleotidyltransferase activity (NiRAN), whose exact role in the viral life cycle remains elusive [67]. SARS-CoV-2 RdRp cryo-EM structure revealed also an additional, unique N-terminal β -hairpin, which was previously not observed in SARS-CoV-1 RdRp [63].



Figure 17.6 Architecture of SARS-CoV-2 nsp12. (a) Schematic diagram outlining the domain organization of SARS-CoV-2 nsp12. (b) Structure of SARS-CoV-2 PDB ID: 6M71. Left: Ribbon diagram of nsp12 showing palm, fingers, and thumb domains, colored as in (a). Nsp7 and nsp8 cofactors are shown as pale green and raspberry ribbons. Right: overview of nsp12 active site, shown as white ribbons, whereas the conserved motifs A–G are colored as in (a). The catalytic Asp760 and Asp761 are shown as white sticks. The template entry, NTP entry, product hybrid exits paths are indicated by orange arrows.

The RdRp active site is composed by seven conserved motifs (A–G) lining a central cavity where the template-directed RNA synthesis takes place (Figure 17.6). In particular, the incoming NTP binds within motif F, whereas the RNA template enters the active site through a channel formed by motifs F and G; motif E and the thumb subdomain sustain the primer strand. The product of RNA synthesis leaves the active site through an RNA exit path, situated at the front side of the polymerase [63].

Moreover, the replicating nsp12–nsp7–nsp8 complex has been solved bound to RNA template-product duplex (PDB ID: 6YYT; Figure 17.7) [68]. The active site cleft of nsp12 binds the first turn of RNA, whereas two nsp8 subunits bind to opposite sides of the cleft, flanking the exiting RNA duplex with long α -helical extensions, called "sliding poles." These nsp8 extensions are rich in positively charged residues and form multiple RNA backbone interactions. The mutation of one of these positively charged residues, namely Arg58, with Ala, was previously reported as lethal in SARS-CoV-1, because of the strong decrease of polymerase activity, leading to a nonviable phenotype [60].

Despite the large amount of structural data produced for the SARS-CoVs RdRp, there was no clear identification of putative allosteric binding pockets for allosteric inhibitors. Furthermore, there are no reports of medicinal chemistry programs aimed at the identification of specifically designed inhibitors of CoVs RdRp, both active site or allosteric inhibitors. Indeed, the only inhibitors identified and characterized derive from repurposing of known broad-spectrum nucleoside inhibitors (NIs) acting as non-obligate chain terminators or mutagens [69]. The first possess a natural base and a 3'-hydroxyl on the sugar and an additional substituent at the C-1' or the C-2' positions of the ribose ring blocking the formation of the phosphodiester linkage with the incoming NTP [69]. The mechanism of lethal mutagenesis involves the inability to recognize the nucleoside analogues as regular



Figure 17.7 Structure of SARS-CoV-2 replicating RdRp–RNA complex (PDB ID: 6YYT). Nsp12 is shown as a molecular surface (color code as in Figure 17.6), the cofactors nsp7 and nsps8 (1 and 2) are shown as pale green and raspberry ribbons. RNA turns are shown as orange ribbons. The positively charged nsp8 residues, proposed to interact with RNA, are shown as sticks.



Figure 17.8 Binding mode of Remdesivir into the SARS-CoV-2 nsp12 active site (PDB ID: 7BV2). Left: nsp12 is shown as a molecular surface, colored according to the schematic diagram in Figure 17.6. For clarity, nsp7 and nsp8 cofactor have been removed. The template and primer RNA are shown as ribbon models and labeled. Right: zoom-in of the nsp12 active site. The covalently bound monophosphate form of Remdesivir and the pyrophosphate group are shown as sticks. Mg²⁺ ions are shown as green spheres. The RNA bases interacting with Remdesivir are shown as orange thin sticks, while protein residues are shown as white thick sticks. Hydrogen bonds are shown as black dashed lines.

nucleobases, thus inducing a mismatch in base pairing and an increase in mutations, ultimately leading to nonviable genomes.

Remdesivir has been deeply characterized for its mode of action and interaction with SARS-CoV-2 RdRp. The cryo-EM structure of Remdesivir monophosphate in complex with SARS-CoV-2 RdRp and a 50-base template-primer RNA (PDB ID: 7BV2) [64] shows the covalent bond to the primer strand at the 3'-end (Figure 17.8) to terminate chain elongation.

Remdesivir is positioned at the center of the catalytic site, with the adenine stacking with the upstream bases U-1 and A-1 of the primer and template strands. In addition, Remdesivir engages three strong H-bonds with the uridine bases U-1 and U+1, whereas the sugar 2'-OH group forms a further H-bond with Asn691. A "delayed" chain termination has been reported for Remdesivir, according to which the molecule blocks CoVs RdRps after the addition of three more nucleotides [70–73]. This hypothesis is consistent with a recently released cryo-EM structure of the SARS-CoV-2 RdRp in complex with RNA, before and after RNA translocation (PDB IDs: 7C2K and 7BZF, respectively) [74]. In the pre-translocated catalytic complex structure, the incorporated Remdesivir has been translocated to the -1 position, whereas the 3'-guanosine occupies the +1 position (Figure 17.8). Remdesivir engages four H-bonds, with the upstream G-2 base, the uridine base U-1, and with Ser759. The primer strand with the incorporated Remdesivir may translocate without obstruction to positions i+1, i+2, or i+3, allowing the incorporation of three subsequent nucleotides. However, at position i+4, a putative steric clash was postulated between the 1'-CN substituent of Remdesivir and Ser861 along the RNA exit tunnel [72]. Ser861 is highly conserved among CoV RdRps and its important role in Remdesivir-induced RdRp inhibition

was ultimately supported by mutagenesis studies, revealing that the Ser861Ala RdRp mutant yields a smaller fraction of i+3 termination compared with the wild-type RdRp, possibly supporting the steric clash hypothesis [74].

Two amino acid substitutions were found in the nsp12 polymerase providing low-level resistance to Remdesivir, corresponding to the SARS-CoV residues Phe480Leu and Val557Leu. Such residues are identical across CoVs, and have been found to cause an impairing of fitness and virulence [75].

Non-nucleoside inhibitors (NNIs) have not yet been reported for CoVs RdRp. The NNIs are characterized by drug-like heterocyclic scaffolds and thus offer the possible advantage to handle molecules with higher optimization potential, despite generally showing lower resistance barrier with respect to NIs. However, no allosteric sites have been mapped on the RdRp protein surface for CoVs. Since viral RdRp shows the typical 3D right hand shape, organized in palm, thumb, and fingers subdomains, the knowledge acquired on other RNA virus polymerase may allow similar considerations also for CoVs RdRp. In this regard, the best characterized viral RdRp is HCV NS5B that, despite working in the absence of a primer in a cellular environment, shows four distinct binding sites for different classes of NNIs that are able to allosterically inhibit the enzyme [76]. A very large number of inhibitors, specifically binding one of these sites, have been reported, with some of them reaching late stages of clinical trials and also approval (for instance, Dasabuvir).

The integrated computational and experimental analysis of the nsp12 surface may point toward the discovery of potential allosteric sites. Moreover, the disruption of the polymerase complex integrity, for example, targeting the interaction between nsp8 and nsp12 would be an interesting avenue to be explored. Unfortunately, also for this approach, there are no molecules reported acting with this mechanism.

17.3.2 The CoVs 3CL^{pro} (or Main Protease): Structure, Function, and Inhibitors

The 3CL^{pro}, known also as main protease or M^{pro}, is a 33.8 kDa cysteine protease able to process the polyprotein at no less than 11 conserved sites (starting with the autolytic cleavage from pp1a and pp1ab) for the release of most of the nsps CoV functional proteins. Due to its key role in viral replication and the lack of human counterparts, over the years, the discovery efforts have focused on 3CL^{pro} to identify antiviral agents against human CoVs. Indeed, a large number of crystal structures of 3CL^{pro} from CoVs have been solved either in their apo forms or in complex with inhibitors. Analysis of crystal structures highlight as SARS-CoV-1 3CL^{pro} (PDB ID: 2BX4) [77] and SARS-CoV-2 3CL^{pro} (PDB ID: 6Y2E) [78] shows almost identical tridimensional spatial organization, and binding pockets, consistent with the 96% sequence identity between the two proteins. The active form of 3CL^{pro} is a dimer, with each protomer formed by three domains [77, 78]: six antiparallel β -barrels form domains I and II (residues 10-99 and 100-182, respectively) and allocate the substrate-binding site, while domain III (residues 198-303) is a globular cluster of five helices that regulates protein dimerization. Arg4 and Glu290 from each protomer establish an ionic bond driving the formation of the dimer, which has a contact interface between the perpendicularly oriented domain II of protomer A and domain III of protomer B (Figure 17.9). The N-terminal tail, called the "N-finger," of molecule B is squeezed in between domains II and III of the parent monomer and domain II of the other one. This



Figure 17.9 (a) The X-ray structure of SARS-CoV-2 3CL^{pro} (PDB ID: 6Y2G), selected as a representative example, is shown as ribbon model. For clarity, the bound inhibitor has been removed. Protomers A (light blue) and B (light orange) associate into a dimer stabilized by a salt bridge between Glu290 and Arg4, while the substrate binding site resides at the interface of domains I and II. The catalytic Cys145 and His41 are highlighted. (b) Surface representation of the active site pocket of SARS-CoV-1 3CL^{pro} bound to a peptide aldehyde inhibitor (dark salmon sticks, PDB ID: 3SNE), chosen as a representative substrate-like inhibitor. The S1–S4 and S1'subsites are indicated with red lines and labeled. The key residues forming the active site pocket are displayed as white sticks; the catalytic Cys145 and His41 are labeled.

peculiar arrangement is stabilized by some key H-bonds, particularly those formed by Ser1 and Glu166 from each unit. The substrate binding site contains the catalytic Cys145-His41 flanked by the subpockets S4, S3, S2, S1, and S1' for the binding of the substrate P4, P3, P2, P1, and P1' amino acids. The substrates of CoVs $3CL^{pro}$ display almost identical recognition motifs, which is consistent with the high structural similarity of the catalytic domains of these enzymes [79]. The $3CL^{pro}$ of several α - and β -CoVs have the Leu-Gln-Ser (Ala, Gly) as preferred P2–P1–P1' sequence, suggesting the possibility to identify very effective broad-spectrum inhibitors. In particular, the requirement for the P1 Gln is almost a unique feature that is common only to enteroviruses (EV) $3C^{pro}$ and is unknown in human proteases, thus increasing the appeal of these viral proteins as targets for safe and selective inhibitors.

The SARS-CoVs 3CL^{pro} substrate affinity and specificity can be reasonably explained by the shape and amino acid composition of the different sub-pockets of the cleavage site. In both SARS-CoVs 3CL^{pro}, the S1 pocket is formed by the side chains of residues Phe140, Asn142, His163, Glu166, and His172 and the main chains of Phe140 and Leu141. Interestingly, the imidazole of His163 is located at the very bottom of the cleft, suitably positioned to donate a H-bond to the side chain carbonyl of substrate/inhibitor P1 Gln. The S2 subsite of SARS-CoVs 3CL^{pro} is a buried hydrophobic pocket that can host bulky alkyl/ aryl substituents as the substrate P2 Leu side chain. This cage is defined by a "lid" comprising the 3₁₀ helix residues 46–51, particularly Met49, three walls defined by the main chain of residues 186–188 and by the side chains of His41, Asp187 and Gln189, and a floor lined by Met165. Notably, the shape and the size of the S2 subsite of SARS-CoVs 3CL^{pro} are highly similar to that of the MERS-CoV homologue. Indeed, only two conservative mutations can be found, specifically the replacement of Met49 and Arg188 in SARS-CoVs with Leu49 and Lys191 in MERS-CoV, respectively. On the other hand, the volume of the S2

subsite in SARS-CoVs $3CL^{pro}$ (252Å³) is significantly larger than in other CoVs' homologues of the α -genus, such as the HCoV-NL63 $3CL^{pro}$ (45Å³) [80]. Compared with S1 and S2, the S4, S3, and the S1' pockets of the $3CL^{pro}$ are more shallow and exposed to the solvent and could accommodate groups of various size and nature. In particular, S3 and S4 are defined by the flexible loops connecting residues 165–168 and 189–192, which can rearrange upon ligand binding, while the S1' cleft is characterized by a number of threonine residues (24–26) which can potentially form either hydrogen bonds or lipophilic contacts with the substrate/inhibitor P1' group.

The targeting of proteases represents a solid route for antiviral drug discovery as demonstrated by the therapeutic success of HIV and HCV proteases inhibitors. The more attractive and time-saving approach to identify protease inhibitors is the design of peptidomimetics endowed with an electrophilic warhead thus acting as covalent irreversible/reversible inhibitors, that are designed based on the substrate sequences. Thus, it is important to gain specificity for target proteins, which reduces their potential side toxicity.

In 2003, the first X-ray structure of the SARS-CoV-1 $3CL^{pro}$ in complex with the peptide inhibitor **2** (Figure 17.10), functionalized with a chloromethyl ketone as warhead, was released (PDB ID 1UK4) [81]. Moreover, the ER $3C^{pro}$ inhibitor Rupintrivir, that successfully completed Phase II trials for human rhinovirus (HRV) infection in 1999 without showing toxicity in common cold patients [82], in the complex with HRV2 $3C^{pro}$ (PDB ID 1CQQ) showed an orientation comparable to that of peptide **2** bound to $3CL^{pro}$ [58, 83]. Although Rupintrivir is not active against CoVs protease [84, 85] likely due to the presence of a P2-*p*-fluorobenzyl group that is too large for the $3CL^{pro}$ S2-pocket [86], this investigational drug has inspired design of CoV $3CL^{pro}$ inhibitors [58]. Indeed, the Gln mimetic γ -lactam, that replaces the substrate P1 Gln in Rupintrivir, constitutes a key structural moiety kept constant in the large majority of the CoVs $3CL^{pro}$ from SARS-CoV-1 and other CoVs have proven to be active also against the SARS-CoV-2 $3CL^{pro}$ and/or have been optimized to develop new inhibitors. This supports the hypothesis to develop a pan-CoVs inhibitor.

In the peptidomimetic covalent reversible/irreversible inhibitors of SARS-CoVs $3CL^{pro}$ reported so far, the explored chemical warheads include Michael acceptors, aldehydes, (halomethyl-, aryl-, hydroxymethyl-, acyloxymethyl-) ketones, and α -ketoamides [57, 58]. In order to efficiently compete with the natural substrates at the catalytic site, most of the inhibitors span from P1 to P4 to establish a considerable number of specific interactions with the binding pockets (namely, the enzyme S1–S4 subsites), to facilitate



Figure 17.10 Chloromethylketone peptide **2** and Rupintrivir as starting point for the design of 3CL^{pro} inhibitors.



Figure 17.11 Schematic representation of the main features of covalent reversible/irreversible 3CL^{pro} inhibitors.

the covalent reaction between the Cys145 and the warhead (Figure 17.11), contributing to the selectivity, affinity, potency, and to the binding specificity. In addition, the reported inhibitors share similar peptide sequence organization in P1–P2 residues, while there are remarkable differences at P3/P4 positions and in the overall compound molecular size (Figure 17.11).

The first publication on compounds specifically designed to target the SARS-CoV-1 3CL^{pro} described a small series of Rupintrivir analogues endowed with a vinyl ester acting as the Michael acceptor warhead, but showing differences in the P2 [83]. Indeed, compound **3** has a smaller benzyl instead of the *p*-F-benzyl of Rupintrivir, enough to equip compound **3** with inhibitory activity against the enzyme ($K_{\text{inact}} = 0.014 \text{ min}^{-1}$), showing also a moderate antiviral activity in a cell-based assay (EC_{50 SARS-CoV} = 45 µM, CC₅₀ > 100 µM) (Figure 17.12).

Subsequently, the tripeptide derivative **4** endowed with an aldehyde warhead and resembling the sequence of compound **2** was reported as potent inhibitor of SARS-CoV-1 $3CL^{pro}$ (Figure 17.12) [87]. In particular, the presence of the cyclohexyl alanine in P2 and of a *tert*-butyl threonine in P3 provided potent enzymatic and antiviral activity ($K_{i \text{ SARS-CoV-1}} = 53 \text{ nM}$, EC_{50 SARS-CoV-1} = 0.6µM). The X-ray structures of compound **4** bound to SARS-CoV-1 $3CL^{pro}$ (PDB ID 2GX4) or SARS-CoV-2 $3CL^{pro}$ (PDB ID 7C8T) showed the formation of the thiohemiacetal derived from the nucleophilic attack of Cys145 to the C-term aldehyde of the inhibitor, with the oxyanion stabilized by H-bond of Gly143 and Cys145.

A small series of *N*-benzyloxy capped dipeptides functionalized with a fluoromethylketone warhead were reported as inhibitors of SARS-CoV-1 replication in cell-based assays, with compound **5** showing $\text{EC}_{50} = 2.4 \,\mu\text{M}$ and low cytotoxicity in Vero cells ($\text{CC}_{50} > 100 \,\mu\text{M}$, SI > 40) while no data on the enzymatic assay are available (Figure 17.12) [88]. Nonetheless, compound **5** was not active against HRV thus hinting to a specific inhibition against CoVs. The main difference with most of the 3CL^{pro} inhibitors is the presence of a *N*,*N*-dimethylglutamine residue in P1, instead of the γ -lactam.

A series of dipeptides having a ketobenzothiazole warhead was reported as potent inhibitors of SARS-CoV-1 3CL^{pro} (Figure 17.12) [89]. Similar to inhibitors reported by Pfizer (Section 17.2.2), these compounds were characterized by the presence of an heterocyclic drug-like N-cap, with the 4-methoxyindolyl derivative **6** showing the best activity in the nM range ($K_i = 6 \text{ nM}$, IC₅₀ = 0.74 µM, $K_D = 16 \text{ nM}$), while no data on antiviral activity in cellbased assays have been described.



Figure 17.12 Compounds **3–6** reported as SARS-CoV-1 3CL^{pro} inhibitors; ^aantiviral activity determined by NR assay; ^bantiviral activity determined by CPE reduction.

In the first released X-ray structure of SARS-CoV-2 3CL^{pro}, the protein was in complex with the N-term isoxazole capped tetrapeptide **7** endowed with a vinyl carboxyl ester acting as the Michael acceptor warhead (PDB ID: 6LU7, superseded by 7BQY; Figures 17.13 and 17.14) [79]. Compound **7** was previously reported as an inhibitor of 3CL proteases of other human CoVs [90, 91], including SARS-CoV-1 ($K_i = 9 \mu M$ [92]) and MERS-CoV (IC₅₀ = 0.3 μM) [93], while no data in an enzymatic assay were reported against SARS-CoV-2 3CL^{pro}. Nonetheless, the high structural identity between SARS-CoVs 3CL proteases would suggest that **7** may reasonably inhibit also SARS-CoV-2 3CL^{pro}. In a cell-based assay, compound **7** inhibits SARS-CoV-2 replication with an EC₅₀ of 16.77 μM , while it has modest selective index with respect to other CoVs (EC₅₀ = 4.0, 8.8, 2.7, and 3.4 μM against HCoV-229E, FIPV, MHV-A59, and MHV, respectively), and most likely the variability depends on subtle differences in amino acid sequences in the different CoV proteases and cellular assay conditions.

Peptidomimetic α -ketoamides have been reported as broad-spectrum inhibitors of 3CL and 3C proteases effective in cell lines against different CoVs and EVs, with derivative **8** showing the most promising activity (Figure 17.13) [80]. Preliminary chemical optimization of **8** by structure-based drug design (SBDD) led to derivative **9**, the first published SARS-CoV-2 3CL^{pro} inhibitors designed *ad-hoc* (Figure 17.13) [78]. The X-ray structure of the complex **9**/SARS-CoV-2 3CL^{pro} (PDB ID: 6Y2F) (Figure 17.14) confirmed the formation of a thiohemiketal intermediate by the nucleophilic attack of the Cys145 over the ketoamide α -carbonyl [78]. The oxyanion group is stabilized by a H-bond with His41, while the amide oxygen of **9** accepts a H-bond from the main-chain amides of Gly143 and Cys145 which, together with Ser144, form



Figure 17.13 Compounds **7–12** with their biological activities; ^aantiviral activity evaluated by plaque assay; ^bdata from Ref. [80]; ^cantiviral activity evaluated by viral RNA qRT-PCR quantification; ^dantiviral activity evaluated by CPE reduction. The main structural differences, significant modifications, and warheads are highlighted.

the so-called "oxyanion hole" typical of cysteine proteases. Ketoamide derivative **9** inhibits SARS-CoV-2 $3CL^{\text{pro}}$ with an $IC_{50} = 0.67 \,\mu\text{M}$, showing similar potency against SARS-CoV-1 and MERS-CoV M^{pro}. **9** showed EC₅₀ ~4–5 μ M against SARS-CoV-2 replication in human lung (Calu3) cells but no data on cytotoxicity were provided (Figure 17.13).

In another study, a SBDD strategy led to the identification of dipeptides **10** and **11** having an aldehyde as warhead and exhibiting excellent inhibitory activity against SARS-CoV-2 3CL^{pro} in enzymatic assays ($IC_{50} = 53$ and 40 nM, respectively; Figures 17.13 and 17.14) and SARS-CoV-2 infection in cell culture ($EC_{50} = 0.53$ and 0.72 μ M, respectively), coupled with low toxicity ($CC_{50} > 100 \,\mu$ M, SIs > 139) [94]. Both compounds have an indolyl acid in P3 and are reminiscent of 3CL^{pro} inhibitors reported by Pfizer,



Figure 17.14 Co-crystallographic pose of compounds PF-00835231, **4**, **7**, **9**, **10**, and **12** in complex with SARS-CoV-2 3CL^{pro}. The compounds' co-crystallographic poses are shown only into the active site of SARS-CoV-2, even for those co-crystallized in SARS-CoV-1, given the high degree of structural similarity between SARS-CoV-1 and SARS-CoV-2 3CL^{pro} enzymes. The key residues forming the active site pocket are displayed as white sticks, water molecules are shown as red spheres, H-bonds are depicted as dashed black lines. (a) PF-00835231 (pink sticks, PDB ID: 6XHM). (b) **4** (blue sticks, PDB ID: 7C87). (c) **7** (violet sticks, PDB ID: 7BQY). (d) **9** (green sticks, PDB ID: 6Y2F). (e) **10** (yellow-orange sticks, PDB ID: 6LZE), (f) **12** (teal sticks, PDB ID: 7BRR).

such as PF-00835231. As expected, compounds **10** and **11** bind to the protein with similar poses, where the carbon of the aldehyde and the sulphur of Cys145 form the thiohemiacetal, the oxygen of the resulting tetrahedral adduct is stabilized by interaction with backbone of residue Cys145 and through a water bridge with the Thr26 side chain.

We report only the co-crystal structure of **10**/SARS-CoV-2 3CL^{pro} as representative example (PDB IDs: 6LZE; Figure 17.14) [94].

The α -hydroxy-bisulfite dipeptide **12** (GC376), originally developed as investigational veterinary drug (usually administered via s.c. route) for feline infectious peritonitis (FIP) [95, 96], is able to potently inhibit SARS-CoV-2 $3CL^{pro}$ (IC₅₀ = 30 nM) and viral replication, even though with almost two orders lower potency ($EC_{50} = 3.37 \,\mu\text{M}$ in CPE assay; $CC_{50} > 100 \,\mu\text{M}$; Figure 5) (Figure 17.13). [85] This inhibitor is a prodrug, converted into the aldehyde form by the removal of the bisulfite group to alkylate the Cys145 of the 3CL^{pro}. Derivative 12 shows activity against multiple 3CL proteases ($IC_{50 \text{ FIPV 3CL}} = 0.72 \,\mu\text{M}$, $IC_{50 \text{ TGEV 3CL}} = 0.82 \,\mu\text{M}$, $IC_{50 \text{ SARS-CoV-1 3CL}} = 4.35 \,\mu\text{M}$, $IC_{50 \text{ MERS-CoV 3CL}} = 1.56 \,\mu\text{M}$) and against CoVs (TGEV, FIPV, MHV, 229E, and BCV) in cell lines with high nM potency [95-97]. 12 resulted also in potent inhibition of EV 3C and norovirus 3CL cysteine proteases and of the replication of several of these viruses, including HRVs [97]. Notably, 12 has been co-crystallized with SARS-CoV-2 3CL^{pro} (PDB IDs: 7BRR [98], 6WTJ [99], 6WTT [85], 7C8U, 7C6U, 7CBT) (Figure 17.14). Recently, a small set of new analogues of compound 12, differing for the N-terminal capping moieties, has been reported as inhibitors of 3CL^{pro} from SARS-CoVs and MERS-CoV [100]. The new compounds showed biological activities in the same range of parent compound 12 and have proven in vivo activity in mice infected with MERS-CoV.

The anti-HCV drug Boceprevir [101] showed low μ M potency against the isolated SARS-CoV-2 3CL^{pro} and the viral replication with no host cell toxicity up to 100 μ M. Despite deviating from 3CL substrate sequence specificity, Boceprevir has been recently co-crystallized with SARS-CoV-2 3CL^{pro} (PDB IDs: 7BRP [98], 6WNP, 7C6S, 6ZRU) providing hints to improve its anti-enzymatic activity (Figure 17.15). To be noted, none of the HIV protease inhibitors, including lopinavir, have shown activity in these studies, thus providing an experimental explanation to the failing of these drugs in clinical trials [41].

In spite of different warheads, most of the reported 3CL^{pro} inhibitors share in P1 a Gln mimetic y-lactam. As shown in the co-crystal structures of compounds PF-00835231, 4, 7, 9, 10, and 12, the P1 γ -lactam specifically fills the S1 subsite establishing key H-bonds with the Phe140 main chain and with the side chains of His163 and Glu166 (Figure 17.14). Notably, Glu166 is involved not only in substrate recognition but also in the substrate-induced dimerization of SARS-CoVs 3CL proteases through the interaction with Ser1 from the other monomer [78, 102]. Therefore, the interaction of the inhibitor γ -lactam with this residue could stabilize the monomeric and inactive form of the enzyme. Literature data on CoVs 3CL^{pro} inhibitors clearly indicate that the P1 γ -lactam enhances the inhibitory potency up to 10-fold, probably because the higher rigidity reduces the loss of entropy upon binding, if compared with the flexible Gln [58]. Paradoxically though, Boceprevir having a cyclobutyl ring in place of the γ -lactam, cannot form any hydrogen bond at the S1 subsite, but nevertheless was able to inhibit the SARS-CoV-2 3CL^{pro}. The presence of a modified proline in the Boceprevir at the P2 position would suggest the possibility to synthesize proline-based analogues with higher affinity and specificity toward the target enzyme. Indeed, in the first trimester of 2021, new P2 bicycloproline-containing compounds derived from Boceprevir have been reported as potent inhibitors of SARS-CoV-2 3CL^{pro} with IC₅₀ in the nM range [103]. Some of the new compounds showed also antiviral activity in cell-based assays and reduced lung viral loads and lung lesions in SARS-CoV-2 infection transgenic mouse model, after oral or ip treatment. A similar design strategy has been exploited for the oral clinical candidate PF-07321332



Figure 17.15 (a) Boceprevir and its biological activity against SARS-CoV-2. ^aAntiviral activity evaluated by viral plaque assay; ^bantiviral activity evaluated by CPE reduction. (b) Co-crystallographic pose of Boceprevir (purple sticks, PDB ID: 6WNP) in complex with SARS-CoV-2 3CL^{pro}. The key residues forming the active site pocket are displayed as white sticks; water molecules are displayed as red spheres. H-bonds are depicted as dashed black lines.

(Figure 17.5), containing the druglike bicycloproline in P2 similarly to Boceprevir, as described in Section 17.2.2. Indeed, the P2 amino acids of the other reported inhibitors show that this position tolerates a wide variety of aliphatic or aromatic residues characterized by similar lipophilicity and size to the substrate Leu and that it is important to modulate activity. The S3 and S4 pockets of SARS-CoVs 3CL^{pro} are less structured and can rearrange upon the binding of distinct P3/P4 residues that are generally modified to modulate both potency and drug-like properties of the inhibitors. For instance, in peptide 7, the P3 lipophilic Val is solvent exposed, although it can establish van der Waals contacts with P1 γ-lactam to stabilize the inhibitor binding conformation, while the P4 N-capped Ala can form some lipophilic contacts with residues Met165, Leu167, and Gln192 in S4 (Figure 17.14). Remarkably, the P3 backbone of the inhibitors 4 and 7 establishes two key hydrogen bonds with the main chain of Glu166. These interactions can also be formed by derivative 9 (Figure 17.14) in which the P2–P3 amide bond is masked by a N-Boc-aminopyridone, likely to prevent cellular proteases cleavage. Indeed, the pyridone CO donates a H-bond to Glu166, which in turn accepts another H-bond from the NH of the ligand carbamate. Similar to 9, the N-terminal P3 indole of derivatives PF-00835321 and 10 does not protrude into the S3 but it is exposed to solvent and stabilized by H-bonds with the amide backbone Glu166 (Figure 17.14). Also, in derivative 12, the P2 Leu is capped by a benzyloxycarbonyl group which can alternatively extend toward the S3/S4 pockets or downward to form intramolecular contacts with the P1 γ -lactam; in both cases, however, this moiety can form only one hydrogen bond with Glu166 with respect to the two formed by the other inhibitors. On the other hand, the role of the P1' substituent when present needs to be further investigated. Worth noting, some of the cocrystallized inhibitors, namely derivatives 4, 7, 12, and Boceprevir, show that the inhibitors fold in such conformation that bring in closer proximity the P3 residue (compounds 3, 7, and Boceprevir) or the N-benzyloxycarbonyl cap (compound 12) and the P1 substituent, thus suggesting that P1-P3 macrocyclization could be explored as optimization strategy. Indeed,

macrocyclization has been widely employed as strategy to preorganize HCV NS3 protease inhibitors in their bioactive conformation leading to BILN-2061 [104, 105] and the approved pan-genotypic drug Grazoprevir [106, 107]. Moreover, macrocyclization could lead to additional advantages such as greater stability and improved PK properties.

Beside PF-00835231 and its prodrug PF-07304814 in clinical Phase 1b, also the new inhibitors 9-11 have been evaluated for their in vivo PK properties in mice. Pyridone 9 upon sc administration (3 mg/kg) showed modest half-life ($T_{1/2}$ < 2 hours) and a rather low C_{max} (around 126 ng/ml), indicating a suboptimal PK profile and the need of further optimization [78]. More interestingly, indole derivatives 10 and 11 were dosed via iv administrations (single dose) at (5 mg/kg), and ip (5 and 20 mg/kg) resulting in reasonable profiles with $T_{1/2}$ ranging from ~ 2 to ~ 5 hours, depending on the administration route and the compounds, CL = 17and 21 ml/min/kg, respectively, via iv, high C_{max} (>2390 ng/ml) and availability >85% [94]. Overall, comparing PK profiles of both compounds after iv administration at 5 mg/kg, 10 appeared metabolically more stable than 11, thus compound 10 was further evaluated in rats (10 mg/kg, iv) and dogs (5 mg/kg, iv) showing good half-life values (rat, 7.6 hours and dog, 5.5 hours), low clearance (rat, 4.01 ml/min/kg and dogs, 5.8 ml/min/kg), and high AUCs (rat, 41 500 h*ng/ml and dog, 14 900 h*ng/ml). The PK profile was considered good enough to proceed to acute and seven-day toxicity studies in rats and dogs. In particular, compound 10 was evaluated for acute toxicity in rats by iv administration at 24 mg/kg (1 rat), 40 mg/kg (10 rats), and 60 mg/kg (4 rats), one rat from the last group died. The seven-day studies were carried out in rats and dogs by via IV administration at 2, 6, 18 mg/kg of the compound to four rats per study. Four dogs were dosed via IV at 10 mg/kg (the first day), 15 mg/kg (the second day), 20 mg/kg (the third day), 25 mg/kg (the fourth day), 25 mg/kg (the fifth to seventh days, randomly two dogs), and 40 mg/kg (the fifth to seventh days, other two dogs). All animals were clinically observed during seven days for toxic signs, which include bodyweight, food intake, and hematology, and no anomalies were observed. Overall, derivative 10 shows a PK profile comparable to PF-00835231. Therefore, the most promising compounds identified as CoVs 3CL^{pro} inhibitors are reversible covalent inhibitors (ketone of aldehyde) relying on small dipeptide sequences N-capped with drug-like indolyl acid, which establish key interaction in the active site of the enzyme and likely provide suitable PK properties. Indeed, as described in Section 17.2.2, the phosphate prodrug of PF-00835231, namely PF-07304814 has entered testing in humans in a Phase Ib clinical trial (NCT04535167). Moreover, a further very promising direction is the design of small peptidomimetic inhibitors based on the bicycloproline moiety as P2 fragment, as in the oral clinical candidate PF-07321332.

17.3.3 The S Protein: Structure, Function, and Inhibition

In the known human CoVs, the S glycoproteins mediate attachment to host cell surface and entry. The S protein is a 1200 aa long homotrimeric class I fusion protein, synthesized in the secretory pathway of the host cells, comprising a large ectodomain with a receptor-binding subunit S1 and a membrane-fusion subunit S2, a single-pass transmembrane anchor, and a short intracellular tail [108]. The N-linked glycans protruding from the trimer surface are responsible for the S folding and for the recognition by neutralizing antibodies [109].

Both SARS-CoVs enter into cells through the binding of S1 RBDs to the host ACE2, even though RBDs show moderate sequence similarity (64% identity) [15]. Conversely, the S2

domain is highly conserved and contains the FP, heptad repeat 1 (HR1), heptad repeat 2 (HR2), transmembrane region (TM), and cytoplasmatic tail (CT) (90% identity, with HR1 and HR2 identity of 92.6 and 100%, respectively) [110].

Cryo-EM structures of SARS-CoV-1 (PDB IDs: 3JCL [111], 5X58 [112]) and SARS-CoV-2 (PDB IDs: 6VXX [113], 6VSB [110]) S-trimer revealed a metastable prefusion ("down") conformation inaccessible to the host receptor, that changes into a less stable receptor-accessible ("up") conformation, exposing the determinants for receptor binding (Figure 17.16). The RBD undergoes a hinge-like movement to bind the peptidase domain of ACE2, resulting in a shedding in S1 and a cleavage in S2' site, with a refolding in S2 to adopt a stable post-fusion hairpin conformation [114, 115]. This six-helix bundle (6-HB) drives viral and cell membranes in close proximity for the fusion event [110].

The X-ray crystal structures of SARS-CoV-1 and SARS-CoV-2 RBD domains in complex with ACE2 (PDB ID: 2AJF and 6M0J, respectively) show the RBD presenting a loop-dominated concave surface with at the basis two-stranded antiparallel β -sheets and two ridges formed by loop interactions (Figure 17.17) [116, 117]. The α 1-helix of the ACE2 protease domain shows the main polar interactions with RBD, which contains nine cysteine residues with eight of them forming four disulfide bridges, in particular three in the RBD core and one in distal loops. Eight identical amino acids (Tyr 449/Tyr436, Tyr453/Tyr440, Asn487/Asn473, Tyr489/Tyr475, Gly496/Gly482, Thr500/Thr486, Gly502/Gly488 and Tyr505/Tyr491) have been found in SARS-CoV-2 and SARS-CoV-1 RBDs, respectively. On the other hand, key amino acids of SARS-CoV-1 RDB, i.e. Tyr442, Leu472, Asn479, Thr487, are replaced by Leu455, Phe486, Gln493, Asn501 in SARS-CoV-2 RDB, establish more interactions with the host receptor, along with a unique salt bridge between S Lys417 and ACE2 Asp30. Altogether, these differences may explain the about 20-fold higher affinity of SARS-CoV-2 S to ACE2 compared with SARS-CoV-1 and this likely is at the base of higher transmissibility of SARS-CoV-2 infection (Figure 17.17) [117, 118].



Figure 17.16 Structure of the SARS-CoV-2 S protein in its pre-fusion conformation, displayed as cartoon (gray, PDB ID: 6VSB) (a) and of a single protomer in the RBD up conformation (b) next to the RBD in the down conformation (c). RBD is colored cyan, the N-terminal domain (NTD) is pale green, subdomains 1 and 2 (SD1 and SD2) are yellow, the S2 domain is white, with HR1 colored salmon and FP hotpink.



Figure 17.17 Structural details of the interface between SARS-CoV-1 (pink, PDB ID: 2AJF) or SARS-CoV-2 RBDs and ACE2 (wheat), displayed as cartoon. The region enclosed by the black dashed lines, encompassing the interface between RBD and ACE2 is illustrated in detail in panel (bottom right). Overlay of the RBD interface residues of SARS-CoV-1 (pink sticks) and SARS-CoV-2 (cyan sticks) and Q493 is shown in two alternate positions (top right). Hydrogen bonds and salt bridges are displayed as dashed red lines.

A comparison between the 6-HB fusion core structure of SARS-CoV-2 and SARS-CoV-1 S proteins, in order to investigate the structural basis for S-mediated membrane fusion, has been carried out analyzing recombinant fusion proteins containing the major parts of HR1 (residues 910-988) and HR2 (residues 1162-1206) merged with a linker (L6, SGGRGG). Comparing the X-ray crystal structures of post-fusion hairpin conformation of SARS-CoV-1 S2 (PDB ID: 1WYY) [119] with the recently solved post-fusion core of SARS-CoV-2 S2 subunit (PDB ID: 6LXT) [120], the overall 6-HB structure of SARS-CoV-2 appears highly conserved with respect to those of other HCoVs, including SARS-CoV-1 and MERS-CoV, presenting hydrophobic residues (Val1164, Leu1166, Ile1169, Ile1172, Ala1174, Val1176, Val1177, Ile1179, Ile1183, Leu1186, Val1189, Leu1193, Leu1197 and Ile1198) in the central fusion core region (Figure 17.18). Fusion core of the SARS-CoV-2 S2 reveals eight different amino acids in the HR1 domain which provide stronger interactions between HR1 and HR2, whereas HR2 domain results fully identical; in particular, Ser929, Arg933, Asp936, Ser943, and Lys947 provide new strong H-bonds and stronger salt-bridge interactions, explaining the higher fusion activity compared with SARS-CoV-1 (Figure 17.17) [120].

The considerable amount of structural data on SARS CoVs RBD/ACE2 interaction provides useful structural information to drive the development of disruptors of the SARS-CoVs/ACE2 interaction. However, considering the high variability of RBD regions within S protein across different CoVs, the RBD motif is not an ideal target for the design of



Figure 17.18 The 6-HB fusion core structures of SARS-CoV-1 and SARS-CoV-2. (a) Overlay of 6-HB structures of SARS-CoV-1 (PDB ID: 1WYY) and SARS-CoV-2 (PDB ID: 6LXT) displayed as ribbon. The HR1 and HR2 domains are colored salmon and slate for SARS-CoV-2 raspberry, and deep teal for SARS-CoV-1, respectively. (b) Side view of SARS-CoV-2 and (c) SARS-CoV-1 6-HB displayed as cartoon. A close-up view of the interactions between HR1 and HR2 is shown on the right side. Key residues are displayed as sticks and labeled; hydrogen bonds and salt bridges are displayed as dashed red lines.

broad-spectrum inhibitors. Otherwise, HR1 and HR2 conserve their sequences and play a key role in viral fusion by forming the 6-HB [121]. Thus, the identification of peptide fusion inhibitors may represent a more advantageous strategy. Based on X-ray crystal structures of MERS-CoV and SARS-CoV-1 S protein fusion cores, several peptides derived from HR1 and HR2 sequences have been reported [120, 121]. However, SARS-CoV-1 and MERS-CoV HR1 peptides did not display antiviral activity likely due to their tendency of aggregation in the absence of HR2 sequences [16]. Interestingly, peptides derived from the HR2 region of SARS-CoV-1 [122] (namely, CP-1: GINASVVNIQKEIDRLNEVAKNLNESLIDLQELGKYE) andMERS-CoV[123](namelyHR2P:SLTQINTTLLDLTYEMLSLQQVVKALNESYIDLKEL) have been previously reported to inhibit HCoV S-mediated cell-cell fusion, using effector cells carrying S-GFP fluorescent fusion protein. Unfortunately, these HR2-derived peptides do not cross-inhibit MERS-CoV and SARS-CoV-1, thus lacking broad-spectrum antiviral activity against heterologous HCoVs. The insertion of some charged amino acids like Glu and Lys, as reported for HIV-1 fusion inhibitors, may increase the solubility and stability of these peptides as well as the introduction of mutations at specific sites not involved in HR1 binding, may enhance the antiviral activity [124]. By applying this strategy to the HR2P peptide derived from the short 6-HB fusion core of OC43-HCoV, EK1 has been identified (Figure 17.19) as a broad-spectrum inhibitor. It shows an IC_{50} ranging from 0.19 to 0.62 μ M in the cell fusion assays for SARS-CoV-1 and MERS-, 229E-, NL63-, OC43-, Rs3367-, WIV1-, and SHC014-CoVs [125], and dose-dependent antiviral activity against live HCoV infections. Preliminary data showed that EK1 peptide was effective against SARS-CoV-2 S protein-mediated membrane fusion and PsV infection in a dose-dependent manner [121].

	EK1	2 5 7 8/9 10 12 15/16 19 22/23 26 29/30 31 33 36 SLDQINVTFLDLEYEMKKLEEAIKKLEESY-IDLKEL			
SARS-HR1	948	LQKVLTNLAQANQNVVDQLKGLATSTTTLSEQIQSIAKNFQNAIQ	904		
OC43-HR1	1050	LQQLLNNLAEADANVVAQIKVLASNTADFGEQIADLANNFANAIL	1006		
MERS-HR1	1040	LESALKSLAQANNNVADQVKRFAENTTTFGTQMAGLAQNFKNAIL	996		
HKU1-HR1	1051	LQQLLSNLAQANSNVVSQIKALASNTASFGNQISLLANNFATAIL	1007		
229E-HR1	851	LQSTLHNLSNGQQNVVDQIKNLATAVTQLAQSTQTIADNVGTFAD	807		
NL63-HR1	1032	LQSTLHNLASGQQNVVDQIKNLAITVTHIAEATQTIADNVSSFSA	988		

Figure 17.19 The interactions of the broad-spectrum peptide inhibitor EK1 with HR1 residues of different HCoVs, including SARS-CoV-1. EK1 and HR1 residues connected with dashed gray lines locate to the same layers on the 3HR1 triple helix. Burying EK1 residues are highlighted in dark pink, and ridge-packing EK1 residues are highlighted in violet. HR1 residues forming conserved side-to-side and side-to-main chain hydrophilic interactions with EK1 residues are indicated with boxes colored blue and purple, respectively. *Source*: Adapted from Ref. [125].

A lipidation strategy was undertaken for optimization of the EK-1 peptide which was derivatized with cholesterol and palmitic acid covalently bound to the C-terminus with a spacer of polyethylene glycol, yielding lipopeptides EK1C and EK1P ($IC_{50} = 48.1$ and 69.2 nM, respectively, in the cell fusion assay using SARS-CoV-2 S) (Figure 17.20) [120]. Moreover, the introduction of a GSGSG linker between EK1 and cholesterol, as in the EK1C4 peptide, increases potency ($IC_{50} = 1.3 \text{ nM}$) in the cell-cell fusion assay. Thermal shift experiments confirmed the specific binding of EK1C4 to S-HR1s of SARS-CoVs and MERS-CoV. Interestingly, EK1C4 exerted very potent activity in cell-based assays against SARS-CoV-2, MERS-CoV, HCoV-OC43, HCoV-229E, and HCoV-NL63 (EC₅₀s ≤ 180 nM), with low cell toxicity ($CC_{50} > 5 \mu M$, SIs > 136). In particular, EK1C4 potently inhibits SARS-CoV-2 replication, with $EC_{50} = 36.5 \text{ nM}$, resulting 67-fold higher potency compared with EK1 (EC₅₀ = $2.47 \,\mu$ M) in the same assay. EK1C4 was also tested in mice infected with HCoV-OC43 (intranasal administration, 0.5 mg/kg) at different time points pre- (0.5, 2, 4, 12, and 24 hours) and post- (0.5 and 2 hours) infection. Pretreatment from 0.5 to 4 hours resulted in the 100% survival rate, at lower dosage than parent EK1 (20 mg/kg), while the protecting effect decreased to 83 and 0% at 12 and 24 hours, respectively. Post-treatment was evaluated at 0.5 and 2 hours postinfection, EK1C4 was highly effective (100%) if administered 0.5 hour postinfection while after 2 hours the effect was much lower (17%).

Other HR2 sequence-based fusion inhibitors have been reported as C-terminally cholesterol-linked IPB01 lipopeptides (Figure 17.21) [126]. IPB01–IPB04 peptides potently inhibit cell fusion in a dual-split protein (DSP)-based cell fusion assay of SARS-CoV-2,

- EK1 SLDQINVTFLDLEYEMKKLEEAIKKLEESYIDLKEL
- EK1P SLDQINVTFLDLEYEMKKLEEAIKKLEESYIDLKEL-PEG4-Palm
- EK1C SLDQINVTFLDLEYEMKKLEEAIKKLEESYIDLKEL-PEG4-Chol
- EK1C4 SLDQINVTFLDLEYEMKKLEEAIKKLEESYIDLKEL-GSGSG-PEG4-Chol
- Figure 17.20 Amino acid sequences of EK-1 and its lipopeptide derivatives.

IPB01 ISGINASVVNIQKEIDRLNEVAKNLNESLIDLQEL

IPB02 ISGINASVVNIQKEIDRLNEVAKNLNESLIDLQELK-Chol

IPB03 INASVVNIQKEIDRLNEVAKNLNESLIDLQELGK-Chol

IPB04 SVVNIQKEIDRLNEVAKNLNESLIDLQELGK-Chol

Figure 17.21 SARS-CoV-2 HR2-derived peptides.

showing IC₅₀ values between 15 and 33 nM. However, only IPB02 maintained comparable activity in cell-based assays against SARS-CoV-2 PV with EC₅₀ = 0.08 μ M while truncation at the N-terminus as in IPB03 and IPB04 resulted in 12- and 3-fold reduction in activity, respectively. IPB01 lacking the cholesterol unit is instead only weakly active (EC₅₀ = 33.7 μ M) in cell-based assay, thus the lipidation strategy is very important to improve activity in cells. In the early 2021, additional HR C-terminus derived lipopeptides have been reported as potent fusion inhibitors of SARS-CoV-2 in vitro and ex vivo. In particular, a dimeric 36-aminoacid lipopeptide showed the highest potency in inhibiting viral entry of SARS-CoV-2, including several emerging variants, of SARS-CoV-1 and of MERS-CoV, and prevented direct-contact SARS-CoV-2 transmission in ferret animal models. [127].

17.4 Conclusions

CoVs are important human pathogens that cause infections with symptoms that range from mild symptoms to a serious or even lethal pathology, with SARS-CoVs and MERS-CoV associated with the most serious diseases. Because of the outbreak of SARS-CoV-1 in 2002 and MERS-CoV in 2011, the scientific community has been involved in characterization of these viruses and exploration of antiviral strategies. However, even if associated with fairy high mortality, the control of SARS-CoV-1 infection in 2003 and the limitation of MERS-CoV spreading to an area of the Middle East have somehow decreased the interest to find drugs against these viruses. The pandemic outbreak of SARS-CoV-2 has now pushed the scientific community in collaboration with private and public funding organization to find effective antiviral agents to stop the COVID-19 crisis.

Drug repurposing has allowed the identification of RdRp NIs as first anti-CoVs agents. Hence, Remdesivir is the first and, at the time of writing this chapter, the only licensed drug to treat the COVID-19 disease and a SARS-CoV infection. However, its efficacy is modest and clinical data are still controversial, while iv administration route represents a limitation for its wide use. Another promising clinical candidate is the oral NI Molnupiravir currently evaluated in a Phase II trial. However, RdRp is still underexplored as antiviral target. Indeed, no specific inhibitors derived from focused medicinal chemistry programs have been reported, although the protein has been deeply characterized from a functional and structural point of view.

The most exploited strategy to identify antiviral agents against CoVs is the inhibition of the 3CL^{pro}, the main protease of these viruses. Indeed, many structural information and several inhibitors, mostly peptidomimetic covalent reversible/irreversible inhibitors, have

been reported for SARS-CoV-1, MERS-CoV, and other HCoVs, thus accelerating drug discovery also against SARS-CoV-2. Indeed, PF-00835231 is a very potent broad-spectrum inhibitor of CoVs 3CL^{pro} originally developed within a SARS-CoV-1 drug discovery campaign that recently has been exploited as precursor for the design of its prodrug PF-07304814 that is being evaluated in a Phase 1b trial in hospitalized participants with COVID-19. A further follow-up is represented by the related clinical candidate PF-07321332, that is the first orally administered inhibitor of SARS-CoV-2 3CL^{pro} and is currently in a Phase 1 clinical trial.

Additionally, the inhibition of the entry of CoVs into host cells is another deeply investigated approach to prevent/treat these infections. In this scenario, fusion inhibitors targeting the conserved HRs represent an approach. In particular, HR2 sequence-based peptides could pave the way for the development of new therapeutics with broad-spectrum anti-CoVs activity.

Finally, another putative emerging target is the papain-like protease, which is nevertheless a more challenging and underexplored protein for the identification of inhibitors. Furthermore, this protease is less conserved across the CoVs family, and for this reason it may be less attractive in view of possible future CoV outbreaks.

The COVID-19 pandemia has emphasized that control of the CoV infection will benefit from broad-spectrum antivirals, which today is still an unmet need. Therefore, stimulating research efforts toward this direction is of pivotal importance in order to face the present pandemic, but also potential future pandemics from new emerging CoVs or other zoonotic RNA viruses.

References

- **1** Woo, P.C.Y., Lau, S.K.P., Lam, C.S.F. et al. (2012). Discovery of seven novel mammalian and avian coronaviruses in the genus deltacoronavirus supports bat coronaviruses as the gene source of alphacoronavirus and betacoronavirus and avian coronaviruses as the gene source of gammacoronavirus and Deltacoronavirus. *J. Virol.* 86: 3995–4008.
- **2** Cui, J., Li, F., and Shi, Z.L. (2019). Origin and evolution of pathogenic coronaviruses. *Nat. Rev. Microbiol.* 17: 181–192.
- **3** Forni, D., Cagliani, R., Clerici, M., and Sironi, M. (2017). Molecular evolution of human coronavirus genomes. *Trends Microbiol.* 25: 35–48.
- **4** Schoeman, D. and Fielding, B.C. (2019). Coronavirus envelope protein: current knowledge. *Virol. J.* 16: 1–22.
- **5** Guan, Y., Zheng, B.J., He, Y.Q. et al. (2003). Isolation and characterization of viruses related to the SARS coronavirus from animals in southern China. *Science* 302: 276–278.
- **6** He, J.F., Peng, G.W., Min, J. et al. (2004). Molecular evolution of the SARS coronavirus, during the course of the SARS epidemic in China. *Science* 303: 1666–1669.
- 7 Erles, K., Toomey, C., Brooks, H.W., and Brownlie, J. (2003). Detection of a group 2 coronavirus in dogs with canine infectious respiratory disease. *Virology* 310: 216–223.
- 8 Wang, Q., Vlasova, A.N., Kenney, S.P., and Saif, L.J. (2019). Emerging and re-emerging coronaviruses in pigs. *Curr. Opin. Virol.* 34: 39–49.

- **9** Paules, C.I., Marston, H.D., and Fauci, A.S. (2020). Coronavirus infections-more than just the common cold. *JAMA* 323: 707–708.
- **10** De Wit, E., Van Doremalen, N., Falzarano, D., and Munster, V.J. (2016). SARS and MERS: recent insights into emerging coronaviruses. *Nat. Rev. Microbiol.* 14: 523–534.
- **11** Chafekar, A. and Fielding, B.C. (2018). MERS-CoV: understanding the latest human coronavirus threat. *Viruses*. 10: 93.
- Lu, R., Zhao, X., Li, J. et al. Genomic characterisation and epidemiology of 2019 novel coronavirus: implications for virus origins and receptor binding. *Lancet* 2020 (395): 565–574.
- **13** Zhang, Y.Z. and Holmes, E.C. (2020). A genomic perspective on the origin and emergence of SARS-CoV-2. *Cell* 181: 223–227.
- **14** Wu, A., Peng, Y., Huang, B. et al. (2020). Genome composition and divergence of the novel coronavirus (2019-NCoV) originating in China. *Cell Host Microbe* 27: 325–328.
- **15** Zhou, P., Yang, X.L., Wang, X.G. et al. (2020). Pneumonia outbreak associated with a new coronavirus of probable bat origin. *Nature* 579: 270–273.
- **16** Cannalire, R., Stefanelli, I., Cerchia, C. et al. (2020). Sars-Cov-2 entry inhibitors: small molecules and peptides targeting virus or host cells. *Int. J. Mol. Sci.* 21: 1–27.
- **17** Irigoyen, N., Firth, A.E., Jones, J.D. et al. (2016). High-resolution analysis of coronavirus gene expression by RNA sequencing and ribosome profiling. *PLoS Pathog.* 12: e1005473.
- **18** Ziebuhr, J., Snijder, E.J., and Gorbalenya, A.E. (2000). Virus-encoded proteinases and proteolytic processing in the Nidovirales. *J. Gen. Virol.*: 853–879.
- **19** Smith, E.C., Sexton, N.R., and Denison, M.R. (2014). Thinking outside the triangle: replication fidelity of the largest RNA viruses. *Annu. Rev. Virol.* 1: 111–132.
- **20** Eckerle, L.D., Becker, M.M., Halpin, R.A. et al. (2010). Infidelity of SARS-CoV Nsp14exonuclease mutant virus replication is revealed by complete genome sequencing. *PLoS Pathog.* 6: 1–15.
- **21** Angelini, M.M., Akhlaghpour, M., Neuman, B.W., and Buchmeier, M.J. (2013). Severe acute respiratory syndrome coronavirus nonstructural proteins 3, 4, and 6 induce double-membrane vesicles. *MBio* 4.
- **22** Bohn, M.K., Hall, A., Sepiashvili, L. et al. (2020). Pathophysiology of COVID-19: mechanisms underlying disease severity and progression. *Physiology* 35: 288–301.
- **23** Romagnoli, S., Peris, A., De Gaudio, A.R., and Geppetti, P. (2020). SARS-CoV-2 and COVID-19: from the bench to the bedside. *Phys. Rev.* 100: 1455–1466.
- **24** Wang, M., Cao, R., Zhang, L. et al. (2020). Remdesivir and chloroquine effectively inhibit the recently emerged novel coronavirus (2019-NCoV) in vitro. *Cell Res.* 30: 269–271.
- **25** Choy, K.T., Wong, A.Y.L., Kaewpreedee, P. et al. (2020). Remdesivir, lopinavir, emetine, and homoharringtonine inhibit SARS-CoV-2 replication in vitro. *Antivir. Res.* 178: 104786.
- 26 Runfeng, L., Yunlong, H., Jicheng, H. et al. (2020). Lianhuaqingwen exerts anti-viral and anti-inflammatory activity against novel coronavirus (SARS-CoV-2). *Pharmacol. Res.* 156: 104761.
- **27** Pruijssers, A.J., George, A.S., Schäfer, A. et al. (2020). Remdesivir inhibits SARS-CoV-2 in human lung cells and chimeric SARS-CoV expressing the SARS-CoV-2 RNA polymerase in mice. *Cell Rep.* 32: 107940.

- 28 Remdesivir EUA. https://www.gilead.com/-/media/files/pdfs/remdesivir/eua-fdaauthorization-letter_01may2020-old.pdf?la=en&hash=2AE25800C7C9612D2F67F28D 3F3B9921 (accessed 9 August 2020).
- **29** Goldman, J.D., Lye, D.C.B., Hui, D.S. et al. (2020). Remdesivir for 5 or 10 days in patients with severe Covid-19. *N. Engl. J. Med.*
- 30 Dolin, R. and Hirsch, M.S. (2020). Remdesivir an important first step. N. Engl. J. Med.
- **31** Beigel, J.H., Tomashek, K.M., Dodd, L.E. et al. (2020). Remdesivir for the treatment of Covid-19 final report. *N. Engl. J. Med.*
- **32** Cho, A., Saunders, O.L., Butler, T. et al. (2012). Synthesis and antiviral activity of a series of 1'-substituted 4-aza-7,9-dideazaadenosine C-nucleosides. *Bioorg. Med. Chem. Lett.* 22: 2705–2707.
- **33** Warren, T.K., Jordan, R., Lo, M.K. et al. (2016). Therapeutic efficacy of the small molecule GS-5734 against Ebola virus in rhesus monkeys. *Nature* 531: 381–385.
- 34 Siegel, D., Hui, H.C., Doerffler, E. et al. (2017). Discovery and synthesis of a phosphoramidate prodrug of a pyrrolo[2,1-f][triazin-4-amino] adenine C-nucleoside (GS-5734) for the treatment of Ebola and emerging viruses. J. Med. Chem. 60: 1648–1661.
- **35** Lo, M.K., Jordan, R., Arvey, A. et al. (2017). GS-5734 and its parent nucleoside Analog inhibit filo-, pneumo-, and paramyxoviruses. *Sci. Rep.* 7: 43395.
- **36** Lo, M.K., Feldmann, F., Gary, J.M. et al. (2019). Remdesivir (GS-5734) protects African Green monkeys from Nipah virus challenge. *Sci. Transl. Med.* 11: eaau9242.
- **37** Mulangu, S., Dodd, L.E., Davey, R.T. et al. (2019). A randomized, controlled trial of Ebola virus disease therapeutics. *N. Engl. J. Med.* 381: 2293–2303.
- **38** Sheahan, T.P., Sims, A.C., Graham, R.L. et al. (2017). Broad-spectrum antiviral GS-5734 inhibits both epidemic and zoonotic coronaviruses. *Sci. Transl. Med.* 9: eaal3653.
- **39** Brown, A.J., Won, J.J., Graham, R.L. et al. (2019). Broad spectrum antiviral remdesivir inhibits human endemic and zoonotic deltacoronaviruses with a highly divergent RNA dependent RNA polymerase. *Antivir. Res.* 169: 104541.
- **40** Williamson, B.N., Feldmann, F., Schwarz, B. et al. (2020). Clinical benefit of Remdesivir in rhesus macaques infected with SARS-CoV-2. *Nature* 585: 273–276.
- **41** WHO Solidarity trial consortium, Hongchao Pan, Richard Peto, et al. (2021). Repurposed antiviral drugs for COVID-19 interim WHO SOLIDARITY trial results. *N Engl. J. Med.* 384: 497–511.
- **42** McCreary, E.K. and Angus, D.C. (2020). Efficacy of Remdesivir in COVID-19. *JAMA* 324: 1041–1042.
- **43** Sheahan, T.P., Sims, A.C., Zhou, S. et al. (2020). An orally bioavailable broad-spectrum antiviral inhibits SARS-CoV-2 in human airway epithelial cell cultures and multiple coronaviruses in mice. *Sci. Transl. Med.* 12: eabb5883.
- **44** Urakova, N., Kuznetsova, V., Crossman, D.K. et al. (2017). β-D-N 4-hydroxycytidine is a potent anti-alphavirus compound that induces a high level of mutations in the viral genome. *J. Virol.* 92: e01965–e01917.
- **45** Toots, M., Yoon, J.J., Cox, R.M. et al. (2019). Characterization of orally efficacious influenza drug with high resistance barrier in ferrets and human airway epithelia. *Sci. Transl. Med.* 11: eaax 5866.
- 46 Agostini, M.L., Pruijssers, A.J., Chappell, J.D. et al. (2019). Small-molecule antiviral β-D-N
 4-hydroxycytidine inhibits a proofreading-intact coronavirus with a high genetic barrier to resistance. *J. Virol.* 93: e01348-19.
- 47 Ehteshami, M., Tao, S., Zandi, K. et al. (2017). Characterization of β-D-N4-hydroxycytidine as a novel inhibitor of Chikungunya virus. *Antimicrob. Agents Chemother*. 61: e02395-16.
- **48** Warren, T.K., Wells, J., Panchal, R.G. et al. (2014). Protection against Filovirus diseases by a novel broad-spectrum nucleoside analogue BCX4430. *Nature* 508: 402–405.
- **49** Taylor, R., Kotian, P., Warren, T. et al. (2016). BCX4430 a broad-spectrum antiviral adenosine nucleoside analog under development for the treatment of Ebola virus disease. *J. Infect. Public Health* 9: 220–226.
- **50** Kania, R.S., Mitchell Jr., L.J., and Nieman, J.A. (2006). Anticoronaviral compounds and compositions, their pharmaceutical uses and materials for their synthesis. WO2005113580A1.
- **51** Kania, R.S., Mitchell Jr., L.J., and Nieman, J.A. (2006). Anticoronaviral compounds and compositions, their pharmaceutical uses and materials for their synthesis. WO2006061714A2.
- 52 Hoffman, R.L., Kania, R.S., Brothers, M.A. et al. (2020). Discovery of ketone-based covalent inhibitors of coronavirus 3CL proteases for the potential therapeutic treatment of COVID-19. *J. Med. Chem.*, acs.jmedchem.0c01063.
- 53 Boras, B., Jones, R.M., Anson, B.J. et al. (2020). Discovery of a novel inhibitor of coronavirus 3CL protease as a clinical candidate for the potential treatment of COVID-19. *bioRxiv*, 2020.09.12.293498.
- **54** Pfizer investor day features significant number of pipeline advances for COVID-19 programs and across numerous therapeutic areas. Pfizer. https://www.pfizer.com/news/ press-release/press-release-detail/pfizer-investor-day-features-significant-number-pipeline (accessed 25 September 2020).
- 55 Gil, C., Ginex, T., Maestro, I. et al. (2020). COVID-19: drug targets and potential treatments. *J. Med. Chem.* 63: 12359–12386.
- 56 Liu, C., Zhou, Q., Li, Y. et al. (2020). Research and development on therapeutic agents and vaccines for COVID-19 and related human coronavirus diseases. ACS Cent. Sci. 6: 315–331.
- 57 Cannalire, R., Cerchia, C., Beccari, A.R. et al. (2020). Targeting SARS-CoV-2 proteases and polymerase for COVID-19 treatment: state of the art and future opportunities. *J. Med. Chem.* doi: https://doi.org/10.1021/acs.jmedchem.0c01140.
- **58** Pillaiyar, T., Manickam, M., Namasivayam, V. et al. (2016). An overview of severe acute respiratory syndrome-coronavirus (SARS-CoV) 3CL protease inhibitors: peptidomimetics and small molecule chemotherapy. *J. Med. Chem.* **59**: 6595–6628.
- **59** Xiu, S., Dick, A., Ju, H. et al. (2020). Inhibitors of SARS-CoV-2 entry: current and future opportunities. *J. Med. Chem.* 63: 12256–12274.
- **60** Subissi, L., Posthuma, C.C., Collet, A. et al. (2014). One severe acute respiratory syndrome coronavirus protein complex integrates processive RNA polymerase and exonuclease activities. *Proc. Natl. Acad. Sci. U. S. A.* 111: E3900–E3909.
- **61** Imbert, I., Guillemot, J.C., Bourhis, J.M. et al. (2006). A second, non-canonical RNAdependent RNA polymerase in SARS coronavirus. *EMBO J.* 25: 4933–4942.
- **62** Kirchdoerfer, R.N. and Ward, A.B. (2019). Structure of the SARS-CoV Nsp12 polymerase bound to Nsp7 and Nsp8 co-factors. *Nat. Commun.* 10: 2342.
- **63** Gao, Y., Yan, L., Huang, Y. et al. (2020). Structure of the RNA-dependent RNA polymerase from COVID-19 virus. *Science* 368: 779–782.
- **64** Yin, W., Mao, C., Luan, X. et al. (2020). Structural basis for the inhibition of the RNAdependent RNA polymerase from SARS-CoV-2 by Remdesivir. *Science* 368: 1499–1504.

- 482 17 A Focus on Severe Acute Respiratory Syndrome (SARS) Coronavirus (SARS-CoVs) 1 and 2
 - 65 Peng, Q., Peng, R., Yuan, B. et al. (2020). Structural and biochemical characterization of Nsp12-Nsp7-Nsp8 core polymerase complex from COVID-19 virus. *BioRxiv*, 2020.04.23.057265.
 - **66** Mcdonald, S.M. (2013). RNA synthetic mechanisms employed by diverse families of RNA viruses. *Wiley Interdiscip. Rev.* 4: 351–367.
 - 67 Lehmann, K.C., Gulyaeva, A., Zevenhoven-Dobbe, J.C. et al. (2015). Discovery of an essential nucleotidylating activity associated with a newly delineated conserved domain in the RNA polymerase-containing protein of all nidoviruses. *Nucleic Acids Res.* 43: 8416–8434.
 - **68** Hillen, H.S., Kokic, G., Farnung, L. et al. (2020). Structure of replicating SARS-CoV-2 polymerase. *Nature* 584: 154–156.
 - **69** De Clercq, E. and Neyts, J. (2009). Antiviral agents acting as DNA or RNA chain terminators. In: *Handbook of Experimental Pharmacology*, vol. 189 (eds. H.-G. Kräusslich and R. Bartenschlager), 53–84. Berlin, Heidelberg: Springer Berlin Heidelberg.
 - **70** Tchesnokov, E.P., Feng, J.Y., Porter, D.P., and Götte, M. (2019). Mechanism of inhibition of Ebola virus RNA-dependent RNA polymerase by Remdesivir. *Viruses* 11: 326.
 - **71** Jordan, P.C., Liu, C., Raynaud, P. et al. (2018). Initiation, extension, and termination of RNA synthesis by a paramyxovirus polymerase. *PLoS Pathog.* 14: e1006889.
 - **72** Gordon, C.J., Tchesnokov, E.P., Woolner, E. et al. (2020). Remdesivir is a direct-acting antiviral that inhibits RNA-dependent RNA polymerase from severe acute respiratory syndrome coronavirus 2 with high potency. *J. Biol. Chem.* 295: 6785–6797.
 - **73** Gordon, C.J., Tchesnokov, E.P., Feng, J.Y. et al. (2020). The antiviral compound Remdesivir potently inhibits RNAdependent RNA polymerase from middle east respiratory syndrome coronavirus. *J. Biol. Chem.* 295: 4773–4779.
 - **74** Wang, Q., Wu, J., Wang, H. et al. (2020). Structural basis for RNA replication by the SARS-CoV-2 polymerase. *Cell* 182: 417–428.e13.
 - **75** Agostini, M.L., Andres, E.L., Sims, A.C. et al. (2018). Coronavirus susceptibility to the antiviral Remdesivir (GS-5734) is mediated by the viral polymerase and the proofreading exoribonuclease. *MBio* 9: e00221-18.
 - 76 Sofia, M.J., Chang, W., Furman, P.A. et al. (2012). Nucleoside, nucleotide, and nonnucleoside inhibitors of hepatitis C virus NS5B RNA-dependent RNA-polymerase. J. Med. Chem. 55: 2481–2531.
 - 77 Tan, J., Verschueren, K.H.G., Anand, K. et al. (2005). PH-dependent conformational flexibility of the SARS-CoV Main proteinase (Mpro) dimer: molecular dynamics simulations and multiple X-Ray structure analyses. J. Mol. Biol. 354: 25–40.
 - **78** Zhang, L., Lin, D., Sun, X. et al. (2020). Crystal Structure of SARS-CoV-2 Main protease provides a basis for design of improved a-ketoamide inhibitors. *Science* 368: 409–412.
 - **79** Jin, Z., Du, X., Xu, Y. et al. (2020). Structure of Mpro from SARS-CoV-2 and discovery of its inhibitors. *Nature* 582: 289–293.
 - 80 Zhang, L., Lin, D., Kusov, Y. et al. (2020). α-Ketoamides as broad-spectrum inhibitors of coronavirus and enterovirus replication: structure-based design, synthesis, and activity assessment. J. Med. Chem. 63: 4562–4578.
 - 81 Yang, H., Yang, M., Ding, Y. et al. (2003). The crystal structures of severe acute respiratory syndrome virus Main protease and its complex with an inhibitor. *Proc. Natl. Acad. Sci.* U. S. A. 100: 13190–13195.

- **82** Hayden, F.G., Turner, R.B., Gwaltney, J.M. et al. (2003). Phase II, randomized, doubleblind, placebo-controlled studies of ruprintrivir nasal spray 2-percent suspension for prevention and treatment of experimentally induced rhinovirus colds in healthy volunteers. *Antimicrob. Agents Chemother.* 47: 3907–3916.
- **83** Ghosh, A.K., Xi, K., Ratia, K. et al. (2005). Design and synthesis of peptidomimetic severe acute respiratory syndrome chymotrypsin-like protease inhibitors. *J. Med. Chem.* 48: 6767–6771.
- **84** Matthews, D.A., Patick, A.K., Baker, R.O. et al. (2004). *In Vitro Antiviral Activity of Human Rhinovirus 3C Protease Inhibitors against the SARS Coronavirus*, 186–193. Washington, DC: National Academy of Sciences.
- **85** Ma, C., Sacco, M.D., Hurst, B. et al. (2020). Boceprevir, GC-376, and calpain inhibitors II, XII inhibit SARS-CoV-2 viral replication by targeting the viral Main protease. *Cell Res.* 30: 678–692.
- **86** Anand, K., Ziebuhr, J., Wadhwani, P. et al. (2003). Coronavirus main proteinase (3CLpro) structure: basis for design of anti-SARS drugs. *Science* 300: 1763–1767.
- 87 Yang, S., Chen, S.J., Hsu, M.F. et al. (2006). Synthesis, crystal structure, structure-activity relationships, and antiviral activity of a potent SARS coronavirus 3CL protease inhibitor. *J. Med. Chem.* 49: 4971–4980.
- **88** Zhang, H.Z., Zhang, H., Kemnitzer, W. et al. (2006). Design and synthesis of dipeptidyl glutaminyl fluoromethyl ketones as potent severe acute respiratory syndrome coronovirus (SARS-CoV) inhibitors. *J. Med. Chem.* 49: 1198–1201.
- **89** Thanigaimalai, P., Konno, S., Yamamoto, T. et al. (2013). Development of potent dipeptidetype SARS-CoV 3CL protease inhibitors with novel P3 scaffolds: design, synthesis, biological evaluation, and docking studies. *Eur. J. Med. Chem.* 68: 372–384.
- **90** Wang, F., Chen, C., Tan, W. et al. (2016). Structure of main protease from human coronavirus NL63: insights for wide spectrum anti-coronavirus drug design. *Sci. Rep.* 6: 22677.
- **91** Ren, Z., Yan, L., Zhang, N. et al. (2013). The newly emerged SARS-like coronavirus HCoV-EMC also has an "Achilles" heel ": current effective inhibitor targeting a 3C-like protease." *Protein Cell* 4: 248–250.
- **92** Xue, X., Yu, H., Yang, H. et al. (2008). Structures of two coronavirus main proteases: implications for substrate binding and antiviral drug design. *J. Virol.* 82: 2515–2527.
- **93** Yang, H., Xie, W., Xue, X. et al. (2005). Design of wide-spectrum inhibitors targeting coronavirus main proteases. *PLoS Biol.* 3: e324.
- **94** Dai, W., Zhang, B., Jiang, X.M. et al. (2020). Structure-based design of antiviral drug candidates targeting the SARS-CoV-2 main protease. *Science* 368: 1331–1335.
- **95** Kim, Y., Liu, H., Galasiti Kankanamalage, A.C. et al. (2016). Reversal of the progression of fatal coronavirus infection in cats by a broad-spectrum coronavirus protease inhibitor. *PLoS Pathog.* 12: 1–18.
- **96** Pedersen, N.C., Kim, Y., Liu, H. et al. (2018). Efficacy of a 3C-like protease inhibitor in treating various forms of acquired feline infectious peritonitis. *J. Feline Med. Surg.* 20: 378–392.
- **97** Kim, Y., Lovell, S., Tiew, K.-C. et al. (2012). Broad-spectrum antivirals against 3C or 3C-like proteases of picornaviruses, noroviruses, and coronaviruses. *J. Virol.* 86: 11754–11762.
- **98** Fu, L., Ye, F., Feng, Y. et al. (2020). Both Boceprevir and GC376 efficaciously inhibit SARS-CoV-2 by targeting its main protease. *Nat. Commun.* 11: 4417.

- 484 17 A Focus on Severe Acute Respiratory Syndrome (SARS) Coronavirus (SARS-CoVs) 1 and 2
 - **99** Vuong, W., Khan, M.B., Fischer, C. et al. (2020). Feline coronavirus drug inhibits the main protease of SARS-CoV-2 and blocks virus replication. *Nat. Commun.* 11: 4282.
 - 100 Rathnayake, A.D., Zheng, J., Kim, Y. et al. (2020). 3C-like protease inhibitors block coronavirus replication in vitro and improve survival in MERS-CoV–infected mice. *Sci. Transl. Med.* 12: eabc5332.
 - 101 Venkatraman, S., Bogen, S.L., Arasappan, A. et al. (2006). Discovery of (1R,5S)-N-[3-amino-1-(cyclobutylmethyl)-2,3-dioxopropyl]-3-[2(S)-[[[(1,1-dimethylethyl)amino]carbonyl]amino]-3,3dimethyl-1-oxobutyl]-6, 6-dimethyl-3-azabicyclo[3.1.0]hexan-2(S)-carboxamide (SCH 503034), a selective, potent, orally Bioavailable. *J. Med. Chem.* 49: 6074–6086.
 - **102** Cheng, S.C., Chang, G.G., and Chou, C.Y. (2010). Mutation of Glu-166 blocks the substrateinduced dimerization of SARS coronavirus main protease. *Biophys. J.* 98: 1327–1336.
 - **103** Qiao, J., Li, Y.-S., Zeng, R. et al. (2021). SARS-CoV-2 Mpro inhibitors with antiviral activity in a transgenic mouse model. *Science* 371: 1374–1378.
 - **104** Lamarre, D., Anderson, P.C., Bailey, M. et al. (2003). An NS3 protease inhibitor with antiviral effects in humans infected with hepatitis C virus. *Nature* 426: 186–189.
 - **105** Llinàs-Brunet, M., Bailey, M.D., Bolger, G. et al. (2004). Structure-activity study on a novel series of macrocyclic inhibitors of the hepatitis C virus NS3 protease leading to the discovery of BILN 2061. *J. Med. Chem.* 47: 1605–1608.
 - **106** Summa, V., Ludmerer, S.W., McCauley, J.A. et al. (2012). MK-5172, a selective inhibitor of hepatitis C virus NS3/4a protease with broad activity across genotypes and resistant variants. *Antimicrob. Agents Chemother.* 56: 4161–4167.
 - **107** Harper, S., McCauley, J.A., Rudd, M.T. et al. (2012). Discovery of MK-5172, a macrocyclic hepatitis C virus NS3/4a protease inhibitor. *ACS Med. Chem. Lett.* 3: 332–336.
 - **108** Bosch, B.J., van der Zee, R., de Haan, C.A.M., and Rottier, P.J.M. (2003). The coronavirus spike protein is a class I virus fusion protein: structural and functional characterization of the fusion core complex. *J. Virol.* 77: 8801–8811.
 - **109** Walls, A.C., Xiong, X., Park, Y.J. et al. (2019). Unexpected receptor functional mimicry elucidates activation of coronavirus fusion. *Cell* 176: 1026–1039.e15.
 - **110** Loganathan, S.K., Schleicher, K., Malik, A. et al. (2020). Rare driver mutations in head and neck squamous cell carcinomas converge on NOTCH signaling. *Science* 367: 1264–1269.
 - **111** Walls, A.C., Tortorici, M.A., Bosch, B.J. et al. (2016). Cryo-electron microscopy structure of a coronavirus spike glycoprotein trimer. *Nature* 531: 114–117.
 - **112** Yuan, Y., Cao, D., Zhang, Y. et al. (2017). Cryo-EM structures of MERS-CoV and SARS-CoV spike glycoproteins reveal the dynamic receptor binding domains. *Nat. Commun.* 8: 1–9.
 - **113** Walls, A.C., Park, Y.J., Tortorici, M.A. et al. (2020). Structure, function, and antigenicity of the SARS-CoV-2 spike glycoprotein. *Cell* 181: 281–292.e6.
 - 114 Walls, A.C., Tortorici, M.A., Snijder, J. et al. (2017). Tectonic conformational changes of a coronavirus spike glycoprotein promote membrane fusion. *Proc. Natl. Acad. Sci. U. S. A.* 114: 11157–11162.
 - 115 Pallesen, J., Wang, N., Corbett, K.S. et al. (2017). Immunogenicity and structures of a rationally designed prefusion MERS-CoV spike antigen. *Proc. Natl. Acad. Sci. U. S. A.* 114: E7348–E7357.
 - 116 Li, F., Li, W., Farzan, M. et al. (2005). Science 309: 1864-1868.
 - **117** Lan, J., Ge, J., Yu, J. et al. (2020). Structure of the SARS-CoV-2 spike receptor-binding domain bound to the ACE2 receptor. *Nature* 581: 215–220.

- Yan, R., Zhang, Y., Li, Y. et al. (2020). Structural basis for the recognition of SARS-CoV-2 by full-length human ACE2. *Science* 367: 1444–1448.
- **119** Duquerroy, S., Vigouroux, A., Rottier, P.J.M. et al. (2005). Central ions and lateral asparagine/glutamine zippers stabilize the post-fusion hairpin conformation of the SARS coronavirus spike glycoprotein. *Virology* 335: 276–285.
- Xia, S., Liu, M., Wang, C. et al. (2020). Inhibition of SARS-CoV-2 (previously 2019-NCoV) infection by a highly potent Pan-coronavirus fusion inhibitor targeting its spike protein that harbors a high capacity to mediate membrane fusion. *Cell Res.* 30: 343–355.
- 121 Xia, S., Zhu, Y., Liu, M. et al. (2020). Fusion mechanism of 2019-NCoV and fusion inhibitors targeting HR1 domain in spike protein. *Cell. Mol. Immunol.* 17: 765–767.
- Liu, S., Xiao, G., Chen, Y. et al. (2004). Interaction between heptad repeat 1 and 2 regions in spike protein of SARS-associated coronavirus: implications for virus fusogenic mechanism and identification of fusion inhibitors. *Lancet* 363: 938–947.
- Lu, L., Liu, Q., Zhu, Y. et al. (2014). Structure-based discovery of middle east respiratory syndrome coronavirus fusion inhibitor. *Nat. Commun.* 53067.
- Pang, W., Tom, S.C., and Zheng, Y.T. (2009). Current peptide HIV Type-1 fusion inhibitors. *Antivir. Chem. Chemother.* 20: 1–18.
- Xia, S., Yan, L., Xu, W. et al. (2019). A pan-coronavirus fusion inhibitor targeting the HR1 domain of human coronavirus spike. *Sci. Adv.* 5: eaav4580.
- Zhu, Y., Yu, D., Yan, H. et al. Design of potent membrane fusion inhibitors against SARS-CoV-2, an emerging coronavirus with high fusogenic activity. *J. Virol.* 2020: 2020.03.26.009233.
- de Vries, R.D., Schmitz, K.S., Bovier, F.T. et al. (2021). Intranasal fusion inhibitory lipopeptide prevents direct-contact SARS-CoV-2 transmission in ferrets. *Science* 371: 1379–1382.

a

Abreva[™] 139, 140, 147 acetyl-CoA carboxylase (ACC) 382, 384 acetylsalicylic acid (ASA) 81 "acquired immunodeficiency syndrome" (AIDS) 1 AIDS-associated KS 194 acute infectious conjunctivitis 285 acute respiratory distress syndrome (ARDS) 453 acyclic guanosine analogues 137-139 acyclic guanosine drug 139 acycloguanosine 137 acyclovir, aciclovir 135, 290, 420 adamantanes and derivatives 63-64 adaptive COVID-19 Treatment Trial (ACTT) 455 adefovir 196 adenain 290 adeno-associated virus (AAV) 298 adeno-virus 300 adenovirus 277 in human stem cell transplantation 279 current therapy strategies and antiviral agents 281-283 incidence, transmission and clinical manifestation 279-283 novel antiviral approaches in clinical development 283-285 receptor 277 replication cycle 280 infections 277-296

current therapy strategies and antiviral agents 285-289 novel antiviral approaches in clinical development 283-285 incidence, transmission and clinical manifestation in stem cell transplanted patients 279 - 281adoptive T cell transfer and immune checkpoint blockade 180 AG7404 356 Aldara[™] 142 alisporivir 386 6-alkoxy-substituted methylene cyclopropane nucleosides 196 alkaline phosphatase 257 allogeneic HSCT 279 alphavirus 367 Alzheimer's disease 144 amantadines and derivatives 62-63 amantadine-resistant A(H3N2) virus 64 Amenalief[™] 140 amenamevir 140 amiloride 357 amino acid (AA) virus capsid protein 255 aminotransferases 257 anaplastic large-cell lymphoma (ALCL) 179 Anelloviridae 297 angiotensin-converting enzyme 2 (ACE2) 451 anticancer drugs 384 anti-CD20 antibody Rituximab 179 antigenic shifts 60

New Drug Development for Known and Emerging Viruses, First Edition. Edited by Helga Rübsamen-Schaeff and Helmut Buschmann. © 2022 WILEY-VCH GmbH. Published 2022 by WILEY-VCH GmbH. anti-herpes simplex viruses drugs acyclic guanosine analogues 137 chemical structures of 138 docosanol 139 foscarnet 139 helicase-primase inhibitors 140 immunomodulators 142 monoclonal antibodies 140, 141 5-substituted 2-deoxyuridine analogues 136 therapeutic vaccine 140-142 vidarabine 137 antiretroviral combination therapy (ART) 1-12, 194, 210antiretroviral therapy 2 antisense therapies 411, 418 antiviral chemotherapy of parvovirus B19-infection 303 antiviral drugs arboviruses 374-375 antiviral strategies against human cytomegalo virus 151–174 APD-209 289 2A protease of enteroviruses 355 arbidol 387 arenaviruses 409, 420 New World 426-429 Old World Arenaviruses 420, 423–426 arthropod-borne viruses (arboviruses) 367 antipyretics 371 antiviral drugs and vaccines 372 antiviral therapy, targets for 374 capsid protein inhibitors 379 chikungunya virus 370 dengue viruses 369 direct acting antivirals 374, 375 endocytosis and membrane fusion inhibitors 387 envelope protein inhibitors 378–379 genome organization 371–373 host cell lipid biosynthesis inhibitors 382-384 host cell nucleoside biosynthesis inhibitors 381 host kinase inhibitors 384-385

host-targeting antivirals 380 inhibitors with nonspecific action 380 methyltransferase inhibitors 379-380 nonsteroidal anti-inflammatory drugs 371 NS4B inhibitors 379 protease/helicase inhibitors 377-378 protein metabolism inhibitors 385-386 RNA-dependent RNA polymerase 374-377 virus replication cycle 372 West Nile virus infections 369-370 vellow fever virus infections 368-369 zika virus infection 368 assembly inhibitors of, enteroviruses (EV) 357-358 asthma, development of 111 asunaprevir 21, 25, 40 atomic interaction network analysis 78 auristatin E conjugated form 179 autoimmune CD4+ T cells 179 autologous EBV transformed B cell lines 180 AUY-922 204, 207 avian hepatitis E virus 254 avian influenza virus H7N9 73 avian influenza virus strains (H5N1) 67.73 azidothymidine (AZT) 199, 201 azithromycin 358

b

bacterial coinfections and influenza 62 baloxavir acid (BXA) 71 baloxavir marboxil (BAM) 60, 62, 70 bat hepatitis E virus 254 B cell depleting therapy 177, 179 BCX-1812 67 BCX4430 413 beclabuvir 21, 24, 40 benzoxazole analogue 378 9-beta-d-arabinofuranosyladenine 137 bevacizumab 202, 206 BI-D1870 203, 205 bilirubin 257

Biosafety Level 4 (BSL4) pathogens arenaviruses 420–429 bunyaviruses 429–433 filoviruses 410-418 henipaviruses 418-420 viral hemorrhagic fever viruses 433-435 BK-Virus 279 boceprevir 25, 29, 30, 35, 470-471 bone marrow failure 196 bortezomib 201, 202, 207, 386 bosutinib 205 Bovine Respiratory Syncytial Virus (BRSV) 107 bovine virus diarrhea virus (BVDV) 14 brachial neuritis 257 brentuximab 179 Brequinar 381 brincidofovir 152, 153, 154, 283, 418 brivudine 137, 196 broadly neutralizing antibodies (BNAPs) 10 Bundibugyo virus (BDBV) 410 bunyaviruses 409 immunotherapy 430 overview 429 small molecules 430–433 Burkitt lymphoma 179 B19V 301-304

С

Calomys musculus 426 canine minute virus 297 capsid assembly for KSHV 209 capsid assembly effectors (CAEs) 234 capsid assembly modulators (CAMs) 232-234 in clinical trials 236, 242-243 first-generation 234-235 modes of action 237-238 resistance profile of 238-242 second-generation 243 structures 235 capsid binders 352, 355, 359 capsid hexons adenoviruses 277 capsid protein(s) parvoviruses 298 capsid protein inhibitors, arboviruses 379 CAPSTONE 1,2 trials 71, 72 castanospermine 384, 386 CD4-cell count 2 CD4-positive T-lymphocyte count 1 CD4-receptor positive cells 3 CD4-receptor positive T-cell 2 CD8+ T cells 180–182 CD26 452 CD46 277 celgosivir 386 cell-cell adhesion molecules 134 cell surface heparan sulfate proteoglycans 134 cellular chaperone heat shock protein 90 (HSP90) 207 cellular immunity 257 cellular topoisomerases 207 Chapare virus (CHPV) 426 chemokine receptor homologue vGCR (ORF74) 207 chikungunya fever (CHIKF) 370 chikungunya virus (CHIKV) 367, 370 genome organization 371 chimeric antigen receptors (CARs) 178, 181 chloromethylketone peptide 465 chloroquine 376, 380 5-(2-Chlorovinyl)-2'-deoxyuridine 196 chronic chikungunya arthritis (CCA) 370 chymotrypsin-like cysteine protease (3CL^{pro}) 454, 459 cidofovir 143, 152, 278, 282, 283, 290, 452 Circoviridae 297 circoviruses 297 classical H274Y NA mutation 74 2'-C-methylcytidine (2-CMC) 259, 261, 317.323-324 cognate antigen 176 combination antiretroviral therapy (cART) 2,4 core protein allosteric modulators for HBV (CpAMs) 234 coronaviruses (CoVs) 264, 449-486 3CL^{pro} 458–459, 463–472 clinical investigational drugs against coronaviruses 456-459

coronaviruses (CoVs) (Contd.) epidemiology of coronaviruses 449-450 licensed drugs against coronaviruses 455-456 corticosteroids 286 COVID-19 disease, phases of 454 disease progression 453 mortality 456 severity 454 coxsackievirus 277 3C protease 355 CRISPR-CASP-technique 10 CRISPR-Cas13 strategy 358 crizotinib 202, 206 CS-8958 69 2C targeting molecules for picorna viruses 356 C-terminal DNA binding domain (DBD) 211 cutavirus (CutaV) 297 cyclooxygenases (COX) 81 cyclophilins (CyPs) 384-386 cyclopropavir derivatives 196 Cymeven[®] 278, 282 cynomolgus macaque model 430 cytokine release syndrome (CRS) 181 cytokine storm 61 cytomegalovirus antiviral strategies 151-174 infections, potential indications 167-168 pUL viral protein kinase (vPK) 196 terminase inhibitors 151-156 cytoplasmatic tail (CT) of SARS CoV 473 cytotoxic CD8⁺ T cells 176 cytotoxic effect 322

d

daclatasvir 21, 33, 34, 36, 38–40
Dandenong virus 420
DAS181 79
dasabuvir 21, 23, 26, 40
dasatinib 202, 205
7-deaza-2'-C-methyladenosine

(7DMA) 319, 323, 324, 376

2,3-dehydro-2-deoxy-*N*-acetylneuraminic acid

(DANA) 64, 65

Denavir[™] 139 dengue viruses 369, 372 Dependoparvovirus 298 deptropine 260, 262 diacetyl 6-deoxy prodrug 139 diffuse large B cell lymphoma 179 dihydroorotate dehydrogenase (DHODH) inhibitors 381, 384 dihydroquinazoline derivatives 156 1,3-dihydroxy-2-propoxymethyl)purine 196 dipeptidyl peptidase 4 (DPP4) 452 direct acting antivirals (DAA) 14 combination therapies against HCV 21, 35 particle maturation 289-291 virus attachment and entry 289-290 virus genome replication 290 17-DMAG 203, 207 DMPK profiling 455, 456 DNA polymerase 196-201, 208-209 docosanol 139, 140 Down's syndrome 98 dual-split protein (DSP) 476 dual target therapeutics, enteroviruss (EV) 358

е

EBNA1 177-182 Ebola virus 73, 410 glycoprotein GP 411 edaphic CD4-receptor 1 EDP-938 106, 119 EIDD-1931 107, 119 EK-1 476 elevated EBNA1 specific antibodies 179 encapsidation of enteroviruses 350 endocytosis, and membrane fusion inhibitors 387 endoplasmic reticulum-golgi intermediate compartment (ERGIC) 452 endosomal sorting complexes required for transport (ESCRT) 256 enteroviruses (EV) assembly and release 350 classification and clinical impact 347

CRISPR-Cas13 strategy 358 directly acting antivirals 352 2C targeting molecules 356 early stage inhibitors 352-355 RNA-dependent RNA polymerase (3D) inhibitors 356-357 viral protease inhibitors 355-356 dual target therapeutics 358-359 genome structure 348 host targeting antivirals 357 assembly inhibitors 357-358 lipid processing, inhibitors of 357 monoclonal antibodies 358 prevention 351 replication cycle stages 351 entry 349-350 translation and replication 350 RNAi strategy 358 virion structure 347-349 enteroviruses of group A (EV-A) 347 enteroviruses of group B (EV-B) 347 enteroviruses of group C (EV-C) 347 envelope protein inhibitors, arboviruses 378 epidemic keratoconjunctivitis (EKC) 278. 279, 285, 286 epidemic Kaposi's sarcoma 194 "episodic" treatment 135 Epstein Barr virus (EBV) 279 adoptive T cell transfer and immune checkpoint blockade 180-181 associated extranodal NK/T cell lymphomas 179 associated lymphomas 179 B cell depleting therapy 179 disease overview 175 open issues 182-183 pharmacological inhibition 177-179 therapeutic interventions 178 vaccination 181-182 virus associated tumorigenesis 176 Epstein Barr virus nuclear antigens (EBNAs) 176 Epstein Barr virus small noncoding RNAs (EBERs) 176 ERK/MAPK pathway 201

erlotinib 383, 384 2-'-exo-methanocarbathymidine 196

f

famciclovir 138, 139 Famvir[™] 139 Fas-associated death domain (FADD) 212 Favipiravir (FP) 60, 73, 74, 119, 357, 413 fecal-orally transmitted Hepatitis E Virus 253 feline calicivirus (FCV) 316 feline panleukopenia virus 297 fetal disease 304 filoviruses 409, 410 antisense therapies 418 EBOV glycoprotein GP 411 host-directed therapies 418 immunotherapies 411-412 postexposure prophylaxis 412-413 recombinant vesicular stomatitis virus 411 small molecules 413-418 virus and disease overview 410-411 flaviviruses 367 Fludase 79 2-'-fluoro-5-iodo-aracytosine 196 fluoxetine 354, 356 follicular conjunctivitis (FC) 285 formalin-inactivated (FI) vaccines 112 foscarnet 139, 196 Foscavir[™] 139 FP ribofuranosyl-5'-triphosphate (FRTP) 73 fucosyltransferases (FUT) 316 fusion peptide (FP) of SARS CoVs 472

g

galidesivir 457–458 γ -glutamyltransferase 257 γ -herpesvirus 175 ganciclovir 152, 157, 158, 177–178, 196–201, 278, 282, 290 geldanamycin 207, 357 genital lesions 133, 140 genotypes (GT) of HCV 14 glecaprevir 30, 33, 34, 40

Global Polio Eradication Initiative (GPEI) 352 GLS4 234, 235, 240, 242, 243 glutathione (GSH) 357 glycoproteins of herpes simplex viruses 134 GPC-N114 259, 260, 261, 354, 357 GS-5734 413, 417, 421 Guanarito virus (GTOV) 426 4-guanidino-2,4-dideoxy-2,3-dehydro-Nacetylneuraminic acid (GG 167) 66, 67 guanidine hydrochloride (GuaHCl) 354, 356 guanosine triphosphate (GTP) 258 Guillain–Barré syndrome 257 gut microbiota 316, 328–330

h

H1PVAT 352 H274Y neuraminidase mutation 67 HBoV1-infection 300, 302 HCMV protein kinase pUL97 282 HCV. See hepatitis C virus (HCV) 13-58 Hecolin[®] 265 helicase of HEV (Hel) 254 helicase-primase inhibitors 136, 140, 143, 144 hemagglutinin (HA) 59, 63, 77 hematopoietic stem cell transplantation (HSCT) 112, 118, 279, 282 heme oxygenase 1 (HO-1) inhibitor (SnPP) 208 hemophagocytic lymphohistiocytosis (HLH) 175, 180 hemophagocytosis 191, 196 Hendra virus (HeV) 418-420 henipaviruses 409, 418 disease overview 419 immunotherapies 420 small molecules 419-420 hepatitis B virus (HBV) 13, 229-252 core protein 232–234 entry inhibitors 232 first-generation CAMs 234-235 hepatitis C virus (HCV) 13-14, 253, 257, 259, 291 drug discovery targets 15

enable drug discovery tools to 14-15 NS3/4A protease acyclic covalent binding inhibitors 25-30 NS3/4A protease acyclic reversible binding inhibitors 28-30 NS3/4A protease acyclic reversible inhibitors 30 NS3/4A protease covalent binding inhibitors 29-30 NS3/4A protease inhibitors 22-24 NS3/4A protease macrocyclic reversible inhibitors 30–32 NS3/4 protease P2-P4 macrocyclic inhibitors 30 NS5A inhibitors 32 RdRp inhibitor sofosbuvir 259 hepatitis E virus (HEV) clinical course of 256 Fecal-orally transmitted 253, 264 genetic diversity and molecular virology of HEV 254-256 HEV therapy 258 novel antivirals against HEV 259 prevention of infection and vaccination strategies for HEV 264 hepatocyte growth factor (HGF) 206 heptad repeat 1 (HR1) 473 heptad repeat 2 (HR2) 473 herpes simplex viruses (HSV) 133-150, 279 acyclic guanosine analogues 137-139 Alzheimer's disease 144 docosanol 138, 139-140 epidemiology and manifestation 142 foscarnet 139, 143 genital lesions 133, 140 helicase-primase inhibitors 138, 140-141 immunomodulators 142 management of resistant 143-144 monoclonal antibodies 136, 141 ocular HSV-1 infections 134 polymerase chain reaction 134 resistance mechanisms 143 replication cycle 134 5-substituted 2-deoxyuridine analogues 136

therapeutic vaccines 141 thymidine kinase 282 treatment of infections 135 HSV Type 1 133 HSV Type 2 133 typical orofacial cold sores 133 vidarabine 137 viral replication cycle 134 herpesvirus entry mediator (HVEM) 134 HEV hepatitis E virus 253-276 hexon 277 histo-blood group antigens (HBGA) 315-318, 328-329 histone deacetylase (HDAC) inhibitors 177 H1N1pdm09 viruses 66, 68, 69 H3N2 and B viruses 68 Hodgkin lymphoma 179 homozygous CCR5- Δ 32/ Δ 32-gene mutations 10 host cell nucleoside biosynthesis inhibitors 381 host-directed therapies 418 host factors 325 host kinase inhibitors, arboviruses 384-385 host-targeting candidates against influenza DAS181 79 LASAG 81 nitazoxanide 80 HPMPA((S)-9-[3-hydroxy-2-(phosphonomethoxy)propyl] adenine 196, 197 HSP70 inhibitor 204, 207, 209 HSP90 inhibitors 212 HSV529 141 HSV UL23 and VZV ORF36 thymidine kinases (TK) 199 human adenoviruses (HAdVs) 277-296 DNA-polymerase 280, 282, 283, 290 serotypes 277, 278, 279, 281, 285 strains 281, 291 human-associated circoviral DNA-sequences 297 human bocavirus (HBoV) 297, 298 human bufavirus (BuV) 297 human corona virus (HCoV) 449-486

diseases natural history 453-454

replication 450-454 human cytomegalovirus (HCMV) 151-174, 279 human γ_2 -herpesvirus 191 human immunodeficiency virus (HIV) 1-12, 291 open inssues in research 10 HIV-infection, natural history of 1 human intestinal enteroids (HIE) 316 human metapneumoviruses 79 human norovirus (HuNoV) 313-346 binding and entry 318-319 disease burden and pathogenesis 313-314 gut microbiota 316, 328-330 host factors 325 in vitro model 316-317 in vivo model 317-318 monoclonal antibodies 327, 330 nitazoxanide 330 non-structural proteins 315, 320 replication cycle 315-316, 325 vaccine development 331-332 viral genome 315 zinc 330-331 human polyomaviruses 279 human protein amphiregulin 79 α -hydroxy-bisulfite dipeptide 470 hydroxypyridonecarboxylic acids 153 α-hydroxytropolone 209 hypercytokinemia 61 hyperimmune globulin 430 hypoxia-inducible factor $1 (HIF1) \alpha$ protein 207, 208

i

iatrogenic, Kaposi's sarcoma (KS) 193 idoxuridine 136–137 IgA 108, 120 imidazoquinolinamine 142 iminosugars 385–386 imiquimod 142–143 immune suppressive cytokine TGF- β 181 immunocompromised patients 303, 304 immunodominant globular head (HA1) 76 immunomodulators 140, 142

immunoprophylaxis 110, 111, 115 immunotherapies 411, 420 inactivated poliovirus vaccine (IPV) 351 infectious mononucleosis (IM) 175, 176, 181 influenza A and B virions 59, 60, 65, 67.68.69 influenza-associated pneumonia 62 influenza A virus 59, 62-64, 66, 68 influenza B virus 59, 62, 63, 66, 75 influenza hemagglutinin glycoprotein (HA) 59, 77, 78 influenza virus 59-96 adamantanes and derivatives 63-64 hemagglutinin 59 host-targeting candidates 79-82 M2 ion channel blockers 62 matrix 2 protein 59 monoclonal antibodies 60, 63, 76–78 neuraminidase 59, 60, 62 neuraminidase inhibitors 60, 64-69 pandemic situation virus-specific vaccine 60 pathogenesis of influenza 60-62 polymerase inhibitors 60, 70-76 ribonucleoprotein 59 seasonal influenza epidemics 59 vaccination coverage 60 inhaled zanamivir 67 inosine-5'-monophosphate dehydrogenase (IMPDH) 258, 264, 381 inosine monophosphate (IMP) 258 integrin α 3 256 interferon (IFN) 181, 325 interferon regulatory factor homologue vIRF3 (ORF K10.5) 207 interleukin 1β-converting enzyme (FLICE)inhibitory proteins (FLIP) 212 Intravenous Immune Globulin (IVIG) 117 in vitro CVB3 replication inhibitors 356 in vitro model 316-317 in vivo model 317-318 murine norovirus 318 5-iodo-2-deoxyuridine 136 ivermectin 376, 380, 388 iVIEW-1201 278, 284, 288

j

JNJ-63623872 75 Junín virus (JUNV) 426

k

Kaposi's sarcoma 192–194 Kaposi's sarcoma-associated herpesvirus (KSHV) 191-227 bone marrow failure 196 capsid assembly 209 DNA polymerase 196-201, 209 epidemiological considerations 191 hemophagocytosis 191, 196 interfering with cellular cytokines 206 kinase inhibitors 201-206 life cycle 206-208, 210 multicentric Castleman disease 191, 194–195 polyclonal post-transplant lymphoproliferative disease 195 preclinical data on other cellular targets 207-208 primary effusion lymphoma (PEL) 194 targeted immunotoxins 210 target productive (lytic) KSHV DNA replication and/or packaging 209-210 13-kDa protein 255 kinase inhibitors 201-206 KPT-8602 205, 208 KSHV-associated inflammatory cytokine syndrome (KICS) 191 KSHV-induced hemophagocytosis 191

l

laninamivir 62–65, 69 laninamivir octanoate 69 LASAG 79, 81–82 Lassa fever disease (LFD) 423 Lassa virus (LASV) 423 latency-associated nuclear antigen (LANA) 211 latent membrane proteins (LMPs) 176 latent viral protein LANA 195, 210 ledipasvir 36, 38, 39 letermovir 154-170 clinical efficacy 156–158 clinical experience 159–164 clinical samples 165, 167 potential indications for 167–169 preclinical models 156-158 resistance-associated variants 164-165 resistance development 164–167 resistance mutations 164–165 structure-activity relationship 156 lipid processing, inhibitors of, enteroviruses (EV) 357 long-acting cART drugs 10 Lonsurf[™] 137 lower respiratory tract infection (LRTI) 97, 110.112 Lujo virus (LUJV) 423 lumicitabine (ALS-008176 or JNJ-64041575) 106, 118 lycorine 375, 379 Lymphocytic Choriomeningitis Virus (LCMV) 420 d,l-lysine acetylsalicylate glycine 81 lysosomotropic agents 387

т

Machupo virus (MACV) 426 marburgvirus genera 410 Maribavir 152-154, 157, 158 MARV nucleoprotein 418 matrix 2 (M2) protein 59 matrix 2 (M2) ion channel blockers 62 Mavyret 30, 34, 35, 39-41 measles vaccine development 113 MEDI8852 76, 77, 78 MEDI-8897 118 MERS-CoV 450, 451 5'-7-methylguanylate 254 methyltransferase (MeT) 254-255 methyltransferase inhibitors, arthropodborne viruses (arboviruses) 379 MHAA4549A 76, 77, 78 mitochondrial antiviral-signaling protein (MAVS) 322 modified vaccinia virus Ankara (MVA) 182

molecular biology of parvo viruses 298-299 Molnupiravir 456, 457 monoclonal antibodies (mAbs) 140, 141, 328, 411 MEDI8852 77-78 MHAA4549A 78 VIS410 78 5-monophosphate 137 Motavizumab (MEDI-524) 117-118 mouse hepatitis virus (MHV) 456 multicentric Castleman's disease (MCD) 191, 194, 195 multimeric gp350 vaccine 182 multiple sclerosis (MS) 175 mycophenolic acid (MPA) 263, 264

n

N-(4-hydroxyphenyl)-retinamide 380 NAI-resistant influenza viruses 60, 70 nasopharyngeal carcinoma (NPC) 178 natural killer (NK) cells 175 nectin-1 134 nectin-2 134 Neu5Ac a(2,3)-and Neu5Ac a(2,6)-Gal linkages 59, 79 neuralgic amyotrophy 257 neuraminidase (NA) 59, 60, 62, 64 neuraminidase inhibitors (NAIs) 60, 64 laninamivir octanoate 64, 65, 69 oseltamivir 65-69 peramivir 64–69 zanamivir 66–67 neutralizing antibody (nAb) 411, 412, 427 New World Arenaviruses 420, 426, 427 N-finger 463 NFkB pathway 212 Niemann-Pick C1 412 Nipah virus (NiV) 418 nitazoxanide (NTZ) 79, 80-82, 119, 314, 327, 330 NITD008 259, 260, 261 NK cells 181 NM107 259 non-A non-B hepatitis (NANBH) 13 non-Hodgkin lymphomas 194, 195, 200

non-nucleoside analogues 324, 325 non-nucleoside HCV NS5B polymerase palm domain inhibitors 19. 23 - 24.26non-nucleoside inhibitors (NNIs) 463 arboviruses 376 non-nucleoside NS5B Inhibitors 19–21, 39 non-nucleoside NS5B polymerase palm domain inhibitors 26 non-nucleoside reverse transcriptase inhibitors (NNRTIs) 290 nonstructural proteins 1 299 3' and 5' nontranslated region (NTR) 14 norovirus 73, 313–346 Norwalk virus 313 nosocomial transmission 423 novel antivirals against HEV 259 novel CAMs 236 NS1/2 315, 316, 320 NS3/4A protease acyclic covalent binding inhibitors 25-30 NS3/4A protease acyclic reversible binding inhibitors 25, 28 NS3/4A protease acyclic reversible inhibitors 28 NS3/4A protease covalent binding inhibitors 29-30 NS3/4A protease inhibitors 22-24, 30 NS3/4A protease macrocyclic reversible binding inhibitors 31NS3/4A protease macrocyclic reversible inhibitors 30-32 NS3/4 protease P2-P4 macrocyclic inhibitors 30 NS3 (NTPase/helicase) 320-321 NS4B inhibitors, arboviruses 379 NS4 (p22) 315, 316, 321 NS5A inhibitors 30, 32 NS5B polymerase inhibitors non-nucleoside 19–22 nucleoside and nucleotide 15-19 NS5 (VPg) 321-322 NS6 (protease) 322 NSC373989 209

N-terminal domain (NTD) 473 N-terminal tail 463 NTPase/helicase domain 23, 315 nucleoside analogues 287, 290, 322-325 arboviruses 376 nucleoside HCV NS5B inhibitors 17.20 nucleoside inhibitors, of NS5B polymerase 15, 19 nucleos(t)ide analogs (NAs) mode of action 231 reverse transcriptase inhibitors 229, 231 nucleotide inhibitors, of NS5B polymerase 15, 18, 20 nucleotide kinase 290 nucleotidyltransferase activity (NiRAN) 460

0

Ocrelizumab 179 ocular HSV-1 infections 134 OKG-0301 288 Old World Arenaviruses 420-423 ombitasvir 36, 38, 40 open reading frames (ORF1-3) of noroviruses 315 oral polio vaccine (OPV) 351-352 ORF1 polyprotein of HEV 254 ORF2 capsid proteins of HEV 255 ORF9/ORF59 complex of KSHV 209 ORF36-encoded viral protein kinase vPK 199-201, 205 Orthohepevirus A 254 Orthohepevirus B 254 Orthohepevirus C 254 Orthohepevirus D 254 oseltamivir 62-67, 69 oxyanion hole 486 oxysterol binding protein (OSBP) 357

р

p22 (NS4) 315 p48/N-terminal protein (or NS1/2) 315 Palivizumab 97, 99, 110, 111, 116–118 palliative treatment 288

pandemic A(H1N1)pdm09 virus 64, 69 pandemic H1N1 influenza virus 64, 73 pandemic situation virus-specific vaccine 60 pandemic strain A H1N1pdm09 66 papain-like cysteine protease (PCP) 254, 255 papain-like protein (PL^{pro}) 452 papilloma-virus 300 parainfluenza viruses 79, 117 paratepravir 30 paritaprevir 31, 32, 40 Parsonage-Turner syndrome 257 particle maturation of adenoviruses 280, 289, 290 Parvoviridae 297-311 parvovirus 4 (PARV4) 297 parvovirus B19 297 parvoviruses 297, 298, 300 particle 298 replication 300 pathogenesis of influenza 60 PD-1 blocking antibodies 181 pegIFN-*α* 258, 260 pegylated-IFN (PEG-IFN) 14 penciclovir 136, 138, 139, 140 peptidomimetic α -ketoamides 467 peramivir 62-65, 67 peripheral blood mononuclear cells (PBMCs) 77 pharyngeal conjunctival fever (PCF) 285 phenylpropenamide (PPA) 234 phenylsulfonamide aminopyridines 156 phosphatidylinositol 4-phosphate (PI4P) lipids-enriched membrane microenvironment 357 phosphorodiamidate morpholino oligomers (PMOs) 418 PI3K/AKT/mTOR pathway 201, 206 PI4KIIIβ inhibitors 357 pibrentasvir 30, 35, 37, 39, 40 pimodivir 75-76 PLA2-enzymatic activity 298, 299, 303 pleconaril phospholipase 352

pneumomediastinum 302 pneumothorax 302 pocapavir 352 poliovirus 73 entry inhibitors 347, 348, 351, 352 vaccines 351 polyclonal post-transplant lymphoproliferative disease 195 polymerase acidic protein (PA) 59 polymerase basic protein 1 (PB1) 59 polymerase basic protein 2 (PB2) 59 polymerase inhibitors against influenzaBaloxavir marboxil 60, 6270–72 favipiravir 62, 73–74 pimodivir 75-76 polyoma-virus 300 polyproteins (pp) of Corona viruses 451, 452 ponatinib 205 porcine parvovirus 297 pORF17 polyprotein of KSHV 209 pORF17 scaffold/protease polyprotein of KSHV 209 pORF45/RSK complex 205 postexposure prophylaxis (PEP) 411–413 post-transplant lymphoproliferative disease (PTLD) 177, 179, 180 potent Hepatitis E Virus-specific T-cell responses 257 potential indications for treating cytomegalovirus infections 167-168 povidone iodine (PVP-I) 287 Presatovir (GS-5806) 118-119 primary effusion lymphoma (PEL) 191, 194 primate bocaparvovirus 297 primate erythroparvovirus B19V 1, 297 pritelivir 138, 140, 144 pro-inflammatory cytokine 72 prolonged B19V-symptoms 304 protease/helicase inhibitors, arthropod-borne viruses (arboviruses) 377-378 protective EBV specific T cell responses 182 protein metabolism inhibitors, arboviruses 385-386

protein-protein interactions 322 provirion 350, 351 PU-H71 203, 207 pulmonary aspergillosis 62 pyrimidine nitrile 291 pyrrolone derivative 378

r

rapid virological response (RVR) 35 rat hepatitis E virus 254 Ravn virus (RAVV) 410 recombinant gp350 vaccine 182 recombinant vesicular stomatitis virus (rVSV) 411 REGN-EB3 411, 412, 418 relaxed circular DNA genome (rcDNA) 232 Remdesivir 117, 119, 120, 413, 417, 420, 455, 456, 462, 463 replication and transcription compartments (RTCs) 207 replication cycle of noro viruses 315-316 resiguimod 138, 142 resistance-associated variants (RAVs) 165 resistant-associated substitutions (RASs) 40 Respiratory Syncytial Virus (RSV) infection 76 attenuated replication-competent vaccines 114-115 challenge of vaccine development 99, 112-114 clinical aspects of adult RSV disease 111-112 conformational breathing affects neutralization sensitivity 113 development of treatments 98–99 epidemiology 111-112 evasion or suppression of immune memory 108-109 on human health burden 97–98 immune evasion or immune suppression 114 immunoprophylaxis impact 110 lack of immunogenicity 114 life cycle 103

longevity 108 measles versus vaccine development 113 M2-1 inhibitor 119 secondary endpoints 114 surface glycoproteins 108 symptoms 110 therapeutics and vaccine research 97 transmission 98 virus and replication cycle 99-104 wheeze 110-111 Reston virus (RESTV) 410 reverse transcriptase (RT) inhibitors 229, 231 rhadinovirus 191 rhinoviruses 347, 355-358 RI-001 105, 117-118 RI-002 105.117 ribavirin 117, 259, 260, 262, 265, 278, 282, 323, 324, 419, 420, 422 ribavirin monophosphate (RMP) 258 ribavirin triphosphate (RTP) 258 ribonucleoprotein (RNP) 59 Rilematovir 119 rimantadine 62-64 Rituximab 179 RNA dependent RNA polymerase (RdRp) 15, 254, 256, 258, 259, 314-316, 320-322, 459-463, 477 arboviruses 367-408 inhibitors, enteroviruses (EV) 352-359 RNA interference (RNAi) strategy 358 RNAse H-like nucleotidyltransferases 209 RSV-F fusion glycoprotein 102–104 RSV-G glycoprotein 102, 108 RSV-IGIV 115-116 RSV-SH glycoprotein 104, 107 rupintrivir 321-322, 355, 356, 465, 466 (+)-rutamarin 204, 207 rVSV-EBOV vaccination 412 RWJ-27020 67

S

S31N substitution 64 S-033188 71 Sabiá virus (SABV) 424, 426 Sapanisertib 203, 206 SARS-CoV-1 450-457 X-ray crystal structures of 473 SARS-CoV-2 450-486 X-ray crystal structures of 473 SARS-CoVs RNA-dependent RNA polymerase 460, 463 saturation transfer difference nuclear magnetic resonance (STD NMR) 318 scaffold (assembly) protein of KSHV 209 seasonal influenza A/H1N1pdm09 59, 60.68.78 seasonal influenza epidemics 59 secondary bacterial coinfection 62 serine protease domain 23 serine protease of KSHV 209SHP640 287 sialic acids (SAs) 277, 289 Sigmodon alstoni 426 siltuximab 202, 206 silvestrol 260, 264 simeprevir 30, 31, 39 single-stranded (ss) DNA-viruses antiviral chemotherapy of parvovirus B19-infection 303 classification 297-298 molecular biology 298-300 parvovirus replication 300 therapeutic options and recommendations 303-304 SiSunatovir 106, 119 sliding poles 461 small interfering RNAs (siRNAs) 418 small molecule 419-420, 421, 423, 424, 430-431 NFkB inhibitors 207, 212 sofosbuvir 17-20, 30, 38-41, 259, 260, 374 solid organ transplantation (SOT) 151, 166, 167, 279 specific cART-drug 2 S protein of Corona virus 463, 472, 473-475, 477 squaric acid dibutyl ester (SADBE) 138, 142

SOX770 138, 142 1-[(2S,4S-2-(hydroxymethyl)-1,3-dioxolan-4yl]5-vinylpyrimidine-2,4(1H,3H)dione 196, 198 sterol regulatory element-binding proteins (SREBP) 382-384 Strategic Timing of Antiretroviral Treatment (START) 10 structural proteins of Parvo viruses 298, 299 structure-based drug design (SBDD) 467 subepithelial corneal infiltrates (SEIs) 286 5-substituted 2-deoxyuridine analogues 136 Sudan virus (SUDV) 410 3-O-sulfonated heparan sulfate 134 sunitinib 383, 384, 389 surrogate systems 14 sustained virological response (SVR) 19, 28 SWISS STATEMENT hypothesis 10 symptomatic HAdV disease 281

t

T-705 73, 74, 319, 324, 448 Taï Forest virus (TAFV) 410 targeted immunotoxins 210 target productive (lytic) KSHV DNA replication and/or packaging 209 T cell 168, 175, 176, 180 T cell mediated immune control 176 T cell receptor (TCR) 178, 181 TCN032 76 Technivie 32.40 tegument-coated capsid 134 telaprevir 25, 29, 30, 35, 354, 356 telbivudine for parvovirus infections 303 terminase inhibitors of CMV 151-156 tetrapeptide nitrile 291 therapeutic vaccines 140, 141 thymidine kinase (TK) 134 TK/ORF21 200 TLR-7 antagonists 142 tocilizumab 202, 206 Togaviridae 367, 370 Toll-Like Receptor (TLR) 112

tomeglovir against cytomegalovirus terminase 156 torque-teno-midi-virus (TTMDV) 297 torque-teno-virus (TTV) 297 transient aplastic crisis (TAC) 302-303 transmembrane region (TM) of coronavirus S protein 473 transmission electron microscopy 100 transplant-associated KS 193 trifluorothymidine 136 trifluridine 136-138 5-triphosphate (TP) 137 trisodium phosphonoformate 139 tryptophan dendrimers 355 tumor necrosis factor (TNF) 178 tumor necrosis factor receptor-related herpesvirus entry receptor 134 tumor necrosis factor (TNF) receptorassociated factors (TRAFs) 179 tusavirus (TusaV) 297 typical orofacial cold sores 133 Tyro3/Axl/Mer TK (TAM) family 206

и

UL23 gene coding 143 UL56 of cytomegalovirus 164–166 UL97 of cytomegalovirus 152 United Nations Program on HIV/AIDS (UNAIDS) 10 upper respiratory tract (URT) infection 116 uracil-DNA glycosylase 134

V

vaccination against EBV 181 vaccine-derived polioviruses (VDPV) 352 vaccines, poliovirus 351 valacyclovir 137 Valcyte[®] 282 valganciclovir (VGCV) 152, 158, 167–169, 199, 200, 278, 282 Valtrex[™] 137 varicella zoster virus (VZV) 133, 279 vascular endothelial growth factor (VEGF) 201, 202, 206 velpatasvir 37, 39, 40 VER-155008 207, 209 vFLIP/IKK γ-dependent NFkB activation 212 vFLIP-interacting protein NEMO/ ΙΚΚγ 207 VH244 107, 119 vidarabine 137, 138 Viekira Pak 32, 35, 38, 40 viral DNA derived palindromic oligonucleotide 177 viral DNA synthesis 134, 154 viral genome 315, 328 Viral genome (H)CMV 151, 154, 164 viral hemorrhagic fever (VHF) 410, 433 viral replication cycle 134 viral replication cycle, kinetics of 154 viral ribonucleoprotein (vRNP) segments 59 viral RNA synthesis 59 viral tegument protein encoded by ORF45 205 Viroptic[™] 136 virulence factor 1 (VF1) 315 virus associated tumorigenesis 176 virus attachment and entry of adeno viruses 289 virus-encoded protein kinases 282 virus genome replication 290 virus-like particles (VLPs) 114, 178, 182 VIS410 78 VP1-unique region 298, 300, 301, 303 VPg (NS5) 315 vPK/ORF36 200, 201 VX-787 75

W

West Nile virus (WNV) 73, 367, 369, 372 wheeze 110-111

X

xanthosine monophosphate (XMP) 258 Ximency 21, 24, 29, 40 X-ray crystallography 318

у

yellow fever virus (YFV) infections 367, 368–369, 372

Ζ

zanamivir 63–69 zanamivir-resistant A(H1N1)pdm09 and H3N2 isolates 67 Zika virus (ZIKV) infection 367, 368, 372 zinc 330–331 Ziresovir (AK0529) 106, 119 ZMapp 411, 412 Zostex[™] 137 Zovirax[™] 137, 139

WILEY END USER LICENSE AGREEMENT

Go to www.wiley.com/go/eula to access Wiley's ebook EULA.